A Survey of Stellar Families: Multiplicity of Solar-Type Stars

A Dissertation

Presented in Partial Fulfillment of Requirements for the Degree of

Doctor of Philosophy

in the College of Arts and Sciences

Georgia State University

2009

by

DEEPAK RAGHAVAN

Committee:

Dr. Harold A. McAlister, Chair

Dr. Douglas R. Gies, Member

Dr. Todd J. Henry, Member

Dr. David W. Latham, Member

Dr. Brian D. Mason, Member

Dr. A. G. Unil Perera, Member

Dr. Russel J. White, Member

Date

Dr. H. Richard Miller, Chair Department of Physics & Astronomy

A Survey of Stellar Families: Multiplicity of Solar-Type Stars

by

Deepak Raghavan

Under the Direction of Harold A. McAlister

Abstract

I present the results of a comprehensive assessment of companions to solar-type stars. The sample of 454 stars, including the Sun, was selected from the *Hipparcos* catalog with $\pi > 40$ mas, $\sigma_{\pi}/\pi < 0.05$, $0.5 \leq B - V \leq 1.0$, and positioned on an HR diagram within a band extending 1.5 magnitudes below and 2 magnitudes above an iterative best-fit main sequence. The resulting sample is an exhaustive set of stars with V-band flux between 0.1 and 10 times that of the Sun, providing a physical basis for the term "solar-type". New observational aspects of this work include a survey for separated fringe packet companions with long-baseline interferometry at the Center for High Angular Resolution Astronomy (CHARA) Array, blinking multi-epoch archival images to identify wide common proper motion companions, and speckle interferometry. Apart from evaluating companion information from the various catalogs covering many different techniques, I also obtained and included unpublished results

from extensive radial velocity monitoring programs. The many sources leveraged enable a thorough search and evaluation for stellar and brown dwarf companions.

The results presented here include eight new companion discoveries, four of which are wide common proper motion pairs discovered by blinking archival images, and four more are from the spectroscopic data. The overall observed fractions of single, double, triple, and higher order systems are $57\% \pm 3\%$, $33\% \pm 2\%$, $8\% \pm 1\%$, and $3\% \pm 1\%$, respectively. If all candidate companions identified are found to be real, the percentages would be $54\% \pm 2\%$, $34\% \pm 2\%$, $9\% \pm 2\%$, and $3\% \pm 1\%$, respectively. The incompleteness analysis indicates that only a few undiscovered companions remain for this well-studied sample. Including these missed companions and estimating the percentage of candidates that are real leads to multiplicity fractions of $55\% \pm 3\%$, $34\% \pm 2\%$, $9\% \pm 2\%$, and $2\% \pm 1\%$ for single, binary, triple, and higher order systems. These results include all stellar and brown dwarf companions, and show that a majority of the solar-type stars are single, in contrast to earlier expectations. This large sample enables a check of the dependence of multiplicity on various physical parameters. Bluer, more massive stars are more likely to have companions than redder, less massive ones, consistent with the trend seen over the entire spectral range, but showing a steeper drop at solar-type stars than previously believed. Consistent with prior studies, younger, more active stars are seen as more likely to have companions than older, less active ones, suggesting that companions may be stripped over time by dynamical interactions. A preliminary, but important indication seen here is that brown dwarfs, like planets, seem to prefer stars with higher metallicity, tentatively suggesting that brown dwarfs may form like planets, at least when they are companions to stars.

The period distribution derived is unimodal and roughly Gaussian with a peak and median value of 300 years. The period-eccentricity relation shows the expected circularization for periods below 12 days, followed by a roughly flat distribution. The mass-ratio distribution shows a clear discontinuity near a value of one, indicating a preference for twins, which are not confined to short orbital periods. The period-mass-ratio relationship shows that multiple formation mechanisms are likely responsible for the formation of stellar systems – fission for like pairs and fragmentation or random associations for longer-period systems of all mass ratios. This distribution also shows a paucity of low-mass companions, confirming that the brown dwarf desert extends out to wide separations. The ratio of planet hosts among single, binary, and multiple systems are statistically indistinguishable, suggesting that planets are as likely to form around single stars as they are around components of binary or multiple systems. Hence, the common occurrence of binary systems is likely to increase the real-estate available for planets, and perhaps life.

INDEX WORDS: Stellar multiplicity, Binary stars, Solar-type stars, Solar neighborhood, Exoplanet systems, Brown dwarfs, Survey, Long baseline interferometry, Radial velocity

A Survey of Stellar Families: Multiplicity of Solar-Type Stars

by

DEEPAK RAGHAVAN

A Dissertation Presented in Partial Fulfillment of Requirements for the Degree of

Doctor of Philosophy

in the College of Arts and Sciences

Georgia State University

2009

Copyright by Deepak Raghavan 2009

A Survey of Stellar Families: Multiplicity of Solar-Type Stars

 $\mathbf{b}\mathbf{y}$

Deepak Raghavan

Major Professor:

Committee:

Harold A. McAlister

Douglas R. Gies Todd J. Henry David W. Latham Brian D. Mason A. G. Unil Perera Russel J. White

Electronic Version Approved:

Office of Graduate Studies College of Arts & Sciences Georgia State University May 2009 To my parents

Thank you for inculcating the spirit of learning in me

Acknowledgments

When I decided to pursue a doctorate in astronomy more than six years ago, it was not with complete certainty, but with a sense of curiosity, wonder, and yes, some apprehension. I had decided to chart a unique course for myself, and to travel down a path with few footsteps, especially given where I was coming from. I would not have been able to complete this journey without the steadfast support of my family and friends, and I would like to extend my most sincere gratitude to them.

I could not have done this without the support of my dear wife, Priya, who has not only always stood by me, but also been such a tremendous source of support and wisdom, letting me probe and explore, while always being there when I needed a listening ear or a supporting hand. My children, Anjan and Aditi, while sometimes curious, and sometimes worried as to what daddy was doing back in school, never wavered in their implicit support.

I thank my parents and siblings for inculcating the spirit of learning in me, which has become a significant driver of my life. My mother led by example, completing her own Ph.D. in Hinduism in 2007. I am very proud of her accomplishment and inspired by it. I appreciate my friends for their indulgence when I would excitedly talk about astronomy and for allowing me to miss parties or leave them early to go to AROC for observing.

My thanks also go out to my committee members and the astronomy faculty members for the invaluable support and guidance they have provided me throughout the process. Special thanks to Todd Henry for directing my master's work on exoplanet systems, for the idea of this thesis work, and for his consistently insightful feedback. I have truly enjoyed working with my advisor, Hal McAlister, and would like to recognize his consistent encouragement, support, and direction, not to mention his pioneering efforts in making the CHARA Array a reality and providing us graduate students with copious observing time on such a worldclass instrument. Special thanks to Dave Latham for allowing my work to benefit from his meticulous radial velocity monitoring over 30 years and for being so supportive, gathering new data as required and reducing the data on hundreds of stars to enhance the results of this effort. Thanks to Brian Mason for always knowing the answer to any question about double stars and for taking me along on the speckle observing runs; and to Doug Gies, whose consistent advice and ideas enabled me to give a better shape to this effort.

This work would not have been possible without the help extended by astronomers around the world, who responded to my persistent requests with support and encouragement. Geoff Marcy graciously allowed me access to his treasure trove of the highest precision radial velocities, enabling me to derive more robust statistics. Artie Hatzes, Bill Cochran, Hugh Jones, and Jason Wright also helped me improve the radial velocity coverage. Bill Hartkopf, Andrei Tokovinin, Dimitri Pourbaix, and Richard Gray were very helpful with their insights and analysis on specific stellar systems.

My fellow graduate students enriched this journey with their humor, wit, and open acceptance of a who some might have considered "an outsider", for which I am indebted to them. Special thanks to Rajesh Deo and Justin Cantrell for taking care of our computers while working on their own research, and to Alvin Das for creating a template that made it easier to navigate the LaTeX dissertation jungle. Thanks also to the staff at the CHARA Array for keeping such a sophisticated instrument working so well that it seemed like I was witnessing science fiction come alive.

I am grateful to have grown up in India, whose society gave me a strong foundation, built with a desire to learn and accomplish. I am also proud to call the the Unites States of America home, and am thankful to my fellow citizens for accepting and embracing me beyond my wildest expectations. There is no other country in the world where I could have traveled the paths I did, charting a uniquely individual course for my life, without once having to justify or explain the twists and turns I decided to take. May India and the USA always be beacons of freedom, innovation, and accomplishment.

This work could not have been possible without various grants and resources, which I acknowledge here. Research at the CHARA Array is supported by the College of Arts and Sciences at Georgia State University and by the National Science Foundation through NSF grant AST-0606958. This research has made use of the SIMBAD literature database, operated at CDS, Strasbourg, France; NASA's Astrophysics Data System; multi-epoch images from the Digitized Sky Survey, which was produced at the Space Telescope Science Institute under U.S. Government grant NAG W-2166; and data products from the Two Micron All Sky Survey (2MASS), which is a joint project of the University of Massachusetts and the Infrared Processing and Analysis Center/California Institute of Technology, funded by the National Aeronautics and Space Administration and the National Science Foundation.

TABLE OF CONTENTS

A	CKNO	VLEDGMENTS	•	\mathbf{v}
Lı	ST OI	TABLES	•	xii
Lı	ST OI	FIGURES	•	xiv
Aı	BBRE	TATIONS AND ACRONYMS	•	xliii
1	Inti	ODUCTION AND BACKGROUND	•	1
	1.1	Motivation		1
	1.2	History of Double Star Observations		5
	1.3	Previous Multiplicity Surveys		9
		1.3.1 Early Multiplicity Surveys		10
		1.3.2 The Duquennoy & Mayor Survey	•	15
		1.3.3 Multiplicity Results Since DM91		17
	1.4	Multiplicity Among Stars with Planets		24
2	SAM	PLE AND SURVEY METHODS	•	26
	2.1	The Sample of Solar-Type Stars		26
		2.1.1 Sample Selection from the Hipparcos Catalog		27
		2.1.2 Comparison with DM91 Sample		54
		2.1.3 A New Reduction of the <i>Hipparcos</i> Data		57
		2.1.4 Sample Analysis & Bias		62
	2.2	Survey Methods		65
3	Lon	g-Baseline Interferometric Results		68
	3.1	Survey of Separated Fringe Packet Binaries		73

		3.1.1	SFP Data Reduction and Results	76
		3.1.2	Astrometry from Separated Fringe Packets	100
	3.2	Visual	Orbits Derived using the CHARA Array	103
		3.2.1	The Visual Orbit of σ^2 CrB (HD 146361)	104
		3.2.2	The Visual Orbit of HD 8997 \ldots	105
		3.2.3	The Visual Orbit of HD 45088	108
		3.2.4	The Visual Orbit of HD 223778	111
4	Con	imon F	PROPER MOTION RESULTS	116
	4.1	Identif	ication and Confirmation of CPM Companions	116
	4.2	Linear	Motion of Field Stars	149
5	Отн	ier As	rrometric Results	160
	5.1	The H	<i>ipparcos</i> Double Stars	160
		5.1.1	Component Solutions	162
		5.1.2	Accelerating Proper Motions	165
		5.1.3	Orbital Solutions	171
		5.1.4	Stochastic Solutions	172
	5.2	Visual	Orbits	172
	5.3	The W	Vashington Double Star Catalog	182
	5.4	The Fe	ourth Interferometric Catalog	187
	5.5	The C	atalog of Nearby Stars	188
6	Spe	CTROSC	COPIC AND PHOTOMETRIC RESULTS	190
	6.1	Consta	ant Radial Velocity Stars	191
	6.2	Stellar	Companions	208
		6.2.1	Single-lined Spectroscopic Binaries	208
		6.2.2	Double-lined Spectroscopic Binaries	213
		6.2.3	Radial Velocity Variations Without Orbits	221
	6.3	Planet	ary Companions	223

	6.4	Eclips	ing Binaries	224
7	Syn	THESIS	OF RESULTS	227
	7.1	Nome	nclature	227
	7.2	Result	ts for Each of the 454 Systems	229
	7.3	Obser	ved Stellar Multiplicity	243
	7.4	Notes	on Individual Systems	248
8	Dise	CUSSIO	n and Analysis	285
	8.1	Comp	arison with DM91 Multiplicity Statistics	285
	8.2	Incom	pleteness Analysis & True Stellar Multiplicity	292
		8.2.1	Missing Spectroscopic Companions	294
		8.2.2	Missing Visual Companions	298
		8.2.3	Very Low-Mass Companions	300
		8.2.4	Results from DM91 Incompleteness Analysis	303
		8.2.5	Missed Companions and True Multiplicity	304
	8.3	Multip	plicity Dependency on Physical Parameters	306
		8.3.1	Multiplicity by Spectral Type and Color	307
		8.3.2	Multiplicity by Chromospheric Activity	311
		8.3.3	Multiplicity by Metallicity	312
	8.4	The D	Distribution of Orbital Elements	316
		8.4.1	Period Distribution	316
		8.4.2	Period-Eccentricity Relationship	320
		8.4.3	Mass-Ratio Distribution	322
	8.5	Multip	plicity in Exoplanetary Systems	327
	8.6	Hierar	chy of Multiple Systems	329
	8.7	Conclu	usions	330
Rı	EFERI	ENCES		343

A	opene	dices		363
\mathbf{A}	Рну	SICAL]	Parameters for Primaries and Companions	364
в	SFP	INVES	TIGATION PLOTS FOR INDIVIDUAL SYSTEMS	392
С	IDL	Prog	RAMS	311
	C.1	CHAR C.1.1 C.1.2	A Observations Planning Tool for Survey Projects 6 Example Input File 6 IDL code 6	512 512 513
	C.2 C.3	Examp SFP D	ple Output File	535 536
	C.4	Blinkiı	ng Archival Images	350
	C.5	Derivit C.5.1 C.5.2 C.5.3 C.5.4 C.5.5	ng a Visual Orbit From Interferometric Visibilities	355 355 361 362 363 366
D	Two) Suns	IN THE SKY: STELLAR MULTIPLICITY IN EXOPLANET SYSTEMS $\dots 6$	369
\mathbf{E}	Тне	Visua	l Orbit σ^2 Coronae Borealis $\ldots \ldots \ldots$	390

LIST OF TABLES

1.1	Results of Previous Multiplicity Surveys	23
2.1	Volume-Limited Sample of 454 Solar-Type Stars	32
2.2	Stars Excluded Due to Large Parallax Errors	51
2.3	Stars Excluded Due to Large Offset from the Main Sequence	53
2.4	Stars excluded from sample, but qualifying in FvL07 data	60
2.5	Sample stars outside selection criteria based on FvL07 data	61
3.1	Observing Nights for the SFP Survey	77
3.2	SFP Observations and Results	83
3.3	Summary of SFP Results	94
3.4	Projected Separations for SFP Measurements	102
3.5	Interferometric Visibilities for HD 8997	105
3.6	Visual Orbit Solution for HD 8997	107
3.7	Interferometric Visibilities for HD 45088	109
3.8	Visual Orbit Solution for HD 45088	111
3.9	Interferometric Visibilities for HD 223778	112
3.10	Visual Orbit Solution for HD 223778	113
4.1	Archival Image Blink Results	125
4.2	Photometry of CPM Candidates	148
4.3	Optical WDS Entries	151

5.1	Hipparcos Component Solutions	163
5.2	Accelerating Proper Motion Solutions	169
5.3	Hipparcos Orbital Solutions	171
5.4	Visual Orbit Solutions	178
5.5	WDS companions confirmed by photometry	184
5.6	WDS pairs demonstrating orbital motion	186
6.1	Stars with no Evidence of Radial Velocity Variations	193
6.2	SB1 Orbits	211
6.3	SB2 Orbits	217
6.4	Stars with Possible RV Variations	222
6.5	Planetary Companions	225
7.1	Stellar and Planetary Companions	233
7.2	Observed Multiplicity Statistics	243
7.3	Classification of Confirmed Companions	246
7.4	Classification Including Candidate Companions	247
8.1	Comparison of Multiplicity Statistics with DM91	290
8.2	Multiplicity Statistics by Spectral Type and Color	308
A.1	Physical Parameters of the Sample Stars	365
A.2	Spectral Type and Mass of the Companions	383

LIST OF FIGURES

2.1	Hipparcos main sequence	27
2.2	Hipparcos distance-limited sample	28
2.3	Hipparcos distance-limited sample close to the main sequence	29
2.4	Volume-limited sample of solar-type stars from <i>Hipparcos</i>	31
2.5	DM91 sample selected from <i>Gliese</i> and <i>Hipparcos</i> catalogs	55
2.6	The current sample plotted along the same axes as Figure 2.5 \ldots .	58
2.7	The current sample from <i>Hipparcos</i> and FvL07	59
2.8	Color-Magnitude diagram of <i>Hipparcos</i> and FvL07 samples	60
2.9	Apparent and absolute magnitude distribution of the sample	63
2.10	Distance distribution of the sample	64
2.11	$B-V$ color distribution of the sample $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	65
2.12	Spectral type distribution of the sample	66
2.13	Correlation between spectral type and color	66
3.1	Intensity of diffraction patterns	70
3.2	Schematic of a basic interferometer	71
3.3	Beam combining at CHARA Classic	73
3.4	Examples of separated fringe packets	75
3.5	Example fringe envelopes for a separated fringe packet binary	80
3.6	Example fringe envelopes for a single fringe packet star	81
3.7	Triangulation of SFP measures	102
3.8	Calibrated visibility measurements for HD 8997	106

3.9	Calibrated visibility measurements for HD 45088	110
3.10	Calibrated visibility measurements for HD 223778	115
4.1	Example of DSS images	117
4.2	Epoch distribution of the images used for blinking	118
4.3	Distribution of the time interval between images blinked	119
4.4	Distribution of the motion seen on blinking	121
4.5	Distribution of the search space for CPM companions	123
4.6	New CPM companion to HD 4391	145
4.7	New CPM companion to HD 43162	146
4.8	New CPM companion to HD 157347	146
4.9	New CPM companion to HD 218868	147
4.10	Examples of the linear motion of unrelated field stars	150
6.1	Number and time coverage of CfA RV measurements	201
6.2	Error profiles of the CfA RV measurements	203
6.3	$P(\chi^2)$ distribution for constant RV stars	205
6.4	$P(\chi^2)$ distribution for constant RV stars as a function of the number of	
	observations	207
6.5	Error distributions as a function of $P(\chi^2)$ for constant RV stars	208
6.6	Single-lined spectroscopic orbits of HD 10307 and HD 17382 \ldots	213
6.7	Single-lined spectroscopic orbits of HD 24409 and HD 32850 \ldots	213
6.8	Single-lined spectroscopic orbits of HD 54371 and HD 79028 \ldots	214
6.9	Single-lined spectroscopic orbits of HD 101206 and HD 112758A	214
6.10	Single-lined spectroscopic orbits of HD 121370 and HD 128642	215
6.11	Single-lined spectroscopic orbits of HD 142267 and HD 185414	215
6.12	Single-lined spectroscopic orbit of HD 224465	216

6.13	Double-lined spectroscopic orbits of HD 8997 and HD 13974 \ldots	219
6.14	Double-lined spectroscopic orbits of HD 45088 and HD 80715 \hdots	220
6.15	Double-lined spectroscopic orbits of HD 111312 and HD 148704 $\ .$	220
6.16	Double-lined spectroscopic orbits of HD 223778	221
7.1	Illustration of companion nomenclature according to the WMC standards	230
7.2	The observed frequency single, binary, triple, and quadruple systems	244
8.1	The DM91 frequency single, binary, triple, and quadruple systems. $\ . \ . \ .$	286
8.2	HR diagram for the DM91 sample	292
8.3	Number and time coverage of CCPS RV measurements	296
8.4	External error distribution in the CCPS RV measurements	297
8.5	Multiplicity statistics by color	309
8.6	Multiplicity statistics by spectral type	310
8.7	Distribution of chromospheric activity by $B - V$ Color $\ldots \ldots \ldots$	312
8.8	Multiplicity statistics by chromospheric activity	313
8.9	Multiplicity statistics by metallicity	314
8.10	Planet metallicity correlation	316
8.11	Period distribution for the 258 confirmed companions	318
8.12	Period-eccentricity relationship	321
8.13	Eccentricity distribution	323
8.14	Mass-ratio distribution	324
8.15	Mass-ratio – period relation	325
8.16	Secondary mass distribution	327
8.17	Mobile diagrams of triple systems (1 of 3)	338
8.18	Mobile diagrams of triple systems (2 of 3)	339
8.19	Mobile diagrams of triple systems (3 of 3)	340

8.20	Mobile diagrams of quadruple systems	341
8.21	Mobile diagrams of quintuple and higher order systems	342
B.1	SFP inspection for HD 000166 with S1-E1 on UT 2007/07/29, seq 001 $$	392
B.2	SFP inspection for HD 000166 with S1-E1 on UT 2007/08/14, seq 001 $$	393
B.3	SFP inspection for HD 000166 with S1-W1 on UT $2007/08/14,\mathrm{seq}~002$	393
B.4	SFP inspection for HD 001461 with S1-E1 on UT 2008/07/22, seq 001 $$	393
B.5	SFP inspection for HD 001461 with S1-W1 on UT $2008/07/27,\mathrm{seq}$ 001	394
B.6	SFP inspection for HD 001461 with S2-W1 on UT $2007/09/17,\mathrm{seq}$ 001	394
B.7	SFP inspection for HD 001562 with S1-E1 on UT 2007/08/16, seq 002 $$	394
B.8	SFP inspection for HD 001562 with S1-E1 on UT 2008/07/09, seq 001 $$	395
B.9	SFP inspection for HD 001562 with S1-W1 on UT 2007/08/16, seq 001	395
B.10	SFP inspection for HD 001562 with S1-W1 on UT $2008/07/08,\mathrm{seq}$ 001	395
B.11	SFP inspection for HD 003196 with S1-E1 on UT 2007/08/14, seq 003 $$	396
B.12	SFP inspection for HD 003196 with S1-E1 on UT 2008/07/22, seq 001 $$	396
B.13	SFP inspection for HD 003196 with S1-W1 on UT $2007/08/14,\mathrm{seq}$ 004	396
B.14	SFP inspection for HD 003196 with S1-W1 on UT $2008/07/07,\mathrm{seq}~001$	397
B.15	SFP inspection for HD 003196 with S2-E2 on UT 2007/09/27, seq 001 $$	397
B.16	SFP inspection for HD 003196 with S2-W1 on UT $2007/09/17,\mathrm{seq}~001$	397
B.17	SFP inspection for HD 003651 with S1-E1 on UT 2007/07/29, seq 001 $$	398
B.18	SFP inspection for HD 003651 with S1-E1 on UT 2007/08/14, seq 001 $$	398
B.19	SFP inspection for HD 003651 with S1-W1 on UT $2007/08/14,\mathrm{seq}~002$	398
B.20	SFP inspection for HD 003765 with S1-E1 on UT 2007/08/16, seq 002 $$	399
B.21	SFP inspection for HD 003765 with S1-E1 on UT 2008/07/09, seq 001 $$	399
B.22	SFP inspection for HD 003765 with S1-W1 on UT 2007/08/16, seq 001	399
B.23	SFP inspection for HD 003765 with S1-W1 on UT $2008/07/08$, seq 001	400

B.24 SFP inspection for HD 004256 with S1-E1 on UT 2007/08/14, seq 001 . . 400B.25 SFP inspection for HD 004256 with S1-W1 on UT 2007/08/14, seq 002 . . 400 B.26 SFP inspection for HD 004628 with S1-E1 on UT 2007/08/14, seq 001 . . 401 B.27 SFP inspection for HD 004628 with S1-W1 on UT 2007/08/14, seq 002 . . 401 B.28 SFP inspection for HD 004635 with S1-E1 on UT 2007/09/16, seq 002 . . 401 B.29 SFP inspection for HD 004635 with S1-W1 on UT 2007/09/16, seq 001 . . 402 B.30 SFP inspection for HD 004635 with S2-E1 on UT 2007/08/20, seq 001 . 402B.31 SFP inspection for HD 004813 with S1-W1 on UT 2008/07/27, seq 001 . . 402 B.32 SFP inspection for HD 004915 with S1-W1 on UT 2008/07/27, seq 001 . . 403 B.33 SFP inspection for HD 004915 with S2-W1 on UT 2007/09/17, seq 001 . . 403B.34 SFP inspection for HD 007590 with S1-E1 on UT 2007/02/04, seq 001 . . 403B.35SFP inspection for HD 007590 with S1-E1 on UT 2007/02/04, seq 002 . . 404 B.36 SFP inspection for HD 007590 with S1-E1 on UT 2007/08/16, seq 002 . . 404 B.37 SFP inspection for HD 007590 with S1-E1 on UT 2008/07/09, seq 001 . . 404B.38 SFP inspection for HD 007590 with S1-W1 on UT 2007/08/16, seq 001 . . 405B.39 SFP inspection for HD 007590 with S1-W1 on UT 2008/07/08, seq 001 . . 405B.40 SFP inspection for HD 007924 with S1-E1 on UT 2007/09/16, seq 002 . . 405B.41 SFP inspection for HD 007924 with S1-W1 on UT 2007/09/16, seq 001 . . 406 B.42 SFP inspection for HD 008997 with S1-E1 on UT 2008/07/23, seq 001 . . 406 B.43 SFP inspection for HD 008997 with S1-W1 on UT 2008/07/07, seq 001 . . 406 B.44 SFP inspection for HD 008997 with S2-E2 on UT 2007/09/18, seq 001 . . 407 B.45SFP inspection for HD 008997 with S2-W1 on UT 2007/09/17, seq 001 . . 407 B.46 SFP inspection for HD 010008 with S2-E2 on UT 2007/09/27, seq 001 . . 407 B.47 SFP inspection for HD 010008 with S2-W1 on UT 2007/09/17, seq 001 . . 408 B.48 SFP inspection for HD 010086 with S1-E1 on UT 2007/08/16, seq 002 . 408 SFP inspection for HD 010086 with S1-W1 on UT 2007/08/16, seq 001 . . B.49 408 B.50 SFP inspection for HD 010476 with S1-E1 on UT 2007/07/29, seq 001 . . 409B.51 SFP inspection for HD 010476 with S1-E1 on UT 2007/08/16, seq 001 . . 409B.52SFP inspection for HD 010476 with S1-W1 on UT 2007/08/14, seq 001 . . 409B.53 SFP inspection for HD 010780 with S1-E1 on UT 2007/09/16, seq 002 . . 410 B.54 SFP inspection for HD 010780 with S1-W1 on UT 2007/09/16, seq 001 . . 410 B.55 SFP inspection for HD 010780 with S2-E1 on UT 2007/08/20, seq 002 . 410B.56 SFP inspection for HD 010780 with S2-W1 on UT 2008/06/24, seq 001 . . 411 B.57 SFP inspection for HD 012051 with S1-E1 on UT 2007/01/26, seq 001 . 411 B.58 SFP inspection for HD 012051 with S2-E2 on UT 2007/09/18, seq 001 . . 411 B.59 SFP inspection for HD 012051 with S2-E2 on UT 2007/09/18, seq 002 . . 412 B.60 SFP inspection for HD 012051 with S2-W1 on UT 2007/09/17, seq 001 . . 412 B.61SFP inspection for HD 012846 with S2-E2 on UT 2007/09/18, seq 001 . . 412B.62SFP inspection for HD 012846 with S2-W1 on UT 2007/09/17, seq 001 . . 413 B.63 SFP inspection for HD 016160 with S1-E1 on UT 2007/08/15, seq 001 . . 413 B.64 SFP inspection for HD 016160 with S2-E2 on UT 2007/09/18, seq 001 . . 413B.65 SFP inspection for HD 016160 with S2-W1 on UT 2007/09/17, seq 001 . 414 B.66 SFP inspection for HD 016287 with S1-E1 on UT 2007/08/15, seq 001 . . 414 B.67 SFP inspection for HD 016287 with S1-E1 on UT 2007/10/31, seq 001 . . 414 B.68 SFP inspection for HD 016673 with S1-E1 on UT 2007/08/15, seq 001 . . 415B.69 SFP inspection for HD 016673 with S2-E2 on UT 2007/09/27, seq 001 . . 415B.70 SFP inspection for HD 016765 with S1-E1 on UT 2007/08/15, seq 001 . . 415 B.71 SFP inspection for HD 016765 with S1-E1 on UT 2007/10/31, seq 001 . . 416 B.72 SFP inspection for HD 016765 with S1-W1 on UT 2007/10/31, seq 002 . . 416 B.73 SFP inspection for HD 016765 with S2-E2 on UT 2007/09/27, seq 001 . . 416 B.74 SFP inspection for HD 016765 with S2-W1 on UT 2007/09/17, seq 001 . . 417 B.75 SFP inspection for HD 017382 with S1-W1 on UT 2008/07/07, seq 001 . . 417 B.76 SFP inspection for HD 017382 with S1-W1 on UT 2008/07/08, seq 001 . . 417B.77 SFP inspection for HD 017382 with S2-E2 on UT 2007/09/18, seq 001 . . 418 B.78 SFP inspection for HD 017382 with S2-W1 on UT 2007/09/17, seq 001 . . 418 B.79 SFP inspection for HD 018143 with S1-E1 on UT 2007/10/31, seq 001 . . 418 B.80 SFP inspection for HD 018143 with S1-E1 on UT 2008/07/23, seq 001 . . 419 SFP inspection for HD 018143 with S1-W1 on UT 2007/11/01, seq 001 . . B.81 419B.82SFP inspection for HD 018143 with S2-W1 on UT 2007/09/17, seq 001 . . 419 B.83 SFP inspection for HD 018632 with S1-E1 on UT 2007/11/01, seq 001 . . 420B.84 SFP inspection for HD 018632 with S1-W1 on UT 2007/11/01, seq 002 . 420 B.85 SFP inspection for HD 018757 with S1-E1 on UT 2007/09/16, seq 003 . . 420 B.86 SFP inspection for HD 018757 with S1-W1 on UT 2007/09/16, seq 002. 421B.87 SFP inspection for HD 018757 with S2-E1 on UT 2007/08/20, seq 001 . . 421B.88 SFP inspection for HD 018803 with S2-E2 on UT 2007/09/18, seq 001 . . 421 B.89 SFP inspection for HD 018803 with S2-W1 on UT 2007/09/17, seq 001 . . 422 B.90 SFP inspection for HD 019994 with S1-E1 on UT 2007/11/01, seq 001 . . 422B.91 SFP inspection for HD 019994 with S1-W1 on UT 2007/11/01, seq 002 . . 422 B.92 SFP inspection for HD 019994 with S2-E2 on UT 2007/09/27, seq 001 . . 423 B.93 SFP inspection for HD 019994 with S2-E2 on UT 2007/09/27, seq 002 . . 423 B.94 SFP inspection for HD 019994 with S2-E2 on UT 2007/09/27, seq 003 . . 423B.95 SFP inspection for HD 020165 with S1-E1 on UT 2007/11/01, seq 001 . . 424B.96 SFP inspection for HD 020165 with S1-W1 on UT 2007/11/01, seq 002 . . 424B.97 SFP inspection for HD 020619 with S1-E1 on UT 2007/11/01, seq 001 . . 424 B.98 SFP inspection for HD 020619 with S1-W1 on UT 2007/11/01, seq 002 . . 425B.99 SFP inspection for HD 022049 with S2-E2 on UT 2007/09/27, seq 001 . . 425B.100 SFP inspection for HD 022049 with S2-E2 on UT 2007/09/27, seq 002 . . 425B.101 SFP inspection for HD 022879 with S1-E1 on UT 2007/11/01, seq 001 . . 426 B.102 SFP inspection for HD 022879 with S2-E2 on UT 2007/09/27, seq 002 426 . . B.103 SFP inspection for HD 022879 with S2-E2 on UT 2007/09/27, seq 003 426 . . B.104 SFP inspection for HD 024238 with S1-E1 on UT 2007/09/16, seq 002 . . 427 B.105 SFP inspection for HD 024238 with S1-W1 on UT 2007/09/16, seq 001 . . 427SFP inspection for HD 024409 with S1-E1 on UT 2007/09/16, seq 002 . B.106 427 B.107 SFP inspection for HD 024409 with S1-W1 on UT 2007/09/16, seq 001 . . 428SFP inspection for HD 024409 with S2-E1 on UT 2007/08/20, seq 001 . . 428 B.108 SFP inspection for HD 024496 with S1-E1 on UT 2007/10/31, seq 001 . . B.109 428B.110 SFP inspection for HD 024496 with S1-W1 on UT 2007/10/31, seq 002 . . 429SFP inspection for HD 024496 with S2-E2 on UT 2007/09/26, seq 001 . . 429 B.111 SFP inspection for HD 024496 with S2-E2 on UT 2007/09/27, seq 001 . . B.112 429B.113 SFP inspection for HD 025457 with S1-E1 on UT 2007/10/31, seq 001 . . 430B.114 SFP inspection for HD 025457 with S1-W1 on UT 2007/10/31, seq 002 . . 430B.115 SFP inspection for HD 025457 with S2-E2 on UT 2007/09/26, seq 001 . . 430 SFP inspection for HD 025457 with S2-E2 on UT 2007/09/27, seq 001 . . B.116 431SFP inspection for HD 025457 with S2-E2 on UT 2007/09/27, seq 002 . B.117 431B.118 SFP inspection for HD 025665 with S1-E1 on UT 2007/09/16, seq 002 . 431SFP inspection for HD 025665 with S1-W1 on UT 2007/09/16, seq 001 . . 432 B.119 SFP inspection for HD 026913 with S1-E1 on UT 2007/10/31, seq 001 . . B.120 432SFP inspection for HD 026913 with S1-W1 on UT 2007/10/31, seq 002 . . B.121 432SFP inspection for HD 026913 with S2-E2 on UT 2007/09/26, seq 001 . . B.122 433B.123 SFP inspection for HD 026923 with S1-E1 on UT 2007/10/31, seq 001 . . 433 B.124 SFP inspection for HD 026923 with S1-W1 on UT 2007/10/31, seq 002 . . 433SFP inspection for HD 026923 with S2-E2 on UT 2007/09/26, seq 001 . . 434 B.125 B.126 SFP inspection for HD 029883 with S2-E2 on UT 2007/09/18, seq 001 . . 434 B.127 SFP inspection for HD 029883 with S2-W1 on UT 2007/09/17, seq 001 . . 434 B.128 SFP inspection for HD 032850 with S1-E1 on UT 2007/10/31, seq 001 . . 435B.129 SFP inspection for HD 032850 with S1-W1 on UT 2007/10/31, seq 002 . . 435B.130 SFP inspection for HD 032923 with S1-E1 on UT 2007/02/25, seq 001 . . 435B.131 SFP inspection for HD 032923 with S1-E1 on UT 2007/02/25, seq 002 . . 436SFP inspection for HD 032923 with S1-E1 on UT 2007/03/11, seq 001 . . B.132 436 B.133 SFP inspection for HD 032923 with S1-E1 on UT 2007/10/31, seq 001 . . 436SFP inspection for HD 032923 with S1-W1 on UT 2007/10/31, seq 002 . . 437 B.134 SFP inspection for HD 032923 with S2-E2 on UT 2007/09/26, seq 001 . . B.135 437 B.136 SFP inspection for HD 032923 with S2-E2 on UT 2007/09/27, seq 001 . . 437SFP inspection for HD 035112 with S1-E1 on UT 2007/10/31, seq 001 . . 438 B.137 SFP inspection for HD 035112 with S1-W1 on UT 2007/10/31, seq 002 . . B.138 438B.139 SFP inspection for HD 037008 with S1-E1 on UT 2007/01/25, seq 001 . . 438B.140 SFP inspection for HD 037008 with S1-W1 on UT 2007/11/01, seq 002 . . 439B.141 SFP inspection for HD 037008 with S2-E2 on UT 2007/11/19, seq 001 . . 439SFP inspection for HD 037008 with S2-E2 on UT 2007/11/19, seq 002 B.142 . . 439B.143 SFP inspection for HD 037008 with S2-E2 on UT 2007/11/27, seq 001 . . 440 B.144 SFP inspection for HD 037394 with S1-E1 on UT 2007/01/25, seq 001 . . 440SFP inspection for HD 037394 with S1-E1 on UT 2007/01/26, seq 002 . . 440 B.145 SFP inspection for HD 037394 with S1-E1 on UT 2007/04/14, seq 001 . . B.146 441 SFP inspection for HD 037394 with S1-W1 on UT 2007/11/01, seq 001 . . 441 B.147 B.148 SFP inspection for HD 037394 with S2-E2 on UT 2007/11/19, seq 003 . . 441 B.149 SFP inspection for HD 037394 with S2-E2 on UT 2007/11/19, seq 004 . . 442 B.150 SFP inspection for HD 037394 with S2-E2 on UT 2007/11/19, seq 005 . . 442 SFP inspection for HD 037394 with S2-E2 on UT 2007/11/27, seq 001 . B.151 442B.152 SFP inspection for HD 038230 with S1-E1 on UT 2007/01/25, seq 001 . . 443 B.153 SFP inspection for HD 038230 with S1-W1 on UT 2007/11/01, seq 001 . . 443 B.154 SFP inspection for HD 038230 with S2-E2 on UT 2007/11/19, seq 002 . . 443B.155 SFP inspection for HD 038230 with S2-E2 on UT 2007/11/27, seq 001 . . 444 B.156 SFP inspection for HD 038858 with S1-E1 on UT 2007/10/31, seq 001 . . 444 B.157 SFP inspection for HD 038858 with S1-W1 on UT 2007/10/31, seq 002 . . 444SFP inspection for HD 038858 with S2-E2 on UT 2007/09/26, seq 001 . B.158 445B.159 SFP inspection for HD 040397 with S1-W1 on UT 2007/11/01, seq 001 . . 445SFP inspection for HD 040397 with S2-E2 on UT 2007/09/26, seq 001 . . 445 B.160 SFP inspection for HD 041593 with S1-E1 on UT 2007/10/31, seq 001 . . B.161 446 B.162 SFP inspection for HD 041593 with S1-W1 on UT 2007/11/01, seq 001 . . 446SFP inspection for HD 041593 with S2-E2 on UT 2007/09/26, seq 002 . . 446 B.163 SFP inspection for HD 042618 with S1-E1 on UT 2007/10/31, seq 001 . . 447 B.164 B.165 SFP inspection for HD 042618 with S1-W1 on UT 2007/11/01, seq 001 . . 447B.166 SFP inspection for HD 042618 with S2-E2 on UT 2007/09/26, seq 001 . . 447 B.167 SFP inspection for HD 046588 with S1-E1 on UT 2007/04/14, seq 001 . . 448 SFP inspection for HD 046588 with S1-E1 on UT 2007/09/16, seq 001 . . B.168 448B.169 SFP inspection for HD 046588 with S1-W1 on UT 2008/04/26, seq 001 . 448 B.170 SFP inspection for HD 051419 with S1-E1 on UT 2007/02/05, seq 002 . 449SFP inspection for HD 051419 with S1-E1 on UT 2007/02/05, seq 003 . . 449 B.171 SFP inspection for HD 051419 with S1-W1 on UT 2008/04/26, seq 001 . . B.172 449B.173 SFP inspection for HD 051866 with S1-E1 on UT 2007/01/25, seq 001 . . 450B.174 SFP inspection for HD 051866 with S1-W1 on UT 2007/11/01, seq 002 . . 450B.175 SFP inspection for HD 051866 with S2-E2 on UT 2007/11/19, seq 001 . . 450 B.176 SFP inspection for HD 051866 with S2-E2 on UT 2007/11/19, seq 002 . 451SFP inspection for HD 054371 with S1-E1 on UT 2007/02/04, seq 001 . . B.177 451B.178 SFP inspection for HD 054371 with S1-E1 on UT 2008/04/25, seq 001 . . 451 B.179 SFP inspection for HD 054371 with S1-W1 on UT 2007/11/01, seq 002 . . 452 B.180 SFP inspection for HD 054371 with S1-W1 on UT 2008/04/26, seq 001 . . 452B.181 SFP inspection for HD 055575 with S1-E1 on UT 2007/01/26, seq 001 . . 452B.182 SFP inspection for HD 055575 with S1-E1 on UT 2007/02/03, seq 001 . . 453B.183 SFP inspection for HD 055575 with S1-W1 on UT 2007/11/01, seq 001 . . 453SFP inspection for HD 055575 with S2-E2 on UT 2007/11/20, seq 001 . B.184 453B.185 SFP inspection for HD 055575 with S2-E2 on UT 2007/11/27, seq 002 . . 454SFP inspection for HD 059747 with S1-W1 on UT 2007/11/01, seq 001 . . 454 B.186 B.187 SFP inspection for HD 063433 with S1-E1 on UT 2007/03/20, seq 001 . . 454B.188 SFP inspection for HD 063433 with S1-E1 on UT 2008/04/13, seq 001 . . 455SFP inspection for HD 063433 with S1-W1 on UT 2007/11/01, seq 001 . . 455 B.189 SFP inspection for HD 063433 with S1-W1 on UT 2008/04/12, seq 001 . . B.190 455SFP inspection for HD 065430 with S1-W1 on UT 2008/04/26, seq 001 . . B.191 456B.192 SFP inspection for HD 065583 with S1-E1 on UT 2008/04/13, seq 001 . . 456B.193 SFP inspection for HD 065583 with S1-W1 on UT 2008/04/26, seq 001 . 456SFP inspection for HD 067228 with S1-E1 on UT 2007/02/06, seq 002 . B.194 457B.195 SFP inspection for HD 067228 with S1-E1 on UT 2007/02/25, seq 001 . . 457B.196 SFP inspection for HD 067228 with S1-E1 on UT 2007/02/25, seq 002 . . 457SFP inspection for HD 067228 with S2-E2 on UT 2007/11/29, seq 003 B.197 . . 458 SFP inspection for HD 068017 with S1-E1 on UT 2008/04/13, seq 001 B.198 . . 458B.199 SFP inspection for HD 068017 with S1-E1 on UT 2008/04/25, seq 001 . . 458B.200 SFP inspection for HD 068017 with S1-W1 on UT 2008/04/12, seq 001 . . 459B.201 SFP inspection for HD 068255 with S1-E1 on UT 2007/02/25, seq 002 459. . B.202 SFP inspection for HD 068255 with S1-E1 on UT 2007/02/25, seq 003 459. . B.203 SFP inspection for HD 079096 with S1-E1 on UT 2007/02/06, seq 003 460. . B.204 SFP inspection for HD 079096 with S1-E1 on UT 2007/02/25, seq 001 460. . B.205 SFP inspection for HD 079096 with S1-E1 on UT 2007/02/25, seq 002 . . 460 B.206 SFP inspection for HD 079096 with S1-E1 on UT 2007/03/09, seq 001 461. . B.207 SFP inspection for HD 079096 with S1-E1 on UT 2007/03/09, seq 002 461B.208 SFP inspection for HD 079096 with S1-E1 on UT 2007/03/10, seq 001 . . 461 B.209 SFP inspection for HD 079096 with S1-E1 on UT 2007/03/11, seq 001 462 SFP inspection for HD 079096 with S1-E1 on UT 2008/04/14, seq 001 B.210 462SFP inspection for HD 079096 with S1-E1 on UT 2008/04/14, seq 002 B.211 . . 462SFP inspection for HD 079096 with S1-E1 on UT 2008/04/14, seq 003 B.212 463. . SFP inspection for HD 079096 with S1-W1 on UT 2008/04/12, seq 001 . . B.213 463B.214 SFP inspection for HD 079096 with S2-E2 on UT 2007/11/29, seq 002 463. . SFP inspection for HD 079969 with S1-E1 on UT 2007/02/06, seq 001 B.215 . . 464SFP inspection for HD 079969 with S1-E1 on UT 2007/02/25, seq 001 B.216 . . 464B.217 SFP inspection for HD 079969 with S1-E1 on UT 2008/04/13, seq 001 . . 464SFP inspection for HD 079969 with S1-W1 on UT 2008/04/12, seq 001 . . B.218 465B.219 SFP inspection for HD 079969 with S2-E2 on UT 2007/11/29, seq 001 . . 465SFP inspection for HD 080715 with S1-E1 on UT 2008/04/13, seq 001 . . B.220 465B.221 SFP inspection for HD 080715 with S1-W1 on UT 2008/04/26, seq 001 . . 466 B.222 SFP inspection for HD 082443 with S1-E1 on UT 2007/02/06, seq 001 . . 466 SFP inspection for HD 082443 with S1-E1 on UT 2008/04/13, seq 001 . . B.223 466 SFP inspection for HD 082443 with S1-E1 on UT 2008/04/13, seq 002 . . B.224 467SFP inspection for HD 082443 with S1-W1 on UT 2008/04/12, seq 001 . . B.225 467B.226 SFP inspection for HD 082443 with S2-E2 on UT 2007/11/29, seq 001 . . 467B.227 SFP inspection for HD 082885 with S1-E1 on UT 2007/01/26, seq 003 . . 468 B.228 SFP inspection for HD 082885 with S1-W1 on UT 2007/04/24, seq 001 . . 468 B.229 SFP inspection for HD 082885 with S1-W1 on UT 2008/04/26, seq 001 . . 468B.230 SFP inspection for HD 082885 with S2-E2 on UT 2007/11/29, seq 001 . . 469B.231 SFP inspection for HD 087883 with S1-E1 on UT 2007/02/06, seq 001 . . 469 B.232 SFP inspection for HD 087883 with S1-W1 on UT 2008/04/26, seq 001 . . 469B.233 SFP inspection for HD 087883 with S2-E2 on UT 2007/11/29, seq 001 . . 470B.234 SFP inspection for HD 089269 with S1-E1 on UT 2007/05/17, seq 002 . . 470B.235 SFP inspection for HD 089269 with S1-W1 on UT 2007/04/24, seq 001 . . 470SFP inspection for HD 090343 with S1-E1 on UT 2008/06/25, seq 001 . . B.236 471SFP inspection for HD 090343 with S2-W1 on UT 2008/06/24, seq 001 . . B.237 471SFP inspection for HD 094765 with S1-E1 on UT 2008/04/14, seq 001 . . 471 B.238 SFP inspection for HD 094765 with S1-W1 on UT 2008/04/15, seq 001 . . 472 B.239 B.240 SFP inspection for HD 096064 with S1-E1 on UT 2008/04/14, seq 001 . . 472 SFP inspection for HD 096064 with S1-W1 on UT 2008/04/15, seq 001 . . 472 B.241 SFP inspection for HD 097334 with S1-E1 on UT 2008/04/12, seq 002 . B.242 473B.243 SFP inspection for HD 097334 with S1-W1 on UT 2008/04/12, seq 001 . . 473SFP inspection for HD 097658 with S1-E1 on UT 2008/04/13, seq 001 . . B.244 473B.245 SFP inspection for HD 097658 with S1-E1 on UT 2008/04/14, seq 001 . . 474 SFP inspection for HD 098230 with S1-E1 on UT 2007/02/06, seq 001 B.246 . . 474 B.247 SFP inspection for HD 098230 with S1-E1 on UT 2007/03/08, seq 001 . . 474B.248 SFP inspection for HD 098230 with S1-E1 on UT 2007/03/08, seq 002 . . 475SFP inspection for HD 098230 with S1-E1 on UT 2007/05/17, seq 001 . . 475 B.249 SFP inspection for HD 098230 with S1-E1 on UT 2007/05/17, seq 002 B.250 . . 475SFP inspection for HD 098230 with S1-E1 on UT 2007/05/17, seq 003 . . B.251 476SFP inspection for HD 098230 with S1-E1 on UT 2008/04/13, seq 001 . . B.252 476B.253 SFP inspection for HD 098230 with S1-W1 on UT 2007/04/24, seq 001 . . 476 B.254 SFP inspection for HD 098230 with S1-W1 on UT 2007/05/28, seq 001 . . 477 SFP inspection for HD 098230 with S1-W1 on UT 2007/06/01, seq 001 . . 477 B.255 B.256 SFP inspection for HD 098230 with S1-W1 on UT 2007/06/01, seq 002 . . 477 B.257 SFP inspection for HD 098230 with S1-W1 on UT 2007/06/01, seq 003 . . 478 B.258 SFP inspection for HD 098230 with S1-W1 on UT 2008/04/12, seq 001 . . 478B.259 SFP inspection for HD 098281 with S1-E1 on UT 2008/04/14, seq 001 . . 478B.260 SFP inspection for HD 098281 with S1-W1 on UT 2008/04/15, seq 001 . . 479 B.261 SFP inspection for HD 099028 with S1-E1 on UT 2007/02/06, seq 003 . . 479SFP inspection for HD 099491 with S1-E1 on UT 2007/03/08, seq 001 . . B.262 479B.263 SFP inspection for HD 099491 with S1-E1 on UT 2008/04/14, seq 001 . . 480SFP inspection for HD 099491 with S1-E1 on UT 2008/04/14, seq 002 . B.264 480B.265 SFP inspection for HD 099491 with S1-W1 on UT 2008/04/15, seq 001 . . 480B.266 SFP inspection for HD 099492 with S1-W1 on UT 2008/04/15, seq 001 . . 481 SFP inspection for HD 100180 with S1-E1 on UT 2007/02/06, seq 001 . . 481 B.267 SFP inspection for HD 100180 with S1-E1 on UT 2007/03/08, seq 001 . . B.268 481 B.269 SFP inspection for HD 100180 with S1-W1 on UT 2007/04/24, seq 001 . . 482B.270 SFP inspection for HD 101177 with S1-E1 on UT 2007/05/17, seq 001 . . 482 B.271 SFP inspection for HD 101177 with S1-W1 on UT 2007/04/24, seq 001 . . 482 SFP inspection for HD 101206 with S1-E1 on UT 2008/04/12, seq 002 . . B.272 483B.273 SFP inspection for HD 101206 with S1-W1 on UT 2008/04/12, seq 001 . . 483B.274 SFP inspection for HD 104304 with S1-E1 on UT 2008/04/14, seq 001 . . 483B.275 SFP inspection for HD 104304 with S1-W1 on UT 2008/04/15, seq 001 . . 484 SFP inspection for HD 105631 with S1-E1 on UT 2007/05/17, seq 001 . . B.276 484SFP inspection for HD 105631 with S1-W1 on UT 2007/04/24, seq 001 . . B.277 484B.278 SFP inspection for HD 105631 with S1-W1 on UT 2007/06/01, seq 001 . . 485B.279 SFP inspection for HD 105631 with S1-W1 on UT 2008/04/26, seq 001 . . 485B.280 SFP inspection for HD 108954 with S1-E1 on UT 2007/04/11, seq 001 . . 485SFP inspection for HD 108954 with S1-E1 on UT 2007/04/11, seq 002 . B.281 486 B.282 SFP inspection for HD 108954 with S1-W1 on UT 2007/04/24, seq 001 . . 486 B.283 SFP inspection for HD 110833 with S1-E1 on UT 2007/04/14, seq 001 . . 486 B.284 SFP inspection for HD 110833 with S1-E1 on UT 2007/07/28, seq 001 . . 487B.285 SFP inspection for HD 110833 with S1-W1 on UT 2007/04/17, seq 001 . . 487B.286 SFP inspection for HD 110833 with S1-W1 on UT 2007/04/17, seq 002. 487 B.287 SFP inspection for HD 110833 with S1-W1 on UT 2007/05/28, seq 002 . . 488 SFP inspection for HD 112758 with S1-E1 on UT 2008/04/14, seq 001 . . B.288 488B.289 SFP inspection for HD 112758 with S1-W1 on UT 2008/04/15, seq 001 . . 488 SFP inspection for HD 113449 with S1-E1 on UT 2008/04/14, seq 001 . . B.290 489B.291 SFP inspection for HD 113449 with S1-W1 on UT 2008/04/15, seq 001 . . 489B.292 SFP inspection for HD 114783 with S1-E1 on UT 2008/04/13, seq 001 . . 489SFP inspection for HD 114783 with S1-E1 on UT 2008/04/25, seq 001 . . B.293 490SFP inspection for HD 114783 with S1-W1 on UT 2008/04/13, seq 002. B.294 490 B.295 SFP inspection for HD 115404 with S1-E1 on UT 2007/02/06, seq 001 . . 490SFP inspection for HD 115404 with S1-E1 on UT 2007/02/06, seq 002 B.296 . . 491 B.297 SFP inspection for HD 115404 with S1-E1 on UT 2007/03/08, seq 001 . . 491B.298 SFP inspection for HD 115404 with S1-W1 on UT 2007/04/24, seq 001 . . 491B.299 SFP inspection for HD 116442 with S1-E1 on UT 2007/03/12, seq 002 . 492B.300 SFP inspection for HD 116442 with S1-W1 on UT 2007/06/01, seq 001 . . 492SFP inspection for HD 116443 with S1-E1 on UT 2007/03/12, seq 001 . . 492 B.301 SFP inspection for HD 116443 with S1-W1 on UT 2007/06/01, seq 001 . . B.302 493B.303 SFP inspection for HD 116956 with S1-E1 on UT 2007/04/03, seq 001 . . 493B.304 SFP inspection for HD 116956 with S1-E1 on UT 2007/04/03, seq 003 . . 493B.305 SFP inspection for HD 116956 with S1-E1 on UT 2008/06/25, seq 001 . . 494B.306 SFP inspection for HD 116956 with S1-W1 on UT 2007/05/28, seq 001 . . 494B.307 SFP inspection for HD 116956 with S1-W1 on UT 2008/04/26, seq 001 . . 494B.308 SFP inspection for HD 116956 with S2-W1 on UT 2008/06/24, seq 001 . . 495SFP inspection for HD 119332 with S1-E1 on UT 2007/04/03, seq 002 . . B.309 495 B.310 SFP inspection for HD 119332 with S1-E1 on UT 2008/04/12, seq 002 . . 495B.311 SFP inspection for HD 119332 with S1-W1 on UT 2008/04/12, seq 001 . . 496B.312 SFP inspection for HD 121560 with S1-E1 on UT 2007/02/05, seq 001 . . 496 B.313 SFP inspection for HD 121560 with S1-E1 on UT 2007/02/05, seq 002 496 . . SFP inspection for HD 121560 with S1-E1 on UT 2007/03/08, seq 001 B.314 497. . B.315 SFP inspection for HD 121560 with S1-W1 on UT 2007/04/24, seq 001 . . 497SFP inspection for HD 124292 with S1-E1 on UT 2007/03/12, seq 002 . . B.316 497 B.317 SFP inspection for HD 124292 with S1-E1 on UT 2008/04/13, seq 001 . . 498 B.318 SFP inspection for HD 124292 with S1-E1 on UT 2008/04/25, seq 001 . . 498SFP inspection for HD 124292 with S1-W1 on UT 2008/04/13, seq 002 . . B.319 498SFP inspection for HD 124850 with S1-E1 on UT 2007/02/16, seq 001 B.320 . . 499SFP inspection for HD 124850 with S1-E1 on UT 2007/03/23, seq 001 B.321 499B.322 SFP inspection for HD 124850 with S1-E1 on UT 2008/04/25, seq 001 499. . B.323 SFP inspection for HD 124850 with S1-E2 on UT 2007/05/31, seq 001 500SFP inspection for HD 124850 with S1-E2 on UT 2008/06/08, seq 001 B.324 500B.325 SFP inspection for HD 124850 with S1-E2 on UT 2008/06/08, seq 002 . . 500B.326 SFP inspection for HD 124850 with S1-W1 on UT 2008/07/08, seq 001 . . 501B.327 SFP inspection for HD 124850 with S2-W2 on UT 2008/06/09, seq 001 . . 501SFP inspection for HD 125455 with S1-E1 on UT 2007/03/12, seq 001 B.328 501B.329 SFP inspection for HD 125455 with S1-E1 on UT 2008/04/14, seq 001 . . 502B.330 SFP inspection for HD 125455 with S1-E1 on UT 2008/04/25, seq 001 . . 502B.331 SFP inspection for HD 125455 with S1-W1 on UT 2008/04/14, seq 002. 502B.332 SFP inspection for HD 127334 with E1-W1 on UT 2007/04/26, seq 001. 503B.333 SFP inspection for HD 127334 with S1-E1 on UT 2007/04/14, seq 001 . . 503B.334 SFP inspection for HD 127334 with S1-W1 on UT 2007/04/17, seq 001 . . 503B.335 SFP inspection for HD 127334 with S1-W1 on UT 2007/04/17, seq 002 . . 504 B.336 SFP inspection for HD 128165 with E1-W1 on UT 2007/04/26, seq 001 . . 504B.337 SFP inspection for HD 128165 with S1-E1 on UT 2007/04/14, seq 001 . . 504B.338 SFP inspection for HD 128165 with S1-E1 on UT 2007/07/28, seq 001 . . 505B.339 SFP inspection for HD 128165 with S1-W1 on UT 2007/05/28, seq 001 . . 505SFP inspection for HD 128311 with S1-E1 on UT 2008/04/13, seq 001 . . B.340 505SFP inspection for HD 128311 with S1-E1 on UT 2008/04/25, seq 001 . . B.341 506 B.342 SFP inspection for HD 128311 with S1-W1 on UT 2008/04/13, seq 002. 506B.343 SFP inspection for HD 128642 with S1-E1 on UT 2008/06/25, seq 001 . . 506B.344 SFP inspection for HD 128642 with S1-E1 on UT 2008/06/25, seq 002 . 507 SFP inspection for HD 128642 with S1-W1 on UT 2008/04/26, seq 001 . . B.345 507SFP inspection for HD 128642 with S1-W1 on UT 2008/04/26, seq 002. B.346 507B.347 SFP inspection for HD 128642 with S2-W1 on UT 2008/06/24, seq 001 . . 508SFP inspection for HD 130004 with S1-E1 on UT 2007/03/10, seq 001 . . B.348 508B.349 SFP inspection for HD 130004 with S1-E1 on UT 2008/04/13, seq 001 . . 508B.350SFP inspection for HD 130004 with S1-E1 on UT 2008/04/25, seq 001 509B.351 SFP inspection for HD 130004 with S1-W1 on UT 2007/05/28, seq 001 . . 509B.352 SFP inspection for HD 130004 with S1-W1 on UT 2008/04/13, seq 002. 509SFP inspection for HD 130307 with S1-E1 on UT 2008/04/13, seq 001 . . B.353 510SFP inspection for HD 130307 with S1-W1 on UT 2008/04/13, seq 002. B.354 510B.355 SFP inspection for HD 132142 with S1-E1 on UT 2008/04/12, seq 002 . . 510B.356 SFP inspection for HD 132142 with S1-E1 on UT 2008/06/25, seq 001 . . 511B.357 SFP inspection for HD 132142 with S1-W1 on UT 2008/04/12, seq 001 . . 511B.358 SFP inspection for HD 132142 with S2-W1 on UT 2008/06/24, seq 001 . . 511B.359 SFP inspection for HD 132254 with E1-W1 on UT 2007/04/26, seq 001 . . 512B.360 SFP inspection for HD 132254 with S1-E1 on UT 2007/04/11, seq 001 . . 512B.361 SFP inspection for HD 132254 with S1-W1 on UT 2007/04/24, seq 001 . . 512 B.362 SFP inspection for HD 135204 with S1-E1 on UT 2007/03/08, seq 001 513. . B.363 SFP inspection for HD 135204 with S1-E1 on UT 2008/06/23, seq 001 . . 513B.364 SFP inspection for HD 135204 with S1-E1 on UT 2008/07/05, seq 001 . . 513B.365SFP inspection for HD 135204 with S1-E1 on UT 2008/07/05, seq 002 514. . SFP inspection for HD 135204 with S1-E2 on UT 2007/05/31, seq 001 . . B.366 514B.367 SFP inspection for HD 135204 with S1-W1 on UT 2008/06/22, seq 001 . . 514SFP inspection for HD 135204 with S1-W1 on UT 2008/06/22, seq 002 . . B.368 515B.369 SFP inspection for HD 135204 with S1-W1 on UT 2008/07/08, seq 001 . . 515B.370 SFP inspection for HD 135204 with S1-W1 on UT 2008/07/08, seq 002 . . 515SFP inspection for HD 135204 with S2-W2 on UT 2008/06/09, seq 001 . . B.371 516SFP inspection for HD 135599 with S1-E1 on UT 2007/03/08, seq 001 . . B.372 516B.373 SFP inspection for HD 135599 with S1-E2 on UT 2007/05/31, seq 001 . . 516SFP inspection for HD 136202 with S1-E1 on UT 2007/02/16, seq 001 B.374 . . 517B.375 SFP inspection for HD 136202 with S1-E1 on UT 2007/03/08, seq 001 . . 517SFP inspection for HD 136202 with S1-E1 on UT 2008/04/13, seq 001 B.376 . . 517SFP inspection for HD 136202 with S1-E2 on UT 2007/05/31, seq 001 B.377 . . 518B.378 SFP inspection for HD 136202 with S1-E2 on UT 2007/06/01, seq 001 . . 518B.379 SFP inspection for HD 136202 with S1-W1 on UT 2008/04/13, seq 002 . . 518SFP inspection for HD 136713 with S1-E1 on UT 2008/04/25, seq 001 . . B.380 519B.381 SFP inspection for HD 136713 with S1-W1 on UT 2008/04/14, seq 001 . . 519B.382 SFP inspection for HD 136923 with S1-E1 on UT 2008/04/13, seq 001 . . 519B.383 SFP inspection for HD 136923 with S1-W1 on UT 2008/04/13, seq 002 . . 520B.384 SFP inspection for HD 137763 with S1-E1 on UT 2008/04/25, seq 001 . . 520B.385 SFP inspection for HD 137763 with S1-E1 on UT 2008/06/23, seq 001 . . 520B.386 SFP inspection for HD 137763 with S1-W1 on UT 2008/04/14, seq 001 . . 521B.387 SFP inspection for HD 137763 with S1-W1 on UT 2008/06/22, seq 001 . . 521

B.388 SFP inspection for HD 137763 with S1-W1 on UT 2008/07/07, seq 001 . . 521B.389 SFP inspection for HD 137763 with S1-W1 on UT 2008/07/07, seq 002 . . 522B.390 SFP inspection for HD 137763 with S2-W2 on UT 2008/06/09, seq 001 . . 522B.391 SFP inspection for HD 137778 with S1-E1 on UT 2008/04/25, seq 001 . . 522SFP inspection for HD 137778 with S1-W1 on UT 2008/04/14, seq 001 . . B.392 523B.393 SFP inspection for HD 137778 with S1-W1 on UT 2008/06/22, seq 001 . . 523SFP inspection for HD 137778 with S1-W1 on UT 2008/07/07, seq 001 . . B.394 523B.395 SFP inspection for HD 137778 with S1-W1 on UT 2008/07/07, seq 002. 524B.396 SFP inspection for HD 139323 with S1-E1 on UT 2008/04/12, seq 002 . 524SFP inspection for HD 139323 with S1-W1 on UT 2008/04/12, seq 001 . . B.397 524SFP inspection for HD 139341 with E1-W1 on UT 2007/04/26, seq 002. B.398 525B.399 SFP inspection for HD 139341 with S1-E1 on UT 2007/04/14, seq 001 . . 525B.400 SFP inspection for HD 139341 with S1-W1 on UT 2007/04/17, seq 001 . . 525B.401 SFP inspection for HD 139341 with S1-W1 on UT 2007/04/17, seq 002. 526B.402 SFP inspection for HD 139341 with S1-W1 on UT 2007/04/17, seq 004 . . 526B.403 SFP inspection for HD 139777 with S1-E1 on UT 2007/04/14, seq 001 . . 526B.404 SFP inspection for HD 139777 with S1-W1 on UT 2007/05/28, seq 001 . . 527B.405 SFP inspection for HD 139777 with S1-W1 on UT 2008/04/26, seq 001 . . 527SFP inspection for HD 139777 with S2-W1 on UT 2008/06/24, seq 001 . . B.406 527B.407 SFP inspection for HD 139813 with S1-E1 on UT 2008/06/25, seq 001 . . 528B.408 SFP inspection for HD 139813 with S1-W1 on UT 2008/04/26, seq 001 . . 528B.409 SFP inspection for HD 139813 with S2-W1 on UT 2008/06/24, seq 001 . . 528B.410 SFP inspection for HD 141272 with S1-E1 on UT 2008/04/25, seq 001 . . 529B.411 SFP inspection for HD 141272 with S1-W1 on UT 2008/04/14, seq 001 . . 529B.412 SFP inspection for HD 142267 with S1-E1 on UT 2007/08/15, seq 001 . . 529B.413 SFP inspection for HD 142267 with S1-E1 on UT 2008/04/13, seq 001 . . 530
B.414 SFP inspection for HD 142267 with S1-E2 on UT 2007/06/01, seq 001 . . 530B.415 SFP inspection for HD 142267 with S1-W1 on UT 2008/04/13, seq 002. 530B.416 SFP inspection for HD 144287 with S1-E1 on UT 2008/04/13, seq 001 . . 531B.417 SFP inspection for HD 144287 with S1-W1 on UT 2008/04/12, seq 001 . . 531SFP inspection for HD 145675 with E1-W1 on UT 2007/04/26, seq 001. B.418 531B.419 SFP inspection for HD 145675 with S1-E1 on UT 2007/05/17, seq 001 . . 532SFP inspection for HD 145675 with S1-W1 on UT 2007/04/24, seq 001 . . B.420 532B.421 SFP inspection for HD 146233 with S1-E1 on UT 2007/03/08, seq 001 . . 532B.422 SFP inspection for HD 146233 with S1-E1 on UT 2008/04/25, seq 001 . . 533B.423 SFP inspection for HD 146233 with S1-E1 on UT 2008/06/23, seq 001 . . 533SFP inspection for HD 146233 with S1-W1 on UT 2008/04/14, seq 001 . . B.424 533SFP inspection for HD 146233 with S1-W1 on UT 2008/06/22, seq 001 . . B.425 534B.426 SFP inspection for HD 148653 with S1-E1 on UT 2007/02/04, seq 001 . . 534B.427 SFP inspection for HD 149661 with S1-E1 on UT 2007/03/08, seq 002 . . 534SFP inspection for HD 149661 with S1-E2 on UT 2007/05/31, seq 001 B.428 535. . B.429 SFP inspection for HD 149806 with S1-E1 on UT 2008/04/25, seq 001 535. . B.430 SFP inspection for HD 149806 with S1-W1 on UT 2008/04/13, seq 001 . . 535B.431 SFP inspection for HD 149806 with S1-W1 on UT 2008/04/14, seq 001 . . 536SFP inspection for HD 151541 with S1-E1 on UT 2008/06/25, seq 001 . . B.432 536 B.433 SFP inspection for HD 151541 with S1-W1 on UT 2008/04/26, seq 001 . . 536 B.434 SFP inspection for HD 151541 with S2-W1 on UT 2008/06/24, seq 001 . . 537B.435 SFP inspection for HD 153557 with S1-E1 on UT 2008/06/25, seq 001 . . 537B.436 SFP inspection for HD 153557 with S1-W1 on UT 2007/05/30, seq 001 . . 537 B.437 SFP inspection for HD 153557 with S1-W1 on UT 2007/08/21, seq 001 . . 538B.438 SFP inspection for HD 153557 with S1-W1 on UT 2008/04/12, seq 001 . . 538B.439 SFP inspection for HD 153557 with S2-W1 on UT 2008/06/24, seq 001 . . 538 B.440 SFP inspection for HD 154345 with E1-W1 on UT 2007/04/26, seq 001. 539B.441 SFP inspection for HD 154345 with S1-E1 on UT 2007/04/14, seq 001 . . 539B.442 SFP inspection for HD 154345 with S1-W1 on UT 2007/04/17, seq 001 . . 539B.443 SFP inspection for HD 154345 with S1-W1 on UT 2007/04/17, seq 002 . . 540SFP inspection for HD 155712 with S1-E1 on UT 2008/04/25, seq 001 . . B.444 540B.445 SFP inspection for HD 155712 with S1-W1 on UT 2008/04/13, seq 001 . . 540SFP inspection for HD 155712 with S1-W1 on UT 2008/04/14, seq 001 . . 541 B.446 B.447 SFP inspection for HD 155712 with S2-W1 on UT 2007/09/17, seq 001 . . 541B.448 SFP inspection for HD 157347 with S1-E1 on UT 2008/04/25, seq 001 . . 541SFP inspection for HD 157347 with S1-W1 on UT 2008/04/13, seq 001 . . B.449 542SFP inspection for HD 157347 with S1-W1 on UT 2008/04/14, seq 001 . . B.450542SFP inspection for HD 158614 with S1-E1 on UT 2008/04/25, seq 001 . . B.451 542B.452 SFP inspection for HD 158614 with S1-E2 on UT 2007/06/01, seq 001 . . 543B.453 SFP inspection for HD 158614 with S1-W1 on UT 2008/04/14, seq 001 . 543SFP inspection for HD 158633 with S1-E1 on UT 2007/04/14, seq 001 . . B.454 543B.455 SFP inspection for HD 158633 with S1-W1 on UT 2007/04/17, seq 001 . . 544B.456 SFP inspection for HD 158633 with S1-W1 on UT 2007/04/17, seq 002 . . 544B.457 SFP inspection for HD 158633 with S1-W1 on UT 2007/04/17, seq 003 . . 544SFP inspection for HD 159062 with S1-E1 on UT 2007/04/14, seq 001 . . B.458 545B.459 SFP inspection for HD 159062 with S1-E1 on UT 2008/06/25, seq 001 . . 545B.460 SFP inspection for HD 159062 with S1-W1 on UT 2007/08/21, seq 002 . . 545B.461 SFP inspection for HD 159062 with S1-W1 on UT 2008/04/12, seq 001 . . 546B.462 SFP inspection for HD 159062 with S2-W1 on UT 2008/06/24, seq 001 . . 546SFP inspection for HD 159222 with S1-E1 on UT 2007/05/17, seq 002 . B.463 546B.464 SFP inspection for HD 159222 with S1-E1 on UT 2008/06/26, seq 001 . . 547B.465 SFP inspection for HD 159222 with S1-W1 on UT 2008/04/12, seq 001 . . 547 B.466 SFP inspection for HD 159222 with S2-W1 on UT 2008/06/24, seq 001 . . 547B.467 SFP inspection for HD 160346 with S1-E1 on UT 2007/07/22, seq 003 . . 548B.468 SFP inspection for HD 160346 with S1-W1 on UT 2007/07/22, seq 001 . . 548B.469 SFP inspection for HD 161198 with S1-E1 on UT 2008/04/25, seq 001 . . 548SFP inspection for HD 161198 with S1-W1 on UT 2007/05/30, seq 001 . . B.470 549B.471 SFP inspection for HD 161198 with S1-W1 on UT 2008/04/13, seq 001 . . 549SFP inspection for HD 161198 with S1-W1 on UT 2008/04/14, seq 001 . . B.472 549B.473 SFP inspection for HD 161198 with S2-W1 on UT 2007/09/17, seq 001 . . 550B.474 SFP inspection for HD 164922 with S1-E1 on UT 2007/05/17, seq 001 . . 550SFP inspection for HD 164922 with S1-E2 on UT 2007/05/31, seq 001 B.475 . . 550SFP inspection for HD 165401 with S1-E1 on UT 2007/07/22, seq 002 B.476 . . 551SFP inspection for HD 165401 with S1-E1 on UT 2008/04/25, seq 001 B.477 551. . SFP inspection for HD 165401 with S1-E1 on UT 2008/06/21, seq 001 B.478 551. . B.479 SFP inspection for HD 165401 with S1-E1 on UT 2008/06/23, seq 001 . . 552B.480 SFP inspection for HD 165401 with S1-E1 on UT 2008/07/05, seq 001 . . 552B.481 SFP inspection for HD 165401 with S1-E1 on UT 2008/07/05, seq 002 552. . B.482 SFP inspection for HD 165401 with S1-W1 on UT 2007/07/22, seq 001 . . 553B.483 SFP inspection for HD 165401 with S1-W1 on UT 2008/06/22, seq 001 . . 553SFP inspection for HD 165401 with S1-W1 on UT 2008/07/07, seq 001 . . B.484 553B.485 SFP inspection for HD 165401 with S1-W1 on UT 2008/07/07, seq 002 . . 554B.486 SFP inspection for HD 166620 with E1-W1 on UT 2007/04/26, seq 001. 554B.487 SFP inspection for HD 166620 with S1-E1 on UT 2007/04/14, seq 001 . . 554B.488 SFP inspection for HD 166620 with S2-E2 on UT 2007/09/18, seq 001 . . 555B.489 SFP inspection for HD 166620 with S2-W1 on UT 2007/09/17, seq 001 . . 555B.490 SFP inspection for HD 175742 with S1-E1 on UT 2008/06/21, seq 001 . . 555B.491 SFP inspection for HD 175742 with S1-E1 on UT 2008/06/23, seq 001 . . 556 B.492 SFP inspection for HD 175742 with S1-E2 on UT 2007/05/29, seq 001 . . 556B.493 SFP inspection for HD 175742 with S1-W1 on UT 2008/04/13, seq 001 . . 556 B.494 SFP inspection for HD 175742 with S1-W1 on UT 2008/04/14, seq 001 . . 557B.495 SFP inspection for HD 175742 with S2-E2 on UT 2007/09/18, seq 003 . . 557SFP inspection for HD 175742 with S2-W1 on UT 2007/09/17, seq 001 . . B.496 557B.497 SFP inspection for HD 175742 with S2-W1 on UT 2008/06/24, seq 001 . . 558SFP inspection for HD 176377 with S1-E1 on UT 2007/04/14, seq 001 . . B.498 558B.499 SFP inspection for HD 176377 with S1-E1 on UT 2008/06/21, seq 001 . . 558B.500 SFP inspection for HD 176377 with S1-E1 on UT 2008/06/23, seq 001 559. . SFP inspection for HD 176377 with S1-E1 on UT 2008/06/26, seq 001 . . B.501 559SFP inspection for HD 176377 with S1-W1 on UT 2008/04/13, seq 001 . . B.502559 B.503 SFP inspection for HD 176377 with S2-W1 on UT 2007/09/17, seq 001 . . 560B.504 SFP inspection for HD 176377 with S2-W1 on UT 2008/06/24, seq 001 . . 560B.505 SFP inspection for HD 178428 with S1-E2 on UT 2007/05/29, seq 001 . . 560SFP inspection for HD 178428 with S1-E2 on UT 2007/06/01, seq 001 B.506 . . 561B.507 SFP inspection for HD 178428 with S1-W1 on UT 2007/05/30, seq 001 . . 561B.508 SFP inspection for HD 179957 with S1-E1 on UT 2007/07/25, seq 002 . 561B.509 SFP inspection for HD 179957 with S1-W1 on UT 2007/05/30, seq 001 . . 562SFP inspection for HD 179957 with S1-W1 on UT 2007/08/18, seq 001. B.510 562B.511 SFP inspection for HD 179957 with S1-W1 on UT 2007/08/21, seq 002. 562B.512 SFP inspection for HD 179957 with S2-E1 on UT 2007/08/20, seq 001 . . 563B.513 SFP inspection for HD 179958 with S1-W1 on UT 2007/05/30, seq 002. 563B.514 SFP inspection for HD 180161 with S1-E1 on UT 2007/04/14, seq 001 . . 563B.515SFP inspection for HD 180161 with S1-E1 on UT 2007/09/16, seq 002 . . 564B.516 SFP inspection for HD 180161 with S1-W1 on UT 2007/08/21, seq 001 . . 564B.517 SFP inspection for HD 180161 with S1-W1 on UT 2007/09/16, seq 001 . . 564 B.518 SFP inspection for HD 180161 with S2-E1 on UT 2007/08/20, seq 001 . . 565B.519 SFP inspection for HD 182488 with S1-E1 on UT 2007/04/14, seq 001 . . 565B.520 SFP inspection for HD 182488 with S1-W1 on UT 2007/08/18, seq 001 . . 565B.521 SFP inspection for HD 182488 with S2-E2 on UT 2007/09/18, seq 001 . . 566 SFP inspection for HD 182488 with S2-W1 on UT 2007/09/17, seq 001 . . B.522 566 B.523 SFP inspection for HD 184385 with S1-E2 on UT 2007/05/29, seq 001 . . 566 B.524SFP inspection for HD 184385 with S1-W1 on UT 2007/05/30, seq 001 . . 567B.525 SFP inspection for HD 185144 with E1-W1 on UT 2007/04/26, seq 001. 567B.526 SFP inspection for HD 185144 with S1-E1 on UT 2007/07/28, seq 001 . . 567 B.527 SFP inspection for HD 185144 with S1-E1 on UT 2007/09/16, seq 002 . . 568SFP inspection for HD 185144 with S1-W1 on UT 2007/05/28, seq 001 . . B.528 568 B.529 SFP inspection for HD 185144 with S1-W1 on UT 2007/09/16, seq 001 . . 568B.530 SFP inspection for HD 185144 with S2-E1 on UT 2007/08/20, seq 001 . . 569B.531 SFP inspection for HD 185414 with S1-E1 on UT 2007/04/14, seq 001 . . 569B.532SFP inspection for HD 185414 with S1-W1 on UT 2007/05/28, seq 001 . . 569B.533 SFP inspection for HD 189340 with S1-E1 on UT 2008/06/21, seq 001 . . 570B.534 SFP inspection for HD 189340 with S1-E1 on UT 2008/06/23, seq 001 . . 570SFP inspection for HD 189340 with S1-E1 on UT 2008/06/23, seq 002 . B.535 570SFP inspection for HD 189340 with S1-W1 on UT 2008/06/22, seq 001 . . B.536 571B.537 SFP inspection for HD 189340 with S2-W2 on UT 2008/06/09, seq 001 . . 571B.538 SFP inspection for HD 189340 with S2-W2 on UT 2008/06/09, seq 002. 571B.539 SFP inspection for HD 189733 with S1-E1 on UT 2007/07/24, seq 001 . . 572B.540 SFP inspection for HD 189733 with S1-W1 on UT 2007/05/30, seq 001 . . 572B.541SFP inspection for HD 189733 with S2-W1 on UT 2007/09/17, seq 001 . . 572B.542 SFP inspection for HD 190067 with S1-E1 on UT 2007/07/22, seq 002 . . 573B.543 SFP inspection for HD 190067 with S1-E1 on UT 2007/07/24, seq 003 . . 573 B.544 SFP inspection for HD 190067 with S1-W1 on UT 2007/07/22, seq 001 . . 573B.545 SFP inspection for HD 190404 with S1-E1 on UT 2007/07/24, seq 001 . . 574B.546 SFP inspection for HD 190404 with S1-W1 on UT 2007/07/24, seq 002. 574B.547 SFP inspection for HD 190470 with S1-E1 on UT 2007/07/24, seq 001 . . 574SFP inspection for HD 190470 with S1-E1 on UT 2007/08/18, seq 002 . . B.548 575B.549 SFP inspection for HD 190470 with S1-W1 on UT 2007/07/24, seq 002 . . 575B.550SFP inspection for HD 190470 with S1-W1 on UT 2007/08/17, seq 001 . . 575B.551 SFP inspection for HD 190470 with S1-W1 on UT 2007/08/18, seq 001 . . 576B.552SFP inspection for HD 190771 with S1-E1 on UT 2007/04/14, seq 001 . . 576 SFP inspection for HD 190771 with S1-E1 on UT 2007/08/18, seq 002 . . B.553 576SFP inspection for HD 190771 with S1-W1 on UT 2007/08/18, seq 001 . . B.554 577B.555 SFP inspection for HD 190771 with S1-W1 on UT 2007/08/21, seq 001 . . 577 SFP inspection for HD 190771 with S2-E1 on UT 2007/08/20, seq 001 . . B.556 577B.557 SFP inspection for HD 191499 with S1-E1 on UT 2007/07/22, seq 002 . . 578B.558 SFP inspection for HD 191499 with S1-E1 on UT 2007/07/24, seq 001 . . 578B.559 SFP inspection for HD 191499 with S1-W1 on UT 2007/07/22, seq 001 . . 578B.560 SFP inspection for HD 191499 with S1-W1 on UT 2007/07/24, seq 002 . . 579SFP inspection for HD 191785 with S1-E1 on UT 2007/07/22, seq 001 . . B.561 579SFP inspection for HD 191785 with S1-E1 on UT 2007/07/24, seq 001 . . B.562 579B.563 SFP inspection for HD 191785 with S1-W1 on UT 2007/07/24, seq 002 . . 580B.564 SFP inspection for HD 192263 with S1-E1 on UT 2007/08/15, seq 002 . . 580B.565 SFP inspection for HD 192263 with S1-E1 on UT 2008/04/25, seq 001 580. . B.566 SFP inspection for HD 192263 with S1-E1 on UT 2008/06/23, seq 001 581. . B.567 SFP inspection for HD 192263 with S1-E1 on UT 2008/07/06, seq 001 581. . B.568 SFP inspection for HD 192263 with S1-E1 on UT 2008/07/06, seq 002 . . 581SFP inspection for HD 192263 with S1-W1 on UT 2008/06/22, seq 001 . . B.569582 B.570 SFP inspection for HD 192263 with S1-W1 on UT 2008/07/07, seq 001 . . 582B.571 SFP inspection for HD 192263 with S1-W1 on UT 2008/07/07, seq 002. 582B.572 SFP inspection for HD 195564 with S1-E1 on UT 2008/06/23, seq 001 . . 583B.573 SFP inspection for HD 195564 with S1-W1 on UT 2008/06/22, seq 001 . . 583SFP inspection for HD 195564 with S2-W2 on UT 2008/06/09, seq 001 . . B.574 583SFP inspection for HD 195564 with S2-W2 on UT 2008/06/09, seq 002. B.575 584SFP inspection for HD 197076 with S1-E1 on UT 2007/07/22, seq 002 . . B.576 584B.577 SFP inspection for HD 197076 with S1-E1 on UT 2007/07/24, seq 001 . . 584B.578 SFP inspection for HD 197076 with S1-W1 on UT 2007/07/22, seq 001 . . 585SFP inspection for HD 197076 with S1-W1 on UT 2007/07/24, seq 002 . . B.579 585SFP inspection for HD 198425 with S1-E1 on UT 2007/08/14, seq 001 . . B.580 585SFP inspection for HD 198425 with S1-E1 on UT 2008/07/23, seq 001 . . B.581 586B.582 SFP inspection for HD 198425 with S1-W1 on UT 2007/08/17, seq 001 . . 586B.583 SFP inspection for HD 198425 with S1-W1 on UT 2008/07/08, seq 001 . . 586B.584 SFP inspection for HD 200560 with S1-E1 on UT 2007/07/28, seq 001 . . 587B.585 SFP inspection for HD 200560 with S1-E1 on UT 2007/08/18, seq 002 . . 587 B.586 SFP inspection for HD 200560 with S1-W1 on UT 2007/08/18, seq 001 . . 587SFP inspection for HD 200560 with S2-E1 on UT 2007/08/20, seq 001 . . B.587 588SFP inspection for HD 202751 with S1-E1 on UT 2007/07/22, seq 001 . . B.588 588 B.589 SFP inspection for HD 202751 with S1-W1 on UT 2007/07/24, seq 001 . . 588 B.590 SFP inspection for HD 202751 with S2-W1 on UT 2007/09/17, seq 001 . . 589B.591 SFP inspection for HD 208038 with S1-E1 on UT 2007/07/24, seq 002 . . 589B.592 SFP inspection for HD 208038 with S1-E1 on UT 2007/08/14, seq 001 589. . B.593 SFP inspection for HD 208038 with S1-W1 on UT 2007/08/14, seq 002 . . 590B.594 SFP inspection for HD 208313 with S1-E1 on UT 2007/07/25, seq 001 . . 590B.595 SFP inspection for HD 208313 with S1-E1 on UT 2007/08/14, seq 001 . . 590 B.596 SFP inspection for HD 208313 with S1-W1 on UT 2007/08/14, seq 002 . . 591B.597 SFP inspection for HD 210277 with S1-E1 on UT 2008/06/23, seq 001 . . 591B.598 SFP inspection for HD 210277 with S1-E1 on UT 2008/07/06, seq 001 591 . . B.599 SFP inspection for HD 210277 with S1-E1 on UT 2008/07/06, seq 002 . . 592 SFP inspection for HD 210277 with S1-W1 on UT 2008/06/22, seq 001 . . B.600 592B.601 SFP inspection for HD 210277 with S1-W1 on UT 2008/07/07, seq 001 . . 592SFP inspection for HD 210277 with S1-W1 on UT 2008/07/07, seq 002 . . B.602 593B.603 SFP inspection for HD 210277 with S2-W1 on UT 2007/09/17, seq 001 . . 593B.604 SFP inspection for HD 210667 with S1-E1 on UT 2007/07/26, seq 001 . . 593SFP inspection for HD 210667 with S1-E1 on UT 2007/07/28, seq 001 . . B.605 594SFP inspection for HD 210667 with S1-W1 on UT 2007/07/26, seq 002 . . B.606 594SFP inspection for HD 211472 with S1-E1 on UT 2007/07/26, seq 001 B.607 594. . B.608 SFP inspection for HD 211472 with S1-E1 on UT 2007/07/28, seq 001 . . 595B.609 SFP inspection for HD 211472 with S1-W1 on UT 2007/07/26, seq 002. 595SFP inspection for HD 215152 with S1-E1 on UT 2008/07/06, seq 001 . . B.610 595B.611 SFP inspection for HD 215152 with S1-W1 on UT 2008/07/07, seq 001 . . 596B.612 SFP inspection for HD 215152 with S2-W1 on UT 2007/09/17, seq 001 . . 596 SFP inspection for HD 216520 with S1-E1 on UT 2007/07/26, seq 001 . . B.613 596SFP inspection for HD 216520 with S1-E1 on UT 2007/09/16, seq 002 B.614 . . 597 B.615 SFP inspection for HD 216520 with S1-W1 on UT 2007/09/16, seq 001 . . 597 B.616 SFP inspection for HD 217107 with S1-E1 on UT 2007/08/14, seq 001 . . 597B.617 SFP inspection for HD 217107 with S1-W1 on UT 2007/08/14, seq 002. 598 B.618 SFP inspection for HD 217813 with S1-E1 on UT 2007/07/22, seq 001 . . 598B.619 SFP inspection for HD 217813 with S1-E1 on UT 2007/08/14, seq 001 . . 598B.620 SFP inspection for HD 217813 with S1-W1 on UT 2007/08/14, seq 002. 599B.621 SFP inspection for HD 218868 with S1-E1 on UT 2007/07/26, seq 001 . . 599 B.622 SFP inspection for HD 218868 with S1-E1 on UT 2007/07/28, seq 002 599B.623 SFP inspection for HD 218868 with S1-E1 on UT 2008/06/25, seq 001 600 B.624 SFP inspection for HD 218868 with S1-E1 on UT 2008/07/21, seq 001 600 . . B.625 SFP inspection for HD 218868 with S1-W1 on UT 2007/07/26, seq 002. 600 SFP inspection for HD 219134 with S1-E1 on UT 2007/07/26, seq 002 B.626 601 B.627 SFP inspection for HD 219134 with S1-W1 on UT 2007/07/26, seq 003 . . 601 B.628 SFP inspection for HD 219538 with S1-E1 on UT 2007/08/14, seq 001 . . 601 B.629 SFP inspection for HD 219538 with S1-W1 on UT 2007/08/14, seq 002. 602 B.630 SFP inspection for HD 219623 with S1-E1 on UT 2007/08/18, seq 002 . . 602 SFP inspection for HD 219623 with S1-W1 on UT 2007/08/18, seq 001 . . B.631 602 SFP inspection for HD 219623 with S2-E1 on UT 2007/08/20, seq 001 B.632 . . 603 SFP inspection for HD 220140 with S1-E1 on UT 2007/09/16, seq 002 B.633 603 SFP inspection for HD 220140 with S1-W1 on UT 2007/09/16, seq 001 . . B.634 603 B.635 SFP inspection for HD 220182 with S1-E1 on UT 2007/08/18, seq 002 . . 604B.636 SFP inspection for HD 220182 with S1-W1 on UT 2007/08/18, seq 001 . . 604 B.637 SFP inspection for HD 220339 with S1-E1 on UT 2008/07/06, seq 001 . . 604B.638 SFP inspection for HD 220339 with S1-W1 on UT 2008/07/07, seq 001 . . 605B.639 SFP inspection for HD 221354 with S1-E1 on UT 2007/09/16, seq 002 . . 605 SFP inspection for HD 221354 with S1-W1 on UT 2007/09/16, seq 001 . . B.640 605 B.641 SFP inspection for HD 221354 with S2-E1 on UT 2007/08/20, seq 002 606 B.642 SFP inspection for HD 221851 with S1-E1 on UT 2007/08/18, seq 002 . . 606 B.643 SFP inspection for HD 221851 with S1-W1 on UT 2007/08/18, seq 001. 606 B.644 SFP inspection for HD 222143 with S1-E1 on UT 2007/07/27, seq 001 607 B.645 SFP inspection for HD 222143 with S1-E1 on UT 2007/08/18, seq 002 607 . . B.646 SFP inspection for HD 222143 with S1-W1 on UT 2007/08/18, seq 001 . . 607B.647 SFP inspection for HD 222143 with S2-E1 on UT 2007/08/20, seq 001 . . 608

B.648	SFP inspection for HD 222	2404 with	S1-E1 on UT $2007/09/16$, seq 003	•••	608
B.649	SFP inspection for HD 222	2404 with	S1-E1 on UT 2008/06/25, seq 001		608
B.650	SFP inspection for HD 222	2404 with	S1-W1 on UT 2007/09/16, seq 002		609
B.651	SFP inspection for HD 222	2404 with	S1-W1 on UT $2008/04/26$, seq 001		609
B.652	SFP inspection for HD 222	2404 with	S2-W1 on UT 2008/06/24, seq 001		609
B.653	SFP inspection for HD 224	465 with	S1-E1 on UT 2007/08/18, seq 002		610
B.654	SFP inspection for HD 22^4	465 with	S1-W1 on UT 2007/08/18, seq 001		610
B.655	SFP inspection for HD 22^4	1465 with	S2-E1 on UT $2007/08/20$, seq 001		610

Abbreviations and Acronyms

AO	Adaptive Optics
CHARA	Center for High Angular Resolution Astronomy
CCPS	California and Carnegie Planet Search
CIV	Calibrated Interferometric Visibility
CfA	The Harvard-Smithsonian Center for Astrophysics
CNS	Catalogs of Nearby Stars
CPM	Common Proper Motion
EB	Eclipsing Binary
FIC	Fourth Catalog of Interferometric Measurements of Binary Stars
GSU	Georgia State University
HST	Hubble Space Telescope
IMF	Initial Mass Function
SB	Spectroscopic Binary
SB1	Single-Lined Spectroscopic Binary
SB1VB	Single-Lined Spectroscopic Binary with a Visual Orbit
SB2	Double-Lined Spectroscopic Binary
SB2VB	Double-Lined Spectroscopic Binary with a Visual Orbit
$\mathbf{SB9}$	The 9th Catalogue of Spectroscopic Binary Orbits

SED	spectral Energy Distribution			
SFP	Separated Fringe Packet			
SNR	Signal-to-Noise Ratio			
SSS	SuperCOSMOS Sky Survey			
IAU	International Astronomical Union			
IDL	Interactive Data Language			
IMF	Initial Mass Function			
IR	Infrared			
LBI	Long-Baseline Interferometry			
NASA	National Aeronautic and Space Administration			
NLTT	New Luyten Catalogue of Stars With Proper Motions Larger than Two			
Tenths of an Arcsecond				
OPLE	Optical Path-Length Equalizer			
PMS	Pre-Main-Sequence			
PTI	Palomar Testbed Interferometer			
POP	Pipes of Pan, the fixed-delay offsets at the CHARA Array			
SFP	Separated Fringe Packets			
SFR	Star Formation Regions			
SMARTS	Small and Moderate Aperture Research Telescope System			
USNO	United States Naval Observatory			
WDS	Washington Double Star Catalog			

VB	Visual Binary
VBO	Visual Binary Orbit
VB6	Sixth Catalog of Orbits of Visual Binary Stars
VLMC	Very Low Mass Companion(s)
WDS	Washington Double Star Catalog
WMC	Washington Multiplicity Catalog
ZAMS	Zero-Age Main Sequence
2MASS	Two Micron All Sky Survey

Sometimes I think we're alone in the universe, and sometimes I think we're not. In either case, the idea is quite staggering. —— Arthur C. Clark

INTRODUCTION AND BACKGROUND

- 1 -

This dissertation presents the results of a comprehensive multiplicity survey of solar-type stars in the solar neighborhood. In this chapter, I discuss my motivation for undertaking this effort, outline the history of double star observations, and summarize the results of prior surveys. Chapter 2 describes my sample selection, compares it with those of significant prior efforts, and outlines the survey methods for identifying and characterizing the various types of companions. Chapter 3–6 detail the results obtained from the individual observing techniques and Chapter 7 synthesizes the results and presents the overall statistics. Finally, Chapter 8 presents an analysis of the results and discusses their implications.

1.1 Motivation

Humans have always been fascinated by the night sky. A view of the sky on a clear night, far from city lights, seems filled with countless stars and is sure to evoke a feeling of amazement and wonder. Indeed, we wonder if among what we are looking at, there might be someone who is looking back at us! This question has always fascinated me: is our Earth the only place in the Universe with life, or is the Universe, like our home planet, teeming with life? The vastness of space and time and the minuscule part of it that we occupy make it very hard for us to experimentally answer this question. Speaking of time, if we imagine the Universe to be one year old and it began with the Big Bang on January 1, the Sun and Earth formed in early September, and life was flourishing on Earth by late September. However, humans arrived very late in the game, at around 9 p.m. on December 31, and the entire human civilization has played out in just the last 30 seconds! Similarly, speaking of space, if the observable Universe was the size of the Earth, galaxies would only stretch from the endzone to midfield on a typical football field and would be separated by some 500 miles of nothingness. Stars would be invisible specs about 2 mm apart, and the Earth would be a millionth of the size of a typical bacteria! So, even though we do not have any evidence for life outside Earth, Carl Sagan's words "absence of evidence is not evidence of absence" is very relevant in this context.

Our innate curiosity does not allow us to abandon hard problems. If we cannot study something directly and in its entirety, we study it indirectly and break it up into manageable chunks. The most accepted attempt of this technique for the study of alien life is the Drake Equation (Equation (1.1) below), which takes a probabilistic approach to determining N, the number of civilizations in the Galaxy whose electromagnetic emissions are detectable.

$$N = R_* f_p n_e f_l f_i f_c L, \tag{1.1}$$

where R_* is the rate of star formation in the Galaxy, f_p is the fraction of those stars with planetary systems, n_e is the number of planets per solar system with conditions suitable for life, f_l is the fraction of suitable planets where life actually begins, f_i is the fraction of life-bearing planets on which intelligent life emerges, f_c is the fraction of civilizations that develop technologies which emit detectable signs of their existence into space, and L is the length of time for which such civilizations emit their signal into space. My thesis effort aims to make a very small contribution to one factor in Equation (1.1), namely f_p , the fraction of stars that are suitable planet hosts. This work is a survey of the habitats of solar-type stars, aiming to better understand the types of objects (planets, brown dwarfs, or other stars) that inhabit their immediate environs. Until recently, it was believed that close binary systems are not conducive environments for planet formation, but more recent observational efforts (e.g., Raghavan et al. 2006; Eggenberger et al. 2007) have shown that planets are quite common in binary systems despite selection biases, and recent theoretical work (e.g., Boss 2006) now suggests that stellar companions as close as ~ 50 AU may actually trigger the formation of gas giant planets out to ~ 20 AU by inducing clumping through gravitational instability. With two-thirds of Sun-like stars thought to live in binary or higher-order multiple systems (Duquennoy & Mayor 1991, hereafter DM91), a better understanding of any relations between stellar and substellar companions to Sun-like stars is critical in better estimating the fraction of planet-host stars, thus impacting the number of potentially habitable places in the Universe.

In addition to this motivation for better understanding life in the Universe, the study of stellar multiplicity has many other important benefits. First, knowing whether a star is single or not is important for many astronomical researches, such as for the selection of calibration stars, which typically need to be single. Second, studies of stellar multiplicity ratios contribute valuable clues about star formation and evolution. It has been a longestablished belief that the majority of stars live in systems of more than one star (e.g., Heintz 1969; Abt & Levy 1976, DM91), and more recent work suggests a dependence of binary fraction on mass (DM91, Henry & McCarthy 1990; Fischer & Marcy 1992; Mason et al. 1998a; Burgasser et al. 2003), and on age (Mason et al. 1998b). Third, a very valuable result of studying stellar multiplicity is the determination of dynamical masses of the component stars. According to the Vogt-Russell theorem, composition and mass are the most important physical parameters of stars, and binary stars remain the only direct means of obtaining accurate masses, enabling an observational basis of testing stellar structure and evolution models.

Yes, stellar multiplicity studies have many benefits, but is yet another effort in this subject warranted at this time? Do we by now not have access to robust multiplicity statistics? Indeed, DM91 represents the seminal work on the multiplicity of solar-type stars, as evidenced by over a thousand citations since publication. Science, however, represents an evolving process of learning, and there are benefits in revisiting important topics from time to time in order to ratify or revise our understanding, as warranted by the new facts available to us. My dissertation is a humble effort at a modern update to DM91, leveraging a larger sample selected using modern measures and the great volume of other multiplicity studies that are constantly evolving our knowledge of this field. Most notably, even though my observational efforts do not focus on substellar companions, a great amount of excellent work is being done in this area, enabling a modern statistical analysis to include these results.

1.2 History of Double Star Observations

The first mention of the term "double star" was made in Ptolemy's Star Catalog (2nd century AD), to identify ν_1 and ν_2 Sagittarii as $\delta \iota \pi \lambda o \nu \varsigma$ (Heintz 1978). While this pair, separated by 14' in our sky, is not physically related, its appearance as a double star is consistent with the modern definition of the term. In fact, when astronomers began telescopically studying double stars, starting with Galileo's observations of the double star Mizar in about the year 1617, which was independently measured and published by Riccioli in 1650, they presumed that these stars were *not* physical associations, but rather chance alignments of stars separated by a great distance along the line of sight. In 1767, John Mitchell first argued that the law of probabilities suggested that double stars are likely to be gravitationally bound (Aitken 1964). The earliest organized studies of double stars were initiated by Christian Mayer and Sir William Herschel in the last quarter of the eighteenth century. Mayer published the first catalog of double stars in his 1779 book, and Herschel undertook a systematic search for double stars at around the same time. Once again, these astronomers assumed that they were studying chance alignments of stars separated by large distances, and they hoped to measure differential proper motions to derive a parallax to the nearer star, as was proposed by Galileo. In any case, this began an era of careful measurements of the separations between double stars using micrometers. By 1802, Herschel, based on his own observations, and the persistent input from Mitchell, separated his observations into optical double stars (chance alignments of unbound stars) and binary stars (gravitationally bound pairs), and his publication in the following year demonstrated that orbital motion was the only reasonable choice in describing his observations for six pairs, including *Castor*.

Double star astronomy continued over the ages through stalwarts such as Sir John Herschel (William's son), Sir James South, and Sherburne W. Burnham to F. G. Wilhelm Struve, who is credited with taking this science to a new level with the publication of the *Mensu*rae Micrometricae, a fundamental catalog of double stars, in 1837. In this catalog, Struve adopted John Herschel's earlier suggestions of recording the epoch of each observation, and measuring the position angle of the pair's separation in degrees from 0 to 360, starting at north and going towards east. These conventions are followed for double star work to this day. The early work of double star astronomy continued with other legends in this field such as Otto Struve, (Wilhelm's son), W. R. Dawes, and Admiral W. H. Smythe through the mid 1800s, and by this time, methods had been developed to derive orbital elements from the observations of Visual Binaries (VB).

The modern era of double star astronomy was heralded by S. W. Burnham, who, over an active 40-year career beginning around 1870, made remarkable contributions to all of the modern developments in double star astronomy, including the discovery and observation of spectroscopic binaries, the demonstration that variable stars are eclipsing binary systems, and the application of photographic methods to the measurement of visual double stars, all while discovering 1,340 new double stars and contributing many thousands of high-quality measures (Aitken 1964). Notably, Burnham measured pairs with separations down to 0".2 and ones with large brightness differences between the components. In 1895, Robert G. Aitken started working in this field and conducted the first systematic study of double stars for the purpose of deriving multiplicity statistics. Visual double star worked flourished in the coming decades with significant contributions from Gerard Kuiper, Robert Jonckheere, Paul Couteau, and Charles Worley (H. McAlister 2008, private communication). Worley transferred the Lick "Index Catalog" to the United States Naval Observatory (USNO), creating the Washington Double Star Catalog¹ (WDS), a current online double star resource that is continually updated under the direction of Brian Mason at the USNO. Meanwhile, lunar occultation events had been used for over a hundred years to determine the angular diameters of stars (Evans 1950, 1952) and vector separations of close binaries (Innes 1901; Nather & Evans 1970; Edwards et al. 1980). Observations of double stars in the southern hemisphere were started in the late nineteenth century by Robert Innes, and continued with significant contributions by Willem van den Bos, William S. Finsen, and Richard Rossiter. Recent notable contributors to visual double stars include Willem Luyten, who systematically measured proper motions of fast-moving stars and discovered about 2,000 common proper motion pairs, Antoine Labeyrie, who introduced the technique of speckle interferometry, and Harold McAlister, who used this technique to discover almost 300 new pairs and contributed about 35,000 double star measures. Recent adaptive optic surveys have enabled the detection of doubles with a very high brightness difference between the components, enabling the detection of substellar companions such as brown dwarfs (e.g., Liu et al. 2002).

The history of double stars traced above relates to visual doubles, one in which the two components of the system are resolved and their angular separation, or a component thereof, measured. As noted, these can be optical or physical. There are however, other techniques of

¹http://ad.usno.navy.mil/wds/

identifying and characterizing binaries, and these involve studying periodic variations in the spectral lines or apparent brightnesses. The first spectroscopic binary was discovered by E. C. Pickering, with the announcement of Mizar as a double-lined spectroscopic binary (SB2) in 1889, in which the spectral lines of an apparently single star were seen as a pair of lines which moved relative to each other in a periodic manner due to the Doppler effect. Essentially, what Pickering saw was the shifting of the light from the components of the binary as they orbited each other, moving towards shorter wavelength (blue-shift) when a component moved towards us, and towards longer wavelengths (red-shift) when the component moved away from us. As the stars orbit their center-of-mass, when one star moves towards us, the other moves away, and so the characteristic pattern of movement is very telling and can be used to deduce characteristics of the stars and their orbits. Many astronomers and observatory programs have contributed to the study of spectroscopic binaries, notable among them, Roger Griffin, who pioneered the field by developing a technique for measuring stellar velocities to 1 km s^{-1} precision (Griffin 1967) and recently published his 200th paper using this technique (Griffin 2008), the CORAVEL survey (Baranne et al. 1979, DM91), and Carney & Latham (1987).

As early as 1782, John Goodricke had discovered that *Algol* (β Persei) showed periodic variations in its apparent brightness and explained this behavior as a partial eclipse by an unseen darker companion in orbit around the brighter primary. In 1889, H. C. Vogel confirmed this theory by demonstrating that the spectral lines of this star showed a periodic shift consistent with the brightness changes. Due to the relative faintness of the companion, its spectral lines were not seen, so only one set of spectral lines moved periodically towards shorter and longer wavelengths, consistent with the eclipses. Hence *Algol* was the first star to be seen as a single-lined spectroscopic binary (SB1).

Studying planets around stars other that the Sun is a relatively new field in astronomy, and the most fruitful techniques involve the detection and characterization of planets by studying minute shifts in the parent star's spectral lines as a result of its wobble around the center of mass of its solar system, or by studying the tiny drop in its brightness due to eclipses from an unseen planet. Radial velocities are now routinely being measured to a few $m s^{-1}$ (Baranne et al. 1996; Butler et al. 1996) and have enabled the detection of almost 300 planets around other stars², enabling statistical analyses of orbital and physical properties (Marcy et al. 2005; Udry & Santos 2007). On the other hand, detecting planets via eclipses or transits requires photometric precision below 1% and is recently gaining momentum, with over 50 planets detected to-date, either as a follow-up of spectroscopically identified planets (Charbonneau et al. 2000) or through photometric surveys (e.g., Udalski et al. 2002), some of which are run with robotic telescopes (Bakos et al. 2002; Pollacco et al. 2006) or from space (Moutou et al. 2008).

1.3 Previous Multiplicity Surveys

While double star astronomers have focused on the discovery and characterization of pairs since William Herschel in the late eighteenth century, it was Robert Aitken who launched

 $^{^{2}}http://exoplanet.eu/catalog.php$

the first systematic survey in 1895, aimed at deriving multiplicity statistics, vastly increasing the number of pairs discovered. Several noteworthy efforts followed, and, as expected, the quality of the results has improved over the years, enabled by better observing instruments and techniques as well as by more robust statistical analyses. Before discussing the seminal work of DM91 on the multiplicity of solar-type stars (§ 1.3.2), let us first review the efforts leading up to it.

1.3.1 Early Multiplicity Surveys

The early double star observers, from William Herschel in the late 1700s to Robert Aitken about a hundred years later, focused on discovering and cataloging thousands of visual double stars. Aitken's Double Star Catalog (Aitken & Doolittle 1932) lists over 3,000 double stars, over 30% of which have angular separations < 0?5, and over half have separations < 1", confirming the improved capabilities, and increasing the confidence in a physical relationship due to proximity. Given the nascent nature of these surveys, however, there is little discussion of orbital elements or multiplicity statistics until a few years later.

With the evidence for a physical association among double stars mounting because of the large number of close doubles discovered, attention turned toward treating them as bound systems whose orbital properties could be studied. Hertzsprung (1922) developed an empirical method of estimating the orbital period of visual binaries by deriving a relationship between the orbital period and the ratio of the radius vector to yearly orbital motion, based on his studies of 13 pairs within 10 pc of the Sun with known orbits. He also noted that of the 15,000 known double stars at that time, only 50 had reliable orbits and an additional 1,000 showed hints of orbital motion. A few years later, Luyten (1927) used matching proper motion measurements of the components of double stars to argue for a physical association. Luyten (1930a) noted that the results of Hertzsprung (1922) are incomplete and arbitrary, and developed a more rigorous statistical method for the estimation of orbital period based on Kepler's Third Law. He measured separation at one epoch, estimated masses from measured luminosities, and statistically estimated the orbital eccentricity and inclination. He also provided observational support for this method by testing it on 15 binaries with known orbits (Luyten 1930b). In perhaps the earliest statistical analysis of a volume-limited sample, Luyten (1930b) presented a tally of 47 visual doubles, 15 of which had reliable orbits, and 5 spectroscopic binaries for the 10 pc sample, which included 105 stars (Luyten & Shapley 1930). Estimating the periods of the visual doubles without orbits, Luyten (1930b) developed a period distribution of the complete sample of known binaries within 10 pc, concluding that the $\log P$ distribution was unimodal with a mean of 2.5 and a dispersion of 1.7, noting that any undetected binaries were likely to have periods larger than the derived mean.

Kuiper (1935a,b) proposed a variety of problems that could be studied with double stars, and addressed some of them, deriving a companion fraction of 80%, i.e. for the 465 primaries studied, he estimated a total of 372 companions. These results were based on an observed companion fraction of 33%, which was adjusted to account for the incompleteness of the survey for large Δ mag pairs. Further, he noted a roughly Gaussian distribution of the semimajor axis, with a peak at about 20 AU. In a follow up work, Kuiper (1942) reported raw multiplicity statistics for a sample of 254 stars with parallax ≥ 0.0095 . He presented multiplicity by spectral type and by total mass. For A-K stars, the Single:Double:Triple:Quadruple (S:D:T:Q) ratios were 44:23:5:1, which yields a companion fraction of 49%. This is significantly higher than his earlier work's results of 33%, but no incompleteness analysis was performed in the later effort. The semi-major axis distribution was confirmed to be Gaussian, but the peak had moved out to 50 AU. Kuiper noted the incompleteness of the survey and suggested that a sample selected from a larger volume of space along with the completion of spectroscopic and visual surveys would provide more reliable results. Heintz (1969) studied 100 stellar systems and presented a companion fraction of 1.0 to 1.1, i.e. the 100 systems contained 200–210 stars. Based on spectroscopic and visual binaries, his work identified 30 single, 47 binary, and 23 multiple systems, with an asymmetric distribution of semi-major axis peaked at 45 AU. However, Heintz also noted that the statistics were limited by selection effects, primarily the discovery probability and confirmation of physical pairs.

Focusing on spectroscopic binaries, Jaschek & Jaschek (1957) derived an observed binary ratio of approximately 12% for F-K stars by studying about 200 stars of varying spectral types and luminosity classes. Binaries were identified as pairs with known orbits, those noted as spectroscopic binaries in earlier efforts, or those with mean velocity differences between observatories of more than 20 km s⁻¹. This threshold was independent of spectral type or mass, and, to account for the dependence of velocity semi-amplitude on mass, they applied a correction factor and reported a corrected spectroscopic binary fraction of about 30%, noting that it was roughly constant along the main sequence, a conclusion earlier noted by Kuiper for visual binaries. Petrie (1960) studied the probable error distributions of radial velocities and estimated that $\sim 51\%$ of F-M stars showed radial velocity variations, higher than previous estimates, but once again confirmed a roughly flat distribution across the main sequence. However, Petrie's work assumed a Gaussian distribution for the observational errors of constant velocity stars, which was later showed to be incorrect (Kirillova & Pavlovskaya 1963). Jaschek & Gómez (1970) reviewed these prior efforts and presented an updated percentage of variable radial velocity stars as about 45% for F-M stars based on 350 stars, once again noting similar results across all spectral types.

Abt & Levy (1976) attempted a comprehensive systematic effort for the multiplicity statistics of solar-type stars based on a sample of 135 F3–G2 IV or V bright field stars (V < 5.5 mag) and about 20 radial velocity measurements for each star. Their results were S:D:T:Q = 42:46:9:2 with a median period of 14 years, considerably shorter than the estimates of 320 years obtained by Luyten (1930b) or 79 years by Kuiper (1935a,b). Based on an incompleteness analysis aimed at accounting for missed binaries, they derived a a companion fraction of 1.4 companions for each primary, and concluded that two-thirds of solar-type stars have stellar companions and the remaining one-third have substellar companions. However, their results have been called into question based on selection effects and the validity of the binaries reported. As analyzed by Branch (1976), their magnitude limited sample suffers from a serious selection effect favoring binaries with bright companions, because unresolved binaries are intrinsically brighter than single stars and will be counted out to a larger volume of space (the Malmquist bias). Further, Morbey & Griffin (1987) pointed out that 24 of the 25 new binaries reported by Abt & Levy are not statistically supported by their data and showed that 21 of these are likely not binaries at all. Another important conclusion of Abt & Levy (1976) was the bimodal distribution of mass-ratio, which they use as evidence of two formation mechanisms of binaries – fission for close systems and cloud fragmentation for wider systems. However, several authors have pointed out that these results are dominated by selection effects and that the true distribution of mass-ratio is unimodal with frequencies increasing towards lower mass-ratios (Scarfe 1986; Trimble 1987, 1990).

Zinnecker (1984) reviewed binary statistics and the distribution of mass-ratios, discussing the various selection effects that hampered such studies and suggesting that none of the binary formations mechanisms (fission, fragmentation, capture, and disintegration of small clusters) could be ruled out. Halbwachs (1986) studied the multiplicity statistics of the stars in the 4th edition of the Yale Catalog of Bright Stars (Hoffleit & Jaschek 1982) and its supplement (Hoffleit et al. 1983). Accounting for selection biases, he reported that the sample of 2591 dwarfs contained 52% single stars, 36% binaries, and 12% multiple systems. Based on an estimation of missed binaries, he concluded that at most 23% of stars are truly single. Halbwachs (1987) analyzed the mass-ratio distribution among spectroscopic binaries in the 7th Catalog of Spectroscopic Binaries (Batten et al. 1978), and concluded that there was no peak at $q \sim 1$ as suggested by Abt & Levy (1976) but rather a possible peak near $q \sim 0.4$. Further, he concluded that there is no difference in the mass-ratio distribution between close-period spectroscopic systems and the long-period visual binaries, suggesting that all binaries may be formed by a single mechanism.

1.3.2 The Duquennoy & Mayor Survey

In the most comprehensive and systematic treatment of the multiplicity of solar-type stars to date, DM91 studied an unbiased sample of 164 stars. In order to minimize selection effects that plagued prior efforts, they chose a volume-limited sample with trigonometric parallaxes from Gliese (1969) $\geq 0.045''$. Choosing to focus on solar-type stars, they limited spectral types to F7–G9 and luminosity classes to IV-V, V, and VI. While they included companions of all known types for statistical analyses, their primary observing program resulted in about 4200 radial velocity measurements obtained over 13 years with a precision of 0.3 km s⁻¹. The location of their observing facility at Haute-Provence Observatory imposed a final restriction on their sample, namely, declination north of -15° . In addition to the 164 primaries in their sample, they could obtain radial velocity observations for 17 of the 30 wide Common Proper Motion (CPM) companions, and included them in the analysis as well.

In all, DM91 presented 82 orbits with derivable orbital periods, 52 of which had spectroscopic orbital solutions, including 6 new orbits, 2 revisions to existing orbits, 17 preliminary orbital solutions, and 12 independent orbits corresponding to existing orbits in the Batten et al. (1989) catalog. The 82 orbits for 164 primaries implies an average of 0.5 companions per primary, significantly lower than the earlier estimates presented above. They also noted that their S:D:T:Q = 57:38:4:1 indicate much fewer multiple systems (triples or higher order) than the 25–50% of binaries found by prior studies (Mayor & Mazeh 1987; Mazeh 1990). They explained the deficiency of multiple systems as additional components that had so far been missed. Including stars without definitive orbits, but with evidence of radial velocity variations, their statistics change to 51:40:7:2. Accounting for an estimation of undetected binaries, they concluded that 57% of solar-type stars have companions with q > 0.1, and about one-third may be real single stars, i.e. with no companion above 0.01 M_{\odot}. This compares to 0% reported in Abt & Levy (1976) and a 23% upper-limit in Halbwachs (1986).

The primary results of DM91 for $M_2/M_1 > 0.1$ are: (i) The orbital period distribution is unimodal and roughly Gaussian, with a median period of 180 years. (ii) The eccentricity distribution follows three different patterns depending on the period: for periods below ~ 11 days, the orbits are circularized due to tidal interactions; for 11 < P < 1000 days, the mean eccentricity is 0.31 ± 0.04 ; and for P > 1000 days, the distribution follows f(e) = 2e. (iii) The mass-ratio distribution is independent of period and is not peaked at $q \sim 1$, but rather rises toward smaller ratios to at least $q \sim 0.3$.

For systems with $M_2/M_1 < 0.1$, DM91 used simulations to identify a binary percentage of 8% ± 6%. They also drew the following conclusions from the 11 SB they found with very low mass companions (VLMC) among CORAVEL data for the International Astronomical Union (IAU) standard stars and M dwarfs plus 7 published astrometric binaries with VLMC candidates: (i) Approximately 10% of IAU standard stars may contain brown dwarf companions, similar to the G dwarf sample. (ii) The mean eccentricity for these VLMC binaries for 11 < P < 1000 days is 0.34 ± 0.07 , similar to the G dwarf sample, implying a similar formation mechanism of stellar and brown dwarf companions, but one different from the formation of planets, given the low eccentricities of Solar System gas giant planets.

1.3.3 Multiplicity Results Since DM91

While DM91 remains the most comprehensive multiplicity survey of solar-type stars, several efforts have looked at multiplicity statistics and analyzed patterns among orbital and physical elements as well as compared the results of subsamples or disparate samples to those of DM91. A brief description of some of these follows.

Analyzing short-period systems (P < 3000 days) from DM91, Mazeh et al. (1992) showed that the mass-ratio distribution monotonically increases from q = 0 to 1. However, with only 23 systems analyzed, their results should be considered preliminary, and a flat or a decreasing function is still consistent with their data (Halbwachs et al. 2003). In any case, this distribution being considerably different than that of DM91 suggests that the mass-ratio distribution might indeed be dependent on the orbital period, as mentioned by Abt & Levy (1976). Heacox (1995) developed a statistical approach to adequately analyze SB1 pairs, and found that the distribution was roughly independent of spectral type. His reanalysis of DM91 data (Heacox 1995, 1998) showed a clear peak around $q \sim 0.2$ and the detection difficulty of lower mass-ratios might imply a rise all the way to q = 0. He also noted a second peak at q = 1, but one of limited statistical significance. Bimodal distributions were also seen in several prior analyses as noted above, but they were generally explained by selection effects alone (Halbwachs 1987). Nevertheless, as noted by Halbwachs et al. (2003), bimodal distributions were also seen in the photometric study of binaries in open clusters (Kähler 1999), but these results could be contaminated by triple systems.

Other studies seem to indicate that the DM91 work might have underestimated the true

multiplicity among solar-type stars. Studies of *Hipparcos* (Perryman & ESA 1997) double stars (more on this in §5.1) seem to indicate a significantly higher binary fraction. Quist & Lindegren (2000) modeled the reliability of *Hipparcos* double stars identifications and concluded that a companion fraction of 0.9–1.2 is implied, as compared to 0.67 of DM91. Söderhjelm (2000) studied *Hipparcos* doubles and once again concluded that the implied multiplicity is about twice as found by DM91. In an effort to recover the primordial binary fraction of intermediate-mass stars, Kouwenhoven et al. (2007) studied the nearby, young Scorpius OB2 association and combined multi-technique observational results with Monte Carlo simulations to determine a binary fraction of at least 70% for intermediate-mass stars, with a fraction of 100% providing the best fit with the data. They also stated that the log-normal period distribution of DM91 agrees with the fraction of VB but significantly underpredicts the number of SB, and a related effort (Kouwenhoven 2006) derives a binary fraction of $\sim 93\%$ for solar-type stars.

Woitas et al. (2001) studied T Tauri binary systems in four nearby star forming regions using near-infrared speckle interferometry and derived mass ratios, concluding that the mass distribution was roughly flat for q > 0.2 even though the results depend on the evolutionary model used. They do not find any relationship between mass-ratio and the primary's mass or the components' separation. Goldberg et al. (2003) studied 129 SBs with periods 1– 2500 days and concluded that the mass-ratio distribution showed a high asymmetric peak at $q \sim 0.2$ characterized by a sharp drop below 0.2 and a slower drop to a minimum at around $q \sim 0.55$, followed by an increase to a lower peak $q \sim 0.8$. An analysis of subsamples confirms a similar distribution among disk stars, but not among halo stars. The distributions also seem to depend on the primary's mass, with the lower mass ($M_p < 0.67 \,\mathrm{M_{\odot}}$) sample showing the two peaks, but the higher mass sample showing an increasing trend towards lower mass-ratios. Halbwachs et al. (2003) studied SBs with orbital periods up to 10 years and saw a mass-ratio distribution with a broad, shallow peak from $q \sim 0.2 - 0.7$ and a sharp high peak for q > 0.8. Combining this with the finding that large mass-ratio systems (twins) have lower eccentricity that other binaries at all periods, they speculated that this points to a different formation mechanism for twins. The relative abundance of twin systems was earlier seen by Tokovinin (2000) and confirmed for *Hipparcos* visual binaries by Söderhjelm (2007). However, Mazeh et al. (2003) obtained infrared spectroscopic observations of 62 disk binaries, including 43 double-lined systems, and derived a flat distribution of mass ratios from $q \sim 0.3 - 1.0$ and a possible increase at lower ratios. They suggested that earlier distributions showing the peaks might be influenced by selection effects.

Meanwhile, surveys based on specific observing techniques have contributed to our growing understanding of binaries. Mason et al. (1998a) surveyed Galactic O-stars via speckle interferometry and found that > 75% of the O-stars in clusters and associations had spectroscopic or visual companions, but the percentages were lower for field and runaway stars. This suggests that most massive stars form in binary systems which may lose companions via dynamic interactions. In a follow-up effort, Mason et al. (2009a) confirmed these findings, once again concluding that at least 75% of the O-type stars in clusters and associations are part of binary or multiple systems. Kobulnicky & Fryer (2007) compared the observed radial velocity of early-type stars in the Cygnus OB2 association with expectations from Monte-Carlo models to conclude that binary fractions are likely greater than 80% and that the companion mass distribution follows the Initial Mass Function (IMF) for only 60% of the companions, with the remainder having $q \sim 1$. At the other end of the main sequence, Fischer & Marcy (1992) find a smaller binary fraction of $42\% \pm 9\%$ among M-dwarfs, consistent with the earlier efforts of Henry & McCarthy (1990), and smaller than the DM91 results for solar type stars. Thus, multiplicity frequency appears to drop with the mass of the primary, perhaps in part due to the shrinking mass available for companions.

Mason et al. (1998b) used speckle interferometry to suggest a decreasing binary fraction from 17.9% \pm 4.6% for chromospherically active and presumably young (1 Gyr) stars to 8.5% \pm 2.7% for inactive and older (4 Gyr) stars, a result confirmed in a follow-up survey of a large volume-limited sample (B. Mason 2008, private communication). In a progress report of an adaptive optics (AO) survey looking for faint companions to solar-type stars within 25 pc, Turner et al. (2001) reported five faint companions, including three new detections. In spectroscopic surveys of high proper motion stars, Goldberg et al. (2002) presented 34 SB2 orbits and Latham et al. (2002) presented 171 SB1 orbits, noting that that binary characteristics such as frequency and period distribution are similar for the halo and the disk populations. In the first of an intended series of papers, Abt & Willmarth (2006) presented the results of a spectroscopic survey of 167 solar-type stars in the solar-neighborhood. While this work presents 39 SB1 and 12 SB2 orbits, analysis of orbital elements has been left to follow-up efforts, yet to come.

Other efforts have studied multiplicity frequencies in specific populations of stars like globular clusters or open clusters. Albrow et al. (2001) conducted a photometric survey using the *Hubble Space Telescope* (HST) in the core of the globular cluster 47 Tucanae, deriving overall percentages of $13\% \pm 6\%$ detached binaries and $14\% \pm 4\%$ W UMa-type contact binaries. In a theoretical effort, Ivanova et al. (2005) predicted that dynamical interactions in the core of clusters would rapidly deplete binary populations, estimating that a cluster starting as 100% binaries would eventually be left with a maximum of $\sim 5-10\%$ binary fraction. They revised the incompleteness analysis of Albrow et al. (2001), primarily questioning the assumption of a flat period distribution, and showed that the resulting expectations are more consistent with recent HST observations. In a recent HST effort, Sollima et al. (2007) used color-magnitude diagram morphologies to estimate the minimum percentage of binaries in 13 low-density clusters, obtaining values of around 6%, larger than estimates for dense clusters, presumably due to the lower level of dynamical interactions. Their analysis of the radial distribution of binaries indicates a concentration towards the core, and a comparison of the results for the individual clusters implies a depletion of binary frequency with age from 6–12 Gyr.

Early multiplicity studies of low-mass (roughly solar-mass and lower) in star forming regions (SFR) yielded significantly higher percentages, suggesting that pre-main-sequence (PMS) stars have a much higher fraction of binaries than nearby solar-type stars (Ghez et al. 1993, 1997; Leinert et al. 1993). Brandner & Koehler (1998) studied 114 weak-line T Tauri stars in the nearby Scorpius-Centaurus OB association to demonstrate that separate

populations exist with different separation distributions, which are more peaked than field star samples, but peak at different values correlated with the number of massive stars in the population. They invoked this finding to explain that binary fractions among T Tauri stars may not be very different compared to older field stars, and results indicating such a conclusion could have extrapolated a sharp peak out to wider ranges of separations. More recent studies have indeed confirmed similar binary fractions among T Tauri and solarneighborhood samples (e.g., Melo 2003). Moving up the evolutionary stage to open clusters, Patience et al. (2002a) conducted a high-angular resolution multiplicity survey of two open clusters and presented the following results: (a) For companions identifiable by their survey, they found a binary ratio consistent with field G-dwarfs, implying that binary ratios do not change appreciably over a few times 10^7 years; (b) The cluster binary separation peaks at $4^{+1}_{-1.5}$ AU, significantly smaller than for field or T Tauri stars, supporting the notion of sub-populations with different sharply peaked separation distributions; (c) The binary fraction in clusters increases with decreasing primary mass, and the mass-ratio distribution increases towards lower values for higher-mass stars, while it is reasonably flat for lower-mass stars; and, (d) solar-type primaries with close companions have higher rotational velocities, suggesting that companions affect the rotational evolution of young stars.

Searching for additional companions to known SBs using AO, Tokovinin et al. (2006) reported 12 new tertiary companions and several interesting conclusions for multiple systems: (a) the period distribution for SBs with and without tertiary companions is significantly different, strongly indicating that binaries without additional companions exist; (b) the
mass-ratio distributions for binary and triple systems are identical; (c) While 63% of SB pairs have tertiary companions, there is a strong dependence on period, with 96% of the close (P < 3 days) pairs having tertiary companions, suggesting an angular momentum exchange; (d) The SB primaries are more massive than the tertiary companion in 83% ± 4% of the systems. In a follow-up effort, Tokovinin (2008) studied a large sample of triple and quadruple systems, concluding that the properties of multiple stars do not correspond to dynamical decay of small clusters, but are rather consistent with a cascading rotationally driven fragmentation followed by a migration of the orbits.

Table 1.1 summarizes the multiplicity statistics of the comprehensive surveys discussed above. It is clear that, while multiplicity studies can offer important clues to star formation and evolution, the results available vary significantly based on the samples, observing techniques, and incompleteness analyses. My Ph.D. effort is able to leverage all these prior efforts, and based on a systematic approach, aims to further improve our understanding of stars through a study of their multiplicity.

TABLE 1.1: Results of Previous Multiplicity Surveys

Survey (1)	Sample Description (2)	Sample Size (3)	Observed S:D:T:Q (4)	Multiplic Observed (5)	ity Ratio ^a Corrected (6)	Comment (7)
Luyten (1930b)	10 pc sample	105		50%		
Kuiper (1935b)	33 pc sample	465		33%	80%	
Kuiper (1942)	10.5 pc sample	465	44:23:5:1	40%		1
Heintz (1969)	Large sample		30:47:23	70%		2
Jaschek & Gómez (1970)	Diverse sample	350		45%		3
Abt & Levy (1976)	Magnitude limited F3-G2 IV-V	135	42:46:9:2	58%	66%	4
Halbwachs (1986)	Large sample of bright dwarfs	2591	52:36:12	48%	77%	5
Duquennoy & Mayor (1991)	Volume-limited F7-G9 IV-VI	164	57:38:4:1	43%	67%	6
Söderhjelm (2000)	Hipparcos Double Stars			70%	100%	7

NOTES.— Column 7 abridged notes (see § 1.3 for more information): (1) Noted by the author as limited and based on incomplete surveys. The ratios reported are for A-K stars. M star and white dwarf results are not included. (2) The ratio is for Single:Double:Multiple systems. At least 1/3 of the non-single stars are estimated to be multiple (more than two components). (3) Estimated binary fraction includes only spectroscopic binaries and stars showing radial-velocity variations. (4) Recognized to suffer from selection effects favoring binaries, and most new binaries reported were

later refuted. (5) Corrected multiplicity ratio is based on the upper-limit of 23% for single stars. (6) Corrected multiplicity ratio is based on the estimated 33% single stars, i.e. stars with no companions with mass greater than 0.01 M_{\odot} . (7) The reference suggests that multiplicity implied by *Hipparcos* is twice that of DM91. Further, Kouwenhoven (2006) estimates a 93% multiplicity ratio for solar-type stars.

^a Multiplicity Ratio is the percentage of non-single stars in a sample.

1.4 Multiplicity Among Stars with Planets

As a majority of the stars are thought to have stellar companions, any attempt to comprehend the availability of habitable real estate in the Universe is incomplete without an understanding of the nature of planetary systems in binaries and multiple star systems. While the stability of planets in binary systems has long been theoretically established (e.g., Harrington 1977; Holman & Wiegert 1999; Pilat-Lohinger et al. 2003; Musielak et al. 2005; Verrier & Evans 2007), observational investigations of planets around tight binaries are very difficult. Hence, most planet-hunting programs avoid close binaries (e.g., Marcy & Butler 1996), and while a couple of nascent programs specifically tackle this problem (Konacki 2005; Muterspaugh 2005), they have yet to produce concrete results. Nevertheless, searches for stellar companions around planet-host stars have produced positive results (e.g., Patience et al. 2002b; Eggenberger et al. 2004; Udry et al. 2004; Mugrauer et al. 2005, and references therein). My 2006 publication (Raghavan et al. 2006), undertaken based on a suggestion by Todd Henry and completed with the help of my collaborators, reported on a comprehensive assessment of stellar companions to exoplanet systems, showing that even against selection biases, at least 23% of the exoplanet systems had stellar companions. As this work was performed as a part of my thesis, I have reproduced this Astrophysical Journal publication in Appendix D. While this effort remains the most comprehensive statistical assessment todate, the number of radial velocity-detected planets has since risen from 131 to 251, and additional efforts have continued to enhance our understanding of specific aspects of planets in binary systems (e.g., Bonavita & Desidera 2007; Desidera & Barbieri 2007; Eggenberger et al. 2007; Fabrycky & Tremaine 2007; Takeda et al. 2008).

Who are we? We find that we live on an insignificant planet of a humdrum star lost in a galaxy tucked away in some forgotten corner of a universe in which there are far more galaxies than people.

—— Carl Sagan

SAMPLE AND SURVEY METHODS

– 2 –

2.1 The Sample of Solar-Type Stars

The Milky Way galaxy is just one of perhaps a trillion galaxies in the universe, and this galaxy alone contains about 200 billion stars. My thesis is aimed at enhancing our understanding of the habitats of solar-type stars in our galaxy, of which there are about 30 billion. I have chosen a sample of 454 primary stars to study comprehensively, from which the results can be extrapolated to the galactic population. In order to minimize selection effects, I have chosen a volume-limited sample, consistent with the approach of DM91. The selection criteria for solar-type stars in the solar neighborhood are as follows:

Solar-type stars: The luminosity of stars in the main-sequence ranges from about 10^{-4} to 10^{+4} L_{\odot}. I limit the selection of solar-type stars to the range 10^{-1} to 10^{+1} L_{\odot}, and further restrict them to a band around the main sequence to include luminosity classes IV, V, and VI, but exclude evolved and degenerate stars. This band extends 1.5 magnitudes below a best-fit main sequence and two magnitudes above it. The wider range above the main sequence avoids discrimination against binaries. As we will see in §2.1.1, these conditions conveniently translate to a color range of 0.5 < B - V < 1.0.

Solar neighborhood: As pointed out by DM91, a volume-limited sample is free of selection effects that plague magnitude-limited samples. Accordingly, I chose a spherical volume of space around the Sun, out to a radius of 25 pc, using *Hipparcos* parallaxes as



FIGURE 2.1: Color-Magnitude diagram of stars in the *Hipparcos* catalog.

described below.

2.1.1 Sample Selection from the Hipparcos Catalog

The *Hipparcos* mission conducted unprecedented astrometry on over 100,000 stars, and the resulting catalog contains accurate information on a large volume of stars. Hence, this became the natural source for my sample selection. Figure 2.1 plots all the stars in the *Hipparcos* catalog on a color-magnitude diagram. Applying the distance selection first, I included all stars in the catalog with parallax $\pi \geq 40$ mas with uncertainty $\sigma_{\pi}/\pi \leq 0.05$, resulting in the selection of 1 231 stars. In making this selection, I leveraged Söderhjelm (1999),



FIGURE 2.2: Color-Magnitude diagram of stars within 25 pc of the Sun from the *Hipparcos* catalog.

an effort that improved the *Hipparcos* parallax for close binaries by combining *Hipparcos* and ground-based observations, resulting in the inclusion of three stars, one (HD 98230, ξ UMa) due to a high-quality parallax estimate where *Hipparcos* had none, and two (HIP 40167 and HD 219834) as a result of revised parallax estimates just above the 40 mas threshold. This effort also resulted in the exclusion of one star (HD 108799), whose parallax was moved just below the 40 mas threshold. Figure 2.2 shows the resulting subset of distance-limited stars on a color-magnitude diagram. HD 219834, included due to the Söderhjelm (1999) effort above was nevertheless later left out of the final sample because it lies too far above the main sequence, and HIP 40167, which represents a close triple system in *Hipparcos* was included through its primary, HD 68257.



FIGURE 2.3: Selection of *Hipparcos* stars within 25 pc of the Sun and within 2 magnitudes above or 1.5 magnitudes below (dotted lines) a best-fit main sequence (long dashed line). The short dashed line is the main sequence based on data from Cox (2000).

Next, I generated a recursive best-fit main-sequence curve fitted to this selection of stars, the first fit using all the distance-limited stars, and the second one excluding the outliers, i.e. all points two magnitudes or more above or below the first fit. The selection of luminosity classes IV, V, and VI was then accomplished by including stars within 1.5 magnitudes below or 2 magnitudes above this main sequence, resulting in 1177 stars (Figure 2.3). The final sample of 462 solar-type stars in the solar neighborhood was obtained by applying a color filter of $0.5 \leq B-V \leq 1.0$, as seen in Figure 2.4. The two small triangles in the figure marked by "x" do not contain any stars, confirming that the above criteria are equivalent to selecting all stars within 2.5 magnitudes, i.e. within a factor of 10 in luminosity, of the Sun. Thus, the term "solar-type" has a physical basis connected to intrinsic luminosity of stars. Nine of the 462 stars selected above are companions to other stars in the sample, yielding a final sample of 454 primaries, including the Sun. Table 2.1 lists the final sample of solar-type primaries, in ascending order of right ascension. Columns 5–8 are from the *Hipparcos* catalog, with the exception of a few updated parallaxes from Söderhjelm (1999) as described in the previous paragraph. Columns 9–10 list the parallax and its error from van Leeuwen (2007b), a recent effort that updated *Hipparcos* astrometry based on a new reduction of the raw data (more on this in § 2.1.3). Column 11 lists the spectral type from Gray et al. (2003, 2006), or when not available from these sources, from *Hipparcos*, respectively coded as G03, G06, and HIP in Column 12.



FIGURE 2.4: The volume-limited sample of solar-type stars from the *Hipparcos* catalog. The symbols for the main-sequence curves are the same as in Figure 2.3. The dotted vertical lines mark the B - V color limits of the sample, and the dotted horizontal lines mark the magnitudes corresponding to one-tenth and ten times the luminosity of the Sun. The two triangles marked by "X" do not contain any stars, confirming that the color-based selection is equivalent to a selection based on luminosity. The solid outline defines the boundary of the final sample of 462 stars, and the large filled circle inside it marks the position of the Sun.

TABLE 2.1: Volume-Limited Sample of 454 Solar-Type Stars

Ref (12)G06 HIP HIP G06 G06G06HIP G06HIP HIP HIP HIP HIP : HIP HIP HIP G06 G06 G06G06 G06 G06 **Lype** G0 F9.5V G5V K0V 38.5V 78V... Spec G5VG0V K3VGOV G7V.. (2.5V)(11)K0V K2V G5V GOV G2VG3V G5VK0V K2V G6V (mas)(10)0.650.320.800.320.620.390.482.23 0.560.580.620.590.160.530.860.110.671.850.510.270.310.51FvL07 (mas)82.17 90.4265.9767.98 73.1543.0247.0545.3446.3734.14 42.6446.5649.5357.1540.33 16.4647.93134.07 64.93 44.25 54.8757.71 $\overline{\mathbf{6}}$ ⊨ (mas) σ_{π} 3.03 1.050.750.780.531.190.850.641.081.151.400.720.980.560.950.730.860.750.62 (\mathbf{s}) 0.810.910.51Hipparcos÷ (mas)57.9080.63 72.9849.1843.4540.2516.3849.0555.5333.78 64.3890.03 45.7645.4366.9234.0441.8067.9949.3056.7642.6747.51E B - V0.6180.6870.7520.7490.6590.9400.5670.7150.8500.9370.6550.9830.6350.8900.5020.6900.7750.6550.6400.5760.5870.6500.674(9)5.883.465.986.594.235.397.922.827.366.555.805.806.077.226.846.476.975.205.578.03 5.745.07<u>6</u> \geq 004628Name 224930003443 004256 004676 000123 000870 001237 001273 001562 001835 002025 003196 003651 004391 001461 001581 002151 003765 004308 000166 004614HD (4)Name 000518 001292 001349 001499 001598 001599 003535 003583 003765 003810 000544 001803 001936 002762003093 003206 003497 001031 002021 002941 003821 D00171 HIP Sun (3) $+58\ 26\ 12.2$ $52 \ 29.2$ -12 12 34.0 $+40\ 11\ 13.8$ $-65\ 38\ 58.3$ $+01\ 47\ 07.9$ -47 33 07.2 $+05\ 16\ 50.2$ $+57 \ 48 \ 54.7$ $+29\ 01\ 17.4$ -575445.4 $-08 \ 03 \ 10.8$ $-27\ 01\ 36.4$ -03 35 34.2 $-24 \ 46 \ 02.2$ $+16\ 56\ 26.3$ -795104.3+38 13 38.6 $-77 \ 15 \ 15.3$ (J2000.0) $+27\ 04\ 56.1$ -52 39 04.1 $+21 \ 15 \ 01.7$ Continued on Next Page. Decl. 6 -6400 45 45.59 00 48 22.98 00 49 06.29 (J2000.0) $00 \ 02 \ 10.16$ 00 06 15.81 $00 \ 06 \ 36.78$ $00\ 12\ 50.25$ 00 16 12.68 $00\ 16\ 53.89$ $00 \ 18 \ 41.87$ $00 \ 20 \ 00.41$ 00 20 04.26 00 22 51.79 $00 \ 24 \ 25.93$ $00 \ 25 \ 45.07$ $00 \ 35 \ 14.88$ 00 37 20.70 00 39 21.81 00 40 49.27 00 44 39.27 00 45 04.8900 48 58.71 R.A. (1)

						Hippe	nrcos	FvI	207		
R.A.	Decl.	HIP	HD			μ.	σ_{π}	μ	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	T_{ype}	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$00 \ 49 \ 26.77$	$-23 \ 12 \ 44.9$	003850	004747	7.15	0.769	53.09	1.02	53.51	0.53	G9V	G06
$00 \ 49 \ 46.48$	$+70\ 26\ 58.1$	003876	004635	7.75	0.900	46.47	0.70	46.23	0.53	$\mathrm{K0}$	HIP
00 50 07.59	$-10 \ 38 \ 39.6$	003909	004813	5.17	0.514	64.69	1.03	63.48	0.35	F7IV-V	HIP
$00 \ 51 \ 10.85$	$-05\ 02\ 21.4$	003979	004915	6.98	0.663	45.27	0.97	46.47	0.66	G0	HIP
00 53 01.13	$-30\ 21\ 24.9$	004148	005133	7.15	0.936	71.01	0.78	70.56	0.61	K2.5V	G06
00 53 04.20	$+61\ 07\ 26.3$	004151	005015	4.80	0.540	53.85	0.60	53.35	0.33	F8V	HIP
$01 \ 08 \ 16.39$	+54 55 13.2	005336	006582	5.17	0.704	132.40	0.60	132.38	0.82	K1V	G03
$01 \ 15 \ 00.99$	$-68 \ 49 \ 08.1$	005842	007693	7.22	1.000	47.36	1.25	46.20	0.82	K2+V	G06
$01 \ 15 \ 11.12$	$-45 \ 31 \ 54.0$	005862	007570	4.97	0.571	66.43	0.64	66.16	0.24	F9V	G06
$01 \ 16 \ 29.25$	+42 56 21.9	005944	007590	6.59	0.594	42.30	0.75	43.11	0.45	G0-V	G03
$01 \ 21 \ 59.12$	$+76\ 42\ 37.0$	006379	007924	7.17	0.826	59.46	0.59	59.49	0.46	$\mathrm{K0}$	HIP
$01 \ 29 \ 04.90$	$+21 \ 43 \ 23.4$	006917	008997	7.74	0.966	43.16	0.93	42.13	0.68	K2.5V	G03
$01 \ 33 \ 15.81$	-24 10 40.7	007235	009540	6.97	0.766	51.27	0.88	52.49	0.46	G8.5V	G06
$01 \ 34 \ 33.26$	+685653.3	007339	009407	6.52	0.686	47.65	0.60	48.41	0.40	G6.5V	G03
$01 \ 35 \ 01.01$	-295437.2	007372	02700	7.11	0.909	42.29	1.47	46.24	3.07	K2V	G06
$01 \ 36 \ 47.84$	$+41 \ 24 \ 19.7$	007513	009826	4.10	0.536	74.25	0.72	74.12	0.19	F8V	HIP
$01 \ 37 \ 35.47$	$-06\ 45\ 37.5$	007576	010008	7.66	0.797	42.35	0.96	41.75	0.74	G9V	G03
$01 \ 39 \ 36.02$	+45 52 40.0	007734	010086	6.60	0.690	46.73	0.80	46.79	0.60	G5V	G03
$01 \ 39 \ 47.54$	$-56\ 11\ 47.0$	007751	010360	5.76	0.880	122.75	1.41	127.84	2.19	K2V	G06
$01 \ 41 \ 47.14$	$+42 \ 36 \ 48.1$	007918	010307	4.96	0.618	79.09	0.83	78.50	0.54	G1V	G03
$01 \ 42 \ 29.32$	$-53 \ 44 \ 27.0$	007978	010647	5.52	0.551	57.63	0.64	57.36	0.25	F9V	G06
$01 \ 42 \ 29.76$	+20 16 06.6	007981	010476	5.24	0.836	133.91	0.91	132.76	0.50	K0V	G03
$01 \ 44 \ 04.08$	-15 56 14.9	008102	010700	3.49	0.727	274.17	0.80	273.96	0.17	G8.5V	G06
$01 \ 47 \ 44.83$	+63 51 09.0	008362	010780	5.63	0.804	100.24	0.68	99.33	0.53	G9V	G03
$01 \ 59 \ 06.63$	+33 12 34.9	009269	012051	7.14	0.773	40.74	0.90	40.03	0.58	G9V	G03
Continued o	n Next Page										

						Hippe	rrcos	FvI	201		
R.A.	Decl.	HIP	HD			μ	σ_{π}	н	σ_{π}	\mathbf{Spec}	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	\mathbf{Type}	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$02 \ 06 \ 30.24$	$+24 \ 20 \ 02.4$	009829	012846	6.89	0.662	43.14	0.94	43.91	0.57	G2V-	G03
$02 \ 10 \ 25.93$	$-50\ 49\ 25.4$	010138	013445	6.12	0.812	91.63	0.61	92.74	0.32	K1V	G06
02 17 03.23	+34 13 27.2	010644	013974	4.84	0.607	92.20	0.84	92.73	0.39	G0V	HIP
$02 \ 18 \ 01.44$	$+01 \ 45 \ 28.1$	010723	014214	5.60	0.588	40.04	0.92	41.06	0.49	G0IV-	G03
$02 \ 18 \ 58.50$	-25 56 44.5	010798	014412	6.33	0.724	78.88	0.72	78.93	0.35	G8V	G06
$02 \ 22 \ 32.55$	$-23\ 48\ 58.8$	011072	014802	5.19	0.608	45.60	0.82	45.53	0.82	G0V	G06
$02 \ 36 \ 04.89$	+06 53 12.7	012114	016160	5.79	0.918	138.72	1.04	139.27	0.45	K3V	G03
$02 \ 36 \ 41.76$	$-03 \ 09 \ 22.1$	012158	016287	8.10	0.944	41.09	1.25	41.44	0.97	K2.5V	G03
$02 \ 40 \ 12.42$	$-09\ 27\ 10.3$	012444	016673	5.79	0.524	46.42	0.82	45.96	0.41	F8V	G03
$02 \ 41 \ 14.00$	$-00 \ 41 \ 44.4$	012530	016765	5.72	0.511	46.24	1.31	44.27	0.84	F7V	G03
$02 \ 42 \ 14.92$	$+40\ 11\ 38.2$	012623	016739	4.91	0.582	40.52	1.25	41.34	0.43	F9IV-V	G03
$02 \ 42 \ 33.47$	$-50\ 48\ 01.1$	012653	017051	5.40	0.561	58.00	0.55	58.25	0.22	F9V	G06
$02 \ 44 \ 11.99$	$+49 \ 13 \ 42.4$	012777	016895	4.10	0.514	89.03	0.79	89.87	0.22	F7V	HIP
$02 \ 48 \ 09.14$	$+27\ 04\ 07.1$	013081	017382	7.56	0.820	44.71	1.15	40.59	1.28	K0V	G03
02 52 32.13	$-12 \ 46 \ 11.0$	013402	017925	6.05	0.862	96.33	0.77	96.60	0.40	K1.5V	G06
02 55 39.06	$+26\ 52\ 23.6$	013642	018143	7.52	0.953	43.71	1.26	42.57	0.84	K2IV	G03
$03 \ 00 \ 02.81$	+07 44 59.1	013976	018632	7.97	0.926	42.66	1.22	41.00	1.12	K2.5V	G03
$03 \ 02 \ 26.03$	$+26\ 36\ 33.3$	014150	018803	6.62	0.696	47.25	0.89	48.45	0.47	G6V	G03
$03 \ 04 \ 09.64$	$+61 \ 42 \ 21.0$	014286	018757	6.64	0.634	43.74	0.84	41.27	0.58	G1.5V	G03
$03 \ 09 \ 04.02$	$+49 \ 36 \ 47.8$	014632	019373	4.05	0.595	94.93	0.67	94.87	0.23	F9.5V	G03
$03 \ 12 \ 04.53$	-28 59 15.4	014879	020010	3.80	0.543	70.86	0.67	70.24	0.45	F6V	G06
$03 \ 12 \ 46.44$	$-01 \ 11 \ 46.0$	014954	019994	5.07	0.575	44.69	0.75	44.29	0.28	F8.5V	G03
$03 \ 14 \ 47.23$	+08 58 50.9	015099	020165	7.83	0.861	44.96	1.09	44.15	0.83	K1V	G03
$03 \ 15 \ 06.39$	$-45 \ 39 \ 53.4$	015131	020407	6.75	0.586	41.05	0.59	41.34	0.40	G5V	G06
$03\ 18\ 12.82$	$-62 \ 30 \ 22.9$	015371	020807	5.24	0.600	82.79	0.53	83.11	0.19	G0V	G06
Continued o	n Next Page										

						Hippe	trcos	FvI	207		
R.A.	Decl.	HIP	HD			π -	σ_{π}	μ	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$03 \ 19 \ 01.89$	-02 50 35.5	015442	020619	7.05	0.655	40.52	0.98	39.65	0.74	G2V	G03
$03 \ 19 \ 21.70$	$+03\ 22\ 12.7$	015457	020630	4.84	0.681	109.18	0.78	109.41	0.27	G5Vvar	HIP
$03 \ 19 \ 55.65$	$-43 \ 04 \ 11.2$	015510	020794	4.26	0.711	165.02	0.55	165.47	0.19	G8V	G06
$03\ 21\ 54.76$	+52 19 53.4	015673	232781	9.05	0.990	44.03	1.24	42.81	1.03	K3.5V	G03
$03 \ 23 \ 35.26$	$-40\ 04\ 35.0$	015799	021175	6.90	0.840	58.53	1.04	57.40	0.67	K1V	G06
$03 \ 32 \ 55.84$	$-09\ 27\ 29.7$	016537	022049	3.72	0.881	310.75	0.85	310.94	0.16	K2V	G03
$03 \ 36 \ 52.38$	$+00\ 24\ 06.0$	016852	022484	4.29	0.575	72.89	0.78	71.62	0.54	F9V	HIP
$03 \ 40 \ 22.06$	$-03 \ 13 \ 01.1$	017147	022879	6.68	0.554	41.07	0.86	39.12	0.56	F9V	HIP
$03 \ 43 \ 55.34$	$-19\ 06\ 39.2$	017420	023356	7.10	0.927	71.17	0.91	71.69	0.67	K2.5V	G06
$03 \ 44 \ 09.17$	-38 16 54.4	017439	023484	6.99	0.870	61.63	0.67	62.39	0.52	K2V	G06
$03 \ 54 \ 28.03$	$+16\ 36\ 57.8$	018267	024496	6.81	0.719	48.36	1.02	48.95	0.70	G7V	G03
03 55 03.84	+61 10 00.5	018324	024238	7.84	0.831	46.95	0.95	47.59	0.84	K2V	G03
$03 \ 56 \ 11.52$	$+59 \ 38 \ 30.8$	018413	024409	6.53	0.698	46.74	0.96	45.49	0.62	G3V	G03
$04 \ 02 \ 36.74$	$-00\ 16\ 08.1$	018859	025457	5.38	0.516	52.00	0.75	53.10	0.32	F7V	G03
$04 \ 03 \ 15.00$	$+35 \ 16 \ 23.8$	018915	025329	8.51	0.863	54.14	1.08	54.68	0.92	K3Vp	G03
$04 \ 05 \ 20.26$	$+22 \ 00 \ 32.1$	019076	025680	5.90	0.620	59.79	0.84	59.04	0.33	G1V	G03
04 07 21.54	$-64 \ 13 \ 20.2$	019233	026491	6.37	0.636	43.12	0.50	42.32	0.28	G1V	G06
$04 \ 08 \ 36.62$	$+38\ 02\ 23.0$	019335	025998	5.52	0.520	46.87	0.77	47.63	0.26	F8V	G03
$04 \ 09 \ 35.04$	+69 32 29.0	019422	025665	7.70	0.952	54.17	0.79	53.33	0.71	K2.5V	G03
$04\ 15\ 16.32$	$-07 \ 39 \ 10.3$	019849	026965	4.43	0.820	198.24	0.84	200.62	0.23	K1V	G03
$04 \ 15 \ 28.80$	$+06\ 11\ 12.7$	019859	026923	6.32	0.570	47.20	1.08	46.88	0.47	G0IV-V	G03
$04 \ 43 \ 35.44$	$+27 \ 41 \ 14.6$	021988	029883	8.00	0.907	44.74	0.99	45.61	0.81	K5III	HIP
$04 \ 45 \ 38.58$	$-50 \ 04 \ 27.2$	022122	030501	7.58	0.875	48.90	0.64	47.93	0.45	K2V	G06
$04\ 47\ 36.29$	-165604.0	022263	030495	5.49	0.632	75.10	0.80	75.32	0.36	G1.5V	G06
$04 \ 49 \ 52.33$	$-35\ 06\ 27.5$	022451	030876	7.49	0.901	55.59	0.71	56.35	0.48	K2V	HIP
Continued o	n Next Page										

35

						Hippe	arcos	FvI	207		
$\mathbf{R.A.}$	Decl.	HIP	HD			μ	σ_{π}	π	σ_{π}	\mathbf{Spec}	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$05 \ 02 \ 17.06$	$-56\ 04\ 49.9$	023437	032778	7.02	0.636	44.94	0.58	44.48	0.36	G7V	G06
$05 \ 05 \ 30.66$	$-57\ 28\ 21.7$	023693	033262	4.71	0.526	85.83	0.46	85.87	0.18	F9V	G06
$05 \ 06 \ 42.22$	$+14\ 26\ 46.4$	023786	032850	7.74	0.804	41.70	1.14	42.24	0.92	G9V	G03
$05 \ 07 \ 27.01$	$+18 \ 38 \ 42.2$	023835	032923	4.91	0.657	63.02	0.93	64.79	0.33	G1V	G03
$05 \ 18 \ 50.47$	$-18\ 07\ 48.2$	024786	034721	5.96	0.572	40.11	0.76	39.96	0.40	F9-V	G06
$05 \ 19 \ 08.47$	$+40\ 05\ 56.6$	024813	034411	4.69	0.630	79.08	0.90	79.17	0.28	G1V	G03
$05\ 22\ 33.53$	+79 13 52.1	025110	033564	5.08	0.506	47.66	0.52	47.88	0.21	F7V	G03
$05\ 22\ 37.49$	$+02 \ 36 \ 11.5$	025119	035112	7.76	0.980	50.24	1.52	49.44	1.17	K2.5V	G03
$05\ 24\ 25.46$	$+17\ 23\ 00.7$	025278	035296	5.00	0.544	68.19	0.94	69.51	0.38	F8V	G03
$05 \ 26 \ 14.74$	$-32 \ 30 \ 17.2$	025421	035854	7.70	0.946	55.76	0.76	56.27	0.61	K3-V	G06
$05\ 27\ 39.35$	$-60\ 24\ 57.6$	025544	036435	6.99	0.755	51.10	0.52	52.08	0.45	G9V	G06
$05\ 28\ 44.83$	$-65\ 26\ 54.9$	025647	036705	6.88	0.830	66.92	0.54	65.93	0.57	K2V	G06
$05 \ 36 \ 56.85$	-47 57 52.9	026373	037572	7.95	0.845	41.90	1.74	39.82	1.36	K1.5V	G06
$05 \ 37 \ 09.89$	$-80\ 28\ 08.8$	026394	039091	5.65	0.600	54.92	0.45	54.60	0.21	G0V	G06
$05 \ 38 \ 11.86$	$+51\ 26\ 44.7$	026505	037008	7.74	0.834	48.72	1.00	49.60	0.72	K1V	G03
$05 \ 41 \ 20.34$	$+53\ 28\ 51.8$	026779	037394	6.21	0.840	81.69	0.83	81.45	0.54	$\mathrm{K}0\mathrm{V}$	G03
$05 \ 46 \ 01.89$	+37 17 04.7	027207	038230	7.34	0.833	48.60	1.03	45.76	0.76	$\mathrm{K}0\mathrm{V}$	G03
$05\ 48\ 34.94$	$-04 \ 05 \ 40.7$	027435	038858	5.97	0.639	64.25	1.19	65.89	0.41	G2V	G03
05 54 04.24	$-60 \ 01 \ 24.5$	027887	040307	7.17	0.935	77.95	0.53	76.95	0.37	K2.5V	G06
05 54 22.98	+20 16 34.2	027913	039587	4.39	0.594	115.43	1.08	115.43	0.27	G0IV-V	G03
$05 \ 54 \ 30.16$	$-19\ 42\ 15.7$	027922	039855	7.51	0.700	43.86	1.19	42.43	0.99	G8V	G06
05 58 21.54	$-04 \ 39 \ 02.4$	028267	040397	6.99	0.720	43.10	0.93	42.46	0.63	G7V	G03
$06 \ 06 \ 40.48$	$+15 \ 32 \ 31.6$	028954	041593	6.76	0.814	64.71	0.91	65.48	0.67	G9V	G03
$06 \ 10 \ 14.47$	$-74 \ 45 \ 11.0$	029271	043834	5.08	0.714	98.54	0.45	98.06	0.14	G7V	G06
$06 \ 12 \ 00.57$	$+06\ 46\ 59.1$	029432	042618	6.85	0.642	43.26	0.87	42.55	0.55	G3V	G03
Continued of	n Next Page										

						Hippo	rcos	FvI	207		
$\mathbf{R.A.}$	Decl.	HIP	HD			π	σ_{π}	μ	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$06 \ 13 \ 12.50$	$+10\ 37\ 37.7$	029525	042807	6.43	0.663	55.20	0.96	55.71	0.44	G5V	G03
$06 \ 13 \ 45.30$	-23 51 43.0	029568	043162	6.37	0.713	59.90	0.75	59.80	0.49	G6.5V	G06
$06\ 17\ 16.14$	$+05\ 06\ 00.4$	029860	043587	5.70	0.610	51.76	0.78	51.95	0.40	G0V	G03
$06\ 22\ 30.94$	$-60 \ 13 \ 07.2$	030314	045270	6.53	0.614	42.56	0.49	42.05	0.27	G0Vp	G06
$06\ 24\ 43.88$	$-28\ 46\ 48.4$	030503	045184	6.37	0.626	45.38	0.63	45.70	0.40	G1.5V	G06
$06\ 26\ 10.25$	$+18 \ 45 \ 24.8$	030630	045088	6.78	0.938	68.20	1.10	67.89	1.53	K3V	G03
$06\ 38\ 00.36$	$-61 \ 32 \ 00.2$	031711	048189	6.15	0.624	46.15	0.64	46.96	0.81	G1V	G06
$06\ 46\ 05.05$	+32 33 20.4	032423	263175	8.80	0.964	40.02	1.22	38.11	1.01	K3V	G03
$06\ 46\ 14.15$	$+79 \ 33 \ 53.3$	032439	046588	5.44	0.525	56.02	0.53	55.95	0.27	F8V	G03
$06\ 46\ 44.34$	$+43 \ 34 \ 38.7$	032480	048682	5.24	0.575	60.56	0.73	59.82	0.30	F9V	G03
06 55 18.67	$+25\ 22\ 32.5$	033277	050692	5.74	0.573	57.89	0.90	58.00	0.41	G0V	G03
06 58 11.75	+22 28 33.2	033537	051419	6.94	0.620	41.25	0.88	40.60	0.53	G5V	G03
065959.66	$-61 \ 20 \ 10.3$	033690	053143	6.81	0.786	54.33	0.54	54.57	0.34	K0IV-V	G06
$07 \ 01 \ 13.74$	-25 56 55.4	033817	052698	6.71	0.882	68.42	0.72	68.27	0.61	K1V	G06
$07 \ 01 \ 38.59$	$+48\ 22\ 43.2$	033852	051866	7.98	0.986	48.96	0.97	49.79	0.84	K3V	G03
$07 \ 03 \ 30.46$	$+29\ 20\ 13.5$	034017	052711	5.93	0.595	52.37	0.84	52.27	0.41	G0V	G03
$07 \ 03 \ 57.32$	$-43 \ 36 \ 28.9$	034065	053705	5.56	0.624	61.54	1.05	60.55	1.04	G0V	G06
$07 \ 08 \ 04.24$	+29 50 04.2	034414	053927	8.32	0.907	44.92	1.43	44.93	0.97	K2.5V	G03
$07 \ 09 \ 35.39$	$+25\ 43\ 43.1$	034567	054371	7.09	0.700	40.68	1.02	39.73	0.54	G6V	G03
$07\ 15\ 50.14$	+47 14 23.9	035136	055575	5.54	0.576	59.31	0.69	59.20	0.33	F9V	G03
$07\ 17\ 29.56$	$-46\ 58\ 45.3$	035296	057095	6.70	0.975	67.69	0.86	68.52	0.56	K2.5V	G06
$07 \ 27 \ 25.47$	$-51 \ 24 \ 09.4$	036210	059468	6.72	0.694	44.43	0.53	44.10	0.36	G6.5V	G06
$07 \ 29 \ 01.77$	+31 59 37.8	036357	:	7.73	0.923	56.98	1.24	56.63	0.93	K2.5V	G03
$07 \ 30 \ 42.51$	$-37\ 20\ 21.7$	036515	059967	6.66	0.641	45.93	0.58	45.84	0.37	G2V	G06
$07 \ 33 \ 00.58$	$+37\ 01\ 47.4$	036704	059747	7.68	0.863	50.80	1.29	50.60	0.94	K1V	G03
Continued o	n Next Page										

						Hippe	ircos	FvI	207		
R.A.	Decl.	HIP	HD			π	σ_{π}	я	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$07 \ 34 \ 26.17$	$-06\ 53\ 48.0$	036827	060491	8.16	0.900	40.32	1.26	40.73	1.00	K2.5V	G03
$07 \ 39 \ 59.33$	$-03 \ 35 \ 51.0$	037349	061606	7.18	0.891	70.44	0.94	70.37	0.64	K3-V	G03
$07 \ 45 \ 35.02$	-34 10 20.5	037853	063077	5.36	0.589	65.79	0.56	65.75	0.51	F9V	G06
$07 \ 49 \ 55.06$	$+27\ 21\ 47.4$	038228	063433	6.90	0.682	45.84	0.89	45.45	0.53	G5V	G03
$07 \ 51 \ 46.30$	-13 53 52.9	038382	064096	5.16	0.600	59.98	0.95	60.59	0.59	G0V	G06
07 54 34.18	$-01 \ 24 \ 44.1$	038625	064606	7.43	0.739	52.01	1.85	49.78	1.85	K0V	G03
07 54 54.07	$+19 \ 14 \ 10.8$	038657	064468	7.76	0.950	50.05	1.05	48.33	0.86	K2.5V	G03
$07 \ 56 \ 17.23$	$+80\ 15\ 55.9$	038784	062613	6.55	0.719	58.67	0.57	58.17	0.36	G8V	HIP
$07 \ 57 \ 46.91$	$-60 \ 18 \ 11.1$	038908	065907	5.59	0.573	61.76	0.51	61.71	0.21	F9.5V	G06
07 59 33.93	+20 50 38.0	039064	065430	7.68	0.833	43.21	0.96	42.15	0.71	K0V	G03
$08 \ 00 \ 32.13$	+29 12 44.5	039157	065583	6.97	0.716	59.52	0.77	59.64	0.56	K0V	G03
$08 \ 02 \ 31.19$	$-66\ 01\ 15.4$	039342	067199	7.18	0.872	57.88	0.58	57.76	0.41	K2V	G06
$08 \ 07 \ 45.86$	$+21 \ 34 \ 54.5$	039780	067228	5.30	0.642	42.86	0.97	42.94	0.30	G2IV	G03
08 11 38.64	+32 27 25.7	040118	068017	6.78	0.679	46.05	0.92	45.90	0.55	G3V	G03
08 12 12.73	$+17 \ 38 \ 52.0$	040167	068257	4.67	0.531	41.10^{a}	0.90^a	39.87	0.82	F8V	G03
$08 \ 18 \ 23.95$	$-12 \ 37 \ 55.8$	040693	069830	5.95	0.754	79.48	0.77	80.04	0.35	G8+V	G06
$08 \ 19 \ 19.05$	$+01 \ 20 \ 19.9$	040774	:	8.35	0.901	42.89	1.32	43.61	1.26	G5	HIP
$08 \ 27 \ 36.79$	$+45 \ 39 \ 10.8$	041484	071148	6.32	0.624	45.89	0.84	44.94	0.46	G1V	G03
$08 \ 32 \ 51.50$	$-31 \ 30 \ 03.1$	041926	072673	6.38	0.780	82.15	0.66	81.91	0.46	G9V	G06
$08 \ 34 \ 31.65$	$-00\ 43\ 33.8$	042074	072760	7.32	0.791	45.95	1.01	47.31	0.72	K0-V	G03
$08 \ 37 \ 50.29$	$-06\ 48\ 24.8$	042333	073350	6.74	0.655	42.32	1.04	41.71	0.70	G5V	G03
$08 \ 39 \ 07.90$	-22 39 42.8	042430	073752	5.05	0.720	50.20	0.98	51.55	0.63	G5IV	G06
$08 \ 39 \ 11.70$	$+65\ 01\ 15.3$	042438	072905	5.63	0.618	70.07	0.71	69.66	0.37	G1.5Vb	HIP
$08 \ 39 \ 50.79$	$+11 \ 31 \ 21.6$	042499	073667	7.61	0.832	53.98	1.04	55.13	0.71	K2V	G03
08 42 07.52	-42 55 46.0	042697	074385	8.11	0.904	44.73	0.79	43.73	0.55	K2+V	G06
Continued o:	n Next Page										

						Hippe	rrcos	FvI	207		
R.A.	Decl.	HIP	HD			μ	σ_{π}	н	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	T_{ype}	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
08 43 18.03	$-38\ 52\ 56.6$	042808	074576	6.58	0.917	89.78	0.56	89.76	0.37	K2.5V	G06
$08 \ 52 \ 16.39$	$+08 \ 03 \ 46.5$	043557	075767	6.57	0.640	41.42	1.19	41.64	1.03	G1.5V	G03
08 52 35.81	+28 19 50.9	043587	075732	5.96	0.869	79.80	0.84	81.03	0.75	V-VI0X	G03
08 54 17.95	$-05\ 26\ 04.1$	043726	076151	6.01	0.661	58.50	0.88	57.52	0.39	G3V	G06
08 58 43.93	$-16\ 07\ 57.8$	044075	076932	5.80	0.521	46.90	0.97	47.54	0.31	G2V	G06
$09 \ 08 \ 51.07$	+33 52 56.0	044897	078366	5.95	0.585	52.25	0.87	52.11	0.33	G0IV-V	G03
$09 \ 12 \ 17.55$	+14 59 45.7	045170	079096	6.49	0.731	48.83	0.92	49.11	0.54	G9V	G03
$09 \ 14 \ 20.54$	$+61\ 25\ 23.9$	045333	079028	5.18	0.605	51.12	0.72	51.10	0.32	G0IV-V	G03
$09\ 17\ 53.46$	$+28 \ 33 \ 37.9$	045617	079969	7.20	0.992	57.05	1.08	57.92	0.76	K3V	G03
$09 \ 22 \ 25.95$	$+40\ 12\ 03.8$	045963	080715	7.69	0.987	41.19	1.08	40.10	0.64	K2.5V	G03
$09 \ 30 \ 28.09$	$-32 \ 06 \ 12.2$	046626	082342	8.31	0.985	51.71	0.91	51.79	0.85	K3.5V	G06
$09 \ 32 \ 25.57$	$-11 \ 11 \ 04.7$	046816	082558	7.82	0.933	54.52	0.99	53.70	0.84	$\mathrm{K0}$	HIP
$09 \ 32 \ 43.76$	+26 59 18.7	046843	082443	7.05	0.779	56.35	0.89	56.20	0.60	G9V	G03
$09 \ 35 \ 39.50$	$+35\ 48\ 36.5$	047080	082885	5.40	0.770	89.45	0.78	87.96	0.32	G8+V	G03
$09\ 42\ 14.42$	-23 54 56.1	047592	084117	4.93	0.534	67.19	0.73	66.61	0.21	F8V	G06
$09 \ 48 \ 35.37$	$+46\ 01\ 15.6$	048113	084737	5.08	0.619	54.26	0.74	54.44	0.28	G0IV-V	G03
$10\ 01\ 00.66$	+31 55 25.2	049081	086728	5.37	0.676	67.14	0.83	66.46	0.32	G4V	G03
$10 \ 04 \ 37.66$	$-11\ 43\ 46.9$	049366	087424	8.15	0.891	43.14	1.11	41.61	0.95	K2V	G06
$10 \ 08 \ 43.14$	+34 14 32.1	049699	087883	7.56	0.965	55.37	0.94	54.93	0.54	K2.5V	G03
$10 \ 13 \ 24.73$	$-33\ 01\ 54.2$	050075	088742	6.38	0.592	43.98	0.72	43.77	0.41	G0V	G06
$10 \ 17 \ 14.54$	$+23\ 06\ 22.4$	050384	089125	5.81	0.500	44.01	0.75	43.85	0.36	F6V	G03
$10 \ 18 \ 51.95$	$+44\ 02\ 54.0$	050505	089269	6.66	0.653	48.45	0.85	49.41	0.50	G4V	G03
$10\ 23\ 55.27$	$-29 \ 38 \ 43.9$	050921	090156	6.92	0.659	45.26	0.75	44.74	0.49	G5V	G06
$10\ 28\ 03.88$	$+48\ 47\ 05.6$	051248	090508	6.42	0.610	42.45	0.77	43.65	0.43	G0V	G03
$10 \ 30 \ 37.58$	+555849.9	051459	090839	4.82	0.541	77.82	0.65	78.25	0.28	F8V	G03
Continued o:	n Next Page										

						Hippe	arcos	FvI	207		
R.A.	Decl.	HIP	HD			μ	σ_{π}	н	σ_{π}	\mathbf{Spec}	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$10 \ 31 \ 21.82$	$-53\ 42\ 55.7$	051523	091324	4.89	0.500	45.72	0.51	45.85	0.19	F9V	G06
$10 \ 35 \ 11.27$	$+84 \ 23 \ 57.6$	051819	090343	7.29	0.819	47.55	0.60	48.24	0.49	$\mathrm{K0}$	HIP
$10\ 36\ 32.38$	$-12 \ 13 \ 48.4$	051933	091889	5.71	0.528	40.67	0.68	39.88	0.37	F8V	G06
$10\ 42\ 13.32$	$-13\ 47\ 15.8$	052369	092719	6.79	0.622	42.73	0.82	41.97	0.47	G1.5V	G06
$10\ 43\ 28.27$	$-29 \ 03 \ 51.4$	052462	092945	7.72	0.873	46.36	0.84	46.73	0.69	K1.5V	G06
$10\ 56\ 30.80$	$+07\ 23\ 18.5$	053486	094765	7.37	0.920	56.98	1.03	57.79	0.87	K2.5V	G03
10 59 27.97	$+40\ 25\ 48.9$	053721	095128	5.03	0.624	71.04	0.66	71.11	0.25	G0V	HIP
$11 \ 04 \ 41.47$	$-04 \ 13 \ 15.9$	054155	096064	7.64	0.770	40.57	1.40	38.06	0.99	G8+V	G03
$11 \ 08 \ 14.01$	$+38 \ 25 \ 35.9$	054426	096612	8.35	0.942	43.91	1.03	44.29	0.84	K3-V	G03
$11 \ 12 \ 01.19$	$-26\ 08\ 12.0$	054704	097343	7.05	0.760	46.22	0.84	45.16	0.50	G8.5V	G06
$11 \ 12 \ 32.35$	$+35\ 48\ 50.7$	054745	097334	6.41	0.600	46.04	0.90	45.61	0.44	G1V	G03
$11 \ 14 \ 33.16$	$+25\ 42\ 37.4$	054906	097658	7.76	0.845	46.95	0.97	47.36	0.75	K1V	G03
$11 \ 18 \ 10.95$	$+31 \ 31 \ 45.7$	055203	098230	3.79	0.606	119.70^{a}	0.80^{a}	:		G0V	HIP
$11 \ 18 \ 22.01$	$-05\ 04\ 02.3$	055210	098281	7.29	0.732	45.48	1.00	46.36	0.64	G8V	HIP
$11\ 26\ 45.32$	$+03 \ 00 \ 47.2$	055846	099491	6.49	0.778	56.59	1.40	56.35	0.75	K0IV	HIP
$11 \ 31 \ 44.95$	$+14\ 21\ 52.2$	056242	100180	6.27	0.570	43.42	1.10	42.87	1.22	F9.5V	G03
$11 \ 34 \ 29.49$	$-32 \ 49 \ 52.8$	056452	100623	5.96	0.811	104.84	0.81	104.61	0.37	$\mathrm{K0-V}$	G06
$11 \ 38 \ 44.90$	$+45\ 06\ 30.3$	056809	101177	6.29	0.566	42.94	0.95	43.01	0.73	F9.5V	G03
$11 \ 38 \ 59.72$	+42 19 43.7	056829	101206	8.22	0.980	50.61	1.15	50.19	1.03	K5V	HIP
$11 \ 41 \ 03.02$	$+34 \ 12 \ 05.9$	056997	101501	5.31	0.723	104.81	0.72	104.04	0.26	G8V	G03
$11 \ 46 \ 31.07$	$-40 \ 30 \ 01.3$	057443	102365	4.89	0.664	108.23	0.70	108.45	0.22	G2V	G06
$11 \ 47 \ 15.81$	$-30 \ 17 \ 11.4$	057507	102438	6.48	0.681	56.26	0.77	57.23	0.41	G6V	G06
$11 \ 50 \ 41.72$	$+01 \ 45 \ 53.0$	057757	102870	3.59	0.518	91.74	0.77	91.50	0.22	F8V	HIP
$11 \ 52 \ 58.77$	+37 43 07.2	057939	103095	6.42	0.754	109.21	0.78	109.99	0.41	K1V	G03
$11 \ 59 \ 10.01$	$-20\ 21\ 13.6$	058451	104067	7.92	0.974	48.04	1.03	47.47	0.90	K3-V	G06
Continued o	n Next Page										

40

						Hippe	urcos	FvI	205		
R.A.	Decl.	HIP	HD			μ	σ_{π}	Я	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$12 \ 00 \ 44.45$	$-10\ 26\ 45.6$	058576	104304	5.54	0.760	77.48	0.80	78.35	0.31	G8IV-V	G03
$12 \ 09 \ 37.26$	$+40\ 15\ 07.4$	059280	105631	7.46	0.794	41.07	0.98	40.77	0.66	G9V	G03
$12 \ 30 \ 50.14$	$+53\ 04\ 35.8$	061053	108954	6.20	0.568	45.58	0.62	45.92	0.35	F9V	G03
$12 \ 33 \ 31.38$	$-68 \ 45 \ 20.9$	061291	109200	7.13	0.836	61.83	0.63	61.82	0.48	K1V	G06
$12 \ 33 \ 44.54$	$+41\ 21\ 26.9$	061317	109358	4.24	0.588	119.46	0.83	118.49	0.20	G0V	G03
$12 \ 41 \ 44.52$	$+55\ 43\ 28.8$	061946	110463	8.27	0.955	43.06	0.82	42.78	0.81	K3V	HIP
$12 \ 44 \ 14.55$	+51 45 33.5	062145	110833	7.01	0.936	66.40	0.78	67.20	0.66	K3V	HIP
$12 \ 44 \ 59.41$	$+39 \ 16 \ 44.1$	062207	110897	5.95	0.557	57.57	0.64	57.55	0.32	F9V	G03
$12 \ 45 \ 14.41$	$-57\ 21\ 28.8$	062229	110810	7.82	0.937	49.71	0.95	48.79	0.88	K2+V	G06
$12 \ 48 \ 32.31$	$-15\ 43\ 10.1$	062505	111312	7.93	0.946	47.19	1.93	41.96	3.00	K2.5V	G06
$12 \ 48 \ 47.05$	+24 50 24.8	062523	111395	6.29	0.703	58.23	0.99	59.06	0.45	G7V	G03
$12 \ 59 \ 01.56$	-09 50 02.7	063366	112758	7.54	0.769	47.60	0.86	47.87	0.90	K2V	G03
$12 \ 59 \ 32.78$	+41 59 12.4	063406	112914	8.60	0.940	41.36	1.48	39.71	1.00	K3-V	G03
$13 \ 03 \ 49.66$	$-05 \ 09 \ 42.5$	063742	113449	7.69	0.847	45.20	1.27	46.10	0.81	K1V	G03
$13 \ 11 \ 52.39$	+275241.5	064394	114710	4.23	0.572	109.23	0.72	109.54	0.17	G0V	HIP
$13 \ 12 \ 03.18$	$-37\ 48\ 10.9$	064408	114613	4.85	0.693	48.83	0.79	48.38	0.29	G4IV	G06
$13 \ 12 \ 43.79$	$-02 \ 15 \ 54.1$	064457	114783	7.56	0.930	48.95	1.06	48.78	0.59	K1V	G03
$13 \ 13 \ 52.23$	$-45 \ 11 \ 08.9$	064550	114853	6.93	0.643	40.87	0.84	40.95	0.56	G1.5V	G06
$13 \ 15 \ 26.45$	-87 33 38.5	064690	113283	7.11	0.710	40.52	0.56	40.70	0.38	G5V	G06
$13 \ 16 \ 46.52$	$+09\ 25\ 27.0$	064792	115383	5.19	0.585	55.71	0.85	56.95	0.26	G0Vs	HIP
$13 \ 16 \ 51.05$	$+17\ 01\ 01.9$	064797	115404	6.49	0.926	89.07	0.99	90.32	0.74	K2.5V	G03
$13 \ 18 \ 24.31$	$-18 \ 18 \ 40.3$	064924	115617	4.74	0.709	117.30	0.71	116.89	0.22	G7V	G06
$13\ 23\ 39.15$	$+02 \ 43 \ 24.0$	065352	116442	7.06	0.780	62.41	1.41	64.73	1.33	G9V	G03
$13\ 25\ 45.53$	+565813.8	065515	116956	7.29	0.804	45.76	0.72	46.31	0.51	G9V	G03
$13\ 25\ 59.86$	$+63 \ 15 \ 40.6$	065530	117043	6.50	0.739	46.86	0.55	47.24	0.31	G6V	HIP
Continued o:	n Next Page										

						Hippe	arcos	FvI	201		
R.A.	Decl.	HIP	HD			π -	σ_{π}	я	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	T_{ype}	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$13\ 28\ 25.81$	$+13 \ 46 \ 43.6$	065721	117176	4.97	0.714	55.22	0.73	55.60	0.24	G5V	HIP
$13\ 41\ 04.17$	$-34\ 27\ 51.0$	066765	118972	6.92	0.855	64.08	0.81	63.88	0.49	K0V	G06
$13\ 41\ 13.40$	$+56\ 43\ 37.8$	066781	119332	7.77	0.830	42.12	0.76	40.59	0.53	V-VI0X	HIP
$13\ 47\ 15.74$	$+17\ 27\ 24.9$	067275	120136	4.50	0.508	64.12	0.70	64.03	0.19	F7V	HIP
$13 \ 51 \ 20.33$	$-24 \ 23 \ 25.3$	067620	120690	6.43	0.703	50.20	0.85	51.35	0.45	G5+V	G06
$13 \ 51 \ 40.40$	$-57\ 26\ 08.4$	067655	120559	7.97	0.663	40.02	1.00	39.42	0.97	G7V	G06
$13 \ 52 \ 35.87$	-50 55 18.3	067742	120780	7.37	0.891	60.86	0.95	58.55	0.68	K2V	G06
$13 \ 54 \ 41.08$	$+18\ 23\ 51.8$	067927	121370	2.68	0.580	88.17	0.75	87.75	1.24	G0IV	HIP
13 55 49.99	$+14 \ 03 \ 23.4$	068030	121560	6.16	0.518	41.28	0.79	40.22	0.37	F6V	HIP
$14\ 03\ 32.35$	$+10\ 47\ 12.4$	068682	122742	6.27	0.733	60.24	0.78	58.88	0.62	G6V	G03
$14 \ 11 \ 46.17$	$-12 \ 36 \ 42.4$	069357	124106	7.93	0.865	43.35	1.40	42.76	1.22	K1V	G06
$14 \ 12 \ 45.24$	$-03 \ 19 \ 12.3$	069414	124292	7.05	0.733	44.89	1.01	45.35	0.54	G8+V	G03
$14\ 15\ 38.68$	$-45\ 00\ 02.7$	069671	124580	6.31	0.596	47.51	0.78	47.13	0.42	G0V	G06
$14 \ 16 \ 00.87$	$-06\ 00\ 02.0$	069701	124850	4.07	0.511	46.74	0.87	44.97	0.19	F7V	HIP
$14 \ 19 \ 00.90$	$-25\ 48\ 55.5$	069965	125276	5.87	0.518	56.23	0.94	55.45	0.82	F9V	G06
$14 \ 19 \ 34.86$	$-05 \ 09 \ 04.3$	070016	125455	7.58	0.867	48.12	1.11	47.89	0.81	K1V	HIP
$14\ 23\ 15.28$	$+01 \ 14 \ 29.6$	070319	126053	6.25	0.639	56.82	1.04	58.17	0.53	G1.5V	G03
$14\ 29\ 22.30$	$+80\ 48\ 35.5$	070857	128642	6.88	0.774	51.04	0.58	50.27	0.48	G5	HIP
$14\ 29\ 36.81$	+41 47 45.3	070873	127334	6.36	0.702	42.43	0.59	42.12	0.38	G5V	G03
$14 \ 33 \ 28.87$	+52 54 31.6	071181	128165	7.24	0.997	74.50	0.69	75.65	0.42	K3V	HIP
$14 \ 36 \ 00.56$	$+09 \ 44 \ 47.5$	071395	128311	7.48	0.973	60.35	0.99	60.60	0.83	K3-V	G03
$14 \ 39 \ 36.50$	-605002.3	071683	128620	-0.01	0.710	742.12	1.40	754.81	4.11	G2V	G06
$14 \ 40 \ 31.11$	$-16 \ 12 \ 33.4$	071743	128987	7.24	0.710	42.43	0.97	42.23	0.54	G8V	G06
$14 \ 41 \ 52.46$	$-75\ 08\ 22.1$	071855	128400	6.73	0.707	49.15	0.64	50.01	0.43	G5V	G06
$14 \ 45 \ 24.18$	+13 50 46.7	072146	130004	7.87	0.931	51.20	0.98	52.92	0.82	K2.5V	G03
Continued of	n Next Page										

						Hippe	rrcos	FvI	207		
$\mathbf{R.A.}$	Decl.	HIP	HD			μ.	σ_{π}	μ	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	${ m Type}$	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$14\ 47\ 16.10$	$+02 \ 42 \ 11.6$	072312	130307	7.76	0.893	50.84	1.04	51.62	0.79	K2.5V	G03
$14 \ 49 \ 23.72$	$-67 \ 14 \ 09.5$	072493	130042	7.26	0.836	41.69	1.24	40.11	0.89	K1V	G06
$14 \ 50 \ 15.81$	+23 54 42.6	072567	130948	5.86	0.576	55.73	0.80	55.03	0.34	G2V	HIP
$14 \ 51 \ 23.38$	$+19\ 06\ 01.7$	072659	131156	4.54	0.720	149.26	0.76	148.98	0.48	G7V	G03
$14 \ 53 \ 23.77$	$+19 \ 09 \ 10.1$	072848	131511	6.00	0.841	86.69	0.81	86.88	0.46	K0V	G03
$14 \ 53 \ 41.57$	$+23 \ 20 \ 42.6$	072875	131582	8.65	0.934	43.66	1.20	42.47	1.12	K3V	HIP
$14 \ 55 \ 11.04$	$+53 \ 40 \ 49.2$	073005	132142	7.77	0.785	41.83	0.63	42.76	0.45	K1V	HIP
$14\ 56\ 23.04$	$+49 \ 37 \ 42.4$	073100	132254	5.63	0.533	40.25	0.54	39.83	0.26	F8-V	G03
$14 \ 58 \ 08.80$	$-48 \ 51 \ 46.8$	073241	131923	6.34	0.708	40.79	0.86	41.93	0.83	G4V	G06
$15 \ 03 \ 47.30$	$+47 \ 39 \ 14.6$	073695	133640	4.83	0.647	78.39	1.03	79.95	1.56	G2V	HIP
$15 \ 10 \ 44.74$	$-61 \ 25 \ 20.3$	074273	134060	6.29	0.623	41.41	0.77	41.32	0.45	G0V	G06
$15 \ 13 \ 50.89$	$-01\ 21\ 05.0$	074537	135204	6.58	0.763	57.80	0.85	56.59	0.49	G9V	G03
$15 \ 15 \ 59.17$	$+00\ 47\ 46.9$	074702	135599	6.92	0.830	64.19	0.97	63.11	0.70	K0V	G03
$15 \ 19 \ 18.80$	$+01 \ 45 \ 55.5$	074975	136202	5.04	0.540	40.46	0.81	39.40	0.29	F8III-IV	HIP
$15\ 21\ 48.15$	$-48 \ 19 \ 03.5$	075181	136352	5.65	0.639	68.70	0.79	67.51	0.39	G2-V	G06
$15\ 22\ 36.69$	$-10 \ 39 \ 40.0$	075253	136713	7.97	0.970	45.83	1.41	45.22	1.12	K3IV-V	G03
$15\ 22\ 46.83$	+18 55 08.3	075277	136923	7.16	0.804	49.67	0.92	51.01	0.64	G9V	G03
$15\ 23\ 12.31$	$+30\ 17\ 16.1$	075312	137107	4.99	0.577	53.70	1.24	55.98	0.78	G2V	HIP
$15\ 28\ 09.61$	$-09 \ 20 \ 53.1$	075718	137763	6.89	0.788	50.34	1.11	48.58	1.33	G9V	G03
$15\ 29\ 11.18$	$+80\ 26\ 55.0$	075809	139777	6.57	0.665	45.32	0.57	45.77	0.37	G1.5V(n)	G03
$15\ 36\ 02.22$	$+39\ 48\ 08.9$	076382	139341	6.78	0.906	45.85	0.79	44.83	0.60	K1V	G03
$15 \ 44 \ 01.82$	$+02 \ 30 \ 54.6$	077052	140538	5.86	0.684	68.16	0.87	68.22	0.66	G5V	HIP
$15\ 46\ 26.61$	$+07\ 21\ 11.1$	077257	141004	4.42	0.604	85.08	0.80	82.48	0.32	G0IV-V	G03
$15\ 47\ 29.10$	-375458.7	077358	140901	6.01	0.715	65.60	0.77	65.13	0.40	G7IV-V	G06
$15\ 48\ 09.46$	$+01 \ 34 \ 18.3$	077408	141272	7.44	0.801	46.84	1.05	46.97	0.80	G9V	G03
Continued o:	n Next Page										

						Hippe	rrcos	FvI	207		
R.A.	Decl.	HIP	HD			μ	σ_{π}	я	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
15 52 40.54	+42 27 05.5	047760	142373	4.60	0.563	63.08	0.54	62.92	0.21	G0V	G03
$15 \ 53 \ 12.10$	+13 11 47.8	077801	142267	6.07	0.598	57.27	0.88	57.64	0.54	G0IV	HIP
$16\ 01\ 02.66$	$+33 \ 18 \ 12.6$	078459	143761	5.39	0.612	57.38	0.71	58.02	0.28	G0V	G03
$16\ 01\ 53.35$	$+58 \ 33 \ 54.9$	078527	144284	4.01	0.528	47.79	0.54	47.54	0.12	F8IV-V	HIP
$16\ 04\ 03.71$	$+25 \ 15 \ 17.4$	078709	144287	7.10	0.771	46.56	0.89	45.01	0.79	G8+V	G03
$16\ 04\ 56.79$	$+39 \ 09 \ 23.4$	078775	144579	6.66	0.734	69.61	0.57	68.87	0.33	K0V	G03
$16\ 06\ 29.60$	$+38 \ 37 \ 56.1$	078913	144872	8.58	0.963	42.57	0.86	42.55	0.77	K3V	G03
$16 \ 09 \ 42.79$	$-56\ 26\ 42.5$	079190	144628	7.11	0.856	69.66	0.90	68.17	0.64	K1V	G06
$16 \ 10 \ 24.31$	+43 49 03.5	079248	145675	6.61	0.877	55.11	0.59	56.91	0.34	V-VI0X	G03
$16 \ 13 \ 18.45$	$+13\ 31\ 36.9$	079492	145958	6.68	0.764	41.05	1.58	42.40	1.12	G9V	G03
$16 \ 13 \ 48.56$	$-57 \ 34 \ 13.8$	079537	145417	7.53	0.815	72.75	0.82	72.01	0.68	K3V	G06
$16 \ 14 \ 11.93$	$-31 \ 39 \ 49.1$	079578	145825	6.55	0.646	45.73	0.95	46.40	0.62	G2V	G06
$16 \ 14 \ 40.85$	+33 51 31.0	079607	146361	5.23	0.599	46.11	0.98	47.44	1.22	G1IV-V	G03
$16 \ 15 \ 37.27$	$-08\ 22\ 10.0$	079672	146233	5.49	0.652	71.30	0.89	71.94	0.37	G2V	G03
$16\ 24\ 01.29$	-39 11 34.7	080337	147513	5.37	0.625	77.69	0.86	78.26	0.37	G1V	G06
$16\ 24\ 19.81$	$-13 \ 38 \ 30.0$	080366	147776	8.40	0.950	46.44	1.20	46.46	1.06	K3-V	G06
$16\ 28\ 28.14$	$-70\ 05\ 03.8$	080686	147584	4.90	0.555	82.61	0.57	82.53	0.52	F9V	G06
$16\ 28\ 52.67$	$+18\ 24\ 50.6$	080725	148653	6.98	0.848	51.20	1.49	50.87	0.80	K2V	G03
$16\ 31\ 30.03$	-39 00 44.2	080925	148704	7.24	0.858	40.60	1.75	40.77	2.01	K1V	G06
$16\ 36\ 21.45$	$-02 \ 19 \ 28.5$	081300	149661	5.77	0.827	102.27	0.85	102.55	0.40	K0V	G03
$16\ 37\ 08.43$	$+00\ 15\ 15.6$	081375	149806	7.09	0.828	49.63	0.92	49.18	0.62	K0V	G03
$16 \ 39 \ 04.14$	$-58\ 15\ 29.5$	081520	149612	7.01	0.616	46.13	0.91	44.54	0.54	G5V	G06
$16\ 42\ 38.58$	$+68\ 06\ 07.8$	081813	151541	7.56	0.769	41.15	0.57	39.97	0.45	K1V	HIP
$16\ 52\ 58.80$	$-00\ 01\ 35.1$	082588	152391	6.65	0.749	59.04	0.87	57.97	0.66	G8+V	G03
$16\ 57\ 53.18$	$+47 \ 22 \ 00.1$	083020	153557	7.76	0.980	55.71	1.21	54.63	0.61	K3V	G03
Continued o:	n Next Page										

						Hippe	rrcos	FvI	207		
R.A.	Decl.	HIP	HD			μ	σ_{π}	μ	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$17\ 02\ 36.40$	+47 04 54.8	083389	154345	6.76	0.728	55.37	0.55	53.80	0.32	G8V	HIP
$17 \ 04 \ 27.84$	$-28 \ 34 \ 57.6$	083541	154088	6.59	0.814	55.31	0.89	56.06	0.50	V-VI0X	G06
$17\ 05\ 16.82$	$+00\ 42\ 09.2$	083601	154417	6.00	0.578	49.06	0.89	48.39	0.40	F9V	G03
$17 \ 10 \ 10.35$	$-60\ 43\ 43.6$	083990	154577	7.38	0.889	73.07	0.91	73.41	0.70	K2.5V	G06
$17 \ 12 \ 37.62$	$+18\ 21\ 04.3$	084195	155712	7.95	0.941	48.69	1.03	47.70	0.93	K2.5V	G03
$17 \ 15 \ 20.98$	$-26\ 36\ 10.2$	084405	155885	4.33	0.855	167.08	1.07	168.54	0.54	K1.5V	G06
$17 \ 19 \ 03.83$	$-46\ 38\ 10.4$	084720	156274	5.47	0.764	113.81	1.36	113.61	0.69	M0V	HIP
$17\ 20\ 39.57$	+32 28 03.9	084862	157214	5.38	0.619	69.48	0.56	69.80	0.25	G0V	HIP
$17\ 22\ 51.29$	$-02 \ 23 \ 17.4$	085042	157347	6.28	0.680	51.39	0.85	51.22	0.40	G3V	G03
$17\ 25\ 00.10$	+67 18 24.1	085235	158633	6.44	0.759	78.14	0.51	78.11	0.30	K0V	HIP
$17 \ 30 \ 16.43$	$+47\ 24\ 07.9$	085653	159062	7.22	0.737	44.77	0.59	44.91	0.50	G9V	G03
$17 \ 30 \ 23.80$	$-01 \ 03 \ 46.5$	085667	158614	5.31	0.715	60.80	1.42	61.19	0.68	G8IV-V	HIP
$17 \ 32 \ 00.99$	$+34 \ 16 \ 16.1$	085810	159222	6.52	0.639	42.20	0.56	41.81	0.35	G1V	G03
$17 \ 34 \ 59.59$	+61 52 28.4	086036	160269	5.23	0.602	70.98	0.55	70.47	0.37	G0V	HIP
$17 \ 39 \ 16.92$	+03 33 18.9	086400	160346	6.53	0.959	93.36	1.25	90.91	0.67	K2.5V	G03
$17\ 41\ 58.10$	+72 09 24.9	086620	162004	5.81	0.530	44.80	1.94	43.36	0.51	G0V	HIP
$17 \ 43 \ 15.64$	$+21 \ 36 \ 33.1$	086722	161198	7.51	0.752	42.45	0.98	44.15	0.89	G9V	G03
$17 \ 44 \ 08.70$	-51 50 02.6	086796	160691	5.12	0.694	65.46	0.80	64.47	0.31	G3IV-V	G06
$17 \ 46 \ 27.53$	+27 43 14.4	086974	161797	3.42	0.750	119.05	0.62	120.33	0.16	G5IV	HIP
17 53 29.94	$+21 \ 19 \ 31.1$	087579	:	8.50	0.940	40.22	1.04	41.06	1.04	K2.5V	G03
$18\ 02\ 30.86$	$+26\ 18\ 46.8$	088348	164922	7.01	0.799	45.61	0.71	45.21	0.54	G9V	G03
$18 \ 05 \ 27.29$	$+02 \ 30 \ 00.4$	088601	165341	4.03	0.860	196.62	1.38	196.72	0.83	K0-V	G03
$18 \ 05 \ 37.46$	$+04 \ 39 \ 25.8$	088622	165401	6.80	0.610	41.00	0.88	41.82	0.59	G0V	G03
$18 \ 06 \ 23.72$	$-36\ 01\ 11.2$	088694	165185	5.94	0.615	57.58	0.77	56.97	0.48	G0V	G06
$18 \ 07 \ 01.54$	$+30 \ 33 \ 43.7$	088745	165908	5.05	0.528	63.88	0.55	63.93	0.34	F7V	HIP
Continued o	n Next Page										

						Hippe	rcos	FvI	207		
R.A.	Decl.	HIP	HD			μ	σ_{π}	μ	σ_{π}	\mathbf{Spec}	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$18 \ 09 \ 37.42$	$+38\ 27\ 28.0$	088972	166620	6.38	0.876	90.11	0.54	90.71	0.30	K2V	G03
$18 \ 10 \ 26.16$	$-62 \ 00 \ 07.9$	089042	165499	5.47	0.592	56.32	0.68	56.78	0.52	G0V	G06
$18 \ 15 \ 32.46$	+45 12 33.5	089474	168009	6.30	0.641	44.08	0.51	43.82	0.29	G1V	G03
$18 \ 19 \ 40.13$	-63 53 11.6	089805	167425	6.17	0.584	43.64	0.72	43.39	0.39	F9.5V	G06
$18 \ 31 \ 18.96$	-18 54 31.7	067000	170657	6.81	0.861	75.71	0.89	75.46	0.70	K2V	G06
$18 \ 38 \ 53.40$	$-21 \ 03 \ 06.7$	091438	172051	5.85	0.673	77.02	0.85	76.43	0.47	G6V	G06
$18 \ 40 \ 54.88$	$+31 \ 31 \ 59.1$	091605	• • •	8.54	0.865	41.88	1.59	42.48	1.11	K2.5V	G03
18 55 18.80	$-37\ 29\ 54.1$	092858	175073	7.98	0.857	41.84	1.19	41.31	0.98	K1V	G06
18 55 53.22	+23 33 23.9	092919	175742	8.16	0.910	46.64	1.03	46.74	0.85	K0V	HIP
$18 \ 57 \ 01.61$	+325404.6	093017	176051	5.20	0.594	66.76	0.54	67.24	0.37	G0V	HIP
$18 \ 58 \ 51.00$	$+30\ 10\ 50.3$	093185	176377	6.80	0.606	42.68	0.64	41.94	0.47	G1V	G03
$19\ 06\ 25.11$	-37 03 48.4	093825	177474	4.23	0.523	55.89	1.94	57.79	0.75	F8V	G06
$19\ 06\ 52.46$	$-37\ 48\ 38.4$	093858	177565	6.15	0.705	58.24	0.91	58.98	0.47	G6V	G06
$19\ 07\ 57.32$	$+16\ 51\ 12.2$	093966	178428	6.08	0.705	47.72	0.77	46.66	0.48	G5IV-V	G03
$19\ 12\ 05.03$	+49 51 20.7	094336	179957	5.85	0.666	40.16	0.83	40.90	0.58	G3V	G03
$19\ 12\ 11.36$	$+57 \ 40 \ 19.1$	094346	180161	7.04	0.804	50.00	0.54	49.96	0.32	G8V	HIP
$19\ 21\ 29.76$	-34 59 00.6	095149	181321	6.48	0.628	47.95	1.28	53.10	1.41	G1V	G06
$19\ 23\ 34.01$	+33 13 19.1	095319	182488	6.37	0.804	64.54	0.60	63.45	0.35	G9+V	G03
$19\ 24\ 58.20$	+115639.9	095447	182572	5.17	0.761	66.01	0.77	65.89	0.26	G8IVvar	HIP
$19 \ 31 \ 07.97$	$+58 \ 35 \ 09.6$	095995	184467	6.60	0.859	59.84	0.64	58.96	0.65	K2V	G03
$19 \ 32 \ 06.70$	$-11 \ 16 \ 29.8$	096085	183870	7.53	0.922	55.50	0.90	56.73	0.72	K2.5V	G06
$19 \ 32 \ 21.59$	$+69 \ 39 \ 40.2$	096100	185144	4.67	0.786	173.41	0.46	173.77	0.18	G9V	G03
$19 \ 33 \ 25.55$	+21 50 25.2	096183	184385	6.89	0.745	49.61	0.94	48.64	0.63	G8V	G03
$19 \ 35 \ 55.61$	+565902.0	096395	185414	6.73	0.636	41.24	0.49	41.48	0.30	GO	HIP
$19\ 41\ 48.95$	$+50\ 31\ 30.2$	096895	186408	5.99	0.643	46.25	0.50	47.44	0.27	G1.5V	G03
Continued o	n Next Page										

46

						Hippe	arcos	FvI	207		
R.A.	Decl.	HIP	HD			π -	σ_{π}	н	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$19\ 45\ 33.53$	$+33 \ 36 \ 07.2$	097222	186858	7.68	1.000	49.09	1.43	47.34	0.82	K3+V	G03
$19 \ 51 \ 01.64$	$+10\ 24\ 56.6$	097675	187691	5.12	0.563	51.57	0.77	52.11	0.29	F8V	HIP
$19 \ 59 \ 47.34$	-09 57 29.7	098416	189340	5.87	0.598	40.75	1.35	45.04	0.99	F9V	G03
$20 \ 00 \ 43.71$	+22 42 39.1	098505	189733	7.67	0.932	51.94	0.87	51.41	0.69	K2V	G03
$20\ 02\ 34.16$	$+15 \ 35 \ 31.5$	779860	190067	7.15	0.714	51.71	0.83	52.71	0.65	$\mathrm{K}0\mathrm{V}$	G03
$20 \ 03 \ 37.41$	+29 53 48.5	098767	190360	5.73	0.749	62.92	0.62	63.06	0.34	G7IV-V	G03
$20 \ 03 \ 52.13$	$+23 \ 20 \ 26.5$	098792	190404	7.28	0.815	64.17	0.85	63.43	0.57	K1V	G03
$20 \ 04 \ 06.22$	$+17\ 04\ 12.6$	098819	190406	5.80	0.600	56.60	0.76	56.28	0.35	G0V	G03
$20 \ 04 \ 10.05$	$+25\ 47\ 24.8$	098828	190470	7.82	0.924	46.28	0.91	45.56	0.77	K2.5V	G03
$20 \ 05 \ 09.78$	$+38 \ 28 \ 42.4$	098921	190771	6.18	0.654	52.99	0.55	53.22	0.36	G2V	G03
$20 \ 05 \ 32.76$	$-67 \ 19 \ 15.2$	098959	189567	6.07	0.648	56.45	0.74	56.41	0.44	G2V	G06
$20\ 07\ 35.09$	$-55\ 00\ 57.6$	099137	190422	6.26	0.530	43.08	0.79	42.68	0.45	F9V	G06
$20\ 08\ 43.61$	$-66\ 10\ 55.4$	099240	190248	3.55	0.751	163.73	0.65	163.71	0.17	G8IV	G06
$20 \ 09 \ 34.30$	$+16\ 48\ 20.8$	099316	191499	7.56	0.810	41.07	1.18	42.26	0.99	G9V	G03
$20\ 11\ 06.07$	$+16 \ 11 \ 16.8$	099452	191785	7.34	0.830	48.83	0.91	49.04	0.65	K0V	G03
$20 \ 11 \ 11.94$	$-36\ 06\ 04.4$	099461	191408	5.32	0.868	165.24	0.90	166.25	0.27	K2.5V	G06
$20\ 13\ 59.85$	-005200.8	099711	192263	7.79	0.938	50.27	1.13	51.77	0.78	K2.5V	G03
$20 \ 15 \ 17.39$	$-27\ 01\ 58.7$	099825	192310	5.73	0.878	113.33	0.89	112.22	0.30	K2+V	G06
$20\ 17\ 31.33$	+665113.3	100017	193664	5.91	0.602	56.92	0.52	56.92	0.24	G0V	G03
$20\ 27\ 44.24$	-305204.2	100925	194640	6.61	0.724	51.50	0.82	51.22	0.54	G8V	G06
$20 \ 32 \ 23.70$	-09 51 12.2	101345	195564	5.66	0.689	41.26	0.87	40.98	0.33	G2V	G03
$20 \ 32 \ 51.64$	+41 53 54.5	101382	195987	7.08	0.796	44.99	0.64	45.35	0.43	G9V	G03
$20 \ 40 \ 02.64$	$-60 \ 32 \ 56.0$	101983	196378	5.11	0.544	41.33	0.73	40.55	0.27	G0V	G06
$20\ 40\ 11.76$	$-23\ 46\ 25.9$	101997	196761	6.36	0.719	68.28	0.82	69.53	0.40	G8V	G06
20 40 45.14	+195607.9	102040	197076	6.43	0.611	47.65	0.76	47.74	0.48	G1V	G03
Continued o:	n Next Page										

						Hippe	rrcos	FvI	207		
R.A.	Decl.	HIP	HD			μ	σ_{π}	я	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$20\ 43\ 16.00$	$-29\ 25\ 26.1$	102264	197214	6.95	0.671	44.57	0.87	44.83	0.91	G6V	G06
$20\ 49\ 16.23$	+32 17 05.2	102766	198425	8.25	0.939	42.23	0.98	41.58	0.83	K2.5V	G03
$20\ 56\ 47.33$	$-26\ 17\ 47.0$	103389	199260	5.70	0.507	47.61	0.95	45.52	0.38	F6V	G06
$20 \ 57 \ 40.07$	$-44\ 07\ 45.7$	103458	199288	6.52	0.587	46.26	0.81	45.17	0.46	G2V	G06
$21 \ 02 \ 40.76$	+45 53 05.2	103859	200560	7.69	0.970	51.65	0.72	51.36	0.63	K2.5V	G03
$21 \ 07 \ 10.38$	-13 55 22.6	104239	200968	7.12	0.901	56.67	1.18	56.90	0.60	G9.5V	G06
$21 \ 09 \ 20.74$	$-82 \ 01 \ 38.1$	104436	199509	6.98	0.619	41.28	0.58	41.95	0.37	G1V	G06
$21 \ 09 \ 22.45$	-73 10 22.7	104440	200525	5.67	0.590	53.38	2.18	50.59	1.52	F9.5V	G06
$21 \ 14 \ 28.82$	$+10\ 00\ 25.1$	104858	202275	4.47	0.529	54.11	0.85	54.09	0.66	F7V	G03
$21 \ 18 \ 02.97$	$+00 \ 09 \ 41.7$	105152	202751	8.15	0.990	52.03	1.23	50.46	1.03	K3V	G03
$21 \ 18 \ 27.27$	$-43 \ 20 \ 04.7$	105184	202628	6.75	0.637	42.04	0.90	40.95	0.46	G1.5V	G06
$21 \ 19 \ 45.62$	$-26\ 21\ 10.4$	105312	202940	6.56	0.737	53.40	1.09	55.65	0.62	G7V	G06
$21 \ 24 \ 40.64$	$-68 \ 13 \ 40.2$	105712	203244	6.98	0.723	48.86	0.81	48.97	0.68	G8V	G06
$21 \ 26 \ 58.45$	$-56\ 07\ 30.9$	105905	203850	8.65	0.924	43.12	1.17	43.47	1.01	K2.5V	G06
$21 \ 27 \ 01.33$	$-44\ 48\ 30.9$	105911	203985	7.49	0.876	42.52	1.29	42.54	1.32	K2III-IV	G06
$21 \ 36 \ 41.24$	-50 50 43.4	106696	205390	7.14	0.879	67.85	0.92	68.40	0.58	K1.5V	G06
$21 \ 40 \ 29.77$	$-74 \ 04 \ 27.4$	107022	205536	7.07	0.755	45.17	0.67	45.41	0.52	G9V	G06
$21 \ 44 \ 08.58$	+28 44 33.5	107310	206826	4.49	0.512	44.64	0.69	44.97	0.43	F6V	G03
$21 \ 44 \ 31.33$	$+14 \ 46 \ 19.0$	107350	206860	5.96	0.587	54.37	0.85	55.91	0.45	G0IV-V	G03
$21 \ 48 \ 00.05$	$-40\ 15\ 21.9$	107625	207144	8.62	0.960	42.12	1.06	42.20	0.93	K3V	G06
$21 \ 48 \ 15.75$	$-47 \ 18 \ 13.0$	107649	207129	5.57	0.601	63.95	0.78	62.52	0.35	G0V	G06
21 53 05.35	+205549.9	108028	208038	8.18	0.937	41.71	0.98	43.40	0.75	K2.5V	G03
21 54 45.04	+32 19 42.9	108156	208313	7.73	0.911	49.21	0.93	50.11	0.80	K2V	G03
$22 \ 09 \ 29.87$	$-07 \ 32 \ 55.1$	109378	210277	6.54	0.773	46.97	0.79	46.38	0.48	G8V	G03
22 11 11.91	$+36\ 15\ 22.8$	109527	210667	7.23	0.812	44.57	0.79	43.67	0.53	G9V	G03
Continued o	n Next Page										

48

						Hippe	trcos	FvI	201		
R.A.	Decl.	HIP	HD			π.	σ_{π}	π	σ_{π}	Spec	
(J2000.0)	(J2000.0)	Name	Name	Λ	B - V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$22 \ 14 \ 38.65$	$-41 \ 22 \ 54.0$	109821	210918	6.23	0.648	45.19	0.71	45.35	0.37	G2V	G06
$22 \ 15 \ 54.14$	$+54 \ 40 \ 22.4$	109926	211472	7.50	0.810	46.62	0.67	46.43	0.50	K0V	G03
$22 \ 18 \ 15.62$	$-53 \ 37 \ 37.5$	110109	211415	5.36	0.614	73.47	0.70	72.54	0.36	G0V	G06
$22 \ 24 \ 56.39$	$-57\ 47\ 50.7$	110649	212330	5.31	0.665	48.81	0.61	48.63	0.34	G2IV-V	G06
$22 \ 25 \ 51.16$	-75 00 56.5	110712	212168	6.12	0.599	43.39	0.96	43.39	0.50	G0V	G06
$22 \ 39 \ 50.77$	$+04\ 06\ 58.0$	111888	214683	8.48	0.938	44.10	1.12	41.49	0.76	K3V	G03
$22\ 42\ 36.88$	$-47 \ 12 \ 38.9$	112117	214953	5.99	0.584	42.47	0.72	42.31	0.40	F9.5V	G06
$22 \ 43 \ 21.30$	$-06\ 24\ 03.0$	112190	215152	8.11	0.966	46.46	1.31	46.47	0.90	K3V	G03
$22 \ 46 \ 41.58$	+12 10 22.4	112447	215648	4.20	0.502	61.54	0.77	61.36	0.19	F7V	HIP
$22 \ 47 \ 31.87$	$+83 \ 41 \ 49.3$	112527	216520	7.53	0.867	50.15	0.64	50.83	0.44	K0V	G03
$22 \ 51 \ 26.36$	+13 58 11.9	112870	216259	8.29	0.849	47.56	1.18	46.99	1.01	K2.5V	G03
22 57 27.98	$+20\ 46\ 07.8$	113357	217014	5.45	0.666	65.10	0.76	64.07	0.38	G3V	G03
$22 \ 58 \ 15.54$	$-02 \ 23 \ 43.4$	113421	217107	6.17	0.744	50.71	0.75	50.36	0.38	G8IV-V	G03
$23 \ 03 \ 04.98$	+20 55 06.9	113829	217813	6.65	0.620	41.19	0.87	40.46	0.57	G1V	G03
$23 \ 10 \ 50.08$	$+45 \ 30 \ 44.2$	114456	218868	6.98	0.750	42.65	0.74	41.15	0.54	G8V	G03
$23 \ 13 \ 16.98$	+57 10 06.1	114622	219134	5.57	1.000	153.24	0.65	152.76	0.29	K3V	G03
$23 \ 16 \ 18.16$	$+30\ 40\ 12.8$	114886	219538	8.07	0.871	41.33	0.97	41.63	0.72	K2V	G03
$23 \ 16 \ 42.30$	+53 12 48.5	114924	219623	5.58	0.556	49.31	0.58	48.77	0.26	F7V	HIP
$23 \ 16 \ 57.69$	$-62 \ 00 \ 04.3$	114948	219482	5.64	0.521	48.60	0.60	48.69	0.33	F6V	G06
$23 \ 19 \ 26.63$	+79 00 12.7	115147	220140	7.53	0.893	50.65	0.64	52.07	0.47	K2V	G03
$23 \ 21 \ 36.51$	$+44\ 05\ 52.4$	115331	220182	7.36	0.801	45.63	0.83	46.46	0.53	G9V	G03
$23 \ 23 \ 04.89$	$-10\ 45\ 51.3$	115445	220339	7.80	0.881	51.37	1.25	52.29	0.86	K2.5V	G03
$23 \ 31 \ 22.21$	$+59 \ 09 \ 55.9$	116085	221354	6.76	0.839	59.31	0.67	59.06	0.45	K0V	G03
$23 \ 35 \ 25.61$	$+31 \ 09 \ 40.7$	116416	221851	7.90	0.845	42.63	0.93	42.00	0.72	K1V	G03
$23 \ 37 \ 58.49$	$+46\ 11\ 58.0$	116613	222143	6.58	0.665	43.26	0.80	42.86	0.42	G3V	G03
Continued o	n Next Page										

						Hippe	urcos	FvI	20'		
R.A.	Decl.	HIP	HD			Я	σ_{π}	π	σ_{π}	\mathbf{Spec}	
(J2000.0)	(J2000.0)	$\mathbf{N}\mathbf{a}\mathbf{m}\mathbf{e}$	Name	Λ	B-V	(mas)	(mas)	(mas)	(mas)	Type	Ref
(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
23 39 37.39	$-72 \ 43 \ 19.8$	116745	222237	7.09	0.989	87.72	0.64	87.56	0.51	K3+V	G06
$23 \ 39 \ 51.31$	$-32 \ 44 \ 36.3$	116763	222335	7.18	0.802	53.52	0.86	53.85	0.63	G9.5V	G06
$23 \ 39 \ 57.04$	$+05\ 37\ 34.6$	116771	222368	4.13	0.507	72.51	0.88	72.92	0.15	F7V	HIP
23 52 25.32	+75 32 40.5	117712	223778	6.36	0.977	92.68	0.55	91.82	0.30	K3V	G03
$23 \ 56 \ 10.67$	$-39 \ 03 \ 08.4$	118008	224228	8.24	0.973	45.28	1.11	45.52	0.93	K2.5V	G06
$23 \ 58 \ 06.82$	$+50\ 26\ 51.6$	118162	224465	6.72	0.694	41.35	0.76	40.77	0.49	G4V	G03

NOTES.—Column 12 lists the source of the spectral types and has the following values: G03=Gray et al. (2003),

G06=Gray et al. (2006), HIP=The *Hipparcos* catalog.

^a The parallax is from Söderhjelm (1999).

Table 2.2 lists the 24 stars that were left out of the sample because their parallax error was greater than 5% of the corresponding parallax value. All columns have the same meaning as Table 2.1. Only four of these stars have a parallax larger than 40 mas to a 3- σ significance, and these were also left out of the sample so as to only include "good-quality" parallax measures from *Hipparcos*. Table 2.3 lists the 15 stars excluded from the final sample due to a large offset from the best-fit main sequence, i.e. being over 2 magnitudes above or 1.5 magnitudes below it. Columns 1–8 have the same meaning as in Table 2.1, Column 9 lists the derived absolute V magnitude from the apparent magnitude and parallax, and Column 10 lists the absolute magnitude of the best-fit main sequence corresponding to the B - Vcolor of the star.

TABLE	2.2:	Stars	Excluded	Due to	Large	Parallax	Errors	$(\sigma_{\pi}/$	π	\geq	0.0)5)
-------	------	-------	----------	--------	-------	----------	--------	------------------	---	--------	-----	-----

						Hipp	arcos
R.A.	Decl.	HIP	HD			π	σ_{π}
(J2000.0)	(J2000.0)	Name	Name	V	B-V	(mas)	(mas)
00 22 23.61	$-27 \ 01 \ 57.3$	001768	001815	8.30	0.888	44.57	7.12
$01 \ 49 \ 23.36$	$-10\ 42\ 12.8$	008486	011131	6.72	0.654	43.47	4.48
$02 \ 15 \ 42.55$	$+67 \ 40 \ 20.2$	010531	013579	7.13	0.920	42.46	2.51
$02 \ 57 \ 14.71$	-24 58 10.2	013772	018455	7.33	0.863	44.49	2.55
$03 \ 47 \ 02.12$	$+41 \ 25 \ 38.2$	017666	023439	7.67	0.796	40.83	2.24
$04 \ 29 \ 44.87$	$-29 \ 01 \ 37.4$	020968		11.42	0.646	120.70	56.47
$04 \ 30 \ 12.58$	$+05 \ 17 \ 55.8$	021000		9.83	0.600	84.76	4.74
$05 \ 44 \ 56.79$	+09 14 31.5	027111	247168	11.35	0.699	44.67	14.98
$07 \ 03 \ 58.92$	$-43 \ 36 \ 40.8$	034069	053706	6.83	0.779	66.29	6.81
$08 \ 35 \ 51.27$	$+06 \ 37 \ 22.0$	042173	072946	7.25	0.710	42.71	4.61
$10\ 04\ 50.59$	$-31 \ 05 \ 28.0$	049376		11.99	0.938	41.59	3.13
$12\ 29\ 55.04$	$+36 \ 26 \ 42.1$	060970		11.91	0.800	43.42	3.62
$12 \ 31 \ 18.92$	$+55 \ 07 \ 07.7$	061100	109011	8.08	0.941	42.13	3.11
12 59 18.98	$+06 \ 30 \ 33.7$	063383		10.69	0.515	45.19	44.19

Continued on Next Page...

TABLE 2.2 – Continued

						Hipp	arcos
R.A. (J2000.0)	Decl. $(J2000.0)$	HIP Name	HD Name	V	B - V	π (mas)	σ_{π} (mas)
13 33 18.71	$-77 \ 34 \ 24.6$	066125		9.31	0.914	52.09	39.91
$15 \ 31 \ 54.05$	$+09 \ 39 \ 26.9$	076051		9.80	0.787	44.59	20.41
$15 \ 38 \ 39.95$	$-08 \ 47 \ 41.0$	076602	139460	6.56	0.520	44.21	4.72
$15 \ 38 \ 40.08$	$-08\ 47\ 29.4$	076603	139461	6.45	0.505	40.19	3.62
$16 \ 19 \ 31.52$	-30 54 06.7	079979	146835	7.29	0.585	56.82	23.38
$22\ 06\ 11.82$	$+10\ 05\ 28.7$	109119		10.20	0.668	93.81	66.11
$22 \ 12 \ 59.72$	$-47 \ 23 \ 11.1$	109670		11.48	0.660	44.20	16.43
$22 \ 26 \ 34.28$	$-16 \ 44 \ 31.7$	110778	212697	5.55	0.618	49.80	2.54
$23 \ 01 \ 51.54$	-03 50 55.4	113718	217580	7.48	0.943	59.04	3.40
23 44 07.36	-27 11 45.8	117081	222834	9.01	0.535	58.45	47.16

Sequence
Main
the
from
Offset
Large (
ue to
Excluded D
S: Stars
2.0
TABLE

						Hippc	<i>ircos</i>		
R.A. (J2000.0)	Decl. (J2000.0)	HIP Name	HD Name	Δ	B - V	π (mas)	σ_{π} (mas)	M_V	${ m MS}~M_V$
$00 \ 49 \ 09.90$	$+05\ 23\ 19.0$	003829		12.37	0.554	226.95	5.35	14.15	4.21
01 55 57.47	$-51 \ 36 \ 32.0$	200000	011937	3.69	0.844	57.19	0.62	2.48	5.72
$03 \ 43 \ 14.90$	$-09 \ 45 \ 48.2$	017378	023249	3.52	0.915	110.58	0.88	3.74	6.08
$05 \ 16 \ 41.36$	+45 59 52.8	024608	034029	0.08	0.795	77.29	0.89	-0.48	5.48
$07 \ 34 \ 27.43$	+62 56 29.4	036834		10.40	0.942	87.01	2.17	10.10	6.22
$07\ 42\ 57.10$	-45 10 23.2	037606	062644	5.04	0.765	41.43	0.81	3.13	5.33
$07 \ 45 \ 18.95$	$+28\ 01\ 34.3$	037826	062509	1.16	0.991	96.74	0.87	1.09	6.47
$16\ 41\ 17.16$	+31 36 09.8	081693	150680	2.81	0.650	92.63	0.60	2.64	4.73
$18\ 21\ 18.60$	-02 53 55.8	089962	168723	3.23	0.941	52.81	0.75	1.84	6.22
19 55 18.79	$+06\ 24\ 24.4$	098036	188512	3.71	0.855	72.95	0.83	3.03	5.78
$19\ 55\ 50.36$	$-26\ 17\ 58.2$	098066	188376	4.70	0.748	42.03	0.94	2.82	5.24
$20 \ 06 \ 21.77$	+35 58 20.9	099031	191026	5.38	0.850	41.34	0.54	3.46	5.75
$20\ 45\ 17.38$	+61 50 19.6	102422	198149	3.41	0.912	69.73	0.49	2.63	6.07
21 58 24.52	$+75 \ 35 \ 20.6$	108467		10.56	0.742	47.95	1.08	8.96	5.21
23 19 06.67	$-13\ 27\ 30.8$	115126	219834	5.20	0.787	40.28^{a}	1.51^{a}	3.23	5.44

^a The parallax is from Söderhjelm (1999).

2.1.2 Comparison with DM91 Sample

In order to minimize selection effects, DM91 chose a volume-limited sample out to 22 pc based on (Gliese 1969) parallaxes. The *Hipparcos* mission, whose results were published well after the DM91 effort, greatly improved the accuracy and completeness of parallaxes. As pointed out by Halbwachs et al. (2003), the Gliese (1969) data becomes increasingly incomplete for parallaxes below 53 mas, and I will show that a sample selected using DM91's criteria from *Hipparcos* is significantly different than their sample.

As described in §1.3.2, the DM91 volume-limited sample of solar-type stars included 164 primaries. Selecting an equivalent sample, subject to the same criteria but from the *Hipparcos* catalog, results in the selection of 148 primaries, but only 92 of these overlap with the DM91 sample. This indicates that 72 (44%) of the DM91 sample is now known to lie outside their selection criteria, and 56 (38%) of the stars meeting their criteria are not included in their sample. Figure 2.5 plots the 164 stars of the DM91 sample (circles) based on their *Hipparcos* parallax and B - V color. The filled circles represent stars that still match their selection criteria based on the *Hipparcos* data, while the open circles identify stars that are now known to violate their criteria.

Figure 2.5 illustrates several points: (i) As expected, the density of stars in the plot drops off with proximity due to the cubed relation of volume to distance. I will use this in $\S2.1.4$ to test the completeness of the current sample. (ii) Many stars included by DM91 based on Gliese parallaxes are now known to reside beyond their distance limit of 22 pc. In fact, the updated *Hipparcos* parallaxes alone account for as many as 68 of the 72 stars



FIGURE 2.5: The DM91 multiplicity survey sample and the sample selected according to their criteria from the *Hipparcos* catalog are plotted against *Hipparcos* parallaxes and B-Vcolors. Filled circles represent DM91 sample stars that still match their criteria based on the *Hipparcos* data while open circles identify DM91 sample stars that no longer fit their criteria. Filled triangles represent stars that were not part of the DM91 sample but meet their criteria based on *Hipparcos* data. The vertical dashed line marks the parallax limit of 45 mas of the DM91 study, and the horizontal dashed lines mark B-V colors of 0.5 and 0.8, which roughly correspond to the DM91 spectral-type limits of F7–G9.

in the DM91 sample that no longer fit their criteria. Most of these distant stars (five have *Hipparcos* parallaxes of less than 10 mas) are likely to be evolved stars. (iii) The *Hipparcos* data indicates that the DM91 sample was fairly incomplete, as evidenced by the number of filled triangles inside the DM91 criteria "window" outlined by the dashed lines, the vertical line corresponding to a parallax of 45 mas and the horizontal lines to B - V color limits of 0.5 and 0.8, which roughly correspond to the DM91 spectral-type limits of F7–G9 (Cox 2000). The filled triangles represent the 56 stars that were not part of the DM91 sample but consistent with their criteria. One outlier (HIP 84123) is not included in the figure because its very red B - V color of 1.6 is beyond the plot's range, but consistent with its spectral type of M3V (Gray et al. 2006). It was however selected as matching the DM91 criteria because of an erroneous *Hipparcos* spectral type of G. (iv) Most of the stars lying outside the DM91 criteria window due to their color are red rather than blue, i.e. there are many more points above the window than below. Furthermore, most of these outliers were not part of the DM91 sample, but their *Hipparcos* spectral types indicate G-dwarfs, which appears incorrect based on their colors. This might be suggestive of the fact that a sample selection is based on the *Hipparcos* B - V color, and we will compare these colors to recent independent assessment of their spectral types in §2.1.4 to check the quality of the *Hipparcos* colors.

We have thus seen that there are substantial differences between the samples selected according to DM91 criteria from Gliese (1969) and *Hipparcos*. The key question, however, is whether these differences introduce biases that would taint their results. One such aspect explored by Halbwachs et al. (2003) was whether the incompleteness of the CNS catalog at farther distances implied that fainter stars were less studied hence not fully represented in the catalog. If this were true, there would be a bias towards binaries due to a similar reasoning as the Malmquist bias, i.e. single stars would be included to a smaller volume of space than binaries. However, Halbwachs et al. (2003) showed that the samples of stars from CNS and *Hipparcos* have similar distributions around the main sequence, arguing against any magnitude bias.

My thesis effort attempts to verify whether the DM91 results were biased due to this sampling difference and obtain updated multiplicity statistics based on a larger sample chosen using the more accurate *Hipparcos* catalog. It also leverages current studies on stellar and substellar companions to solar-type stars, enabling the exploration of a key aspect of my motivation, namely, the correlations between stellar and substellar companions to solar-type stars. While no substellar companions were confirmed before the DM91 effort, well over 300 such companions have been published to-date, offering a wealth of observational results to conduct this analysis. Figure 2.6 plots the thesis sample of 454 primaries based on their *Hipparcos* parallax and B - V color.

2.1.3 A New Reduction of the *Hipparcos* Data

van Leeuwen (2007a) recently published the results of a new reduction of the *Hipparcos* raw data based on an improved model of the satellite attitude, yielding improved astrometry but making no updates to the photometry. A more recent update was posted online¹ (van Leeuwen 2007b, hereafter FvL07), which corrected some errors relating to the goodness of fit calculation in the original work. While these results came well after my sample construction and bulk of the observational work, I discuss below their possible effects on the multiplicity statistics derived in this work. Selecting the thesis sample using the FvL07 results yields 459

 $^{^{1}}http://webviz.u-strasbg.fr/viz-bin/VizieR-3$


FIGURE 2.6: The current sample plotted along the same axes as Figure 2.5

stars compared to the 462 selected from *Hipparcos*. Both these samples include the same nine stars that are removed from the sample of primaries because they are companions to other primary stars in the sample. Thus, the final sample of stars selected from FvL07 would contain 451 primary stars including the Sun, compared to the *Hipparcos*-based sample of 454 stars. The total number of stars in both samples is almost the same, and a check of specific stars reveals the 15 new stars that would have been included, and the 18 stars that would have been left out if the sample were selected using FvL07data. Tables 2.4 and 2.5 list these specific stars with the individual reasons for exclusion or inclusion. Compared to the sample size, these differences are small, and most of them are due to small changes



FIGURE 2.7: The current sample plotted using the updated FvL07 parallaxes. Solid circles identify stars of the current sample that qualify selection using FvL07, open circles identify the ones which do not qualify selection using the updated data, and filled triangles are stars not in the current sample but match the criteria based on the FvL07 data. The size of the solid circles have been suppressed to enable easier spotting of the other symbols. Tables 2.4 and 2.5 detail these two sets of stars.

in the parallax value near the 40 mas limit. Accordingly, while the updated work presents improved astrometry which will be used in the determination of orbital parameters and mass determinations, they are not expected to have a significant impact on the derived multiplicity statistics. Figure 2.7 compares the *Hipparcos* and FvL07 samples by parallax and color, and Figure 2.8 shows these stars on a color-magnitude diagram, demonstrating that the two samples are largely consistent for statistical analyses. One star included in the



FIGURE 2.8: Color-Magnitude diagram for the current sample using *Hipparcos* and FvL07 data. Symbols have the same meaning as in Figure 2.7.

thesis sample, HIP 55203, was selected based on Söderhjelm (1999) parallax but does not exist in *Hipparcos* or FvL07 catalogs and hence is plotted in the figure using its Söderhjelm (1999) parallax.

TABLE 2.4: Stars excluded from sample, but qualifying in FvL07 data

HIP		Hipp	oarcos	\mathbf{Fv}	L07	Reason for
Name	B - V	π (mas)	σ_{π} (mas)	π (mas)	σ_{π} (mas)	Exclusion
003170	0.564	39.26	0.56	40.07	0.34	1
010531	0.920	42.46	2.51	53.82	1.74	2
013388	0.827	38.45	0.90	40.83	0.74	1
013772	0.863	44.49	2.55	44.51	2.09	2

Continued on Next Page...

HIP		Hipp	arcos	\mathbf{Fv}	L07	Reason for
Name	B - V	π (mas)	σ_{π} (mas)	π (mas)	σ_{π} (mas)	Exclusion
017336	0.659	38.75	0.99	40.55	1.11	1
021223	0.990	39.93	0.93	40.36	0.78	1
030862	0.613	38.98	0.90	40.69	0.62	1
034950	0.827	39.72	1.29	40.56	0.90	1
035872	0.959	39.91	1.14	41.72	0.86	1
042418	0.897	39.28	1.21	40.27	1.14	1
060994	0.591	39.6^{a}	0.9^a	40.57	0.60	3
088945	0.633	39.62	0.68	40.29	0.49	1
110778	0.618	49.80	2.54	49.50	1.23	2
113718	0.943	59.04	3.40	58.70	0.92	2
114378	0.607	38.67	0.83	40.28	0.53	1

TABLE 2.4 - Continued

NOTES.—Column 7 notes: (1) = Hipparcos parallax less than the 40 mas threshold, (2)

= Hipparcos parallax error greater than the 5% threshold, (3) = Parallax from Söderhjelm

(1999) less than the 40 mas threshold.

^a The parallax is from Söderhjelm (1999).

TABLE 2.5: Sample stars outside selection criteria based on FvL07 data

HIP		Hipp	oarcos	\mathbf{Fv}	L07	Reason for
Name	B - V	π (mas)	σ_{π} (mas)	π (mas)	σ_{π} (mas)	Exclusion
007372	0.909	42.29	1.47	46.24	3.07	1
015442	0.655	40.52	0.98	39.65	0.74	2
017147	0.554	41.07	0.86	39.12	0.56	2
024786	0.572	40.11	0.76	39.96	0.40	2
026373	0.845	41.90	1.74	39.82	1.36	2
032423	0.964	40.02	1.22	38.11	1.01	2
034567	0.700	40.68	1.02	39.73	0.54	2
040167	0.531	39.11	1.38	39.87	0.82	2
051933	0.528	40.67	0.68	39.88	0.37	2
054155	0.770	40.57	1.40	38.06	0.99	2
055203	0.606	119.70^{a}	0.80^{a}		•••	3

Continued on Next Page...

HIP		Hipp	oarcos	\mathbf{Fv}	L07	Reason for
Name	B - V	π (mas)	σ_{π} (mas)	π (mas)	σ_{π} (mas)	Exclusion
062505	0.946	47.19	1.93	41.96	3.00	1
063406	0.940	41.36	1.48	39.71	1.00	2
067655	0.663	40.02	1.00	39.42	0.97	2
069701	0.511	46.74	0.87	44.97	0.19	4
073100	0.533	40.25	0.54	39.83	0.26	2
074975	0.540	40.46	0.81	39.40	0.29	2
081813	0.769	41.15	0.57	39.97	0.45	2

TABLE 2.5 - Continued

NOTES.—Column 7 notes: (1) = FvL07 parallax error greater than the 5% threshold, (2) = FvL07 parallax less than the 40 mas threshold, (3) = Not in *Hipparcos* or FvL07 catalogs. Selected based on Söderhjelm (1999) parallax, (4) = More than 2 magnitudes above FvL07 main-sequence.

^a The parallax is from Söderhjelm (1999).

2.1.4 Sample Analysis & Bias

The targets for this survey were selected as a distance-limited sample from the *Hipparcos* catalog to minimize selection effects. In this section, I analyze the sample based on parallax, magnitude, color, and spectral type as a way of of determining the sample completeness and discussing any remaining biases. Söderhjelm (2000) analyzed the completeness of the *Hipparcos* catalog based on brightness and presented V magnitude limits based on the V - Icolor and galactic latitude. These relations indicate that the catalog is complete to $V \sim 9$ for solar-type stars with galactic latitudes of at least 40° and to $V \sim 8.2$ for all latitudes. Figure 2.9 shows the distribution of apparent (solid) and absolute (dashed) V magnitudes of the sample. The largest apparent V magnitude is 9.05 and only 30 of the 454 stars have V > 8.2, indicating that the sample is fairly complete. As an additional check, Figure 2.10 shows the sample's distance distribution along with a curve representing the theoretical cubed-dependence of the number of stars in a given volume of space based on distance, using the number of stars within 11 pc as the reference (dotted line). The actual distribution of stars closely follows the expected curve out to the distance-limit of 25 pc, improving the confidence in the completeness of the sample.



FIGURE 2.9: Apparent magnitude (solid line) and absolute magnitude (dashed line) distributions of the sample.

Unlike DM91's sample selection based on spectral type, I used the *Hipparcos* B-V color. Figure 2.11 shows the color-distribution of the sample, indicating a peak around B-V = 0.6,



FIGURE 2.10: Distance Distribution of the Sample. The dotted line represents the theoretical distribution of the number of stars in a given volume of space, based on its cubed relation to distance. The number of sample stars within 11 pc is used as the reference for this curve.

just blueward of the Sun with a tapering distribution out to the limit of 1.0. In contrast, Figure 2.12 shows a double-peaked distribution with a strong peak at K2 and a smaller peak at G0. To further analyze this, Figure 2.13 shows the correlation between color and spectral type. While the figure indicates a fairly good correlation between these parameters, the range of colors for many spectral types is quite broad. These figures suggest that while color is determined directly from photometric measurements, spectral identifications are more complex and involve the interpretation of spectral lines, which may not always be easy. The difficulty in identifying certain spectral types is likely responsible for the low counts for the mid-G stars and a high count for K2.



FIGURE 2.11: B - V Color Distribution of the Sample.

2.2 Survey Methods

The primary observational efforts of this work include a survey for Separated Fringe Packet (SFP) companions using long baseline interferometry (LBI) at the Center for High Angular Resolution Astronomy (CHARA) Array, a search for wide CPM companions by blinking multi-epoch archival images, follow-up photometry for the candidates thus revealed, and speckle interferometric observations to achieve a near 100% completion using this technique. The CHARA Array was also used to develop visual orbits for select short-period spectroscopic binaries. The results of the LBI efforts are covered in Chapter 3. The search for wide CPM



FIGURE 2.12: Spectral type distribution of the sample.



FIGURE 2.13: Correlation between spectral type and color for the sample of stars.

companions and the follow-up photometry are discussed in Chapter 4, which also covers the

evaluation of entries in the Washington Double Star Catalog² (WDS) as CPM pairs or chance alignments of optical pairs.

Apart from the targeted observations, this survey benefits from the tremendous amount of attention these nearby solar-type stars have garnered over the years. A comprehensive synthesis effort brings together information about previously known and suspected companions, which are evaluated individually. Chapter 5 addresses astrometric companions discovered by the *Hipparcos* mission, the resolved-pair and photocentric-motion visual binaries cataloged in the Sixth Catalog of Orbits of Visual Binary Stars³ (VB6), visual resolutions cataloged in the WDS and the Fourth Catalog of Interferometric Measurements of Binary Stars⁴ (FIC), and the CPM as well as spectroscopic binaries listed in the Catalogs of Nearby Stars (Gliese 1969; Gliese & Jahreiß 1979, 1991, hereafter CNS). Finally, Chapter 6 covers the spectroscopic companions from publications as well as from the unpublished data made available to this effort. This chapter also mentions the few eclipsing binaries in the sample of stars of this study.

 $^{^{2}}http://ad.usno.navy.mil/wds/$

 $^{^{3}}http://ad.usno.navy.mil/wds/orb6.html$

 $^{^{4}}http://ad.usno.navy.mil/wds/int4.html$

If it was easy, someone would have already done it! — Harold A. McAlister

LONG-BASELINE INTERFEROMETRIC RESULTS

– 3 –

The primary observing method of this thesis effort aims to leverage the high resolving power of interferometers, especially the world's longest baselines for optical and near-infrared interferometry of the CHARA Array. A brief overview of the technique and instrument is outlined below. For a more complete description, see ten Brummelaar et al. (2005).

Ever since the days of William Herschel, visual observers have continually been in pursuit of better angular resolution. For single-aperture techniques, the resolution limit or Rayleigh Criterion (θ) depends on the wavelength of light collected (λ) and the aperture of the telescope (D), and is given, in radians, by $\theta = 1.22 \lambda/D$, which corresponds to the situation when the first diffraction minimum of one source coincides with the maximum of another. The actual resolution of observations can, at best, equal the above limit, but is often worse due to atmospheric and instrumental effects. Techniques such as AO and speckle interferometry have been successfully used to mitigate astrometric effects and approach the resolution limit. The pursuit of even higher resolutions has led to the use of innovative techniques such as Long-Baseline Interferometry (LBI), which involve the simultaneous observation of a target with two or more telescopes. By leveraging multiple telescopes, this technique effectively uses the distance between the telescopes, or *baseline*, in place of aperture in the above relationship to get far better resolutions than the single telescopes for resolved stellar disks (see Michelson 1890, 1920). Further, Michelson also showed that unresolved point sources could be detected at a resolving power of 2.4 times the above value. This resolution of an interferometer is related to the wavelength (λ) and baseline (B) as

$$\theta_{\rm int} = \lambda/2B \tag{3.1}$$

Let us look at the above in a bit more detail. From Chapter 3 of Lawson (2000), the intensity of an image after diffraction though a circular aperture of diameter D for a wavelength λ is given by

$$I_{\rm tel}(\theta) = \left[\frac{2J_1(\pi\theta D/\lambda)}{\pi\theta D/\lambda}\right]^2 D^2,\tag{3.2}$$

where J_1 is the first-order Bessel function and θ , the independent variable, is the angle from the peak intensity in the center. This intensity function is plotted in Figure 3.1 as the solid outer envelope. The condition corresponding to the Rayleigh Criterion is when the peak of one source coincides with the first zero of another, and this corresponds to $\theta_{tel} = 1.22\lambda/D$ as seen in the figure. Now, when we use two apertures of diameter D, separated by a distance B, the resultant intensity function for an interferometer becomes (Lawson 2000)

$$I_{\rm int}(\theta) = 2I_{\rm tel}(\theta) \left[1 + \cos(2\pi\theta B/\lambda)\right]$$
(3.3)

Figure 3.1 also plots this function for B = 3D as the narrow-inner fringe envelope. Similar to the discussion above, the resolution of an interferometer is considered to be the first zero of this function, which corresponds to $\theta_{int} = \lambda/2B$ or $\theta_{int} = \lambda/6D$ as seen in the figure.

While long baseline interferometers can, in general, use any number of telescopes greater than one, this effort used only two-telescope interferometry, and so the discussion here will be limited to this situation. In order to obtain interferometric fringes, light wavefronts from the



FIGURE 3.1: Intensity of the diffraction patterns from single and two-telescope images. The broad, thick envelope represents a single-telescope diffraction pattern and the narrow fringes are for a two-telescope setup with B = 3D. The single aperture and interferometer resolution limits are marked.

two telescopes must be combined, or allowed to interfere. One aspect that makes this difficult is the fact that while the difference in distance from the star to the two telescopes is negligible when compared to the distance to either telescope, it is not zero. Hence, a wavefront from the star reaches one telescope a bit sooner than it does the other. Interferometers are designed to compensate for this delay to sub-micron precision to generate the fringe. A simple pictorial depiction of this is shown in Figure 3.2. CHARA uses a two-stage delay compensation method including Pipes of Pan (POP) mirrors which offer one of five fixed delays for each telescope between 0 and 143 meters in roughly 37-meter increments, and Optical Path-Length Equalizer (OPLE) carts that run on 46-m rails to compensate for the delay in real-time to a precision of about 10-nm (ten Brummelaar et al. 2005). Because the



FIGURE 3.2: Schematic of a basic interferometer showing the delay compensation (diagram by H. McAlister).

target star is rising or setting during an observation, its differential distance to the telescopes changes constantly, and the carts move actively during an observation to keep up with this change.

The final step in generating interference fringes is combining the delay-compensated beams from the two telescopes. The beam combiner used in this effort, CHARA Classic, is a pupil-plane interferometer, which passes the light from the two telescopes through the two sides of a half-reflecting and half-transmitting piece of glass to enable interference, producing two beams with the interference pattern, one obtained by combining the reflected light from one telescope and the transmitted light from the other, and the other with the transmitted light from the first telescope and the reflected light from the second. This technique of beam combination is referred to as a *Michelson* interferometer because it uses the same method as used by Michelson in 1893 to demonstrate that the speed of light is independent of the observer's motion. When a fringe is found and tracked, the observations need to record a sampling of the fringe pattern over a range of delay space around the peak of the fringe. This is accomplished by an oscillating mirror (Dither mirror) that changes the path length of the light from one telescope with respect to the other over a range of roughly 80 microns for "short-scan" mode to 160 microns for "long-scan" sample at a velocity of about 0.3 mm s⁻¹. Figure 3.3 shows a picture of the CHARA Classic beam combination setup, with captions and lines representing the light beams added for clarity. For the Classic setup, the two telescope beams are sent along beams 5 and 6 of the six-telescope arrangement, and as seen in the figure, beam 5 is reflected off the Dither mirror before combination. Fringe data are then recorded from both the interferences (B5R+B6T and B5T+B6R) at a sampling rate of 250 Hz, 500 Hz, 750 Hz, or 1000 Hz, which is typically 5 times the fringe frequency to adequately sample each fringe and is observer-selected based on seeing conditions. In the above notations, "T" stands for transmitted light, and "R" for reflected light.

Using Equation (3.1), observations at the K' band of 2.15 microns with the longest CHARA baseline of 330 meters have a resolution of 0.7 mas. However, the definition of the first zero corresponding to the resolution is somewhat arbitrary, and simulations show resolution capabilities down to about 0.4 mas for binaries (H. McAlister 2008, private communication). These resolutions are valuable for resolving short-period double-lined spectroscopic pairs, and for periods less than about 10 days, the Array's longest baselines are uniquely suited. This technique has been utilized to resolve some short-period binaries as discussed in § 3.2. First, § 3.1 below covers the survey effort in the search for new companions at wider separations (10–120 mas) using the Separated Fringe Packet (SFP) approach.



FIGURE 3.3: A picture of the beam combining equipment at CHARA Classic. Captions and lines representing light beams have been added for clarity. B5 and B6 represent light from the two telescopes, which are sent along beams 5 and 6 of a six-telescope arrangement, and the interference pattern emerging from both sides of the beam combiner are labeled with suffixes R and T representing reflected and transmitted light, respectively.

3.1 Survey of Separated Fringe Packet Binaries

A hypothesis tested by this survey effort via SFP analysis is whether this technique, leveraging the long baselines of the CHARA Array, can address a gap in binary separations between spectroscopic techniques and traditional visual methods. While spectroscopic efforts are suited to short-period systems, typically from hours to a few dozen years, visual techniques are ideal for more widely separated systems with orbital periods of many-tens to thousands of years. Speckle interferometry using 4-m class telescopes routinely resolve binaries with separations down to 30 mas (Horch et al. 2008; Mason et al. 2009a) and AO can resolve high Δ mag binaries to sub-arcsec separations (e.g., Tokovinin et al. 2006; Turner et al. 2008). However, as seen above, LBI allows us to probe closer-in on visual binaries. While the resolution limits of LBI described in the preceding section are for *overlapping* fringe envelopes separated at least by θ_{int} in Figure 3.1, wider binaries may have spatially *separated* fringe packets that fit within the delay-space scanning window. This technique was first demonstrated as a viable means of studying binary stars by Dyck et al. (1995) and later used with the CHARA Array to derive a precise orbit and component masses for 12 Persei (Bagnuolo et al. 2006).

While SFP studies of binary stars are an effective way of characterizing orbits and the component stars, they are especially suited to searching a large list of stars for new companions. In contrast to Calibrated Interferometric Visibility (CIV) measurements (more on this in § 3.2), SFP analysis can identify a companion with just a single observation and can confirm the absence of a companion within detection limits with just two observations taken over a short time gap, one each along orthogonal baselines. The CHARA Array can detect SFPs for binaries with separations of 10–120 mas (Bagnuolo et al. 2006) and is sensitive to $\Delta K' \leq 2.5$. Figure 3.4 shows some theoretical examples of separated fringes for various separations and intensity ratios.

Targets for the SFP survey were selected from the *Hipparcos* sample of 462 stars (including 9 companions), based on the magnitude and declination limits of CHARA. Specifically, the Array can reach targets north of a declination of -10° and has magnitude limits of $V \leq 9$ for tip-tilt tracking of the target and $K' \leq 6$ for fringe recording in moderate seeing conditions. These criteria selected 288 of the 462 targets, 92 of which were being observed by



FIGURE 3.4: Examples of theoretical separated fringe packets for different values of intensity ratios (IR above the figures) and separations (offsets above the figures in units of wavelength).

Chris Farrington as part of his Ph.D. effort (Farrington 2008) with the same instrument and technique, leaving me with 196 targets. The observations for this survey were carried out between 2007 January and 2008 July. Following the loss of allocated time in the Fall of 2006 due to camera issues at the Array, make-up time was assigned during 2007 January–March, the normal maintenance time due to unfavorable weather conditions on Mount Wilson, as a shared pool to various survey projects of Georgia State University (GSU) graduate students, enabling about 20 nights of data-gathering. Additionally, the equivalent of approximately 60 nights were awarded in total to this survey during the observing seasons of 2007 April–August, 2007 September–December, and 2008 April–August, allowing for the completion of a majority of the survey. The individual observing nights along with some notes regarding

observing conditions are included in Table 3.1. All but 12 (6%) of the 196 targets have so far been observed at least once, and 161 (82%) have been observed on orthogonal baselines and are hence considered done.

3.1.1 SFP Data Reduction and Results

The CHARA Classic observing software was changed to generate FITS files beginning 2008, before which, text files in a proprietary format were generated. The reduction code for 2007 data used a text file output containing some header information and three arrays for each observation. Each array entry represents an interpolated sampling of the readout to correspond to a frequency of 1000 Hz. So, for every millisecond, the arrays record the dither mirror position offset and the intensity of the light of the interference pattern on each side of the beam combiner as shown in Figure 3.3. The FITS files generated since 2008 contain similar information, but in a more standard form, and with an added array containing the time offset for every sampling interval. This is required because the data captured now corresponds to the actual sampling rate rather than an interpolated fixed rate. Also, the FITS files parse the data into two-dimensional arrays with the each row representing a particular scan of delay space, i.e. the motion of the Dither mirror from one end to the other, and the columns corresponding to the sampling step within the scan. The observing software records data until 200 good Signal-to-Noise Ratio (SNR) scans, as described below, are recorded, so a typical data packet contains over 200 scans.

vey	
ie SFP sur	
ghts for th	
3.1: Observing Ni	
TABLE	

Comment (7)		High minde	Poor, variable seeing early	High winds	High winds	Worked well mostly!	Clouds Clouds	Variable seeing	High RH, terrible seeing	Wind, Dust, Ash	Clouds	Clouds. Rain. Snow	Clouds	Variable seeing	Network issues	Variable seeing	11 brackets + 9 SFP	6 brackets of HD 146361	Clear & nice!	Includes 6 brackets of HD 146361 Cood mosther	Windv. some dust. bright moon. iet stream?	Light cloud cover	Calm, cool	Ciouds all nignt, drizzie later Partiv cioudy variable seeing	Many technical issues	Many Tip-Tilt issues	Delayed start due to MIRC offsets	Thin, high clouds, poor to moderate seeing Includes 6 hundles of HD 146361	Few hours spent on ToO Nova Vul 07	Few hours spent on ToO Nova Vul 07	Clear but hazy, steady stream of particulates	Little data due to time on 100 & smoke later Clear & nice!	Smoke, Haze, Clouds later	Clear & nice! Spent some time on ToO Nova Vul 07		Clear & nicel	Cool & dry, light winds	Clear	Fog, Clouds, High KH MIRC offsets and R1 metrology issues	Late start due to weather & technical issues	Mostly clear, passing clouds	High RH early, CA fires subsided, no smoke Sneut some time on ToO Comet Holmes	Very smooth, observed Yamina & Ellyn's targets	Clear	High winds, Malibu fire restarts!	
Baseline(s) (6)			S1(1)-E1(5)			SI(1)-EI(5)		S1(1)-W1(5)		:	:			S1(1)-W1(5)		E1(5)-W1(5) E1(1) S1(1)	$E_1(1)$ - $S_1(1)$	E2(4)-S1(1)	W1(5)-S1(1)	E2(4)-S1(1)	W1(1)-S1(1)	W1(5)-S1(2); E2(1)-S1(2)	W1(1)-S1(5); S1(5)-E1(1)	S1(5)-E1(1), S1(5)-W1(1)	S1(1)-E1(1)	S1(1)-E1(5); W1(5)-S1(1)	S1(1)-E1(4)	S1(1)-E1(4) S1(1)-E1(1)	S1(1)-E1(1); W1(1)-S1(1)	S1(5)-E1(1)	S1(5)-E1(1); S1(5)-W1(1)	W 1(5) - S1(5) W1(5) - S1(5); $S1(5) - E1(5)$		S2(2)-E1(5) W1(4)-S1(2)		W1(5)-S1(1) · S1(1)-E1(5)	$W_1(5)$ -S2(5)	E2(5)-S2(5)	:	S2(1)-E2(1)	E2(1)-S2(1)	S1(5)-E1(5), S1(5)-W1(5) F2(1)-W1(5), S1(5)-E1(5), S1(5)-W1(5)	E2(4)-S2(1)	E2(4)- $S2(1)$	E2(4)-S2(1)	
Data Sets RecordedFotalSFP(4)(5)			 16 4			27 17	: :	17 11			• • •	: :		14 12		25 2 2	36 36	26	23	26 3 10 7	10 6	23 10	26 12	1 27 16	8 8 9 9	18 9	8 ș	16 16	25 22 25 22	14 6	17 9 2	3 2 19 19	4	21 13 22 5		26 26	23 23	12 12		6 6	13 13	22 22 24 21	64			
I hare Observer 7 (2) (3)	August 2007	ап	H DR	H DR	H DR	H DR H	н DR	H PJ	H PJ	H PJ	ц Сц Сц Сц С	H CF	H CF	H DR	H DR	H DR DD	F DB	H CF	H CF	H PJ H	H DR	H DR	H CF	н Сг	H DR	H DR	H DR	H DK H	F DR	F DR	F DR T	F DR	H CF	H CF H PJ	ber–December 2007	Ъ	F CF	F PJ	F CF	F TB	F NT	F DR DR	Q DR	Q PJ	LA C	ext Page
UT Date S. (1)	2007-01: April-A	2006/01/10	04/11/2007	04/12/2007	04/13/2007	04/14/2007	04/16/2007	04/17/2007	04/18/2007	04/19/2007	04/20/2007	04/22/2007	04/23/2007	04/24/2007	04/25/2007	04/26/2007 05/16/2007	05/17/2007	05/27/2007	05/28/2007	05/29/2007 05/30/2007	05/31/2007	06/01/2007	07/22/2007	01/23/2007	07/25/2007	07/26/2007	07/27/2007	07/28/2007	08/14/2007	08/15/2007	08/16/2007	08/11/2007	08/19/2007	08/20/2007 $08/21/2007$	2007-02: Septem	09/16/2007	09/17/2007	09/18/2007	09/24/2007 09/25/2007	09/26/2007	09/27/2007	10/31/2007	11/18/2007	11/22/2007	$\frac{11/24}{2007}$	Continued on INE

Comment (7)	Power outage on the mountain! Power issues and bad weather Cloudy in the latter part of the night Clouds for half the night Clouds for half the night Clouds Clouds High humidity	Clear & nice Clear & nice Clear & nice Clear & nice Clear & nice Clear & nice Clear & nich High RH, then windy & dusty Hard to find/keep fringes for half the night Hard to find/keep fringes for half the night Smoke at the base of the mountain To dusty in the first half-night Clear, windy dusty Clear, windy dusty Clear, windy Clear & nice Hazy, high thin clouds Dust & ash Nice night Nice night Nice night Nice night OPLE issues throughout the night Clear & nice
Baseline(s) (6)	$\begin{array}{c} {\rm E2}(4){\rm -S2}(1)\\ \cdots\\ \cdots\\ \cdots\\ \end{array}$	$ \begin{array}{c} {\rm S1}(1)-{\rm W1}(1); {\rm S1}(1)-{\rm E1}(1)\\ {\rm S1}(1)-{\rm E1}(1); {\rm S1}(1)-{\rm W1}(1)\\ {\rm S1}(5)-{\rm W1}(1); {\rm S1}(5)-{\rm W1}(1)\\ {\rm S1}(5)-{\rm W1}(1)\\ {\rm S1}(5)-{\rm W1}(1)\\ {\rm S1}(5)-{\rm W1}(1)\\ {\rm S1}(5)-{\rm E1}(1)\\ {\rm S1}(5)-{\rm S1}(5)-{\rm S2}(5)\\ {\rm S1}(5)-{\rm E1}(1)\\ {\rm S1}(5)-{\rm E1}(1)\\ {\rm S1}(5)-{\rm E1}(1)\\ {\rm S1}(5)-{\rm E1}(1)\\ {\rm S1}(5)-{\rm S1}(2)\\ {\rm S1}(1)-{\rm S2}(5)\\ {\rm S1}(1)-{\rm S2}(5)\\ {\rm S1}(1)-{\rm S2}(1)\\ {\rm S1}(1)-{\rm S2}(1)\\ {\rm S1}(1)-{\rm S1}(5)-{\rm S2}(5)\\ {\rm S1}(1)-{\rm E1}(1)\\ {\rm S1}(5)-{\rm E1}(1)\\ {\rm S1}(5)-{\rm C1}(1)\\ {\rm S1}(1)-{\rm E1}(5)\\ {\rm$
s Recorded SFP (5)	· · · 4 · · 9 · · · · ·	22 232 332 332 332 33 33 33 33 33 33 33
Data Sets Total (4)	ວີ ວີ	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Observer (3)	DR EB EB TB TB PJ DR DR	Н С С С С С С С С С С С С С С С С С С С
Share (2)	Q Q Q F F F F F F F F	
UT Date (1)	11/25/2007 11/26/2007 11/26/2007 11/28/2007 11/39/2007 11/39/2007 12/01/2007 12/01/2007 12/02/2007	04/12/2008 04/12/2008 04/15/2008 04/15/2008 04/25/2008 04/25/2008 04/25/2008 06/22/2008 06/22/2008 06/22/2008 06/22/2008 06/22/2008 06/26/2008 07/05/2008 07/05/2008 07/07/2008 07/07/2008 07/07/2008

TABLE 3.1 - Continued

1

NOTES.—Column 2 denotes the fraction of the night allocated to the SFP survey: F = Full night, H = Half-night, Q = Quarter-night. Column 3 observer codes are: CF = C. Farrington, DR = D. Raghavan, EB = E. Baines, NT = N. Turner, PJ = PJ Goldfinger, TB = T. Boyajian, YT = Y. Touhami. Counts

in Column 5 are data files obtained for this survey and are sometimes less than Column 4 values because some data was obtained for CIV work or for other

observers. Column 6 contains the baseline(s) used during the night. The numbers within the brackets are the POP setting for the telescopes.

The reduction process involves using the shutter sequences to subtract dark noise and balance the beams from each telescope, combining the beams from both sides of the beam combiner after normalization, then taking a Fourier transform of the data to work in frequency domain. Next, "good" fringes are identified by comparing the integrated power in a bandwidth around the fringe frequency to the power off the fringe, and the qualifying scans are low-pass and band-pass filtered to reduce noise. Then, fringe envelopes are fitted to each fringe. The detection of a second separated fringe packet is enabled by three summary fringe envelope plots, described below, as well as a visual inspection of each fringe envelope. The Interactive Data Language (IDL) code written for this reduction process is included in §C.3. The three summary fringe envelope plots are -(1) a normalized shift-and-add plot which determines the shift by cross-correlating each fringe with a reference fringe. (2) a normalized sum of the autocorrelation of each fringe envelope, which produces a symmetric plot with a central peak, and when there is a separated fringe, a second peak on either side, and (3) a simple normalized shift-and-add plot by aligning the peaks of each fringe envelope. While all three plots serve the same purpose, i.e., to enable detection of a second fringe when present, they have some differences, and experience has shown it useful to inspect all three. For example, when two fringes are found, the third plot above is the best for follow-up astrometry as it best preserves the directional orientation of the fringes allowing for differential brightness assessments and elimination of the 180° ambiguity of the other methods. However, the simple shift-and-add plot is often better at noise cancellation of weak side-lobes that are not separate fringes, while the cross correlation plots tends to accentuate them. Figures 3.5



FIGURE 3.5: Example fringe envelopes for a separated fringe packet binary HD 79096. The observation was made on UT 20070309 on the S1-E1 baseline. The summary plots in the top panel from left to right are the cross-correlation shifted sum, the autocorrelation sum, and the peak-align shifted sum. The bottom panels shows the 14 strongest individual fringe envelopes as determined by the integrated power in a bandwidth around the fringe frequency when compared to off-fringe power. The X-axis of all plots is the relative offset in the dither mirror position as represented by the count of the sampling interval bin, the Y-axis is the relative intensity of the fringe envelope. While all plots clearly show the double fringe envelopes, and the top left plot is most effective at identifying the magnitude difference and resolving the 180° ambiguity of the companion's position angle.

and 3.6 show examples of the plot included in *Appendix* B for each target observed in this effort. Figure 3.5 is an example of a star with separated fringe packets and Figure 3.6 is an example of an apparently single star. The top three panels of the plots show the summary plots discussed above and the bottom panels shows the individual envelopes corresponding to the 14 strongest fringes as determined by the integrated power method described above.

Overall, this effort largely yielded null results, with only a few resolutions of known pairs and no new detections. Table 3.2 lists the results of the SFP data reduction for each observation and Table 3.3 identifies the status of each target on every baseline observed. Only three stars showed definitive evidence of separated fringes – HD 3196 on 5 observations



FIGURE 3.6: Example fringe envelopes for a single fringe packet star HD 1461 observed on UT 20080727 on the S1-W1 baseline, in the same format as Figure 3.5. These plots do not show any evidence of a separated fringe packet.

over two baselines, HD 79096 on 10 observations along the S-E baseline, three of which were too wide to fit in the scan window, and HD 137763 on three observations along S1-W1, two of which were too wide to fit in the scan window. All detections pertain to known binary systems (HD 3196 is a 6.9-year SB2VB, HD 79096 is a 2.6-year SB2VB, and HD 137763 is a 2.4-year SB2VB), so no new companions were discovered by this survey. Additionally, HD 98230 (ξ UMa) is a quadruple system with a wide 1".6 separation pair, each component of which is a single-lined spectroscopic binary. Upon searching wide for fringes by moving the OPLE carts around the first fringe, the widely separated fringe was seen on several occasions, but they were always only a single fringe, indicating that the magnitude difference of a single-lined spectroscopic binary is perhaps too large for detection by the Array. Additionally, Bagnuolo et al. (2006) derived a visual orbit for HD 16739 (12 Per) using SFP detection at CHARA and C. Farrington detected SFPs for three more stars of this sample Farrington (2008) – HD 4676 and 202275, which are known SB2VBs, and HD 131511, which is a known SB1 with a photocentric-motion orbit for which Farrington saw a second fringe on one of three observations (Figure B.24 in Farrington (2008)).

TABLE 3.2: SFP Observations and Results

	TABLE 3.2	- Continuea					0.40 0.40		Entrary 0	00000				Dadrotion
000000000000000000000000000000000000	me	UT Date	ß	Baseline	Obs	\mathbf{Red}	Date Reduced	s	Good	Total	FF	$\langle {\bf F} {\bf W} \rangle$	$\sigma_{\rm FW}$	Notes
0 0	846 846	09/17/2007		W1-S2	CF	DR	11/14/2007	ωu	199	199	65 103	12.0	6.4	Multi-peaked FE in most scans; asymmetric SFE
0 0	540 160	09/18/2007 08/15/2007		E2-52 E1-51	L T A C	н 1 1 1 1 1 1 1	11/12/2007	n n	194 198	102	103	1.b 44 1	0.8 21.6	Multi-peaked FE in most scans; asymmetric SFE
	160	09/17/2007		W1-S2	CF.	DR	11/14/2007	n v	198	199	156	10.6	6.1	
	160	09/18/2007	1	E2-S2	ЪJ	DR	11/14/2007	S	203	207	102	5.8	4.6	
	287	08/15/2007		E1-S1 E1 -S1	DR DR	DR DR	11/12/2007	0.' D'	207	232	102	0 0 0 0	1.8	Noticeable double peaks in SFE plots
	673	10/31/2007 08/15/2007		E1-31	H U H U	R L L L L	11/12/2007	n c	198	108	102	9.0 30.6	0.0 15.4	LOW DINK, WEAK ITINGE Rumny FR. multi-meaked SFR
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	673	09/27/2007		E2-S2	ΞL	DR	11/14/2007	ζω	199	199	65	14.3	5.7	Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	765	08/15/2007	ч	E1-S1	$_{\rm DR}$	$_{\rm DR}$	11/12/2007	S	198	198	102	25.6	16.1	
	765	09/17/2007	·	W1-S2	CF.	DR	11/14/2007	0 0	206	206	65	23.0	15.2	
	765 765	10/27/2007		EZ-SZ 51 C1		אר אר	11/14/2007	νo	205	206	65 1 E E	11.7	0.0 0	Multi-peaked FE in several scans; asymmetric SFE
	202	10/31/2007	- c	12-13			11/14/2007	ΩIJ	007 806	009 016	103	ч с 1 с	0 F	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	382	10/31/2007 09/17/2007	۰.	W1-S2	CF CF	DR	11/14/2007 11/14/2007	າທ	232	232	107	6.6	- 8.0	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	382	09/18/2007		E2-S2	ΡJ	DR	11/14/2007	S.'S	180	452	102	0.9	0.5	 Multi-peaked FE in most scans; two peaks in SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	382	07/07/2008	П	m W1-S1	DR	DR	07/07/2008	S	154	246	150	2.6	1.2	
	382	07/08/2008	1	W1-S1	$_{\rm DR}$	$_{\rm DR}$	07/08/2008	S	146	324	150	2.3	1.4	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	143	09/17/2007	1	W1-S2	$_{\rm CF}$	DR	11/14/2007	S	206	206	65	11.1	5.8	Multi-peaked FE in most scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	143	10/31/2007	1	E1-S1	$_{ m DR}$	$_{\rm DR}$	11/14/2007	S.'	230	258	102	2.2	1.0	Multi-peaked FE in several scans; double-peaked SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	143	11/01/2007	1	W1-S1	DR	DR	11/15/2007	S.'	210	307	102	1.5	1.0	Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	143	07/23/2008		E1-S1	TB	DR	07/24/2008	s l	160	224	100	2.5	1.1	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	032	11/01/2007	- 0	EI-SI	HU HU	HU HU	11/15/2007	N o	157	355	102	1.1	0.7	Low SNR; Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	101	1006/06/11/11/2007	۰- ۱	10-1 M	4 E	4 A C	2006/61/11	οv	077	000	102	с н г с	ით ით	Multi-peaked FE III Several Scaus; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	57	09/16/2007	- 0	W_{1-S1}	CF C	DR	11/12/2007	n or	205	216	102	4.9	4.0	 Multi-peaked FE in several scans
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	57	09/16/2007	ŝ	E1-S1	CF	DR	11/12/2007	S	203	229	102	2.9	2.0	Multi-peaked FE in several scans
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	03	09/17/2007	1	W1-S2	CF	DR	11/14/2007	S	217	239	102	2.7	1.7	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	03	09/18/2007		E2-S2	ЪJ	DR	11/14/2007	S C	185	206	102	2.5	1.3	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	94	09/27/2007	- 0	E2-S2		н Ч Ц	11/14/2007	νc	199	199	65	24.1	11.5	Multi-peaked FE in several scans
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	94 94	09/27/2007	1 03	E2-32 E2-S2	t L Z	DR.	11/14/2007 11/14/2007	ດທ	198	198	102	32.7	10.9 18.2	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	94	11/01/2007	1	E1-S1	DR	DR	11/15/2007	S	213	273	102	3.1	2.8	Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	94	11/01/2007	2	W1-S1	DR	$_{\rm DR}$	11/15/2007	S	198	200	102	11.8	9.0	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	65	11/01/2007	- 0	E1-S1	DR	DR	11/15/2007	S CO	175	371	102	1.1	0.7	Low SNR; Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0.5	11/01/2007	N -	W1-S1	н Ч Ц	אט מע	11/15/2007	νo	216	450	102	1.1 6.3	0.7	Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	6 I	11/01/2007	- 0	W1-S1	H H H	H H H	11/15/2007	o v	240 213	283	102	6.0 6.1	0.0	Multi-peaked FE in several scans; asymmetric of E Multi-meaked FE in several scans
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	49	09/27/2007	. –	E2-S2	ΕZ	DR	11/14/2007	ŝ	201	208	156	3.4	1.5	Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	49	09/27/2007	7	E2-S2	ΤN	$_{\rm DR}$	11/14/2007	S	199	199	102	11.7	6.7	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	62	09/27/2007	0	E2-S2	£ E	DR	11/14/2007	0 0	196	202	102	6.2	3.3 9	Multi-peaked FE in several scans
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	6	09/27/2007	ი -	E2-S2 E1 51		H H H H H H H	11/14/2007	n f	211	220	102	4.5	9 i 9 i 9	Multi-peaked FE in several scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	n x	11/01/2007		W1-S1	4 G G G	R R C	11/12/2007	n v	24/	309	00 100	- 0. 1. 0.	0.0	Nuuri-peakeu r.b. iii severai scaiis; Asymmetric SRF
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	200	09/16/2007	2	E1-S1	CF.	DR	11/12/2007	S.S.	208	359	102	1.3	0.8	Asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	60	08/20/2007	1	E1-S2	$_{\rm CF}$	DR	11/12/2007	S	200	201	102	7.3	4.0	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	60	09/16/2007	1	$\rm W1{-}S1$	CF	DR	11/12/2007	S	198	225	102	3.5	2.6	
$ \begin{array}{l c c c c c c c c c c c c c c c c c c c$	60	09/16/2007	0,	E1-S1	CF CF	DR G	11/12/2007	8	197	226	102	2.7	5.0	Asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	90 90	00/20/20/2007		E2-52 F9_63	a E	אם ב מ	11/14/2007	. ç	199 206	222	100	- 7	1.4 1.0	Multi-peaked FE in several scans; asymmetric SFE Multi montod EF in sourced scans, asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	96	10/31/2007		E1-S1	DB	DR	11/14/2007	, v	199	199	102	4.6	1.5	Multiproved F.E. III Several Scaus, asymmetric DF E
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	96	10/31/2007	0	$ m W1{-}S1$	DR	DR	11/14/2007	S	202	203	102	9.2	6.6	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	57	09/26/2007		E2-S2	ΤB	DR	11/14/2007	S	198	200	156	11.1	5.2	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	21	09/27/2007	N 7	E2-S2		нц	11/14/2007	N C	196 816	198	102	7.0	5.4	Clouds during after-shutter sequence
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	- 1-2	10/31/2007	- ¢	W1-S1	H C	R C	11/14/2007	n v	203	203	102	0.0	0.4 14.1	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$. 22	09/16/2007		W_{1-S1}	CF C	DR	11/12/2007	n or	200	233	102	2.7	1.9	Asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	65	09/16/2007	101	E1-S1	CF.	DR	11/12/2007	S.S.	211	287	102	1.9	1.4	Asymmetric SFE
13 10/31/2007 1 E1-S1 DR DR 11/14/2007 S 203 212 102 3.4 1.7 13 10/31/2007 2 W1-S1 DR DR 11/14/2007 S 196 203 102 7.5 4.3 23 09/56/2007 1 E2-S2 TB DR 11/14/2007 S 198 304 156 1.6 1.1 Multi-peaked FE in several scans; asymmetric SFE	13	09/26/2007	1	E2-S2	$^{\mathrm{TB}}$	DR	11/14/2007	S	196	219	156	3.2	1.9	Multi-peaked FE in several scans; asymmetric SFE
13 10/31/2007 Z W1-51 DK DK 11/14/2007 S 199 203 102 7.5 4.5 23 09/5007 1 E2-S2 TB DR 11/14/2007 S 198 304 156 1.6 1.1 Multi-peaked FE in several scans; asymmetric SFE inned on Next Pare	13	10/31/2007	0	E1-S1	DR	DR	11/14/2007	s s	203	212	102	3.4 1.4	1.7	
ана од жај да се из кај	13	10/31/2007		W1-S1 E2-S2	лн Л	ли И	11/14/2007	N V	196 198	203 304	156	7.5 1.6	4.3	 Multi-nealed RR in coveral scans: asymmetric SRR
	tinned	on Next Page.	•		1			2	0	100	001	0.1		T THE OUTPOINT AND A DAMAGE TO THE ADDRESS OF A DAMAGE AND A

I	1	l																																																						8	5
	Reduction Notes	:	 Multi-naabad RR in most scans: asymmatric SRR		Asymmetric SFE		Good SNR	Good SNR		 Multi-naabad RR in samaal seans: sermmatric SRR	Multi-peared Fil III several seates, asymmetric DF E		Multi-neaked FE in several scans: asymmetric/double-neaked SFE		Second peak (0.95) at $\sim +25$, one more small bump (0.4) $\sim +55$	Multi-peaked FE in several scans			Poor SNR, bad PS	Distinct second peak in SFE plots	Weak fringe, shutter seq very uneven, but clear double peak seen!	No secondary seen in SFE here!	Multi-peaked FE in several scans			 Shame alasa damhla maab libo athan ahsammatians af this night	Duows crose uounte pean inte ouner observations of vints might Multi-neaked FR in several scans			Multi-peaked FE in several scans; asymmetric SFE			Low SNR; Multi-peaked FE in several scans; asymmetric SFE		Asymmetric SFE Asymmetric SEE: Asymptotic ACODD alot	Asymmetric of D; uouble-peaked COOMIN prot	Multi-peaked FE in several scans: asymmetric SFE	Low SNR; Multi-peaked FE in several scans; asymmetric SFE	Multi-peaked FE in several scans				Door CND mode fuirme in mour conne	T OOT DIVIL, WOON HINGS IN MOUNT POWER	Weak fringes: hint of secondary peak in SFE	Multi-peaked FE in several scans	۰ ۱	Weak fringe;		 CODB has sub nooks closed to mimany hut OK fouthon out	COUNT has sub-peaks close to primary, but ON laturer out Asymmetric SFE	Hint of second peak (0.4) at ± 20 like other obs of this night		Multi-peaked FE in several scans	Foor SNR, many scans nave a weak iringe		
	$\sigma_{\rm FW}$	2.3	0 r 0 r	0.7	1.0	2.6	:	:	. a : a	о и о и	12.1	27.8	0.8	3.0	:	1.6	3.6	1.8	1.0	:	:	. h . (0.01 1.10	0.10	о т о т	# 5	. 6 . 6	0 00 0 00	1.5	1.5	8.9	16.9	0.7	7 X	н н 4 г	4.0	0.9	0.7	2.8	:	0.0 0.0	9.9 9	:	2.5	i :	1.6	2.0	1.0		4. C	2.4		:	14.7	1.4 4.0	2.1	
	$\langle {\bf F} {f W} angle$	4.2	0 0 0 10	1.3	1.9	4.9	÷	:		10.0	19.9	61.7	1.5	5.6	:	2.8	7.0	2.9	1.5	:	:	. 1 : 1	10.0 1	0.02	0.0T	0.0	5.7	6.7	2.3	2.0	15.0	26.4	1.2	0.0	- 1 7 7	0 10 9	1.5	1.0	4.7	:	10.7	7.7	:	1.4		2.4	3.2	1.7	. 0	0.0	2.5) .	:	20.5	1.4 5.9	3.1	
	FΓ	102	707 929	102	102	102	155	155	155	103	102	102	102	102	155	102	102	102	102	155	155	155	102	707	102	155	102	102	102	156	102	102	156	102	100	102	156	102	102	155	102	100	1 5 0 1 2 0	100	155	102	102	102	155	100	100	155	155	102	102	102	
	Scans Total	218	010	343	249	202	201	206	238	108	199	198	333	203	643	240	206	222	275	268	588	252	66T	06T	0000	044	707	198	266	293	203	214	373	200	206	202	280	543	204	234	198	018	161	889	208	261	214	291	256	202	417	362	253	200 265	200 214	232	
	Fringe 5 Good	213	012	201	204	197	200	202	204	107	199	198	243	200	210	217	201	195	185	210	220	224	100	100	190	50%	201	198	222	201	203	200	213	199	190	200	198	224	200	211	198	312	010	231	197	211	201	210	203 100	145	140 272	206	203	200	199 207	207	
	ß	ωu	οv	n w	s	S	s s	N C	νo	οu	2 02	n oc	S S	ŝ	D?	S	S	s	S	0.5 D		N.	ΩC	οo	ου		ζ α	n vo	S	s	s	S	S. 5	νc	n S	i v	ŝ	S?	s	s	s s	νo	ου	2 00	s S	s so	S	S	ωc	οu	o o	S.'S	S	ທເ	oα	ŝ	
	Date Reduced	11/14/2007	11/14/2007	11/14/2007	11/14/2007	11/14/2007	05/09/2007	05/09/2007	05/09/2007	11/14/2007	11/14/2007	11/14/2007	11/14/2007	11/14/2007	05/09/2007	11/15/2007	12/05/2007	12/05/2007	12/05/2007	05/09/2007	05/09/2007	05/09/2007	2008/20/01	10/02/00/21	10/02/00/21	1002/00/21	11/15/2007	12/05/2007	12/05/2007	11/14/2007	11/14/2007	11/14/2007	11/14/2007	11/15/2007	11/14/2007	11/15/2007	11/14/2007	11/14/2007	11/15/2007	05/09/2007	11/12/2007	06/09/2008	02/00/2001	06/05/2008	05/09/2007	11/15/2007	12/05/2007	12/05/2007	05/09/2007	/007/01/11	06/05/2008	05/09/2007	05/09/2007	11/15/2007	12/05/2007	11/15/2007	
	\mathbf{Red}	DR	H H H	DR	$_{\rm DR}$	DR	DR	ЦЦ	DR H		DR	DR	DB	DR	$_{\rm DR}$	DR	$_{ m DR}$	$_{\rm DR}$	DR	DR	ЧЦ	ЦЦ	אר ה ה		ה ה ה		H H H	DR	DR	$_{\rm DR}$	$_{\rm DR}$	$_{ m DR}$	DR	нц	ת ת ה	H H H	DR	$_{\rm DR}$	DR	$_{ m DR}$	DR	a c C	ה ה ה	55	DR	DR	$_{\rm DR}$	$_{ m DR}$	DR	н ц	38	DR	$_{\rm DR}$	DR rr	DR DR	DR	
	Obs	DR	2 G	PJ	$_{\rm DR}$	DR	E E	H.F.	DR	a E	DR	DR	DB	DR	EB	$_{ m DR}$	CF	CF	EB	EB	БН	нц	בים ביום	50	ч С Г			CF	EB	TB	$_{\rm DR}$	$_{ m DR}$	1B	лн н	ם ב	H H H	ΠB	$_{\rm DR}$	$_{ m DR}$	$_{ m DR}$	CF CF	2 L L	0 0 0 0	D.B.C.	EB	DR	CF	CF	CF C	אם ר מים	DR.	EB	CF	DR 'c	БB	DR	
	aseline	E1-S1	05-1M	E2-S2	E1-S1	W1-S1	E1-S1	E1-S1	E1-S1	E2-52	E1-S1	M^{-S1}	E1-S1	W1-S1	E1-S1	W1-S1	E2-S2	E2-S2	E2-S2	E1-S1	E1-S1	EI-SI		20-20	10-21 20-21	F1_C1	1S-17	E2-S2	E2-S2	E2-S2	E1-S1	W1-S1	E2-S2	WI-SI	20-24 101 01	W1-S1	E2-S2	E1-S1	W1-S1	E1-S1	E1-S1	WI-SI	10110	W1-S1	E1-S1	W1-S1	E2-S2	E2-S2	E1-S1	10-1 M	W1-S1	E1-S1	E1-S1	W1-S1	E2-52 E2-S2	w_{1-S1}	
	s B	1	7 -		1	61		. 17				- 2		5	1	5	1	5	1		. 17		-	о л О	 - c			- 0		1	1	5		 	N -			1	1	1			10	c		- 7	1	2		N -			1	 н,	- 0	-	
nued	ıte	200	1002	200	200	200	200	200	200	1002	200	007	007	200	200	200	200	200	200	200	200	200	100	100	200	200	200	007	200	200	200	200	200	200	100	200	007	200	200	200	200	200	200	008	200	007	200	200	200	100	008	007	200	200	007	007	t Page.
2 – Contii	Obs UT Da	10/31/2	7/10/01/2/01	09/18/20	10/31/2	10/31/2	02/25/2	02/25/2	03/11/2	7/07/60 07/00	10/31/20	10/31/20	10/31/20	10/31/20	01/25/2	11/01/2	11/19/2	11/19/2	11/27/2	01/25/2	01/26/2	04/14/2	7/10/11	7/61/11	7/61/11 10/20/11	10/20/11 01/22/01	11/01/20	11/19/20	11/27/20	09/26/20	10/31/2	10/31/2	09/26/2	11/01/2	7/07/60	7/10/01	09/26/20	10/31/20	11/01/2	04/14/2	09/16/2	04/26/2	10/20/20	02/00/2	01/25/20	11/01/2	11/19/20	11/19/2	02/04/2	7/10/11	04/26/20	01/26/20	02/03/2	11/01/2	11/27/20	11/01/2	d on Next
TABLE 3.	HD Name	026923	020923	029883	032850	032850	032923	032923	032923	0362620	032923	032923	035112	035112	037008	037008	037008	037008	037008	037394	037394	037394	03/394	460/00 100400	1027204	126220	038230	038230	038230	038858	038858	038858	040397	040397	041503	041593	042618	042618	042618	046588	046588	040588	051419 051410	051419	051866	051866	051866	051866	054371	054371 054271	0.54371	055575	055575	055575	055575 055575	059747	Continue

ľ	1																																																				C	Ĩ
	Reduction Notes	Hint of secondary peak in SFE	Double-peaked PS				Uneven shutter sequence	Hint of a bulge at sep ~ -40 in CCOR		:	Noisy PS		Deer UND and much filmer	I DOL DIVIL WILL WEAK ITTIGE Hint of a comp close to primary	Companion clearly seen	Companion clearly seen	Companion clearly seen	Companion clearly seen	No comp seen at this epoch!! Probably too wide	Companion clearly seen	Many scans have only one fringe	T	Double fringes too wide – this is A Double fringes too wide – this is A	Double fringes too wide – this is B	ACOB flat part a little wayv	Slight hint of comp peak in SFE	:	:		:					 Comione double nook eimiler to other cheamatione of this date	Spurious double peak, summar to other observations of time date	Strong fringe		Small bump (0.35) at ± 65 in ccorr sum			Slight asymmetry on the right on both shift-and-add plots	:			 A component, no fringe found for B			:			Primary fringe on the right end of window	Primary tringe on the left end of window	
	$\sigma_{\rm FW}$	 4.6	1.2	2.1	1.3	1.1			:	1.9	0.0	2.5	:				:	:	:	:	1.6	0.1. 0.0	0.0 0.0	510	1		1.2	1.7	1.5	ο. - c	7	1.7	3.1	2.7	L.4		8.9	16.6		1.4	:	:	4.7	1.1	11.0	1.7	4.5	2.6	1.5	1.5 1.5	: :	÷	:;;	:
	$\mathbf{F}\mathbf{W}\rangle$		1.2	2.2	1.0	1.3 3.7			÷	2.7	1.0	2.2	:	: :			:	:	:	:	2.2	1.7	- o c	5.7	1		1.9	1.8	1.5	ю и г	0.1	3.3	3.1	2.8	1.4	: :	12.9	13.8		+ 8°0	÷	:	5.4	1.2	3.U 16.7	10.7 2.4	6.4	1.9	1.4	1.4 1.4		÷	: :	
	FF <	155	150	100	100	100	155	155	155	156	150	31	200	155	155		155	155	155		102	150	150	150	22	155	102	100	150		155	102	100	150	100	155	102	001	155	700	155	155	100	001	150	150	150	001	150	150	155	155	155 155	201
	ns tal	31	- 67	56	59	17	20	0.0	21	40	57		1 10	- 20	89		01	. 66	81		89		12 00	108		53	67	34	97	 	00	15	21	41	100	0.22	02	86	44	23	06	59	35	60 i	10	46	44	24	1 5	52	03	45	18	2
	ge Sca I To	с, с	1 03	0	9	ω 4	10	101	0	0	n d	00	Ин	04	• 4		7	4	ŝ		010	21 0	N C	10	10	100	0	ŝ	010	1 1	- 0	101	ŝ	010	2 12	103	0	ŝ	C1 0	s oc	0	0	0	n (21 0	10	0	4.0	m 0	100	0	00	10	1
	Frin Good	195 200	158	167	245	$141 \\ 284$	202	203	200	197	128	242	202	194	199		199	217	208		221	164	183	185	192	203	210	202	153	211 211	1186	206	223	168	2006	200	201	289	204	243	218	213	200	157	189	192	230	215	164	173	198	206	661 199	1
	ß	ς. α	n oc	S	s s	n n	n vi	S.2	S	S	w a	Ωŭ	υŝ		; D	D	D	D	S	D	Ωō	Ωŭ	Ωŭ	o o	n oc	ŝ	S	S	s s	ΩŪ	o o	n w	S	S CO	υŝ	v	S	S	S. S	. v	S	S	S	s s	ωu	ດທ	S	ω	νc	υთ	ŝ	Ωŭ	v %	2
	ate uced	/2007	/2008	/2008	/2008	/2008	/2007	/2007	/2007	/2007	/2008	2008/	1002/	/2007	/2007	/2007	/2007	/2007	/2007	/2007	/2007	/.2008	0007/	/2008	/2002	/2007	/2007	/2008	/2008	2006/	/ 2005	/2007	/2008	/2008	2006/	/2007	/2007	/2008	/2007	/2008	/2007	/2007	/2008	/2008	/2008	/2008	/2008	/2008	/2008	/2008	/2007	/2007	/2007	
	$_{\rm D_i}$ Red	05/05	04/21	04/21	06/05	04/21	05/06	05/06	02/06	12/05	04/21	06/04	20/00	0.5/05	05/06	05/06	05/05	05/05	05/05	05/06	12/05	04/21	12/20	04/21	05/10	05/10	12/05	04/21	04/21	06/05	05/10	12/05	04/21	04/21	01/21	05/10	12/05	06/05	19/05	06/05	05/10	05/21	06/30	10/20	04/21	04/21 04/21	04/21	04/21	04/21	04/21	05/10	05/10	05/10	/ ~ ~
	\mathbf{Red}	DR DR	DB	DR	GB	H C H C	DR	DR	DR	DR	DR	CB CB	ה ה ה	DR	DR	DR	DR	DR	DR	DR	DR	ЧЧ		H C	H C	DR	DR	DR	DR	H C		DR	DR	DR	א ח ה	DR.	DR	CB	HU HU	CB CB	DR	DR	CB	GB	HU AU	DR	DR	DR	DR	Ч DR	DR	DR	л И И И И	1
	Obs	TB	H H H	DR	DR	н Н П Н П	EB	TB	$^{\mathrm{TB}}$	ΤB	DR	ЦК	1 C	E E E	H H	TB	DR	DR	DR	DR	ΤB	ЧЧ		UR DR	E E E	TB	$^{\mathrm{TB}}$	DR	DR	ה ה ה	4 H H	TB	$_{\rm DR}$	DR	д ц И	DB	$^{\mathrm{TB}}$	DR	н Ц Ц Ц Ц	DR	DR	DR	DR	DR	H L L L L L	DR	DR	DR	DR H	ч Л В Л	EB	DR	лв ЛВ	;
	line	S1 S1	5	S1	S1	12 IV	S1	S1	S1	S_2	S1	225	10	1 5	S12	S1	S1	$\mathbf{S1}$	S1	S1	S2	ν. Γ	5 5	1 5	s s	SI	S_2	-S1	S1	10 I	10	S2	-S1	SI	2 10 10		S_2	S1	s. SI	4 K	-S1	$\mathbf{S1}$	S2	S1	2 N	S1	S1	S1	5 S1	21 S1	S1	S1	2 Z Z	1
	Base	E1-	-1-M	E1–	W1-	EI- M1-	E1-	E1-	E1–	E2-	-1 M	- 5	- - - - - -		Е1- 1-	E1-	E1-	E1-	E_{1-}	E1–	E2-	- N		1 1 1	12	E1-	E2-	W1-	Е1- 1	-13	L 2	E2-	W1-	-1- 1- 1-		M1-1	E2-	W_{1-}	- 13 1 - 13	N1-	W1-	E1-	W1-	- 1 2 1	E1-	E1-	$W1^-$	W1-	- 13 6	 	E1-	E1- E	 5 E	1
	ß	1 -		-			- 0	-	7	e		- 0	10			2	1	0	5	1	0,						1	1					1	,	- 0	o	1				1	7	1				-	0	cı -			- 0	- 1	•
ontinued	Obs CDate	20/2007	12/2008	13/2008	26/2008	26/2008	06/2007	25/2007	25/2007	29/2007	12/2008	25/2008	20/2007	007/2007	25/2007	25/2007	09/2007	09/2007	10/2007	11/2007	29/2007	12/2008	14/2008	14/2008	002/2002	25/2007	29/2007	12/2008	13/2008	13/2008	2002/02	29/2007	12/2008	13/2008	13/2005	24/2007	29/2007	26/2008	06/2007	26/2008	24/2007	17/2007	24/2008	25/2008	14/2008	13/2008	15/2008	12/2008	12/2008	13/2008	06/2007	21/2007	21/2007 08/2007	
3.2 - C	LU	03/: 11/(04/1	04/.	04/:	04/. 04/5	02/(02/7	02/.	11/2	04/.	04/:	/70	02/(02/2	02/2	03/(03/(03/.	03/2	11/:	04/.	04/1	1/10	02/(02/2	11/	04/.	04/.	04/	02/0	11/2	04/.	04/.	04/.	04/5	11/2	04/;	02/	04/2	04/5	05/2	06/2	:/90	04/	04/1	04/5	04/:	04/.	04/j	02/(02/:	03/0	122 222
TABLE	HD Name	063433	063433	063433	065430	065583	067228	067228	067228	067228	068017	068017	062200	079096	0200020	079096	079096	079096	079096	079096	079096	079096	070006	060610	020000	079969	079969	079969	079969	080715	082443	082443	082443	082443	082443	082885	082885	082885	087883	087883	089269	089269	090343	090343	094765	096064	096064	097334	097334	097658	098230	098230	098230	

																					-add plots																											of the night				2	
	Reduction Notes	Hint of peak in SFE	Hint of peak in SFE	Primary fringe at left edge	s summary minge at fight edge Servo-tracked primary fringe		Comp1; weak fringe,	Comp2	Comp2				Hint of second peak in SFE	A component	 R component (99492)	n component (20102)					Slight asymmetry on the right on both shift-and	Bumpy PS	inuisy uata			Noisy data, weak fringe	Weak fringe, poor PS, SFE shows double-peak	Climbt commutur of chiffod DD cum plate	Sugni asymmetry of smitted F E sum proce		Slight asymmetry & secondary peak in SFE	Slight bump in SFE		Poor SNB. Multi-peaked FE in several scans						Hint of second peak in SFE	:		 Weak fringe, poor PS		Weak fringe, poor PS		 Noisv data: Second peak in SFE	SFE shows multiple peaks, similar to other obs	:	 Hint of a second neak in SFE		Timt of a correct in SER	THE OF GOOD AND A STATE AND A STATE
	$\sigma_{\rm FW}$:	:	:	: :	16.5	0.9	9.7 19 E	17.0	2.0	2.8	6.9	. (. (5 7 7 7 7 7	0.01 6.8	000		:	:	:		1.9 0	0.0	25.5	:	:	0.0	1.4	: :		:	:	: 0	4.7 1.0	1.3	4.0	1.3	0.9 4.6	:	:	:	:			0.9	:	0.6	2.6	1.2	7.7	2.5	0.9	:
	$\langle {f F} {f W} angle$:	:	:	: :	27.0	1.1	10.0	15.7	2.0	4.7	8.5	:	4.1	1.07	13.5		:	:	:	: 0	7 0 7 0	0.0	34.1	:	:	1.2	1.9	: :	: :	:	:	: 0	0.9 1.4	1.4	4.6	1.7	2.0	:	:	:	:	1.3		1.3	:	0.8	3.0	1.5	7.0	2.8	1.0	:
	ΕF	155	155	155	155	156	156	156	100	100	150	150	155	150	150	150	155	155	155	155	155	100	150	150	155	155	156	1 1 00	155	155	155	155	155	156 156	150	150	150	100	155	155	155	155	156	155	156	155	156	100	100	100 155	150	150 155	7007
	Scans Total	198	208	217	215	213	524	214	359	270	216	248	302	215	204	202	331	276	234	306	269	322	935	237	356	459	401	309 26.6	244	269	343	278	240	372	321	281	288	257	209	208	202	242	355	290	326	207	197	289	271	207	267	336 245	08.7
	Fringe Good	198	205	214	213	210	217	2112	208	160	205	214	200	184	677	219	203	206	212	226	213	2.76	006	233	222	204	207 215	215	221	219	237	235	223	222	174	244	187	214 196	199	201	197	2.7.2	185	199	187	196	214	223	160	166 204	192	144 108	FOC
	ß	S?	D?	ω ⁰	. v	S	s: S:	n S	υ α	o o	ŝ	S	S?	s c	οœ	2 07	ŝ	S	S	S	s s	ωo	o o	n v	S	s l	o	u fi	i v	ŝ	S?	S.'	s c	o w	S	S o	νc	o w	S	S^{2}	S C	νo	ດທ	S	S	ωu	S.'S	S?	S C	n ç	ς Ω	ω f	i
	Date Reduced	05/10/2007	05/10/2007	05/21/2007	05/21/2007	06/27/2007	06/27/2007	06/27/2007	06/27/2007	04/21/2008	04/21/2008	04/21/2008	05/10/2007	04/21/2008	04/21/2008	04/21/2008	05/10/2007	05/10/2007	05/10/2007	05/10/2007	05/21/2007	04/21/2008	04/21/2008	04/21/2008	05/10/2007	05/21/2007	06/27/2007 06/05/2008	002/00/00/00	05/10/2007	05/10/2007	05/10/2007	05/10/2007	05/10/2007	11/09/2007	04/21/2008	04/21/2008	04/21/2008	04/21/2008 04/21/2008	05/10/2007	05/10/2007	05/10/2007	05/10/2007	06/27/2007	05/10/2007	06/27/2007	05/10/2007	06/27/2007	06/05/2008	06/30/2008	07/01/2008 05/10/2007	04/21/2008	04/21/2008	100#/0T/00
	\mathbf{Red}	$_{ m DR}$	DR	DR DR	DR	DR	DR	HU AU	H H H	DR	DR	DR	DR	DR.	L L L	H H H	DR	DR	$_{\rm DR}$	DR	DR	HU AU	H C	DR	DR	DR	DR DR		UR H	DR	$_{\rm DR}$	DR	DR	DR.	DR	DR	лк ч	л Л Л Л Л	$_{\rm DR}$	$_{\rm DR}$	DR	HU HU	DR	DR	DR	HU HU	DR	CB	GB	а С В С С С С С С	DR	DR ar	יות
	Obs	$_{ m DR}$	DR	DR GR	DR	CF	DR	אר האר	H C	DR	DR	$_{\rm DR}$	DR	DR G	L L L	H C	EB	DR	DR	DR	DR	אט אטר	H L L	DR	DR	DR	H H H H H H H H H H H H H H H H H H H	אר היו	A C	DR	$_{\rm DR}$	ЪJ	ГЧ цс	DR	DR	DR	HU HU HU	DR.	EB	EB	DR	н Н Ц	DR	CF	DR	L L L	CF.	$_{\rm DR}$	DR	лц Г.	DR	DR 5D	1
	Baseline	E1-S1	W1-S1	E1-S1 E1 S1	E1-S1	W1-S1	W1-S1	W1-51	W1-51 W1-51	E1-S1	E1-S1	W1-S1	E1-S1	E1-S1	W1-31 E1-S1	M1-S1	E1-S1	E1-S1	W1-S1	W1-S1	E1-S1	W1-S1 51 61	E1-21	W1-S1	W1-S1	E1-S1	W1-S1	W1-51	E1-21 E1-S1	W_{1-S1}	E1-S1	W1-S1	W1-S1	W1-51 E1-S1	E1-S1	W1-S1	EI-SI	W_{1-S1}	E1-S1	E1-S1	E1-S1	W1-S1	W1-S1	E1-S1	W1-S1	EI-SI 51 SI	W1-S1	W_{1-S1}	W1-S2	E1-21 E1-31	W_{1-S1}	E1-S1 E1_S1	
	s	2	1	- 0	100	1	0	. 10	η - α			1						1	1	1		c	- ۱		1				- 6	. –	1		010	ч –				- 0	1	7		- 0	۱ ۲		-	- 0	. –			- 6	. –	- 12	+
- Continued	Obs UT Date	03/08/2007	04/24/2007	05/17/2007	05/17/2007	05/28/2007	06/01/2007	06/01/2007	06/01/2007 04/12/2008	04/13/2008	04/14/2008	04/15/2008	03/08/2007	04/14/2008	04/13/2008 04/14/2008	04/15/2008	02/06/2007	03/08/2007	04/24/2007	04/24/2007	05/17/2007	04/12/2008	04/12/2008	04/15/2008	04/24/2007	05/17/2007	06/01/2007	04/20/2008	04/11/2007	04/24/2007	04/14/2007	04/17/2007	04/17/2007	07/28/2007	04/14/2008	04/15/2008	04/14/2008	04/13/2008 04/13/2008	02/06/2007	02/06/2007	03/08/2007	04/24/2007	06/01/2007	03/12/2007	06/01/2007	04/03/2007	05/28/2007	04/26/2008	06/24/2008	06/25/2008 04/03/2007	04/12/2008	04/12/2008	on Nevt Page
TABLE 3.2	HD Name	098230	098230	098230	098230	098230	098230	098230	098230	098230	098281	098281	099491	099491	039491	09492	100180	100180	100180	101177	101177	101206	104304	104304	105631	105631	105631	108054	108954	108954	110833	110833	110833	110833	112758	112758	113449	114783	115404	115404	115404	115404	116442 116442	116443	116443	116956 116056	116956	116956	116956	116956 119332	119332	119332	Continued

	duction tes						low freq variation! Small second peak in SFE	scan has peak on the left edge	sral RR, show double-neak									vll second peak in SFE	all second peak in SFE	i shoulders in SFE plots	all peaks in SFE, probably just noise	ht asymmetries, but nothing more	" CND Multi mached EE in contant accura	TOTAL STATE DESCRIPTION FOR SEASON FOR SEASON STATES	3 shows multiple peaks - similar to other obs of the night	? shows multiple peaks - similar to other obs of the night	shows multiple peaks - similar to other obs of the night	shows multiple peaks	a nigner rrequency	ht broadening at base in SFE plots	y noisy data.	sy data, asymmetric sum plots) chama multipla maalaa - aimilan ta athan aha af tha nimht	s snows multiple peaks - similar to other obs of the might			t of second peak in SFE		ter part of signal scans drop in intensity - passing clouds?	rd low freq oscillation in Beam 5 data										. CND. CEE nlate cham hint of second neak	THIN DE E PIONS SHOW HIM OF SCORES POWER	- DC COODD what shows out work of a 1.25	Γ EQ, COULUL PION SHOWS SHUPPED as $\alpha_{V} \sim \tau \sigma_{U}$
	v No	:	:	:	: :		Big	Rei					:	:	:	-	: :	Sm	Sm	Min	Sn	, Sli	С	5	SF1	2 SFI	SEI SEI	SE	so use	Slig	5 Ver	ION (2	1 0 F			Hir (: :	Lat	We	:	:	: :	: :	: :	:	:	: :	:		:	Э	5 -
	$\sigma_{\rm FW}$:	:	:	6	1.8	:		οr N.C	5	4	21.1	24.(0.0	-		:	:	:	:	: 0	N -	15.5	5.0	2.2	01 0	x c	0.0		0.5	1.0	x, -		oo	1.8	0.0	7.0.1		:	:	. c	10.	1.9	5.2	000	0 C 5 M		:	5.0	: :	10.6	
	$\langle {\bf F} {\bf W} angle$:	:	:	15.1	1.8	:		0.9 9	100	3.9	25.9	56.2		14.8	0.1	: :	:	:	:	:	: •	4.4	23.1	2.3	3.0	3.4	10.4	0.0		0.8	2.4	10.4	2.0	12.0	1.8	1.1	0.2	:	:	:	. °	5.0	2.5	4.7	11.4	4 4	5.1	:	3.4	: :	11.5	1
	FF	155	155	155 155	100	100	155	155	100	100	100	100	150	155	1001	155	155	155	155	155	155	155	001	100	100	100	100	100	100	155	156	100	100	100	100	150	150	100	155	155	155	155 156	100	150	100	100	150	150	155	156 155	155	156 156	F CO
	cans Total	283	260	298	218	452	202	635	222	348	349	217	201	263	102	275	441	233	229	274	247	393	077	219	547	310	318	219	420 245	433	069	242	700	241	223	326	341	241 398	235	226	235	198	403	252	295	237	204	204	199	240 132	198	270 215	oto
	Fringe S Good	200	204	233	208	272	199	213	210	271	274	210	200	200	200	218	229	226	220	218	216	230	017	215	352	246	259	191	11/1	195	229	180	207	070 173	198	188	144	125	216	211	224	198	358	198	225	220	209 198	192	199	212 107	198	198 216	017
	s	S	S C	n n	o o	S	S	s s	n n	n w	ŝ	s	S C	N C	n n	a or	n N	S?	S	S.'	S 0	N. C	υu	n v	D?	D?			n v	n vo	S?	S.?	ν Ç	ی ا	S	s s		n w	S	S	s s	νu	n vi	ŝ	D?	s s	n v	o v	S	ωŞ	. ç.	ω į	ċ
	Date Reduced	05/10/2007	05/10/2007	05/10/2007 05/10/2007	04/21/2008	06/04/2008	05/10/2007	05/10/2007	06/27/2007	00/04/2008 07/01/2008	07/01/2008	07/01/2008	07/08/2008	05/10/2007	04/21/2008 06/04/2008	05/10/2005	05/10/2007	05/10/2007	05/10/2007	05/10/2007	05/10/2007	05/21/2007	11/00/21/2007	04/21/2008	06/04/2008	06/05/2008	06/05/2008	06/30/2008	07/01/2008	05/14/2007	06/27/2007	04/21/2008	04/21/2008	00/04/2008 04/21/2008	04/21/2008	04/21/2008	04/21/2008	07/01/2008	05/14/2007	05/14/2007	05/14/2007	05/14/2007	06/04/2008	07/01/2008	06/30/2008	07/05/2008	07/08/2008	07/08/2008	05/14/2007	06/27/2007	05/14/2007	06/27/2007	100/21/200
	\mathbf{Red}	DR	DR	H H H H H H H H H H H H H H H H H H H	DR	CB	DR	DR	H H H	38	CB	CB	DR	н Ч	H H H	a n B	DR	$_{\rm DR}$	$_{\rm DR}$	$_{\rm DR}$	DR	DR G	אר היו	DR	CB	CB	GB	9 E	98	DR	$_{\rm DR}$	DR	HU HU	DB.	DR	DR	лн	98	DR	DR	DR	אט	CB	CB	CB	DR	н И И И И	DR	DR	DR ar	DR	DR	ייר
	Obs	EB	DR	н Сн Сн С	DR	DR	DR	ΠB	н Ч Ч	EB.	EB	EB	DR	CF.	אר הר	H H H	PJ	ЪJ	$_{\rm DR}$	$_{\rm DR}$	DR	DR		DR.	DR	$_{\rm DR}$	DR	нц	H H H	DR	\mathbf{CF}	DR	ЯU Ч	DR.	DR	DR	лч	DR.	DR	DR	DR	אט	DR.	EB	$_{\rm DR}$	DR	H H H	DR	$_{\rm DR}$	DR ar	DR	DR	ייח
	Baseline	E1-S1	E1-S1	W1-S1 E1-S1	W_{1-S1}	E1-S1	E1-S1	E1-S1	W1-31 E1_S1	E2-S1	E2-S1	W2-S2	W1-S1	EI-SI	W1-51	E1-S1	W_{1-S1}	W1-S1	W1-E1	E1-S1	W_{1-E1}	E1-S1	W1-51	W1-S1	E1-S1	W1-S1	W1-S1	W1-S2	E1-21 E1-S1	E1-S1	W1-S1	E1-S1	W1-S1 51 51	E1-31 E1-S1	W_{1-S1}	W1-S1	EI-SI	W1-32 E1-S1	E1-S1	W1-S1	W1-E1	EI-SI W1 C1	E1-S1	W2-S2	E1-S1	E1-S1	W1-S1	W_{1-S1}	E1-S1	W1-S1 F1-S1	E1-S1	W1-S1	10-201
	s	2		- 6	101	1	1				0	1						2	1	1				- 6	-	1	01		- 6		1		cı -		5	1	°1 -			1					1	- 0	v –	101	1				- : •
- Continued	Obs UT Date	02/05/2007	03/08/2007	04/24/2007	04/13/2008	04/25/2008	02/16/2007	03/23/2007	05/31/2007	04/20/2008	06/08/2008	06/09/2008	07/08/2008	03/12/2007	04/14/2008 04/95/2008	04/14/2005	04/17/2007	04/17/2007	04/26/2007	04/14/2007	04/26/2007	05/16/2007	1002/28/2002	04/13/2008	04/25/2008	04/26/2008	04/26/2008	06/24/2008	06/25/2008	03/10/2007	05/28/2007	04/13/2008	04/13/2008	04/13/2008	04/13/2008	04/12/2008	04/12/2008	06/25/2008	04/11/2007	04/24/2007	04/26/2007	03/08/2007	04/25/2008	06/09/2008	06/23/2008	07/05/2008	07/08/2008	07/08/2008	03/08/2007	05/31/2007 02/16/2007	03/08/2007	05/31/2007	on Next Page
TABLE 3.2	HD Name	121560	121560	121560	124292	124292	124850	124850	124850	124850	124850	124850	124850	125455	125455 195455	127334	127334	127334	127334	128165	128165	128165	201021	128311	128311	128642	128642	128642	128642	130004	130004	130004	130004	130307	130307	132142	132142	132142 132142	132254	132254	132254	135204	135204	135204	135204	135204	135204	135404	135599	135599	136202	136202	Continued

TABLE 3.	2 - Continued												
$_{ m Name}^{ m HD}$	Obs UT Date	s	Baseline	Obs	\mathbf{Red}	Date Reduced	s	Fringe Good	Scans Total	FF	$\langle {f F} {f W} angle$	$\sigma_{\rm FW}$	Reduction Notes
136202	04/13/2008	1 0	E1-S1	DR	DR	04/21/2008	ωu	200	218	100	6.9 60.0	5.1 12.7	
136713	04/14/2008	1 1	W1-S1	DR	DR	04/21/2008	o o	201	202	150	10.8	4.9	
136713	04/25/2008	-	E1-S1	DR	CB	06/04/2008	S	271	384	100	2.4	2.1	
136923	04/13/2008	-	E1-S1	DR	DR	04/21/2008	S I	192	237	100	2.3	1.7	:
136923	04/13/2008	c1 -	W1-S1	DR A	DR ac	04/21/2008	υ Ç	158	275	100	9.6	x. x 5. 3	Three second is CED
137763	04/14/2008 04/25/2008		W1-51 F:1-S1	R U R U	H H	04/21/2008	Ξ v	1 99 2 6 2	202 405	1001	12.9 3.3	10.1 7 8	I wo peaks seen in SFE
137763	06/09/2008		W2-S2	EB	88	07/01/2008	n w	195	303	100	1.8	1.5	
137763	06/23/2008	г	E1-S1	DR	CB	06/30/2008	S	222	315	100	2.8	2.9	
137763	07/07/2008	1	W1-S1	DR	DR	07/07/2008	S	196	211	150	9.4	7.4	Fringe 1 of a wide SFP – too wide to fit in window
137763	07/07/2008	0,	W1-S1	DR	DR	07/07/2008	s s	203	208	150	11.9	7.0	Fringe 2 of a wide SFP – too wide to fit in window
137778	04/14/2008		W1-S1	HU HU	нц	04/21/2008	ν ⁰	200	200	150	13.7	0.0 0	ст. СББ аналистиција стора – ajmejjan 4.2 а4нас ана аб 44на ејени.
137778	04/25/2008		W_{1-S1}	н Ч Ц		00/04/2008	ŭ v	201	328 203	150	0.0 12.3	n x n y	SFE snows multiple peaks - similar to other obs of the night
137778	07/07/2008	10	W1-S1	DR	DR	07/07/2008	o o	200	202	150	13.5	2.6	
139323	04/12/2008		W1-S1	DR	DR	04/21/2008	ŝ	173	325	150	1.7	1.9	
139323	04/12/2008	7	E1-S1	$_{\rm DR}$	$_{\rm DR}$	04/21/2008	S	131	334	100	1.1	1.1	
139341	04/14/2007	1	E1-S1	DR	$_{ m DR}$	05/14/2007	S?	117	338	155	:	:	Poor SNR, scan stopped midway; weak fringe, noisy envelopes
139341	04/17/2007		W1-S1	Γ	DR	05/14/2007	S	223	350	155	:	÷	Hint of peak in SFE
139341	04/17/2007	4 0	W1-S1	L d	DR G	05/14/2007	Ωŭ	222	440	155	:	:	VB comp of $a=0.8$ " not seen!
140801 14061	1005/07/7002/	N -	13-1 M	ה ח ה	ה ח ה	002/14/2002	ου	077	000	100 1 100	:	:	VID COMP OI A=U.S. NOU Seen! Mini and month of 1 AF
139777	05/28/2007		W1-S1	4 C 4 G	H C H	0.0/14/2007	0.02	216	260	156	2.6		MIIII SUD-PEAK (0.9) at 740
139777	04/26/2008		W1-S1	DR	B	06/05/2008	n vo	234	416	100	2.5	3.0	
139777	06/24/2008		W1-S2	DR	CB	06/30/2008	S	203	219	100	14.0	11.1	
139813	04/26/2008	1	W1-S1	$_{\rm DR}$	CB	06/05/2008	S	218	329	100	1.8	1.6	:
139813	06/24/2008	1	W1-S2	$_{\rm DR}$	CB	06/30/2008	S	196	279	100	6.0	6.3	
139813	06/25/2008	1	E1-S1	DR	CB	07/01/2008	S	202	292	100	2.3	2.1	
141272	04/14/2008		W1-S1	DR	DR	04/21/2008	S C	201	202	150	19.8	8.0 0.0	
141272	04/25/2008		E1-S1	DК	CB CB	06/04/2008	s c	256	325	100	ເດັ 1 ເບັ	3.2 7 5	
142267	00711/2002		EZ-SI D1 21	אח	אר ה	11/12/2007	νç	210	332	156	۲.1 ۲.1	7.7 	Big spike in PS
142267	00/13/2007		E1-21	ч с ч с	R C	017/12/2007	. v	194	130 933	100	0.1 9 P	9 T	wusy uara, pout r.o. assymmetric
142267	04/13/2008	10	W_{1-S1}	DR	DR	04/21/2008	n w	209	212	100	38.2	26.5	
144287	04/12/2008		W1-S1	DR	DR	04/21/2008	S	199	273	150	3.0	3.2	
144287	04/13/2008	1	E1-S1	$_{\rm DR}$	$_{\rm DR}$	04/21/2008	S	175	258	100	2.1	1.8	
145675	04/24/2007	1	W1-S1	DR	DR	05/14/2007	S.?	222	245	155	:	÷	Weird low-freq noise in beam 5
145675	04/26/2007		W_{1-E1}	DR	DR	05/14/2007	S I	233	347	155	:	:	
1400/0	2006/21/2002/20		E1-01	ה ח ה	אר ר איר	2006/17/2002/02/27/20	u S	212	142	100 1 FE	:	:	$\mathbf{U}_{int} = e_{ini} \mathbf{F}_{nonle} \left(0, 0 \right) = e_{ini} \mathbf{F}_{nonle} \left(0$
146233 146233	03/06/2007		W1-S1	H C	H C	0.0/14/200/	. v	201	201	150	73.1	86	IIIII OI SUUPEAN (U.Z) at TOU
146233	04/25/2008		E1-S1	DR	en e	06/04/2008	s S	265	285	100	14.1	11.8	SFE shows multiple peaks - similar to other obs of the night
146233	06/23/2008		E1-S1	DR	GB	06/30/2008	ς Ω	246	279	100	10.4	10.0	
146361	05/17/2007	1	E1-S1	DR	$_{\rm DR}$	06/27/2007	S	214	216	156	13.4	8.1	
146361	05/17/2007	0	E1-S1	$_{\rm DR}$	$_{ m DR}$	06/27/2007	S	214	219	156	16.2	10.5	
146361	05/17/2007	ς. Έ	E1-S1	DR	DR	06/27/2007	s o	216	218	156	13.6_{-2}	9.0 5	
146361	05/17/2007	4 L	EI-SI	ЧЧ	H H H H H H	06/27/2007	νc	218	222	156	9.7 1.7	9.0 1	
140301	7006/21/2002/21/20	ດປ	E1-21	2 C	אם ר א	7006/26/90	no	912 210	239	150	7.0 2.7	0.0 0.0	
146361	03/11/2007	-1 C	E1-21	4 A C	R R C	100/2/2001	o o	208	700	156	0.4 7	0.0 7 0	
146361	05/17/2007	- 6	E1-S1	DR	DR	06/27/2007	o o	209	211	156	30.5	15.2	
146361	05/17/2007	10	E1-S1	DR	DR	06/27/2007	S	205	214	156	29.0	13.4	
146361	05/17/2007	12	E1-S1	$_{\rm DR}$	$_{ m DR}$	06/27/2007	S	212	215	156	26.8	11.3	:
146361	05/17/2007	14	E1-S1	DR	DR G	06/27/2007	N C	216	234	156	4.2	0 5 0 5	
140301	1005/25/90	c	E2-31	5 C	אם ר א	2006/26/20	νo	212	774	150	0 0 -	с.4 0.4	
146361	05/27/2007	1 03	E2-S1 E2-S1	CF CF	DR.	06/27/2007	o w	219	230	156		6.9	
146361	05/27/2007	4	E2-S1	CF	DR	06/27/2007	S	205	226	156	4.3	3.8	
146361	05/27/2007	ю	E2-S1	CF	DR	06/27/2007	S (194	325	156	1.6	1.2	
146361	05/27/2007 05/20/2007	9 -	E2-S1 F2-S1	CF DI	DR A	06/27/2007 06/07/2007	ω u	240 213	914 247	156 156	0.7	0.5 0.5	Tiny fringe, very noisy data
Continue	A nn Navt Page	-	TC-7.1	F G	חוו	100/21/200	c	717	117	TOO	4.0	4.4	
COMMINIA	DI TNAVE I ORE												

I		1																																																								(90 	
	Reduction Notes			 Weak fringe	Weak fringe		Double moderal CED	Double-peaked SFE	Many weak Iringes	INTAILY WEAK ILLINGS		Very weak fringes very low SNR	Very weak minibes, very iow since									Widening ACOR plot	Multi-peaked FE in several scans					:		Additional weak peaks in SFE	:		Foor tringe quality, multi-peaked FE in several scans			 Weak fringe faw good scans	Weak IIIIIge, Iew good scalls					Hint of additional peaks in SFE	Multi-peaked FE in several scans						Poor SNR	:		Widening ACOR plot Multi nooloof PD in most connection SDD	MULTIPEARED F P. III 111081 SCAUS; ASYMMETIC OF P.	 Side lobes on both sides of fringe in both sum plots				Poor SNR, broadened ACOR, hint of second peak in shift plot		
	$\sigma_{\rm FW}$	2.1	1.7 1.6	0.7	0.6	9.9	7.2	0.1 7	4 0 8 0	10.5	0.01	# 0 0	0.0		. m	10.7	- 20	0.0	, re	3.5	1.2	0.8	1.0	2.2	2.4		:	:	:	2.8	8.7	ი. ი	0.9 101	14.0 14.4	14.4 19.1	7.0	18.7	4.6	:	:	:	:	0.9	2.1	3.3	:	3.6	9.2	0.7	5.8	0.0	о и - С	 999	5.0	1.6	:	1.3	0.7	0.1	
	$\langle \mathbf{F} \mathbf{W} \rangle$	2.9	4 C	1.1	0.9	9.7	6.1	1 I 1 i	- 2 - 2	0.0	19.3	0.4T	;	:	4.0	0.4	25.3	5.4	, rc	4.8	1.0	1.5	2.0	2.3	2.8	:	:	:	÷	6.6	11.6	0.0	0.0 1	24.1 33 R	0.00 9	0.0	40.2	4.3	:	:	:	:	1.6	2.0	4.0	÷	3.4	11.9	0.7	3.0	9.1	11.6	6.11 7.6	0.4	1.7	:	2.2	1.0	2.0	
	ΕF	156	156	156	156	105	105	201	105	105	105	105	155	155	156	100	150	100	100	100	100	156	156	100	100	155	155	155	155	65	100	150	100	150	100	156	150	100	155	155	155	155	156	150	100	155	150	100	100	156	156	100 1	100	150	100	155	156	156	00T	
	cans Fotal	256 266	207 074	474	629	218	261 204	394 910	203	000	202	404 717	305	198	230	266	202	270	348	227	358	285	198	299	251	217	229	230	235	224	231	202	488	617 200	211	102	205	305	219	306	222	240	198	293	232	223	284	220	690	255	198	000	230	217	339	228	250	608	612	
	Fringe So food	211	522 110	231	231	206	204	100	174	500 200	100	193 015	203	198	201	202	202	238	192	192	129	193	161	197	184	215	217	218	217	224	209	199	1.00 0.10	212	102	007	201	256	218	225	222	215	146	173	189	216	204	196	180	197	198	217	107	194	186	221	211	230	199	
	s	s s	οv	n va	S	S	ω ;		Ωυ	00	o o	2 S	; v	2 02	n U	n or	n or	2 07.	o v	S	S	S?	S	S	S	S	S	s	S	S?	S I	Ω (n	οu	סמ	2 S	. v	n so	S	S	S	s	s	S	s	s	s	s	s	S I	s g	ŭ o	0.00	5.5	. v	S	s S	S.'	a	
	Date Reduced	5/27/2007	5/27/2007	5/27/2007	5/27/2007	1/09/2007	1/09/2007	1/00/2007	1/00/2007	1/09/2007	1/09/2001	1/15/2007	5/14/2007	5/14/2007	5/97/9007	1/21/2008	1/21/2008	5/04/2008	5/05/2008	5/30/2008	7/01/2008	5/27/2007	1/12/2007	4/21/2008	3/30/2008	5/14/2007	5/14/2007	5/14/2007	5/14/2007	1/14/2007	4/21/2008	$\frac{1}{21}$	0/ 04/2008	4/21/2008	#/ 71 / 2008	5/97/9007	1/21/2008	5/04/2008	5/14/2007	5/14/2007	5/14/2007	5/14/2007	1/12/2007	4/21/2008	3/30/2008	5/21/2007	4/21/2008	3/30/2008	7/01/2008	1/08/2007	1/08/2007	2005/17/002/17/0	1/21/2008	1/21/2008	5/04/2008	5/21/2007	5/27/2007	1/08/2007	1/110/2011	
	I per	DR 00	н с и с	OR 00	OR 00	JR I:	CR 1		н н н н	4 A			E E E	B B C	an a	2B 02	B B C	5 8 5 8	100	B B	CB 00	JR 00	JR I:	$0R = 0^{-1}$	CB 00	JR 08	JR 05	JR 05	JR 03	JR II	OR 0	DR B	n n n		d d		DB 00	e B	OR 05	DR 08	JR 0!	JR 05	JR I	JR 0 ²	CB CB	JR 03	$OR = 0^2$	B B	B	JR I	DR 1 2	אר ה היו	H Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z	DB DE		OR 05	OR 06	CR CR	л. т.	
	bs F	L L L L L L L		. L	PJ I	JR I	H A A	ביים ביים	ב ר ה				1 E		; e				E C	OR 0	OR	DR I	PJ I	DR I	OR (DR I	P.J I	PJ I	OR	GF	JR I	H L	בי בי בי				H H H	R C R C	DR I	P.J I	P.J I	JR I	PJ I	JR I	OR	OR I	OR	OR	OR	E I	E E	בים בים	- L		- HO	DR I	J. I	E E	- 40	
	3aseline C	E2-S1	E2-51	E2-S1	E2-S1	E1-S1]	E1-S1	E1-21	E1-01	E1-21	E1_S1	E9-W1	E1-S1	E1-S1	W1-S1	W1-S1	W1-S1	E1-S1	W1-S1	W1-S2	E1-S1 1	W1-S1	W1-S1	W1-S1	W1-S2]	E1-S1 1	W1-S1	W1-S1	W1-E1	W1-S2	W1-S1	W1-S1	EI-SI	TC-TA		E2-S1	W1-S1	E1-S1	E1-S1 i	W1-S1	W1-S1	E1-S1	W1-S1	W1–S1	W1-S2	E1-S1	W1-S1	W1-S2	E1-S1	W1-S1	E1-S1	10-1M	W1-S2	W1-S1	E1-S1	E1-S1 1	W1-S1	W1-S1	10-13	
	S	5	04	9	7	1	0 0	ກມ	റഴ	1 0				- 6	ı –	·		. –	. –	1	1	1	1	1	1	1	1	2	1	1	1	,							1	1	3	1	5	1	1	5	1	1	1	1	ი. -					1	1	0		
- Communed	Obs UT Date	05/29/2007	05/29/2007 05/99/2007	05/29/2007	05/29/2007	07/29/2007	07/29/2007 07/30/3007	1002/62/10	1002/62/10	1002/62/10	1002/20/2002	1007/67/10	02/04/2007	02/04/2007	05/31/2007	04/13/2008	04/14/2008	04/11/25/2008	04/26/2008	06/24/2008	06/25/2008	05/30/2007	08/21/2007	04/12/2008	06/24/2008	04/14/2007	04/17/2007	04/17/2007	04/26/2007	09/17/2007	04/13/2008	04/14/2008	04/25/2008	04/13/2008	04/14/2008	014/23/2005	00/01/2001	04/25/2008	04/14/2007	04/17/2007	04/17/2007	04/14/2007	08/21/2007	04/12/2008	06/24/2008	05/17/2007	04/12/2008	06/24/2008	06/26/2008	07/22/2007	07/22/2007	1002/02/20	09/11/2007	04/13/2008 04/14/2008	04/25/2008	05/17/2007	05/31/2007	07/22/2007	101/22/2001 on Next Page.	- D
TABLE 0.4	$_{\rm Name}^{\rm HD}$	146361	140501	146361	146361	146361	146361	140301	140301	140301	176361	146361	148653	149661	149661	149806	149806	149806	151541	151541	151541	153557	153557	153557	153557	154345	154345	154345	154345	155712	155712	155712	21/001	157347	157247	158614	158614	158614	158633	158633	158633	159062	159062	159062	159062	159222	159222	159222	159222	160346	160346	061101	161198	161198	161198	164922	164922	165401	L034U1 Continued	

HD Obs Name UT Date S Baseli UT Date S 165401 $0/25/2008$ 1 El-s 165401 $0/23/2008$ 1 El-s 165401 $0/7/05/2008$ 1 El-s 165401 $07/05/2008$ 1 El-s 166620 $09/17/2007$ 1 W1-s 166620 $09/17/2007$ 1 W1-s 166620 $09/17/2007$ 1 W1-s 175742 $06/21/2008$ 1 El-s 175742 $06/21/2008$ 1 W1-s 175742 $06/21/2008$ 1 W1-s 175742 $06/21/2008$ 1 W1-s 175742 $06/21/2008$ 1 W1-s 175742 $06/21/2007$ 1 W1-s 175742 $06/21/2007$ 1 W1-s 175742 $06/21/2007$ 1 W1-s 175742 $06/21/2007$ 1 W1-s <th></th>										
	aseline	Obs I	ted	Date Reduced	s	Fringe Scans Good Tota	l FF	$\langle {f F} {f W} angle$	$\sigma_{\rm FW}$	Reduction Notes
	E1-S1 E1-S1	DR DR	CB	06/04/2008 $06/30/2008$	თ თ	226 269 264 506	100 100	3.7 1.3	$3.1 \\ 1.3$	 Several FE are double peaked
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	CB	06/30/2008	ωc	208 385	100	1.7	1.8	
	E1-51	DR	DR N	07/05/2008	νv	172 509 173 353	100	2.2	1.4	weak tringes, double-peaked FE in many scans Weak fringes, double-peaked FE in many scans
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	N1-S1 N1-S1	DR	DR	07/07/2008 07/07/2008	w w	199 205 206 205	150	15.3 14.2	0.0 8	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	DR	05/14/2007	n vo	221 224	155		· ·	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	V1-E1	DR	DR	05/14/2007	ω ç	222 231	155	. c	. a . c	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	N 1-52 E2-S2	PJ PJ	DR DR	11/14/2007	v	206 214 214	102	0.9 4.0	0.0 10.0	Muuti-peaked FE in most scans; asymmetric SFE; tew good scans
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	CB	06/30/2008	ŝ	235 431	100	2.1	2.4	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E2-S1	L H	DR	06/27/2007	S: 5	222 312	156	1.5	0.8	Weak fringe, strong side-lobes
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	N1-52 N1-51	DB.	DR	11/14/2007 04/21/2008	. v	210 210 181 255	00 100	2.9	0.0	LWIN-peaked SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	N1-S1	DR	DR	04/21/2008	ŝ	158 257	150	1.8	1.6	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	CB	06/30/2008	S: 5	127 615	100	0.6	0.6	Poor SNR
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1 M1-S2	H H H H H H H	n e	06/30/2008 06/30/2008	n n	74 821 169 306	100	0.4	0.4 1.6	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	DR	05/14/2007	o o	209 226	155	: :	₽ ₩	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	N1-S2	CF	DR	11/14/2007	S	213 213	65	15.4	10.9	Multi-peaked FE in most scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	V1-S1	DR	DR	04/21/2008	s s	203 242	100	5.7	5.4	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	61-51 M1-82	H C H C	98	06/30/2008 06/30/2008	n n	176 2407 176 240	100	9.0 7	0.0 9.0	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	E2-S1	FJ	DR	06/27/2007	n v	209 215	156	5.1	, v v v	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	N_{1-S1}	DR	DR	06/27/2007	S	207 212	156	15.1	8.5	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E2-S1	DR	DR	06/27/2007	S: 3	193 299	156	1.6	1.2	Double-peaked SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	V1-S1 81_S1	H L L	H C C	06/27/2007 11/09/2007		221 376	156	1.3	0.0	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	N1-S1	DR	DR	11/12/2007	o oo	208 283	156	2.4	1.8	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	E1-S2	CF	$_{\rm DR}$	11/12/2007	S	221 379	102	1.2	0.7	Weak fringe
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	V1-S1	L d	DR	11/12/2007	ω ξ	118 198	102	1.5	1.3	Low SNR
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	N1-31 E1-S1	H C	H H H	06/27/2007 05/14/2007	n v	215 252	155	×.	D.I	 Slicht asymmetry at base of neak on left
	E1-S2	CF.	DR	11/12/2007	o v	198 199	102	5.9	2.8	the administry as parts of point on the
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	N1-S1	PJ	DR	11/12/2007	S: 3	120 198	102	1.4	0.8	Asymmetric SFE
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	V1-S1 51 21	E E	H H H H H H H	11/12/2007	νo	205 224 205 224	102	3.1	2.4	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	E1-S1	DB	U H	11/12/2007 05/14/2007	οთ	212 213 221	155	4.4	0.0	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$_{\rm N1-S1}$	DR	DR	11/12/2007	S	195 224	156	3.0	2.2	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	N1-S2	CF CF	DR	11/14/2007	S S	210 221	102	7.7	6.3	Multi-peaked FE in most scans; asymmetric SFE
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E2-S2 F2-S1	L L L	UK AU	11/14/2007 06/27/2007	ωu	207 237 208 237	156	57 5 7	1.4 4.1	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	N1-S1	DR	DR	06/27/2007	n v	215 219	156	0.0 9	3.0	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	v_{1-E1}	DR	DR	05/14/2007	s	228 236	155	:	:	
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	V1-S1	CF	DR	06/27/2007	თ წ	208 208	156	23.7	15.6	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-31	2 5	אם הם	11/19/2007	. u	213 344 107 200	102	2 0 C	13.7 0 E	HINT OF LOW SECOND PEAK IN SFE
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	N1-S1	CF C	DR	11/12/2007	າທ	198 198	102	29.5	21.3	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	E1-S1	CF	DR	11/12/2007	S	203 250	102	6.5	5.6	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	DR	05/14/2007	S	223 245	155			
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	VI-SI	ъ. С.F.	H L L	06/27/2007	n f	213 310 166 290	156	2.4	2.0	Decent CND
189340 06/21/2008 1 E1-S 189340 06/23/2008 1 E1-S 189340 06/23/2008 1 E1-S 189373 05/30/2007 1 W1-S 189733 07/24/2007 1 W1-S 189773 09/17/2007 1 W1-S 190067 07/22/2007 1 W1-S 190067 07/22/2007 1 W1-S 190067 07/20/2007 1 W1-S	N2-S2	E E	38	07/01/2008	. v	196 235	100	3.4	2.5	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	CB	06/30/2008	s	253 420	100	1.6	1.4	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	E1-S1	DR	B	06/30/2008	S o	250 866	100	0.9	1.4	:
189733 07/24/2007 1 E1-S 189733 07/24/2007 1 E1-S 189733 09/17/2007 1 W1-S 190067 07/22/2007 1 W1-S 100067 07/22/2007 1 W1-S	ET-ST MT-ST	אח אח		06/30/2008 06/37/2008	νu	228 702	156	0.9 3.6	1.2	
189733 09/17/2007 1 W1-5 190067 07/22/2007 1 W1-5 100067 07/05/0007 0 W1-5	E1-S1	PJ	DR	11/08/2007	n v	203 228	156	2.4	1.3	
190067 $07/22/2007$ 1 W1-S	N1-S2	CF	DR	11/14/2007	S	208 208	65	11.9	8.0	Multi-peaked FE in most scans; asymmetric SFE
	N1-S1 E1_S1	CF G	DR	11/08/2007	ω u	200 209 201 209	156	7.2 13.5	3.7 6.7	
Continued on Next Page	1	5			1)) 4	2	1	

	Reduction Notes				roor minge quainy, muur-peaked r E in several scans Multi-peaked FE in several scans	· · · · · · · · · · · · · · · · · · ·	Weak, choppy fringe; asymmetric summary SFE	:		Asymmetric SFE		Broad ACOR plot	Low fringe quality	Multi-peaked FE in several scans		Multi-peaked FE in several scans		Noisy data, poor PS, assym SFE	mh - 6	I he fringes are very very messy, this is inconclusive data					:		Double peaked CCORR				Flat-topped FE peak in several scans	•	Poor-quality data, very low SNR, weird PS	:				Weak choppy tringe	 Multi-neaked FE in several scans	Asymmetric or twin-peaked SFE plots	Multi-peaked FE in several scans	Weak fringe	 Multinoolod FR in sources score				:	:					:
	$\sigma_{\rm FW}$	1.9	3.7	1.2	1.1	1.7	1.4	. ×	2.6	1.6	5.7	5 0 5 0	- 5 - 7	1.9	4.0	2.4	3.5	1.4	1.1 2.2	0 u 0 u	0.0	5.3 I	9.1	8.2	5.2	7.3	4.5	2.7	0 L 1 U	5 6 17 6	4.6	0.8	1.3	x 0	1.4	2.0	1.1	0.0 0	0.4 1	5.3	1.0	1.1	0.9	1 1	1.2	6.2	0.8	3.1 2.1	4.4 6.5	5.7	2.1	1.7	;
	$\langle \mathbf{F} \mathbf{W} \rangle$	2.7	6.5	2.1	3.9	3.4	4.0		5.1	2.6	11.6	1.2	9 U	5 2 2	9.6	3.8	4.5	1.9	1.1 7	0.0 V	1 oc	3.6	12.9	11.1	7.1	8.4 9.7	8.9 8.0	2.7	16.8 16.8	0.07 8.6	7.8	2.0	3.6	x o x o	2.9	5.3	2.3	о г х. с	1.0	1 00 1 00	1.6	2.0	1.4	2.7	2.5	12.3	0.4	4-х -	7.8 7.8	7.6	4.8	2.4 1.6	2
	ΕF	156 156	102	156	150 156	102	65 1 E E	102	$102 \\ 102$	102	102	102	156 156	102	156	156	102	102	100	00 T	150	150	150	150	151	100	100	100	156 156	156	156	156	65	100	102	102	102	156	156	65	156	156	156 156	156	156	102	100	150 150	150 150	150	156	156 102	40.4
	cans Total	248 206	198	243 406	490 198	214	239	235 198	205	254	200	198	198 260	198	198	224	198	198	470	1016 1016	237	228	217	214	213	$^{240}_{240}$	$^{246}_{240}$	349	108	229 229	198	198	273	522	213	198	231	411	198	229	309	198	198 264	198	198	199	697	225	211 211	217	203	270 258	2
	ringe S ood	206 1 98	193	200	112	201	239	223	200	233	200	197	198	161	198	205	181	134	191	89 115	061	174	198	202	206	212	241	248	108	203	198	176	273	142 65	199	197	211	126	178	228	213	168	132	183	181	198	193	191	180 201	205	199	211 180	201
	I O	w w	2 02	S	n n	S I	S O	ກັນ	0.00	n on	S	s co	n n	200	S	S	S	2.0	ΩŞ		2 00	n v	S	S	S	s s	··· (ω α	Ωŭ	າທ	S	S	<u></u>	νu	0.00	S	S S		0 0	ç.,	S	s so	ົ້	2 02	n vo	S	ິດເ	ממ	n n	n on	S	s S S	;
	q	07 70	07	04	20	07	10	20	07	07	07	04	20	20	07	07	07	07	x	x x	200	80	08	08	08	808	808	2 0	- 10	20	07	07	07	x x	20	07	10	200	20	07	02	07	20	-20	07	07	808	2 2	080	080	07	- 0 0	
	Date Reduce	$\frac{11/08/20}{11/08/20}$	11/08/20	11/08/20	11/12/20	11/12/20	11/12/20	05/14/20 11/12/20	11/12/20	11/12/20	11/12/20	11/08/20	11/08/20	11/08/20	11/08/20	11/08/20	11/08/20	11/12/20	06/04/20	06/30/20	02/00/20	07/07/20	07/07/20	07/07/20	07/01/20	07/01/20	06/30/20	06/30/20	02/90/11	11/08/20	11/08/20	11/09/20	$\frac{11}{12}20$	02/20/20	11/09/20	11/12/20	11/12/20	11/12/20	11/08/20	11/14/20	11/08/20	11/09/20	11/09/20	11/09/20	11/09/20	11/14/20	06/30/20	02/20/20	07/07/20	07/07/20	11/09/20	11/09/20	/ / + +
	\mathbf{Red}	DR	DR	DR	DR	DR	DR DR	лч Ля	DR	DR	DR	DR	ж Ц Ц	DR	DR	$_{\rm DR}$	DR	DR	9 E	9 8 2 8	a a a	DR	$_{\rm DR}$	$_{\rm DR}$	CB	B	GB	CB CB	2 C	DR	DR	DR	DR	אם ר א	DR	DR	DR	אם ר א	H C H	DR	$_{\rm DR}$	DR	н Ч	UR.	DR	DR	CB CB	א ה א	ч Л Л Л	DR	DR	DR BR	
	Obs	ЪЛ	ΡJ	ЪЛ	DR	DR	DR DR	H C H H	DR	CF.	ЪJ	GF GF	P CF	ΡJ	CF	ЪJ	ЪJ	DR	H H H H H H	н н н н	H C	DR	DR	DR	EB	EB	DR	л Ч	50	PJ	ЪJ	DR	DR	лч н	DR	DR	DR	сF С	P.I	CF.	ЪJ	DR	н Ч Ц	DB DB	DR	CF	DR	אר ה איר	чл DR	DR	DR	DR B	
	Baseline	E1-S1 E1_S1	W_{1-S1}	E1-S1	W1-S1	W1-S1	E1-S1 E1 S1	E1-51 W1-S1	E1-S1	E1-S2	W1-S1	W1-S1	E1-21 E1-21	W1-S1	E1-S1	E1-S1	W1-S1	E1-S1	EI-SI	W1-51 F1_51	E1-S1	E1-S1	m W1-S1	W1-S1	W2-S2	W2-S2	W1-S1	E1-S1	W1-21 F21-21	E1-S1	W1-S1	E1-S1	W1-S1	W1-51 E1_61	E1-S1	m W1-S1	E1-S1	E1-52 E1-52	M1-S1	W1-S2	E1-S1	E1-S1	W1-S1 F1_S1	E1-S1	W_{1-S1}	W1-S2	E1-S1	E1-21 51-21	W1-S1	W1-S1	E1-S1	W1-S1 E1-S1	1
	s	с -	101	0	7		01-		- 0	. –	1	- 0	- N	- 01	-	1	0	0,				101	1	0	-	01			- c	1	0	1				1	0,				7		:N -		0	1		- c	- ۱	0		- 10	• : •
- Continued	Obs UT Date	07/24/2007	07/24/2007	07/24/2007	08/17/2007	08/18/2007	08/18/2007	04/14/2007 08/18/2007	08/18/2007	08/20/2007	08/21/2007	07/22/2007	07/22/2007	07/24/2007	07/22/2007	07/24/2007	07/24/2007	08/15/2007	04/25/2008	06/22/2008	00/23/2008	07/06/2008	07/07/2008	07/07/2008	06/09/2008	06/09/2008	06/22/2008	06/23/2008	1002/27/10	07/24/2007	07/24/2007	08/14/2007	$\frac{08}{17}$	07/93/2008 07/93/2008	07/28/2007	08/18/2007	08/18/2007	08/20/2007	01/24/2007	09/17/2007	07/24/2007	08/14/2007	08/14/2007	08/14/2007	08/14/2007	09/17/2007	06/23/2008	07/06/2008	07/07/2008	07/07/2008	07/26/2007	07/26/2007 07/28/2007	on Next Pag
TABLE 3.2	HD Name	190067 190404	190404	190470	190470	190470	190470	177001	190771	190771	190771	191499	191499 191499	191499	191785	191785	191785	192263	192263	102263	192263	192263	192263	192263	195564	195564	195564	195564	197076 197076	197076	197076	198425	198425	198425 108495	200560	200560	200560	200560	202751	202751	208038	208038	208038	208313	208313	210277	210277	22012	210277	210277	210667	210667 210667	Continued
	Reduction Notes				Multi-peaked FE in most scans; asymmetric SFE	:	:	Poor SNR; multi-peaked FE in several scans	:	Multi-peaked FE in several scans	:	:	Beam intensities weird during fringes; few scans show two peaks	:	:	:	Multi-peaked FE in several scans	:	:	Good SNR	Good SNR		Weak fringe, Choppy FE	:		Noisy data		:				 Multi-peaked FE in several scans		Multi-peaked FE in several scans	:	Multi-peaked FE in several scans	Poor SNR, but fringes strong when seen			Very choppy tringe		Tiny tringes, weak PS											
-------------	---	------------	------------	-------------	---	---------------	---------------	--	---------------	----------------------------------	-------------	---------------	---	------------	------------	------------	----------------------------------	------------	------------------	-------------	-------------	---------------	------------------------	------------	---------------	------------	------------	------------	------------	----------------	------------------	--------------------------------------	---------------	----------------------------------	---------------	----------------------------------	--	------------	------------------	--------------------	--------------	-----------------------	--------------	---------------	------------------	--	----------------------------						
	$\sigma_{\rm FW}$	1.7	1.3	0.7	1.8	1.2	1.9	0.9	2.4	3.1	4.3	3.4	12.2	2.8	2.2	3.1	1.3	0.5	4.1	12.1	11.2	1.0	0.9	11.9	8.6	1.2	3.1	3.1	9 I 7 7	1.1	1.1	9.0	3.6	1.5	1.0	5.2	4.3	4.2	0.0 0.0	1.9 1.9	9.	0.1 1	10.0	17.7	4.4	0 9 ₽ 0	5						
	$\langle \mathbf{F} \mathbf{W} \rangle$	2.1	2.1	1.2	4.1	2.3	2.6	1.5	2.9	4.4	6.3	7.4	19.6	4.9	6.3	5.7	2.6	0.4	6.4	35.7	18.4	1.9	1.9	26.0	27.2	2.2	3.9	 	4.9	4.T	4 e	9.6	5.0	3.4	1.5	4.2	1.3	10.6	×.	4.0	14.U	4.0	0.0	13.8	۵.0 10		2						
	ЪF	156	156	156	102	150	150	156	102	102	156	156	102	156	156	156	102	100	150	156	156	156	156	102	102	102	102	102	102	150	150	65	102	102	102	102	102	102	707	65	150	156	1 nu	100	102 103	102	1						
	cans Total	259	243	373	203	313	279	323	234	222	198	198	233	198	198	223	224	616	218	198	198	198	198	198	199	236	214	205	213	102	2570	201	210	204	189	208	610	198	797 797	137	002	239	105	272	102	253)) 4						
	ringe S	84	198	203	198	166	167	219	204	197	192	197	198	192	198	200	203	70	197	198	198	159	172	198	661	205	661	197	202	200	50	201	661	201	88	102	79	98	50	137	- 66 - 52	102	242	602	002	154							
	т. С			с.				~.					ç.,																	0.7															•••	0.70							
	о Т	20	. 20	07 S	27	38	38	07 S	27	27	27	27	07 S	70	27	22	27	38	38	27	27	27	37 20	70	70	70	27	20		200	20	20	27 5	27	07	22	20					20	x	x t	20		-						
	Date Reduce	11/09/200	11/09/200	11/09/200	11/14/200	07/07/200	07/07/200	11/09/200	11/12/200	11/12/200	11/09/200	11/09/200	11/08/200	11/09/200	11/09/200	11/09/200	11/09/200	07/01/200	07/24/200	11/09/200	11/09/200	11/09/200	11/09/200	11/12/200	11/12/200	11/12/200	11/12/200	11/12/200	11/12/200	02/21/11	02/02/20/20	11/12/200	11/12/200	11/12/200	11/12/200	11/12/200	11/09/200	11/12/200	07/21/11	11/12/200	11/12/21/11	11/12/20	107/00/00/00	007/02/00/200	11/12/200	11/12/200	/== /==						
	\mathbf{Red}	DB	DR	DR	DR	DR	DR	DR	DR	DR	$_{\rm DR}$	DR	$_{ m DR}$	DR	$_{ m DR}$	$_{ m DR}$	$_{\rm DR}$	CB	$_{\rm DR}$	$_{\rm DR}$	$_{\rm DR}$	DR	DR	DR	DR	DR	DR	ЦЧ	нц	אר היו	H C	DR	DR	DR	DR	DR	ЦЧ	DR.	H L L L	ЧЧ	нц	н	n e		ч п п	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	;						
	Obs	DB	DR	$_{\rm DR}$	CF	$_{\rm DR}$	DR	DR	CF	CF	$_{ m DR}$	DR	CF	DR	$_{ m DR}$	$_{ m DR}$	$_{ m DR}$	$_{ m DR}$	$^{\mathrm{TB}}$	$_{ m DR}$	$_{ m DR}$	$_{\rm DR}$	DR	DR	DR	CF	CF	GF.	нц	אם ב	H C A	CF CF	$_{\rm CF}$	CF	$_{ m DR}$	DR	нц	DR	ЦЧ Н	GF.	CF.	ЧС С	בת ה	אם ביים	ת ח ה ה	1 E E E	;						
	Baseline	E1-S1	W1-S1	E1-S1	W1-S2	E1-S1	W1-S1	E1-S1	W1-S1	E1-S1	E1-S1	W1-S1	E1-S1	E1-S1	W1-S1	E1-S1	E1-S1	E1-S1	$E_{1-S_{1}}$	E1-S1	W1-S1	E1-S1	$\rm W1{-}S1$	W1-S1	E1-S1	E1-S2	W1-S1	E1-S1	W1-S1	EI-NI 51 CI	M1-S1	E1-S2	W1-S1	E1-S1	W1-S1	E1-S1	E1-S1	W1-S1	EI-NI	E1-S2	EI-VI	E1-S1	M L-DL	72-T M	W1-01 E1-S1	E1-S2	1 2 4 4						
	s	-	101	1	1	1	1	1	1	7	1	0	1	-	0	1	7	1	1	7	ŝ	1	7	-	7	-	-	21	- 0	N -		- 01	1	7	1	2		- 0	. 1	- 0	21 0	، מ			- د	، ۱	,						
- Continued	Obs UT Date	07/26/2007	07/26/2007	07/28/2007	09/17/2007	07/06/2008	07/07/2008	07/26/2007	09/16/2007	09/16/2007	08/14/2007	08/14/2007	07/22/2007	08/14/2007	08/14/2007	07/26/2007	07/28/2007	06/25/2008	07/21/2008	07/26/2007	07/26/2007	08/14/2007	08/14/2007	08/18/2007	08/18/2007	08/20/2007	09/16/2007	09/16/2007	08/18/2007	02/06/2006	07/07/2008	08/20/2007	09/16/2007	09/16/2007	08/18/2007	08/18/2007	07/27/2007	08/18/2007	1002/21/20	08/20/2007	06/16/2007	09/16/2007	04/20/2000	06/24/2008	08/10/2007	08/20/2007	· · · · · / · - / · ·						
TVDDD 017	HD Name	211472	211472	211472	215152	215152	215152	216520	216520	216520	217107	217107	217813	217813	217813	218868	218868	218868	218868	219134	219134	219538	219538	219623	219623	219623	220140	220140	220182	220122	220339 220339	221354	221354	221354	221851	221851	222143	222143	222143	222143	222404	222404	222404	222404	224400 994465	224465)) 4 4 4 8						

are DR as above and CB = C. Black. The last column contains notes made when data were reduced and analyzed. It contains some abbreviations that stand for: ACOR = Autocorrelation NOTES.— Column 5 observer codes are: CF = C. Farrington, DR = D. Raghavan, EB = E. Baines, NT = N. Turner, PJ = PJ Goldfinger, TB = T. Boyajian. Column 6 reducer codes summary plot, CCORR = Cross-correlation shifted summary plot, FE = Fringe envelope, PS = Power spectrum, and SFE = Summary fringe envelope. When secondary fringes are mentioned, their intensity relative to the primary fringe is given parenthetically, and the offset in sampling interval unit is given as a positive or negative number, indicating the position of the secondary

:- : :- : : : : : : :- :			
	i∾ : : : : : : : : : : : : : : : : : : :		
: :- : : : : : :- :	8	R : : : : : : : : : : : : : : : : : : :	
5			
, I -			
5			
1			
1			
1			
1 1			· · · · · · · · · · · · · · · · · · ·
1		· · · · · · · · · · · · · · · · · · ·	
1		· · · · · · · · · · · · · · · · · · ·	
			: : : : : : : : : : : : : : : : : : :
			• = : : : : : : : : : : : : : : : : : :
	::::::::	• : : : : : : : : : : : : : : : : : : :	
	:::::::		: : : : : : : : : : : : : : : : : : :
			: : : : : : : : : : : : : : : : : : :
: :	:::::		· · · · · · · · · · · · · · · · · · ·
1 1	::::		
1	:::		· · · · · · · · · · · · · · · · · · ·
1	 		· · · · · · · · · · · · · · · · · · ·
1	:		
	:		::::: :::::
		::::	: : : : : : : : :-
	:	: : :	· · ·
1	:	· · · · · · · · · · · · · · · · · · ·	
	:		-
	:		
-			2
-			
	:		
:	:		
1	:	:	
1	:		
1	:		:
1			
1			1
		-	
	:	: + c	
	:	: 1	· · · · · · · · · · · · · · · · · · ·
T	:	T	··· T ···
1	:		
1	:		
-			-
	:		··· ·· T
1	:		
1	:		
1			1
	:	•••••	· · · ·
··· T	:		T
	:		
1	:	••••	

ontinued	
Ŭ	
3.3	
ABLE	

) Name	Total Obs	==== Single	=== >-E] Single?	Double	====== Double?	==== Single	=== >-w Single?	Baselines = Double	====== Double?	=== Single	=== E- W Single?	Double	Double.
61606		:	:	:	:	:	:	:	:	:			:
63433	4	1	1	:	:	2	:	:	:	:	:	:	:
64606	:	:	:	:	:	:	:	:	:	:	:	:	:
54468	: •	:	:	:	:	: •	:	:	:	:	:	:	:
05450	- 6	:-	:	:	:		:	:	:	:	:	:	:
57058	1 -	- °	:-		•	-		:			:		:
38017	* 6	c	4			:		:			:		:
38255	10		: -			4							
72760	1	• :	• .			: :		: :					
73350								: :	: :				
73667		:	:	:	:			:	:		:		:
5767	:	:	:	:	:	:	:	:	:	:	:	:	:
6151	:	:	:	:	:	:	:	:	:	:	:	:	:
9606.	12	4	:	1	9	1	:	:	:	:	:	:	:
6966.	5	4	:	:	:	1	:	:	:	:	:	:	:
0715	2	1	:	:	:	1	:	:	:	:	:	:	:
2443	Ð	4	:	:	:	1	:	:	:	:	:	:	:
2885	4	1	1	:	:	7	:	:	:	:	:	:	:
17883	ი	:	7	:	:	1	:	:	:	:	:	:	:
9269	2	1	:	:	:	1	:	:	:	:	:	:	:
0343	7	1	:	:	:	1	:	:	:	:	÷	÷	:
4765	2	1	÷	:	÷	1	÷	:	:	:	÷	÷	:
6064	7		:	:	:		:	:	:	:	:	:	:
7334	7	1	:	:	:	1	:	:	:	:	:	:	:
7658	c1 i	010	:	:	:	:	: 0	: ,	:	:	:	:	:
8230	CI O	0,	n	:	:	، 0	2	T	:	:	:	:	:
10701	21		:-	:	:		:	:	:	:	:	:	:
19491	00		T	:	:		:	:	:	:	:	:	:
2676	71 0	c	:	:	:		:	:	:	:	:	:	:
0120	n a	.N -	:	:	:		:	:	:	:	:	:	:
2000	21 0		:	:	:		:	:	:	:	:	:	:
0071	71 0		:	:	:		:	:	:	:	:	:	:
4004	4 -		:	:	:	- c	:-	:	:	:	:	:	:
2051 8051	* 0		: -	:	:	1-	-	:					:
0833	о LC	- -				- 6	-						
2758	0		' :			ı –			: :				
3449	2	1	:	:	:	1	:	:	:	:	:	:	:
4783	1	:	:	:	:	1	:	:	:	:	:	:	:
5404	4	2	1	:	:	1	:	:	:	:	:	:	:
6442	2	1	:	:	:	1	:	:	:	:	:	:	:
6443	0	1	:	:	:	1	:	:	:	:	:	:	:
6956 2222	9	ю,	: •	:	:		61	:	÷	:	:	:	:
9332	ю -	- 0		:	:		:	:	:	:	:	:	:
1560	4 0	21 0	Т	:	•		:	:	:	•	•	•	:
4232	10	4 1	:	:	:	⊣ c	:	:	:	:	:	:	:
1403U	- 0	00	:	:	:	v -	:	:	:	:	:	:	:
8642	с	10				-		. 6					:
7334	0 4	ı –				-		1		-			
8165	ŋ	1	7	:		1	:	:		:	1		:
8311	2	:	:	1	:	1	:	:	:	:	:	:	:
80004	ъ	1	1	1	:	1	1	:	:	:	:	:	:
0307	- 12		: '	:	:	- 0	:	:	:	:	:	:	:
52142	4 0		Т	:	•	- 17	:	:	:	-	•	•	:
107704	10		:	: -	:	- c	:	:	:	T	:	:	:
5599	- 6	۲		-		- ہ							
6202	9		n			101			: :				
6713	c	,				•							
	4	T		:		-	:	:		:	:	:	:

ŏ
Ē
Ц
÷.
Ę
<u>,</u>
\circ
1
ŝ
÷
田
1
1
1

D Name	Total Obs	====: Single	=== S-E I Single?	3aselines = Double	 Double?	=== Single	=== >-w Single?	Baselines : Double	====== Double?	===== Single	=== E-W Single?	Double	======= Double?
137763	9	2	:	:		ę	:	1	:	:	:	:	:
137778	4	:	1	:	:	ŝ	:	:	:	:	:	:	:
139777	4	,	:	:	:	с (:	:	:	:	:	:	:
139813	m c		÷	•	÷	cı -	:	:	:	:	:	÷	:
139341	14	- :	:				:						
141272	- 01	- 1	• :		: :					• :			
142267	4	2	1	:	:	1	:	:	:	:	:	:	:
144287	2	1	:	:	:	1	:	:	:	:	:	:	:
145675	ŝ	1	:	:	:	:	1	:	:	1	÷	:	:
145958	: 0	: 0	:,	:	:	:	:	:	:	:	:	:	:
146361	30	29	- 0	:	:	: -	:	:	:	:	:	:	:
146233	4 -		N	:	:	Ч	:	:	:	:	:	:	:
148033	- 6		:	:	:	-	:	:	:	:	:	:	:
149806	100					10							
151541	000					101							
153557	4	:	:	:	:	33	1	:	:	:	:	:	:
154345	4	1	:	:	:	2	:	:	:	1	:	:	:
155712	4	1	:	:	:	2	1	:	:	:	:	:	:
157347	ŝ	1	:	:	:	5	:	:	:	:	:	:	:
158633	ο ·		:	:	:	0	:	:	:	:	:	:	:
159062	4 c		:-	:	:		:	:	:	:	:	:	:
159222	0.4	- 6	-	:		- 6	:	:			:	:	:
160346	10	1			: :	1		: :	: :		: :		
61198	5	1	:	:	:	2	7	:	:	:	:	:	:
64922	7		:	:	:	1	:	:	:	:	:	:	:
.65401	6,	9	:	:	:	7		:	:	:•	:	:	:
75740	41	N -	: c	:	:	: •		:	:	Т	:	:	:
76377	- v:	7 6	N		:	0 0	-	:				:	
78428	6		1										
79957	5	2	:	:	:	2	1	:	:	:	:	:	:
80161	ю.	с (:	:	:		1	:	:	:	:	:	:
82488	4 0	21 0	: -	:	:	21 0	:	:	:	:-	:	:	:
84385	00	7 -	Т	:	:	7 -	:	:	:	T	:	:	:
85414	10												
86858	1	• :				• :			: :				
89340	5				: :	1	1						
89733	co	1	:	:	:	2	:	:	:	:	:	:	:
29006	ŝ	2	:	:	:	1	:	:	:	:	:	:	:
90404	61		:	:	:		:	:	:	:	:	:	:
90470 00771	ю ı	21 0	:	:	:	n	:	:	:	:	:	:	:
90771 01400	0 <	0 C	:	:	:	10	:	:	:	:	:	:	:
01785	† cr	10				4 -		:			:		
92263	c oc	14	:	: :	: :	- 0	:	: :	: :		: :	: :	
95564	4	1	:	:	:	2	1	:	:	:	:	:	:
97076	4	2	:	:	:	2	:	:	:	:	:	:	:
98425	4	2	:	:	:	1	1	:	:	:	:	:	:
00560	4	0 -	1	:	:		: •	:	:	:	:	:	:
167203	n c	c	:	:	:		Т	:	:	•	:	:	:
08313	0 m	40	:	:	:		:	:	:	:	:	:	:
10277		10		:		H CT					:		
210667	ი	- 1	. 1		: :			: :	: :		: :		
211472	ę	1	1	:	:	1	:	:	:	:	:	:	:
215152	0	1	:	:	:	2	:	:	:	:	:	:	:
216520	33	1				-							
107111			4	:	:		:	:	:	:	:	:	:

led	
ntin	
- Co	
3.3	
ABLE	

	Total		=== S-E]	Baselines =			== S-W	Baselines -			== E-W	Baselines	
HD Name	Obs	\mathbf{Single}	Single?	Double	Double?	Single	Single?	Double	Double?	Single	Single?	Double	Double?
217813	3	1	1	:	:	1	:	:	:	:	:		:
218868	4	4	:	:	:	:	:	:	:	:	:	:	:
219134	5	1	:	:	:	1	:	:	:	:	:	:	:
219538	5	1	:	:	:	1	:	:	:	:	:	:	:
219623	ŝ	2	:	:	:	1	:	:	:	:	:	:	:
220140	2	1	:	:	:	1	:	:	:	:	:	:	:
220182	2	1	:	:	:	1	:	:	:	:	:	:	:
220339	5	1	:	:	:	1	:	:	:	:	:	:	:
221354	ę	2	:	:	:	1	:	:	:	:	:	:	:
221851	2	1	:	:		1	:	:	:	:	:	:	:
222143	4	ę	:	:	:	1	:	:	:	:	:	:	:
224465	c,	2	:	:	:	1	:	:	:	:	:	:	:

Given the Δ mag limit of CHARA of about 2.5 in K-band, it is no surprise that all but one of the systems detected are double-lined spectroscopic binaries. Two additional doublelined binaries escaped detection – HD 158614, the 46.4-year SB2VB whose separation has closed-in from 0".6 in 1829 to 0".3 in 2005 (WDS), and HD 189340, the 4.9-year SB2VB with a 0".2 separation. The former was observed twice along S-E baselines and once along a S-W baseline, and the latter thrice along S-E baselines and twice along S-W baselines, with no evidence of separated fringes. Perhaps the orientation of the binary caused these fringes to be too wide.

Several single-lined spectroscopic binaries, and/or photocentric motion binaries (HD 17382, HD 24409, HD 32850, HD 64468, HD 64606, HD 65430, HD 110833, HD 128642, HD 142267, HD 144287, HD 113449, HD 160346, HD 161198) could have separations in the SFP detection range, but none of these were seen as SFP binaries. This is not surprising because the magnitude difference that causes spectroscopy or visual techniques to only pick up one star is likely too large for detection using CHARA. Finally, five more systems were resolved with speckle or other visual techniques, (HD 32923, HD 100180, HD 135204, HD 145958, HD 217107) but were not seen as SFP binaries. Many of these also have null detections by other observers, so these might be large-contrast pairs or have too-wide a separation for SFP detection.

So why did this study not detect more SFP binaries, filling-in the gap between spectroscopic and visual binaries? This gap is evident from other studies (Bouvier et al. 1997; Mason et al. 1998a), but not seen here. The reason is likely that this study is limited to nearby stars within 25 pc, for which the angular separations probed by visual techniques translate to smaller linear separations as compared to more distant systems, allowing visual techniques to approach separations studied spectroscopically and thus closing the gap. Additionally, this sample of nearby solar-type stars represents one of the most extensively studied samples. Long-standing campaigns of high-precision radial velocity measurements (e.g., Udry et al. 1998; Marcy et al. 2004) as well as high-resolution visual techniques such as speckle interferometry (e.g., Mason et al. 1998b) have targeted such systems. Planet search teams have now collected radial velocities for many of these stars over some 12 years at a few $m s^{-1}$ precision, and other surveys such as CORALIE and the CfA efforts have almost 30 years of coverage with $\sim 0.5 \text{ km s}^{-1}$ precision. With such intensive scrutiny, binaries can be detected for periods out to a few tens of years, and several such examples are covered in Chapter 6. As an example, a binary with a 30-year period can easily be identified by such surveys and, for a 1.5 M_{\odot} total mass, will have a semi-major axis of 11". At the median distance of my sample of about 20 pc, this translates to an angular semimajor axis of 0.6 for a face-on orientation. For a 45° inclination, this separation gets projected down to 0.4. These limits are well beyond the limits of speckle and of SFP analysis, closing the gap for nearby systems. However, gaps between spectroscopic and visual techniques likely do exist for more distant systems, as the detection limit discussed above shrinks to about 80 mas for systems at a distance of 100 pc. Quist & Lindegren (2000) and references therein show evidence that while gaps may exist for distant systems, they are pretty slim, if at all present, for nearby ones. While limiting the chances for new discoveries, this gives us greater confidence in the multiplicity statistics derived for such a well-studied nearby sample.

3.1.2 Astrometry from Separated Fringe Packets

For the few instances where separated fringes were seen, C. Farrington helped me derive the projected separations of the two stars in the sky and, where a pair of observations on different baselines was available, did the triangulation to derive actual separations and position angles. The procedure for this is the same as that described in Farrington (2008) but is briefly outlined below. First, the fringe envelope is extracted for each scan and added across all scans of an observation to obtain the summary fringe envelope, as discussed in the previous section. When two fringe peaks are recorded in a scan, an IDL code written by T. ten Brummelaar is used to fit Gaussian profiles to each peak, from which their offset is measured in microns and translated to angular separation in the sky using the projected baseline. This separation corresponds to the projection along the baseline's orientation as interferometry is not sensitive to offsets perpendicular to the baseline. For single-baseline measurements, this would be the final deliverable, i.e. a vector separation between the stars. In instances when multiple observations are available in a time span sufficiently short such that the actual binary motion is negligible, multiple vector projections can be used to triangulate the actual relative positions of the two stars, yielding a separation and position angle at an observational epoch. While the projection vectors rotate over multiple observations over the same baseline due to change in its orientation over time as viewed by the star, the triangulation is most effective if the observations are taken over different, especially roughly orthogonal, baselines.

Of the three stars for which SFP data were recorded (HD 3196, 79096, and 137763),

the first two could be analyzed for astrometry, while the third has only a single observation that clearly shows two fringes, but the data quality was not good enough for measurements. Table 3.4 lists the four vector separations for HD 3196 and six measures for HD 79096 that could be obtained in the manner described here. One of the measures of HD 79096 corresponds to the failure to detect two fringes on 2008 March 10. Based on the observations on the previous and subsequent day, which detected separated packets, I conclude that they were superimposed, and accordingly take this as a measure of zero separation because the orientation of the baseline was nearly perpendicular to the binary separation. As seen below, this fits the data points rather well. Column 1 of the table is the HD number with a component designation of the primary component of the SFP binary. Column 2 is the epoch of mid-exposure in MJD, Column 3 is the projected baseline in meters, and Column 4 is the baseline orientation in degrees measured from North towards East. Column 5 lists the vector separation projected along the baseline, measured from the primary to the secondary along the direction in Column 4. The first two measures of HD 3196 and the last four measures of HD 79096 could be used for triangulation to determine true offsets, as shown in Figure 3.7. The HD 3196 pair has a relative separation of $\rho = 106.9$ mas and $\theta = 335^{\circ}3$ at 2007.618. No error bars could be determined because the triangulation was only based on two points. The HD 79096 pair has a relative position of $\rho = 63.70 \pm 0.32$ and $\theta = 285.48 \pm 0.28$ at 2007.187. These new measures are consistent with the known visual orbits for these pairs.



FIGURE 3.7: Triangulation of SFP measures for HD 3196A (left) and HD 79096A (right) corresponding to measures in Table 3.4. The primary is indicated by the square at the origin, and the secondary is the other square connected by the dotted line. The solid lines are perpendicular to the baseline orientation for each measure, and the secondary is at or near their intersections.

HD Name	MJD	Baseline Length (m)	Baseline Angle (deg)	Proj Sep (mas)
003196A	54326.350	321.783	40.32	70.321
003196A	54326.489	258.564	130.13	121.311
003196A	54360.424	244.127	125.50	63.402
003196A	54370.255	230.305	35.33	58.127
079096A	54156.375	297.271	1.18	-37.993
079096A	54156.429	301.887	167.35	57.323
079096A	54168.334	297.568	3.40	-19.573
079096A	54168.424	308.000	160.68	54.324
079096A	54169.283	304.197	15.49	0.000
079096A	54170.337	297.265	1.09	-22.343

TABLE 3.4: Projected Separations for SFP Measurements

3.2 Visual Orbits Derived using the CHARA Array

LBI has been regularly used to resolve short-period binaries and determine their orbits, yielding physical parameters such as mass, radius and luminosity of the component stars (e.g., Hummel et al. 1993; Boden et al. 1999; Torres et al. 2002). However, the shortest period spectroscopic binaries are out of reach of other interferometers, and hence have remained unresolved, yielding only partial orbital solutions and mass functions, but no component masses. An analysis of the targets investigated by this work revealed five double-lined spectroscopic binaries with periods ranging 0.27–10.98 days that had not yet been resolved interferometrically, while longer-period systems had been resolved with other long-baseline interferometers such as the Palomar Testbed Interferometer (PTI) (e.g., Torres et al. 2002) and the Mark III Interferometer (e.g., Hummel et al. 1995), or by other visual techniques (e.g., Hartkopf et al. 1996; Pourbaix 2000). Four of the five systems were studied with CHARA, the results from which are presented in the following subsections.

The shortest-period system (HD 133640B, 0.27-day period) among the five binaries selected as above could not be effectively studied because the spectroscopic binary is the secondary component of a longer period visual binary. The primary, HD 133640 (44 Bootis), is about 4" away and about 1.3 magnitudes brighter in the V-band. Despite many attempts, I could not consistently lock the secondary with the CHARA Tip-Tilt system, as it would often lock on the brighter primary. This binary is a W UMa type contact system with an increasing orbital period and has been extensively studied photometrically (Rovithis & Rovithis-Livaniou 1990, and references therein). However, due to observational and modeling challenges of this system, I was unable to pursue it further.

During the course of this work, five additional systems have been determined to be SB2 without corresponding visual orbits, three of which are suitable candidates for future CHARA observations, while the other two (HD 111312 and HD 148704) are too far south. HD 80715 was reported as a binary but without a full orbital solution (Gliese & Jahreiß 1991; Halbwachs et al. 2003), but the CfA survey efforts have now derived a 3.8-day SB2 solution (see § 6.2.2). HD 101177B is the secondary of a 9" visual pair and itself a 23.5-day spectroscopic binary with a single-lined solution in DM91 and a double-lined solution (Mazeh et al. 1997) based on three velocity measurements of the secondary. HD 144284 is a 3.1day SB2 (Mazeh et al. 2002) composed of an F8 primary and an M-dwarf companion. The estimated ΔK of 2.7 for this q = 0.4 pair is high, but within the range for which visibility modulations could be seen. Each of these systems is planned to be observed during 2009.

The following sections present the results for the four systems which were observed with CHARA to obtain visual orbits and component masses as described in Raghavan et al. (2009). These results, with the exception of the publication for HD 146361 (Raghavan et al. 2009), are all preliminary and need further analysis and, in some case, more observations.

3.2.1 The Visual Orbit of σ^2 CrB (HD 146361)

The results for this system have been published in the Astrophysical Journal (Raghavan et al. 2009), representing the first visual orbit derived using CHARA visibilities. This work leveraged spectroscopic results from D. Latham and G. Torres at CfA and utilized the longest baselines of the CHARA Array to resolve the 1.1-day spectroscopic binary, the shortestperiod system yet resolved. Because this work was undertaken and completed as part of this thesis effort, the entire text of that publication is reproduced in *Appendix* E.

3.2.2 The Visual Orbit of HD 8997

Forty three calibrated visibility measurements were obtained over five nights for this star using the CHARA Array's E1–E2 baseline in 2008 October and November. For the first two nights, I used HD 10697, a planet host star, as a calibrator. While there is no evidence of this star being a binary, the work done for Baines et al. (2008a) showed a possible pattern in the residuals for its diameter calculation, so I used HD 10477 for the final three nights, which was the calibrator used by Baines et al. and believed to be well-behaved. The data and a preliminary orbital solution are presented below, but this star needs more observations.

MJD	Obs V	σ_V	$\mathbf{Model}\;V$	$(O-C)_V$	u (m)	v (m)	HA (h)
54769.31961	0.664	0.054	0.809	-0.145	-47.98	-39.24	0.91
54769.32614	0.753	0.081	0.832	-0.079	-46.49	-39.96	1.07
54769.34215	0.771	0.078	0.886	-0.115	-42.50	-41.62	1.45
54769.34819	0.827	0.090	0.905	-0.078	-40.89	-42.21	1.60
54769.35428	0.858	0.074	0.922	-0.064	-39.20	-42.78	1.74
54769.36051	0.872	0.068	0.939	-0.067	-37.40	-43.34	1.89
54769.36796	0.935	0.076	0.956	-0.021	-35.20	-43.96	2.07
54769.37529	0.849	0.083	0.971	-0.122	-32.94	-44.55	2.25
54769.38220	0.861	0.088	0.982	-0.121	-30.75	-45.06	2.42
54769.38929	0.947	0.104	0.990	-0.043	-28.45	-45.55	2.59
54793.19142	0.334	0.058	0.378	-0.044	-57.86	-31.39	-0.60
54793.20114	0.332	0.024	0.367	-0.035	-56.88	-32.70	-0.36
54793.20992	0.340	0.028	0.360	-0.020	-55.82	-33.86	-0.15
54793.21943	0.346	0.044	0.354	-0.008	-54.48	-35.08	0.07
54793.23076	0.344	0.051	0.350	-0.007	-52.65	-36.48	0.35

TABLE 3.5: Interferometric Visibilities for HD 8997

Continued on Next Page...



FIGURE 3.8: Calibrated visibility measurements for HD 8997 versus the projected baseline. The plus signs are the calibrated visibilities with vertical error bars, and the diamonds are the calculated visibilities for the best-fit orbit. Table 3.5 lists the numeric values corresponding to this plot.

MJD	$\mathbf{Obs}~V$	σ_V	$\mathbf{Model}\; V$	$(O-C)_V$	<i>u</i> (m)	v (m)	HA (h)
54793.24114	0.384	0.068	0.349	0.036	-50.73	-37.73	0.59
54793.25135	0.401	0.061	0.348	0.053	-48.60	-39.92	0.84
54793.27051	0.439	0.069	0.348	0.091	-44.13	-40.98	1.30
54794.19223	0.685	0.078	0.627	0.058	-57.52	-31.88	-0.51
54794.20040	0.760	0.048	0.632	0.128	-56.65	-32.97	-0.32
54794.21060	0.843	0.075	0.633	0.210	-55.36	-34.30	-0.07
54794.22001	0.720	0.043	0.631	0.089	-53.96	-35.50	0.06
54794.22764	0.702	0.068	0.626	0.076	-52.69	-36.45	0.34
54794.23816	0.686	0.060	0.614	0.071	-50.74	-37.72	0.59

TABLE 3.5 - Continued

Table 3.5 lists the 24 observed visibilities over the three nights with HD 10477 as calibrator along with the corresponding epoch at mid-exposure, uncertainty of the visibility measurement, model visibility for the best-fit orbit and residuals, baseline projections in meters along East-West (u) and North-South (v) directions, and the hour angle of the target. Similar to the approach in Raghavan et al. (2009), the visual orbit solution uses the spectroscopic elements from the CfA measurements, which are presented in § 6.2.2, propagates their 1 σ errors to derive the uncertainty of the visual parameters derived here, namely α , *i*, Ω , and $\Delta K'$, constraining α and *i* by the $a \sin i$ from spectroscopy and the FvL07 parallax. Figure 3.8 plots the observed visibilities along with their uncertainties and corresponding model values. There appear to be some systematic patterns in the residuals and the phase coverage of the observations is weak. Table 3.6 lists the visual orbit elements for the above solution, which lead to component mass estimates of 1.446 ± 0.122 and 1.193 ± 0.101, too high for these K dwarfs. More observations are needed to determine the visual orbit for this pair, perhaps with additional calibrators.

Orbital Parameter Value Source Adopted values Period (days) 10.98372 ± 0.00026 1 0.0368 ± 0.0025 1 49085.08 ± 0.16 T_0 (JD - 2400000) 1 ω (deg) 179.2 ± 5.1 1 2 $\theta_{\rm p} \ ({\rm mas})$ 0.31 ± 0.03 $\theta_{\rm s}$ (mas) 0.26 ± 0.03 2Visual orbit parameters 5.63 ± 0.13 3 $\alpha \ (mas)$

TABLE 3.6: Visual Orbit Solution for HD 8997

Continued on Next Page...

TABLE 3.6 – Continued

Orbital Parameter	Value	Source
$i (\mathrm{deg})$	41.2 ± 1.4	3
$\Omega ~(\mathrm{deg})$	50 ± 14	3
$\Delta K'$	0.74 ± 0.03	3
Reduced χ^2	0.66	3
Physical parameters		
M_p (M_{\odot})	1.446 ± 0.122	3
$M_s~({ m M}_{\odot})$	1.193 ± 0.101	3

NOTES.—Source codes: 1 = From the spectroscopic solution in §6.2.2; 2 = Estimated using Gray et al. (2003) spectral types and FvL07 parallaxes, with an adopted 10% error; 3 = This work.

3.2.3 The Visual Orbit of HD 45088

Forty calibrated visibility measurements were obtained over four nights for this star using the CHARA Array's E1–E2 baseline in 2008 October and November. The calibrator used for all nights was HD 43947, but, for the last two nights, HD 43042 was also used as an additional calibrator for a total of nine brackets. On final data analysis, the fit to the second calibrator points were not as good, so they were eliminated from the final solution. Table 3.7 lists the 31 observed visibilities used in deriving the visual orbit, in the same format as in Table 3.5. Figure 3.9 plots the observed visibilities along with their uncertainties and corresponding model values for the visual orbit derived as described in § 3.2.2.

ABLE 3.7: Interferometric Visibilities for HD 45088	

MJD	Obs V	σ_V	$\mathbf{Model}\ V$	$(O-C)_V$	u (m)	v (m)	HA (h)
54767.41212	0.779	0.100	0.788	-0.008	-59.16	-24.51	-1.95
54767.41988	0.865	0.100	0.790	0.075	-59.42	-25.45	-1.76
54767.42666	0.861	0.073	0.794	0.067	-59.53	-26.26	-1.60
54767.43298	0.820	0.104	0.799	0.021	-59.53	-27.02	-1.44
54767.43920	0.914	0.088	0.805	0.110	-59.45	-27.77	-1.30
54767.46480	0.602	0.098	0.839	-0.237	-58.14	-30.83	-0.68
54767.47392	0.857	0.077	0.855	0.002	-57.30	-31.90	-0.46
54767.48083	0.718	0.115	0.868	-0.150	-56.55	-32.69	-0.29
54767.48751	0.826	0.095	0.881	-0.055	-55.71	-33.45	-0.13
54767.49464	0.825	0.094	0.895	-0.070	-54.71	-34.25	0.04
54767.50635	0.852	0.113	0.918	-0.066	-52.82	-35.53	0.32
54767.51346	0.936	0.172	0.932	0.004	-51.54	-36.28	0.49
54767.51978	0.803	0.100	0.944	-0.141	-50.31	-36.93	0.64
54767.52609	0.902	0.113	0.956	-0.053	-49.01	-37.57	0.80
54767.53284	1.119	0.172	0.966	0.153	-47.53	-38.23	0.96
54768.42211	0.245	0.026	0.247	-0.002	-59.51	-26.04	-1.64
54768.43432	0.256	0.029	0.244	0.011	-59.49	-27.51	-1.35
54768.44509	0.228	0.020	0.245	-0.016	-59.18	-28.81	-1.09
54768.45324	0.245	0.021	0.249	-0.004	-58.76	-29.79	-0.89
54768.46295	0.276	0.035	0.260	0.015	-58.06	-30.93	-0.66
54768.47167	0.342	0.059	0.278	0.063	-57.25	-31.95	-0.45
54793.43328	0.595	0.077	0.576	0.019	-53.18	-35.30	0.27
54793.44167	0.566	0.117	0.618	-0.052	-51.69	-36.20	0.47
54793.44837	0.785	0.115	0.652	0.133	-50.40	-36.89	0.63
54793.45545	0.705	0.088	0.689	0.017	-48.93	-37.60	0.80
54793.46250	0.682	0.074	0.725	-0.044	-47.38	-38.29	0.97
54794.48268	0.710	0.079	0.815	-0.105	-41.71	-40.36	1.52
54794.48894	0.866	0.110	0.822	0.044	-39.99	-40.88	1.68
54794.49560	0.825	0.124	0.829	-0.005	-38.11	-40.40	1.84
54794.50473	0.810	0.136	0.839	-0.029	-35.42	-42.08	2.06
54794.51207	0.807	0.121	0.846	-0.038	-33.16	-42.59	2.23

Table 3.8 lists the visual orbit elements for the above solution. To obtain a good fit to the visibility data, I had to vary T_0 wider than the 1σ limits of the spectroscopic solution. The



FIGURE 3.9: Calibrated visibility measurements for HD 45088 versus the projected baseline. The plus signs are the calibrated visibilities with vertical error bars, and the diamonds are the calculated visibilities for the best-fit orbit. Table 3.7 lists the numeric values corresponding to this plot.

value of epoch of periastron obtained from the visual orbit is 0.57 days or about 2σ away using the uncertainties from the visual orbit but about 20σ away using the lower uncertainties from the spectroscopic solution. Limiting the epoch to within 1σ of the spectroscopic solution leads to a higher χ^2 , but the visual orbit parameters remain well-constrained to the values presented here to within 1σ . While the resulting mass estimates are consistent with the spectral types, the uncertainties are large, due to the relatively large errors for the derived angular semimajor axis (3.4%) as well as for the FvL07 parallax (2.3%).

Orbital Parameter	Value	Source
Adopted values		
Period (days)	6.991888 ± 0.000010	1
e	0.1461 ± 0.0025	1
$\omega ~({ m deg})$	78.9 ± 1.0	1
$\theta_{\rm p} \ ({\rm mas})$	0.49 ± 0.05	2
$\theta_{\rm s}$ (mas)	0.43 ± 0.04	2
Visual orbit parameter	rs	
$T_0 (\text{JD} - 2400000)$	52228.8 ± 0.3	1
$\alpha ~({\rm mas})$	5.61 ± 0.19	3
$i \; (deg)$	108.5 ± 2.5	3
$\Omega ~(\mathrm{deg})$	125 ± 4	3
$\Delta K'$	0.53 ± 0.04	3
Reduced χ^2	0.75	3
Physical parameters		
$M_p~({ m M}_{\odot})$	0.831 ± 0.101	3
M_s (${ m M}_{\odot}$)	0.709 ± 0.087	3

TABLE3.8: Visual Orbit Solution for HD 45088

NOTES.—Source codes: 1 = From the spectroscopic solution in §6.2.2; 2 = Estimated using Gray et al. (2003) spectral types and FvL07 parallaxes, with an adopted 10% error; 3 = This work.

3.2.4 The Visual Orbit of HD 223778

Forty eight calibrated visibility measurements were obtained over nine nights for this star using the CHARA Array's S1–S2 and E1–E2 baselines in 2008 June–November. The calibrator used for all nights was HD 219485, an apparent single star with an estimated angular diameter of 0.224 ± 0.003 mas, obtained by fitting BVRIJHK photometry to spectral energy distribution (SED) models from R. L. Kurucz, available at http://cfaku5.cfa.harvard.edu. Table 3.9 lists the 48 observed visibilities that were used in deriving the visual orbit, in the same format as in Table 3.5. Figure 3.10 plots the observed visibilities along with their uncertainties and corresponding model values, showing the excellent phase coverage of these data.

MJD	Obs V	σ_V	Model V	$(O-C)_V$	<i>u</i> (m)	v (m)	HA (h)
54639.43052	0.663	0.067	0.639	0.024	10.54	22.61	-3.37
54639.44842	0.646	0.070	0.661	-0.015	8.66	23.65	-2.94
54639.46030	0.659	0.078	0.681	-0.022	7.34	24.24	-2.66
54639.47252	0.710	0.093	0.704	0.005	5.95	24.73	-2.36
54639.48446	0.711	0.068	0.730	-0.019	4.57	25.12	-2.07
54639.49592	0.695	0.060	0.756	-0.062	3.21	25.39	-1.80
54640.42254	0.798	0.068	0.922	-0.124	11.07	22.26	-3.50
54640.43569	0.827	0.066	0.950	-0.123	9.73	23.09	-3.19
54640.44800	0.878	0.084	0.972	-0.094	8.41	23.77	-2.89
54640.46041	0.835	0.107	0.988	-0.153	7.04	24.35	-2.59
54640.47249	0.859	0.106	0.996	-0.137	5.66	24.82	-2.30
54640.48496	0.857	0.075	0.998	-0.141	4.19	25.20	-2.00
54640.49713	0.865	0.095	0.992	-0.126	2.75	25.46	-1.71
54642.48679	0.540	0.038	0.496	0.044	3.32	25.37	-1.82
54653.43650	0.561	0.048	0.610	-0.049	5.61	24.84	-2.29
54653.44289	0.534	0.044	0.583	-0.049	4.86	25.04	-2.13
54653.44951	0.471	0.078	0.556	-0.085	4.09	25.22	-1.98
54653.45650	0.550	0.061	0.528	0.022	3.25	25.38	-1.81
54653.46249	0.510	0.041	0.504	0.006	2.53	25.49	-1.66
54653.46877	0.438	0.048	0.481	-0.044	1.78	25.57	-1.51
54653.47551	0.448	0.060	0.458	-0.010	0.96	25.62	-1.35
54653.48168	0.483	0.044	0.437	0.046	0.21	25.65	-1.20
54653.48768	0.457	0.051	0.418	0.039	-0.52	25.64	-1.06
54653.49392	0.469	0.048	0.400	0.069	-1.27	25.61	-0.51
54654.39421	0.677	0.126	0.574	0.103	9.97	22.95	-3.24
54654.40082	0.590	0.084	0.565	0.024	9.28	23.34	-3.08
54654.40754	0.547	0.043	0.558	-0.011	8.55	23.70	-2.92
54654.41317	0.578	0.058	0.553	0.025	7.94	23.99	-2.78
54654.41906	0.542	0.030	0.549	-0.007	7.28	24.26	-2.64
54674.31861	0.634	0.053	0.651	-0.017	12.05	21.54	-3.75
54674.32518	0.660	0.062	0.655	0.005	11.42	22.01	-3.59

TABLE 3.9: Interferometric Visibilities for HD 223778

Continued on Next Page...

MJD	$\mathbf{Obs}~V$	σ_V	$\mathbf{Model}\ V$	$(O-C)_V$	u (m)	v (m)	HA (h)
54674.33108	0.564	0.045	0.660	-0.097	10.84	22.41	-3.45
54674.33718	0.654	0.049	0.666	-0.012	10.22	22.80	-3.30
54674.34314	0.702	0.042	0.673	0.029	9.60	23.16	-3.16
54674.40283	0.743	0.046	0.783	-0.040	2.81	25.45	-2.72
54674.40866	0.719	0.072	0.796	-0.078	2.11	25.54	-1.58
54674.41447	0.772	0.078	0.810	-0.038	1.41	25.60	-1.44
54792.20403	0.698	0.042	0.614	0.084	-44.68	-45.15	1.25
54792.21701	0.762	0.052	0.806	-0.045	-41.28	-48.59	1.56
54793.14329	1.000	0.102	0.961	0.039	-55.78	-27.22	-0.15
54793.15142	0.973	0.097	0.986	-0.013	-54.64	-29.96	0.05
54793.15817	0.907	0.076	0.986	-0.078	-53.58	-32.19	0.21
54793.16409	0.917	0.075	0.969	-0.052	-52.57	-34.10	0.36
54793.17011	0.932	0.052	0.937	-0.005	-51.48	-36.02	0.50
54794.12873	0.141	0.015	0.147	-0.006	-57.17	-23.14	-0.43
54794.13731	0.295	0.044	0.188	0.107	-56.20	-26.07	-0.23
54794.15074	0.303	0.034	0.308	-0.005	-54.32	-30.64	0.10
54794.16061	0.357	0.032	0.398	-0.041	-52.71	-33.86	0.34

TABLE 3.9 - Continued

Table 3.10 lists the visual orbit elements for the above solution. If wider excursions are allowed for P, e, and T_0 , the visual orbit seems to prefer a smaller period and larger eccentricity, but the visual orbit elements of interest remain very well constrained to the values presented here, which were obtained by limiting the spectroscopic parameters to 1σ variation. The resulting component mass estimates have uncertainties below 2%.

TABLE 3.10: Visual Orbit Solution for HD 223778

Value	Source
	_
7.75442 ± 0.00052	1
0.0162 ± 0.0035	1
	Value 7.75442 ± 0.00052 0.0162 ± 0.0035

Continued on Next Page...

TABLE 3.10 - Continued

Orbital Parameter	Value	Source
$T_0 (\text{JD} - 2400000)$	52232.28 ± 0.23	1
$\omega (\mathrm{deg})$	279 ± 11	1
$\theta_{\rm p} \ ({\rm mas})$	0.66 ± 0.06	2
$\theta_{\rm s}$ (mas)	0.62 ± 0.06	2
Visual orbit parameters		
$\alpha \ ({\rm mas})$	8.12 ± 0.04	3
i (deg)	50.2 ± 0.4	3
$\Omega ~(\mathrm{deg})$	126.8 ± 1.3	3
$\Delta K'$	0.33 ± 0.03	3
Reduced χ^2	1.12	3
Physical parameters		
M_p (M_{\odot})	0.786 ± 0.014	3
$\dot{M_s}$ (${ m M}_{\odot}$)	0.748 ± 0.014	3

NOTES.—Source codes: 1 = From the spectroscopic solution in §6.2.2; 2 = Estimated using Gray et al. (2003) spectral types and FvL07 parallaxes, with an adopted 10% error; 3 = This work.



FIGURE 3.10: Calibrated visibility measurements for HD 223778 versus the orbital phase. The plus signs are the calibrated visibilities with vertical error bars, and the diamonds are the calculated visibilities for the best-fit orbit. Table 3.9 lists the numeric values corresponding to this plot.

The true delight is in the finding out rather than in the knowing. —— Isaac Asimov

Common Proper Motion Results

– 4 –

Visual companions span the widest range of separations among binaries, with companions detected within a milli-arcsec by the high resolution techniques covered in the previous chapter and out to several hundred arcseconds by techniques such as common proper motion (CPM) detection. This chapter describes the methods of this effort for observing these wide binaries and presents their results.

4.1 Identification and Confirmation of CPM Companions

Nearby stars, like the ones of the current sample, generally have a larger proper motion than their distant counterparts. This is due to relative motion between the the Sun and the stars, which is visually more pronounced for nearby stars, much like the larger apparent motion of closer trees when riding in a train. This large proper motion becomes a handy tool in identifying widely separated companions, which can be readily identified as stars with similar apparent movement across the sky. The technique involves blinking two digital images, taken many years apart, of the sky around each primary to identify CPM pairs. Candidate companions thus revealed can then be confirmed of refuted by follow-up photometry. This method is not effective for discovering close companions because the bright primaries saturate the images out to several accesconds. For the wide pairs suited to this technique, orbital



FIGURE 4.1: Example of the images blinked to identify CPM companions. The 10' square images from the DSS are for HD 9826 (v And) with north up and east to the left. The epoch of the left image is 1953.71 and for the right image is 1989.77. The arrow marks the CPM companion at a separation of 56" and the lines identify WDS entries which are field stars. The primary's proper motion is 0".42 yr⁻¹ at 204°.

periods are upwards of several thousand years, and any orbital motion is negligible over the time interval between the two images. Hence, gravitationally bound stars are seen as linked pairs moving in unison across the field of generally more distant and largely stationary stars in the images. Figure 4.1 shows the two images blinked for HD 9826 (v And), allowing the identification of the CPM companion (marked by the arrow), as well as the detection of two WDS entries as unrelated field stars with minimal proper motion (marked by vertical lines). The effect, while noticeable in these printed images, is readily apparent when they are blinked numerous times in quick succession.

The primary source of the archival images leveraged in this effort is the multi-epoch STScI Digitized Sky Survey¹ (DSS). This resource provides all-sky images for multiple epochs, which

¹http://stdatu.stsci.edu/cgi-bin/dss_form



FIGURE 4.2: The epoch distribution of the DSS and SSS images used for blinking to identify CPM companions. A pair of images were blinked for each target, with the earlier epoch identified by the solid line and the later epoch by the dashed line.

can be downloaded in FITS format for a specified central position and size. In some cases, when the time interval between the two DSS images was not sufficient to easily identify the proper motion of the primary star, SuperCOSMOS Sky Survey (Hambly et al. 2001, SSS) images were used for the earlier epoch, significantly increasing the apparent motion seen upon blinking. Figure 4.2 shows the epoch-distribution of the images blinked. The earlier epoch distribution (solid lines) is bimodal, with 266 frames from 1949–1957 with a median of 1953.77 and 187 frames from 1974–1989 with a median of 1979.71. The later epoch (dashed

line) is more tightly constrained, with all 453 frames obtained during 1984–2000 with a median 1992.28.



FIGURE 4.3: The distribution of the time interval between the images blinked for each target.

The relevant parameters that determine the effectiveness of this method are the proper motion of the primary star and the time interval between the blinked images for each target, which result in the total motion seen upon blinking. While the characteristics of the individual images such as the relative translation or rotation between the frames and the field of stars available determine the limits of motion detectable upon blinking, experience has shown that total motions under 2" are seldom detectable and those under ~ 4 " are often only marginally detectable. Figure 4.3 shows the distribution of the time interval between the images blinked, which again is bimodal. For 187 targets, the images used span a time interval of 1–22 years with a median of 14.1 years, and the remaining 266 targets span of 28–49 years with a median of 38.1 years. Figure 4.4 shows the distribution of the primary's total transverse motion over the duration spanning the images blinked, excluding 12 targets with motions greater than 50". Forty six targets have transverse motions under 2" and 97 targets have motions under 4". As discussed above, detectability of the motion is also affected by individual characteristics of the images used, so these numbers only serve as a rough guide and as a test of the results. When blinking the images, I made individual notes about the detectability of the motion. Of the 453 image-pairs blinked, 44 were flagged as targets whose motion could not be detected and an additional 43 were noted as marginally detectable. These numbers correspond well to the limits identified above. For 366 of the 453 targets investigated, this technique proved to be fully effective in determining CPM companions, and somewhat effective for an additional 43, providing excellent coverage.

This technique is well suited to search for CPM companions wider than $\sim 15-30''$ from the primary, as companions closer than this are often buried in the saturation around the bright primaries. However, in many instances, bright companions inside the saturation region can be identified by twin, comoving diffraction spikes. Choosing an outer limit is a matter of picking the optimum size of the images blinked. A larger image size will naturally cover a larger search space, but this gain is offset by a diminishing detection capability due to a smaller apparent motion. Additionally, while the reduction code attempts to compensate



FIGURE 4.4: The distribution of the transverse motion of the primary star seen on blinking the images.

for any translation or rotation between the images blinked, the residual effects could be more pronounced for larger images, causing even the stationary stars to wobble, hindering the detection of CPM candidates. To optimize across these effects, I followed a two-step process as described below. First, I blinked images, 10' on each side, to thoroughly inspect a search radius of 300" around each primary. Then, systems closer than 20 pc were blinked again using 22' square images to expand the search region out to 660", while systems at a distance of 20–25 pc were blinked using 15' square images to explore out to 450". The larger angular size searched for the nearby systems was chosen to enable a similar-sized *linear* search space for all stars. The subsample of stars within 20 pc contains 238 primaries and has a median distance of 15.6 pc, translating the 660" angular radius searched to a linear radius of approximately 10,000 AU, while the farther subsample of 215 primaries has a median distance of 22.4 pc, also corresponding to a linear radius of $\sim 10,000$ AU for the angular radius of 450". Images larger than 15' were blinked as four sub-images of 15' square from each corner of the image, allowing for a closer inspection. Figure 4.5 shows the distribution of the linear projected separation searched around each primary, demonstrating that most systems were effectively searched out to a separation of ~ 10000 AU. While companions farther than this separation limit have been reported (e.g., Latham et al. 1991; Poveda et al. 1994), the range selected here enables an effective search of the region containing the vast majority of companions. Only 18 of the closest systems were inspected out to less than a 5000 AU separation from the primary. The blinking was performed with an IDL code that compensated for any relative translation or rotation between the images, scaled them appropriately, displayed a subtracted image and blinked them 15 times with a 0.1 second interval at the user's request to help identify companions. The relative position of the candidate companions thus identified were then captured by clicking first on the primary and then on the companion. The code is included in Appendix C.4.

The 366 primaries adequately inspected using this technique revealed 78 candidate companions among 73 primaries, while the remaining 293 primaries revealed no CPM candidates. Fifty three of these were noted as compelling, and the remaining 25 were identified as possibilities deserving follow-up efforts. Table 4.1 summarizes the results of this effort, listing each



FIGURE 4.5: The distribution of the linear projected separation from the primary searched for CPM companions.

primary (Column 1) along with the epochs of the images used for blinking (Columns 2 and 3), the FvL07 proper motion of the primary (Columns 4 and 5), the total motion between the image-epochs (Column 6), and the blink result (Column 7) identifying whether the motion was detectable and if any candidate companions were identified (see the table footnote for an explanation of the codes used in this column). Column 8 lists the final status of the companion, confirmed companions are flagged as "Yes", and refuted candidates as "No". The methods used to make this determination are described in the following paragraph, and individual footnotes in the table identify the reason for each situation. The last column

identifies the candidate companion either by its HD, HIP, or other name, or by its position relative to the primary. The separation and position angles listed for the companions are measured from the more recent image and are approximate. They are good to within an arcsecond and a degree, respectively for wide companions, but the errors could be larger for close pairs identified by overlapping PSF or twin diffraction spikes. A few pairs, wider than the limits inspected as described above, have nevertheless been published by other efforts, especially in the CNS catalogs. An adequately large image was blinked to check if they exhibit CPM as well, and pairs passing this test are also included in Table 4.1.

	Image	Epochs	μ_{lpha}	μ_{δ}	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$({ m mas}~{ m yr}^{-1})$	$({ m mas}~{ m yr}^{-1})$	μ	Result	Status	ID
HD 000123	1954.75	1992.73	247.24 ± 0.48	15.68 ± 0.40	9.4	NComp	:	: .
HD 000166	1954.83	1989.83	380.98 ± 0.57	-178.68 ± 0.29	14.7	Comp?	No^a	$311'' { m at} 147^{\circ}$
HD 000870	1975.84	1991.85	-118.80 ± 0.48	35.12 ± 0.57	2.0	Marg		
HD 001237	1977.77	1997.58	433.92 ± 0.30	-56.74 ± 0.29	8.7	NComp	•	
HD 001273	1980.79	1996.84	314.38 ± 0.42	182.44 ± 0.57	5.8	NComp		
HD 001461	1983.77	1996.84	416.87 ± 0.63	-143.83 ± 0.26	5.8	NComp		
HD 001562	1954.76	1989.75	-140.48 ± 0.57	-274.97 ± 0.48	10.9	NComp		
HD 001581	1977.87	1993.64	1707.42 ± 0.15	1164.30 ± 0.16	32.6	Comp?	No^b	$87'' \text{ at } 343^{\circ}$
HD 001835	1983.77	1996.84	394.45 ± 0.49	61.14 ± 0.29	5.2	NComp		
HD 002025	1977.70	1994.62	665.64 ± 1.03	83.67 ± 0.52	11.4	NComp	:	
HD 002151	1977.77	1997.58	2219.54 ± 0.11	324.09 ± 0.11	44.4	NComp	:	
HD 003196	1982.71	1995.90	408.34 ± 0.81	-35.22 ± 0.32	5.4	NComp	:	
HD 003443	1980.63	1989.74	1450.34 ± 3.77	-19.38 ± 1.77	13.0	NComp	:	
HD 003651	1953.91	1987.65	-461.32 ± 0.33	-370.02 ± 0.28	20.0	NComp	:	
HD 003765	1985.96	1989.75	355.07 ± 0.65	-668.18 ± 0.57	2.9	NoPM	:	
HD 004256	1954.67^{c}	1987.73	-49.08 ± 0.60	-573.23 ± 0.47	19.0	NComp	:	
HD 004308	1977.78	1987.82	157.49 ± 0.27	-742.32 ± 0.31	7.6	NComp	:	
HD 004391	1975.82	1993.70	183.99 ± 0.34	78.81 ± 0.30	3.6	Comp	$Y_{\mathrm{eS}}{}^{d,e}$	$49''$ at 240°
HD 004614	1952.71	1991.68	1086.59 ± 0.40	-559.43 ± 0.33	47.7	NComp	÷	•
HD 004628	1953.83	1994.61	757.11 ± 0.48	-1141.33 ± 0.34	55.9	NComp	:	• •
HD 004635	1954.73	1995.59	369.83 ± 0.52	201.35 ± 0.39	17.2	NComp	:	
HD 004676	1954.00	1990.70	-2.87 ± 0.29	-202.05 ± 0.24	7.4	NComp	:	• •
HD 004747	1980.63	1989.74	516.92 ± 0.55	120.05 ± 0.45	4.8	NComp	÷	• •
HD 004813	1954.68^{c}	1989.83	-226.91 ± 0.36	-229.75 ± 0.23	11.2	NComp	:	
HD 004915	1982.71	1995.90	261.10 ± 0.60	-119.49 ± 0.45	3.8	NComp	•	
Continued or	1 Next Page							

TABLE 4.1: Archival Image Blink Results

125

	Image	Enochs	Ш.	511	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(\max yr^{-1})$	$(\max {\mathbf{yr}}^{-1})$	μ	Result	Status	ID
HD 005015	1952.71	1991.68	-68.95 ± 0.23	170.15 ± 0.22	7.1	NComp		
HD 005133	1978.82	1988.76	620.35 ± 0.63	30.26 ± 0.48	6.2	NComp		:
HD 006582	1954.01	1990.79	3422.23 ± 0.60	-1598.93 ± 0.89	138.9	NComp	:	:
HD 007570	1979.88	1993.70	665.13 ± 0.20	177.63 ± 0.19	9.5	NComp	:	:
HD 007590	1952.81	1989.75	-112.73 ± 0.35	-31.28 ± 0.37	4.3	Marg		:
HD 007693	1975.83	1989.73	417.23 ± 0.81	75.32 ± 0.78	5.9	Comp	Yes^{f}	HD 7788
HD 007924	1954.72	1995.81	-34.23 ± 0.43	-33.28 ± 0.48	2.0	NoPM	:	:
HD 008997	1954.74	1990.71	456.37 ± 0.86	-183.83 ± 0.57	17.7	NComp		:
HD 009407	1953.83	1991.68	-377.76 ± 0.31	113.87 ± 0.34	15.0	NComp	:	:
HD 009540	1983.75	1996.72	271.95 ± 0.52	-158.99 ± 0.32	4.1	Comp?	NO^{g}	$336^{\prime\prime} \mathrm{~at~} 220^{\circ}$
HD 009770	1977.92	1996.61	115.48 ± 3.47	115.49 ± 1.97	2.4	Marg		:
HD 009826	1953.71	1989.77	-173.33 ± 0.20	-381.80 ± 0.13	15.1	Comp	Yes^h	$55''$ at 150°
HD 010008	1982.65	1997.90	171.30 ± 0.67	-98.50 ± 0.53	3.0	Marg		:
HD 010086	1953.78	1989.75	217.49 ± 0.48	-228.62 ± 0.56	11.3	NComp	:	:
HD 010307	1953.71	1989.77	791.47 ± 0.48	-180.80 ± 0.36	29.3	NComp	:	:
HD 010360	1977.92	1997.61	282.16 ± 2.16	10.56 ± 1.88	5.6	Comp	${ m Yes}^{i,j}$	$13^{\prime\prime}\mathrm{at}5^{\circ}$
HD 010476	1954.89	1986.69	-302.42 ± 0.60	-678.88 ± 0.41	23.6	NComp	:	:
HD 010647	1977.92	1997.61	166.32 ± 0.24	-106.52 ± 0.27	3.9	Marg	:	:
HD 010700	1982.73	1991.68	-1721.05 ± 0.18	854.16 ± 0.15	17.2	NComp	:	:
HD 010780	1953.83	1991.70	582.03 ± 0.35	-246.93 ± 0.43	23.9	NComp	:	:
HD 012051	1951.84	1989.92	243.58 ± 0.59	-352.17 ± 0.48	16.3	NComp	:	:
HD 012846	1953.78	1990.87	5.80 ± 0.65	-147.35 ± 0.44	5.5	Comp?	NO^{g}	$304'' \text{ at } 336^{\circ}$
HD 013445	1975.85	1988.91	2092.86 ± 0.27	653.21 ± 0.30	28.6	NComp	:	:
HD 013974	1951.84	1986.91	1151.83 ± 0.40	-246.89 ± 0.30	41.3	NComp	:	:
HD 014214	1982.66	1995.66	366.69 ± 0.52	370.63 ± 0.36	6.8	NComp	:	:
HD 014412	1976.89	1995.82	-217.55 ± 0.37	444.44 ± 0.28	9.4	NComp	:	:
HD 014802	1976.89	1995.82	196.61 ± 0.81	-4.98 ± 0.58	3.7	Marg		:
Continued or	ı Next Pag∈							

TABLE 4.1 – Continued

TABLE 4.1 -	- Continued							
Name	Image Epoch 1	Epochs Epoch 2	μ_lpha (mas ${f yr}^{-1}$)	$\mu_{\delta} \ ({ m max} \ { m yr}^{-1})$	$\begin{array}{c} \text{Total} \\ \mu \end{array}$	Blink Result	Comp Status	anion ID
HD 016160	1954.97	1990.73	1807.78 ± 0.89	1444.02 ± 0.40	82.7	Comp	Y_{es^d}	$162'' { m at} 110^{\circ}$
HD 016287	1982.79	1997.74	323.60 ± 1.10	58.00 ± 1.16	4.9	NComp	:	:
HD 016673	1979.72	1993.85	-138.88 ± 0.47	-79.97 ± 0.56	2.3	Marg	:	
HD 016739	1951.97	1989.67	-17.20 ± 0.43	-183.30 ± 0.43	6.9	NComp		:
HD 016765	1982.78	1995.95	216.51 ± 1.23	-129.33 ± 1.09	3.3	Marg	:	•
HD 016895	1953.77	1990.94	334.66 ± 0.17	-89.99 ± 0.17	12.9	NComp		:
HD 017051	1977.77	1997.81	332.82 ± 0.19	218.74 ± 0.18	8.0	NComp		:
HD 017382	1950.94	1989.98	266.58 ± 1.29	-124.97 ± 0.77	11.5	Comp	Y_{es}^{d}	$21^{\prime\prime}{ m at}27^{\circ}$
HD 017925	1982.71	1995.90	397.41 ± 0.45	-189.32 ± 0.36	5.8	NComp		:
HD 018143	1950.94	1989.98	264.78 ± 1.17	-193.91 ± 0.76	12.7	Comp	Yes^{f}	NLTT 9303
HD 018632	1955.04	1990.81	327.42 ± 1.59	19.31 ± 1.30	11.8	NComp		:
HD 018757	1954.07	1993.94	720.57 ± 0.60	-695.55 ± 0.61	39.9	Comp	Y_{es}^d	$262^{\prime\prime} \mathrm{~at~} 68^{\circ}$
HD 018803	1954.89	1989.98	233.13 ± 0.45	-168.88 ± 0.36	10.1	NComp	:	•
HD 019373	1953.77	1989.76	1262.41 ± 0.17	-91.50 ± 0.19	45.6	NComp	:	•
HD 019994	1951.68	1997.85	194.56 ± 0.37	-69.01 ± 0.30	9.5	NComp	:	•
HD 020010	1977.95	1991.93	370.87 ± 0.30	611.33 ± 0.42	10.0	NComp	:	•
HD 020165	1955.87	1991.86	399.63 ± 1.05	-405.51 ± 0.87	20.5	NComp	:	•
HD 020407	1977.78	1995.00	-132.28 ± 0.36	142.00 ± 0.48	3.3	Marg	:	•
HD 020619	1982.73	1998.97	253.64 ± 0.96	-101.49 ± 0.72	4.4	NComp	:	•
HD 020630	1951.68	1990.74	269.30 ± 0.24	93.75 ± 0.22	11.1	NComp	:	•
HD 020794	1977.78	1995.00	3038.34 ± 0.20	726.58 ± 0.21	53.8	NComp	:	•
HD 020807	1977.78	1996.87	1330.74 ± 0.21	647.11 ± 0.19	28.2	Comp	Y_{esf}	HD 20766
HD 021175	1976.96	1991.79	40.76 ± 0.61	43.06 ± 0.62	0.9	NoPM	:	•
HD 022049	1982.79	1998.97	-975.17 ± 0.21	19.49 ± 0.20	15.8	NComp	:	•
HD 022484	1954.00^{c}	1990.80	-232.60 ± 0.59	-481.92 ± 0.54	19.7	NComp	:	•
HD 022879	1955.88^{c}	1987.87	689.15 ± 0.83	-213.18 ± 0.61	23.0	NComp	:	•
HD 023356	1955.88^{c}	1991.91	309.90 ± 0.67	157.77 ± 0.66	12.5	NComp		:
Continued on	n Next Page							
TABLE 4.1	– Continuec	J						
-------------	------------------	-------------------	----------------------------------	--	-------	-----------------	--------------------------	---
Name	Image Enoch 1	Epochs Enoch 2	μ_lpha (mas ${ m vr}^{-1}$)	μ_{δ} (mas vr^{-1})	Total	Blink Besult	Comp	anion TD
		- Hooder	(If contra)	(If control	d.		anon a	1
HD 023484	1977.71	1991.91	208.60 ± 0.41	288.38 ± 0.55	5.0	Marg	:	•
HD 024238	1954.01	1992.84	437.05 ± 0.77	-245.46 ± 0.84	19.5	NComp	:	•
HD 024409	1954.01	1992.84	-285.44 ± 0.58	159.82 ± 0.61	12.6	NComp	:	:
HD 024496	1951.91	1992.84	217.60 ± 0.78	-165.28 ± 0.52	11.2	NComp		:
HD 025329	1954.97	1993.71	1732.84 ± 1.19	-1365.72 ± 0.90	85.5	NComp		:
HD 025457	1951.01^{c}	1989.99	149.04 ± 0.42	-253.02 ± 0.43	11.5	NComp	:	
HD 025665	1953.79	1993.72	72.59 ± 0.38	-298.28 ± 0.64	12.3	Comp?	No^a	$287'' { m at} 134^{\circ}$
HD 025680	1949.97	1989.97	172.47 ± 0.40	-131.27 ± 0.34	8.6	Comp?	NO^{a}	$174^{\prime\prime} { m at} { m 0}^{\circ}$
HD 025998	1955.81	1989.76	164.10 ± 0.25	-202.60 ± 0.20	8.9	Comp	γ_{es}^{f}	HD 25893
HD 026491	1979.87	1989.98	185.73 ± 0.28	336.98 ± 0.34	3.9	Marg	•	•
HD 026923	1954.97	1986.77	-109.46 ± 0.48	-108.25 ± 0.43	4.9	Comp?	No^{g}	$223^{\prime\prime} \mathrm{~at~} 63^{\circ}$
		:	:		:	Comp	Y_{es}^{f}	HD 26913
HD 026965	1982.82	1985.96	-2240.12 ± 0.23	-3420.27 ± 0.20	12.8	Comp	γ_{es}	LHS 25
:		:				Comp	${ m Y}_{{ m es}^{i,j}}$	$80'' ext{ at } 100^{\circ}$
HD 029883	1954.99	1989.90	53.34 ± 0.87	-265.47 ± 0.60	9.5	NComp	:	•
HD 030495	1984.91	1990.90	130.04 ± 0.37	169.27 ± 0.28	1.3	NoPM	:	•
HD 030501	1983.01	1997.02	-446.09 ± 0.46	-336.70 ± 0.52	7.8	NComp	:	•
HD 030876	1977.85	1997.01	-26.13 ± 0.35	-115.26 ± 0.52	2.2	Marg	:	•
HD 032778	1986.97	1989.98	-45.43 ± 0.41	731.74 ± 0.42	2.2	Comp?	Yes^d	$81'' \text{ at } 145^{\circ}$
HD 032850	1955.96	1989.84	281.27 ± 0.95	-239.00 ± 0.66	12.5	NComp	:	•
HD 032923	1955.88	1992.64	534.73 ± 0.41	17.93 ± 0.18	19.7	NComp	:	•
HD 033262	1980.79	1989.03	-30.97 ± 0.19	117.22 ± 0.19	1.0	NoPM	:	•
HD 033564	1955.04	1998.00	-78.48 ± 0.15	161.95 ± 0.22	7.7	NComp	:	•
HD 034411	1953.12	1988.85	518.99 ± 0.26	-665.06 ± 0.13	30.1	NComp	:	•
HD 034721	1980.94	1997.02	386.13 ± 0.35	61.03 ± 0.36	6.3	NComp	:	•
HD 035112	1953.91	1991.79	56.17 ± 1.29	-139.35 ± 0.77	5.7	NComp	:	•
HD 035296	1953.02	1989.91	251.05 ± 0.38	-7.99 ± 0.21	9.2	Comp	Yes^{f}	HD 35171
Continued o	n Next Pag	e						

	Image	Epochs	μ_{α}	μδ	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(\max \mathbf{yr}^{-1})$	$(\max yr^{-1})$	μ	Result	Status	ID
HD 035854	1976.99	1992.00	245.67 ± 0.37	-91.06 ± 0.65	3.9	Marg	:	:
HD 036435	1976.83	1994.10	-148.35 ± 0.53	-93.38 ± 0.60	3.0	Marg		
HD 036705	1987.07	1993.89	33.16 ± 0.39	150.83 ± 0.73	1.1	NoPM	•	•
HD 037008	1953.11	1989.77	-549.16 ± 0.86	106.64 ± 0.62	20.5	NComp		
HD 037394	1953.11	1991.11	1.82 ± 0.48	-523.99 ± 0.31	19.9	Comp	Yes^{f}	HD 233153
HD 037572	1976.82	1997.02	25.40 ± 1.65	-3.38 ± 1.45	0.5	Comp?	Yes^{f}	HIP 26369
HD 038230	1954.90	1989.76	486.82 ± 0.84	-509.74 ± 0.50	24.6	NComp		
HD 038858	1984.01	1990.97	60.84 ± 0.41	-228.35 ± 0.33	1.7	Marg	:	
HD 039091	1978.03	1989.99	312.01 ± 0.24	1050.38 ± 0.26	13.1	NComp	:	•
HD 039587	1951.91	1990.81	-162.54 ± 0.28	-99.51 ± 0.16	7.4	NComp		
HD 039855	1982.87	1992.08	93.36 ± 0.72	-27.10 ± 0.77	0.9	NoPM		
HD 040307	1976.83	1994.10	-52.65 ± 0.46	-60.46 ± 0.42	1.4	NoPM	•	•
HD 040397	1983.99	1989.02	70.52 ± 0.69	-203.16 ± 0.37	1.1	NoPM	:	•
HD 041593	1955.90	1990.87	-120.46 ± 0.71	-103.21 ± 0.43	5.6	NComp	:	•
HD 042618	1950.94	1989.85	197.44 ± 0.60	-253.65 ± 0.39	12.5	NComp	÷	•
HD 042807	1955.94	1989.85	77.38 ± 0.58	-298.00 ± 0.34	10.4	NComp	:	•
HD 043162	1981.02	1996.04	-47.05 ± 0.28	110.87 ± 0.42	1.8	Comp?	$Y_{es}^{d,e}$	$164'' { m at} 171^{\circ}$
HD 043587	1950.94	1990.82	-187.72 ± 0.37	170.69 ± 0.28	10.2	Comp	Y_{es}^{d}	$102^{\prime\prime} { m at} 306^{\circ}$
HD 043834	1977.11	1990.00	121.80 ± 0.14	-212.34 ± 0.16	3.2	NComp	:	:
HD 045088	1955.86	1996.77	-117.60 ± 1.23	-163.48 ± 0.84	8.3	NComp	:	•
HD 045184	1979.00	1992.99	-164.99 ± 0.31	-121.77 ± 0.39	2.9	NComp	:	•
HD 045270	1979.97	1995.07	-11.29 ± 0.35	64.24 ± 0.30	1.0	NoPM		•
HD 046588	1954.76	1997.18	-99.16 ± 0.19	-603.76 ± 0.26	25.9	NComp		
HD 048189	1979.97	1995.07	-47.84 ± 1.04	72.73 ± 0.87	1.3	NoPM		•
HD 048682	1953.19	1989.90	-1.07 ± 0.36	164.25 ± 0.23	6.1	NComp	:	•
HD 050692	1956.27	1994.03	-35.91 ± 0.47	25.65 ± 0.29	1.7	Marg	:	•
HD 051419	1956.27	1989.84	48.74 ± 0.80	99.14 ± 0.56	3.7	NComp	•	••••
Continued on	n Next Page							

TABLE 4.1	- Continuec	ł						
t e	Image	Epochs	μ_{α}	$\mu_{\delta} = 0$	Total	Blink	Comp	anion
Name	Epocn 1	Epoch 2	(mas yr ⁺)	(mas yr ⁻)	μ	Kesult	Status	Π
HD 051866	1953.19	1989.93	544.86 ± 0.92	-430.52 ± 0.65	25.6	NComp	:	:
HD 052698	1979.97	1994.20	206.42 ± 0.36	40.82 ± 0.60	3.0	NComp		:
HD 052711	1955.15	1988.13	155.15 ± 0.51	-828.66 ± 0.34	27.9	NComp	:	
HD 053143	1978.04	1992.17	-161.59 ± 0.35	264.67 ± 0.41	4.4	NComp	:	:
HD 053705	1980.12	1994.93	-104.10 ± 0.91	389.07 ± 1.32	6.0	Comp	Y_{es}^{f}	HD 53680
	:	:	:			Comp	γ_{esf}	HD 53706
HD 053927	1955.15	1988.13	-157.81 ± 1.16	-294.71 ± 0.70	11.0	NComp	:	
HD 054371	1956.27	1994.03	-122.89 ± 0.56	-174.66 ± 0.40	8.1	NComp	:	
HD 055575	1953.12	1989.92	29.06 ± 0.36	-186.43 ± 0.23	6.9	Marg	:	
HD 057095	1978.17	1992.17	-18.24 ± 0.62	584.72 ± 0.63	8.2	NComp	:	:
HD 059468	1978.02	1992.18	-286.36 ± 0.40	-2.50 ± 0.42	4.0	NComp	:	
HD 059747	1955.12	1990.08	-51.41 ± 1.58	7.65 ± 0.76	1.8	NoPM		:
HD 059967	1977.07	1991.88	-87.27 ± 0.27	53.93 ± 0.34	1.5	NoPM	:	
HD 060491	1984.01	1985.96	-81.17 ± 1.26	-42.66 ± 0.66	0.2	NoPM	:	
HD 061606	1955.90^{c}	1985.96	69.90 ± 0.71	-278.33 ± 0.31	8.6	Comp	Y_{es^d}	$57''$ at 112°
HD 062613	1954.90	1998.96	-476.24 ± 0.27	88.83 ± 0.32	21.3	NComp	:	
HD 063077	1977.07	1991.88	-221.54 ± 0.41	1722.11 ± 0.51	25.7	Comp	Y_{es^d}	$914^{\prime\prime} { m at} 6^{\circ}$
HD 063433	1955.20	1999.20	-8.31 ± 0.65	-10.48 ± 0.46	0.7	NoPM	÷	•
HD 064096	1955.95^{c}	1984.98	-67.75 ± 0.73	-346.66 ± 0.66	10.2	NComp	:	•
HD 064468	1955.94	1998.00	93.00 ± 0.97	-454.43 ± 0.53	19.5	NComp	:	
HD 064606	1983.04	1988.19	-250.14 ± 2.22	-59.77 ± 1.71	1.3	NoPM	:	• •
HD 065430	1950.94	1997.18	181.25 ± 0.63	-544.95 ± 0.34	26.5	NComp	:	
HD 065583	1955.20	1989.84	-171.26 ± 0.61	-1165.34 ± 0.49	40.8	NComp	:	• •
HD 065907	1977.22	1991.21	517.51 ± 0.28	120.02 ± 0.25	7.4	Comp	Y_{es^d}	$61'' { m at} 74^{\circ}$
HD 067199	1976.25	1991.13	-158.06 ± 0.32	-130.33 ± 0.49	3.0	Marg	÷	•
HD 067228	1950.94	1997.18	23.00 ± 0.33	-66.42 ± 0.24	3.3	NoPM	:	
HD 068017	1955.22	1988.95	-462.57 ± 0.72	-644.20 ± 0.38	26.7	NComp	÷	•
Continued of	n Next Pag	e						

	Image	Epochs	π_{α}	Шs	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(\max yr^{-1})$	$(\max yr^{-1})$	μ	Result	Status	ID
HD 068257	1951.09	1997.18	27.61 ± 1.17	-151.73 ± 0.99	7.1	Comp?	$\mathrm{Y}_{\mathrm{es}^{i,j}}$	$6'' \text{ at } 90^{\circ}$
•		•			:	Comp?	No^a	$372^{\prime\prime} { m at} 107^{\circ}$
HD 069830	1955.88^{c}	1986.19	278.99 ± 0.25	-987.59 ± 0.29	31.2	NComp	:	•
HD 071148	1953.12	1989.16	-20.53 ± 0.58	-351.78 ± 0.37	12.7	NComp	:	
HD 072673	1977.21	1991.26	-1113.37 ± 0.35	761.57 ± 0.32	19.0	NComp	:	:
HD 072760	1954.97	1992.04	-192.18 ± 0.70	24.11 ± 0.58	7.3	NComp		
HD 072905	1954.90	1999.19	-27.44 ± 0.31	88.13 ± 0.26	4.1	Marg		•
HD 073350	1954.24^{c}	1986.25	-297.10 ± 0.74	43.38 ± 0.50	9.7	NComp		•
HD 073667	1951.99	1997.10	-108.92 ± 0.85	-501.23 ± 0.55	23.1	Comp?	No^{g}	$335^{\prime\prime} { m at} 207^{\circ}$
HD 073752	1978.12	1992.09	-267.52 ± 0.61	417.61 ± 0.46	7.0	NComp		•
HD 074385	1979.00	1994.18	-269.82 ± 0.52	-86.99 ± 0.57	4.3	Comp?	Y_{es}^{d}	$46'' { m at} 184^{\circ}$
HD 074576	1977.06	1991.27	-300.84 ± 0.26	339.75 ± 0.29	6.5	NComp	:	•
HD 075732	1953.94	1998.29	-485.80 ± 0.97	-234.05 ± 0.68	23.9	Comp	Y_{es^d}	$84'' \text{ at } 128^{\circ}$
HD 075767	1955.22	1997.10	153.24 ± 1.02	-234.86 ± 0.62	11.8	Comp?	NO^{g}	$385^{\prime\prime} \mathrm{at} 41^{\circ}$
HD 076151	1982.14	1992.04	-412.02 ± 0.31	30.00 ± 0.18	4.1	NComp	:	:
HD 076932	1982.01	1992.04	244.14 ± 0.23	213.94 ± 0.15	3.3	NComp	:	•
HD 078366	1955.19	1998.30	-190.31 ± 0.48	-114.96 ± 0.21	9.6	NComp		
HD 079028	1954.90	1996.21	-6.98 ± 0.18	-32.15 ± 0.23	1.4	NoPM	•	•
HD 079096	1950.22	1987.00	-524.40 ± 0.60	245.64 ± 0.29	21.3	NComp	:	
HD 079969	1955.23	1998.30	49.09 ± 0.86	-507.37 ± 0.46	22.0	NComp		•
HD 080715	1953.29	1992.09	-341.40 ± 0.87	-359.04 ± 0.39	19.2	NComp		
HD 082342	1979.00	1992.19	-66.82 ± 0.68	314.76 ± 0.82	4.3	Comp?	Y_{esd}	$11'' { m at} 204^{\circ}$
HD 082443	1952.08	1998.31	-147.90 ± 0.60	-246.77 ± 0.33	13.3	Comp	Yes^d	$65^{\prime\prime}\mathrm{at}68^{\circ}$
HD 082558	1983.20	1987.31	-247.92 ± 0.81	33.43 ± 0.61	1.0	NoPM	•	•
HD 082885	1953.94	1998.29	-728.71 ± 0.34	-259.81 ± 0.18	34.4	Comp?	No^a	$328^{\prime\prime} { m at} { m 333^{\circ}}$
HD 084117	1980.06	1995.09	-399.57 ± 0.17	262.92 ± 0.15	7.2	Comp?	No^a	$722^{\prime\prime} { m at} { m 331^{\circ}}$
HD 084737	1953.12	1995.24	221.66 ± 0.23	-92.47 ± 0.15	10.1	NComp	•	••••
Continued or	1 Next Page							

TABLE 4.1	- Continuec							
	Image	Epochs	μ_{α} -1/	$\mu_{\delta} = 1$	Total	Blink	Com	banion T
Name	Epocn 1	Epocn 2	(mas yr ⁻)	(mas yr ⁻)	ή	Result	Status	IIJ
HD 086728	1955.29	1989.93	-527.63 ± 0.30	-429.42 ± 0.18	23.6	Comp	$\mathbf{Y}^{\mathrm{es}d}$	$133'' { m at} 278^\circ$
HD 087424	1983.11	1986.26	-189.77 ± 1.05	-25.57 ± 0.86	0.6	NoPM		:
HD 087883	1955.20	1988.95	-64.04 ± 0.53	-60.51 ± 0.34	3.0	Marg		:
HD 088742	1978.03	1992.24	-370.25 ± 0.28	65.35 ± 0.35	5.3	NComp	:	•
HD 089125	1955.22	1990.01	-414.15 ± 0.37	-97.66 ± 0.19	14.8	NComp		:
HD 089269	1953.21	1989.92	62.87 ± 0.56	-299.40 ± 0.36	11.2	NComp		:
HD 090156	1975.36	1990.10	-38.02 ± 0.46	99.61 ± 0.47	1.6	NoPM	:	:
HD 090343	1953.29	1997.19	17.16 ± 0.63	46.36 ± 0.54	2.3	NoPM	:	:
HD 090508	1955.21	1998.22	80.51 ± 0.32	-880.38 ± 0.27	38.1	NComp		:
HD 090839	1955.08	1998.30	-176.71 ± 0.22	-33.21 ± 0.18	7.8	Comp	γ_{es}	HD 237903
HD 091324	1987.19	1996.16	-419.27 ± 0.16	209.18 ± 0.14	4.2	Marg	:	
HD 091889	1955.96^{c}	1986.03	268.46 ± 0.30	-672.57 ± 0.29	21.8	NComp	:	:
HD 092719	1955.96^{c}	1986.25	235.35 ± 0.47	-172.56 ± 0.42	8.8	NComp	:	: : :
HD 092945	1979.38	1993.16	-215.23 ± 0.43	-50.04 ± 0.59	3.0	Marg	:	
HD 094765	1953.29	1989.94	-258.38 ± 1.12	-77.42 ± 0.81	9.9	NComp	÷	
HD 095128	1955.21	1998.37	-317.01 ± 0.22	54.64 ± 0.20	13.8	NComp	:	
HD 096064	1984.18	1985.06	-176.60 ± 0.91	-103.81 ± 0.71	0.2	Comp?	${ m Y}_{{ m eS}^{d,j}}$	$13'' \text{ at } 218^{\circ}$
		•	:		:	Comp ?	NO^{g}	$296'' { m at} 142^{\circ}$
HD 096612	1955.21	1989.04	-213.59 ± 0.75	40.69 ± 0.63	7.3	NComp	:	
HD 097334	1953.19	1992.09	-248.91 ± 0.37	-150.98 ± 0.34	11.3	NComp	:	:
HD 097343	1978.20	1990.32	272.41 ± 0.37	-65.72 ± 0.33	3.4	Marg	:	:
HD 097658	1955.22	1992.32	-106.48 ± 0.63	48.82 ± 0.51	4.3	Comp?	NO^{g}	$346'' { m at} 139^{\circ}$
HD 098230	1950.34	1998.38	-453.7 ± 4.4^k	-591.4 ± 4.1^k	35.0	NComp	:	•
HD 098281	1954.24^{c}	1986.20	796.04 ± 0.78	-151.07 ± 0.47	25.9	NComp	:	
HD 099491	1955.29	1996.27	-725.74 ± 0.85	180.67 ± 0.87	30.7	Comp	Yes^{f}	HD 99492
HD 100180	1955.31	1993.08	-329.26 ± 1.28	-190.01 ± 0.98	14.3	Comp	Yes^d	$17'' \text{ at } 332^{\circ}$
HD 100623	1975.19	1991.12	-671.54 ± 0.39	823.85 ± 0.24	16.9	NComp	•	••••
Continued o	n Next Pag	e						

	Image	F toolog			T _o t₀	Dlimly	5	
Name	Epoch 1	Epoch 2	$\max_{\mu} \mathbf{y} \mathbf{r}^{-1})$	$(\max {f yr}^{-1})$	μ	Result	Status	ID
HD 101177	1953.19	1997.28	-594.65 ± 0.62	15.78 ± 0.38	26.2	Comp	${ m Y}_{{ m eS}^{i,j}}$	$8''$ at 270°
HD 101206	1953.28	1997.28	-129.45 ± 0.81	435.83 ± 0.70	20.0	NComp		
HD 101501	1950.37	1999.34	-12.55 ± 0.25	-380.75 ± 0.21	18.6	NComp		
HD 102365	1977.07	1996.22	-1530.99 ± 0.17	403.67 ± 0.18	30.3	NComp		
HD 102438	1979.18	1995.11	-265.17 ± 0.46	-227.79 ± 0.32	5.6	NComp		
HD 102870	1984.39	1998.30	740.23 ± 0.23	-270.43 ± 0.18	11.0	NComp		
HD 103095	1950.37	1997.42	4003.98 ± 0.37	-5813.62 ± 0.23	332.1	NComp		
HD 104067	1983.13	1996.16	143.31 ± 0.80	-423.67 ± 0.48	5.8	NComp		
HD 104304	1954.25^{c}	1989.18	141.75 ± 0.29	-483.64 ± 0.19	17.6	NComp	:	
HD 105631	1955.23	1989.18	-313.55 ± 0.43	-51.62 ± 0.50	10.8	NComp		
HD 108954	1953.29	1996.35	19.78 ± 0.29	182.47 ± 0.29	7.9	NComp	:	
HD 109200	1987.19	1998.38	-545.97 ± 0.44	-312.32 ± 0.42	7.0	NComp		
HD 109358	1952.64	1990.01	-704.75 ± 0.13	292.74 ± 0.14	28.5	NComp		
HD 110463	1953.29	1996.35	121.51 ± 0.67	-4.90 ± 0.68	5.2	NComp		
HD 110810	1976.48	1998.38	-200.58 ± 0.61	-130.26 ± 0.51	5.3	Marg	:	
HD 110833	1953.29	1994.34	-379.08 ± 0.52	-183.68 ± 0.46	17.3	NComp		
HD 110897	1952.64	1990.01	-359.87 ± 0.29	140.16 ± 0.24	14.4	NComp		
HD 111312	1982.40	1992.40	82.27 ± 3.01	44.97 ± 2.37	0.9	NoPM	:	
HD 111395	1955.39	1992.33	-334.58 ± 0.43	-105.52 ± 0.34	13.0	NComp		
HD 112758	1982.30	1991.12	-825.24 ± 0.80	196.16 ± 0.65	7.5	NComp		
HD 112914	1950.43	1990.07	-234.94 ± 0.63	181.12 ± 0.78	11.8	NComp	:	
HD 113283	1978.33	1993.25	-221.11 ± 0.41	-156.05 ± 0.41	4.0	Marg	:	
HD 113449	1956.35^{c}	1985.16	-191.13 ± 0.86	-218.73 ± 0.68	8.4	NComp	:	
HD 114613	1979.16	1994.26	-381.72 ± 0.31	45.75 ± 0.20	5.8	NComp	:	
HD 114710	1955.29	1993.29	-801.44 ± 0.14	882.04 ± 0.10	45.3	NComp	:	
HD 114783	1956.27	1996.23	-138.66 ± 0.56	10.57 ± 0.57	5.5	Comp	NO^{g}	$240^{\prime\prime} { m at} { m 46^{\circ}}$
HD 114853	1985.36	1992.12	-110.14 ± 0.37	-110.75 ± 0.35	1.1	NoPM	•	
Continued or	n Next Page	····						

	Image	Epochs	μ_{α}	μ,8	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(\max \mathbf{yr}^{-1})$	$(\max yr^{-1})$	μ	Result	Status	ID
HD 115383	1955.37	1997.28	-333.83 ± 0.25	190.24 ± 0.17	16.1	NComp	•	
HD 115404	1950.29	1997.35	632.60 ± 0.67	-262.21 ± 0.48	32.1	NComp		
HD 115617	1976.49	1993.22	-1070.36 ± 0.22	-1063.69 ± 0.13	25.2	NComp		:
HD 116442	1956.19	1997.18	13.54 ± 1.13	196.94 ± 0.74	8.2	Comp	Yes^{f}	HD 116443
HD 116956	1953.20	1993.22	-216.61 ± 0.50	11.17 ± 0.46	8.7	NComp		:
HD 117043	1953.28	1997.28	-392.35 ± 0.31	220.73 ± 0.27	19.8	NComp		:
HD 117176	1955.37	1997.35	-236.02 ± 0.24	-575.73 ± 0.19	26.1	NComp		:
HD 118972	1975.27	1991.21	205.63 ± 0.50	-166.77 ± 0.26	4.2	Marg		:
HD 119332	1953.20	1993.31	-16.62 ± 0.55	70.21 ± 0.48	2.9	NoPM		:
HD 120136	1954.25	1992.19	-479.53 ± 0.16	53.49 ± 0.13	18.3	NComp		:
HD 120559	1987.26	1994.19	-360.04 ± 0.85	-412.39 ± 0.67	3.8	NoPM		:
HD 120690	1976.48	1991.28	-580.89 ± 0.57	-246.00 ± 0.34	9.3	NComp		
HD 120780	1976.27	1990.45	-583.66 ± 0.58	-61.29 ± 0.39	8.3	NComp	:	:
HD 121370	1954.25	1991.34	-60.95 ± 1.14	-356.29 ± 0.73	13.5	NComp		:
HD 121560	1954.34	1996.30	-291.30 ± 0.32	8.92 ± 0.28	12.2	NComp	:	
HD 122742	1954.34	1989.27	84.41 ± 0.52	-304.40 ± 0.36	11.0	NComp		:
HD 124106	1988.15	1992.41	-255.06 ± 1.24	-179.11 ± 0.79	1.3	NoPM	:	
HD 124292	1955.37^{c}	1994.19	-160.88 ± 0.59	-321.90 ± 0.38	14.0	NComp	:	:
HD 124580	1976.33	1993.30	127.85 ± 0.42	-138.05 ± 0.37	3.2	Marg	:	:
HD 124850	1983.50	1994.19	-26.31 ± 0.19	-419.38 ± 0.14	4.5	NComp		:
HD 125276	1976.41	1992.25	-356.34 ± 0.90	367.55 ± 0.74	8.1	NComp	:	:
HD 125455	1983.50	1994.19	-632.65 ± 0.90	-121.44 ± 0.64	6.9	Comp?	Yes^l	$14'' \text{ at } 104^{\circ}$
HD 126053	1979.46	1994.36	223.79 ± 0.39	-477.36 ± 0.36	7.9	NComp	:	:
HD 127334	1950.45	1995.47	161.12 ± 0.29	-219.91 ± 0.29	12.3	NComp	:	:
HD 128165	1953.28	1996.44	-191.79 ± 0.38	248.58 ± 0.37	13.5	NComp	:	:
HD 128311	1950.27	1989.25	204.74 ± 0.74	-249.98 ± 0.61	12.6	NComp	:	:
HD 128400	1976.18	1994.45	119.48 ± 0.36	-18.62 ± 0.38	2.2	Marg		•
Continued on	n Next Page							

	Image	Epochs	Πο	Ця	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(\max \mathbf{yr}^{-1})$	$(\max yr^{-1})$	μ	Result	Status	ID
HD 128620	1976.19	1997.19	-3679.25 ± 3.89	473.67 ± 3.24	77.9	Comp	${\rm Y}_{{\rm es}^{i,j}}$	$15'' { m at} 213^{\circ}$
:	:	:	: :			Comp	${ m Yes}^f$	HIP 70890
HD 128642	1955.22	1994.44	-74.49 ± 0.39	-132.26 ± 0.45	6.0	NComp		:
HD 128987	1983.36	1993.25	-111.98 ± 0.50	-64.97 ± 0.35	1.3	NoPM		
HD 130004	1954.42	1993.36	-233.40 ± 0.96	-224.97 ± 0.87	12.6	NComp		:
HD 130042	1987.39	1997.19	-107.42 ± 0.51	-320.00 ± 0.69	3.3	Marg		
HD 130307	1955.37	1989.24	-288.02 ± 0.89	-78.88 ± 0.67	10.2	NComp		:
HD 130948	1950.35	1994.18	143.91 ± 0.37	32.69 ± 0.34	6.5	NComp		
HD 131156	1950.21	1992.19	154.98 ± 0.40	-66.43 ± 0.45	7.1	NComp		:
HD 131511	1950.21	1992.19	-442.23 ± 0.36	217.61 ± 0.43	20.7	NComp		:
HD 131582	1950.35	1991.37	-825.30 ± 1.20	3.09 ± 1.06	33.8	NComp		
HD 131923	1988.43	1997.17	-16.82 ± 0.90	-335.04 ± 0.91	2.9	Marg		:
HD 132142	1953.28	1991.44	-971.06 ± 0.41	479.89 ± 0.38	41.3	NComp	:	:
HD 132254	1955.23	1996.44	111.37 ± 0.26	-225.83 ± 0.22	10.4	NComp		:
HD 133640	1955.23	1996.44	-445.84 ± 1.44	19.86 ± 1.67	18.0	NComp	:	:
HD 134060	1987.65	1997.17	-184.91 ± 0.44	-10.66 ± 0.42	1.8	NoPM		:
HD 135204	1955.30	1994.36	-1269.73 ± 0.59	-503.41 ± 0.41	53.3	NComp		:
HD 135599	1955.30	1994.36	178.35 ± 0.66	-137.52 ± 0.62	8.8	NComp		:
HD 136202	1979.54	1994.36	372.21 ± 0.32	-513.59 ± 0.22	9.4	NComp		:
HD 136352	1987.31	1992.56	-1622.61 ± 0.37	-275.62 ± 0.36	8.6	NComp		:
HD 136713	1981.34	1991.21	-59.29 ± 1.48	-201.52 ± 1.12	2.1	Marg	:	:
HD 136923	1950.30	1992.49	-230.58 ± 0.63	76.30 ± 0.72	10.3	NComp	:	:
HD 137107	1954.48	1987.30	116.83 ± 0.40	-171.37 ± 0.49	7.1	NComp	:	:
HD 137763	1981.34	1991.21	87.09 ± 1.62	-373.67 ± 1.36	3.7	Comp	γ_{es}	HD 137778
HD 139341	1955.25	1993.29	-481.44 ± 0.52	28.68 ± 0.61	18.4	Comp	Y_{es}^{f}	HD 139323
HD 139777	1955.22	1994.44	-224.25 ± 0.40	108.19 ± 0.40	9.8	Comp	Y_{es}^{f}	HD 139813
HD 140538	1982.52	1993.25	-43.11 ± 0.79	-143.57 ± 0.56	1.6	Marg	•	•••••
Continued on	n Next Page							

TABLE 4.1	- Continued	l						
	Image	Epochs	μ_{α}	μ_{δ}	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(mas yr^{-1})$	$(mas yr^{-1})$	μ	Result	Status	ID
HD 140901	1988.30	1997.29	-414.91 ± 0.39	-215.05 ± 0.33	4.2	Comp	$\mathbf{Y}^{\mathrm{es}l}$	$14'' \text{ at } 138^{\circ}$
HD 141004	1954.42	1993.25	-224.00 ± 0.29	-70.64 ± 0.27	9.2	Comp	No^{g}	$235^{\prime\prime}\mathrm{at}200^{\circ}$
HD 141272	1982.52	1993.45	-177.08 ± 0.89	-165.63 ± 0.96	2.7	Marg	:	
HD 142267	1950.44	1994.44	-150.28 ± 0.74	-561.91 ± 0.53	25.6	NComp	:	
HD 142373	1955.23	1989.19	438.90 ± 0.18	629.70 ± 0.20	26.1	NComp	:	
HD 143761	1950.28	1992.64	-196.63 ± 0.24	-773.02 ± 0.21	33.8	NComp	:	
HD 144284	1954.49	1994.42	-319.51 ± 0.11	334.97 ± 0.13	18.5	NComp		:
HD 144287	1950.37	1996.53	-489.51 ± 0.48	697.39 ± 0.64	39.3	NComp	:	
HD 144579	1955.23	1991.27	-571.08 ± 0.29	52.34 ± 0.29	20.7	Comp	Y_{es^d}	$70'' \text{ at } 280^{\circ}$
HD 144628	1988.43	1991.22	-135.23 ± 0.62	333.37 ± 0.58	1.0	NoPM		
HD 144872	1955.23	1991.27	232.34 ± 0.57	-535.74 ± 0.62	21.1	NComp	:	
HD 145417	1988.43	1997.32	-853.82 ± 0.61	-1410.71 ± 0.61	14.7	NComp		
HD 145675	1955.23	1991.43	131.83 ± 0.32	-297.54 ± 0.36	11.8	NComp	:	
HD 145825	1977.52	1992.25	-80.18 ± 0.60	-252.44 ± 0.65	3.9	Marg	:	
HD 145958	1950.53	1989.27	182.93 ± 0.94	-420.02 ± 0.93	17.7	Comp	${ m Yes}^{i,j}$	$3''$ at 135°
HD 146233	1983.45	1992.64	230.77 ± 0.51	-495.53 ± 0.33	5.0	NComp	:	
HD 146361	1954.48	1991.28	-263.39 ± 0.93	-92.67 ± 1.31	10.3	Comp	${ m Yes}^{i,j}$	$6'' \text{ at } 270^{\circ}$
•						Comp	γ_{esf}	HIP 79551
HD 147513	1987.39	1993.25	71.89 ± 0.47	5.26 ± 0.32	0.4	NoPM		
HD 147584	1976.25	1993.30	199.97 ± 0.25	110.97 ± 0.43	3.9	NComp	:	
HD 147776	1983.50	1995.62	-220.35 ± 1.06	-205.09 ± 0.70	3.7	NComp	:	
HD 148653	1950.21	1991.43	-353.30 ± 1.01	388.45 ± 0.80	21.4	NComp	:	
HD 148704	1987.39	1993.25	-431.25 ± 1.89	-331.16 ± 1.81	3.2	Marg	:	
HD 149612	1987.39	1997.32	-227.00 ± 0.43	-284.80 ± 0.41	3.6	NComp	:	
HD 149661	1950.45	1988.61	456.04 ± 0.38	-309.63 ± 0.34	21.0	NComp	:	
HD 149806	1980.59	1994.36	93.69 ± 0.67	76.83 ± 0.55	1.6	NoPM	÷	
HD 151541	1955.31	1992.28	-283.07 ± 0.48	426.91 ± 0.57	18.9	NComp	÷	
Continued o	n Next Page	e						

TABLE 4.1 -	- Continuec	I						
Name	Image Epoch 1	Epochs Epoch 2	$\mu_lpha \ ({ m mas}\ { m yr}^{-1})$	$\mu_{\delta} \ ({ m max}\ { m yr}^{-1})$	Total μ	Blink Result	Comp Status	anion ID
HD 152391	1950.46	1988.45	-711.70 ± 0.87	-1483.65 ± 0.48	62.5	NComp	:	
HD 153557	1955.23	1991.30	-147.62 ± 0.58	270.80 ± 0.68	11.2	Comp	Yes^f	HD 153525
HD 154088	1988.39	1997.33	83.76 ± 0.64	-268.69 ± 0.36	2.5	NoPM		:
HD 154345	1955.23	1991.30	123.27 ± 0.35	853.63 ± 0.36	31.1	NComp		:
HD 154417	1982.53	1992.41	-17.51 ± 0.50	-335.11 ± 0.28	3.3	NComp	:	:
HD 154577	1976.64	1993.25	70.96 ± 0.55	589.86 ± 0.49	9.9	NComp	•	
HD 155712	1954.51	1996.45	101.86 ± 0.60	-116.10 ± 0.65	6.5	NComp	:	:
HD 155885	1987.65	1997.33	-465.31 ± 0.55	-1140.97 ± 0.34	12.0	Comp	$\mathrm{Yes}^{i,j}$	$3''$ at 180°
		:	:	:	:	Comp	Yes^f	HD 156026
HD 156274	1987.70	1992.43	1037.56 ± 0.73	108.99 ± 0.32	4.9	Comp?	$\mathrm{Yes}^{i,j}$	$15'' \text{ at } 231^{\circ}$
HD 157214	1950.46	1993.46	135.55 ± 0.17	-1040.49 ± 0.31	45.1	NComp	:	:
HD 157347	1954.58	1992.35	49.39 ± 0.51	-107.16 ± 0.25	4.5	Comp	$Y_{es}^{d,e}$	$49'' \text{ at } 147^{\circ}$
HD 158614	1981.27	1992.35	-127.77 ± 0.87	-168.61 ± 0.48	2.4	Marg		:
HD 158633	1953.44	1991.44	-530.81 ± 0.31	4.85 ± 0.38	20.2	NComp	:	:
HD 159062	1953.53	1991.45	174.67 ± 0.62	75.82 ± 0.55	7.2	NComp	:	:
HD 159222	1954.51	1993.61	-239.80 ± 0.26	63.08 ± 0.32	9.7	NComp	:	:
HD 160269	1954.57	1992.36	277.02 ± 0.36	-524.88 ± 0.46	22.5	NComp	:	:
			:	:	:	Comp	γ_{es^f}	HIP 86087
HD 160346	1953.61	1993.62	-179.30 ± 0.44	-96.60 ± 0.32	8.2	NComp	:	:
HD 160691	1987.70	1992.59	-16.85 ± 0.40	-190.60 ± 0.23	0.9	NoPM	:	:
HD 161198	1951.49	1992.57	-123.20 ± 0.85	-619.51 ± 0.86	26.0	NComp	:	:
HD 161797	1952.55	1993.48	-291.66 ± 0.12	-749.60 ± 0.15	32.9	NComp	:	:
HD 162004	1953.67	1993.63	34.89 ± 0.48	-275.94 ± 0.59	11.0	Comp	Y_{es}^{f}	HD 162003
HD 164922	1951.49	1991.37	389.41 ± 0.36	-602.03 ± 0.52	28.6	NComp	:	:
HD 165185	1987.71	1996.69	105.05 ± 0.60	7.95 ± 0.32	0.9	NoPM	:	:
HD 165341	1953.53	1988.59	124.16 ± 0.81	-962.82 ± 0.64	34.0	NComp	:	:
HD 165401	1950.52	1991.46	-31.94 ± 0.66	-321.65 ± 0.57	13.2	NComp	•	•••••
Continued on	n Next Pag	e						

	,						1	.
t de	Image	Epochs	μ_{α}	$\mu_{\delta} = 1$	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(mas yr^{-1})$	$(mas yr^{-1})$	μ	Kesult	Status	II
HD 165499	1975.66	1995.65	-77.69 ± 0.39	234.43 ± 0.29	4.9	NComp	:	:
HD 165908	1951.52	1989.35	-100.32 ± 0.28	110.08 ± 0.34	5.7	Marg		•
HD 166620	1950.60	1992.48	-316.44 ± 0.28	-468.47 ± 0.31	23.7	NComp		:
HD 167425	1976.34	1993.61	39.93 ± 0.33	-276.60 ± 0.30	4.8	NComp		•
HD 168009	1950.38	1991.60	-76.55 ± 0.33	-114.46 ± 0.29	5.7	NComp		:
HD 170657	1987.47	1992.57	-138.73 ± 0.84	-195.99 ± 0.59	1.2	NoPM		•
HD 172051	1987.47	1996.69	-74.85 ± 0.77	-152.00 ± 0.47	1.6	NoPM		:
HD 175073	1974.56	1989.51	143.53 ± 1.32	-357.50 ± 0.78	5.7	NComp		:
HD 175742	1951.53	1992.42	131.31 ± 0.49	-283.72 ± 0.63	12.8	NComp	:	•
HD 176051	1950.46	1992.64	201.96 ± 0.28	-145.46 ± 0.35	10.5	NComp	:	:
HD 176377	1950.46	1989.34	56.74 ± 0.33	194.45 ± 0.50	7.9	NComp	:	•
HD 177474	1977.53	1992.57	96.74 ± 1.05	-281.71 ± 0.58	4.5	NComp		:
HD 177565	1974.56	1989.51	-187.45 ± 0.85	-366.74 ± 0.47	6.1	NComp		:
HD 178428	1953.62	1990.63	65.56 ± 0.37	-304.79 ± 0.41	11.5	NComp		:
HD 179957	1953.69	1992.65	-210.20 ± 0.62	621.79 ± 0.55	25.6	Comp	${ m Yes}^{i,j}$	$7''$ at 180°
HD 180161	1953.75	1991.66	217.40 ± 0.30	407.79 ± 0.39	17.5	NComp	:	•
HD 181321	1977.53	1992.57	68.54 ± 1.94	-98.78 ± 1.24	2.0	NoPM	:	•
HD 182488	1955.38	1992.44	83.40 ± 0.34	162.32 ± 0.30	6.8	NComp	•	:
HD 182572	1952.40	1992.59	721.02 ± 0.21	642.49 ± 0.17	38.9	NComp	:	•
HD 183870	1987.41	1988.45	234.50 ± 0.65	18.14 ± 0.37	0.2	NoPM	:	•
HD 184385	1950.54	1986.67	-20.39 ± 0.67	-204.32 ± 0.67	7.4	NComp	:	•
HD 184467	1952.62	1991.67	-509.98 ± 0.65	-397.67 ± 0.69	25.3	NComp	:	•
HD 185144	1954.57	1991.60	598.07 ± 0.17	-1738.40 ± 0.19	68.1	NComp	:	•
HD 185414	1953.61	1989.64	0.32 ± 0.34	-200.51 ± 0.33	7.2	NComp	:	•
HD 186408	1951.52	1991.52	-147.82 ± 0.30	-159.01 ± 0.28	8.7	Comp	Y_{es}^{f}	HD 186427
HD 186858	1952.54	1992.67	7.34 ± 0.64	-436.03 ± 0.90	17.7	Comp	Yes^f	HD 187013
	•••••		••••		•	Comp	Yes^d	HD 225732
Continued of	n Next Page							

	Image	Epochs	μ_{α}	μ_{δ}	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(\max yr^{-1})$	$(\max yr^{-1})$	μ	Result	Status	ID
HD 187691	1953.62	1991.53	242.28 ± 0.27	-136.48 ± 0.23	10.4	Comp	$\mathbf{Y}^{\mathrm{es}d}$	$23^{\prime\prime} { m ~at} { m ~} 221^{\circ}$
HD 189340	1951.72^{c}	1984.65	-251.27 ± 1.14	-398.65 ± 1.01	15.2	NComp		:
HD 189567	1979.71	1995.65	844.50 ± 0.25	-673.62 ± 0.29	17.2	NComp	:	:
HD 189733	1951.53	1996.53	-2.14 ± 0.53	-251.40 ± 0.40	11.3	NComp	:	:
HD 190067	1951.72	1992.74	-159.88 ± 0.54	-582.87 ± 0.62	24.8	NComp	:	:
HD 190248	1979.71	1995.65	1211.03 ± 0.13	-1130.05 ± 0.13	26.4	NComp		:
HD 190360	1953.53	1992.49	683.94 ± 0.22	-524.70 ± 0.27	33.6	Comp	Yes^{f}	LHS 3509
HD 190404	1951.53	1992.72	-1002.97 ± 0.42	-913.19 ± 0.35	55.8	NComp	:	:
HD 190406	1951.72	1992.74	-394.57 ± 0.27	-407.82 ± 0.33	23.2	NComp	:	:
HD 190422	1976.50	1991.66	18.30 ± 0.45	35.88 ± 0.44	0.6	NoPM	:	:
HD 190470	1951.53	1992.72	-75.76 ± 0.65	-39.75 ± 0.53	3.5	Marg	:	:
HD 190771	1950.61	1991.60	262.77 ± 0.29	111.84 ± 0.33	11.7	NComp	:	:
HD 191408	1976.57	1990.72	456.99 ± 0.33	-1574.64 ± 0.22	23.2	NComp		:
HD 191499	1951.72	1992.74	4.65 ± 0.78	174.51 ± 0.85	7.2	NComp		:
HD 191785	1951.72	1992.74	-414.75 ± 0.50	398.26 ± 0.56	23.6	Comp	Y_{esd}	$102^{\prime\prime} \mathrm{at} 95^{\circ}$
HD 192263	1951.58	1988.67	-61.13 ± 1.21	261.37 ± 0.50	10.0	NComp	:	:
HD 192310	1977.55	1992.43	1241.85 ± 0.34	-180.96 ± 0.21	18.7	NComp	:	:
HD 193664	1953.61	1991.67	468.58 ± 0.26	296.61 ± 0.22	21.1	NComp	:	:
HD 194640	1977.60	1991.74	-14.62 ± 0.51	-521.82 ± 0.36	7.4	NComp	:	:
HD 195564	1953.77^{c}	1986.65	307.35 ± 0.42	106.94 ± 0.33	10.7	NComp	:	:
HD 195987	1953.45	1992.49	-156.76 ± 0.34	452.84 ± 0.34	18.7	NComp	:	:
HD 196378	1976.50	1991.68	313.48 ± 0.21	-569.91 ± 0.24	9.9	NComp	:	:
HD 196761	1974.54	1992.43	501.45 ± 0.56	461.36 ± 0.39	12.2	NComp		:
HD 197076	1951.51	1992.67	118.14 ± 0.30	312.63 ± 0.26	13.8	Comp	Y_{es^d}	$125^{\prime\prime} { m at} 184^{\circ}$
HD 197214	1974.47	1990.73	-42.69 ± 1.09	-208.55 ± 0.82	3.4	NComp	:	:
HD 198425	1953.45	1992.56	-161.13 ± 0.49	-266.74 ± 0.56	12.2	Comp	Yes^d	$33''$ at 247°
HD 199260	1976.41	1994.50	93.79 ± 0.43	-64.78 ± 0.37	2.0	NoPM	•	•
Continued or	ι Next Page	····						

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	TADLE 4.1								
Name Epoch 1 Epoch 2 (mas yr ⁻¹) (mas yr ⁻¹) μ Result Status ID HD 19928 1976.71 1990.78 -515.34 ± 0.42 -975.52 ± 0.22 15.5 N Comp \dots \dots HD 199288 1976.71 1990.78 -515.34 ± 0.42 -975.52 ± 0.22 15.5 N Comp \dots \dots HD 200503 1953.652 1984.50 328.85 \pm 0.72 -46.70 ± 1.29 11.10 N Comp \dots \dots \dots HD 200503 1953.652 1991.56 -582.85 ± 0.72 -46.70 ± 0.29 11.11 N Comp \dots \dots \dots \dots HD 202505 1991.65 1941.93 \pm 0.54 1581.44 \pm 0.91 10.35 $N Comp$ \dots <t< th=""><th></th><th>Image</th><th>Epochs</th><th>μ_{lpha}</th><th>μ_{δ}</th><th>Total</th><th>Blink</th><th>Comp</th><th>anion</th></t<>		Image	Epochs	μ_{lpha}	μ_{δ}	Total	Blink	Comp	anion
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Name	Epoch 1	Epoch 2	$({ m mas}~{ m yr}^{-1})$	$({ m mas}~{ m yr}^{-1})$	μ	Result	Status	ID
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	HD 199288	1976.71	1990.78	-515.34 ± 0.42	-975.52 ± 0.22	15.5	NComp		
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 199509	1976.66	1992.64	264.65 ± 0.42	-49.01 ± 0.33	4.3	Marg	•	:
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 200525	1975.68	1992.43	460.33 ± 1.46	-285.96 ± 1.39	9.1	NComp		:
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 200968	1953.62^{c}	1984.59	382.85 ± 0.72	-46.70 ± 0.29	11.9	NComp		:
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 200560	1952.55	1990.70	403.83 ± 0.58	140.93 ± 0.53	16.3	NComp	:	:
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 202275	1951.58	1987.74	42.39 ± 0.68	-304.19 ± 0.42	11.1	NComp	:	:
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 202628	1980.61	1993.61	240.89 ± 0.34	21.00 ± 0.29	3.2	Marg	:	:
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	HD 202751	1952.65	1991.54	468.58 ± 1.02	-187.74 ± 0.91	19.6	NComp		:
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	HD 202940	1979.77	1994.65	-582.58 ± 0.78	-358.93 ± 0.52	10.2	NComp		:
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	HD 203244	1985.44	1990.79	141.93 ± 0.54	168.40 ± 0.55	1.2	Marg	:	:
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	HD 203850	1975.45	1991.68	661.43 ± 1.05	139.55 ± 0.71	11.0	NComp	:	:
HD 205390 1976.48 1992.64 424.24 ± 0.57 -199.70 ± 0.35 7.6 NCompHD 205536 1977.78 1996.78 -139.36 ± 0.51 212.55 ± 0.45 4.8 NCompHD 206826 1954.72 1987.65 260.72 ± 0.36 -243.21 ± 0.31 11.7 NCompHD 206860 1953.63 1990.80 229.93 ± 0.55 -113.46 ± 0.32 9.6 $Comp?$ No^9 $591'' at 10.61$ HD 207129 1974.64 1990.72 1064.43 ± 0.30 -295.37 ± 0.22 6.0 $NComp$ HD 207144 1974.64 1990.72 107.69 ± 0.93 -352.31 ± 0.53 5.9 $NComp$ HD 207144 1974.64 1990.72 -107.69 ± 0.93 -352.31 ± 0.53 5.9 $NComp$ HD 207144 1974.64 1990.72 -101.58 ± 0.76 4.4 $NComp$ HD 208133 1954.72 1987.65 211.06 ± 0.67 -233.89 ± 0.76 10.4 $NComp$ HD 208033 1954.72 1987.79 85.07 ± 0.48 -251.98 ± 0.47 9.8 $NComp$ HD 210167 1987.79 85.07 ± 0.48 -251.98 ± 0.47 9.8 $NComp$ HD 210918 1980.55 1992	HD 203985	1980.61	1993.61	258.62 ± 0.97	187.57 ± 0.74	4.2	Comp	Y_{esd}	$88'' \text{ at } 259^{\circ}$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	HD 205390	1976.48	1992.64	424.24 ± 0.57	-199.70 ± 0.35	7.6	NComp	:	:
HD 206826 1954.72 1987.65 260.72 ± 0.36 -243.21 ± 0.31 11.7 NComp \dots \dots HD 206860 1953.63 1990.80 229.93 ± 0.55 -113.46 ± 0.32 9.6 $Comp?$ No^9 $591''at16$ HD 207129 1978.81 1996.69 164.43 ± 0.30 -295.37 ± 0.22 6.0 $NComp$ \dots \dots HD 207144 1974.64 1990.72 107.69 ± 0.93 -295.37 ± 0.22 6.0 $NComp$ \dots \dots HD 207144 1974.64 1990.72 107.69 ± 0.93 -295.37 ± 0.22 6.0 $NComp$ \dots \dots HD 207144 1974.64 1990.72 107.69 ± 0.93 -295.37 ± 0.22 6.0 $NComp$ \dots \dots HD 207144 1974.64 1990.72 107.69 ± 0.93 -295.37 ± 0.72 6.0 $NComp$ \dots \dots HD 20038 1951.73 1994.52 -5.96 ± 0.64 -101.58 ± 0.76 4.4 $NComp$ \dots \dots HD 20038 1951.60^{c} 1987.79 85.07 ± 0.46 -233.89 ± 0.77 10.4 $NComp$ \dots \dots HD 210277 1951.60^{c} 1987.79 85.07 ± 0.48 -251.98 ± 0.47 9.8 $NComp$ \dots \dots HD 210918 1980.55 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 $NComp$ \dots \dots HD 211472 1978.80 1993.69 233.52 ± 0.44 <	HD 205536	1977.78	1996.78	-139.36 ± 0.51	212.55 ± 0.45	4.8	NComp		•
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 206826	1954.72	1987.65	260.72 ± 0.36	-243.21 ± 0.31	11.7	NComp	•	
HD 207129 1978.81 1996.69 164.43 ± 0.30 -295.37 ± 0.22 6.0 NCompHD 207144 1974.64 1990.72 107.69 ± 0.93 -352.31 ± 0.53 5.9 NCompHD 207144 1974.64 1990.72 107.69 ± 0.93 -352.31 ± 0.53 5.9 NCompHD 208313 1954.72 1994.52 -5.96 ± 0.64 -101.58 ± 0.76 4.4 NCompHD 208313 1954.72 1987.65 211.06 ± 0.67 -233.89 ± 0.79 10.4 NCompHD 210277 1951.60^c 1987.79 85.07 ± 0.46 -449.74 ± 0.30 16.6 NCompHD 210267 1953.69 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 210918 1980.55 19992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 210918 1980.55 19992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 211472 1978.80 19992.67 235.52 ± 0.48 -6022.46 ± 0.22 11.7 NCompHD 211472 1978.80 19992.67 213.52 ± 0.29 -632.46 ± 0.22 11.7 NCompHD 211472 1977.78 1996.78 57.79 ± 0.50 12.25 ± 0.39 <td< td=""><td>HD 206860</td><td>1953.63</td><td>1990.80</td><td>229.93 ± 0.55</td><td>-113.46 ± 0.32</td><td>9.6</td><td>Comp?</td><td>NO^{g}</td><td>$591^{\prime\prime} { m at} 16^{\circ}$</td></td<>	HD 206860	1953.63	1990.80	229.93 ± 0.55	-113.46 ± 0.32	9.6	Comp?	NO^{g}	$591^{\prime\prime} { m at} 16^{\circ}$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	HD 207129	1978.81	1996.69	164.43 ± 0.30	-295.37 ± 0.22	6.0	NComp	:	:
HD 208038 1951.73 1994.52 -5.96 ± 0.64 -101.58 ± 0.76 4.4 NCompHD 208313 1954.72 1987.65 211.06 ± 0.67 -233.89 ± 0.79 10.4 NCompHD 210277 1951.60° 1987.79 85.07 ± 0.46 -449.74 ± 0.30 16.6 NCompHD 210277 1951.60° 1987.79 85.07 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 210067 1953.69 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 210918 1980.55 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 211018 1980.55 1993.69 438.75 ± 0.29 -632.46 ± 0.22 114.7 NCompHD 211415 1978.80 1993.69 438.75 ± 0.29 -632.46 ± 0.22 111.5 NCompHD 211472 1952.70 1991.60 213.52 ± 0.44 69.87 ± 0.43 8.7 CompHD 211472 1952.70 1991.60 213.52 ± 0.24 -632.46 ± 0.22 11.5 Yes ^d 76^{d} $20^{''}$ at $10^{'}$ HD 211472 1952.70 1991.60 213.52 ± 0.24 -225 ± 0.39 1.1 CompYes ^d $20^{''}$ at 93 HD 212168 1977.78 1996.78 $57.79 $	HD 207144	1974.64	1990.72	107.69 ± 0.93	-352.31 ± 0.53	5.9	NComp	:	•
HD 208313 1954.72 1987.65 211.06 ± 0.67 -233.89 ± 0.79 10.4 NComp \dots \dots HD 210277 1951.60^c 1987.79 85.07 ± 0.46 -449.74 ± 0.30 16.6 NComp \dots \dots HD 210277 1951.60^c 1987.79 85.07 ± 0.46 -449.74 ± 0.30 16.6 NComp \dots \dots HD 210067 1953.69 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NComp \dots \dots HD 210918 1980.55 1992.63 571.11 ± 0.36 -789.84 ± 0.32 14.7 NComp \dots \dots HD 211415 1978.80 1993.69 438.75 ± 0.29 -632.46 ± 0.22 11.5 NComp \dots \dots HD 211472 1978.80 1993.69 438.75 ± 0.29 -632.46 ± 0.22 11.5 NComp \dots \dots HD 211472 1978.80 1991.60 213.52 ± 0.44 69.87 ± 0.43 8.7 Comp \dots \dots HD 211472 1952.70 1991.60 213.52 ± 0.29 -12.25 ± 0.39 1.1 Comp? Yes^d $20''$ at 93 HD 212168 1977.78 1996.78 57.79 ± 0.50 -12.25 ± 0.29 1.1 Comp? Yes^d $20''$ at 93 HD 212168 1977.78 1996.78 57.79 ± 0.23 -331.66 ± 0.20 5.6 NComp? Yes^d $20''$ at 93 HD 212330 1980.60 1995.51 $180.$	HD 208038	1951.73	1994.52	-5.96 ± 0.64	-101.58 ± 0.76	4.4	NComp	:	•
HD 210277 1951.60^{c} 1987.79 85.07 ± 0.46 -449.74 ± 0.30 16.6 NCompHD 210667 1953.69 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 210667 1953.69 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 210918 1980.55 1992.67 25.52 ± 0.48 -789.84 ± 0.32 14.7 NCompHD 211415 1978.80 1993.69 438.75 ± 0.29 -632.46 ± 0.22 11.5 NCompHD 211472 1978.80 1993.69 438.75 ± 0.29 -632.46 ± 0.22 11.5 NCompHD 211472 1978.80 1991.60 213.52 ± 0.44 69.87 ± 0.43 8.7 CompHD 211472 1977.78 1991.60 213.52 ± 0.29 -12.25 ± 0.39 1.1 Comp? Yes^d $76''$ at $10''$ HD 212168 1977.78 1996.78 57.79 ± 0.50 12.25 ± 0.39 1.1 Comp? Yes^d $20''$ at 93 HD 212330 1980.60 1995.51 180.06 ± 0.23 -331.66 ± 0.20 5.6 NComp	HD 208313	1954.72	1987.65	211.06 ± 0.67	-233.89 ± 0.79	10.4	NComp	:	•
HD 210667 1953.69 1992.67 25.52 ± 0.48 -251.98 ± 0.47 9.8 NCompHD 210918 1980.55 1995.63 571.11 ± 0.36 -789.84 ± 0.32 14.7 NCompHD 211415 1978.80 1993.69 438.75 ± 0.29 -632.46 ± 0.22 11.5 NCompHD 211472 1978.80 1993.69 438.75 ± 0.29 -632.46 ± 0.22 11.5 NCompHD 211472 1978.80 1991.60 213.52 ± 0.44 69.87 ± 0.43 8.7 CompHD 211472 1977.78 1991.60 213.52 ± 0.44 69.87 ± 0.43 8.7 CompHD 212168 1977.78 1996.78 57.79 ± 0.50 12.25 ± 0.39 1.1 Comp? Yes^d $20''$ at 93 HD 212330 1980.60 1995.51 180.06 ± 0.23 -331.66 ± 0.20 5.6 NComp	HD 210277	1951.60^{c}	1987.79	85.07 ± 0.46	-449.74 ± 0.30	16.6	NComp	:	•
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	HD 210667	1953.69	1992.67	25.52 ± 0.48	-251.98 ± 0.47	9.8	NComp		•
HD2114151978.801993.69 438.75 ± 0.29 -632.46 ± 0.22 11.5 NCompHD2114721952.701991.60 213.52 ± 0.44 69.87 ± 0.43 8.7 Comp Yes^d $76''$ at 105HD2114721952.701991.60 213.52 ± 0.44 69.87 ± 0.43 8.7 Comp Yes^d $76''$ at 105HD2121681977.781996.78 57.79 ± 0.50 12.25 ± 0.39 1.1 $Comp?$ Yes^d $20''$ at 93HD2123301980.601995.51 180.06 ± 0.23 -331.66 ± 0.20 5.6 $NComp$ \dots \dots	HD 210918	1980.55	1995.63	571.11 ± 0.36	-789.84 ± 0.32	14.7	NComp		
HD 2114721952.701991.60213.52 \pm 0.4469.87 \pm 0.438.7CompYes ^d 76" at 10?HD 2121681977.781996.7857.79 \pm 0.5012.25 \pm 0.391.1Comp?Yes ^d 20" at 93HD 2123301980.601995.51180.06 \pm 0.23 $-331.66 \pm$ 0.205.6NComp	HD 211415	1978.80	1993.69	438.75 ± 0.29	-632.46 ± 0.22	11.5	NComp		
HD 212168 1977.78 1996.78 57.79 \pm 0.50 12.25 \pm 0.39 1.1 Comp? Yes ^d 20" at 93 HD 212330 1980.60 1995.51 180.06 \pm 0.23 $-331.66 \pm$ 0.20 5.6 NComp	HD 211472	1952.70	1991.60	213.52 ± 0.44	69.87 ± 0.43	8.7	Comp	Y^{esd}	$76''$ at 105°
HD 212330 1980.60 1995.51 180.06 ± 0.23 -331.66 ± 0.20 5.6 NComp	HD 212168	1977.78	1996.78	57.79 ± 0.50	12.25 ± 0.39	1.1	Comp?	Y^{esd}	$20^{\prime\prime}\mathrm{at}93^{\circ}$
	HD 212330	1980.60	1995.51	180.06 ± 0.23	-331.66 ± 0.20	5.6	NComp	:	:

	Image	Epochs	μ_{α}	μδ	Total	Blink	Comp	anion
Name	Epoch 1	Epoch 2	$(\max yr^{-1})$	$(\max yr^{-1})$	μ	Result	Status	ID
HD 214683	1954.58	1991.77	176.80 ± 0.82	107.79 ± 0.71	7.8	NComp	:	:
HD 214953	1976.74	1995.65	6.24 ± 0.37	-331.18 ± 0.29	6.3	NComp	:	
HD 215152	1954.52^{c}	1989.80	-155.75 ± 1.20	-290.81 ± 0.97	11.6	NComp	:	
HD 215648	1953.63	1990.64	234.18 ± 0.21	-493.29 ± 0.17	20.1	NComp	:	
HD 216259	1953.63	1991.70	405.69 ± 1.02	202.72 ± 0.97	17.2	NComp		
HD 216520	1954.73	1996.64	-150.23 ± 0.56	116.22 ± 0.40	8.0	NComp	:	
HD 217014	1954.58	1990.79	207.25 ± 0.31	60.34 ± 0.30	7.8	NComp	:	
HD 217107	1982.80	1991.68	-6.35 ± 0.46	-15.80 ± 0.31	0.2	NoPM		:
HD 217813	1954.58	1991.75	-117.70 ± 0.59	-27.66 ± 0.49	4.4	Marg		
HD 218868	1953.83	1989.68	-86.52 ± 0.29	-287.42 ± 0.36	10.7	Comp	$Y_{\mathrm{eS}}{}^{d,e}$	$50^{\prime\prime} { m at} 90^{\circ}$
HD 219134	1952.71	1990.79	2075.07 ± 0.33	295.45 ± 0.25	79.8	NComp		
HD 219482	1979.51	1994.51	175.24 ± 0.30	-27.24 ± 0.25	2.7	NoPM	•	
HD 219538	1954.66	1991.76	359.78 ± 0.57	89.75 ± 0.42	13.7	NComp	:	
HD 219623	1952.71	1990.79	111.87 ± 0.22	-236.51 ± 0.21	10.0	NComp		: : :
HD 220140	1954.72	1996.64	201.56 ± 0.50	71.59 ± 0.41	9.0	NComp	:	
HD 220182	1953.83	1989.68	636.54 ± 0.38	219.49 ± 0.38	24.1	NComp		
HD 220339	1983.82	1991.76	453.40 ± 1.09	260.15 ± 0.79	4.1	NComp	:	
HD 221354	1954.60	1991.60	1105.17 ± 0.46	113.28 ± 0.51	41.1	NComp		
HD 221851	1953.93	1991.76	-197.70 ± 0.72	-281.07 ± 0.40	13.0	NComp	:	
HD 222143	1951.69	1990.64	357.60 ± 0.33	-12.14 ± 0.30	13.9	NComp	:	
HD 222237	1977.78	1987.71	141.94 ± 0.42	-736.82 ± 0.42	7.4	NComp	:	•
HD 222335	1985.55	1997.66	133.37 ± 0.58	-304.64 ± 0.54	4.0	Marg		
HD 222368	1953.62	1986.77	377.15 ± 0.19	-437.43 ± 0.15	19.1	NComp	:	
HD 223778	1952.64	1992.76	341.01 ± 0.32	41.50 ± 0.29	13.8	NComp	:	
HD 224228	1977.71	1996.72	206.72 ± 0.51	-185.07 ± 0.54	5.3	NComp	÷	•
HD 224465	1953.75	1991.68	-45.02 ± 0.40	244.82 ± 0.39	9.5	NComp	÷	•
HD 224930	1950.61	1991.77	780.22 ± 2.01	-917.75 ± 1.20	49.6	NComp	•	
Continued or	n Next Page							

TABLE 4.1 –	- Continued							
Name	Image . Epoch 1	Epochs Epoch 2	$\mu_lpha \ ({ m mas}\ { m yr}^{-1})$	$\mu_{\delta} \ ({ m mas} \ { m yr}^{-1})$	Total μ	Blink Result	Comp. Status	anion ID
HD 232781 HD 263175 HIP 036357	$\begin{array}{c} 1954.74 \\ 1953.93 \\ 1953 13 \end{array}$	1989.77 1989.84 1988 93	$-311.48 \pm 1.08 \\ -452.64 \pm 1.41 \\ 157.85 \pm 1.28 $	$\begin{array}{c} -266.47 \pm 0.97 \\ 99.28 \pm 0.81 \\ 175.95 \pm 0.73 \end{array}$	14.4 16.7 8.5	NComp Comp Comp	$\sum_{\mathrm{Ves}^{d}}^{\dots}$	 30″ at 101° HD 58046
HIP 040774 HIP 087579	1983.04 1951.49	1987.02 1992.57	-164.53 ± 1.35 -73.12 ± 0.85	-52.64 ± 0.85 56.21 ± 0.88	0.7 3.8	NoPM NComp		
HIP 091605	1951.51	1989.34	86.87 ± 0.85	-837.41 ± 1.16	31.9	Comp	$\mathrm{Yes}^{j,l}$	$9''$ at 150°
NOTES.—(Column 7 v	alues: NoPN	1 = Primary's prope	er motion undetect	able, Mar	.g = Prim	ary's proj	per motion
marginal, NCo	omp = No	CPM candic	lates identified, Com	1p = Candidate wi	th visual	ly compel	ing CPM	identified,
Comp? = Can	ndidate with	ı possible CP	M identified					
^a Primary	/ and candi	idate compar	iion have significantl	ly different proper	motions 1	from the l	iterature	(see § 7.4).
^b The can	ndidate com	panion is a no	on-stellar artifact suc	h as a plate defect.	° Ep	och-1 imag	se is from	SuperCOS-
MOS Sky Surv	rey in order	to increase th	ie time interval betwe	een the two images.	d Ph	lotometric	distance t	o the CPM
candidate mat	ches the Hi_l	<i>pparcos</i> distaı	nce to the primary (se	ee Table 4.2). e	New (ca	ndidate) co	mpanion	discovered.
f Parallax	t and proper	r motion for t	he CPM candidate fr	om literature match	the corre	sponding I	rimary's	values from
Hipparcos.	^g Photom	etric distance	to the CPM candida	te is significantly dif	ferent tha	un primary	's Hipparc	os distance

^k This star ^j Identified as twin, comoving diffraction spikes. ⁱ Known companion with a published orbit.

^h Spectroscopic distance to the CPM candidate matches the Hipparcos distance to the primary.

(see Table 4.2).

is not in *Hipparcos* or FvL07 catalogs. It was selected based on Söderhjelm (1999) parallax. Proper motion is from Høg ¹ Companionship confirmed based on matching proper motion and proximity to the primary. et al. (2000).

While CPM is necessary for physical association, it is by no means sufficient. In addition to exhibiting CPM, a physically bound companion must pass additional criteria. The attributes I used to make this determination are the magnitude of the proper motion and the proximity to the primary, or a comparison of the estimated distance to the companion with that of the primary from *Hipparcos*. For example, a candidate companion at a small angular separation and sharing a high proper motion with the primary is likely to be physically bound, simply based on the laws of probability. Similarly, if an independent distance estimate to the candidate matches the *Hipparcos* distance to the primary, one can be quite certain that the stars are bound by gravity. Accordingly, candidates were confirmed as true companions if they passed one or more of the following tests.

1. The candidate has independent published measures of trigonometric parallaxes and proper motions matching the corresponding *Hipparcos* values of the primary. Each of these companions is identified by its HD, HIP, or other identifier in the last column. Twenty four of the 28 companions confirmed by this method are *Hipparcos* stars themselves, allowing for a check of independent measures of the astrometry by the same instrument and technique. The remaining four non-*Hipparcos* stars, NLTT 9303, LHS 25, HD 237903, and LHS 3509, companions to HD 18143, 26965, 90839, and 190360, respectively, have proper motion measurements from Lépine & Shara (2005) or Salim & Gould (2003) and parallax measures from ground-based efforts (Harrington & Dahn 1980; van Altena et al. 1995; Oppenheimer et al. 2001) that match the corresponding *Hipparcos* values for the primaries within the uncertainties. 2. The candidate has published orbits based on many resolved measures listed in the WDS. All 10 companions confirmed by this method were well-within the saturation region around the primary and identified by twin, comoving diffraction spikes, seen due to the low magnitude-difference between the components. While some of the orbital solutions may be preliminary, they are all good enough to confirm that the measures represent a gravitationally-bound pair.



FIGURE 4.6: A new CPM companion to HD 4391 was discovered by blinking the above images and confirmed with photometric distance estimates (see Table 4.2). The companion is marked by the arrows and the primary is the bright star at the center of the images, which are 10' on each side and oriented with North up and East to the left. The image on the left is the earlier epoch and on the right is the later epoch of DSS images as listed in Table 4.1.

3. The candidate has photometric distance estimates and published proper motions that match the primary's *Hipparcos* values. The distance estimates were derived using the candidates' $VRIJHK_S$ photometry and fitting various colors to the M_{K_S} -color relations from Henry et al. (2004) for red dwarfs and M_V to V - I relation from Salim et al. (2004) for white dwarfs. The VRI photometry were obtained at the Cerro Tololo Inter-American Observatory's (CTIO) 0.9-m telescope, a facility of the Small and Moderate Aperture Re-



FIGURE 4.7: A new CPM companion to HD 43162 was discovered by blinking the above images. See the caption for Figure 4.6 for further details.



FIGURE 4.8: A new CPM companion to HD 157347 was discovered by blinking the above images. See the caption for Figure 4.6 for further details.

search Telescope System (SMARTS) Consortium, or from publications accessed via VizieR on SIMBAD, and the JHK_S magnitudes were extracted from the Two Micron All Sky Survey Catalog (2MASS). This method enabled the confirmation of 28 candidates. An additional candidate (v And B) was confirmed based on its infrared magnitudes and spectral type. These candidates are listed in Table 4.2 along with their spectral types, proper motions, and photometry along with their references, and distance estimates. The #1 or #2 notation after



FIGURE 4.9: A new CPM companion to HD 218868 was discovered by blinking the above images. See the caption for Figure 4.6 for further details.

the HD or HIP name in Column 1 denotes the sequence of the companion's listing for the corresponding primary in Table 4.1. While many of these companions have been recognized as companions in publications, several of them are confirmed indisputably for the first time here by matching not only proper motions, but also distances. These results also contain four new companion discoveries, for HD 4391, 43162, and 157347, and HD 218868, which are shown in Figures 4.6–4.9.

;	Spectra.	-	$\mu_{\alpha}^{\mu_{\alpha}}$	$\mu_{\delta} = 1$	ſ	 ;	C C	CD M	agnitu	ides ==	f	Nbr 0	Infrare	d Magn	itudes	۵	Error
Name	Type	4	(mas yr ⁻)	(mas yr ⁻)	ਖ	Λ	z	н	z	I	r	Obs	ſ	Н	К	(bc)	(bc)
			Con	firmed Compan	ions												
HD 004391#1	:		:			14.36	-	13.14	-	11.46	1	ŝ	9.88	9.34	9.03	19.3	3.1
HD 016160#1	:	:	:	:	:	11.68	1	10.47	1	8.87	1	1	7.33	6.79	6.57	6.9	1.1
HD 017382#1	:	:	275	-123	7	16.5	n	13.89	4	:	:	:	10.73	10.17	9.87	22.1	9.0
HD 018757#1	:	:	717	-697	7	12.65	ы	:	:	10.07	9	:	8.88	8.33	8.10	22.6	5.2
HD 032778#1	:	:	:	:	:	10.49	1	9.64	-	8.85	1	1	7.86	7.32	7.06	23.9	3.9
HD $043162 \# 1$:	:	:	:	÷	12.96		11.78		10.21			8.72	8.16	7.87	13.2	2.0
HD 043587#1	:	:			:'	13.29	- 0	12.11	- 0	10.58	- 0	-	9.09 2.00	8.56	8.27	16.5	5.0 1
HD $061606#1$. (: (29	-286	r 0	8.93	× •	8.09	× •	7.34	×,	:,	6.38	5.70	5.57	12.5	2.1
HD 063077#1 HD 065007#1	nc	C1	-246	1054 103	00	10.6U 0.33	- 0	19.97	-	15.39 7.47	I S	-	14.78 6 01	14.55 6.28	14.40 6.06	10.3 17 5	1.1 8 F
HD 074385#1	:	:	-294	-84 -84	n 0	12.68	-1 1 م	:	:	17.1	5	:	1 <i>6</i> .0	8 42	000 8 17	22.0	0. 1 .0
HD 075732#1	: :	: :	F		2 :	13.26		11.91	:	10.24	:	. 0	8.56	7.93	7.67	8.7	1.4
HD 082342#1		-	-75	386	10	13.14	1	12.03	-	10.58	1		9.27	8.77	8.50	23.7	4.2
HD 082443#1	:	:	-134	-242	7	16.8	3	14.39	4	12.07	9	:	10.36	9.86	9.47	12.6	5.1
HD 086728#1	M6.5V	12	-408	-327	÷	:	:	:	:	÷	:	:	10.26	9.64	9.27	:	:
HD 096064#1	:	:			:	10.0	<i>ი</i>	. :	:	:	÷	:	7.27	6.62	6.42	17.0^{a}	2.6
HD 100180#1	:	:	-328	-189	0	9.24	13	7.75	9	:	:	:	7.04	6.52	6.37	23.5_{-1}	4.3
HD 144579#1		:	-547	55	0	14.23	17	12.75	17	11.47	17	:	0.90	9.45	9.16	25.4^{o}	7.3
HD 157347#1	M3.0V	-	:	:	:	12.18	., .	11.06	·	9.64	_, ,	ς, το	8.26	7.68	7.44	13.4	2.1
HD 187691 $\#1$: •	:	:	:	12.67		11.55	н,	10.04			8.89	8.30	8.01	20.0	8.6
HD 191785#1	MI3.5V	-	•		÷	13.93		12.73		11.14			9.63 e 1 <i>e</i>	9.11 7.65	x x x	20.9	4.0 1.4
HD 197076#1 HD 108496#1	:	:	150		: °	19.6		15.47	- u	9.47 12.60	ч	-	8.10 11 79	00.7 91.11	10.86	10.8 95.7	0 0 1 17
HD 203985 ± 1	M3.5V	-			1	13.49	, c	12.29	o	10.71			9.22	8.62	8.35	15.9	2.4
HD 211472#1	:	:	205	99	0	13.93	14	:	:	:	:	:	9.72	9.19	8.93	22.3	3.5
HD 212168#1	:	:	:	:	:	8.80	1	8.09	1	7.43	1	1	6.56	5.94	5.81	16.9	2.8
HD 218868#1	:	:	-50	-322	9	15.32	14	:	:	12.57	9	:	10.84	10.23	9.90	25.9	6.0
HD 263175#1	M0.5V	œ	:	:	:	12.15	œ	11.18	œ	10.11	œ	:	8.99	8.43	8.18	30.6	4.8
			Refuted	Candidate Com	Ipanior	IS											
UD 000540441						10.76	-	10.01	-	01.11	-	-	01 01	11	90 O	60 1	1
HD 012846#1				-102	: 9	11.48	+ %	10.62	- 8	9.84	- 6	-	10.18	08.0	9.66	81.7	13.2
HD 026923#1	:		-15	-131	:	9.99	- 1	9.50	1	9.06		1	8.43	8.04	7.88	55.6	8.9
HD 073667#1	:	:	:	:	:	15.35	1	14.27	1	12.96	1	1	11.70	11.18	10.92	82.4	13.4
HD 075767#1	:	:	58	-178	7	13.77	1	12.74	1	11.51	1	1	10.22	9.57	9.30	38.1^{b}	6.8
HD $096064#2$:	:	:	:	÷	16.20	.,	15.36	., .,	14.60	,		13.63	13.06	12.85	355.8	58.7
HD 097658#1	:	:	:	:	÷	15.94 0 70		14.95		13.61		- 0	12.32	11.74	099.11 7 70	109.4 7.1.0	18.3 1
HD 14/05#1 HD 141004#1	:	:	:	:	:	9.70 18.35		9.31 16.92		0.90 15 11		10	0.02 13 40	12.85	19.50	04.0 76 9	13.0
HD 206860#1	: :	: :	111	-89	19	15.04	18	14.07	9	13.63	9	1	12.77	12.15	12.06	300.0	83.9
OTES — Reference	ordes for co	sumule	≈ 3. 6, 8, 10, and 1.	2: 1 = CTIO of	rervat	ions obta	ined fi	or this v	vork. 5	$\lambda = The I$	N Mds.	orth cats	ulog (Lén	ine & Shi	ara 2005).	3 = Visua	Double Sta
OTED: TOTOTO			+ + + + + + + + + + + + + + + + + + +	;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;	····		• • • • • • •			,) = + a			AVE VEVE		())) = = = = = = = = = = = = = = = = = =		

TABLE 4.2: Spectral Type, Proper Motion, and Photometry of CPM Candidates

ars in ^a While this distance is too low compared to the primary's Hipparcos distance of 24.6 pc, the companion is a tight, roughly equal-brightness binary. Adjusting the Hipparcos and 2MASS Hipparcos (Dommanget & Nys 2000), 4 = The Guide Star Catalog, Version 2.2.01 (1/271), 5 = Catalog of stars with high proper motions (1/306A), 6 = The USNO B 1.0 Catalog (Monet et al. 2003), 7 = The Revised NLTT Catalog (Salim & Gould 2003), 8 = Reid et al. (2004), 9 = All-sky Compiled Catalog (Kharchenko 2001), 10 = NLTT Catalog (Luyten 1979), 11 = The Catalog of Nearby Stars (Gliese & Jahreiß 1991), 12 = Gizis et al. (2000), 13 = An Astrometric Catalog (Rapaport et al. 2001), 14 = The Guide Star Catalog, Version 2.3.2 (Lasker et al. 2008), 15 = Kunkel et al. (1984), 16 = The Tycho-2 Catalog (Høg et al. 2000), 17 = Weis (1996), 18 = The NOMAD Catalog (Zacharias et al. 2004), 19 = The DENIS Consortium (The 2005). ^b See § 7.4. magnitudes accordingly changes the distance estimate to 24.0 ± 3.7 , a much better match with the distance to the primary. ž

4. The candidate has a large, matching proper motion with the primary. Three candidates (LHS 2995, NLTT 41169, and LHS 3402, companions to HD 125455, HD 140901, and HIP 91605, respectively) were confirmed by matching proper motions with the primary in the $0''.5 \text{ yr}^{-1} - 0''.8 \text{ yr}^{-1}$ range and proximity within 15". One additional candidate (LHS 25, companion to HD 26965) was confirmed despite a large separation of 83" because of the unusually large proper motion of $4''.1 \text{ yr}^{-1}$ which matches the primary's *Hipparcos* measure. As an additional support, these pairs have been recognized as gravitationally bound in many publications.

Several candidates were also refuted as physical associations upon a closer inspection. Six candidates were refuted because, despite an apparently matching proper motion with the primary seen on blinking, published proper motion values for the candidates were significantly different than the primary's *Hipparcos* value. One candidate was identified as a non-stellar plate defect because no star was found at the telescope at the expected position, and 10 more were refuted after the photometry obtained indicated a significantly different distance than the primary's *Hipparcos* value. In summary, this method revealed 78 candidate CPM companions, 61 of which were confirmed and 17 refuted.

4.2 Linear Motion of Field Stars

In addition to identifying CPM companions, blinking archival images is an effective method for verifying the status of double star measures in catalogs such as the WDS. The WDS is a *double* star catalog, not a *binary* star catalog, and as such explicitly contains many entries of unrelated pairs with relative separation measures. The blinking of archival images affords a simple way of identifying these pairs as potential companions with matching proper motions or unrelated field stars with differing motions. For distant field stars that have many WDS measures over an extended period, the relative separations will change in a linear fashion, reflecting the high proper motion of the primary. Figure 4.10 shows such an example of two WDS entries for HD 146361 (σ^2 CrB) that are clearly unrelated field stars, and Table 4.3 lists all the WDS entries for the stars of this study that were identified as field stars with a chance alignment but a differing transverse motion.



FIGURE 4.10: Examples of the linear motion of unrelated field stars that are listed in the WDS catalog as double star entries for HD 146361. The left panel is for WDS component C (ADS 9979C) and the right panel is for WDS component D (ADS 9979D). Plus signs indicate micrometric observations, asterisks indicate photographic measures, and filled circles represent Tycho measures. O-C lines connect each measure to its predicted position along the linear fit (thick solid line). The thick dashed line is the predicted movement based on the differential proper motions. The long dashed line connected to the origin indicates the predicted closest apparent position. The scale is in seconds of arc. An arrow in the lower right corner by the north and east direction indicators shows the direction of motion of the star.

WDS ID	Disc Desig	Comp ID	Primary Name	Nbr Obs	θ (deg)	ρ (")	Epoch
00022 + 2705	DII 722		11D 994090	64	205	161 7	2000
00022 ± 2705	DU 700 DU 700	AC AD	ПD 224950 ПD 224950	04 15	320 206	101.7	2000
00022 ± 2705	DU 755 HCW 1	AD AE	HD 224950	10	290	109.9	1921
00022 ± 2703 00066 ± 2001	ISW 1 FNC 1		HD 224950		309 106	149.4	1990
00000+2901	ENG I	A-CD	HD 000100	9 15	190	142.4 196 7	1991
00000+2901	511 049 DII 1999	AD CD	HD 000100	10	209	100.7	2002
00000 ± 2901	DU 1550 C 204		HD 000100	0 49	210	3.2	1999
00200 + 3814	5 304 C 204		HD 001502	45	22	100.7	2005
00200 + 5814	5 304 DUD 6	AC	HD 001302	4	200	22.0	1998
00229 - 1213	BUP 0		HD 001835	0 1 E	294 294	213.9	1998
00352 - 0330	BU 490	AB-U	HD 003190	15	324	24.4	1998
00394 + 2115	$511\ 550$	AB	HD 003051	9	80	107.0	1997
00484 + 0517	BUP 10	Aa-B	HD 004628	4	241	158.5	2000
00490 + 1656	BUP 12	AB	HD 004676	4	330	82.9	1998
00490 + 1050	BUP 12	AC	HD 004676	4	162	04.2	1998
00491 + 5749	STF 60	AC	HD 004614	4	258	216.0	1991
00491 + 5749	STF 60	AD	HD 004614	5	1	177.0	1991
00491 + 5749	STF 60	AE	HD 004614	12	127	91.8	2002
00491 + 5749	STF 60	AF	HD 004614	2	275	369.3	1991
00491 + 5749	STF 60	AG	HD 004614	15	256	409.8	2002
00491 + 5749	STF 60	AH	HD 004614	2	355	689.2	1991
00491 + 5749	STF 60	BC	HD 004614	3	237	152.0	1921
00491 + 5749	STF 60	BE	HD 004614	3	121	220.8	1991
00491 + 5749	STF 60	BH	HD 004614	3	356	679.9	1991
00491 + 5749	STF 60	FG	HD 004614	5	188	142.3	2000
00498 + 7027	ENG 2		$HD \ 004635$	9	277	90.9	1999
00531 + 6107	BU 497	AB	$HD \ 005015$	22	172	145.0	2003
00531 + 6107	BU 497	AD	$HD \ 005015$	3	145	105.6	1991
00531 + 6107	BU 497	AE	$HD \ 005015$	2	42	127.4	1959
00531 + 6107	BU 497	BC	$HD \ 005015$	7	162	0.9	1946
01083 + 5455	STT 551	Aa-B	$HD \ 006582$	14	270	428.9	1998
01083 + 5455	BUP 14	Aa-C	$HD \ 006582$	2	258	175.6	1991
01083 + 5455	BUP 14	Aa-E	HD 006582	1	145	87.7	1907
01083 + 5455	STT 551	Aa-F	HD 006582	2	328	53.2	1854
01083 + 5455	BUP 14	CD	$HD \ 006582$	2	115	4.3	1998
01291 + 2143	HO 9	AB	HD 008997	24	47	55.7	2003

TABLE4.3: Optical WDS entries

TABLE 4.3 – Continued

WDS ID	Disc Desig	Comp ID	Primary Name	Nbr Obs	(heta $(ext{deg})$	ρ (")	Epoch
$01201 \pm 21/3$	HO 0	AD	HD 008007	11	917	<u> </u>	1000
01291+2143 01291+2143	HO 9	RC	HD 008997	$\frac{11}{23}$	217 01	27	1999 2001
01251 + 2145 01350 - 2955	BU 1000	AR-D	HD 000557	20 12	10	$\frac{2.1}{132.2}$	1998
01368 + 4124	BUP 23	AB	HD 009826	1	128	102.2	1909
01368 + 4124	STT 554	AC	HD 009826	9	291	271.6	2006
01000 + 1121 01425 + 2016	HJ 2071	AB	HD 010476	6	10	53.2	1998
01425 + 2016	HJ 2071	AC	HD 010476	13	4	154.0	1998
01441 - 1556	BUP 25		HD 010700	5	157	137.0	2000
01477 + 6351	ENG 7		HD 010780	10	176	45.9	2003
01591 + 3313	ENG 9	AB	HD 012051	11	137	92.8	2002
01591 + 3313	BUP 28	AC	HD 012051	3	26	91.2	1934
02171+3413	DOR 66	AB	HD 013974	2	337	65.4	1907
02442 + 4914	STF 296	AC	HD 016895	18	229	77.2	1924
02442 + 4914	STF 296	BC	HD 016895	13	215	73.5	1924
03042 + 6142	KUI 11	AB	HD 018757	1	132	12.7	1931
03091 + 4937	BUP 38		HD 019373	1	132	146.2	1911
03194 + 0322	STT 557	AB	HD 020630	16	166	266.6	2002
03194 + 0322	BUP 42	BC	HD 020630	6	272	214.2	2000
03329 - 0927	MBA 1	AB	HD 022049	1	326	17.1	2001
03329 - 0927	MBA 1	AC	HD 022049	1	15	17.6	2001
03329 - 0927	MBA 1	AD	HD 022049	1	355	44.3	2001
03329 - 0927	MBA 1	AE	HD 022049	1	69	28.7	2001
03329 - 0927	MBA 1	AF	HD 022049	1	70	41.3	2001
03329 - 0927	MBA 1	AG	HD 022049	1	119	41.1	2001
03329 - 0927	MBA 1	AH	HD 022049	1	321	21.0	2001
03329 - 0927	MBA 1	AI	HD 022049	1	294	34.0	2001
03329 - 0927	MBA 1	AJ	HD 022049	1	145	27.9	2001
03329 - 0927	MBA 1	AK	HD 022049	1	208	38.5	2001
03562 + 5939	ENG 16	AB	HD 024409	10	7	134.7	1999
03562 + 5939	ENG 16	AC	HD 024409	7	50	193.4	1999
03562 + 5939	BUP 48	AF	HD 024409	2	75	37.6	1925
04033 + 3516	OSO 16		HD 025329	3	240	16.0	1994
04053 + 2201	STT 559	$Aa-B^a$	HD 025680	20	359	176.8	2003
04053 + 2201	STT 559	Aa-C	HD 025680	5	17	149.2	1997
04053 + 2201	STT 559	BC	HD 025680	8	124	58.2	1997
04076 + 3804	STT 531	AC	HD 025998	15	218	225.2	2002
04076 + 3804	BU 545	CD	HD 025998	15	305	1.3	1991

TABLE 4.3 – Continued

$\begin{array}{c} \text{WDS} \\ \text{ID} \end{array}$	Disc Desig	Comp ID	Primary Name	${f Nbr} {f Obs}$	$(heta \ (ext{deg})$	ρ (")	Epoch
04153 - 0739	STF 518	AD	HD 026965	18	97	77.9	1992
04153 - 0739	STF 518	AE	HD 026965	10	8	211.0	1907
04153 - 0739	STF 518	BD	HD 026965	1	196	147.0	1922
04153 - 0739	STF 518	BE	HD 026965	1	356	279.5	1922
04155 + 0611	H 4 98	AC	HD 026923	8	48	230.1	2003
04155 + 0611	H 4 98	CD	HD 026923	7	316	54.3	1987
05188 - 1808	SEE 50	AB	HD 034721	4	234	45.5	1951
05188 - 1808	SEE 50	BC	HD 034721	2	101	15.8	1951
05191 + 4006	STFB 3	AB	HD 034411	1	274	29.1	1900
05191 + 4006	STFB 3	AC	HD 034411	6	268	41.7	1934
05191 + 4006	STFB 3	AD	HD 034411	29	349	203.4	2003
$05191 {+} 4006$	DOB 4	AE	HD 034411	5	34	174.8	2003
$05191 {+} 4006$	KUI 20	CB	HD 034411	2	351	27.2	1934
$05191 {+} 4006$	DOB 4	DE	HD 034411	10	112	147.7	2003
05226 + 7914	STF 634	AB	HD 033564	86	135	25.8	2001
05226 + 7914	STF 634	AC	HD 033564	3	333	173.1	1999
05244 + 1723	S 478	AB	HD 035296	35	271	102.7	2002
05413 + 5329	BUP 82	AC	HD 037394	3	307	87.1	1925
05413 + 5329	BUP 82	AD	HD 037394	2	262	688.0	1909
05413 + 5329	BUP 82	BE	HD 037394	2	159	129.7	1910
05460 + 3717	BLL 16		HD 038230	3	104	104.0	1954
05584 - 0439	A 322	AC	HD 040397	13	304	195.5	2002
06173 + 0506	ENG 26	AB	HD 043587	8	241	179.7	2002
06173 + 0506	BUP 87	AC	HD 043587	1	265	58.5	1911
06173 + 0506	BUP 87	AD	HD 043587	1	231	69.3	1911
06467 + 4335	SHJ 75	AB	HD 048682	47	40	30.1	2007
06467 + 4335	WAL 47	AC	HD 048682	1	330	80.0	1944
07040 - 4337	WRH 38	AD	HD 053705	2	268	33.3	1999
07096 + 2544	HO 519	AB	HD 054371	6	103	22.2	1927
07096 + 2544	STTA 83	AC	HD 054371	32	80	120.5	2002
07291 + 3147	A 2124	AC	HIP 036357	14	285	221.7	1998
07291 + 3147	A 2124	CD	HIP 036357	3	270	102.3	1998
07549 + 1914	ENG 33	AB	HD 064468	12	285	96.8	2007
07549 + 1914	ENG 33	AC	HD 064468	10	65	125.4	2007
07549 + 1914	BUP 109	AD	HD 064468	3	28	43.1	1925
08116 + 3227	STT 564	AB	HD 068017	6	327	55.0	1915
08116 + 3227	STT 564	AC	HD 068017	10	66	289.3	2007

TABLE 4.3 – Continued

WDS ID	Disc Desig	Comp ID	Primary Name	Nbr Obs	(heta (deg)	ρ (")	Epoch
08122+1739	STF1196	AB-D	HD 068257	12	107	275.2	2007
08122 + 1739 08122 + 1739	ENH 1	AB-E	HD 068257	3	26	557.7	1991
08122 + 1739	ENH 1	AB-F	HD 068257	2	47	629.2	1894
08122 + 1739	ENH 1	AB-G	HD 068257	3	332	664.4	1991
08122 + 1739	STF1196	CD	HD 068257	5	107	274.8	1991
08122 + 1739	ENH 1	EF	HD 068257	6	106	218.7	1997
08379 - 0648	HJ 99	AB	HD 073350	15	181	60.2	2002
08379 - 0648	HJ 99	BC	HD 073350	3	215	9.8	1999
08391 - 2240	BU 208	AC	HD 073752	6	186	113.7	1999
08398 + 1131	ENG 36	AB	HD 073667	9	337	142.3	2002
08398 + 1131	BUP 119	BC	HD 073667	10	13	30.6	2002
09123 + 1500	BUP 125	$Aa-B^b$	HD 079096	1	118	35.4	1907
09123 + 1500	STT 569	Aa-C	HD 079096	11	217	204.4	2007
09123 + 1500	SLE 478	$Aa-D^b$	HD 079096	1	118	80.1	1984
09179 + 2834	ABT 6	AB-C	HD 079969	1	56	152.4	1921
09179 + 2834	ABT 6	AB-D	HD 079969	3	133	166.4	1999
10189 + 4403	ENG 43		HD 089269	14	97	145.2	2002
10306 + 5559	ARN 4	AC	HD 090839	3	294	233.1	2002
10314 - 5343	HJ 4329		HD 091324	28	103	73.9	2000
10365 - 1214	KUI 51		HD 091889	6	0	37.5	2007
11125 + 3549	STTA108	AB	$HD \ 097334$	23	67	156.5	2004
11125 + 3549	STTA108	AC	HD 097334	4	144	86.5	1998
11125 + 3549	STTA108	BD	$HD \ 097334$	2	88	34.9	1910
11182 + 3132	POP1219	AC	HD 098230	3	324	56.4	2007
11268 + 0301	STF1540	AC	HD 099491	3	187	90.5	1937
11387 + 4507	STF1561	AC	HD 101177	22	90	164.9	2006
11387 + 4507	STF1561	AD	HD 101177	2	76	704.2	1991
11387 + 4507	STF1561	AE	HD 101177	6	331	64.5	2001
11387 + 4507	STF1561	BC	HD 101177	17	89	173.3	2002
11387 + 4507	STF1561	BE	HD 101177	5	340	64.0	2000
11387 + 4507	STF1561	CD	HD 101177	3	72	552.9	1991
11411 + 3412	STT 574		HD 101501	11	88	158.4	1998
11507 + 0146	STT 576	AB	HD 102870	11	285	305.3	1984
11507 + 0146	STT 576	AC	HD 102870	4	80	421.7	2002
12337 + 4121	BU 1433	Aa-B	HD 109358	6	200	264.7	2007
13119 + 2753	STT 578		HD 114710	11	238	85.8	1924
13168 + 0925	KUI 62		HD 115383	1	89	34.3	1958

TABLE 4.3 – Continued

$\begin{array}{c} \text{WDS} \\ \text{ID} \end{array}$	Disc Desig	Comp ID	Primary Name	${f Nbr} {f Obs}$	$egin{pmatrix} heta\ (\mathrm{deg}) \end{bmatrix}$	ρ (")	Epoch
13169 + 1701	BU 800	AC	HD 115404	7	339	120.6	1999
13169 + 1701	BU 800	AD	HD 115404	1	88	50.8	1990
13184 - 1819	H 6 90		HD 115617	13	37	376.2	2007
13284 + 1347	STT 579	AB	HD 117176	14	127	268.6	2002
13284 + 1347	DIC 3	AC	HD 117176	1	263	325.5	1923
13547 + 1824	SHJ 169		HD 121370	27	87	113.3	2007
14190 - 2549	BU 1246	AB	HD 125276	4	217	8.0	1936
14190 - 2549	BU 1246	AC	HD 125276	9	117	82.6	1999
14514 + 1906	STF1888	AC	HD 131156	8	342	68.6	2000
14514 + 1906	STF1888	AD	HD 131156	7	286	159.6	2007
14514 + 1906	ARN 11	AE	HD 131156	6	99	269.2	2007
14514 + 1906	ARN 12	AF	HD 131156	5	38	333.7	2007
14514 + 1906	STF1888	BC	HD 131156	6	347	60.3	1953
14514 + 1906	STF1888	BE	HD 131156	10	99	274.5	2007
14537 + 2321	COU 101		HD 131582	3	68	51.0	2000
15193 + 0146	STF1930	AC	$HD \ 136202$	5	40	127.2	1924
15193 + 0146	STF1930	AD	HD 136202	4	268	674.7	1911
15232 + 3017	STF1937	AB-C	$HD \ 137107$	8	0	69.2	1984
15232 + 3017	STF1937	AB-D	$HD \ 137107$	6	41	217.5	2000
15282 - 0921	SHJ 202	BC	HD 137763	4	99	152.8	1999
15292 + 8027	STF1972	AC	HD 139777	3	102	153.8	1983
15360 + 3948	STT 298	AB-D	HD 139341	4	232	189.3	1934
15360 + 3948	STT 298	AB-E	HD 139341	4	335	456.4	1999
15360 + 3948	STT 298	CE	HD 139341	4	337	335.8	1999
15440 + 0231	A 2230	AC	HD 140538	9	208	195.5	2002
15440 + 0231	A 2230	AD	HD 140538	17	285	172.8	2002
15440 + 0231	A 2230	CE	HD 140538	14	235	171.0	2002
15475 - 3755	SEE 249	AC	HD 140901	6	125	8.4	1956
15532 + 1312	STT 583		HD 142267	9	86	102.4	1998
16010 + 3318	S 676		HD 143761	23	49	135.3	2002
16133 + 1332	STF2021	Aa-C	HD 145958	12	118	206.7	1998
16147 + 3352	STF2032	AC	HD 146361	11	93	24.2	2007
16147 + 3352	STF2032	AD	HD 146361	107	82	90.5	2006
16147 + 3352	STF2032	BD	HD 146361	66	81	95.0	1998
16156 - 0822	BUP 165		HD 146233	2	280	25.8	1958
16243 - 1338	BUP 169	AB	HD 147776	3	281	103.0	1909
16289 + 1825	STF2052	AC	HD 148653	2	29	143.3	1925

TABLE 4.3 – Continued

WDS ID	Disc Desig	Comp ID	Primary Name	Nbr Obs	(θ)	ρ (")	Epoch
	20018	12		0.00	(408)	()	<u> </u>
16315 - 3901	HDS2335		HD 148704	2	221	4.1	1991
16364 - 0219	BUP 171	•••	HD 149661	1	245	100.3	1910
17153 - 2636	SHJ 243	AD	HD 155885	11	338	276.6	1998
17153 - 2636	SHJ 243	AE	HD 155885	3	312	38.4	1998
17153 - 2636	SHJ 243	BD	HD 155885	6	339	284.9	1987
17191 - 4638	BSO 13	AC	HD 156274	2	279	41.8	1900
17191 - 4638	BSO 13	AD	HD 156274	1	30	47.0	1900
17207 + 3228	DOR 1	AB	HD 157214	12	340	308.1	2002
17207 + 3228	ARN 14	AD	HD 157214	1	59	395.0	2002
17207 + 3228	ARN 14	AE	HD 157214	3	51	302.2	2002
17207 + 3228	ARN 14	AF	HD 157214	3	104	383.2	2002
17207 + 3228	DOR 1	BC	HD 157214	2	216	8.8	1911
17350 + 6153	SDR 1	AB-D	HD 160269	2	245	23.9	1999
17419 + 7209	STF2241	AC	HD 162004	12	108	79.2	1999
17419 + 7209	STF2241	AD	HD 162004	1	84	100.5	1905
17419 + 7209	STF2241	CD	HD 162004	1	19	67.6	1908
17465 + 2743	ABT 14	Aa-D	HD 161797	1	0	256.1	1921
17465 + 2743	ABT 14	BC-D	HD 161797	1	7	272.6	1921
18025 + 2619	HO 564	AB	HD 164922	6	326	96.1	1999
18025 + 2619	HO 564	AC	HD 164922	2	57	80.3	1924
18055 + 0230	STF2272	AC	HD 165341	53	282	34.9	1947
18055 + 0230	STF2272	AD	HD 165341	18	324	88.1	2007
18055 + 0230	STF2272	AR	HD 165341	24	29	155.0	2000
18055 + 0230	STF2272	AS	HD 165341	35	11	193.2	2000
18055 + 0230	STF2272	AT	HD 165341	23	48	125.9	2007
18055 + 0230	STF2272	AU	HD 165341	15	334	184.5	1945
18055 + 0230	STF2272	AV	HD 165341	37	246	138.6	1946
18055 + 0230	STF2272	AY	HD 165341	2	354	231.9	2000
18055 + 0230	STF2272	BC	HD 165341	4	252	32.4	1932
18055 + 0230	STF2272	BD	HD 165341	1	247	69.3	1900
18055 + 0230	STF2272	\mathbf{BR}	HD 165341	22	50	121.1	2000
18055 + 0230	STF2272	BZ	HD 165341	2	163	68.3	2000
18055 + 0230	STF2272	VT	HD 165341	21	72	247.3	1946
18055 + 0230	STF2272	VW	HD 165341	4	270	180.9	1910
18055 + 0230	STF2272	VX	HD 165341	2	254	17.4	2002
18070 + 3034	AC 15	AC	HD 165908	6	59	96.2	1998
18070 + 3034	AC 15	AD	HD 165908	4	103	140.0	1998

TABLE 4.3 – Continued

$\begin{array}{c} \text{WDS} \\ \text{ID} \end{array}$	Disc Desig	Comp ID	Primary Name	Nbr Obs	$\left(egin{smallmatrix} heta \ (ext{deg}) \end{smallmatrix} ight)$	ρ (")	Epoch
18070 + 3034	AC 15	AE	HD 165908	4	78	168.8	1998
18070 + 3034	AC 15	AF	HD 165908	2	165	162.5	1998
18070 + 3034	AC 15	AG	HD 165908	5	360	177.0	1998
18570 + 3254	BU 648	AB-C	HD 176051	7	289	64.7	1960
18570 + 3254	BU 648	AB-D	HD 176051	4	198	85.0	1998
18570 + 3254	BU 648	AE	HD 176051	3	320	105.3	1934
18570 + 3254	BU 648	AF	HD 176051	3	87	100.3	1934
19080 + 1651	ENG 66	Aa-B	HD 178428	10	288	132.3	2007
19121 + 4951	STF2486	AC	HD 179957	4	101	26.8	2005
19121 + 4951	STF2486	AD	HD 179957	6	102	192.9	2000
19250 + 1157	STT 588	AB	HD 182572	29	286	101.7	2006
19250 + 1157	STT 588	AC	HD 182572	7	282	140.6	2000
19250 + 1157	COM 7	AD	HD 182572	3	140	78.8	1914
19250 + 1157	STT 588	BC	HD 182572	24	266	44.2	2006
19324 + 6940	STT 590		HD 185144	6	339	492.8	1999
19456 + 3337	STF2576	\mathbf{AC}	$HD \ 186858$	3	348	51.8	1998
19456 + 3337	STF2576	AD	HD 186858	7	333	35.1	1998
19456 + 3337	TKA 1	AE	HD 186858	2	249	25.9	2006
19456 + 3337	STF2576	BD	HD 186858	2	257	15.9	1919
19464 + 3344	STF2580	FH	HD 186858	32	125	110.5	2005
19464 + 3344	KPR 4	\mathbf{FI}	HD 186858	4	63	105.6	2005
19464 + 3344	KPR 4	FJ	HD 186858	1	61	40.9	1960
19464 + 3344	STF2580	GH	HD 186858	21	139	101.3	1999
19510 + 1025	J 124	AB	HD 187691	4	203	14.4	1958
19510 + 1025	POP1228	AD	HD 187691	3	121	53.5	2002
19510 + 1025	POP1228	AE	HD 187691	3	147	84.6	2002
20041 + 1704	STT 592	AB	HD 190406	20	289	166.9	2002
20041 + 1704	STT 592	AC	HD 190406	22	333	213.9	2001
20041 + 1704	BUP 202	AD	HD 190406	6	360	83.7	2002
20041 + 1704	BUP 202	AE	HD 190406	3	51	169.0	2000
20041 + 1704	BUP 202	AF	HD 190406	2	312	142.3	2000
20041 + 1704	ENG 69	BC	HD 190406	19	204	151.5	2000
20041 + 1704	STTA202	BG	HD 190406	15	232	182.7	2002
20041 + 1704	BUP 202	CH	HD 190406	3	185	95.4	2000
20052 + 3829	BU 1481	AB	HD 190771	1	230	12.4	1906
20052 + 3829	WAL 126	AC	HD 190771	3	180	40.0	1944
20096 + 1648	STF2634	AC	HD 191499	2	312	74.8	1924

TABLE 4.3 – Continued

WDS ID	Disc Desig	Comp ID	Primary Name	Nbr Obs	$\left(egin{smallmatrix} heta \ (ext{deg}) \end{smallmatrix} ight)$	ρ (")	Epoch
20111 + 1611	ENG 71	AB	HD 191785	9	148	205.4	2002
20111 + 1611	HZG 15	AD	HD 191785	7	267	40.8	1998
20111 + 1611	BUP 205	BC	HD 191785	4	273	61.7	2002
20140 - 0052	$BU \ 1485$	A-BC	HD 192263	19	102	73.1	2003
20140 - 0052	ABT 15	AD	HD 192263	1	244	71.3	1921
20140 - 0052	J 551	BC	HD 192263	4	265	0.2	1949
20140 - 0052	$BU \ 1485$	BC-D	HD 192263	8	65	23.5	1998
20324 - 0951	BU 668	AC	HD 195564	6	200	103.2	1921
20408 + 1956	BUP 215	AB	HD 197076	2	25	93.7	1924
21028 + 4551	BU 1138	AB	HD 200560	54	170	0.1	1985
21028 + 4551	BU 1138	CA	HD 200560	19	150	153.1	2002
21028 + 4551	BU 1138	CE	$HD \ 200560$	6	250	5.6	1962
21072 - 1355	BU 157	AC	HD 200968	15	287	26.2	1999
21072 - 1355	KPR 5	AD	HD 200968	1	78	276.2	1950
21145 + 1000	STF2777	AB-C	HD 202275	80	6	72.5	2005
21180 + 0010	ENG 82	AB	HD 202751	10	44	157.0	2003
21180 + 0010	TOB 317	AD	HD 202751	2	56	129.1	2000
21180 + 0010	BUP 228	BC	HD 202751	3	117	12.3	2000
21180 + 0010	LYS 44	BD	HD 202751	2	182	41.8	2000
21198 - 2621	BU 271	AC	HD 202940	3	72	81.7	1909
21198 - 2621	BU 271	AD	HD 202940	1	62	247.2	1917
21198 - 2621	BU 271	AE	HD 202940	2	32	180.6	1999
21441 + 2845	STF2822	AC	HD 206826	14	290	72.6	1999
21441 + 2845	STF2822	AD	HD 206826	46	45	197.5	2001
21441 + 2845	STF2822	BD	HD 206826	23	46	198.7	1991
21441 + 2845	$\mathrm{ES}\ 521$	DE	HD 206826	3	284	16.9	1999
21483 - 4718	BSO 15		HD 207129	29	351	75.2	1999
22159 + 5440	BU 377	AB	HD 211472	17	62	38.0	2006
22159 + 5440	BU 377	AC	HD 211472	6	52	35.2	2006
22159 + 5440	BU 377	AD	HD 211472	5	158	22.2	2006
22159 + 5440	BU 377	AQ	HD 211472	3	259	56.6	1999
22159 + 5440	BU 377	AR	HD 211472	2	256	62.7	1999
22159 + 5440	BU 377	AS	HD 211472	3	336	55.4	2006
22159 + 5440	BU 377	BC	HD 211472	11	303	6.7	2007
22159 + 5440	BU 377	QR	HD 211472	2	233	6.6	1999
22249 - 5748	I 383		HD 212330	2	237	81.2	1914
22467 + 1210	HJ 301	AC	HD 215648	5	15	145.0	1924

TABLE 4.3 – Continued

WDS ID	Disc Desig	Comp ID	Primary Name	${f Nbr} {f Obs}$	$\left(egin{smallmatrix} heta \ (ext{deg}) \end{smallmatrix} ight)$	ρ (")	Epoch
22514 + 1358	STT 597	AB	HD 216259	12	329	201.2	1998
22514 + 1358	BUP 233	BC	HD 216259	3	194	114.1	1998
23108 + 4531	HJ 1853		HD 218868	2	281	31.4	1905
23133 + 5710	STT 599		HD 219134	12	244	271.6	2002
23167 + 5313	BUP 235		HD 219623	2	320	129.3	1930
23399 + 0538	BUP 240	AB	HD 222368	4	305	119.2	2002
23399 + 0538	BUP 240	AC	HD 222368	3	21	307.3	2002
23524+7533	BU 996	AC	HD 223778	11	141	145.7	2000

^a Candidate companion is HIP 19075 which is a G-dwarf with a *Hipparcos* parallax indicating a distance of 220 pc, clearly unrelated to the primary as also seen upon blinking the DSS images. ^b The pairs Aa-B and Aa-D are in fact measures of the same pair, 74 years apart.

The large-scale homogeneity of the universe makes it very difficult to believe that the structure of the universe is determined by anything so peripheral as some complicated molecular structure on a minor planet orbiting a very average star in the outer suburbs of a fairly typical galaxy. — Stephen W. Hawking

OTHER ASTROMETRIC RESULTS

-5 -

5.1 The *Hipparcos* Double Stars

The *Hipparcos* mission obtained accurate astrometry of over 100,000 stars from space, and the cataloged results have revolutionized the understanding of all types of stars. The current effort has greatly benefited from this catalog, not only in defining a more accurate and complete sample as discussed in Chapter 2, but also in a more thorough investigation of companions. While astrometric companions were historically limited to wide CPM pairs as discussed in Chapter 4, the precise measurements of *Hipparcos* enabled one to probe significantly closer to the primary stars by, for example, deriving orbits from the observed photocentric motion around an implied center-of-mass or identifying unseen companions by the deviation of proper motion from a linear path. This section covers the analysis of the *Hipparcos* double stars and their evaluation for the multiplicity statistics derived.

The primary identification of companion stars in the *Hipparcos* catalog is done via field H59, with further details in the *Double and Multiple Systems Annex*. Potential companions are identified as one of five types by this multiplicity flag: C (component solutions) where a nearby source is resolved, G (acceleration observed in the proper motion implying an unseen companion), O (orbital solutions obtained using *Hipparcos* data), V (movement detected in the photocenter based on the variability of one or more components), or X (stochastic solutions, implying that an acceptable astrometric solution was not obtained as either a single or a double star). None of the targets of this work are flagged as 'V' in field H59, and hence this indicator is not discussed further. In addition to this multiplicity flag, field H61 of the main catalog lists an 'S' for suspected non-single stars, based on the astrometric fit obtained, although a satisfactory double star solution could not be obtained. Some of these correspond to the 'X' entries of field H59. Also, field H52 contains a 'D' (duplicity-induced variability) identifying photometric variability presumably caused by a companion. While these two indicators contribute 16 possible companions that are not identified by field H59, I do not consider them reliable companion detection indicators for this work without an independent confirmation. There are several additional flags in the catalog that can imply a companion, such as H10, which contains a reference flag for components of double or multiple systems, and H62, which is a component designation for double or multiple systems. However, these indicators overlap with the flags discussed above in all cases, and are not discussed further because they do not contribute any new companions. The following subsections treat each of the companion indicators considered and describe the methods of this survey in evaluating their verity.

While the *Hipparcos* identification of companions is a very useful source, it is neither definite nor complete. Quist & Lindegren (2000) modeled a binary distribution based on the DM91 statistics and found that the number of orbits presented in the *Hipparcos* catalog is deficient by a factor of about three. On the other hand, the description of the catalog points out that several of the companion identification flags may correspond to optical pairs (more on this in the subsections below), and follow-up efforts have had limited success in resolving suspected companions. Mason et al. (1999) studied a majority of the unresolved *Hipparcos* double stars via speckle interferometry and were able to resolve components for only 13% of the binaries flagged as 'G' in field H59. Mason et al. (2001) confirm this low detection fraction, pointing out that the multiplicity fraction observed among *Hipparcos* double stars is far less than that of a random selection of bright field stars, and speculate that speckle interferometry may not adequately address the separation- Δ mag space occupied by the *Hipparcos* double stars. Falin & Mignard (1999) reanalyzed the *Hipparcos* raw data with a specific view to improving the double or multiple star information presented and found that many well-observed systems retain poor solutions. They speculated that this could be due to one of three reasons – they are binaries with periods of 1-10 years, or they are components of multiple systems of weak hierarchy with motions too complicated to model with *Hipparcos* data, or that some of the data relates to poor pointing. Finally, one notable example involving Arcturus has been published, concluding that instrumental and random errors resulted in the mimicking of a spurious binary (Soderhjelm & Mignard 1998). Due to these factors, I have not taken every *Hipparcos* companion designation as real, but rather evaluated them individually depending on their type, as described below.

5.1.1 Component Solutions

These companions, identified by a 'C' in field H59, represent double stars resolved by *Hipparcos* as separated components which could be modeled as single stars, usually with an assumed common parallax. For these systems, field H61 gives an indication of the reliability of the double or multiple star solution with a quality of A (good), B (fair), C (poor), and D
(uncertain). Among the sample studied here, 62 stars were flagged as *Hipparcos* component solutions, and 59 of these have independent supporting evidence and are hence considered real. Of the remaining three, two are flagged as quality 'C', or poor solution, in field H61 (HD 64606 and 111312) and are retained as candidate companions, and an additional one (HD 148704) has been refuted by this work despite its quality 'A' designation in *Hipparcos* based on evidence of differential proper motion (see §7.4 for more information). Table 5.1 lists all entries of this type identified by the HD and HIP names and the companion ID as in the Annex, along with the quality flag (field H61), the final status used in the multiplicity statistics derived here, and the reason for this conclusion.

HD Name	HIP Name	Companion ID	Solution Quality	Companion Status	Reason
000123	000518	В	А	YES	1
003196	002762	В	А	YES	2
003443	002941	В	А	YES	2
004614	003821	В	А	YES	1
007693	005842	D	А	YES	1
009770	007372	В	\mathbf{C}	YES	1
010360	007751	А	D	YES	1
016765	012530	В	В	YES	3
018143	013642	В	А	YES	1
020010	014879	В	А	YES	1
024409	018413	D	А	YES	4
025893	019255	В	А	YES	1
035112	025119	В	А	YES	1
037572	026373	В	В	YES	6
039855	027922	В	А	YES	6
048189	031711	В	А	YES	7
053705	034065	В	А	YES	6

TABLE 5.1: *Hipparcos* Component Solutions

Continued on Next Page...

TABLE 5.1 - Continued

HD Name	HIP Name	Companion ID	Solution Quality	Companion Status	Reason
057095	035296	В	А	YES	1
064096	038382	В	А	YES	2
064606	038625	В	\mathbf{C}	MAY	4
068255	040167	В	В	YES	1
068255	040167	\mathbf{C}	В	YES	1
073752	042430	В	А	YES	1
096064	054155	В	\mathbf{C}	YES	1
096064	054155	\mathbf{C}	\mathbf{C}	YES	6
099491	055846	В	D	YES	6
100180	056242	В	А	YES	6
101177	056809	В	А	YES	1
111312	062505	В	\mathbf{C}	MAY	10
115404	064797	В	А	YES	1
116442	065352	В	А	YES	6
128620	071683	В	D	YES	2
130042	072493	В	А	YES	7
131156	072659	В	А	YES	1
133640	073695	В	В	YES	1
137107	075312	В	А	YES	2
139341	076382	В	А	YES	1
145958	079492	В	А	YES	1
146361	079607	В	В	YES	2
148653	080725	В	А	YES	1
148704	080925	\mathbf{C}	А	NO	8
153557	083020	В	В	YES	7
155885	084405	А	А	YES	1
156274	084720	В	А	YES	9
158614	085667	В	А	YES	2
160269	086036	В	А	YES	9
162004	086620	В	D	YES	6
165341	088601	В	D	YES	2
165908	088745	В	А	YES	1
	091605	В	В	YES	6
176051	093017	В	А	YES	9
177474	093825	В	А	YES	1
179957	094336	В	А	YES	1
184467	095995	\mathbf{S}	В	YES	2

HD Name	HIP Name	Companion ID	Solution Quality	Companion Status	Reason
186858	097222	В	А	YES	1
189340	098416	В	А	YES	2
191499	099316	В	А	YES	10
200968	104239	В	А	YES	3
202275	104858	В	А	YES	2
202940	105312	В	А	YES	9
206826	107310	В	А	YES	1
212168	110712	В	А	YES	11

TABLE 5.1 - Continued

NOTES.—Column 6 notes: (1) = Visual orbit exists (see §5.2); (2) = Double-lined spectroscopic and visual orbits exist (see §5.2 and §6.2); (3) = Measurements in the WDS confirm orbital motion; (4 = Single-lined spectroscopic orbit exists (see §6.2); (5) = Radial velocity variations observed (see §6.2); (6) = Companion has matching independent parallax and proper motion with primary (see §4.1); (7) = Matching proper motion and proximity to primary confirm companionship; (8) = Proper motions do not match (see §7.4); (9) = Single-lined spectroscopic and visual orbits exist (see §5.2 and §6.2); (10) = See individual system notes in §7.4; (11) = Companion has matching proper motion and distances with the primary (see §4.1).

5.1.2 Accelerating Proper Motions

Due to the high precision of the *Hipparcos* astrometry, deviations of proper motion from a linear path can be detected during the approximately three years of observations, indicating orbital motion around a center of mass with an unseen companion with orbital periods larger than some 10 years. Such companions, identified as ones requiring higher-order terms

of proper motion to obtain an acceptable fit with the observations, are identified with a value of 'G' in field H59. As noted above, follow-up efforts with speckle interferometry (Mason et al. 1999, 2001) have failed to resolve the majority of these companions. In a complementary approach, several studies (e.g., Makarov & Kaplan 2005; Frankowski et al. 2007, and references therein) have compared the *Hipparcos* proper motions with compilations of ground-based measurements over a much longer period, such as the Tycho-2 catalog (Høg et al. 2000). Proper motions in the Tycho-2 catalog are the average of measurements over many decades, and typically over 100 years, so these values should average out any curvilinear motion from orbits of a few decades and represent the transverse space motion. In comparison, the *Hipparcos* data obtained over some three years should be significantly influenced by such orbits. On the other hand, orbits of many decades to a few centuries will affect the Tycho-2 measurements, but hardly influence the *Hipparcos* data. So, differences in the proper motions listed in these two catalogs can contain clues about companions, some of which might already be known via spectroscopic or visual techniques. Makarov & Kaplan (2005) followed this approach in identifying companions based on a $3.5-\sigma$ difference between the proper motion in either coordinate from these two catalogs and estimated a minimum companion mass. Frankowski et al. (2007) developed a χ^2 test to minimize false identification of companions, and identified 3565 proper-motion binaries with a greater than 99.99% confidence level, which they estimate to be an order of magnitude better than that of Makarov & Kaplan (2005).

Table 5.2 lists all stars from the sample studied here which either have a 'G' designation in

167

proper motions. Taking a conservative approach, I divide the absolute difference between the proper motion from the two catalogs in each coordinate by the larger of the corresponding uncertainties, representing the x- σ difference in each coordinate. The root-sum-square of these two values is then used to test for the 3- σ difference and this value is also listed in column 8 of the table. For the purposes of the multiplicity statistics derived here, companions which are flagged as 'G' in field H59 are only considered real if they are confirmed by a $3-\sigma$ difference between *Hipparcos* and Tycho-2 proper motions or by other factors such as the existence of spectroscopic or visual orbits of a few decades. Entries lacking definite supporting evidence, but passing the Frankowski et al. (2007) χ^2 test are retained as candidates for further investigation, and *Hipparcos* entries without any supporting evidence are refuted as false detections. On the other hand, companions not flagged as 'G' in *Hipparcos* field H59, but with a 3- σ difference in the *Hipparcos* and Tycho-2 proper motions are also evaluated and considered real if they passed the χ^2 test developed by Frankowski et al. (2007) or are supported by other evidence of the companion such as a visual or spectroscopic orbit, or the presence of a nearby companion which could induce curvilinear motion over a hundred years such that it would have been picked up by the Tycho-2 values. The remaining such detections are left as candidates for further investigations.

In addition to the HD and HIP designation of the primaries, Table 5.2 lists the H59 value, the *Hipparcos* and Tycho-2 proper motions, an indicator if the star was identified as a proper-motion binary in Makarov & Kaplan (2005) or Frankowski et al. (2007), a final

companion status of YES (confirmed) or MAY (candidate), and a reason for this status. Among the stars of the current sample, 18 are flagged as 'G' in field H59, all but two of which are supported by a greater than 3- σ difference between the proper motions. One of the remaining two (HIP 36357) is supported by other evidence of companionship, while the other (HD 25998) is retained as a candidate. Most of the entries in the table have supporting visual or spectroscopic evidence and were independently detected by these methods, but nine companions were identified solely by the 'G' flag and supported by proper motion differences of 3.5–58.4 σ . An additional 11 systems were identified solely based on proper motion differences without the *Hipparcos* 'G' flag designation. Five of these are supported by the Frankowski et al. (2007) χ^2 test and have differences of 5.3–30.0 σ . The remaining six are identified as candidates for follow-up work and have proper motion differences of 3.1–4.4 σ .

Solutions
Motion
Proper
Accelerating
5.2:
TABLE

	Reason	- 0	N C	103	1	4	н к	, c 1	4	1	3	9	с, .	- 1	0 4	т ю	3	2	ı ت	7	r- 1	n oc	5 0	n	ę	6	6 j	10	o -		1	4	11	_ م 1	с <u>т</u> н	- 1	0	2	4 0	4 a	т LC	- 2	3	- 0	- 12	+ 10) m	3	0.	4 -	11	
Companion	Status	YES	Y ES V P S	YES	YES	YES	YES VFC	VES VES	YES	YES	YES	MAY	YES	YES	2 P V V P V V V V	YES	YES	\mathbf{YES}	YES	YES	YES	Y EV VES	YES	YES	\mathbf{YES}	MAY	MAY	Y EX V D C	V FS	YES	\mathbf{YES}	YES	YES	MAY VFS	YES	YES	YES	YES	YES	VES	YES	YES	YES	YES	Y ES V FS	YES	YES	YES	YES	Y ES V FS	YES	
	$\mathbf{F07}^{b}$:	:	: :	:	Y	:>	-	Y	:	:	Y	Y		× >	•	Y	Y	;;	Υ	Y	: }	• :		Y	:	:	:>	н	÷	:	÷	:	:	: :	Y	Y	:	÷			Y	Υ	Y	:		Y	÷	÷	÷	: :	
	$MK05^{a}$:			:	Y	:>	-	Y	:	:	:	Y		- >		Υ	:	:	:	:	:>			Y	:	:	:>	I	÷	:	:	:	:		Y	Υ	:	:	:		Y	Y	:	:		ł	:	:	:	: :	
μ difference	(<i>σ</i>)	16.2	22.6	4.0	6.6	31.8	24.8 24.3	6.4.0	9.6	7.1	23.5	2.4	10.1	14.5	16.8	3.7	8.9	6.0	6.3	3.D	14.9	11.7	9.6	10.4	6.4	3.1	4.1	28.7	6.0 9 9	10.6	21.2	11.6	10.9	0.0 0.0	6.3	8.1	15.4	18.7	5.7	4.0	5.9	8.1	7.7	8.0 1	5.5 20 1	4.4	45.8	4.4	7.5	2, 2 2, 2	4.0 5.1	
-2 ======	$\mu_{\delta}~(\mathrm{mas}~\mathrm{yr}^{-1})$	30.1 ± 1.7	-23.2 ± 0.5 -130 ± 2.3	-124.7 ± 1.4	108.3 ± 3.0	-152.2 ± 1.0	-14.1 ± 1.3 672 0 \pm 0 4	1.7 T 7.010	-122.6 ± 1.1	-185.4 ± 1.6	169.7 ± 1.0	-203.1 ± 1.3	333.7 ± 1.2	-152.1 ± 1.3	7.1 ± 0.761	-206.5 ± 1.1	164.6 ± 1.0	-167.8 ± 0.8	72.4 ± 3.4	37.5 ± 1.3	395.3 ± 1.1	1687.0 ± 1.0	-338.9 ± 0.7	-47.7 ± 1.4	-550.8 ± 1.0	-126.7 ± 1.3	-646.5 ± 0.8	-129.4 ± 1.7	-510.5 ± 1.1	-247.8 ± 1.4	1.5 ± 1.0	-176.9 ± 1.2	39.2 ± 1.7	-100.4 ± 2.0 -223.2 ± 0.8	-259.0 ± 1.1	50.4 ± 0.4	-290.9 ± 3.0	-45.9 ± 1.0	-307.5 ± 1.2	-120.6 ± 0.0	-129.0 ± 0.0 -330.8 ± 1.7	11.0 ± 0.8	-327.8 ± 1.2	9.9 ± 1.2	-359.0 ± 1.0 51 0 \pm 0 9	-147.2 ± 0.8	683.0 ± 1.1	-256.9 ± 1.2	-85.1 ± 2.8	111.5 ± 0.7 $282 2 \pm 1.2$	-326.2 ± 1.2	
===== Tycho	μ_{lpha} (mas ${f yr}^{-1}$)	270.6 ± 1.6	415.9 ± 0.5 1 201 0 \pm 2 2	518.8 ± 1.4	404.9 ± 3.3	806.6 ± 1.0	302.6 ± 1.4 2150.2 ± 2.5	0.7 T 0.0017	274.5 ± 1.1	274.0 ± 1.7	-265.1 ± 0.9	166.8 ± 1.2	196.7 ± 1.1	69.7 ± 1.3	40.9 H 1.0	73.6 ± 1.0	-195.4 ± 1.0	-115.3 ± 0.8	-26.0 ± 3.8	203.3 ± 1.4	-93.0 ± 1.2	137.2 ± 0.0	-60.0 + 0.7	-259.1 ± 1.5	180.1 ± 1.1	-155.9 ± 1.4	-464.7 ± 0.8	79.8 ± 1.7	- TA/ 0 H 10	-723.0 ± 1.3	-577.3 ± 1.0	-389.8 ± 1.1	59.9 ± 1.9	$-220 \pm 1.022 - 180 \pm 0.7$	-103.0 ± 0.0 623.9 ± 1.2	-480.8 ± 0.4	-584.2 ± 3.0	-596.4 ± 1.1	91.1 ± 1.2 951.0 ± 1.1	-720 + 0.8	-12.0 ± 0.0 -109.4 ± 1.7	-824.4 ± 0.8	-15.5 ± 1.0	-443.7 ± 1.2	76.4 ± 1.0 - 455 2 + 0.8	-48.0 ± 0.8	-532.9 ± 1.0	-78.4 ± 1.2	-289.0 ± 3.0	197.8 ± 0.7	-329.3 ± 1.2 -427.6 ± 1.1	
cos =====	$\mu_{\delta}~(\mathrm{mas}~\mathrm{yr}^{-1})$	17.77 ± 0.70	-36.47 ± 0.61 -1715 ± 1.23	-10.10 ± 1.20 119.52 ± 0.72	127.43 ± 0.48	-180.16 ± 0.47	16.66 ± 1.41 654.32 ± 0.55	-4.39 ± 0.50	-127.75 ± 0.81	-191.44 ± 1.15	159.32 ± 0.96	-203.52 ± 0.55	336.76 ± 0.51	-139.41 ± 0.94	-08 03 + 0.60	-203.34 ± 0.66	171.18 ± 0.50	-164.06 ± 0.76	72.69 ± 0.66	40.89 ± 0.72	401.32 ± 2.10	172280 ± 0.55 172280 + 0.55	-344.83 ± 1.03	-62.07 ± 1.48	-544.36 ± 0.50	-130.52 ± 0.66	-644.64 ± 0.61	-150.94 ± 1.15	-50762 ± 0.51	-260.62 ± 0.46	14.80 ± 0.52	-183.86 ± 0.60	44.66 ± 1.46	-100.49 ± 0.07 -910.55 ± 1.10	-260.84 ± 0.63	54.18 ± 0.47	-244.87 ± 0.71	-60.27 ± 0.59	-304.04 ± 0.47	-13203 ± 0.60	-320.91 ± 0.95	2.29 ± 1.07	-337.07 ± 0.88	18.94 ± 1.17	-363.37 ± 0.79 27.52 ± 1.47	-144.73 ± 0.78	696.64 ± 0.73	-252.61 ± 1.04	-86.88 ± 1.12	110.77 ± 0.51	-333.41 ± 1.43	
==== Hippar	$\mu_{lpha}~({ m mas}~{ m yr}^{-1})$	247.36 ± 0.81	407.68 ± 1.31	516.74 ± 1.04	411.11 ± 0.50	791.35 ± 0.65	286.10 ± 1.01	2032.64 ± 0.30 197.34 + 0.77	264.17 ± 1.24	262.72 ± 1.86	-284.06 ± 0.90	163.93 ± 0.65	185.91 ± 0.46	54.20 ± 1.44	02.14 ± 0.03	71.23 ± 0.93	-189.37 ± 0.71	-119.32 ± 1.06	-50.08 ± 0.73	206.58 ± 0.48	-75.43 ± 0.75	-220.83 ± 0.20	-68.46 ± 1.11	-251.57 ± 2.07	180.46 ± 0.91	-157.34 ± 0.47	-460.69 ± 1.17	28.29 ± 2.00	- 194.20 X 1.03 49 78 + 1 15	-730.05 ± 0.71	-593.87 ± 0.68	-378.76 ± 0.59	79.70 ± 1.90	-221.30 ± 0.05 - 180 70 ± 1.16	-103.13 ± 1.10 631.21 ± 0.90	-480.33 ± 0.61	-580.93 ± 0.95	-583.27 ± 0.91	85.26 ± 0.64	-330.33 ± 0.65 -73.73 ± 0.65	-107.28 ± 0.70	-824.15 ± 1.27	-15.47 ± 0.89	-436.24 ± 1.20	72.69 ± 1.09 - 482 47 + 2.25	-44.96 ± 0.95	-488.79 ± 0.58	-81.41 ± 0.89	-266.47 ± 0.86	199.89 ± 0.31 245.02 ± 1.56	-340.33 ± 1.00 -428.05 ± 1.47	
	H59	С	ວເ		U	U	U	: C	00	U	U	U	:	00	5		:	:	υ	ت	υ	: C	50	U	:	:	÷	C	:	: :	U	0	C	: 0	00	:	÷	ۍ ۱	0	÷C	00	ŭ	IJ	00	30)	: U	÷	U	:0	JO	Page
HIP	Name	000518	002041	003850	005896	007918	007751	011072	013081	013642	018413	019335	019233	025119	02204/	028267	029860	030630	031711	033817	034052	037853	038382	038625	039064	039342	040118	040167	045617	047080	056809	062145	062505	063773	064797	067275	067620	067742	068682	070857	072493	072875	073241	073695	075718 076382	077052	078709	079578	020000	080686	080925	on Next
HD	Name	000123	003196	004747	007788^{c}	010307	010360	014802	017382	018143	024409	025998	026491	035112	030587	040397	043587	045088	048189	052698	053680^{a}	053077 063077	064096	064606	065430	067199	068017	068255	001210	082885	101177	110833	111312	113283	115404	120136	120690	120780	122742 195976	128642	120042	131582	131923	133640	137763 139341	140538	144287	145825	146361	147584	148704	Continued

Continued
L
2
Ŋ
TABLE

	D	HIP		===== $Hipp$	arcos =====	===== Tyck	10-2 ======	μ difference			Companion	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		Name	H59	$\mu_{lpha} \; (\mathrm{mas} \; \mathrm{yr}^{-1})$	$\mu_{\delta} \;(\mathrm{mas}\;\mathrm{yr}^{-1})$	$\mu_{lpha}~({ m mas}~{ m yr}^{-1})$	$\mu_{\delta} \;(\mathrm{mas}\;\mathrm{yr}^{-1})$	(α)	$\mathbf{MK05}^{a}$	$\mathbf{F07}^{b}$	Status	Reason
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		083020	C	-146.90 ± 1.15	272.21 ± 1.35	-153.9 ± 1.3	267.8 ± 1.4	6.2			YES	5
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		084720	U	1035.25 ± 1.38	109.22 ± 0.65	1053.5 ± 2.1	144.0 ± 2.0	19.4	:	:	YES	ŝ
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		085667	υ	-126.64 ± 1.72	-172.00 ± 0.91	-126.7 ± 1.1	-179.6 ± 1.1	6.9	:	:	YES	2
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		086036	υ	277.38 ± 0.54	-525.62 ± 0.60	265.4 ± 3.3	-520.9 ± 3.2	3.9	:	:	YES	4
		086722	Ċ	-123.15 ± 1.00	-619.84 ± 0.88	-123.1 ± 1.1	-628.0 ± 1.0	8.2	Y	Y	YES	4
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		086974	:	-291.42 ± 0.49	-750.00 ± 0.53	-310.3 ± 0.4	-750.3 ± 0.4	38.5	Υ	Y	YES	1
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		088601	U	124.56 ± 1.15	-962.66 ± 0.91	276.3 ± 2.3	-1091.8 ± 2.3	86.6	:	:	YES	2
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		088622	:	-30.66 ± 0.88	-322.06 ± 0.75	-26.0 ± 1.1	-316.7 ± 1.1	6.5	Y	Y	YES	6
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		089042	U	-77.60 ± 0.59	234.68 ± 0.44	-81.5 ± 0.9	221.2 ± 0.9	15.6	Y	Y	YES	7
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		089805	:	38.89 ± 0.57	-276.16 ± 0.51	37.9 ± 1.4	-280.4 ± 1.4	3.1	:	:	YES	5
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		093825	U	96.93 ± 2.44	-279.67 ± 1.34	87.7 ± 1.2	-283.9 ± 1.2	4.9	:	:	YES	1
		094336	U	-205.02 ± 0.97	624.33 ± 0.89	-209.5 ± 1.3	622.2 ± 1.4	3.8	:	:	YES	1
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		095149	Ċ	78.88 ± 4.08	-108.93 ± 2.50	87.6 ± 1.2	-86.4 ± 1.3	9.3	Y	Y	YES	7
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		097222	υ	13.30 ± 1.07	-440.57 ± 1.35	18.9 ± 2.6	-445.8 ± 2.4	3.1	:	:	YES	1
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		098416	υ	-246.73 ± 2.31	-392.36 ± 1.63	-282.0 ± 1.0	-399.7 ± 1.0	15.9	:	:	YES	2
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		098921	:	263.35 ± 0.46	111.57 ± 0.46	259.2 ± 1.1	115.7 ± 1.1	5.3	Y	Y	YES	6
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		099461	:	456.89 ± 0.89	-1574.91 ± 0.61	458.4 ± 1.1	-1569.3 ± 1.1	5.3	Y	Y	YES	5
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		099316	υ	3.79 ± 1.00	175.79 ± 1.02	3.9 ± 1.4	167.9 ± 1.4	5.6	:	:	YES	S
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		100017	:	468.52 ± 0.55	296.81 ± 0.41	472.5 ± 1.1	296.1 ± 1.2	3.7	:	:	MAY	6
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		101345	:	307.59 ± 0.88	106.07 ± 0.66	307.2 ± 0.7	103.7 ± 0.7	3.4	:	:	YES	ъ
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		101382	0	-156.89 ± 0.53	452.80 ± 0.47	-154.3 ± 0.9	454.5 ± 0.9	3.4	:	:	YES	7
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		103859	:	402.30 ± 0.66	141.72 ± 0.57	396.1 ± 0.8	141.0 ± 0.8	7.8	Y	Y	YES	ъ
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		104239	υ	382.32 ± 1.31	-46.55 ± 0.60	382.3 ± 0.9	-39.9 ± 0.8	8.3	:	:	\mathbf{YES}	ъ
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		105312	υ	-582.35 ± 1.11	-357.67 ± 0.62	-568.3 ± 1.5	-353.7 ± 1.5	9.7	:	:	YES	4
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		105911	IJ	264.07 ± 1.28	184.71 ± 0.97	253.2 ± 1.1	179.6 ± 1.2	9.5	Y	Y	YES	7
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		107310	υ	260.33 ± 0.61	-242.73 ± 0.57	277.4 ± 2.7	-251.1 ± 2.6	7.1	:	:	YES	1
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		110109	:	439.86 ± 0.53	-632.60 ± 0.43	436.8 ± 0.9	-632.8 ± 0.9	3.4	:	:	YES	ъ
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		110649	:	180.71 ± 0.41	-331.27 ± 0.38	150.6 ± 1.1	-344.9 ± 1.1	30.0	Y	Y	YES	6
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		112117	:	6.15 ± 0.63	-331.43 ± 0.51	3.1 ± 1.1	-326.9 ± 1.0	5.3	Y	Y	YES	ъ
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		117712	:	341.82 ± 0.53	41.88 ± 0.47	325.8 ± 1.0	45.6 ± 1.1	16.4	Y	Y	YES	7
036357 G 160.40 ± 1.56 174.68 ± 1.02 159.7 ± 1.0 175.8 ± 1.1 1.1 YES 5		000171	×	778.59 ± 2.81	-918.72 ± 1.81	829.9 ± 1.2	-989.4 ± 1.1	43.1	Y	Y	YES	4
		036357	Ċ	160.40 ± 1.56	174.68 ± 1.02	159.7 ± 1.0	175.8 ± 1.1	1.1	:	:	YES	5 C

NOTES.—Column 12 notes: (1) = Visual orbit exists (see § 5.2); (2) = Double-lined spectroscopic and visual orbits exist (see § 5.2 and § 6.2); (3) = Single-lined spectroscopic orbit exists

decades; (6) = *Hipparcos* G flag and the χ^2 test in Frankowski et al. (2007) suggest an unseen companion, but because the *Hipparcos* and Tycho-2 proper motions differ by less than 3σ , this variations indicate a spectroscopic binary, but not enough observations exist to derive an orbit. (9) = Greater than 3σ difference in proper motion is the only evidence of a companion. Companion is considered confirmed if it also passed the χ^2 test in Frankowski et al. (2007), otherwise is retained as a candidate; (10) = Thesis primary star is HD 68257. At least a quintuple (2000) show that the *Hipparcos* motion is significantly different from that of the Fifth Fundamental Catalog (Fricke et al. 1988, FK5) and present a convincing photocentric orbit; (13) = See^a 'Y' in this column indicates that this was identified as a proper-motion binary in Makarov & Kaplan (2005) ^b 'Y' in this column indicates that this was identified as a proper-(see § 6.2); (4) = Single-lined spectroscopic and visual orbits exist (see § 5.2 and § 6.2); (5) = Nearby resolved companion exists, likely causing detectable proper motion deviation over several is retained as a candidate for further investigations. (7) = Companionship is confirmed based on Hipparcos, G' flag and a greater than 3σ difference in proper motion; (8) = Radial velocity system with two visual and spectroscopic orbits; (11) = Double-lined spectroscopic orbit exists (see § 6.2); (12) = Tycho-2 does not have a proper motion for this star, but Gontcharov et al. Table 5.3.

^e Wide companion, 756" away $^{\rm d}$ Wide companion, 185 $^{\prime\prime}$ away from to HD 53705. ^c Wide companion, 319" away from to HD 7693. motion binary in Frankowski et al. (2007) from to HD 36357.

5.1.3 Orbital Solutions

Nineteen of the targets studied in this work have orbital solutions in the *Hipparcos* catalog, for which the *Hipparcos* data enabled the determination of at least one of the orbital parameters. Table 5.3 lists these stars along with their orbital period from *Hipparcos*, the final status for the multiplicity statistics derived, and a corresponding reason for this conclusion. As seen in the table, 15 of the 19 are confirmed companions, three are refuted, and one is still a candidate.

HD Name	HIP Name	Period (days)	Companion Status	Reason
001273	001349	411.4	YES	1
006582	005336	7816.0	YES	1
010476	007981	207.3	NO	2
013974	010644	10.0	YES	3
014214	010723	93.5	YES	1
016739	012623	331.0	YES	3
032850	023786	204.4	YES	1
110833	062145	270.2	YES	1
112914	063406	736.8	YES	1
113449	063742	231.2	YES	4
121370	067927	494.2	YES	1
122742	068682	3614.9	YES	1
128642	070857	179.7	YES	1
131511	072848	125.4	YES	1
142373	077760	51.3	NO	5
143761	078459	78.0	NO	4
160346	086400	83.9	YES	1
195987	101382	57.3	YES	3
203244	105712	1060.6	MAY	4

TABLE 5.3: Hipparcos Orbital Solutions

NOTES.—Column 5 notes: (1) = Single-lined spectroscopic and visual orbits exist (see § 5.2 and § 6.2); (2) = *Hipparcos* orbital solution has $i = 89^{\circ} \pm 23^{\circ}$, but the radial velocity variations of less than 0.1 km s⁻¹ over 3.8 years or 6.7 orbital periods (Nidever et al. 2002) raise substantial doubts about the *Hipparcos* orbit; (3) = Double-lined spectroscopic and visual orbits exist (see § 5.2 and § 6.2); (4) = See individual system notes in § 7.4; (5) = *Hipparcos* orbital solution has $i = 132^{\circ} \pm 28^{\circ}$, but radial velocity variations of less than 0.1 km s⁻¹ over 9.7 years or 69.4 orbital periods (Nidever et al. 2002) raise substantial doubts about the *Hipparcos* orbit.

5.1.4 Stochastic Solutions

Three of the sample stars have a stochastic solution designation in *Hipparcos*, implying that they had neither a satisfactory single or double-star solution. The three stars are HD 21175, 200525, and 224930, and all three also have the suspected non-single flag set in field H61. HD 21175 and 224930 have published visual and/or spectroscopic orbits (see § 5.2 and § 6.2) and are hence confirmed as true companions. For HD 200525, Goldin & Makarov (2006) present a photocentric orbit, confirming companionship (see § 7.4).

5.2 Visual Orbits

The Sixth Catalog of Orbits of Visual Binary Stars¹ (Hartkopf et al. 2001, hereafter VB6) is maintained by the United States Naval Observatory (USNO) and is a frequently updated online resource containing over 2,000 visual orbits. Table 5.4 contains 98 orbits from this

 $^{^{1}}http://ad.usno.navy.mil/wds/orb6.html$

source for the current sample, selected from the catalog as of July 8, 2008. Each orbit is listed on two lines. The first line contains the fields described below and the second line lists the corresponding uncertainties. The first column lists the WDS coordinates of the pair, and the next two columns identify the star's HD and *Hipparcos* identifiers, respectively. Some of these are the primaries listed in Table 2.1 while others are their companions with visual orbits of their own. Column 4 lists the WDS discovery designation or other catalog designations as listed in the catalog along with an identifier of the specific pair for which the orbit is listed. The next five columns list the orbital elements from the catalog. The orbital period (P) is listed in days (d) or years (y). The semi-major axis (a) is the orbital semimajor axis for all grades except 9, for which it is the photocentric-motion semi-major axis, and is listed in arcsec (a) or mill-arcsec (m). The epoch of periastron passage (T_0) is listed in modified Julian date (d) or fractional Besselian year (y). These units immediately follow their corresponding values. The next two columns list the grade of the orbit and the reference code as in the catalog. The catalog's website provides a link identifying these references, so they have not been reproduced in the bibliography of this work. Column 14 contains a code identifying whether the orbit also has a published single-lined (1) or double-lined (2)spectroscopic orbit, if it is refuted (R) by other evidence, or if it is an unconfirmed candidate (M) for the multiplicity statistics derived here. Finally, the last two columns list the number of observations and the time range of these observations, in years, from the WDS catalog as of July 8, 2008. For some pairs, the VB6 catalog lists multiple, presumably equally possible, orbital solutions. I have evaluated these and only included one per pair, selecting the more robust solution when that determination could be made, otherwise selecting the more recent one.

The VB6 contains orbits of resolved pairs based on measurements of their separation and position angle (grades 1–5 as described below), pairs resolved using LBI by measuring their interferometric visibility ("grade" 8), and photocentric orbits of unresolved pairs ("grade" 9). The description of grades 1–5 are reproduced below from the catalog's web site.

1 = Definitive. Well-distributed coverage exceeding one revolution; no revisions expected except for minor adjustments.

2 =Good. Most of a revolution, well observed, with sufficient curvature to give considerable confidence in the derived elements. No major changes in the elements likely.

3 = Reliable. At least half of the orbit defined, but the lesser coverage (in number or distribution) or data consistency leaves the possibility of larger errors than in grade 2.

4 = Preliminary. Individual elements entitled to little weight, and may be subject to substantial revisions. The quantity $3 \log(a) - 2 \log(P)$ should not be grossly erroneous. This class contains: orbits with less than half the ellipse defined; orbits with weak or inconsistent data; and orbits showing deteriorating representations of recent data.

5 = Indeterminate. The elements may not even be approximately correct. The observed arc is usually too short, with little curvature, and frequently there are large residuals associated with the computations.

The orbits listed in Table 5.4 are separated into three groups by their orbit classification. Orbits with grades 1–4 and grade 8 are listed in the first group, and represent orbital solutions of excellent (grade 1 and 8) to preliminary (grade 4) quality. Each of these pairs, with one notable exception (HD 32923, see $\S7.4$), is a reliable physical association for the multiplicity statistics derived here, supported, in most cases, by many tens to over a thousand observations measured over tens to a few hundred years, as listed in the WDS. The few exceptions with less than 10 measures are each supported by a matching spectroscopic orbits, as identified in Column 14. HD 4676 (64 Psc) does not have any measures because the orbit was derived based on interferometric visibility (Boden et al. 1999), similar to the results presented in $\S3.2$. While these observations do not directly measure the pair's separation and position angles, which is the criteria for a pair to be listed in the WDS, they are nevertheless very robust solutions, as mentioned in the online description of the orbit catalog. The two remaining grade 8 orbits (HD 16739 and 195987) have independent measurement(s) listed in the WDS, but the orbital solutions are from LBI observations (Torres et al. 2002; Bagnuolo et al. 2006). Not surprisingly, there is a strong correlation between an orbit's grade and its period. Grade 8 orbits represent the shortest-period orbits, with periods ranging from 13–57 days for orbits from visibility measurements, and 331 days for HD 16739 (12 Per), whose orbit was derived from astrometry obtained by studying spatially resolved, separated fringe packets (Bagnuolo et al. 2006). An additional orbit based on LBI visibility measurements is listed for HD 13974 in the table, but this is listed as grade 1 because Hummel et al. (1995), when presenting the orbital solution, also gave the separations and position angles derived from their visibility measurements. For grades 1–4 excluding these LBI pairs, orbital period increases with increasing grade because shorter-period systems can be more comprehensively studied with the measurements recorded over some 200 years since the days of William Herschel. Grade 1 orbits range from 1–88 years, grade 2 orbits from 5–232 years, grade 3 orbits from 76–480 years excluding one orbit of 1.2 years, and grade 4 orbits range from 20–3100 years. The one short-period orbit for grade 3 (HD 32923 with an orbital period of 1.2 years) is poorly sampled. While the WDS lists 19 observations for this similar-brightness pair separated by 0.1, the 4th interferometry catalog (see § 5.4) also lists over 20 null-results with speckle interferometry.

Orbits of grade 5 are listed in the second group because their orbital elements are extremely preliminary and subject to significant modifications. In fact, a few of these orbits may even be solutions fitted to the linear motions of unrelated stars, and hence each of these is individually evaluated to confirm a physical association. Most of the entries in this group in Table 5.4 are again supported by more than, and in many cases, much more than 10 measurements over many decades to a few centuries, and are confirmed as physical based on evidence of curved orbital motion or, in the case of very long-period systems, as linear motions consistent with a small arc of a large orbit, but definitely inconsistent with that of unrelated field stars as discussed in § 4.2. This method confirms all but five systems of this group, and three of these exceptions (HD 43587, 68256, and 161198) are confirmed by matching spectroscopic orbits. The two remaining exceptions (HD 13445 and 21175) are confirmed by published results (see § 7.4). Only five of these orbits have periods less than 200 years, and each of these is poorly sampled. The periods of the rest range from 201 years to 320 centuries. Grade 9 orbits, listed in the last group of Table 5.4, are binaries for which the photocentric motion of an unresolved point of light is modeled as an orbit. Seventeen of the 26 orbits of this type have supporting spectroscopic orbits, as indicated by a 1 (single-lined) or 2 (double-lined) in Column 14. Three, flagged with an 'R' in Column 14 were refuted using proper motion and/or radial velocity data (see § 7.4) and one, flagged with an 'M' is still a candidate (see § 7.4). The remaining five entries with a blank in Column 14 are all physical associations, as confirmed by evidence presented in the indicated references.

$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{lcccccccccccccccccccccccccccccccccccc$
$ \begin{array}{lcccccccccccccccccccccccccccccccccccc$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$\begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{lcccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
$ \begin{array}{lcccccccccccccccccccccccccccccccccccc$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
3 9.002 0.01 0.02 0.01 0
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
8 1986.6 y 0.33 309.2 3 Hei1990c 8 1979.99 y 0.43 351.0 1 Msn1996a 8 1979.99 y 0.43 351.0 1 Msn1996a 1981.0 y 0.32 311 1 Sod1999 7 1995.22 y 0.12 111.2 2 Doc2001e
8 1979.99 y 0.43 351.0 ii Msn1996a 1981.0 y 0.32 311 1 Sod1999 7 1995.22 y 0.12 111.2 2 Doc2001e
1981.0 y 0.32 311 1 solid solid
.7 1995.22 y 0.12 111.2 2 Doc2001e
9 1935.20 y 0.40 127.9 1 M_{sn1995}
7 1875.0 y 0.12 277.0 4 Hle1994
9 1875.66 y 0.52 231.7 2 Pbx2002
1909.3 y 0.51 203 2 Sod1999

TABLE 5.4: Visual Orbit Solutions

TABLE 5.4 - C	Continued															
W DS Coord	HD Name	HIP Name	DD	Ρ	a	i (deg)	رر (deg)	T_0		в	$^{\omega}_{ m (deg)}$	U	Ref	$^{\mathrm{SB}}$	N	ΔT (years)
15038 + 4739	133640	073695	STF1909	 206 y	3.8 a	84		2013	y	0.55		: 01	 Sod1999	::		
15232 + 3017	137107	075312	STF1937AB	 41.56 y	 0.87 a		203.2	1975.46	y	0.27	 38.8	÷⊓	 WSI2006b	5	1002	
15360 + 3948	139341	076382	$^{}_{STT}$ 298AB	0.01 55.6 y	0.00 0.79 a	0.1 63	0.1	0.03 1993.99	y	$0.00 \\ 0.59$	0.3 23	Ē	 Sod1999	::	515	164
16133 ± 1332	145958	079492	STF2021Aa-B	 1354 y	 5.09 a	 58.7	148.5	1754	у	0.39	111.4	.4	 Hop1964b	::	395	224
16147 + 3352	146361	029607	STF2032AB	 888.99 y	 5.93 a	 31.8	16.9	1826.95	y	0.76	72.2	.4	Sca1979	::	1041	226
16289 ± 1825	148653	080725	STF2052AB	224 y	 2.21 a			1921.1	у	0.75		: c1	Sod1999	::	528	
17153 - 2636	155885	084405	 SHJ 243AB	 470.9 y	 13.0 a	 99.8	 94.2	1677.86	y	0.92	 89.8	. 4	 Irw1996	: :	264	230
${17304-0104}$	158614	085667	STF2173	 46.40 y	 0.98 а	 99.1		1962.46	y	0.18	327.5	÷⊓	$\frac{1}{1000}$ Heil994a	5	692	176
17350+6153	160269	086036	BU 962AB	 76.1 y	 1.53 a	104	 151	 1947	у	0.18	 307	÷ε	Sod1999	: T	 129	
17465+2743	161797	086794	AC 7BC	 43.20 y	 1.36 a	 66.2		1965.40	y	0.18	174.0	5	 Cou1960b	::	343	
18055 ± 0230	165341	088601	STF2272AB	 88.38 y	 4.55 a	121.2	302.1	1895.94	y	0.50	14.0	-	 Pbx2000b	6		230
${18070+3034}$	165908	088745	AC 15AB	0.02 56.4 y	0.01 1.0 a	$^{0.1}$	$^{0.1}_{216}$	0.02 1998	y	0.00 0.75	$^{0.1}_{301}$	50	 Sod1999	: :	205	 146
${18570+3254}$		093017	 BU 648AB	 61.18 y	 1.24 a	115.1	 49.0	 1971.82	y	0.25	280.3	5	Doc2008f	÷п	 332	 127
19064 - 3704	${177474}$	093825	HJ 5084	 121.76 y	 1.90 a	149.6	50.3	2000.64	У	0.32	349.0	: c1	 Hei1986b	::	260	
19121 + 4951	179958	094336	STF2486AB	 3100 y	 12.75 a	119.1	255.2	2520	х	0.50	186.1	. 4	 Hle1994	: :	258	
19311 + 5835	184467		MCA 56	 1.35 y	 86 m	144	243	1985.27	y	0.36	356	-	 Pbx2000b	50	 28	 21
19418 + 5032	186408	096895	STFA 46Aa, B	0.00 18212 y	1.40 40.79 a	2.4 135.4	1.5 313.4	$0.00 \\ 43.64$	У	$0.01 \\ 0.86$	2.1 31.7	.4	 Mrc1999	::	534	206
19456 + 3337	186858	097222	STF2576AB	 232 y	 2.07 a	 156	 91	1945.3	y	0.77	 128	50	 Sod1999	: :	 426	
19598 - 0957	189340	098416	 HO 276	 4.90 y	0.15 a	: 9 :		1982.81	y	0.59	$\frac{142}{142}$	5	 Pbx2000b	50		
20329 ± 4154	195987	101382	 BLA 8	0.00 57.32 d	0.00 15.38 m	20 99.4	335.0	0.03 51353.81	q	$0.01 \\ 0.31$	4.3 357.4	: ∞	Trr2002	5	: 1	. 0 :
21145 + 1000	202275	104858	STT 535AB	0.00 2084.03 d	0.03 231.97 m	0.1 99.4	$0.1 \\ 23.4$	0.04 53112.07	q	$0.00 \\ 0.44$	0.3 7.7	÷⊓	 Mut2008	. 2		
21441 + 2845	206826	107310	STF2822AB	0.10 789 y	0.01 5.32 a	0.0 75.5	0.0 110.1	0.05 1958.0	v	0.00 0.66	$0.0 \\ 145.7$.4	 Hei1995	::		
:	:	:	:	:	:	:	:	:		:	:	:	:	÷	:	:
					Grad	le 5 visual	orbit solu	tions								
01158 - 6853	007788	005896	HJ 3423AB	857.0 y	5.96 a 0.00	127.1	10.3	1763.50	У	0.38	284.9 0.0	ъ	Sca2005b	÷	70	165
01398 - 5612	010361	007751	DUN 5	 483.66 y	7.82 a	142.8	13.1	1813.49	v	0.53	18.4	5	vAb1957	: :	159	177
02104 - 5049	013445	010138	ESG 1	 69.7 y	 1.69 а	150	335	1983	у	0.40	342.0	. ю	Lgr2006	: :		. ю :
02442 + 4914	016895	012777	STF 296AB	2720 y	 22.29 а		128.0	1613	у	0.13	100.6	си :	 Hop1958	: :		224
02556+2652	018143	013642	STF 326AB	: :	: :	81.5	34.5	2248.0	у	1.0	308.2	С	 Hop1967	: :		175
03128 - 0112	019994	014954	HJ 663	 1420 y	 6.77 a	114.1	84.1	1982	y	0.26	247.7		Hle1994	: :		154
03236 - 4005	021175	015799	 I 468	 111 y	 1.8 a			1982	У	0.20	 348		Sod1999	::	. °° :	
Continued on	Next Page		:	:	:	:	:	:		:	:	:	:	:	:	:

04076+3804 06173+0506 06262+1845 08122+1739 08122+1739		Name	DD	P	a	i (deg)	Ω (deg)	T_0		в	$^{\omega}$ (deg)	IJ	\mathbf{Ref}	$^{\mathrm{SB}}$	N	ΔT (years)
06173+0506 06262+1845 08122+1739	025893	019255	STT 531AB	590 y	3.87 a	104.0	160.3	2070	у	0.25	195.0	2	Hei1986b	:	171	153
06262+1845 08122+1739 08122+1739	043587	029860	CAT 1Aa	28.8 y	0.62 a	35.4 35.4	166.8	1998.05	у	0.78	75.0	си :	Cat2006	-	4	. ო :
08122+1739	045088	030630	 BU 1191	1.1 600 y	0.02 3.23 a	2.2 76.0	$1.3 \\ 139.8$	0.00 1896	y	0.25	0.3 64.7	со	Hle1994	: :	$\frac{14}{14}$	106
	068256	040167	HUT 1Ca	 17.32 y	 0.18 a	142		1984.06	y	0.08		2	 Hei1996b	÷	2	
09357+3549	 082885	047080	 HU 1128	 201 y	 3.84 a	117.0	41.3	1948.0	y		 170.0	си :	 Hei1988d	::		101
${10281 + 4847}$	090508	051248	KUI 50	 765 y	 7.08 a	72.0	 18.9	1997	у	0.53		си	 Hle1994	::		
11268 ± 0301	099491	055846	STF1540AB	 32000 y	 40.76 a	126.6		5200	y	0.46		20	 Hop1960a	::	$\frac{120}{120}$	227
11387 + 4507	101177	056809	STF1561AB	 2050 y	 10.33 a	125.9	271.2	2500	y	0.05	129.6	20	 Hle1994	::	$\frac{110}{110}$	224
${13473+1727}$	120136		$^{}_{STT}$ 270	 2000 y	 14.39 a	50.7		2017	У	0.91			 Hle1994	::		
$17191 - 4638$	156274	084720	BSO 13AB	 693.24 y	 10.42 a		 131.8		у	0.78	 333.4	. ro	 Wie1957	::	 127	
${17419+7209}$	162003	086614	STF2241AB	12500 y	 55.2 a		. 1	 942	y	$0.00 \\ 0.54$	0.0 312	20	$\frac{\dots}{\operatorname{Rmn1994}}$::	$\frac{141}{141}$	207
${17433+2137}$	161198	086722	 DUQ 1	 2558.4 d	 0.17 a		309.2	49422.53	q	0.94	129.6	си :	 Duq1996	÷	. 9 :	. ი :
21198 - 2621	202940	105312	BU 271AB	10.4 261.62 y	0.01 3.01 a	3.7 74.1	1.5 59.1	$0.10 \\ 1845.99$	y	$0.00 \\ 0.42$	0.6 57.6	; n	Jas 1997	::	 49	
23524 + 7533	223778	 117712	BU 996AB	 290.0 y	 4.14 a	 49.6	 93.9	2015.0	Y	0.55	134.1	: vo	 Hle1994	: :		
•	:	:	•	•	:	:	:	:		÷	:	÷	:	:	:	:
					Photoce	atric-moti	on orbits (grade 9)								
00169 - 5239	001273	001349	GJ 13	411.4 d	20.98 m	74.7	352.3	34233.3	q	0.57	4.7	6	Jnc2005	1	÷	÷
01425 + 2016	010476	007981	107 Psc	207.27 d	4.55 m	2.1	3.3 12.7	48213.79	q	0.00	0.0	6	 HIP1997d	E H	: :	: :
02180 ± 0145	014214	010723	GC 2770	6.18 93.29 d	2.10 6.07 m	23.0 110.0	$^{23.8}_{172}$	11.96 53338.33	p	0.52	104.0	6	Fek2007	- -	: :	::
02225 - 2349	014802	011072	κ For	0.00 26.5 y	0.38 0.54 a	8.4 56	$14 \\ 133$	0.01 1983.6	у	$0.00 \\ 0.2$	$^{0.1}_{22}$	6	Gon 2000	::	: :	: :
02361 + 0653	016160	012114	PLQ 32Aa,P	 61 y	 0.17 a			1940	y	0.67	249	6	 Hei1994b	: :	. 9 :	. ი :
${02481+2704}$	017382	013081	 GC 3359	 15.5 y	 0.04 a	 84	54	1986.5	y	0.00	:0	. 6	 Hei1990d	÷	: :	: :
05067 ± 1427	032850	023786	GJ 3330	204.38 d	10.14 m	29.4	153.0	48444.84	p	0.14	293.1	. 6	 HIP1997d	÷	::	::
05544 + 2017	039587	027913	χ^{1} Ori	2.44 5156.29 d	1.29 89.66 m	95.9	126.4	41.29 51468.2	q	0.24 0.45	111.5	6	HaI2002	: -	: :	: :
${11182+3132}$	098231	055203	ξ UMa Aa	2.51 1.83 y	0.88 0.05 a	0.8 91	0.6 318	3.08 1992.29	у	$0.00 \\ 0.61$	0.2 324	6	Heil996b	Ē	: :	: :
${12442+5146}$	110833	062145	GC 17326	 270.21 d	 7.19 m	48.9	154.9	$\frac{1}{48401.43}$	р	0.00 0.00	0.0	6	 HIP1997d	- -	: :	: :
12595 ± 4159	${112914}$	063406	13738 LTT 13738	3.31 710.6 d	0.71 14.12 m	10.2 81.8	13.0 101.3	8.95 49220.0	q	0.33		. 6	Jnc2005	-	::	::
13038 - 0510	113449	063742	 GC 17714	231.23 d	0.47 11.21 m	5.0 126.5	4.9	48369.75	q	0.51	42.0	6	 HIP1997d	::	::	::
13546 + 1825	121370	067927	$\eta \operatorname{Boo}$	1.96 494.2 d	1.57 $36.02 m$	10.8 115.7	14.5 75.2	10.27 28136.2	φ	$0.21 \\ 0.26$	21.8 326.3	6	Jnc2005	: T	: :	: :
14035 ± 1047	122742	068682	GJ 538	 3614.89 d	0.56 105.28 m	$1.6 \\ 93.5$	1.4 252.3	33998.20	q	0.55	189.0	6	 HIP1997d	÷	: :	::
14160 - 0600	124850	 069701	ιVir	 55 y	2.02 0.87 a		: ო	1950.7	х	0.1		6	Gon 2003	M	: :	: :
14294 + 8049	128642	070857	GC 19630	 179.73 d	 13.07 m	54.8	 80.4	48361.86	р	0.15	158.6	: 6 :	 HIP1997d	÷	: :	: :

TABLE $5.4 - C$	Jontinued															
WDS Coord	HD Name	HIP Name	WDS DD	Ρ	a	$_{i}^{i}$ (deg)	Ω (deg)	T_0		e	$^{\omega}_{(\mathrm{deg})}$	ŭ	Ref	\mathbf{SB}	Ν	ΔT (years)
:	:	:		0.76	0.78	5.2	6.3	35.04		0.12	69.5	:		:	:	:
14534 ± 1909	131511	072848	DE Boo	125.4 d	16.54 m	93.4	248.3	50203.4	q	0.51	219	6	Jnc2005	1	÷	:
:	:	:	:	:	0.18	4.2	3.6	:		:	:	:	:	:	÷	:
15282 - 0921	137763	075718	BAG 25Aa,Ab	889.6 d	38.6 m	60.3	95.8	47967.5	q	0.97	253.9	6	Jnc2005	0	1	0
:	:	:		:	1.0	1.8	3.5	:		:	:	:	:	:	:	:
15527 + 4227	142373	077760	χ Her	51.29 d	0.96 m	131.7	51.7	48349.00	q	0.00	0.0	6	HIP1997d	Ч	:	:
:	:	:		0.41	0.58	27.6	38.0	4.44		:	:	:	:	:	:	:
16010 + 3318	143761	078459	$\rho \text{ CrB}$	0.11 y	1.66 m	0.5	30.5	1997.57	y	0.13	30	6	Gat2001a	:	:	
	:	:			0.35		12.3	:		:	:	:	::	:	:	
16147 + 3352	146361	079607	σ CrB C	52 y	0.11 a	59	30	1963.0	У	0.36	127	6	Hei1990d	:	:	:
:	:	:		:	:	:	:	:		:	:	:	:	:	:	:
16285 - 7005	147584	080686	ζTrA	12.9 d	2.71 m	16.0	2.1	18103.6	q	0.06	274.5	6	Jnc2005	-	:	:
:	:	:		:	0.44	2.5	6.8	:		:	:	:	:	:	:	:
17393 ± 0333	160346	086400	GJ 688	83.70 d	12.71 m	18.4	274.2	47724.9	q	0.23	140.5	6	Jnc2005	1	:	:
:	:	:		:	0.78	1.1	2.7	:		:	:	:	:	:	:	:
17465 ± 2743	161797	086974	TRN $2Aa$	65 y	0.27 a	68	81.8	1951.0	У	0.32	92	6	Hei1994a	:	2	0
:	:	:		:	:	:	:	:		:	:	:	:	:	:	:
21094 - 7310	200525	104440	I 379A	2145 d	68 m	90	13	49422	q	0.70	148	6	Gln2006	:	4	34
:	:	:		1975	54	7	92	1513		0.09	84	:	::	:	:	
21247 - 6814	203244	105712	GC 29928	1060.61 d	31.78 m	110.3	70.7	48167.89	q	0.29	11.9	6	HIP1997d	ч	:	:
:	:	:		54.98	2.28	2.5	2.2	28.32		0.04	13.2	:	:	:	:	:

5.3 The Washington Double Star Catalog

The WDS is a frequently updated online² catalog maintained by the USNO. It contains measurements of double stars spanning the entire history of their observations. The online catalog contains summary information for each pair giving the earliest and latest separation and position angle measures along with their epochs, and the full catalog contains detailed information on each published measure. While the catalog contains measures of unrelated field stars as discussed in § 4.2, it is a very useful source in identifying and examining candidate companions. Cross-referencing the catalog as of July 8, 2008 with the stars of this work, I extracted measures for 503 pairs for 210 distinct primaries. In § 4.2, I described the method of blinking archival images, which identified 302 of these pairs as optical, and hence not gravitational bound. This section analyzes the remaining 201 pairs and utilizes the WDS to identify, confirm, or refute the candidate companions.

Seventy eight of the WDS pairs have supporting spectroscopic or visual orbits, or both, and these are included in the discussions in §5.1.3 or §6.2. An additional 35 pairs were confirmed by the matching proper motions and photometric distance estimates of the companions to the corresponding *Hipparcos* values of their primaries. Twenty two of these were also independently identified by blinking the archival images and are hence included in Table 4.2, and the remaining 13 are listed in Table 5.5. The first column identifies the primary star and the next three columns identify the companion by the separation, position angle, and epoch of the most recent observation listed in the WDS. Columns 5 and 6 list the

 $^{2}http://ad.usno.navy.mil/wds/$

number of measurements of the pair listed in the WDS and the number of years they span. The next three columns list the proper motion of the companion, if available, along with its reference. This is followed by VRI photometry from the literature if available, again with the reference identified, and then the JHK_S magnitudes from 2MASS. The last two columns list the photometric distance estimate indicated by these magnitudes and its corresponding uncertainty. As in §4.1, the distance estimates were derived by fitting various colors to the M_{K_S} -color relations from Henry et al. (2004).

ΗD	\mathbf{Sep}	\mathbf{PA}	Epoch		$\Delta \mathbf{T}$	μ_{α}	μ_{δ}			= CCD	Magni	tudes =		Infrar	ed Magn	itudes	р	Error
\mathbf{Name}	(arcsec)	(deg)	(year)	z	(years)	$(mas \ yr^{-1})$	$(mas \ yr^{-1})$	R	V 1	R	R R	Ι	R	Л	Η	K	(bc)	(bc)
004391	16.6	307	1993	3	98				12.7 5			:		8.44	7.95	7.64	12.0	1.9
024409	8.9	228	1999	ς,	92	:	:		12.9 2	•	:	:		9.22	8.66	8.50	26.3	4.1
036705	9.2	345	1998	co	69	:			13.0 2		:	:		8.17	7.66	7.34	7.2^{a}	1.1
039855	10.6	20	2003	12	127	:	:		:	•	:	9.13	ŝ	7.99	7.40	7.22	19.0	2.9
040397	89.3	313	2000	0	40	78	-216	e	:	1	5.15 4	13.28	ę	11.11	10.64	10.31	19.0	3.6
089125	7.7	299	2005	30	154	:			11.4 2		:	:		8.36	7.79	7.59	24.9	3.9
136202	11.4	35	2000	45	175	376	-517	1	10.2 2	•	:	:		7.49	6.91	6.75	20.6	3.5
149806	6.3	19	2000	e	54	:			13.0 2	•	:	:		8.09	7.57	7.28	6.7^a	1.0
161797	34.9	248	2007	108	226	- 333	-753	1	10.4 2		:	:		6.52^{b}	5.92^{b}	5.70^{b}	6.2	0.9
167425	7.8	352	2000	9	103	:	:		10.8 2	•	:	:		8.00	7.42	7.19	23.7	3.7
186858	26.0	68	2006	117	184	19	-447	1	9.2 2	•	:	:		6.64	6.12	6.00	15.9	3.3
191499	4.0	14	2003	51	221	:	:		9.4 2	•	:	:		6.21	5.88	5.95	12.9^{a}	3.5
214953	7.5	125	1998	15	104	:	:		10.3 2		:	:		7.44	6.86	6.63	17.8	2.8

TABLE 5.5: WDS companions confirmed by photometry

NOTES.—Reference codes for columns 9, 11, 13, 15: 1 = The LSPM North catalog (Lépine & Shara 2005), 2 = Visual Double Stars in Hipparcos (Dommanget & Nys 2000), 3 = The USNO

B 1.0 Catalog (Monet et al. 2003), 4 = The NOMAD Catalog (Zacharias et al. 2004), 5 = Differential photometry with CTIO 0.9m V-band image

^b The companion is a visual binary. The *Hipparcos* input catalog lists individual V magnitudes of 10.2 and 10.7 for the components. 2MASS lists combined magnitudes ^a See § 7.4.

for the two stars. The 2MASS magnitudes have been accordingly adjusted for one component.

Twenty seven pairs from the WDS contain bright companions with independent measures of parallaxes and proper motions that were compared with the primaries' *Hipparcos* measures to confirm a physical association and are accordingly listed in Table 4.1. Twenty additional pairs were confirmed based on proximity to the primary and matching proper motions. While an independent confirmation of distance could not be obtained for these pairs, mainly because obtaining reliable photometry is difficult for tight binaries, the proximity and CPM imply a physical association. Two of these 20 were also identified by blinking archival images as twin diffraction spikes (HIP 91605) or overlapping PSF (HD 125455), and were discussed in $\S4.1$. The remaining 18 could not be seen in the archival images because the companion was lost in the saturation around the primary due to the proximity and magnitude difference. All of these companions are within 23'' of the primary, and all but four are within 7''. Each companion has a matching proper motion with its primary in the range of $0''_{1}$ yr⁻¹ to $1^{\prime\prime}_{..}6 \text{ yr}^{-1}$. Further details for each of these stars, as well as 13 additional pairs which were confirmed based on published evidence, are included in §7.4. Finally, 10 pairs listed in the WDS were confirmed as physical associations because the measures clearly demonstrate not only CPM, but also orbital motion, although they lack sufficient observations to derive an orbit. Table 5.6 lists these 10 pairs with their summary information from the WDS.

motion
orbital
demonstrating
pairs
WDS
5.6:
TABLE

		Epoch N	Epoch N 2006 115	Epoch N 2006 115 1991 7	Epoch N 2006 115 1991 7 1986 17	Epoch N 2006 115 1991 7 1986 17 1974 15	Epoch N 2006 115 1991 7 1986 17 1974 15 1987 22	Epoch N 2006 115 1991 7 1986 17 1974 15 1987 22 1999 29	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	EpochN 2006 115 1991 7 1986 17 1974 15 1987 22 1999 29 1988 20 2007 23	EpochN 2006 115 1991 7 1994 15 1974 15 1987 22 1999 29 1998 20 1935 5
= Last ====	PA (deg) F		306	306 248	306 248 39	$306 \\ 248 \\ 39 \\ 47$	306 248 39 47 126	$306 \\ 248 \\ 39 \\ 47 \\ 126 \\ 177$	306 248 39 47 177 39 39	306 248 39 47 126 39 39	306 248 39 47 177 39 8 8
$h \operatorname{Sep}('')$		3.3		1.5	0.1	1.5 0.1 4.4	1.5 0.1 5.3	1.5 0.1 5.3 3.7	1.5 0.1 5.3 3.7 5.0	$\begin{array}{c} 1.5 \\ 0.1 \\ 5.3 \\ 3.7 \\ 11.1 \end{array}$	1.5 0.1 5.3 3.7 3.4 3.4 3.4
g) Epoch 1829	1829		1929		1904	1904 1910	$1904 \\ 1910 \\ 1834$	1904 1910 1834 1827	1904 1910 1834 1827 1894	$1904 \\1910 \\1834 \\1827 \\1894 \\1828 \\1828 \\$	$1904 \\ 1910 \\ 1834 \\ 1827 \\ 1894 \\ 1828 \\ 1910 \\ 1910 \\ $
') PA (deg) 335	335		64	226		103	$103 \\ 119$	103 119 145	103 119 145 10	$103 \\ 119 \\ 145 \\ 10 \\ 123 $	103 119 145 10 123 11
) Sep (")] 4.6 1.0	4.6 1.0	1.0		0.1	J C	0.0	3.0 15.0	$ \frac{5.0}{5.2} $	$ \begin{array}{c} 5.0 \\ 5.2 \\ 2.5 \end{array} $	$ \begin{array}{c} 5.0 \\ 5.2 \\ 15.0 \\ 15.0 \\ 15.0 \\ \end{array} $	$ \begin{array}{c} 15.0\\ 15.0\\ 15.0\\ 2.5\\ 2.8\\ 2.8\\ \end{array} $
Pair ID	:		:	:	A D	AD	GA :	AB	AB : AB : .	AB : AB : AB	AB : AB : AB AB : AB
DD		STF 295	DON 680	A 691		A 2230	A 2230 HJ 5173	A 2230 HJ 5173 STF2752	A 2230 HJ 5173 STF2752 HDO 298	A 2230 HJ 5173 STF2752 HDO 295 HJ 301	A 2230 HJ 5173 STF2752 HDO 298 HJ 301 A 2124
	Coord	02412 - 0042	14494 - 6714	15138 - 0121	15440 - 0031		20112 - 3606	20112 - 3606 21072 - 1355	20112 - 3606 20112 - 3606 21072 - 1355 22183 - 5338	20112-3606 20112-3606 21072-1355 22183-5338 22467+1210	$\begin{array}{c} 10440 \pm 0.021 \\ 20112 - 3606 \\ 21072 - 1355 \\ 22183 - 5338 \\ 22167 \pm 1210 \\ 07291 + 3147 \end{array}$
	Name	HD 016765	HD 130042	HD 135204	HD 140538		HD 191408	HD 191408 HD 200968	HD 191408 HD 200968 HD 211415	HD 191408 HD 200968 HD 211415 HD 215648	HD 191408 HD 200968 HD 211415 HD 215648 HIP 036357

In addition to the 302 WDS pairs discussed in §4.2 that were refuted as physical associations by demonstrating that the companions are field stars, 11 additional pairs were refuted based on a closer inspection. Two of these (HD 9540 and HD 26923 CD) had photometric distance estimates that were significantly different from the primary's *Hipparcos* distance and are included in Table 4.2. The other nine pairs (HD 20807 Aa, HD 22049 Aa-Ab, HD 64606 AB, HD 109358 Aa, HD 128620 Ca, HD 145958 Aa, HD 147776 AE, HD 178428Aa, and HD 186408 BC) were refuted based on published evidence and/or observations and are detailed in §7.4. Finally, seven WDS pairs do not have sufficient evidence to be confirmed or refuted, and hence remain candidates. These are HD 4628 Aa, HD 45270, HD 100180 Aa, HD 111312 Aa-B, HD 147776 AC, HD 165908 Aa-Ab, and HD 217107, individual notes for which are included in §7.4. The candidate companion for HD 217107 may in fact be the "exoplanet" reported by radial velocity surveys, but in order to be resolved by speckle, must be at least as massive as an early M-dwarf.

5.4 The Fourth Interferometric Catalog

While the WDS contains all measurements of pairs with separation and position angle measurements, the Fourth Catalog of Interferometric Measurements of Binary Stars³ (FIC), maintained by the USNO as an online database, contains the results of observations using high-resolution techniques such as speckle interferometry, CCD imaging, adaptive optics, and long-baseline interferometry. This online catalog lists the details of every observation of

 $^{^{3}}http://ad.usno.navy.mil/wds/int4.html$

a pair including information such as the epoch of observation, the separation and position angle measured when available, and the reference. A feature of this catalog is that it also lists null results identifying the detection limits of the technique used. This source was very useful in deciding the status of many pairs, especially based on the null results published. Rather than list the information for individual pairs in this section, I mention this source in discussing the status of pairs when they are being evaluated, and the bulk of these references are in the notes to individual systems (§ 7.4).

5.5 The Catalog of Nearby Stars

The Catalog of Nearby Stars (Gliese 1969; Gliese & Jahreiß 1979, 1991, hereafter CNS) contains valuable information on stars within 25 pc. While it is dated and its astrometry has been superseded by *Hipparcos*, the information it contains about suspected or confirmed stellar companions was evaluated and included in this effort. Wide pairs are identified with measurement details and often confirmed via CPM. These pairs are typically also found in other sources such as the WDS, although a few really wide pairs (separations greater than some 10') are listed only in the CNS. The catalog also identifies several stars as "SB", "SB?", "RV-Var", or "RV-Var?", but additional supporting information is often not included. Each of these was evaluated based on independent information that would allow their confirmation or rejection as physically bound.

For the targets studied here, the CNS indicates 198 possible companions. Most of these entries are also listed in one or more of the other sources checked and a majority of the companions listed solely in the CNS have been found to be spurious. However, two wide companions (15' from HD 63077 and 20' from 137763) were identified solely based on their CNS entries, both of which were subsequently verified by blinking large enough images and confirmed by matching proper motions and parallaxes. Ninety one companions listed in the CNS have supporting spectroscopic and/or visual orbits, 51 were confirmed based on CPM and matching trigonometric or photometric parallaxes, and 22 were confirmed based on evidence of orbital motion or CPM and proximity. Twenty three of the CNS companions were refuted based on results of modern radial velocity surveys that negate claims of velocity variation or spectroscopic companions, and an additional eight were refuted by other methods utilized here. Finally, three companions (HD 20010, 23484, and 90839) have "RV-Var" or "RV-Var?" designations in the CNS for which I could not find any evidence to confirm or refute, and hence are retained as candidates. In summary, this catalog was found to be very reliable for CPM companions, with only six (4%) of the 148 CPM companions refuted, but less so for pairs identified as spectroscopic binaries, 23 (46%) of which were refuted.

Science is spectral analysis. Art is light synthesis. — Karl Kraus

Spectroscopic and Photometric Results

Spectroscopic techniques have been very useful in discovering and characterizing many stellar and most planetary companions. Photometric techniques have also been productive in detecting close pairs and estimating the components' physical parameters. This chapter presents the results from these techniques. With all my observing programs focused on astrometric techniques, I leveraged catalogs, publications, and collaborations to include the results from these methods.

To address spectroscopic companions, I first extracted information on known pairs from the 9th Catalogue of Spectroscopic Binary Orbits¹ (SB9), the exoplanet catalogs discussed in § 6.3, and publications. However, in order to present a credible update to the DM91 results, I needed to include the results of a more comprehensive effort. I have been able to achieve this goal with tremendous support from David W. Latham at the Harvard-Smithsonian Center for Astrophysics (CfA), Geoffrey W. Marcy at the University of California, Berkeley, Artie P. Hatzes at Thüringer Landessternwarte in Tautenburg, Germany, and William D. Cochran at the McDonald Observatory. Dr. Hatzes sent me the measured radial velocities on 51 stars of this work that were observed as part of his planet search program so that I could look for signatures of possible new stellar companions. Dr. Cochran looked at his planet search database and sent comments on 13 of my stars, confirming that they did not contain evidence for new stellar companions. Information from these surveys is presented in §6.1.

 $^{^{1}}http://sb9.astro.ulb.ac.be/$

Dr. Marcy sent me radial velocity measurements on 255 overlapping stars, obtained as part of planet-search efforts of the California and Carnegie Planet Search (CCPS) program, for the purpose of statistical analyses of companions. The results from these data are discussed in the statistical and incompleteness analyses in Chapter 8. I am grateful to each of these collaborators for enabling a much more comprehensive multiplicity survey. I am particularly indebted to Dr. Latham for sharing with me the spectroscopic results on over 300 overlapping stars based on over 20 years of radial-velocity measurements at the CfA for various surveys (Carney & Latham 1987; Udry et al. 1998). Sections 6.1 & 6.2 present these results for constant velocity stars as well as spectroscopic binaries.

This chapter also includes results from photometric techniques in \S 6.4. While this method has been very useful in identifying and characterizing eclipsing stellar and sub-stellar components, most photometric discoveries are for distant systems with very short periods, and the overlap with this study is small.

6.1 Constant Radial Velocity Stars

Table 6.1 lists 269 stars that do not show evidence of radial velocity variations. The results are primarily from the CfA surveys but also include information from the high-precision surveys of A. Hatzes, W. Cochran, and Nidever et al. (2002). Stars with zero or one CfA observation have been excluded unless they have evidence of stable radial velocities from the high-precision measures, resulting in the inclusion of five with no observations and 11 with one observation. All but 23 of the stars in the table have at least four CfA observations ob-

tained over at least five years, or have stable velocities from the high-precision searches. The first column identifies the star by its HD or HIP identification and a component designation when required. Columns 2–7 pertain to the CfA observations. Column 2 lists the number of velocity measurements and Column 3 the number of years (y) or days (d) spanned by these measurements. The next column lists the mean radial velocity in $\mathrm{km\,s^{-1}}$, followed by the external error, a measure of the rms scatter around the mean. Column 6 lists the $P(\chi^2)$ value of these measures, indicating the probability that the scatter around the mean is solely due to measurement errors. Column 7 lists the ratio of external to internal error, serving as another diagnostic for variability. All stars listed in this table are considered as constant velocity stars for this work. Column 8 lists a "C" if this star was identified as a constant velocity star in the Hatzes or Cochran surveys, "F" if the survey had too-few observations to tell, "L" if the data showed a linear trend, and "V" if the data showed variations. These surveys typically have a few dozen observations for each star, measured over a few (1–4) years. The last column lists these same flags from Nidever et al. (2002), where a "C" designation is assigned to stars listed in Table 1, i.e. those with a velocity scatter of less than 0.1 km s^{-1} and "V" to stars listed in Table 2, i.e. stars with a scatter greater than 0.1 km s⁻¹. Any "L" or "V" values are individually footnoted and these variations relate to known planet(s) or to visual companions for which the high-precision data show a slow linear drift. Note that some of these stars may be planet hosts, and in that sense show periodic variations in their velocities, but for the purposes of this section, we are concerned about stellar companions and include planet hosts that do not show any evidence of longterm trends.

 TABLE
 6.1: Stars with no Evidence of Radial Velocity Variations

Nama	λ	ΛT	$\langle \mathbf{RV} \rangle$	Ext Err $(1 \text{ mm } a^{-1})$	$D(x^2)$	F/I	пс	N: 409
Iname	11	ΔI	(kms)	(kms)	$P(\chi)$	E /1	п,С	INIGUZ
HD 000166	75	15.05 y	-6.71	0.46	0.00310	1.2	\mathbf{C}	С
HD 001461	49	13.08 y	-10.15	0.58	0.00001	1.4	\mathbf{C}	С
HD 001562	10	11.29 y	15.06	0.21	0.95538	0.5		
HD 001835	51	$15.05 { m y}$	-2.51	0.40	0.55195	0.9		\mathbf{C}
HD 002025								\mathbf{C}
HD 003651	23	$25.45 { m y}$	-33.13	0.77	0.00000	2.3		\mathbf{C}
HD 003765	180	$26.78 { m y}$	-63.37	0.53	0.00000	1.4		\mathbf{C}
HD 004256	4	2.11 y	9.39	0.28	0.75400	0.6		\mathbf{C}
HD 004614A	4	6.96 y	7.61	0.35	0.30936	1.0		\mathbf{C}
HD 004614B	89	18.74 y	10.71	0.55	0.00254	1.2		\mathbf{C}
HD 004628	23	$12.41 { m y}$	-10.08	0.56	0.00337	1.5		\mathbf{C}
HD 004635	18	8.46 y	-31.83	0.44	0.17025	1.2		
HD 004813	4	11.16 y	7.98	0.48	0.04838	1.7		
HD 004915	10	8.36 y	-3.77	0.26	0.92060	0.6		\mathbf{C}
HD 005015	4	7.90 y	20.23	0.40	0.24600	1.1		
HD 005133								\mathbf{C}
HD 007590	10	$11.25 { m y}$	-13.30	0.48	0.43457	1.0	\mathbf{C}	С
HD 007924	4	8.49 y	-22.56	0.62	0.05050	1.7		
HD 009407	12	8.50 y	-33.26	0.30	0.67109	0.9	\mathbf{F}	С
HD 009540	3	12.04 y	2.72	0.16	0.91613	0.3		
HD 009826	14	21.20 y	-28.64	0.87	0.00107	2.0		С
HD 010008	7	$5.99 { m y}$	11.42	0.77	0.00188	1.7		
HD 010086	3	1.08 y	1.76	0.34	0.35239	0.9		\mathbf{C}
HD 010476	20	$25.45 { m y}$	-33.67	0.66	0.00000	1.9		С
HD 010700	29	18.32 y	-16.66	0.50	0.01893	1.3		\mathbf{C}
HD 010780	72	22.40 y	2.73	0.49	0.00000	1.4	\mathbf{C}	\mathbf{C}
HD 012051	11	8.33 y	-35.08	0.49	0.41728	1.0		\mathbf{C}
HD 012846	11	11.37 y	-4.53	0.56	0.40119	1.2		\mathbf{C}
HD 014412	4	3.94 y	7.34	0.40	0.19585	1.1		С
HD 016160	19	$12.61 { m y}$	26.03	0.42	0.07031	1.2		С
HD 016895A	37	$17.47 { m y}$	24.12	0.36	0.37098	1.0		С
HD 016895B	34	$14.25 { m y}$	25.81	0.93	0.33501	0.7		С
HD 017925	41	10.36 y	18.04	0.43	0.57961	1.0	\mathbf{C}	С
HD 018143	11	8.34 y	31.96	0.56	0.12744	1.3		С

Continued on Next Page...

			$\langle \mathbf{RV} \rangle$	$\mathbf{Ext} \ \mathbf{Err}$				
Name	N	ΔT	(kms^{-1})	$(\mathrm{km}\mathrm{s}^{-1})$	$P(\chi^2)$	E/I	$_{\mathrm{H,C}}$	Nid02
HD 018632	8	17.45 y	28.79	0.31	0.79244	0.7		С
HD 018757	7	11.40 y	-2.55	0.24	0.87338	0.6		
HD 018803	6	$1.59 { m y}$	9.82	0.38	0.30351	1.1		С
HD 019373	29	13.16 y	49.42	0.62	0.00000	1.7		С
HD 019994	3	3.61 y	19.47	0.44	0.26507	1.2		С
HD 020165	10	14.49 y	-16.82	0.39	0.79352	0.7		\mathbf{C}
HD 020619	3	$3.85 { m y}$	23.03	0.75	0.03899	1.7		С
HD 020630	79	13.03 y	19.00	0.46	0.00049	1.3	\mathbf{C}	\mathbf{C}
HD 022049	32	$17.14 { m y}$	16.33	0.57	0.00017	1.5		С
HD 022484	183	23.44 y	28.01	0.40	0.00031	1.1		\mathbf{C}
HD 022879	70	16.85 y	120.15	0.55	0.17919	1.0		С
HD 023356	1		25.24		1.00000			\mathbf{C}
HD 024238	10	8.42 y	38.49	0.43	0.72252	0.9		С
HD 024496	8	11.99 y	18.65	0.30	0.78679	0.7		С
HD 025329	35	14.08 y	-25.76	0.68	0.78191	0.9		\mathbf{V}^{a}
$HD \ 025665$								\mathbf{C}
HD 025680	9	10.93 y	23.98	0.35	0.52202	1.0		С
HD 025998	12	$1.87 { m y}$	25.35	0.42	0.94963	0.7		
HD 026913	25	16.23 y	-8.02	0.50	0.02160	1.2	\mathbf{C}	
HD 026923	9	11.96 y	-7.28	0.21	0.95513	0.5	\mathbf{C}	
HD 026965	49	17.90 y	-42.24	0.51	0.00011	1.4		С
HD 029883	4	2.12 y	17.20	0.63	0.06590	1.6		С
HD 030495	4	2.41 y	21.70	0.29	0.54368	0.8		
HD 030876	3	7.11 y	20.59	0.34	0.37180	1.0		
HD 032923	94	16.96 y	20.49	0.52	0.00000	1.4		С
HD 034411	4	11.15 y	66.23	0.23	0.64464	0.7	\mathbf{C}	\mathbf{C}
HD 034721	4	$3.82 { m y}$	40.34	0.20	0.79302	0.6		С
HD 035112A	10	$13.25 { m y}$	36.31	0.63	0.05785	1.4		
HD 035296	61	13.03 y	37.59	0.53	0.03684	1.1	\mathbf{C}	
HD 037008	3	$1.94 { m y}$	-45.88	0.23	0.74151	0.6		
HD 037394	20	11.91 y	1.11	0.38	0.85286	0.8	\mathbf{C}	С
HD 038230	10	11.87 y	-29.53	0.44	0.60342	1.0	\mathbf{C}	С
HD 038858	5	2.96 y	31.25	0.49	0.09018	1.3		С
HD 039855	5	$5.01 { m y}$	42.58	0.71	0.07564	1.3		
HD 040397AB	6	7.06 y	143.48	0.31	0.85357	0.7	\mathbf{C}	С
HD 041593	14	$10.47 { m y}$	-9.83	0.40	0.67405	0.9	\mathbf{C}	
HD 042618	10	12.39 y	-53.95	0.58	0.44411	1.1		С

TABLE 6.1 - Continued

_

			$\langle {f RV} angle$	Ext Err				
Name	N	ΔT	(kms^{-1})	$(\mathrm{km}\mathrm{s}^{-1})$	$P(\chi^2)$	E/I	$_{\mathrm{H,C}}$	Nid02
HD 042807	14	6.86 y	5.86	0.31	0.83090	0.8		
HD 043162	7	7.23 y	21.86	0.23	0.97369	0.5		
HD 045184	3	3.13 y	-4.27	0.45	0.09803	1.5		С
HD 046588	2	16.65 y	15.21	0.03	0.97903	0.1		
HD 048682	5	2.81 y	-23.83	0.68	0.01704	1.8	\mathbf{C}	\mathbf{C}
HD 050692	7	11.05 y	-15.11	0.37	0.36320	1.0		\mathbf{C}
HD 051419	9	12.44 y	-27.16	0.46	0.19319	1.1	\mathbf{C}	
HD 051866	4	2.83 y	-21.75	0.27	0.77038	0.6		\mathbf{C}
HD 052711	11	4.26 y	24.43	0.24	0.99114	0.6		\mathbf{C}
HD 055575	9	14.90 y	84.29	0.27	0.77500	0.7		С
HD 059747	5	3.25 y	-15.97	0.29	0.70835	0.7		\mathbf{C}
HD 059967	6	5.89 y	9.15	0.65	0.00303	1.7		
HD 060491	4	1.96 y	-9.96	0.35	0.74222	0.5		С
HD 061606	4	11.70 y	-18.23	0.65	0.07653	1.5		\mathbf{C}
HD 063433	9	11.35 y	-16.05	0.44	0.48748	0.9	\mathbf{C}	С
$HD \ 065583$	328	26.80 y	14.71	0.58	0.00000	1.3		\mathbf{C}
HD 067228	3	2.18 y	-36.27	0.33	0.31834	1.0		\mathbf{C}
HD 068017	12	11.96 y	29.68	0.45	0.37359	1.0		С
HD 068255	23	11.14 y	-6.50	0.90	0.00000	2.8		
HD 068257	21	11.14 y	-4.88	1.05	0.00000	3.1		
HD 069830	4	$3.72 { m y}$	30.04	0.61	0.14309	1.5		
HD 071148	7	$2.41 { m y}$	-32.39	0.46	0.19645	1.3		\mathbf{C}
HD 072673	3	$2.98 { m y}$	14.51	0.38	0.31705	1.1		С
HD 072760	8	9.20 y	35.05	0.24	0.96311	0.5		С
HD 072905	40	18.02 y	-12.86	0.67	0.00000	1.9	\mathbf{C}	
HD 073350	4	14.81 y	35.52	0.55	0.02966	1.4		С
HD 073667	13	$15.02 \mathrm{~y}$	-12.11	0.64	0.00346	1.4		С
HD 074576	2	$5.91 { m y}$	12.26	0.29	0.40422	0.8		
HD 075732	6	3.98 y	27.52	0.45	0.19116	1.1		\mathbf{C}
HD 076151	5	7.09 y	32.26	0.31	0.66501	0.8		\mathbf{C}
HD 076932	6	$11.67 { m y}$	119.32	0.55	0.51166	1.0		
HD 078366	3	$90.00~\mathrm{d}$	25.84	0.63	0.15253	1.3		С
HD 079969	6	16.76 y	-20.41	0.26	0.85587	0.7		
HD 082443	77	14.44 y	8.31	0.45	0.27239	1.0		
HD 082558							\mathbf{C}^{b}	
HD 082885	9	16.18 y	14.47	0.29	0.72885	0.8		
HD 084737	1		4.62		1.00000			С

TABLE 6.1 - Continued

			$\langle {f RV} angle$	Ext Err				
Name	N	ΔT	(kms^{-1})	$(\mathrm{km}\mathrm{s}^{-1})$	$P(\chi^2)$	E/I	$_{\mathrm{H,C}}$	Nid02
HD 086728	7	6.03 y	56.10	0.35	0.29604	1.0		С
HD 087424	1		-12.22		1.00000			\mathbf{C}
HD 087883	5	3.06 y	9.44	0.31	0.70012	0.7		\mathbf{C}
HD $089125A$	7	$5.91 { m y}$	36.79	0.30	0.58788	0.8		
HD 089269	15	$16.34 { m y}$	-7.62	0.34	0.73340	0.9	С	С
HD 090156	5	$5.03 \mathrm{y}$	26.50	0.45	0.09886	1.4		\mathbf{C}
HD 090508	7	2.40 y	-7.64	0.22	0.90382	0.6		
HD 090839A	44	13.24 y	8.40	0.47	0.00016	1.4		\mathbf{C}
HD 091889	16	$13.74 { m y}$	-6.30	0.65	0.00040	1.7		
HD 092719	3	12.76 y	-18.25	0.06	0.97882	0.1		
HD 092945								\mathbf{C}
HD 094765	6	7.68 y	5.57	0.68	0.05949	1.4		С
HD 095128	6	$5.58 { m y}$	11.47	0.42	0.21593	1.2		С
HD 096064A	22	7.75 y	18.56	0.55	0.02311	1.3		
HD 096612	3	$5.78 { m y}$	-36.37	0.35	0.53940	0.8		
HD 097334	50	3.21 y	-3.72	0.37	0.90191	0.9	\mathbf{C}	С
HD 097343	6	3.23 y	39.25	0.42	0.07846	1.3		С
HD 097658	4	2.12 y	-1.34	0.84	0.05214	1.5		С
HD 098281	3	$1.07 { m y}$	13.36	0.36	0.49961	0.8		С
HD 099491	11	$11.25 { m y}$	4.20	0.34	0.78680	0.8		С
HD 099492	6	2.82 y	3.95	0.42	0.65762	0.9		С
HD 100180A	11	12.05 y	-4.52	0.53	0.04321	1.4		С
HD 100180B	9	$11.21 { m y}$	-4.45	0.47	0.29165	1.0		
HD 100623	4	3.15 y	-22.44	0.29	0.47680	0.9		С
HD 101177	3	$361 \mathrm{d}$	-16.86	0.32	0.40733	1.8		С
HD 101501	23	12.86 y	-5.82	0.52	0.00007	1.6		С
HD 102870	96	16.20 y	4.52	0.58	0.00000	1.6	\mathbf{C}	\mathbf{C}
HD 103095	348	$27.17 { m y}$	-98.49	0.48	0.99717	0.9	\mathbf{C}	С
HD 104067	1		14.73		1.00000			С
HD 104304	6	6.02 y	0.49	0.61	0.01160	1.6		
HD 105631	6	9.06 y	-2.43	0.40	0.72285	0.8		С
HD 108954	10	11.02 y	-21.66	0.48	0.16086	1.3		
HD 109358	9	16.10 y	6.02	0.45	0.06352	1.4		С
HD 110897	17	$10.67 { m y}$	80.13	0.39	0.54730	0.9		
HD 111395	126	16.11 y	-9.12	0.53	0.00000	1.4		
HD 114710	138	$25.87 { m y}$	5.22	0.54	0.00000	1.5		С
HD 114783	11	$4.22 { m y}$	-11.54	0.81	0.00389	1.8		\mathbf{C}

TABLE 6.1 - Continued

			$\langle \mathbf{RV} \rangle$	Ext Err				
Name	N	ΔT	(kms^{-1})	$(\mathrm{km}\mathrm{s}^{-1})$	$P(\chi^2)$	E/I	н,С	Nid02
HD 115383	24	10.87 y	-27.23	0.76	0.00000	2.0	С	\mathbf{C}
HD 115404A	24	$10.87 { m y}$	7.57	0.54	0.00449	1.4		
HD 115404B	3	8.06 y	6.80	0.47	0.48180	0.8		
HD 115617	33	13.06 y	-7.77	0.53	0.00001	1.5		\mathbf{C}
HD 116442	18	11.25 y	27.99	0.53	0.01646	1.3		С
HD 116443	9	11.25 y	27.59	0.36	0.49441	0.9		\mathbf{C}
HD 116956	9	$10.91 { m y}$	-12.15	0.39	0.55529	0.9	\mathbf{C}	
HD 117043	3	16.58 y	-30.89	0.28	0.58756	0.7		
HD 117176	6	5.70 y	4.83	0.36	0.30112	1.0		\mathbf{V}^{c}
HD 118972	5	$4.72 { m y}$	-7.83	0.34	0.70987	0.8		
HD 119332	6	9.78 y	-8.50	0.29	0.76803	0.7		
HD 121560	1		-11.44		1.00000			С
HD 124106	1		1.95		1.00000			С
HD 124292	8	8.71 y	37.58	0.37	0.38476	0.9		С
HD 125455	9	14.81 y	-10.32	0.91	0.00039	2.2		\mathbf{C}
HD 126053	131	$27.40 { m y}$	-19.44	0.54	0.00017	1.2		С
HD 127334	5	2.19 y	-0.65	0.44	0.42005	1.0		\mathbf{C}
HD 128311	1		-10.00		1.00000			С
HD 128987	4	$1.83 { m y}$	-22.71	0.44	0.55664	0.8		
HD 130307	5	$3.52 \mathrm{y}$	13.15	0.47	0.31642	1.1		С
HD 130948	135	16.30 y	-2.70	0.53	0.00000	1.2	\mathbf{C}	С
HD 131156	120	13.48 y	1.37	0.48	0.00000	1.4	\mathcal{L}^d	С
HD 131582	3	$7.95 { m y}$	-31.95	0.07	0.97192	0.1		
HD 132142	14	21.69 y	-14.90	0.46	0.24205	1.0		С
HD 132254	5	2.76 y	-15.59	0.34	0.46292	1.0		
HD 135204	5	12.85 y	-69.25	0.60	0.05294	1.4	\mathbf{C}	
HD 135599	4	$2.21 { m y}$	-3.30	0.84	0.00614	2.0	\mathbf{C}	С
HD 136202A	272	22.98 y	54.49	0.47	0.00000	1.4		
HD 136202B	37	16.64 y	55.08	0.60	0.55414	1.0		
HD 136713	2	$54.00 {\rm d}$	-6.12	0.39	0.43150	0.8	\mathbf{C}	С
HD 136923	11	8.67 y	-7.25	0.52	0.09426	1.2		С
HD 137778	3	$76.00 { m d}$	7.98	0.51	0.52087	0.9		С
HD 139323	7	$2.87 { m y}$	-66.84	0.69	0.15562	1.5		
HD 139777	13	14.30 y	-16.71	0.78	0.00000	1.9		
HD 139813	9	$12.05 { m y}$	-15.97	0.67	0.27068	1.4	• • •	•••
HD $140538A$	85	13.17 y	19.10	0.48	0.00004	1.3		•••
HD 140538B	4	$7.00 \mathrm{~d}$	20.10	0.46	0.84604	0.5		

TABLE 6.1 - Continued

_
			$\langle \mathbf{RV} \rangle$	Ext Err				
Name	N	ΔT	$(\mathrm{kms}^{'1})$	$(\mathrm{km}\mathrm{s}^{-1})$	$P(\chi^2)$	E/I	н,С	Nid02
HD 141004	9	14.87 y	-66.43	0.45	0.04026	1.3		С
HD 141272	13	15.00 y	-26.56	0.52	0.33443	1.1		
HD 142373	8	18.78 y	-56.44	0.55	0.19632	1.4		\mathbf{C}
HD 143761	9	13.01 y	18.00	0.35	0.41648	1.0		С
HD 144579	132	$25.13 { m y}$	-59.63	0.48	0.04102	1.0		С
HD 144872	6	8.88 y	23.63	0.51	0.25412	1.2		
HD 145675	11	$5.95~\mathrm{y}$	-14.07	0.49	0.08689	1.3	\mathcal{L}^{e}	С
HD 145958A	9	2.88 y	18.53	0.37	0.53610	1.0		\mathbf{C}
HD $145958B$	10	2.88 y	18.26	0.39	0.51763	1.3		С
HD 146233	60	13.56 y	11.68	0.34	0.92058	0.9		\mathbf{C}
HD 146362	18	8.17 y	-14.72	0.46	0.00770	1.3		С
HD 147776	2	$69.00 \ d$	6.84	0.41	0.33917	0.9		С
HD 148653	41	$12.27 { m y}$	-31.26	0.39	0.90813	0.9		
HD 149661	6	12.88 y	-12.65	0.46	0.34464	1.1		С
HD 149806	4	$3.51 { m y}$	10.46	0.18	0.90865	0.4		\mathbf{C}
HD 151541	10	22.48 y	9.16	0.33	0.47753	0.9		С
HD 152391	23	$17.97 { m y}$	45.06	0.45	0.05422	1.2	\mathbf{C}	\mathbf{C}
HD 153525	5	8.73 y	-7.19	0.42	0.36486	1.1		
HD 153557	5	8.73 y	-7.07	0.37	0.54220	0.9		
HD 154345	5	12.85 y	-47.29	0.48	0.26233	1.0		С
HD 154417	183	26.23 y	-16.89	0.49	0.00000	1.2		С
HD $155885A$	18	$12.97 { m y}$	0.20	0.71	0.00000	2.0		
HD $155885B$	19	$12.97 { m y}$	0.17	0.72	0.00000	2.0		
HD 156026	3	11.96 y	-0.96	0.60	0.04082	1.8		С
HD 157214	7	12.88 y	-78.81	0.38	0.20595	1.1		С
HD 157347	5	3.79 y	-35.70	0.26	0.79978	0.7		С
HD 158633	12	$12.77 { m y}$	-38.92	0.58	0.00186	1.7		
HD 159062	11	$12.51 { m y}$	-84.18	0.43	0.67373	0.9		
HD 159222	5	12.88 y	-51.49	0.41	0.62860	0.9		С
HD 162004	3	5.26 y	-10.65	1.28	0.00000	3.5		
HD 164922	6	8.64 y	20.34	0.39	0.36942	1.0		С
HD 166620	7	11.00 y	-19.38	0.26	0.70265	0.7		С
HD 168009	10	16.10 y	-64.82	0.53	0.08996	1.3		С
HD 170657	4	8.94 y	-42.95	0.37	0.73030	0.7		С
HD 172051	5	$12.15 { m y}$	37.05	0.10	0.99095	0.2		С
HD 176377	15	$12.67 { m y}$	-40.78	0.65	0.00634	1.5		С
HD 179957A	5	2.66 y	-41.13	0.22	0.80528	0.6		С

TABLE 6.1 – Continued

Continued on Next Page...

			$\langle \mathbf{RV} \rangle$	Ext Err				
Name	N	ΔT	(kms^{-1})	$(\mathrm{km}\mathrm{s}^{-1})$	$P(\chi^2)$	E/I	$_{\mathrm{H,C}}$	Nid02
HD 179957B	5	2.66 y	-41.73	0.29	0.59761	0.7		С
HD 182488	10	$12.52 { m y}$	-21.82	0.32	0.76760	0.8		\mathbf{C}
HD 182572	357	23.53 y	-100.33	0.43	0.00000	1.2	\mathbf{C}	
HD 183870	1		-49.74		1.00000			С
HD 184385	2	$182.00 {\rm d}$	11.29	0.49	0.15298	1.4		С
HD 185144	6	8.17 y	26.05	0.47	0.11850	1.3		С
HD 186427	8	24.90 y	-28.18	0.87	0.00000	2.4		\mathbf{C}
HD 187691	211	23.93 y	0.06	0.43	0.00000	1.2	\mathbf{C}	
HD 190067	8	$5.51 { m y}$	19.98	0.39	0.68373	0.9		С
HD 190360	128	$16.02 { m y}$	-45.38	0.53	0.00000	1.4		\mathbf{C}
HD 190404	15	21.21 y	-2.80	0.59	0.15044	1.3		
HD 190406	146	$11.97 { m y}$	4.89	0.37	0.15516	1.0		\mathcal{L}^{f}
HD 190470	5	21.31 y	-7.56	0.20	0.86806	0.5		
HD 191785	6	$2.52 \mathrm{y}$	-49.12	0.68	0.01536	1.5		С
HD 192263	1		-10.71		1.00000			\mathbf{C}
HD 193664	3	$352.00 {\rm ~d}$	-5.40	0.77	0.00192	2.5		
HD 194640	4	12.30 y	-0.45	0.38	0.33068	1.0		
HD 195564	3	3.12 y	9.23	0.16	0.70966	0.7		С
HD 196761	4	$12.02 { m y}$	-41.77	0.18	0.94571	0.3		С
HD 197076	159	17.14 y	-35.51	0.45	0.00000	1.2		\mathbf{C}
HD 199260	2	$84.00 \ d$	-16.55	0.34	0.33843	1.0		
HD 202751	4	2.84 y	-27.49	0.32	0.66950	0.8		С
HD 206860	28	$21.17 { m y}$	-16.92	0.48	0.04936	1.1	\mathbf{C}	С
HD 208038	9	9.62 y	13.63	0.65	0.03537	1.4		
HD 208313	5	$2.51 { m y}$	-13.45	0.69	0.05007	1.4		С
HD 210277	1		-19.68		1.00000			С
HD 210667	6	$2.50 { m y}$	-19.44	0.22	0.94965	0.5		С
HD 211472	11	12.03 y	-7.55	0.66	0.05574	1.5		
HD 214683	5	6.52 y	22.24	0.90	0.00182	2.0		
HD 215152	1		19.18		1.00000			С
HD 215648	5	11.25 y	-5.62	0.48	0.11565	1.4		С
HD 216259	11	2.11 y	1.44	1.05	0.00782	1.8		С
HD 217014	130	16.27 y	-33.17	0.49	0.00000	1.3		С
HD 217107	3	2.99 y	-12.95	0.48	0.36033	1.1	\mathcal{V}^{g}	С
HD 217813	11	12.13 y	1.95	0.55	0.00681	1.5	\mathbf{C}	\mathbf{C}
HD 218868	5	$2.21 { m y}$	-30.62	0.47	0.36929	1.1		
HD 219134	31	14.95 y	-18.57	0.56	0.00000	1.6		

TABLE 6.1 - Continued

Continued on Next Page...

Name	N	ΔT	$\langle {f RV} angle \ ({f kms^{-1}})$	${f Ext} {f Err} ({f km}{f s}^{-1})$	$P(\chi^2)$	$\mathrm{E/I}$	$_{\mathrm{H,C}}$	Nid02
HD 219538	5	2.10 y	9.85	0.57	0.18438	1.2		С
HD 219623	10	12.51 y	-27.29	0.49	0.09274	1.3		
HD 220182	25	18.84 y	3.43	0.46	0.08394	1.1		
HD 220339	2	$30 \mathrm{d}$	33.62	1.05	0.03781	2.0		\mathbf{C}
HD 221354	12	8.73 y	-25.37	0.40	0.32416	0.9		С
HD 221851	13	9.20 y	-21.13	0.33	0.88510	0.8		
HD 222143	12	8.45 y	-0.01	0.50	0.17576	1.2		С
HD 222368	142	23.45 y	5.56	0.46	0.01885	1.2		
HD 232781	5	9.11 y	-40.17	0.42	0.43936	0.9		
HD 263175	5	16.69 y	-31.93	0.57	0.16046	1.2		
HIP 036357	20	18.84 y	-4.14	0.58	0.01862	1.4		
HIP 040774	4	8.02 y	27.28	0.64	0.24353	1.3		
HIP 087579	21	16.57 y	-13.10	0.49	0.27335	1.0		С

TABLE 6.1 - Continued

Identified in Nidever et al. (2002) as $\sigma_{\rm RV} > 0.1 \ \rm km \, s^{-1}$ based on three observations, а but the 35 CfA measures over 14 years do not show any periodic variation consistent with While this star shows an increased scatter in 19 high-precision b a stellar companion. observations over 5 years, there is no periodic signature and the scatter is believed to be due to spots (A. Hatzes 2008, private communication) ^c The $\sigma_{\rm RV} > 0.1 \ {\rm km \, s^{-1}}$ variations noted in Nidever et al. (2002) are consistent with a published 7.5 M_J planet. d A linear trend with $\sigma_{\rm RV} < 0.1 \ {\rm km \, s^{-1}}$ is seen in 54 observations over 4.2 years in the high-precision data from A. Hatzes, but is likely due to the known 150-year VB. е A linear trend with $\sigma_{\rm RV} < 0.1 \text{ km s}^{-1}$, consistent with the known 5 M_J planet, is seen in 25 measures over 1.3 years by A. Hatzes. ^f A linear trend with a slope of $-0.066 \text{ m s}^{-1} \text{ day}^{-1}$ over 58 observations is mentioned in Nidever et al. (2002), and these variations are consistent with a brown dwarf discovered 0".8 away by Liu et al. (2002). g A. Hatzes measured variations with $\sigma_{\rm RV} < 0.1 \ {\rm km \, s^{-1}}$, consistent with the published planets of 1.4 M_J and 2.5 M_J.



FIGURE 6.1: The distributions of the number of radial velocity measurements per star (left) and their time span (right) in the CfA survey.

Figure 6.1 shows the extensive radial velocity coverage of these stable velocity stars. Thirty-nine percent of the systems listed have over 10 observations and 23% have over 20 observations. As seen in right panel of the figure, a majority of the stars have radial velocity measurements gathered over more than 10 years, and almost 10% of the systems observed have over 20 years of coverage. The left panel of Figure 6.2 shows that the distribution of the external error is roughly normal around a strong peak at 0.4–0.5 km s⁻¹, the typical measurement errors of the CfA observations, as expected for constant velocity stars. Only five stars (HD 68255, 68257, 162004, 216259, and 220339) have external errors larger than 1 km s⁻¹. The first two stars are separated by 1" and belong to a system with the largest number of stellar components (five, maybe six, see §7.4) for the sample studied here. The large error for these stars could be due to contamination from nearby sources (D. Latham 2008, private communication). HD 162004 appears to vary in a linear fashion in the 3 CfA

observations over 5.3 years, but the first observation is 2 km s^{-1} away from the other two. Moreover, eight early observations from Duquennoy et al. (1991) and 9 later measures from Abt & Willmarth (2006) are all consistent with the two latter CfA measures. While it is possible that this star is a highly eccentric binary, it appears to be a constant velocity star if the first CfA observation is excluded. HD 216529 has 11 observations over two years, the first of which is about 3 km s^{-1} away from the other observations, causing the large error. However, Nidever et al. (2002) reports this star with a scatter of less than 0.1 km s^{-1} based on 4 years of measurements at an earlier epoch, and their reported velocity agrees well with the CfA mean. It appears that the outlier point is erroneous and not indicative of variability. Finally, the two CfA measures for HD 220339, 30 days apart, vary by 1.5 km s⁻¹, but Nidever et al. (2002) show it to be of stable velocity in observations over almost 4 years. The right panel of Figure 6.2 shows that the distribution of the ratio of external to internal error has a broad peak at $1.0-1.5 \text{ km s}^{-1}$, again as expected for constant velocity stars. Fourteen stars have a ratio ≥ 2 , and an inspection of these reveals that it is due to one or two outliers and/or possible contamination from close sources. Some of these are exoplanet hosts exhibiting an increased scatter in the CfA data which is not supported by the higher precision measures from the planet search efforts.

For a fit with a given χ^2 value, the probability that the scatter around the fit is solely due to observational effects is given by $P(\chi^2)$, defined by Equation (6.1) (Barlow 1999). This metric has been used as an indicator of radial velocity variability in prior studies (DM91,



FIGURE 6.2: The distributions of the external errors (left) and the ratio of external to internal errors (right) for the CfA measurements.

Nordström et al. 2004).

$$P(\chi^2; N) = \int_{\chi^2}^{\infty} p(\chi'^2; N) d\chi'^2, \tag{6.1}$$

where N is the degrees of freedom of the fit, and $p(\chi'^2; N)$ is the probability for a specific χ'^2 value, given by Equation (6.2).

$$p(\chi^2; N) = \frac{2^{-N/2}}{\Gamma(N/2)} \chi^{N-2} e^{-\chi^2/2}$$
(6.2)

Figure 6.3 shows the $P(\chi^2)$ distribution, which is similar to that in DM91, with a clear discontinuity at $P(\chi^2) = 0.01$, to the left of which is a strong peak followed by a fairly flat distribution to the right. However, there is one key difference. While DM91 included all stars studied in their plot, I have only plotted stars that I believe are of constant velocity based on an examination of the individual velocity plots and consistent evidence from the highprecision measures. Moreover, the distribution of other metrics such as the external error and the external-to-internal error ratio are fully consistent with what would be expected for a set of single stars, as discussed in the paragraph above. Hence, it appears that a low $P(\chi^2)$ value does not, by itself, indicate variability due to binary orbital motion. Duquennoy et al. (1991) state that the $P(\chi^2)$ distribution is expected to be flat from 0 to 1 for single stars, while for variable radial velocity stars, it should be strongly peaked towards lower values. DM91 then use the discontinuity at 0.01 to argue that values smaller than this limit imply variability which, at least in some cases, is due to an unseen companion. Nordström et al. (2004) also use the same criteria of $P(\chi^2) < 0.01$ for "certain velocity variability – mostly due to binary orbital motion." However, given the similar appearance of Figure 6.3 for constant velocity stars to the DM91 plot of all stars, $P(\chi^2)$ does not appear to be a good indicator of variability. For instance, stars HD 3651 (a planet host), 4628, 90839, 126053, and 130948 have $P(\chi^2) < 0.01$, but a visual inspection of the velocity plots indicates either random scatter over many observations, or a couple of outlier points, but no evidence of periodic variation. Furthermore, each of these stars has a scatter of less than 0.1 km s⁻¹ in Nidever et al. (2002) across observations spanning at least 3000 days, increasing confidence in the conclusion that these are indeed stable radial velocity stars.

DM91 discuss the example of one star (HD 102870), an IAU velocity standard, which has $P(\chi^2) < 0.01$, and they point out that this low value is due to the large number of observations rather than due to variability, which they believe is clearly ruled out by a low $\sigma_{\rm RV} = 0.32 \text{ km s}^{-1}$. However, they also point out three other examples where a low $P(\chi^2)$ led to the suspicion of a companion, and they note that in each case companions were later reported in the sub-stellar regime. However, their study could not benefit from the highprecision measurements of the current planet search programs, which yield results accurate



FIGURE 6.3: $P(\chi^2)$ distribution for constant radial velocity stars from CfA measurements. to within a few m s⁻¹ and clearly show many of the low $P(\chi^2)$ stars to be of stable velocities. Figure 6.4 clearly shows the correlation between $P(\chi^2)$ and the number of observations. While only 4% of stars with fewer than 10 observations have $P(\chi^2) < 0.01$, 50% of the stars with greater than 10 observations and almost 80% of the stars with greater than 100 observations satisfy this criterion. To explore this further, let's look at the set of stars with greater than 10 observations and see if the subset with $P(\chi^2) < 0.01$ and $P(\chi^2) > 0.01$ differ in their external or internal error distributions as shown in Figure 6.5. First, note that the 11 points with the lowest values of external error belong exclusively to the high $P(\chi^2)$

subset, which follows from the fact that a low external error implies a low χ^2 and hence a high $P(\chi^2)$. In fact each of these has $P(\chi^2) > 0.15$. More importantly, the figure clearly shows that the low $P(\chi^2)$ subset has internal errors fairly well constrained between 0.3 and 0.5 km s⁻¹, while the high $P(\chi^2)$ subset shows a greater excursion in internal errors. If one were to fit lines to the two distributions, the circles will be best fit by a flat line while the triangles will be fit with a line that has a significant positive slope. This is a direct result of the fact that when internal and external errors are both high, the χ^2 value is reduced and hence the $P(\chi^2)$ value increases. This emphasizes that an accurate estimate of the floor error and a Gaussian distribution of the internal errors are essential in arriving at a reliable $P(\chi^2)$ value (D. Latham 2009, private communication), which apparently gets trickier as the number of observations increases. In any case, the large overlap between the two subsets for external error under 0.7 ${\rm km\,s^{-1}}$ illustrates the point that the cutoff of $P(\chi^2)\,<\,0.01$ does not necessarily imply velocity variability. So, it does appear that $P(\chi^2)$ is not a good measure when the number of observations is large. A visual inspection of the velocity and power spectrum plots as followed in this work, is a better approach in these cases. This has implications on the incompleteness analysis of DM91, and I will return to this issue in §8.2.

For stars with few observations, a $P(\chi^2) < 0.001$ can indicate evidence of variability, suggesting follow-up observations (D. Latham 2009, private communication). Three stars in Figure 6.4 satisfy these criteria – HD 125455, 162004, and 186427, each of which has one outlying measure by about 2 km s⁻¹ that seems to be responsible for the low $P(\chi^2)$. HD 125455 has one outlier among 9 CfA observations, but Nidever et al. (2002) shows it to



FIGURE 6.4: $P(\chi^2)$ distribution for constant velocity stars from the CfA measurements as a function of the number of observations. Stars with zero $P(\chi^2)$ values down to the precision limit of the calculations are plotted one order of magnitude below that precision limit, i.e. 0.000001. The horizontal dashed line marks the $P(\chi^2)$ value of 0.01, and the vertical dashed line delineates stars with 10 or more observations from stars with fewer observations.

be of stable radial velocity over almost 4 years with a velocity that is consistent with the remaining CfA measures. HD 162044 is discussed above in this section and is also likely of stable radial velocity. HD 186427 is the planet host 16 Cyg B, and shows no evidence of longterm variability beyond the reported planet, and is reported in Nidever et al. (2002) as a scatter less than 0.1 km s⁻¹ over 7 years of coverage. So, upon a closer inspection, these three candidates with low $P(\chi^2)$ and few observations also appear to be constant velocity stars.



FIGURE 6.5: External and internal error distributions for constant velocity stars with more than. Open circles are for stars with $P(\chi^2) < 0.01$ and filled triangles are for stars with $P(\chi^2) \ge 0.01$.

6.2 Stellar Companions

6.2.1 Single-lined Spectroscopic Binaries

Table 6.2 lists the 48 stars of the sample that have a single-lined spectroscopic orbit. Each star is listed on two lines with the first line specifying the orbital parameters, below which are listed their corresponding errors. The first column lists the HD name, with a suffix when needed, of the primary of the orbit. The next eight columns specify the parameters from the spectroscopic orbital solution, namely the period in days, the systemic radial velocity in km s⁻¹, the velocity semiamplitude of the primary in km s⁻¹, the eccentricity, angle of periastron in degrees, the epoch of periastron as HJD-2400000, the $a \sin i$ of the primary in gigameters, and the mass function in solar-mass units. Columns 10–12 specify the number of observations, their time-span in days, and the rms of the fit in km s⁻¹. Listed below the time span is the number of orbital cycles it represents. Column 13 lists the grade of the corresponding visual orbit, if one exists, as described in §5.2. Columns 14–16 list the orbital inclination in degrees and the primary and secondary stars' masses, in solar units, if available. Column 17 lists the reference for the spectroscopic orbit on the first line and the reference for the visual orbit, if available and different from the spectroscopic orbit, on the second line. Reference codes are expanded at the bottom of the table, except for visual orbit references, which are the same as in Table 5.4.

The table includes two new companion discoveries by the CfA surveys corresponding to the orbits of HD 185414 and 224465, two new spectroscopic orbits from the CfA surveys for *Hipparcos* photocentric motion orbits (HD 32850 and 128642) and one new preliminary spectroscopic orbit for HD 22409 that likely corresponds to a *Hipparcos* double star resolution (H59 flag = 'C'). An additional eight systems have the reference listed as "CfA", and for these the orbital solutions presented here from the CfA measurements are improvements over published solutions. The radial velocities and orbital solutions for these 13 binaries are presented in Figures 6.6–6.12. Because accurate parallaxes are available from FvL07 for all stars of this study, component masses can be estimated when the spectroscopic orbit has a corresponding visual orbit for resolved components. This is done by using the parallax and the visual orbit's angular semi-major axis to estimate the linear semimajor axis, then using the period from either solution and Newton's generalization of Kepler's Third Law to estimate the mass-sum. Then, the spectroscopic orbit's mass function along with inclination of the visual orbit lead to the component masses. Masses were determined in this manner only for HD 6582 and 10307, and these values are included in the table along with corresponding uncertainties. The other visual orbits were either for photocentric motions or were so poorly constrained that meaningful masses could not be estimated.

HD Name	Р (d)	$\gamma (\mathrm{kms^{-1}})$	K (km s ⁻¹)	е	ω (deg)	$T_0 \\ \text{HJD-2400000}$	$a_A \sin i$ (Gm)	f(M)	Ν	$\Delta T(\mathbf{d})$ cycl	$rms (km s^{-1})$	>	i (deg)	(\mathbf{M}_{1}	(\mathbf{M}_2	Ref
000123B	47.685	-11.70	10.2	0.610	290	49891.50	5.31	0.0026	71	29328.8	0.57	÷	:	:	:	Gri1999
001273	0.003 411.449	-14.32	0.5 13.88	0.024 0.567	4 4.68	0.22 34233.31	0.27 64.7	0.0639		4505.8	 1.83	. 6	74.7	: :	: :	Bop1970
	0.076	0.08	0.10	0.004	0.79	0.77				11.0		:	2.1	:	:	Jnc2005
003196A	2.082 0.000	11.23	43.98 0.39	0.0 fixed	0.0 fixed	43400.45 0.00	1.26	0.01844 0.00046	46	3732.9 1792.9	2.65	: :	: :	: :	: :	Duq1991b
004747	6832	9.97	0.65	0.64	257	50453	1000^{a}	0.0000	21	1731.2	0.00	:		0.83	0.04^{b}	Nid2002
0065234	653 7237	0.3	0.1 2.68	0.06	5 147 0	473 19601				0.3			106.9			Due 1001b
WZ OCODO	fixed	0.09	2.00 0.18	10.0	4.8	42034 fixed	15	0.0015	10	0.5	0.44	1 .	0.90 0.90	0.06	0.02	Dru1995
010307	7383	3.34	2.58	0.441	214.77	58037	234.9	0.00947	56	8157.9	0.51	4	105	0.81	0.25	CfA
1 101 10	653	0.13	0.13	0.053	5.92	667 73335 33	13.4	0.00169		1.1	: 0	: 0		0.40	0.07	Sod1999
014214	93.290 0.001	25.72	19.28	0.000	104.04 0.05	0.01 0.01	c0.12	0.00006	187	11.7	0.08	ה ת י ת	110.0 8.4	01.1	0.53	Fek2007
016287	14.838	22.10	10.53	0.216	11.3	47103.16	2.10	0.00168		5427.2	: :	: :		: :	: :	Imb2006
	0.000	0.06	0.10	0.009	2.2	0.09	0.02	0.00005	- 0	365.8	: 1	: 0	: .	:	:	
017382A	5575.8 61 1	9.14 0.07	2.89	0.034	114.4 4.6	48041.6	161.0	0.00535	92	7153.7 13	0.58	6	84	:	:	CtA Heitgand
024409	12625	-18.05	4.19	0.554	258.5	53916.9	605.5	0.0555	40	5396.1	0.44	: :	: :	: :	: :	CfA
	5700	1.27	0.68	0.067	15.1	123.9	103.8	0.0373	÷	0.5	:	:	÷	÷	:	•
026491	7008	:	2.67	0.51	215	50475	1143	÷	×	1879.8	0.00	:	÷	÷	:	Jon 2008
032850	798 205.684	23.70	0.30 10.42	0.324	556.27	54 49985 20	283 27,89	0.0204		0.3 4334.1	0.41	Ξσ	2.9.4	:	:	CfA
	0.084	0.12	0.19	0.018	3.10	1.49	0.54	0.0011	: :	21.1		, .	35.8			Hipparcos
039587	5136	-13.47	1.85	0.45	111	51463	883^{a}	0.0024	38	4579.7	0.03	6	95.9	0.89	0.14^{b}	Nid2002
	12	0.3	0.02	0.01	- 1	17		:	: ;	0.0	: 0	÷	0.8			$\operatorname{Hal2002}_{11}$
043587A	12325 500	:	4.32	0.80	75	50832	1735 "	:	14	1637.6 0.1	0.01	:	:	1.0	0.34°	Vog2002
054371	32.807	19.65	25.05	0.054	$^{1}_{48.62}$	51151.93	11.28	0.0532	: 1	5909.1	0.24	: :	: :	: :	: :	CfA
	0.000	0.14	0.20	0.007	4.61	0.43	0.09	0.0013	÷	180.1	:	:	:	÷	:	:
064468	161.2	:	5.73	0.262	328.1	50457.0	11.2^{a}	÷	13	1540.9	0.02	:	÷	0.78	0.13^{b}	Vog2002
064606	0.1	102.22	0.01	0.002	2.20.8	0.2	34.4	0.00798		9.6 5025.3	0.57	:	: :	:	:	 Lat 2002
	4.3	0.18	0.24	0.031	5.7	80.00	1.4	0.00097		11.2		: :	: :	: :	: :	
065430	3138	-28.43	1.11	0.32	77	3267	598^{a}	0.00038	26	1846.0	0.01	:	:	0.78	0.07^{b}	Nid2002
068257C	342 6302	0.3 -7.93	0.2 4.28	0.02	1 307	302 44696		0.050	103	7810	0.40	; 10	 142	: :	: :	 Gri2000
	59	0.08	0.12	0.018	80	135	13	0.005	÷	1.2		:	÷	:	:	Hei1996b
075767A	10.248	5.96	23.59	0.0	:	47212.16	3.32	0.01396	97	24448.6	:	:	:	:	:	Gri1991
079028	16.239	-14.91	35.31	0.102	137.7	53010.32	7.844	0.0729		1431.1	0.41	: :	: :	: :	: :	CfA
	0.000	0.15	0.19	0.006	3.4	0.16	0.043	0.0012		88.1	: •	: 0	: :	:	:	
098230A	670.24 0.09	:	8.95 0.12	0.008	314.1	47389.1 2.2	09.8 1 D	0.0013	40	3812.7	0.43	n	Tß	:	:	Gr11998 Hei1996h
098230B	3.981	: :	4.83	0.0	i : :	42441.42	0.26	0.00005	45	3812.7	0.62	: :	: :	: :	: :	Gri1998
101906	0.000		0.14 37 76		016 E	0.02	0.01	0.00000	: 6	958.0 623 2		:	÷	:	:	
907101	0.001	12.07	60.0	0.003	519.5 19.4	49109.98 0.70	4.93	0.0003	43	033.2 49.0	0.42	: :	: :	: :	: :	CIA
110833	271.165			0.784	264.43	50607.41	2.48	:		:		6	48.9	0.72	0.14	Hal2000
1107E0 A C	0.472			0.010	1.31	1.26	0.07		1 :			:	10.2		0.01	Hipparcos
H001711	0.026	4.20 0.03	1.92	0.032	09.5 29.5	7.83	2.70	0.00000		4429.0	en.u	: :	: :	0.19	0.20	UIA Hal2003
112914	710.6	27.51	5.61	0.326	65.0	49220	51.91	0.0110	88	11785.8	0.39	6	81.8	: :	: :	Gri2002b
000000	0.4	0.05	0.08	0.013	2.5	4	0.75	0.0005	: 0	16.6	:	:	5.0	:	:	Jnc2005
120690	3762 12	5.93 0.15	4.15 0.30	0.20	161 8	55176	210.3	0.02626	21	1187	•	:	:	:	:	Abt2006
121370	489.74	-0.62	7.87	0.327	323.22	52350.0	50.08	0.0209		1387.1	0.48	. 6	115.7	: :	: :	CfA
	3.30	0.15	0.22	0.023	5.08	5.6	1.46	0.0018		2.8	: ;	:	1.6			Jnc2005
122742 Continued o	3617 n Nevt Page	-11.20	6.41	0.480	183.0	52030	792.9^{a}	0.0670	24	3545.2	0.02	6	93.5	0.96	0.52''	Nid2002

Orbits
Binary
Spectroscopic
Single-lined
6.2:
TABLE

HD Name	Р (d)	(kms^{-1})	$K (\mathbf{km s^{-1}})$	υ	ω (deg)	T_0 HJD-2400000	$a_A \sin i$ (Gm)	f(M)	N	$\Delta T(\mathbf{d})$ cycl	$rms (km s^{-1})$	>	i (deg)	(\mathbf{M}_1	(${ m M}_2$	Ref
128642	7 178.780 0.031	0.3 - 39.12 0.13	0.01 13.46 0.21	$0.001 \\ 0.171 \\ 0.016 \\ 0.001 \\ 0.00$	$\begin{array}{c} 0.1 \\ 91.74 \\ 4.07 \end{array}$	$\frac{3}{51366.54}$	 32.60 0.51	0.0432	23	1.0 4491.8 25.1	0.58	6	54.8 5.2	: : :		Hipparcos CfA Hinnarcos
131511	125.396	-31.27	19.10	0.510	219.0	50203.41 0.00	77.794	0.0580	20	2866.1 22 q	0.03	6	93.4 4.2	0.78	0.43^{b}	Nid2002 1nc2005
131923	5431 1025		4.91 0.67	0.064	304 7	50512 38	1028 387		. 6	1917						Jon2008
142267	138.562 0.068	34.89 0.14	6.86 0.31	0.561 0.034	264.86 4.19	51782.99 1.28	10.81 0.58	0.00262	14	4791.0 34.6	0.40		:::		:::	CfA
144287	4450.8 6.8	-48.17	5.30	0.683 0.007	18.8 1.0	47679.7	237.1	0.0269 0.0019	68	4602.6 1.0	0.30	: :		 	0.32^{d}	Duq1991b
145825	2609 370		0.58	0.21	86 16	50718 36	568 145		12	1796			::		: :	Jon2008
147584	12.977 0.000	 6.39 0.01	0.04 7.50 0.00	0.014	252.98 0.80	52752.32 0.00	0.4 T	: :	228	156.1 12.0	: :	6	16.0	1.12	0.09-0.45	Sku2004 Jnc2005
156274	88.033 0.001		16.26 0.01	0.00	306.9 0.1	51628.33	58.34 4.49		. 11	496.7						Jon2008
160269A	27087 fixed	-16.30 fixed	$3.84 \\ 0.41$	0.019 fixed	126.4 4.2	33184 fixed	$1405 \\ 150$	$0.151 \\ 0.048$	12	2813.6 0.1	0.27	т г	104	1.00	0.82	$\operatorname{Duq1991b}$ Sod1999
160346	$83.728 \\ 0.037$	21.19 0.08	5.7 0.1	0.226 0.021	140.5 4.9	47724.9 1.2	: :	0.00135 0.00011	35	4535.35	0.42	6	: :	: :	0.09^{d}	Tok1991
161198	2558.4 10.4	23.88 0.04	9.09	0.936	129.6 0.6	49422.530.10	: :	0.00867	138	3196.3	0.33	ю	42.8 3.7	0.78	0.32	Duq1996
175742	2.879	10.31	49.46	0.003	266.7	43675.58 0.25	1.96	0.0362	76	12826.7 4455.3	:	:				Imb1979
176051	22423 fixed	-45.82 0.69	3.51 0.74	0.25 fixed	102 fixed	41398 fixed	1047 222	0.091	16		0.36	7	115.1	1.00	0.71	$\operatorname{Duq1991b}$ $\operatorname{Doc2008f}$
178428A	21.955 0.001	14.36 0.08	13.42 0.12	$0.08 \\ 0.01$	57.4 4.9	45592.23 0.29	4.04	0.00546 0.00016	17	2256.8 102.8	0.28	: :				Duq1991b
185414	4778.0 1119.1	$-16.70 \\ 0.18$	$1.65 \\ 0.14$	$0.670 \\ 0.070$	$136.84 \\ 9.77$	55462.6 1133.6	80.31 9.73	0.00090 0.00034	47	5698.4 1.2	0.47	: :	: :	: :	::	CfA
202940A	21.346 0.001	-17.19 0.11	23.98 0.15	$0.251 \\ 0.007$	89.48 1.34	19224.11 0.12	6.81	0.0277	22	19373.6 907.6	2.98	: :	::	::	::	Bop 1970
224465	52.413 0.004	1.60	11.10 0.17	0.143 0.016	164.83	50600.64	7.92	0.00720 0.00033	19	4136.0 78.9	0.47	:	:	: :	: :	CfA
224930	9610 fixed	-36.22 0.03	$4.49 \\ 0.05$	$0.372 \\ 0.009$	285.0 1.6	47738 32	551 6	$0.0722 \\ 0.0024$	126	12519.8 1.3	0.25	6	49	: : :		Gri2004 Sod1999

TABLE 6.2 - Continued

^c Halbwachs et al. (2003) reported an orbit for this star with some parameters, but the reference cited (Udry et al. 2002) was not found. The orbit presented here is a preliminary solution based on only ^b $M \sin i$ estimated using equation (1) in Butler et al. (2006). ^a Estimated assuming a low-mass companion according to equation (2) in Butler et al. (2006). seven CfA observations. ^d Minimum-mass estimates in the cited references.

REFERENCES.— Abt & Willmarth (2006); Bop1970 = Bopp et al. (1970); CfA = D. Latham 2008 (private communication); Duq1991b = DM91; Duq1996 = Duquennoy et al. (1996); Fek2007 = Fekel et al. (2007); Gri1991 = Griffin (1991); Gri1998 = Griffin (1998); Gri1999 = Griffin (1999); Gri2000 = Griffin (2000); Gri2002b = Griffin (2002); Gri2004 = Griffin (2004); Hal2000 = Halbwachs et al. (2000); Hal2003 = Halbwachs et al. (2003); Imb1979 = Imbert (1979); Imb2006 = Imbert (2006); Lat2002 = Latham et al. (2002); Jon2008 = H. Jones 2008, (private communication); Nid2002 = Nidever et al. (2002); Sku2004 = Skuljan et al. (2004); Tok1991 = Tokovinin (1991); Vog2002 = Vogt et al. (2002).



FIGURE 6.6: Single-lined spectroscopic orbits of HD 10307 (left) and HD 17382 (right)



FIGURE 6.7: Single-lined spectroscopic orbits of HD 24409 (left) and HD 32850 (right)

6.2.2 Double-lined Spectroscopic Binaries

Table 6.3 lists the 26 double-lined spectroscopic binary solutions for the stars of this sample. Additionally, Fuhrmann et al. (2005) identified HD 75767B itself as a double-lined binary by resolving pairs of lines in their spectra (see§ 7.4), but they do not have enough observations or



FIGURE 6.8: Single-lined spectroscopic orbits of HD 54371 (left) and HD 79028 (right)



FIGURE 6.9: Single-lined spectroscopic orbits of HD 101206 (left) and HD 112758A (right) radial velocity measurements to derive an orbit, so it is not listed in the table. The table lists each star on two lines, the first line contains orbital and physical parameters, and the second line lists the corresponding uncertainties. The first column identifies the spectroscopic pair by its HD number and, when required, its component designation. Columns 2–10 list the spectroscopic parameters, namely, period in years (y) or days (d), systemic velocity in km s⁻¹,



FIGURE 6.10: Single-lined spectroscopic orbits of HD 121370 (left) and HD 128642 (right)



FIGURE 6.11: Single-lined spectroscopic orbits of HD 142267 (left) and HD 185414 (right) the primary and secondary semiamplitudes in km s⁻¹, mass ratio, eccentricity, angle and epoch of periastron in degrees and HJD-2400000 respectively, and $a \sin i = (a_1 + a_2) \sin i$ in gigameters. Column 11 lists the number of velocity measures for the primary and secondary on the first line, and their time span in days below it. The next column lists the residuals for the primary and secondary from the orbital fit in km s⁻¹. Column 13 lists the grade of



FIGURE 6.12: Single-lined spectroscopic orbits of HD 224465

the visual orbit, if one exists, as described in §5.2, following which appear the inclination in degrees and the longitude of the ascending node in degrees, from the visual-orbit solution or a joint spectroscopic-visual solution. The next two columns list the component masses, again from a joint or the visual orbit solution. The last column lists the reference for the spectroscopic or joint solution on the first line, and the reference for the visual orbit below it, if different. Reference codes are expanded at the bottom of the table, except for visual orbit references, which are the same as in Table 5.4. Parameters are preferentially listed from joint solutions over isolated spectroscopic and visual results.

Orbits
Binary
Spectroscopic
Double-lined
6.3:
TABLE

Ref	Duq1991b	Pou2000	:	Bod1999	CfA		CfA MT-m1000	Poi:2000	Bag2006	ČfA	§ 3.2.3 Pou2000		Pou2000	CfA		Maz1997		CfA	 Pou1999	:	Lu2001		Z66TbnG	Pou2000		Maz2002	Rag2009	; :	CfA	 Pou2000		Pou2000	Doc. 2000	r ou zouo	Pou2000	:	Tor2002	 Poi:2000		CfA	§ 3.2.4
$^{M_2}_{ m (\ M_\odot)}$	1.00	0.70	0.08	1.17	70.0	:	:		0.02	0.71	80.0 0.9	0.1	0.85	0.03		:	:	:		0.03	0.76	0.01	:	1.05	0.05	0.46	1.09	0.04	:	06.0	0.05	0.78	0.04	0.1		:	0.67	1 1 2	0.03	0.75	0.01
(\mathbf{M}_{0}	1.53	0.94	0.09	1.22		:	:	138	0.02	0.83	0.10	0.08	0.89	0.03	: :	:	:	:	1.16	0.03	0.37	0.00	÷	1.19	0.07	1.2	1.14	0.04	:		0.05	0.90	0.07	0.2	; :	:	0.84	0.U2 1 19	0.03	0.79	0.01
Ω (deg)	149.	291.8	0.5	63.6 0 °	o.	:	15.	9. 260.3		125.0	4.0	0.3	317.6	0.5	: :	:	:	÷	204.8	0.1	:	:	:	22.9	0.2	:	207.9	0.7	÷	332.3	0.1	302.1	0.1	0 1 2	$^{-}_{147}$	ę	335.0	1.U 203.8	0.3	126.8	1.3
i (deg)	49.4		0.3	73.8	ю.	:	167.	3. 128.2	0.1	108.0	3.0 80.4	0.2	124.1	0.6	: :	:	:	:	79.2	0.1	:	:	:	58.7	0.2	:	28.1	0.3	:		0.1	121.2	0.1	144 2	' - - -	:	99.4 0.1	1.U	0.4	50.2	0.4
>	1	-	:	œ	: :	:	-	; x		x	: ~	ı :	1	: :	: :	÷	:	:	. 61	:	:	:	:	п	:	:	: ∞	:	:	:	:	1	:-	-	2	:	×	:	• :	x	:
$egin{array}{c} \sigma_1 \ \sigma_2 \ (\mathbf{km}\mathbf{s}^{-1}) \end{array}$	0.7	1.0	0.8	:	0.4	1.4	0.6	1.3	1.9	0.5	2.3	0.7	0.8	1.1	0.6	2.2	2.3	0.5	1.1	1.2	5.5 5	ი. ი. ი	0.3 2 0	0.3	0.6 1	2.7	1.0	1.1	0.3	0.0	0.6	0.6	0.0	0.7	0.4	0.9	0.3	0 T O	0.7	0.5	0.7
${N_1/N_2 \over \Delta T}$ (d)	46/46	3733 25/25	7019	:	25/25	744	29/29	6867. 78/68	35142	20/20	6159 26/18	25282	104/103	5412 $36/36$	688	32/3		23/23	283/152	31946	65/63	10	97/10 4402	31/31	7614	4/4	46/46	2610	11/11	98U 29/29	9146	91/5	34050 26 /36	4016 4016	25/25	5136	73/73 6450	0408 46/46	5562	18/18	198
asin i (Gm)	606.4	1511.9	52.8	22.6	13.1	0.1	3.4	1.0	0.5	11.7	0.0 1452.6	63.3	349.5	4.8	0.0	:	:	277.9 8 0	3570.5	13.8	:		201.2	2343.5	42.4	:	4.2	0.0	27.8	2396.1	40.1	3529.9	76.9	15.6	184.5	6.2	49.9	0.0 631.0	9.1	10.2	0.0
T_0 HJD-24e5	46864.8	3.9 14482.7	60.2	50906.4	49085.1	0.2	50772.8	0.0 52765 1	0.1	52480.9	0.0 46400.1	7.4	45222.0	1.5 49191	93	46998.2	0.1	52687.8 26	35330.0	6.9	50944.7	0.0	47967.5 0 0	12214.2	11.0	45972.0 0 1	50127.6	0.0	53644.2	37898.0	56.7	13539.3	6.0 16161 0	40104.9	45265	11	49404.8	0.0 44778	4	52232.3	0.2
ω (deg)	109.0	3.1 317	ŝ	203.6	179.2	5.1	336	23 80 0	0.3	78.9	1.0	0.4	350.7	8.0 339	339	354.0	0.8	148.5	231.8	0.2	:		253.9	219.2	0.4	63 15	2 -	:	167.1	148 148	1	14.0	0.1 956	200	$^{-}_{142}$	4	357.4	x		279	11
ø	0.77	пхеd 0.235	0.010	0.238	0.037	0.003	0.034	0.014	0.002	0.146	0.003	0.007	0.433	0.003	0.002	0.402	0.007	0.502	0.519	0.001	0.0	• • • •	67.6.0 0.000.0	0.277	0.001	0.039	0.0	:	0.165	0.168	0.003	0.499	0.000	0.008	0.592	0.009	0.306	0.440	0.005	0.016	0.004
5	0.66	$0.04 \\ 0.72$	0.05	:	0.825	0.009	0.671	0.20.0	0.005	0.854	0.005	0.04	0.96	0.02 0.996	0.003	0.48	0.03	0.798	0.82	0.04	0.49	0.01	0.006	0.89	0.03	0.380	0.959	0.005	0.949	0.03	0.03	0.85	0.04	0.02	0.99	0.07	0.79	10.0	0.02	0.951	0.006
K_2	16.44	0.43 6.93	0.22	59.95 0.33	47.35	0.45	14.89	0.34 23 73	0.09	66.48	0.27 9.69	0.26	12.13	0.14 71.31	0.15	50.5	2.7	13.28	5.5	0.1	112.70	0.46	55.5U 0.43	5.28	0.14	66.U 2.2	63.89	0.22	32.95	0.24 5.32	0.12	4.19	0.16	0.15	4.64	0.28	36.73	12.52	0.15	48.91	0.24
$\frac{K_1}{[\mathbf{km s}^{-1}]}$	10.90	0.59 5.13	0.29	57.35 0.21	39.08	0.12	9.99	20.91	0.09	56.76	0.20	0.63	11.49	0.11	0.16	24.41	0.15	10.60	4.6	0.1	231.31	0.65	37.14 0.12	4.71	0.10	25.10	61.25	0.21	31.28	0.10 4.93	0.11	3.66	0.08	9.30 0.14	4.69	0.13	28.94 0.05	0.00 11 73	0.15	46.54	0.18
۔ ۱	8.83	0.22 18.4	0.1	:	21.26	0.08	-6.42	0.11 - 23 03	0.04	-8.50	-21.3	0.2	49.82	-4.23	0.08	-18.71	0.18	-2.88	-21.87	0.05	-17.89	0.40	0.04	-7.41	0.05	-8.23	-13.03	0.11	-50.94	-77.18	0.07	-6.87	0.08	01.0	30.02	0.09	-5.87	- 15 85	0.07	4.80	0.10
Р (d)	2527.165	пхеd 9164.123	10.958	13.825	10.984	0.000	10.020	330 991	0.004	6.992	0.000 8291.175	10.958	988.001	0.365 3.804	0.000	23.542	0.001	978.5 2.0	29183.48	3.65	0.27		889.62 0 1 2	15190.75	3.65	3.071	1.140	0.000	31.865	0.000 16925.69	7.31	44253.69	29.22 404.183	494.103	1787.899	1.826	57.322	2083-021	2.557	7.754	0.001
HD Name	003196AB	003443		004676	008997		013974	016739		045088	064096		079096A	080715		101177B		111312	128620AB		133640B		137763	137107AB		144284	146361A		148704	158614		165341	101101	104401	189340^{a}		195987	202275		223778	

he took another look at this system adopting FvL07 parallaxes, but still could not constrain the mass estimates to be useful (D. Pourbaix, 2008, private communication). Apparently this

^a Pourbaix (2000) notes that a meaningful inclination or component masses could not be derived because of the very large correlation between inclination and parallax. At my request,

system needs more observations to derive useful physical parameters.

REFERENCES.— Bag2006 = Bagnuolo et al. (2006); Bod1999 = Boden et al. (1999); CfA = D. Latham 2008 (private communication); Duq1991b = Duquennoy & Mayor (1991); Duq1992 = Duquennoy et al. (1992); Lu2001 = Lu et al. (2001); Maz1997 = Mazeh et al. (1997); Maz2002 = Mazeh et al. (2002); Pou1999 = Pourbaix et al. (1999); Pou2000 = Pourbaix (2000); Rag2009 = Raghavan et al. (2009); Tor2002 = Torres et al. (2002). The table includes seven orbital solutions from the CfA results presented here, including five improved solutions, one new orbit for a new companion detection (HD 111312), and one double-lined solution for a previous SB1 pair (HD 148704). The velocities along with the orbital fit and residuals for each of these are shown in Figures 6.13–6.16. Most entries in Table 6.3 have mass estimates from the references. In cases where mass estimates do not exist, one can determine them using the separate spectroscopic and visual solutions, specifically, $M_{1,2} \sin i^3$ from spectroscopy and the inclination from the visual solution. However, this method resulted in uncertainties too large to be astrophysically useful. A better estimate of the companion's mass can be derived using the mass ratio and primary's mass estimates from other sources, and this will be the approach used in Chapter 8.



FIGURE 6.13: Double-lined spectroscopic orbits of HD 8997 (left) and HD 13974 (right)

In addition to radial velocity measurements, double-lined binaries can be identified by even a single observation if their spectrum shows the signatures of two stars. Hynek (1938) reported on the results of a comprehensive survey, presenting 566 such composite-spectrum



FIGURE 6.14: Double-lined spectroscopic orbits of HD 45088 (left) and HD 80715 (right)



FIGURE 6.15: Double-lined spectroscopic orbits of HD 111312 (left) and HD 148704 (right) binaries. While this compilation has proved to be a happy hunting ground for visually resolving the component stars through speckle interferometry (e.g., McAlister 1978b, 1979), only one star from the list belongs to the sample of this study. The overlapping star, HD 109358, is now recognized as one that does not show any evidence of binarity (see § 7.4).



FIGURE 6.16: Double-lined spectroscopic orbits of HD 223778

6.2.3 Radial Velocity Variations Without Orbits

In some instances, radial velocities show a clear or suggestive variation consistent with a spectroscopic binary, but there are insufficient observations to develop even a preliminary orbit. Table 6.4 lists 16 such stars, eight of which, listed in the first section, show velocity variations consistent with companions known by other means as discussed in previous chapters, such as visual resolutions with evidence of orbital motion or a variation in proper motion, suggesting an unseen companion. The second section lists one star that shows convincing variations in the CfA measures indicative of a new companion, and the final section includes seven additional candidate companions, four of which suggest a linear trend over a few CfA observations, and three of which were identified in the CNS catalogs as radial velocity variables with no additional information and no independent confirmation or refutation.

HD		ΔT	$\langle {f RV} angle$	Ext Err			
Name	N	(y)	(kms^{-1})	$(\mathrm{kms^{-1}})$	$P(\chi^2)$	E/I	\mathbf{Ref}
	Va	riations co	orresponding	to known c	ompanion	S	
016765	3	7.87	8.10	1.19	0.14208	1.5	CfA
043834							Egg2007
052698	6	7.12	12.03	0.75	0.00018	2.1	CfA
063077	4	3.95	102.54	1.97	0.00000	4.6	CfA
096064BC	19	4.98	16.91	1.67	0.00000	2.3	CfA
161797	134	16.92	-16.67	0.57	0.00000	1.6	CfA
186408	4	24.90	-27.15	0.85	0.00151	2.2	CfA
190771	13	27.48	-25.65	0.78	0.00000	2.1	CfA
		T T •					
		Variatio	ns implying	a new comp	Danion		
016673	6	11.23	-7.06	7.44	0.00000	18.6	CfA
0-0000	Ū						
	Va	riations su	ggesting a p	ossible new	companio	n	
020010B							CNS
020010D	• • •	•••	• • •	• • •		• • •	CNS
020404 000830R	• • •	•••		• • •		•••	CNS
102438		 19 14	 13.64	0.44	0.13325	 13	CfA
110463	3	4 83	-9.43	0.44	0.16020	1.0	CfA
186858	3	16 56	475	0.70	0.00400 0.05576	1.7	CfA
198425	4	8.08	10.46	0.78	0.07981	1.6	CfA
	-	0.00	-0.10				

TABLE 6.4: Stars with Possible RV Variations

REFERENCES.— CfA = D. Latham 2008 (private communication); CNS = Gliese (1969); Gliese & Jahreiß (1979, 1991); Egg2007 = Eggenberger et al. (2007).

Column 1 of the table lists the HD number and component designation, when required. Columns 2 and 3 list the number of observations and their time span in years, if available. The next two columns list the mean radial velocity and its external error in km s⁻¹ followed by the $P(\chi^2)$ and the external to internal error ratio from the CfA results. The last column identifies the source of the velocity variation.

6.3 Planetary Companions

In addition to the stellar companions discussed in the previous section, the spectroscopic technique is responsible for identifying a majority of the planetary companions discovered to-date. As one of the objectives of this effort is studying the correlation between stellar and planetary companions, it is essential to include a current list of planets around the stars of this study. The two online sources which maintain frequently updated results on planetary companions, the Extrasolar Planets Encyclopedia² and the CCPS program's website³ aided this piece of the effort. While the former source is updated much more regularly, almost on a daily basis, it does not list the uncertainties of the parameters. I have listed the 50 planetary companions in 33 systems from the latter source in Table 6.5, downloaded on December 8, 2008, but with a last update date of January 26, 2008. I also cross-checked against the Extrasolar Planets Encyclopedia to ensure completeness. Given the large volume of work focused in this field, the list presented in this table is certain to be obsolete before it is printed, but provides enough information to enable the statistical analysis in later chapters.

The first column of the table lists the HD number of the planet host and the next column gives the planet's designation. Columns 3–7 list the orbital elements, namely the period

 $^{^{2}}http://exoplanet.eu/$

 $^{^{3}}http://exoplanets.org/$

in days, velocity semiamplitude in $m s^{-1}$, eccentricity, and the argument and epoch of the periastron in degrees and JD-2440000 respectively. The second line for each planet lists the corresponding uncertainties. Columns 8 and 9 list the minimum companion mass and semimajor axis, as described in Butler et al. (2006). The last two columns list the number of velocity measurements used to determine the orbit, and the reference as listed in the website.

6.4 Eclipsing Binaries

A search of the All Sky Automated Survey⁴ for variable stars and Malkov et al. (2006), who cataloged 6330 stars from the General Catalog of Variable Stars, revealed only three potential eclipsing stellar companions to the stars of this study: HD 123, 9770, and 133640, the first of which was later refuted while the other two are real. HD 123 is listed in the catalogs without a period, but the entry likely corresponds to a suspected brightness variability with a 1-day period, which was refuted by Griffin (1999) (see § 7.4). For HD 9770, Cutispoto et al. (1997) present an eclipsing light curve and a corresponding orbit with a period of 11.4 hours. HD 133640 is listed in Malkov et al. (2006) with a period of 6.4 hours, which matches that of the double-lined spectroscopic orbit. Only one of the extrasolar planets of this sample, HD 189733b, also has a transiting light-curve solution.

 $^{^{4}}http://archive.princeton.edu/~asas/$

Companions
Planetary
6.5:
TABLE

II

п	ž	а (, K1 		з,	710 01 10000)	Msini	amin	rms		ç
Iname	Flanet Name	(a)	(s m)	e	(geb)	(0000777-777)	(^O INF)		(su)	A7	LIGI
1.27.100	ни 1237 в	133.71 0.20	10.7.01 4.0	0.017	167	0.6 0.6	3.37	0.495	БТ	10	Naef 2001
003651	HD 3651 b	62.241	16.1	0.590	238	12190.7	0.235	0.296	5.7	118	Butler 2006
004308	HD 4308 b	0.026 15.560	1.2 4.1	0.051 0.0		0.7	0.047	0.118		$\frac{1}{41}$	Udrv 2005
000000	-	0.020	0.2	fixed			· 10	· 1			
009826	v And b	1296 12	1.5	0.267 0.041	170	210125 21	3.97	2.55	12	197	Butler 2006
009826	$v \ And \ c$	241.31	55.5	0.258	250	10158.5	1.98	0.832	12	251	Butler 2006
009826	v And d	$0.24 \\ 4.617$	1.4 69.7	0.022	180 58	3.1 11807.2	0.686	0.060		251	Butler 2006
010647	HD 10647 b	0.000	1.1	0.016 0.16	336	0.6 10960	 0.920	 0.03		 80	 Butler 2006
1E00TO	0 1E001 /TTT	56	4.6	0.22		160			Ħ	0 ·	
013445	HD 13445 b	15.765	376.7	0.042	269 10	11903.4	3.91	0.113	12	42	Butler 2006
017051	t Hor b	0.000 302.8	2.9 57.1	0.007 0.14	$^{16}_{346}$	0.6 11227	2.08	0.93			Butler 2006
		2.3	5.2	0.13	: -	46	- 0				
019994	HD 19994 b	535.7 3 1	36.2 1 9	0.300	41 8	12	1.69	1.43	8.1	48	Mayor 2004
022049	ϵ Eri b	2630	10.8	0.35	148	9780	0.603	3.5	8.6	107	Butler 2006
033564	HD 33564 h	260 388 0	3.7	0.20		340 19602 D		1		. <u>1</u> : -	Collor Duck
#00000	11 P0000 UII	3.0	5.0	0.240	604 4	0.0021 8.0	от. 	71.1		2 : -	Callallu 2000
039091	HD 39091 b	2151	196.4	0.641	330	7820	10.3	3.38	5.5	42	Butler 2006
040307	HD 40307h	85 4 312	1.3	0.007		170 14562 8	0.013	0		1 25 1 25	 Mayor 2008
00000		0.001	0.1	fixed	fixed	0.1			2 :		
040307	HD 40307c	9.620	2.5	0.0	0	14551.5	0.021	0.081	0.9	135	Mayor 2008
202020	HD 403074	0.002 20.46	0.1 1 6	fixed	fixed	0.2			. c		Merror 2008
10e0#0		20.40	4.0 0.1	u.u fixed	u fixed	0.3		+er.0	с. 		Mayur 2000
069830	HD 69830 b	8.667	3.5	0.100	340	13496.8	0.032	0.079	0.8	74	Lovis 2006
069830	HD 69830 c	0.003 31 560	0.2	0.040	26 231	0.1 13460 6	0.037	0.187	.∝ .⊂		T. ords 2006
0000000	1117 02000 TH	0.040	0.2	0.060	35	2.8			o	<u>r</u> :	
069830	HD 69830 d	197.0	2.2	0.070	224	13358	0.057	0.633	0.8	74	Lovis 2006
075732	55 Che h	3.0 14.651	0.2 71.8	0.070	$61 \\ 164$	34 7572.0	0.816	0.114			Fischer 2007b
		0.001	0.4	0.008	30	1.2					
075732	55 Cnc c	44.379	10.1	0.053	57	7547.5	0.165	0.238	6.7	636	Fischer 2007b
075732	55 Cnc d	0.007 5370	0.4 47.2	0.063	29 163	3.3 6860	3.84	5.84			Fischer 2007b
		230	1.8	0.030	32	230	÷	÷		:	
075732	55 Cnc e	2.797	3.7	0.264	157	7578.2	0.024	0.038	6.7	636	Fischer 2007b
075732	55 Cnc f	260.7	4.8	0.00	206	7488.0	0.141	0.775		636	Fischer 2007b
006190	47 TIME F	1.1 1005 0	0.0	0.20	60	1.1					 Eischer 2003
071000	T DIMIG D	2.9	0.9	0.0 fixed	: :	: :	70.7	# · ·	ť :	701	
095128	47 UMa c	2190	7.0	0.220	180	11581	0.46	3.39	7.4	182	Fischer 2002
099492	HD 99492 b	460 17.043	2.3 9.8	0.068	170 219	$^{40}_{10468.7}$	0.109	0.123	3.6		Butler 2006
		0.005	1.0	0.092	22	1.4	÷	:	:	:	:
114783	HD 114783 b	494.3	30.2	0.122	95 75	10850	1.06	1.17	4.1	57	Butler 2006
117176	70 Vir b	116.688	316.3	0.401	358	7239.8	7.49	0.484	7.4		Butler 2006
901001		0.004	1.7	0.004	1 00 1	0.2					 D10000
001071	1 DOO 1	0.000	401.1 7.6	0.015	00	0.5	···	0.040		o .	DULLET 2000
Continue	4 on Nevt Page										

НD		D	K_1		Э	T_0	$M \sin i$	a_{\min}	rms		
Jame	Planet Name	(P)	$(m s^{-1})$	в	(deg)	(JD-2440000)	(M ₀)	(AU)	$(m s^{-1})$	N	\mathbf{Ref}
28311	HD 128311 b	456.7	71.7	0.217	110	10218	2.37	1.1	17	86	Vogt 2005
		1.5	8.8	0.081	160	17	:	:	:	:	:
28311	HD 128311 c	912.9	77.9	0.24	198	10060	3.23	1.74	17	86	Vogt 2005
		6.8	4.3	0.13	:	160	:	:	:	:	:
13761	$\rho \operatorname{CrB} \mathrm{b}^a$	39.845	64.9	0.057	303	10563.2	1.09	0.229	6.9	26	Butler 200
		0.006	2.4	0.028	:	4.1	:	:	:	:	:
15675	14 Her b	1755.3	91.8	0.373	19	11368.1	5.07	2.85	7.6	51	Butler 200
		3.2	1.0	0.009	7	5.9			:	:	
7513	HD 147513 b	528.4	29.3	0.260	282	11123	1.18	1.31	5.7	30	Mayor 200
		6.3	1.8	0.050	6	20	:	:	:	:	:
4345	HD 154345 b	3322	14.3	0.036	113	13230	0.963	4.18	2.7	41	Butler 200
		93	0.8	0.046	:	330	:	:	:	:	:
0691	μ Ara b	630.0	37.4	0.271	260	10881	1.67	1.51	4.7	108	Butler 200
		6.2	1.6	0.040	7	28	:	:	:	:	:
0691	μ Ara c	2490	18.1	0.463	184	11030	1.18	3.78	4.7	108	Butler 200
		100	1.1	0.053	œ	110	:	:	:	:	:
0691	μ Ara d	9.550	4.1	0.0	:	:	0.047	0.092	0.9	24	Santos 2004
		0.030	0.2	fixed	:	:	:	:	:	:	:
0691	μ Ara e	310.55	14.9	0.067	6	12708.7	0.546	0.942	0.9	24	Santos 2004
		0.83	0.6	0.012	2	8.3	:	:	:	:	:
4922	HD 164922 b	1155	7.3	0.05	195	11100	0.36	2.11	3.7	64	Butler 200
		23	1.2	0.14	:	280	:	:	:	:	:
6427	16 Cyg B b	798.5	50.5	0.681	86	6549.1	1.68	1.68	7.3	95	Butler 200
		1.0	1.6	0.017	7	6.6	:	:	:	:	:
9733	HD 189733 b	2.219	205.0	0.0	:	:	1.14	0.031	15	:	Knutson 20
		0.001	6.0	fixed	:	:		:		:	
0360	HD 190360 b	2925	23.2	0.307	10	10632	1.56	4.02	3.1	105	Vogt 2005
		28	0.D	0.021	80	32			:	:	
0360	HD 190360 c	17.112	4.6	0.004	168	9999.8	0.058	0.13	3.1	105	Vogt 2005
		0.008	0.4	0.004	:	0.1	:	:	:	:	:
2263	HD $192263 b$	24.356	51.9	0.055	200	10994.3	0.641	0.153	7.7	31	Butler 200
		0.005	2.6	0.039	:	3.9	:	:	:	:	:
0277	HD 210277 b	442.19	38.9	0.476	119	10104.3	1.29	1.14	3.8	69	Butler 200
		0.50	0.8	0.017	ი	2.6	:	:	:	:	:
7014	51 Peg b	4.231	55.9	0.013	58	10001.5	0.472	0.053	7	256	Butler 200
		0.000	0.7	0.012	:	0.6	:	:	:	:	:
17107	HD 217107 b	7.127	139.7	0.129	20	:	1.4	0.075	13	297	Vogt 2005
		0.000	1.0	0.006	60	:	:	:	:	:	:
17107	HD $217107 c^a$	4070	34.7	0.529	200	11133	2.46	5.15	13	297	Vogt 2005
		140	-	1000	007	00					

letails.
further o
for
8 7.4
See
companion.
lar e
stel
be a
fact
н
may
"planet"
This
-

A kind of synthesis, but with some elements that perhaps you wouldn't have expected in advance. I always like that when that happens, when something comes that is more than the sum of the parts.

— Evan Parker

-7 -

Synthesis of Results

Having explored the many techniques for identifying stellar and substellar companions in the previous chapters, it is now time to consolidate the results and present the observed multiplicity of each system, which is the main content of this chapter. After presenting the results, I discuss specific notes for certain systems which need an explanation about confirmed, candidate, or refuted companions.

7.1 Nomenclature

The stars of the current sample have been the subjects of extensive monitoring with many observational techniques, resulting in a comprehensive assessment of stellar, brown dwarf, and planetary companions. These components are named by their discoverers using historical and evolving standards, and as such, there is some confusion when it comes to the names of components in specific systems. Much effort has gone into standardizing the nomenclature, culminating in the adoption of the Washington Multiplicity Catalog¹ (WMC) standards by the IAU via a resolution of C Type for Commissions 5 and 26 in Special Session 3 of the XXVth General Assembly in Sydney, Australia, in 2003. In this work, I follow the WMC standards for stellar and brown dwarf companions as prescribed in Hartkopf & Mason (2004), my adaptation of which is described in the following paragraph. While the IAU

 $^{^{1}}http://ad.usno.navy.mil/wds/wmc.html$

pronouncement applied to companions of all types, a separate de-facto standard has evolved for naming planetary companions, including some brown dwarf companions discovered by the planet search teams. This method attaches a lowercase alphabetic suffix, separated by a blank, to the host star's name for each substellar companion, starting with "b" for the first discovery and incremented alphabetically for each subsequent discovery. It is this standard that is used in the two primary online catalogs mentioned in § 6.3, and followed in the the large volume of work ongoing in this field, thus becoming the widely-accepted standard. Reassigning names to planets according to the WMC prescription at this stage will cause too much confusion, so I follow the lowercase letters as described above for the planetary companions.

The WMC standard uses a hierarchical approach to naming the components of multiple systems, using a combination of uppercase and lowercase alphabets, and in the case of exceptionally complex systems, numbers as well. It is best understood by following the illustration of a fictitious example in Figure 7.1, which is adapted from a similar example in Hartkopf & Mason (2004) and described in the figure's caption. When reviewing the example, it will help to keep one point in mind. The study of binary stars, like all scientific endeavors, is an evolving process. We will hence encounter components discovered and assigned to systems that were later realized to be unrelated stars. As seen in §4.2, the WDS, which is the current comprehensive catalog of visual double stars and the predecessor to the WMC still under construction, lists many double star entries which are optical rather than physical. When a component is understood to not belong to a system, what should

one do about the name it has already been assigned? For example, if component B of a three-component system ABC is found to be optical and component C is determined to be physically associated to A, should C be renamed B? No, because this would cause too much confusion with ever-changing component names. Therefore, we will have situations where, as in the above example, AC represents a binary system, while component B is only optically associated. The designations used here follow the WMC standards, but inherit the actual WDS designations, when available, with only minor cosmetic tweaks to conform to the WMC standards. For systems which have only two components in the WDS, the pair ID is left blank as there is only one, which is meant to imply AB (W. Hartkopf 2009, private communication), and used as such here. It is also relevant to note that the WDS is not a comprehensive multiplicity catalog and only includes pairs with at least one visual measure with separation and position angle measures, but the WMC catalog will contain companions of all kind. The results presented here follow the WMC standards for all stellar and brown dwarf companions, and the nomenclature described in the previous paragraph for planetary components.

7.2 Results for Each of the 454 Systems

Table 7.1 summarizes the stellar and planetary companions to each star in the sample studied in this work. Note that this table only includes confirmed or candidate companions, leaving out components that are now known to be physically unassociated with these systems. Such components are discussed in the preceding chapters and in \S 7.4. The stars of the current



FIGURE 7.1: Illustration of companion nomenclature according to the WMC standards for a fictitious system that grows more complex over time. The following hypothetical events took place as components of this system were discovered. 1850: A telescopic inspection of this star reveals three visual components. The companions are resolved 3" and 12" away, and labeled B and C, respectively, while the primary is designated as A. 1930: A wide CPM companion is discovered and labeled D. 1975: Component C is discovered to be a spectroscopic binary and the original designation is split into Ca and Cb for the new components, while C refers to the pair. 1978: A new measure of the AB pair shows that it is optical rather than physical, but the components names previously assigned are retained. 1985: Component D is split by speckle interferometry, and the components are named Da and Db. 1990: An additional speckle component is found in D and named Dc; 1998: AO reveals a faint source near A, which is identified as a brown dwarf companion by follow-up spectroscopy, splitting A into Aa, referring to the stellar primary, and Ab, identifying the brown dwarf. 2003: Another brown dwarf is discovered near Ca, so the star and brown dwarf are named Ca1 and Ca2 respectively.

sample are listed in order of right ascension, immediately followed in subsequent lines by

the stellar and planetary companions of that system. Note that a few stars in this sample

are actually not the primary component of their system, but are still listed first within their group. Stellar companions are listed in order of proximity and planetary companions are listed directly under the star they orbit, also sequenced by proximity to the star. For ease of readability, the right ascension (Column 1) and declination (Column 2) are only listed at the beginning of each group and correspond to the sample star. Column 3 lists the HD number of the star, when available, and Column 4 lists an alternate name of the star or the name of the planet. For the first line of each group, Column 5 identifies the component designation of the sample star according to the nomenclature described above, unless it has no companions, in which case the column is empty. For stellar companions, this column identifies the pair whose period and separation are listed in Columns 6 and 7 respectively, again according to the WMC standards. The components are separated by a comma, which is suppressed when two components are each an uppercase alphabet (e.g., AB, AB, C, and Aa, Ab are valid pair designations). For planetary companions, this column is left empty. The period is listed from the spectroscopic orbit, if one exists, otherwise from a visual orbit, if available, along with its unit – h for hours, d for days, and y for years. The separation is listed typically from the latest measure in the WDS, and in cases of no WDS measurements, from other published references, or is my approximate measure from the archival images for new CPM companions, and is listed in milli-arcseconds (m) or arcseconds (a) for angular measures, or AU (A) for linear separations. Column 8 lists the status of companionship used for the multiplicity statistics derived here – "Y" indicates a confirmed companion and "M" implies an unconfirmed candidate. The next five columns list additional information about the techniques used to identify each companion and are described below.

Column 9 corresponds to visual orbits and contains an "O" for robust orbits of grade 1, 2, 3, or 8, "P" for preliminary orbits of grade 4 or 5, and "U" for unresolved photocentricmotion orbits (see $\S5.2$ for details). Additionally, two of the LBI visual orbits from $\S3.2$ (HD 146361 and 223778) are counted as robust orbits, one (HD 45088) is included as preliminary, and the trial orbit for HD 8997 is not included due to its low quality. Column 10 identifies spectroscopic companions as a "1" for single-lined orbits, "2" for double-lined orbits, and "V" for radial velocity variations indicating a companion, but without an orbital solution (see Chapter 6). Column 11 identifies CPM companions as close pairs with matching proper motions ("M", see $\S5.3$), pairs with evidence of orbital motion ("O", see $\S5.3$), companions with matching proper motions and photometric distances ("P", see Tables 4.2 and 5.5), close pairs with published evidence of companionship ("R", see §7.4), companions with matching proper motions and spectral type identifications that are consistent with the primary's distance ("S", see §7.4), or pairs with independently-measured matching proper motions and trigonometric parallaxes ("T", see §5.3). Column 12 identifies unresolved companions other than spectroscopic or visual ones which were covered in Columns 9 and 10 as eclipsing binaries ("E", see $\S6.4$), companions indicated by an overluminous star ("L", see $\S7.4$), or implied by proper motion accelerations ("M", see $\S5.1.2$). Finally, Column 13 identifies companions seen by CHARA LBI as SFP ("S", see $\S 3.1$), or as visibility-modulation binaries ("V", see § 3.2).
TABLE 7.1: Stellar and Planetary Companions

R.A. (J2000.0) (1)	Decl. (J2000.0) (2)	HD Name (3)	Other Name (4)	Comp ID (5)	Period (6)	${f Sep}\ (7)$	Sts (8)	VB (9)	SB (10)	CP (11)	UR (12)	CH (13)
			Sun	А								
			Mercury		0.24 y	0.39 A	Υ					
			Venus		0.62 y	0.72 A	Υ					
			Earth		1.00 y	1.00 A	Y					
			Mars		1.88 y	1.52 A	Y					
			Jupiter		11.86 y	5.20 A	Y					
			Jaturn		29.42 y 84.01 y	9.54 A	Y V					
			Neptune		164.8 v	30.06 A	Y					
00.02.10.16	+27 04 56.1	224930	85 Peg	A	10110 9							
				AB	26.31 y	0.8 a	Y	0	1		Μ	
$00 \ 06 \ 15.81$	$+58 \ 26 \ 12.2$	000123	HIP 000518	Α								
				AB	106.7 y	1.46 a	Y	0			Μ	
				$_{\rm Ba,Bb}$	47.69 d		Y		1			
00 06 36.78	+29 01 17.4	000166	HIP 000544									
00 12 50.25	-57 54 45.4	000870	HIP 001031									
00 10 12.08	-79 51 04.5	001257	CI 3021 b	A	133 82 d	0.49.4	v		1			
			HD 1237B	AB	135.82 u	3.87 a	v		1	S		
00 16 53.89	-52 39 04.1	001273	HIP 001349	A		0.01 u						
				Aa,Ab	1.13 y		Y	U	1			
$00\ 18\ 41.87$	$-08 \ 03 \ 10.8$	001461	HIP 001499		"							
$00\ 20\ 00.41$	+38 13 38.6	001562	HIP 001598									
$00 \ 20 \ 04.26$	$-64\ 52\ 29.2$	001581	ζ Tuc									
00 22 51.79	$-12\ 12\ 34.0$	001835	HIP 001803									
00 24 25.93	-27 01 36.4	002025	HIP 001936									
00 25 45.07	-77 15 15.3	002151	β Hy1 12 Cot					• • •				
00 55 14.88	-03 33 34.2	003190	15 Cet	A A A A A	2.08 d		v		1			
				AB	6.92 v	0.28 a	Y	0	2		M	S
00 37 20.70	-24 46 02.2	003443	HIP 002941	A	0.02 9	0.20 u			-			
			GJ 25 B	AB	25.09 v	0.74 a	Y	0	2		Μ	
$00 \ 39 \ 21.81$	+21 15 01.7	003651	54 Psc	Α								
			HD 003651 b		62.23 d	0.28 A	Υ		1			
			HD 3651B	AB		$43.07 \ a$	Y			S		
$00 \ 40 \ 49.27$	$+40\ 11\ 13.8$	003765	HIP 003206	• • •								
$00 \ 44 \ 39.27$	$-65 \ 38 \ 58.3$	004308	HIP 003497	А								
		004956	HD 004308 b		15.56 d	0.11 A	Ŷ		1			
$00 \ 45 \ 04.89$ $00 \ 45 \ 45 \ 59$	+01 47 07.9 -47 33 07 2	004256	HIP 003535 HIP 003583	Δ			• • •					
00 45 45.59	-47 33 07.2	004391	1111 003585	AB		1662	v					
				AC		49.0 a	Ŷ			P		
00 48 22.98	$+05\ 16\ 50.2$	004628	HIP 003765	A								
				Aa,Ab		2.7 a	Μ			R		
$00 \ 48 \ 58.71$	+16 56 26.3	004676	64 Psc	À								
				Aa,Ab	13.82 d	6.53 m	Y	0	2			S
$00 \ 49 \ 06.29$	+57 48 54.7	004614	η Cas	A								
			LHS 122	AB	480 y	12.49 a	Y	0			• • •	
$00 \ 49 \ 26.77$	$-23 \ 12 \ 44.9$	004747	HIP 003850	A			· · · ·					
	1 70 26 58 1	004625	 HID 002876	Aa,Ab	18.7 y		Ŷ		1		1/1	
00 49 40.48	$-10\ 20\ 38\ 30\ 6$	004033	HIP 003000									
$00\ 50\ 07.59$ $00\ 51\ 10.85$	$-05\ 02\ 21.4$	004915	HIP 003979									
00 53 01.13	$-30\ 21\ 24.9$	005133	HIP 004148									
00 53 04.20	$+61 \ 07 \ 26.3$	005015	HIP 004151									
$01 \ 08 \ 16.39$	+54 55 13.2	006582	μ Cas	Α								
			μ Cas B	Aa,Ab	21.43 y	0.8 a	Υ	Р	1			
01 15 00.99	$-68 \ 49 \ 08.1$	007693	HIP 005842	C^a								
•••			GJ 55.1B		85.2 y	0.91 a	Y	0		· · · · T	М	
			HD 7788 GI 55 2P	A,CD	857	318 a	Y V	 Р		T.		
01 15 11 19	-45 31 54 0	007570	v Phe	AD	697 y	ыс	I	r				
01 16 29 25	+4256219	007590	HIP 005944									
$01 \ 21 \ 59.12$	+76 42 37.0	007924	HIP 006379									
01 29 04.90	+21 43 23.4	008997	HIP 006917	A								
				Aa,Ab	$10.98 \ d$		Υ		2			V
$01 \ 33 \ 15.81$	-24 10 40.7	009540	HIP 007235									
01 34 33.26	+68 56 53.3	009407	HIP 007339									
$01 \ 35 \ 01.01$	-29 54 37.2	009770	HIP 007372	A								
				AB	4.56 y	0.18 a	Y	0				
				Ba,Bb	11.50 h	0.01 A	Y	 P			E	
01 36 47 84	$+41 \ 24 \ 10 \ 7$	009826	u And	AB,C	111.8 y	1.7 a	Ŷ	Р				
01 00 41.04	⊤41 24 19.7	003620	v And b	A	4.62 d	0.06 A	v		1			
			v And c		241.52 d	0.83 A	Ý		1			
			v And d		3.49 v	2.51 A	Ŷ		1			
			v And B	AD		55 a	Υ			S		
$01 \ 37 \ 35.47$	$-06 \ 45 \ 37.5$	010008	HIP 007576									
01 39 36.02	+45 52 40.0	010086	HIP 007734									
Continued or	Novt Page	-				-						

Continued on Next Page...

Table 7.1 - Continued

R.A.	Decl.	HD	Other	Comp								
(J2000.0)	(J2000.0)	Name	Name	ID	Period	\mathbf{Sep}	\mathbf{Sts}	VB	\mathbf{SB}	\mathbf{CP}	\mathbf{UR}	\mathbf{CH}
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)
01 39 47 54	-56 11 47 0	010360	HIP 007751	\mathbf{P}^{a}								
01 39 47.34	-30 11 47.0	010300	UD 10261		192 66	11.4 0	v	 р			 M	
		010207	HD 10301	AD	485.00 y	11.4 a	I	Г			1/1	
$01 \ 41 \ 47.14$	$+42 \ 30 \ 48.1$	010307	HIP 007918	A				· · · ·				
				Aa,Ab	20.21 y	0.3 a	Y	Р	1		M	
$01 \ 42 \ 29.32$	-53 44 27.0	010647	q^1 Eri	A								
			HD 010647 b		2.85 y	2.1 a	Y		1			
$01 \ 42 \ 29.76$	$+20\ 16\ 06.6$	010476	107 Psc									
$01 \ 44 \ 04.08$	$-15\ 56\ 14.9$	010700	τ Cet									
01 47 44.83	$+63\ 51\ 09.0$	010780	HIP 008362									
01 59 06 63	+33 12 34 9	012051	HIP 009269									
02 06 30 24	+24 20 02 4	012846	HIP 009829									
02 10 25 02	50 40 25 4	012040	UID 010129									
02 10 25.95	-30 49 23.4	013443		A	15 77 -	0.11.4	· · · · V					
			GI 60 D		15.77 0	0.11 A	1	· · · ·	1			
			GJ 86 B	AB	69.7 y	1.97 a	Ŷ	Р			M	
$02 \ 17 \ 03.23$	+34 13 27.2	013974	ð Tri	. A								
				Aa,Ab	10.02 d		Y	0	2			
$02 \ 18 \ 01.44$	$+01 \ 45 \ 28.1$	014214	HIP 010723	A								
				Aa,Ab	93.29 d		Y	U	1			
$02 \ 18 \ 58.50$	$-25\ 56\ 44.5$	014412	HIP 010798									
$02 \ 22 \ 32.55$	-23 48 58.8	014802	κ For	Α								
				AB	26.5 v	0.5 a	Y	U		R	Μ	
02 36 04 89	± 0653127	016160	HIP 012114	Δ	_0.0 5							
02 00 04.00	100 00 12.1	010100	1111 012114	AaAb	61 v	329	v	11		 В		
			NI TT 9455	As P	01 y	164 a	v	U		D		
02 26 41 76	02 00 00 1	016997	NL11 6400	Аа, Б		104 a	I			г		
02 30 41.70	$-03\ 09\ 22.1$	016287	HIP 012158	A								
				Aa,Ab	14.84 d	0.01 a	Y		1			
$02 \ 40 \ 12.42$	$-09\ 27\ 10.3$	016673	HIP 012444	A								
				Aa,Ab			Y		V			
$02 \ 41 \ 14.00$	$-00 \ 41 \ 44.4$	016765	HIP 012530	А								
				AB		3.3 a	Y		V	0		
$02 \ 42 \ 14.92$	$+40\ 11\ 38.2$	016739	12 Per	Α								
				Aa.Ab	330.99 d	0.1 a	Y	0	2			S
$02 \ 42 \ 33.47$	-50 48 01.1	017051	4 Hor	A								
			HB 810 b		311 29 d	0.91 A	Y		1			
02 44 11 99	$\pm 49 13 42 4$	016895	A Per	Δ	0111 2 0 u	0.01 11			-			
02 44 11.00	15 10 12.1	010050	NI TT 9797	AD	2720	10.6 a	v	 Ъ				
02 48 00 14	197.04.07.1	017299	NLT 1 0/07	AD	2720 y	19.0 a	1	г				
02 48 09.14	+270407.1	017362	HIF 015081	A	15.07		· · · ·					
				Aa,Ab	15.27 y		Y V	U	1	· · · ·	IVI	
			NLTT 8996	AB		20.7 a	Ŷ			Р		
$02 \ 52 \ 32.13$	$-12 \ 46 \ 11.0$	017925	HIP 013402									
02 55 39.06	$+26\ 52\ 23.6$	018143	HIP 013642	A								
			HD 18143 B	AB		5 a	Y	Р			M	
			NLTT 9303	AC		44 a	Y			т		
$03 \ 00 \ 02.81$	+07 44 59.1	018632	HIP 013976									
$03 \ 02 \ 26.03$	$+26\ 36\ 33.3$	018803	51 Ari									
03 04 09 64	+61 42 21 0	018757	HIP 014286	А								
00 01 00.01	101 12 21:0	010101	NLTT 0726	AC		263.2 2	v			P		
02 00 04 02	140 26 47 8	010272	, Por	AO		200.2 a	1			1		
03 09 04.02	$\pm 49 30 47.8$	019373										
$03\ 12\ 04.53$	-28 59 15.4	020010	12 Eri	A				· · · ·				
			GJ 127 B	AB	269 y	4.8 a	Ŷ	Р				
				$_{\rm Ba,Bb}$			Μ		V			
$03 \ 12 \ 46.44$	$-01 \ 11 \ 46.0$	019994	94 Cet	А								
			HD 019994 b		1.2 y	1.3 a	Y		1			
				AB	1420 y	2.5 a	Y	Р				
$03 \ 14 \ 47.23$	+08 58 50.9	020165	HIP 015099									
$03 \ 15 \ 06.39$	$-45 \ 39 \ 53.4$	020407	HIP 015131									
03 18 12 82	-62 30 22 9	020807	ζ^2 Bet	А								
00 10 12.02	52 56 22.3	020001	HD 20766	AR		309.2 2	v			 т		
03 10 01 90	-02 50 25 5	020610	HIP 015449			005.2 a	1			T		
03 13 01.03	-02 00 00.0	020019										
03 19 21.70	+03 22 12.7	020030	K Cet				• • •					• • •
03 19 33.03	-45 04 11.2	020794	HID 015250				• • •					• • •
$03 \ 21 \ 54.76$	+52 19 53.4	232781	HIP 015673	• • •								
$03 \ 23 \ 35.26$	$-40 \ 04 \ 35.0$	021175	HIP 015799	A								
				AB	111 y	1.3 a	Y	Р				
$03 \ 32 \ 55.84$	$-09 \ 27 \ 29.7$	022049	ϵ Eri	A								
			ϵ Eri b		6.85 y	3.39 A	Υ		1			
$03 \ 36 \ 52.38$	+00 24 06.0	022484	10 Tau									
03 40 22.06	-03 13 01.1	022879	HIP 017147									
03 43 55.34	$-19\ 06\ 39.2$	023356	HIP 017420									
03 44 09 17	-38 16 54 4	023484	HIP 017430	Δ								
00 11 00.11	00 10 01.1	010101	011100	Aa Ab			M		v			
03 54 29 02	 ⊥16 36 57 9	024406	HIP 019967	Λ			141		v			
00 04 20.00	LT0 00 01.0	024430	1111 010207				v			 М		
0.2 55 0.2 0.4			 IIID 010204	AD		⊿.1 a	I			1/1		
03 55 03.84	+61 10 00.5	024238	HIP 018324									
03 56 11.52	+59 38 30.8	024409	HIP 018413	A					• • •			
				AD	34.57 y	0.4 a	Y		1	· · · ·	M	
				AE		8.9 a	Y			Р		
$04 \ 02 \ 36.74$	$-00\ 16\ 08.1$	025457	HIP 018859									
$04 \ 03 \ 15.00$	$+35 \ 16 \ 23.8$	025329	HIP 018915									
$04 \ 05 \ 20.26$	+22 00 32.1	025680	39 Tau	Α								
<u> </u>	<u>.</u>	<u>.</u>		Aa,Ab		0.4 a	Y			Μ	<u></u> .	<u>.</u>

-

Continued on Next Page...

TABLE 7.1 - Continued

R.A.	Decl.	HD	Other	\mathbf{Comp}								
(J2000.0)	(J2000.0)	Name	Name	ID	Period	Sep	Sts	VB	\mathbf{SB}	\mathbf{CP}	UR	CH
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)
04 07 21 54	-64 13 20 2	026491	HIP 019233	А								
04 01 21.04	04 10 20.2	020401	1111 010200	Aa Ab	19 19 v		Y		1		M	
04 08 26 62	1 28 02 22 0	025008	UID 010225	E ^b	10110 9				-			
04 08 30.02	$\pm 38 02 23.0$	025998	1111 019555	E2 Eb			 М				 M	
			HD 25893	AE		746 a	V			т	141	
			11D 20030	AB	590 v	24 a	v	 Р		1		
04 09 35 04	$\pm 69.32.29.0$	025665	HIP 019422	AD	550 y	2.4 a	1	1				
04 15 16 32	-07 39 10 3	026965	HIP 019849	A								
01 10 10.02	01 00 1010	020000	LHS 25	ABC		83 a	Y			т		
			HD 26976	BC	252.1 v	69a	Ŷ	 Р		1		
04 15 28.80	$+06\ 11\ 12.7$	026923	HIP 019859	A								
01 10 20.00	100 11 12.1	020020	HD 26913	AB		64.5 a	Y			т		
04 43 35.44	+27 41 14.6	029883	HIP 021988									
04 45 38 58	-50.04.27.2	030501	HIP 022122									
04 47 36.29	-165604.0	030495	HIP 022263									
04 49 52 33	-35,06,27,5	030876	HIP 022451									
05 02 17.06	$-56\ 04\ 49.9$	032778	HIP 023437	A								
00 02 11100	00 01 1010	002110	NLTT 14447	AB		79.1 a	v			P		
05 05 30 66	-57 28 21 7	033262	(Dor	nD		10.1 a	1			1		
05 06 42 22	± 14 26 46 4	032850	HIP 023786	Δ								
00 00 42.22	111 20 10.1	002000	1111 020100	AaAb	205.68 d		v	 11	1			
05 07 27 01	 ⊥18.38.42.2	032023	104 Tau	ла,по	200.08 u		1	0	1			
05 18 50 47	18 07 48 2	024721	UID 024786									
05 10 08 47	-18 07 48.2	024411	HID 024780									
05 19 08.47	+40 03 30.0	022564	HIP 024813									
00 22 00.00	± 79 15 52.1	055504	HIF 025110	A	1.06	1.1	· · · · V					
05 22 27 40	100.26.11 5	025119	HD 055504 D		1.00 y	1.1 a	ĩ		1			
05 22 57.49	± 02 50 11.5	055112	HIF 025119	A		1.0	· · · ·	 D				
05 94 95 46		025206	 IIID 095979	AB	93 y	1.2 a	Ŷ	Р			IVI	
05 24 25.40	+17 23 00.7	035290	HIF 025276	A		707.0	· · · ·					
05 06 14 74		095054	HD 35171	AC		707.2 a	Ŷ			1		
05 26 14.74	-32 30 17.2	035854	HIP 025421									
05 27 39.35	-60 24 57.6	036435	HIP 025544									
$05\ 28\ 44.83$	$-65\ 26\ 54.9$	036705	AB Dor	A								
				Aa,Ab		0.16 a	Y			S	M	
				AB		9.2 a	Y		• • •	P		• • •
				Ba,Bb		0.07 a	Y			R		
05 36 56.85	$-47\ 57\ 52.9$	037572	UY Pic	A						· · · ·		
			HIP 26369	AB		18.3 a	Y			Т		
$05 \ 37 \ 09.89$	$-80\ 28\ 08.8$	039091	π Men	A								
			HD 039091 b		5.65 y	3.29 A	Y		1			
$05 \ 38 \ 11.86$	$+51 \ 26 \ 44.7$	037008	HIP 026505	• • •								
$05 \ 41 \ 20.34$	$+53 \ 28 \ 51.8$	037394	HIP 026779	А								
			HD 233153	AB		98.8 a	Y			Т		
$05 \ 46 \ 01.89$	+37 17 04.7	038230	HIP 027207									
$05 \ 48 \ 34.94$	$-04 \ 05 \ 40.7$	038858	HIP 027435									
$05 \ 54 \ 04.24$	$-60\ 01\ 24.5$	040307	HIP 027887	A								
			HD 040307 b		4.31 d	0.05 A	Y		1			
			HD 040307 c		9.62 d	0.08 A	Y		1			
			HD 040307 d		20.46 d	0.13 A	Y		1			
$05 \ 54 \ 22.98$	$+20 \ 16 \ 34.2$	039587	χ^1 Ori	Α								
				Aa,Ab	14.06 y	0.5 a	Y	U	1	R	Μ	
05 54 30.16	$-19 \ 42 \ 15.7$	039855	HIP 027922	À								
			BD-19 1297B	AB		10.6 a	Y			Р		
05 58 21.54	-04 39 02.4	040397	HIP 028267	A								
				AB		3.9 a	Y			Μ	Μ	
			NLTT 15867	AD		89.3 a	Y			Р		
$06 \ 06 \ 40.48$	$+15 \ 32 \ 31.6$	041593	HIP 028954									
$06 \ 10 \ 14.47$	-74 45 11.0	043834	α Men	А								
				Aa,Ab		3.05 a	Υ		V	Μ		
$06\ 12\ 00.57$	$+06 \ 46 \ 59.1$	042618	HIP 029432									
$06 \ 13 \ 12.50$	$+10 \ 37 \ 37.7$	042807	HIP 029525									
$06\ 13\ 45.30$	-23 51 43.0	043162	HIP 029568	Α								
				AB		164 a	Y			Р		
$06\ 17\ 16.14$	$+05 \ 06 \ 00.4$	043587	HIP 029860	А								
				Aa,Ab	33.74 y	0.7 a	Y		1		Μ	
			NLTT 16333	AE		103.1 a	Y			Р		
$06 \ 22 \ 30.94$	$-60\ 13\ 07.2$	045270	HIP 030314	A								
				AB		16.2 a	Μ			R		
$06 \ 24 \ 43.88$	-28 46 48.4	045184	HIP 030503									
06 26 10.25	$+18 \ 45 \ 24.8$	045088	OU Gem	А								
				Aa,Ab	6.99 d		Y	Р	2			V
				AB	600 v	2.4 a	Y	Р			М	
06 38 00.36	$-61 \ 32 \ 00.2$	048189	HIP 031711	А								
				AB		0.3 a	Y			Μ	М	
$06 \ 46 \ 05.05$	$+32 \ 33 \ 20.4$	263175	HIP 032423	А								
			HD 263175B	AB		30 a	Υ			Р		
$06 \ 46 \ 14.15$	$+79 \ 33 \ 53.3$	046588	HIP 032439									
$06 \ 46 \ 44.34$	+43 34 38.7	048682	ψ^5 Aur									
06 55 18.67	+25 22 32.5	050692	HIP 033277									
06 58 11.75	+22 28 33.2	051419	HIP 033537									
06 59 59.66	$-61\ 20\ 10.3$	053143	HIP 033690									
Continued on	Next Page											
Jonnaca Oli												

TABLE 7.1 - Continued

R.A.	Decl.	HD	Other	Comp								
(J2000.0)	(J2000.0)	Name	Name	ID	Period	\mathbf{Sep}	Sts	VB	\mathbf{SB}	\mathbf{CP}	UR	CH
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)
07 01 13.74	$-25\ 56\ 55.4$	052698	HIP 033817	А								
				Aa,Ab			Υ		V		Μ	
$07 \ 01 \ 38.59$	+48 22 43.2	051866	HIP 033852									
$07 \ 03 \ 30.46$	$+29 \ 20 \ 13.5$	052711	HIP 034017									
$07 \ 03 \ 57.32$	$-43 \ 36 \ 28.9$	053705	HIP 034065	A								
			HD 53706	AB		20.9 a	Y			Т		
			HD 53680	\mathbf{AC}		184.9 a	Y			Т		
				$_{\rm Ca,Cb}$			Y				M	
$07 \ 08 \ 04.24$	+29 50 04.2	053927	HIP 034414	• • •								
07 09 35.39	+25 43 43.1	054371	HIP 034567	A								
				Aa,Ab	32.81 d		Ŷ	• • •	1			
07 15 50.14	+47 14 23.9	055575	HIP 035130									
07 17 29.56	$-40\ 58\ 45.3$	057095	HIP 035296	A			· · · · V	 D				
07 07 05 47	51 24 00 4	050468	LID 026210	AD	94 y	0.8 a	I	Г				
07 20 01 77	-312403.4 ± 3150378	055408	HIP 036357	\mathbf{F}^{a}								
01 20 01.11	101 00 01.0		HD 58946	AE		756 1 a	Y			т		
			GJ 274B	AB		3.4 a	Ŷ			ō	M	
07 30 42.51	$-37\ 20\ 21.7$	059967	HIP 036515									
07 33 00.58	$+37\ 01\ 47.4$	059747	HIP 036704									
$07 \ 34 \ 26.17$	$-06\ 53\ 48.0$	060491	HIP 036827									
07 39 59.33	$-03 \ 35 \ 51.0$	061606	HIP 037349	A								
			NLTT 18260	AB		58.3 a	Υ			Р		
$07 \ 45 \ 35.02$	$-34 \ 10 \ 20.5$	063077	HIP 037853	Α								
				Aa,Ab			Y		V		Μ	
			NLTT 18414	AB		914 a	Y			Р		
$07 \ 49 \ 55.06$	$+27 \ 21 \ 47.4$	063433	HIP 038228									
$07 \ 51 \ 46.30$	-13 53 52.9	064096	9 Pup	Α								
				AB	22.7 y	0.3 a	Y	0	2		M	
07 54 34.18	$-01 \ 24 \ 44.1$	064606	HIP 038625	A					• • •			
				Aa,Ab	1.23 y		Y		1		Μ	
07 54 54.07	+19 14 10.8	064468	HIP 038657	A				• • •				
				Aa,Ab	161.2 d		Ŷ		1			• • •
07 50 17.23	+80 15 55.9	062013	HIP 038784									
07 57 40.91	-00 18 11.1	065907	LUS 1060	A		60.2 5	· · · · V			 р		
			L115 1900	BC		00.3 a	v			M		
07 59 33 93	±20 50 38 0	065430	HIP 039064	A		5 4	1			141		
01 03 00.30	± 20 50 50.0	000430	1111 055004	AaAb	8 59 v		v		1		M	
08 00 32 13	+29 12 44 5	065583	HIP 039157	ла,по	0.03 y		1		1		101	
08 02 31 19	$-66\ 01\ 15\ 4$	067199	HIP 039342	A								
				Aa.Ab			M				M	
$08 \ 07 \ 45.86$	+21 34 54.5	067228	μ Cnc									
$08 \ 11 \ 38.64$	+32 27 25.7	068017	HIP 040118	А								
				Aa,Ab			Μ				Μ	
$08 \ 12 \ 12.73$	$+17 \ 38 \ 52.0$	068257	ζ Cnc C	À								
			HD 68255	AB	59.58 y	1 a	Y	0			Μ	
			HD 68256	AB,C	1115 y	6.2 a	Υ	Р			Μ	
				$_{\rm Ca,Cb}$	17.25 y	0.3 a	Υ	Р	1			
				Cb1, Cb2			Y				\mathbf{L}	
				Cb1, Cb3			Μ			R		
$08 \ 18 \ 23.95$	$-12 \ 37 \ 55.8$	069830	HIP 040693	A								
			HD 069830 b		8.67 d	0.08 A	Y		1			
			HD 069830 c		31.56 d	0.19 A	Y		1			
			HD 069830 d		197 d	0.63 A	Ŷ	• • •	1			
08 19 19.05	+01 20 19.9	071149	HIP 040774									
08 32 51 50.79	-31 30 02 1 -31 30 02 1	071148	HIP 041484									
08 34 31 65	-00 43 33 8	072760	HIP 041920	Δ								
00 04 01.00	-00 43 33.8	012100	1111 042074	Aa. Ab		0.96 a	v			в	м	
08 37 50 29	-06 48 24 8	073350	HIP 042333	110,110		0.50 a	-			10	101	
08 39 07.90	-22 39 42.8	073752	HIP 042430	A								
				AB	123 v	1.3 a	Y	0				
08 39 11.70	+65 01 15.3	072905	π^1 UMa	-	,							
08 39 50.79	+11 31 21.6	073667	HIP 042499									
$08 \ 42 \ 07.52$	-425546.0	074385	HIP 042697	A								
			NLTT 20102	AB		45 a	Υ			Р		
$08 \ 43 \ 18.03$	-38 52 56.6	074576	HIP 042808									
$08 \ 52 \ 16.39$	+08 03 46.5	075767	HIP 043557	А								
				Aa,Ab	$10.25 \ d$		Υ		1			
				AB		3.4 a	Υ			Μ		
				$_{\rm Ba,Bb}$			Y		2			
$08 \ 52 \ 35.81$	+28 19 50.9	075732	55 Cnc	Α								
			55 Cnc b		14.65 d	0.12 A	Y		1			
			55 Cnc c		44.34 d	0.24 A	Y		1			
			55 Cnc d		14.29 y	5.77 A	Y		1			
			oo Unc e		2.82 d	0.04 A	Y		1		• • •	
			JUG DOG2	 A D	200 a	0.18 A	r V		T	 P		
08 54 17 05	-05 26 04 1	076151	HIP 043796	AD		04.1 a	I	• • • •		1.	• • •	
08 58 43 93	$-16\ 07\ 57\ 8$	076932	HIP 044075									
0.0040.00	10 01 01.0	010304	1111 044010									

Continued on Next Page...

TABLE 7.1 - Continued

R.A.	Decl.	HD	Other	\mathbf{Comp}								
(J2000.0)	(J2000.0)	Name	Name	ID (5)	Period	Sep	Sts	VB	SB	CP	UR	CH
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)
$09 \ 08 \ 51.07$	+33 52 56.0	078366	HIP 044897									
$09\ 12\ 17.55$	+14 59 45.7	079096	81 Cnc	Α								
				Aa,Ab	2.7 y	0.1 a	Y	0	2	· · ·		\mathbf{S}
			GI 337C	AE		43 a	Y			P		
		070028	 IIID 045222	Ea,ED		0.53 a	Ŷ			IVI		• • •
$09\ 14\ 20.34$	$\pm 01 20 25.9$	079028	HIF 040505	A A A A A	16.24 d		v		1			
$09\ 17\ 53\ 46$	+28 33 37 9	079969	HIP 045617	A	10.24 u		1		1			
05 11 00.40	1 20 00 01.0	015505	1111 040017	AB	34.17 v	0.8 a	Y	0			M	
$09 \ 22 \ 25.95$	$+40\ 12\ 03.8$	080715	HIP 045963	A								
				Aa,Ab	3.8 d		Υ		2			
$09 \ 30 \ 28.09$	$-32 \ 06 \ 12.2$	082342	HIP 046626	Α								
				AB		12 a	Y			Р		
$09 \ 32 \ 25.57$	-11 11 04.7	082558	HIP 046816									
$09 \ 32 \ 43.76$	$+26\ 59\ 18.7$	082443	HIP 046843	A		CE 0 .				 D		
00 25 20 50	125 49 26 5	092995	NLI I 22015 HID 047080	AB		65.2 a	Ŷ			Р		
09 33 39.30	$\pm 30.48 \ 30.3$	082885	1111 047080	AB	201 v	58 a	v	 Р			 М	
$09 \ 42 \ 14.42$	-235456.1	084117	HIP 047592		201 y	0.0 a						
09 48 35.37	$+46\ 01\ 15.6$	084737	HIP 048113									
$10 \ 01 \ 00.66$	+31 55 25.2	086728	HIP 049081	А								
			GJ 376 B	AB		133 a	Υ			Р		
				$_{\rm Ba,Bb}$			Μ				\mathbf{L}	
$10 \ 04 \ 37.66$	$-11 \ 43 \ 46.9$	087424	HIP 049366									
$10\ 08\ 43.14$	+34 14 32.1	087883	HIP 049699									
10 13 24.73	$-33\ 01\ 54.2$	088742	HIP 050075									
10 17 14.54	$+23 \ 06 \ 22.4$	089125	HIP 050384	A			· · · · V			 D		
10 19 51 05		080260	UID OFOFOF	AD		1.1 a	1			г		
$10\ 18\ 51.95$ $10\ 23\ 55\ 27$	-29 38 43 9	089209	HIP 050921									
10 28 03.88	+48 47 05.6	090508	HIP 051248	A								
			LHS 2266	AB	765 v	4.7 a	Y	Р				
$10 \ 30 \ 37.58$	+55 58 49.9	090839	HIP 051459	А								
			HD 237903	AB		122.5 a	Υ			Т		
				$_{\rm Ba,Bb}$			Μ		V			
$10 \ 31 \ 21.82$	-53 42 55.7	091324	HIP 051523									
10 35 11.27	+84 23 57.6	090343	HIP 051819									
10 36 32.38	-12 13 48.4	091889	HIP 051933									
10 42 13.32	-13 47 15.8	092719	HIP 052369									
10 45 28.27	$-29\ 03\ 51.4$ $\pm 07\ 23\ 18\ 5$	092945	HIP 052402									
$10\ 50\ 30.80$ $10\ 59\ 27\ 97$	+07 25 18.3 +40 25 48 9	095128	47 UMa	Α								
10 00 2000	1 10 20 10:0	000120	47 UMa b		2.97 v	2.11 A	Y		1			
			47 UMa c		6.00 v	3.39 A	Ŷ		1			
$11 \ 04 \ 41.47$	$-04 \ 13 \ 15.9$	096064	HIP 054155	А								
			NLTT 26194	$^{\rm A,BC}$		11.8 a	Υ			Р		
			BD-033040C	BC	23.23 y	0.3 a	Y	0	V			
$11 \ 08 \ 14.01$	$+38 \ 25 \ 35.9$	096612	HIP 054426									
11 12 01.19	$-26\ 08\ 12.0$	097343	HIP 054704	· · ·								
11 12 32.35	+35 48 50.7	097334	HIP 054745	A						· · ·		
			GI 417B	AE Fa Fb		89.7 a	Y		• • •	K M		
	± 25 42 37 4	097658	HIP 054006	Ea,ED		0.1 a	1		• • •	1/1		
11 14 33.10	+20 42 37.4 +31 31 45 7	098230	£ UMa B	\mathbf{B}^{a}								
11 10 10.55	101 01 40.1	000200	ς Oma D	Ba.Bb	3.98 d	26 m	Y		1			
			HD 98231	AB	59.88 v	1.6 a	Ŷ	0				S
				Aa,Ab	1.84 y		Y	U	1			
$11\ 18\ 22.01$	$-05 \ 04 \ 02.3$	098281	HIP 055210									
$11\ 26\ 45.32$	$+03 \ 00 \ 47.2$	099491	83 Leo	A		•••				· <u>· ·</u>		
			HD 99492	AB	32000 y	28 a	Y		• • •	Т		
11 21 44 05		100190	нр 099492 Б		17.04 d	0.12 A	Ŷ		1			
11 31 44.95	+14 21 52.2	100180	88 Leo	A			 M			 D		•••
			NLTT 27656	AR		15.3 a	V			P		
11 34 29.49	-32 49 52.8	100623	HIP 056452	A		10.0 a						
11 01 20110	02 10 02.0	100020	LHS 309	AB		17 a	Y			M		
$11 \ 38 \ 44.90$	$+45 \ 06 \ 30.3$	101177	HIP 056809	А								
			LHS 2436	AB	2050 y	9.3 a	Υ	Р			Μ	
				$_{\rm Ba,Bb}$	23.54 d		Υ		2			
$11 \ 38 \ 59.72$	+42 19 43.7	101206	HIP 056829	A								
		101501		Aa,Ab	12.92 d		Υ		1			
11 41 03.02	+34 12 05.9	101501	61 UMa				• • •					
11 40 31.07	$-40\ 30\ 01.3$	102365	HIP 057443	A A D		22.0 0	· · · · V			 М		
 11 47 15 81	-30 17 11 4	102438	HIP 057507	AD		44.9 a	I			1/1		
11 41 10.01	00 17 11.4	102400		Aa Ab			м		v			
11 50 41.72	+01 45 53.0	102870	β Vir						•			
11 52 58.77	$+37 \ 43 \ 07.2$	103095	HIP 057939									
$11 \ 59 \ 10.01$	$-20\ 21\ 13.6$	104067	HIP 058451									
$12 \ 00 \ 44.45$	$-10\ 26\ 45.6$	104304	HIP 058576									
12 09 37.26	$+40\ 15\ 07.4$	105631	HIP 059280									
Continued on	Next Page											

TABLE 7.1 - Continued

R.A.	Decl.	HD	Other	Comp		~	<i>a</i> .		~ ~	6D		
(J2000.0)	(J2000.0)	Name (3)	Name (4)	ID (5)	Period (6)	Sep (7)	Sts (8)	VB (9)	SB (10)	CP (11)	(12)	CH (13)
12 20 50 14	152 04 25 8	108054	UID 061052	(0)	(0)	(.)	(0)	(0)	(10)	(11)	(1-)	(10)
$12 \ 30 \ 30.14$ $12 \ 33 \ 31.38$	-68 45 20.9	108934 109200	HIP 061291									
$12 \ 33 \ 44.54$	$+41 \ 21 \ 26.9$	109358	HIP 061317									
$12 \ 41 \ 44.52$	$+55 \ 43 \ 28.8$	110463	HIP 061946	А								
				Aa,Ab			Μ		V			
12 44 14.55	+51 45 33.5	110833	HIP 062145	A Ab	 271 17 d		· · · · V				 М	
12 44 59.41	$+39\ 16\ 44.1$	110897	HIP 062207	Aa,Ab	271.17 u		1		1		101	
12 45 14.41	$-57\ 21\ 28.8$	110810	HIP 062229									
$12 \ 48 \ 32.31$	$-15 \ 43 \ 10.1$	111312	HIP 062505	Α								
				Aa,Ab	2.68 y	0.1 a	Y		2	· <u>·</u> ·	Μ	
10 40 47 05		111205		Aa,B		2.7 a	М			R	М	
12 48 47.05	$+24 \ 50 \ 24.8$ $-09 \ 50 \ 02 \ 7$	111395	HIP 062523 HIP 063366	Δ				• • •			• • •	
12 33 01.30	-03 50 02.1	112756		Aa.Ab	103.17 d		Y		1			
				AB		0.8 a	Y			Μ		
12 59 32.78	+41 59 12.4	112914	HIP 063406	А								
				Aa,Ab	1.95 y		Y	U	1			
$13 \ 03 \ 49.66$	$-05 \ 09 \ 42.5$	113449	HIP 063742	A			· · · · V		• • •	 D	 M	
13 11 52 30	± 2752415	114710	G Com	Aa,Ab	231.23 d		Ŷ	U		к	IVI	
$13\ 11\ 02.03$ $13\ 12\ 03.18$	-37 48 10.9	114613	HIP 064408									
$13 \ 12 \ 43.79$	$-02 \ 15 \ 54.1$	114783	HIP 064457	А								
			HD 114783 b		1.37 y	1.2 a	Y		1			
13 13 52.23	$-45\ 11\ 08.9$	114853	HIP 064550									
$13 \ 15 \ 26.45$	-87 33 38.5	113283	HIP 064690	A			 М				 М	
13 16 46 52	$+09\ 25\ 27\ 0$	115383	59 Vir	Aa,AD			101				101	
13 16 51.05	+17 01 01.9	115404	HIP 064797	A								
			LHS 2714	AB	770 y	7.5 a	Υ	Р			Μ	
$13 \ 18 \ 24.31$	$-18 \ 18 \ 40.3$	115617	61 Vir									
$13 \ 23 \ 39.15$	+02 43 24.0	116442	HIP 065352	A					• • •			
12 25 45 52	156 59 12 9	116056	HD 110443 HID 065515	AB		20.2 a	Ŷ	• • •		1		
$13\ 25\ 45.55$ $13\ 25\ 59.86$	+63 15 40.6	117043	HIP 065530									
13 28 25.81	+13 46 43.6	117176	70 Vir	А								
			70 Vir b		116.69 d	0.48 A	Y		1			
$13 \ 41 \ 04.17$	$-34 \ 27 \ 51.0$	118972	HIP 066765									
13 41 13.40	+56 43 37.8	119332	HIP 066781					• • •	• • •			
15 47 15.74	+17 27 24.9	120130	τ Boo b	A	3 31 d	0.05 4	v		1			
			HD 120136B	AB	2000 v	2.8 a	Ý	P			M	
$13 \ 51 \ 20.33$	-24 23 25.3	120690	HIP 067620	Α								
				Aa,Ab	10.3 y		Υ		1		Μ	
13 51 40.40	$-57\ 26\ 08.4$	120559	HIP 067655									
$13 \ 52 \ 35.87$	$-50\ 55\ 18.3$	120780	HIP 067742	A			· · · · V				 М	
				AB		 58 a	Y			м	101	
13 54 41.08	$+18\ 23\ 51.8$	121370	η Boo	A								
			, 	Aa,Ab	1.34 y		Υ	U	1			
$13 \ 55 \ 49.99$	$+14 \ 03 \ 23.4$	121560	HIP 068030									
$14 \ 03 \ 32.35$	$+10\ 47\ 12.4$	122742	HIP 068682	A								
	-12 36 42 4	124106	HIP 060357	Aa,Ab	9.9 y		Ŷ	U	1		IVI	
14 11 40.17 14 12 45.24	-03 19 12.3	124292	HIP 069414									
14 15 38.68	$-45\ 00\ 02.7$	124580	HIP 069671									
$14 \ 16 \ 00.87$	$-06 \ 00 \ 02.0$	124850	ι Vir	Α								
				Aa,Ab	55 y		Μ	U				
14 19 00.90	-25 48 55.5	125276	HIP 069965	A Ab			 M				 M	
14 19 34.86	-05 09 04.3	125455	HIP 070016	A			101				101	
			LHS 2895	AB		8.1 a	Y			М		
$14 \ 23 \ 15.28$	$+01 \ 14 \ 29.6$	126053	HIP 070319									
$14 \ 29 \ 22.30$	+80 48 35.5	128642	HIP 070857	A					• • •			
		107224		Aa,Ab	178.78 d		Y	U	1		М	
14 29 36.81 14 33 28 87	$+41 \ 47 \ 45.3$ $\pm 52 \ 54 \ 31 \ 6$	127334	HIP 070873 HIP 071181									
14 36 00.56	+09 44 47.5	128311	HIP 071395	Α								
			HD 128311 b		1.23 y	1.1 a	Y		1			
			HD 128311 c $$		2.52 y	1.76 A	Υ		1			
$14 \ 39 \ 36.50$	$-60\ 50\ 02.3$	128620	α Cen	A								
			HD 128621 Province Cor	AB	79.9 y	8.8 a 7867 c	Y V	0	2	 т		
$14 \ 40 \ 31 \ 11$	-16 12 33 4	128987	HIP 071743	AC		1001 a	r			1		
$14 \ 41 \ 52.46$	$-75\ 08\ 22.1$	128400	HIP 071855									
$14 \ 45 \ 24.18$	+13 50 46.7	130004	HIP 072146									
$14\ 47\ 16.10$	+02 42 11.6	130307	HIP 072312	•••								
$14 \ 49 \ 23.72$	$-67 \ 14 \ 09.5$	130042	HIP 072493	A			· · · ·				· · · ·	
	 123 54 42 6	130048	 HIP 072567	AB		1.5 a	Ŷ			0	IVI	
14 00 10.01	F20 04 42.0	100940	HD 130948 B	AB		2.6 a	Y			м		
Continued on	Next Page						-					

TABLE 7.1 - Continued

R.A.	Decl.	HD	Other	Comp		~	<i>a</i> .		~ ~	a b		
(J2000.0)	(J2000.0)	Name	Name	1D (5)	Period	Sep	Sts	VB (0)	SB (10)	(11)	UR (12)	(12)
(1)	(2)	(3)	(4)	(5)	(6)	(1)	(8)	(9)	(10)	(11)	(12)	(13)
			HD 130948 C	BC		0.1 a	Y			Μ		
$14 \ 51 \ 23.38$	$+19 \ 06 \ 01.7$	131156	ξ Boo	A								
			HD 131156B	AB	151.6 y	6.3 a	Y	0				
14 53 23.77	+19 09 10.1	131511	HIP 072848	A	195 4 d		· · · · V					 C
14 53 41 57	 ±23.20.42.6	131582	HIP 072875	Aa,AD	125.4 d		I	0	1			5
14 00 41.07	720 20 42.0	101002	1111 072075	Aa Ab			Y				M	
14 55 11.04	+53 40 49.2	132142	HIP 073005									
14 56 23.04	+49 37 42.4	132254	HIP 073100									
14 58 08.80	-48 51 46.8	131923	HIP 073241	А								
				Aa,Ab	14.87 y		Y		1		Μ	
$15 \ 03 \ 47.30$	+47 39 14.6	133640	44 Boo	Α								
			NLTT 39210	AB	206 y	1.8 a	Y	0			M	
				$_{\rm Ba,Bb}$	6.43 h		Y		2		\mathbf{E}	
15 10 44.74	$-61\ 25\ 20.3$	134060	HIP 074273	• • •								
15 13 50.89	$-01\ 21\ 05.0$	135204	HIP 074537	A			· · · · V					
15 15 50 17	100 47 46 0	125500	HID 074702	AD		0.1 a	I			0		
15 10 19 90	+00 47 40.9	126202	HIP 074702									
15 19 18.80	± 01 45 55.5	130202	LHS 3060	AB		11 4 a	v			 Р		
15 21 48 15	-48 19 03 5	136352	HIP 075181	AD		11.4 a	1			1		
15 22 36 69	-1039400	136713	HIP 075253									
15 22 46.83	+185508.3	136923	HIP 075277									
$15\ 23\ 12.31$	$+30\ 17\ 16.1$	137107	n CrB	A								
			HD 137108	AB	41.59 v	0.6 a	Y	0	2			
			GJ 584 C	AB,E		193.5 a	Υ			S		
$15\ 28\ 09.61$	$-09 \ 20 \ 53.1$	137763	HIP 075718	A								
				Aa,Ab	2.44 y	0.1 a	Y	U	2	R	Μ	S
			HD 137778	AB		52.3 a	Υ			Т		
			GJ 586C	AC		1212 a	Y			Т		
$15 \ 29 \ 11.18$	$+80 \ 26 \ 55.0$	139777	HIP 075809	Α								
			HD 139813	AB		31.3 a	Y			Т		
$15 \ 36 \ 02.22$	+39 48 08.9	139341	HIP 076382	Α				• • •				
				AB	55.6 y	0.9 a	Y	0			М	
			HD 139323	AB,C		121.5 a	Ŷ			Т		
$15 \ 44 \ 01.82$	$+02 \ 30 \ 54.6$	140538	ψ Ser	A			· · · ·					
15 40 00 01		141004		AB		4.4 a	Ŷ			0	M	
15 46 26.61	+07 21 11.1	141004	HIP 077250									
15 47 29.10	-3/ 54 58./	140901	HIP 077358 NUTT 41160	A		15 0	· · · · V			 M		
15 48 00 46	$\pm 01 34 18 3$	141979	HIP 077408	AD		10 a	1			101		
15 48 09.40	± 01.34 10.3	141272	1111 077408	AB		179 a	v			S		
15 52 40 54	$+42\ 27\ 05\ 5$	142373	v Her	пъ		11.5 a				0		
15 52 40.04 15 53 12 10	+13 11 47 8	142267	39 Ser	A								
10 00 12.10	, 10 11 100	112201		Aa.Ab	138.56 d		Y		1			
$16 \ 01 \ 02.66$	$+33\ 18\ 12.6$	143761	ρ CrB	A								
			ρ CrB b		39.85 d	0.22 A	Μ		1			
			·	Aa,Ab	40.18 d		Μ	U				
$16 \ 01 \ 53.35$	+58 33 54.9	144284	θ Dra	À								
				Aa,Ab	3.07 d		Υ		2			
$16 \ 04 \ 03.71$	$+25 \ 15 \ 17.4$	144287	HIP 078709	Α								
				Aa,Ab	12.19 y	0.2 a	Y		1		M	
$16 \ 04 \ 56.79$	$+39 \ 09 \ 23.4$	144579	HIP 078775	A								
			LHS 3150	AB		70.3 a	Y			Р		
16 06 29.60	$+38 \ 37 \ 56.1$	144872	HIP 078913									
16 09 42.79	$-56\ 26\ 42.5$	144628	HIP 079190									
10 10 24.31	+43 49 03.5	140070	14 Her 14 Her	А	1 96	 2 77 A	· · · · · V					
 16 13 18 45	 ⊥13 31 36 0	145058	HIP 070409	Δ	4.00 y	2.11 A	I		1			
10 10 10.40	LT0 01 00.9	140300	NLTT 49979	AB	1354 v	412	v	 Р				
			1111 12212	AD	1004 y	1623 26 a	M			S		
$16\ 13\ 48.56$	$-57\ 34\ 13.8$	145417	HIP 079537			1020.20 a						
$16\ 14\ 11.93$	-31 39 49.1	145825	HIP 079578	A								
				Aa,Ab	7.14 y		Y		1		М	
$16 \ 14 \ 40.85$	+33 51 31.0	146361	$\sigma^2 \text{ CrB}$	A								
				Aa,Ab	1.14 d		Y	0	2			V
			HD 146362	AB	889 y	7.1 a	Y	Р			М	
			HIP 79551	AE	· · · · "	633.7 a	Y			Т		
			$\sigma~{\rm CrB}~{\rm D}$	$_{\rm Ea,Eb}$	52 y		Y	U				
$16\ 15\ 37.27$	$-08 \ 22 \ 10.0$	146233	18 Sco									
$16\ 24\ 01.29$	$-39 \ 11 \ 34.7$	147513	HIP 080337	Α								
			HD 147513 b		1.48 y	1.26 A	Y		1			
				AB		345 a	Y			Т		
$16\ 24\ 19.81$	$-13 \ 38 \ 30.0$	147776	HIP 080366	A								
				AC		6.4 a	M			R		
16 00 00 14	70.05.09.0	147504	····	AD		9.7 a	Ŷ			M		• • •
10 28 28.14	-70 05 03.8	14/584	ζırA	A	12 00 -		· · · · V				 M	
16 28 52 67	 ⊥18 24 50 €	1/8652	HIP 080725	Aa,AD	12.98 d		I	U	T		11/1	
10 20 02.07	T10 24 00.0	140000	LHS 3204	AR	224 v	 99a	v	0			м	
16 31 30 03	-39 00 44 2	148704	HIP 080025	Δ	224 y	4.4 a	1	0			141	
Continued on	Next Page		000020									

TABLE 7.1 - Continued

R.A.	Decl.	HD	Other	\mathbf{Comp}								
(J2000.0)	(J2000.0)	Name	Name	ID	Period	Sep	Sts	VB	SB	CP	UR	СН
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)
				Aa,Ab	31.86 d		Υ		2		М	
$16 \ 36 \ 21.45$	$-02 \ 19 \ 28.5$	149661	HIP 081300									
$16 \ 37 \ 08.43$	$+00 \ 15 \ 15.6$	149806	HIP 081375	А								
				AB		6.3 a	Υ			Р		
$16 \ 39 \ 04.14$	$-58\ 15\ 29.5$	149612	HIP 081520									
$16 \ 42 \ 38.58$	$+68 \ 06 \ 07.8$	151541	HIP 081813									
$16 \ 52 \ 58.80$	$-00 \ 01 \ 35.1$	152391	HIP 082588									
16 57 53.18	$+47 \ 22 \ 00.1$	153557	HIP 083020	A								
				AB		4.9 a	Y			Μ	Μ	
			HD 153525	AC		112.1 a	Y			Т		
$17 \ 02 \ 36.40$	$+47 \ 04 \ 54.8$	154345	HIP 083389	Α								
			HD 154345 b		9.14 y	4.19 A	Y		1			
$17 \ 04 \ 27.84$	$-28 \ 34 \ 57.6$	154088	HIP 083541									
17 05 16.82	$+00\ 42\ 09.2$	154417	HIP 083601									
17 10 10.35	$-60\ 43\ 43.6$	154577	HIP 083990									
17 12 37.62	$+18\ 21\ 04.3$	155712	HIP 084195									
17 15 20.98	$-26 \ 36 \ 10.2$	155885	36 Opn	A			· · · ·	 D				
				AB	470.9 y	5.1 a	Y	Р				
17 10 02 82		156974	HD 156026	AC		731.6 a	Ŷ	• • •		.T.		
17 19 03.83	$-40\ 38\ 10.4$	156274	41 Ara	A			· · · ·					
			NUTT 44FOF	Aa,Ab	88.03 d		Y	 D	1		IVI	
17 20 20 57	1 22 28 02 0	157014	NL11 44525 79 Her	AB	693.24 y	15 a	Ŷ	Р				
17 20 39.37	+32 28 03.9	157214	12 ner									
17 22 31.29	-02 23 17.4	13/34/	UD 6465	A			· · · · V			 р		
17 25 00 10		159622	HID 085225	AD		49 a	I			г		
$17 \ 20 \ 16 \ 43$	± 47 24.1 ± 47 24.07.0	150062	HIP 085653									
17 30 23 80	-01 03 46 5	158614	HIP 085667	Δ								
17 30 23.80	-01 03 40.5	158014	1111 085007	AB	46 34 v	032	 V				 M	
17 32 00 99	± 34 16 16 1	159222	HIP 085810	AD	40.34 y	0.5 a	1	0	4		101	
17 34 59 59	+61 52 28 4	160269	26 Dra	Δ								
11 04 00.00	101 02 20.4	100205	20 D14	AB	74.16 v	1.55 a	Y	0	1		M	
			HIP 86087	ABC	11110 9	737 6 a	Ŷ	0	-	т		
17 39 16 92	$\pm 03 33 18 9$	160346	HIP 086400	A A		rono a				-		
11 00 10.02	100 00 1010	100010		Aa.Ab	83.73 d		Y	U	1			
17 41 58.10	+72 09 24.9	162004	31 Dra B	B^a								
			HD 162003	AB	12500 v	31 a	Y	Р		Т		
$17 \ 43 \ 15.64$	+21 36 33.1	161198	HIP 086722	А								
				Aa.Ab	7.0 v	0.1 a	Y	Р	1		Μ	
$17 \ 44 \ 08.70$	-51 50 02.6	160691	μ Ara	A								
			HD 160691 b		1.79 v	1.5 a	Y		1			
			HD 160691 c		8.18 y	4.17 A	Y		1			
			HD 160691 d		9.55 d	0.09 A	Y		1			
			HD 160691 e		310.55 d	0.92 A	Y		1			
$17 \ 46 \ 27.53$	$+27 \ 43 \ 14.4$	161797	μ Her A	А								
				Aa,Ab	65.0 v	1.43 a	Y	U	V	R	Μ	
			NLTT 45430	Aa,BC		34 a	Y			Р		
				BC	43.2 v	1.1 a	Y	0				
17 53 29.94	+21 19 31.1		HIP 087579									
18 02 30.86	$+26\ 18\ 46.8$	164922	HIP 088348	А								
			HD 164922 b		3.16 v	2.11 A	Y		1			
$18 \ 05 \ 27.29$	$+02 \ 30 \ 00.4$	165341	70 Oph	А								
			NLTT 45900	AB	121.2 y	1.59 a	Y	0	2		Μ	
$18 \ 05 \ 37.46$	+04 39 25.8	165401	HIP 088622	А								
				Aa,Ab			Y				Μ	
$18 \ 06 \ 23.72$	$-36 \ 01 \ 11.2$	165185	HIP 088694									
$18 \ 07 \ 01.54$	$+30 \ 33 \ 43.7$	165908	HIP 088745	A								
				Aa,Ab		0.2 a	Μ			\mathbf{R}		
				AB	56.4 y	$0.97 \ a$	Y	0				
$18 \ 09 \ 37.42$	$+38 \ 27 \ 28.0$	166620	HIP 088972									
18 10 26.16	$-62 \ 00 \ 07.9$	165499	ι Pav	A								
				Aa,Ab			Y				М	
$18 \ 15 \ 32.46$	$+45 \ 12 \ 33.5$	168009	HIP 089474									
$18 \ 19 \ 40.13$	-63 53 11.6	167425	HIP 089805	A								
				AB		7.8 a	Y			Р	М	
$18 \ 31 \ 18.96$	$-18\ 54\ 31.7$	170657	HIP 090790									
$18 \ 38 \ 53.40$	$-21 \ 03 \ 06.7$	172051	HIP 091438									
$18 \ 40 \ 54.88$	$+31 \ 31 \ 59.1$		HIP 091605	A								
			LHS 3402	AB		9.3 a	Y			0		
18 55 18.80	-37 29 54.1	175073	HIP 092858	• • •								
18 55 53.22	$+23 \ 33 \ 23.9$	175742	HIP 092919	A					• • •			
				Aa,Ab	2.88 d		Y		1			
18 57 01.61	+32 54 04.6	176051	HIP 093017	A					• • •			
10 50 51 00		170055		AB	61.39 y	0.8 a	Y	0	1			• • •
18 58 51.00	$+30\ 10\ 50.3$	176377	HIP 093185	• • •								
19 06 25.11	-37 03 48.4	177474	γ UrA A	A	101.9	1.01						
10.06.50.46		177505	IIID 0000F0	AB	121.8 y	1.21 a	Ŷ	0			11/1	
19 00 52.46	-374838.4	178408	HIP 093858									
19 01 01.02	-T10 51 12.2	110420	1111 039900	A Aa Ab	21.06.4		v		1			
10 12 05 02	± 4951207	170057	HID 004336	Aa,AD	21.90 U		I		T			
Continued on		113301	1111 034000	А								
Commuted Off	i age											

TABLE 7.1 - Continued

R.A.	Decl.	HD	Other	\mathbf{Comp}								
(J2000.0)	(J2000.0)	Name	Name	ID (5)	Period	Sep	Sts	VB	SB	CP	UR	CH
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)
			HD 179958	AB	3100 y	$7.65~{\rm a}$	Y	Р			Μ	
$19\ 12\ 11.36$	+57 40 19.1	180161	HIP 094346							• • •		
19 21 29.76	-34 59 00.6	181321	HIP 095149	A Aa Ab			v				 M	
19 23 34 01	+33 13 19 1	182488	HIP 095319	Aa,Ab			1				101	
19 20 54.01 19 24 58.20	+115639.9	182572	31 Agl									
19 31 07.97	+58 35 09.6	184467	HIP 095995	A								
				AB	1.35 y	0.1 a	Y	0	2			
$19 \ 32 \ 06.70$	$-11 \ 16 \ 29.8$	183870	HIP 096085									
$19 \ 32 \ 21.59$	$+69 \ 39 \ 40.2$	185144	σ Dra									
$19 \ 33 \ 25.55$	+21 50 25.2	184385	HIP 096183	• • •								
$19 \ 35 \ 55.61$	$+56\ 59\ 02.0$	185414	HIP 096395	A								
				Aa,Ab	13.08 y		Ŷ		1			
19 41 48.95	$+50 \ 31 \ 30.2$	186408	16 Cyg A	A			· · · · V		 V	 D		
			HD 186427	Aa B	18212 1	40.0 a	v		v	T		
			16 Cvg B b	ла, Б	2.19 v	1.68 A	Ý		1	1		
19 45 33.53	+33 36 07.2	186858	HIP 097222	\mathbf{F}^{a}	2.10 9							
				Fa,Fb			Μ		V			
			HD 225732	\mathbf{FG}		26 a	Y			Р		
			HD 187013	AF		780 a	Υ			Т		
				AB	232 y	2.46 a	Y	0			Μ	
$19 \ 51 \ 01.64$	+10 24 56.6	187691	HIP 097675	Α								
				AC		22.5 a	Y			Р		
19 59 47.34	$-09\ 57\ 29.7$	189340	HIP 098416	A								
	1 22 42 20 1	190722	IIID OOSEOF	AB	4.90 y	0.2 a	Ŷ	0	2		M	
20 00 43.71	+22 42 39.1	109755	HD 180733 b	А	2 22 d	0.03 4	v		1		 F	
20 02 34 16	± 15 35 31 5	190067	HIP 098677	Δ	2.22 u	0.05 A	1		1		Б	
20 02 04.10	110 00 01.0	100001		AB		2.9 a	Y			M		
$20 \ 03 \ 37.41$	+29 53 48.5	190360	HIP 098767	А								
			HD 190360 b		7.92 y	3.92 A	Y		1			
			HD 190360 c		$17.10 \ d$	0.13 A	Υ		1			
			LHS 3509	AB		178.2 a	Y			Т		
$20 \ 03 \ 52.13$	+23 20 26.5	190404	HIP 098792	• • •								
$20\ 04\ 06.22$	$+17 \ 04 \ 12.6$	190406	HIP 098819	A								
		100470	HD 354613	Aa,Ab		0.79 a	Ŷ			M		
20 04 10.05	+23 47 24.8	190470	HIP 098828									
20 05 09.78	$+30\ 20\ 42.4$	190771	HIF 098921	Aa Ab			v		v		м	
20 05 32 76	$-67\ 19\ 15\ 2$	189567	HIP 098959	114,115			-		•		101	
20 07 35.09	$-55\ 00\ 57.6$	190422	HIP 099137									
20 08 43.61	$-66\ 10\ 55.4$	190248	δ Pav									
20 09 34.30	+16 48 20.8	191499	HIP 099316	А								
			ADS 13434B	AB		4.45 a	Υ			Р	Μ	
$20\ 11\ 06.07$	$+16 \ 11 \ 16.8$	191785	HIP 099452	А								
				AE		103.8 a	Y			Р		
$20\ 11\ 11.94$	$-36\ 06\ 04.4$	191408	HIP 099461	A								
 20 12 50 85		100060	LHS 487	AB		7.1 a	Ŷ			0	М	
20 13 59.85	-00 52 00.8	192203	HIF 099711 HD 102262 b	A	94.95 d	0.15 A	· · · · V		1			
20 15 17 39	-27 01 58 7	192310	HIP 099825		24.55 u	0.15 A	1		1			
20 17 31.33	$+66\ 51\ 13.3$	193664	HIP 100017	A								
20 11 01:00	100 01 1010	100001		Aa.Ab			M				М	
$20\ 27\ 44.24$	-30 52 04.2	194640	HIP 100925									
$20 \ 32 \ 23.70$	$-09 \ 51 \ 12.2$	195564	HIP 101345	Α								
			LTT 8128	AB		4.4 a	Υ			Μ	Μ	
$20 \ 32 \ 51.64$	+41 53 54.5	195987	HIP 101382	A			• • • •		• • •			
				Aa,Ab	57.32 d		Y	0	2		М	
20 40 02.64	$-60\ 32\ 56.0$	196378	ϕ^2 Pav									
20 40 11.76	-23 46 25.9	196761	HIP 101997									
20 40 45.14	+195607.9	197076	HIP 102040	A		105.1 -	· · · · V			 р	• • •	
20.43.16.00	-29 25 26 1	197214	HIP 102264	AC		120.1 d	1			r		
20 49 16.23	$+32\ 17\ 05.2$	198425	HIP 102264	A								
				Aa.Ab			М		V			
			NLTT 49961	AB		33.0 a	Y			Р		
20 56 47.33	$-26\ 17\ 47.0$	199260	HIP 103389									
$20\ 57\ 40.07$	$-44 \ 07 \ 45.7$	199288	HIP 103458									
$21 \ 02 \ 40.76$	+45 53 05.2	200560	HIP 103859	C								
			GJ 816.1B	CD^{c}		3.3 a	Υ			Μ	М	
21 07 10.38	$-13\ 55\ 22.6$	200968	HIP 104239	A		4.91	· · · ·				 \\ f	
21 00 20 74	···	100500	GJ 819B	AB		4.31 a	Ŷ			0	IVI	
21 09 20.74 21 09 22.74	-62 01 38.1 -73 10 22 7	199509	HIP 104430	Δ			• • • •					
21 00 22.40	10 10 22.1	200020		ÅB	5.87 v	02 a	v	U		в		
			NLTT 50542	AB,C		7.2 a	Ŷ			M		
$21 \ 14 \ 28.82$	+10 00 25.1	202275	δ Equ	A								
				AB	5.70 y	0.3 a	Υ	0	2			\mathbf{S}
$21 \ 18 \ 02.97$	$+00 \ 09 \ 41.7$	202751	HIP 105152									
$21 \ 18 \ 27.27$	$-43 \ 20 \ 04.7$	202628	HIP 105184									
Continued on	Next Page											

TABLE 7.1 - Continued

R.A. (J2000.0) (1)	Decl. (J2000.0) (2)	HD Name (3)	Other Name (4)	Comp ID (5)	Period (6)	Sep (7)	${ m Sts}\ (8)$	VB (9)	SB (10)	CP (11)	UR (12)	CH (13)
21 19 45.62	$-26\ 21\ 10.4$	202940	HIP 105312	А								
				Aa,Ab	21.35 d		Y		1			
			LHS 3656	AB	261.62 y	3.2 a	Υ	Р			Μ	
$21 \ 24 \ 40.64$	$-68\ 13\ 40.2$	203244	HIP 105712									
$21 \ 26 \ 58.45$	$-56\ 07\ 30.9$	203850	HIP 105905	• • •								
$21 \ 27 \ 01.33$	-44 48 30.9	203985	HIP 105911	A								
				Aa,Ab			Y			· · · ·	M	
			LTT 8515	AB		88.0 a	Ŷ		• • •	Р		
$21 \ 30 \ 41.24$ $21 \ 40 \ 20 \ 77$	$-50\ 50\ 45.4$ $-74\ 04\ 27\ 4$	205536	HIP 107022									
21 40 23.11	-740427.4 ± 2844335	205550	IIII 107022	Δ								
21 11 00.00	120 11 00.0	200020	HD 206827	AB	789 v	1.9 a	Y	P			M	
$21 \ 44 \ 31.33$	+14 46 19.0	206860	HIP 107350	А								
			HN Peg B	AB		43.2 a	Y			Μ		
$21 \ 48 \ 00.05$	$-40\ 15\ 21.9$	207144	HIP 107625									
$21 \ 48 \ 15.75$	$-47 \ 18 \ 13.0$	207129	HIP 107649									
21 53 05.35	$+20\ 55\ 49.9$	208038	HIP 108028									
21 54 45.04	+32 19 42.9	208313	HIP 108156									
22 09 29.87	-07 32 55.1	210277	HIP 109378	А	1.01		· · · · V					
	126 15 22 8	210667	HID 210277 D		1.21 y	1.1 a	I		1			
22 11 11.91	-41 22 54 0	210007	HIP 109527									
22 14 00.00 22 15 54.14	+54 40 22.4	211472	HIP 109021	Α								
	101 10 2211		GJ 4269	AT		77.2 a	Y			P		
$22 \ 18 \ 15.62$	$-53 \ 37 \ 37.5$	211415	HIP 110109	А								
				AB		3.4 a	Y			0	Μ	
$22 \ 24 \ 56.39$	$-57 \ 47 \ 50.7$	212330	HIP 110649	А								
				Aa,Ab			Y				Μ	
$22 \ 25 \ 51.16$	$-75 \ 00 \ 56.5$	212168	HIP 110712	A						· <u>· ·</u> ·		
			HIP 110719	AB		20.8 a	Y			Р		
22 39 50.77	$+04\ 06\ 58.0$	214683	HIP 111888	A								
22 42 30.88	-47 12 58.9	214955	NITT 54607	AD			v			 р	 M	
22 43 21 30	-06 24 03 0	215152	HIP 112100	AD		1.0 a	1			r	111	
$22 \ 45 \ 21.50$ $22 \ 46 \ 41.58$	$+12\ 10\ 22.4$	215152	£ Peg	A								
				AB		11.1 a	Y			0		
22 47 31.87	$+83 \ 41 \ 49.3$	216520	HIP 112527									
$22 \ 51 \ 26.36$	+13 58 11.9	216259	HIP 112870									
$22 \ 57 \ 27.98$	$+20 \ 46 \ 07.8$	217014	51 Peg	А								
			51 Peg b		4.23 d	0.05 A	Y		1			
22 58 15.54	$-02 \ 23 \ 43.4$	217107	HIP 113421	A					• • •			
			HD 217107 b		7.13 d	0.07 A	Y		1			
			HD 217107 c	 A D	9.18 y	4.41 A	M	• • •	1	 D		
22 02 04 08	1.20 55 06 0		UID 112820	AB		0.3 a	IVI			к		
23 03 04.98 23 10 50 08	+20 35 00.9 +45 30 44 2	217813	HIP 114456	Δ								
20 10 00.00	140 00 44.2	210000		AB		50 a	Y			P		
23 13 16.98	$+57\ 10\ 06.1$	219134	HIP 114622									
$23 \ 16 \ 18.16$	$+30 \ 40 \ 12.8$	219538	HIP 114886									
$23 \ 16 \ 42.30$	$+53 \ 12 \ 48.5$	219623	HIP 114924									
$23 \ 16 \ 57.69$	$-62 \ 00 \ 04.3$	219482	HIP 114948									
$23 \ 19 \ 26.63$	+79 00 12.7	220140	HIP 115147	А								
			NLTT 56532	AB		10.8 a	Y		• • •	M		
				AC		962.6 a	Ŷ			Т		
23 21 30.51 23 23 04 90	+44 05 52.4 -10.45 51 2	220182	ПГР 115331 НГР 115445									
23 23 04.89 23 31 22 21	+59 09 55 9	220339 221354	HIP 116085									
23 35 25.61	+31 09 40 7	221851	HIP 116416									
$23 \ 37 \ 58.49$	$+46\ 11\ 58.0$	222143	HIP 116613									
23 39 37.39	$-72\ 43\ 19.8$	222237	HIP 116745									
$23 \ 39 \ 51.31$	-32 44 36.3	222335	HIP 116763									
$23 \ 39 \ 57.04$	$+05 \ 37 \ 34.6$	222368	ι Psc									
$23 \ 52 \ 25.32$	$+75 \ 32 \ 40.5$	223778	HIP 117712	A								
				Aa,Ab	7.75 d		Y	O	2			V
 02 EG 10 CF	20.02.00.1			AB	290 y	4.6 a	Y	Р			Μ	
23 00 10.07 23 58 06 82	-39 03 08.4 $\pm 50 26 51 6$	224228	HIP 118008	Α.								
20 00 00.82	100 20 01.0	224400		AaAb	52.41 d		v		1			
					02.11 4		-		-			

^a The sample star is not the system's primary, which is identified as component A below. ^b The brightest component of the system is HD 25998, but is designated as component E in the WDS. The A component is the wide CPM companion, HD 25893, which is about 2 magnitudes fainter and itself a visual binary. I have retained the component designations of the WDS, so the fainter visual pair is AB and the wide CPM companion is E. WDS components C and D are optical, and E itself might have a close companion, as evidenced by its accelerating proper motion (see Table 5.2). ^c WDS lists these entries for HD 200595, a bright binary 153" away from the sample star HD 200560, but one that is not physically associated to it. HD 200560 is itself is a close CPM pair and listed in the WDS as CD. I have retained the WDS designations, which makes C and D the only physically associated components of this system.

7.3 Observed Stellar Multiplicity

The multiplicity of the 454 solar-type stars studied here is summarized in Table 7.2 below. Column 1 identifies the multiplicity order and Column 2 lists the corresponding number of stars from observations. Column 3 contains the implied percentage of stars along with its uncertainty, obtained by bootstrap analysis as described below. Columns 4 and 5 contain the number and percentages when all candidate companions are considered to be real.

The uncertainties were estimated by performing a bootstrap analysis on the data over 10,000 iterations. This analysis involves a resampling of the observed statistics. For each iteration, a new sample of 454 stars is created by randomly selecting stars from the observed sample of 454 stars. Every such sample may include the same star more than once and leave out other stars altogether. The multiplicity frequencies are then computed for each iteration. The distribution shown in Figure 7.2 is the result for the 10,000 iterations. The mean and uncertainty of the resulting statistic are the the mean and standard deviation of the fitted Gaussian curves.

Stellar	Conf	irmed	Including	Candidates
Multiplicity	Number	Percent	Number	Percent
Single	257	56.6 ± 2.4	244	53.7 ± 2.4
Binary	150	33.0 ± 2.2	154	33.9 ± 2.2
Triple	35	7.7 ± 1.3	43	9.5 ± 1.4
Quadruple	10	2.0 ± 0.8	10	2.0 ± 0.8
Quintuple	2	0.7 ± 0.3	2	0.7 ± 0.3
Sextuple			1	0.6 ± 0.3

TABLE 7.2: Observed Multiplicity Statistics



FIGURE 7.2: The observed frequency of single, binary, triple, and quadruple systems. These results were obtained by a bootstrap analysis with 10,000 iterations. The solid line and dashed curve represents confirmed systems and the dashed line and dotted curve includes candidate companions. The uncertainties are estimated as the standard deviation corresponding to the fitted Gaussian curve.

Table 7.3 lists the numbers of pairs detected and confirmed as physical associations by the various techniques studied here. The information is presented in a matrix form showing the overlap between various techniques. Column 1 contains a code identifying the method which confirms a physical association and is composed of a prefix corresponding to the column headings for columns 8–13 in Table 7.1 and a suffix which is the value listed for the pairs in the column in Table 7.1. The values for CPM companions are listed in order of priority

in confirming the companion. For example, the first row lists CPM pairs with matching independent measures of proper motion and trigonometric parallax, followed by matching proper motion and photometric parallax, and so on. Column 2 contains a short description of this code. Column 3 identifies the number of pairs uniquely confirmed by this method alone. The remaining columns make up a symmetric matrix whose diagonal lists the total number of pairs identified by this method and other cells show the number of pairs overlapping with other techniques. For example, The SB-1 row indicates that 48 pairs were confirmed based on evidence of a single-lined spectroscopic orbit, of which three have corresponding good-quality visuals orbits, four have preliminary visual orbits, 14 have photocentric-motion orbits, 21 show evidence of proper motion acceleration, and two were detected as SFP binaries. Sixteen of the 48 SB1 pairs have no other evidence of companionship. Table 7.4 shows similar statistics when all candidate companions are also included.

10
ars
$\mathbf{s}_{\mathbf{t}}$
/pe
É
lar
so
154
of 2
le
du
sa
$_{\rm the}$
in
sue
nic
ıpa
on
Р
me
цIJ
Gor
8
5
fo
ior
cat
sifi
Clas
÷
7.3
3LE
TAE

Code	Description	D	VBO	VBP	VBU	SB1	SB2	SBV	\mathbf{CPT}	CPP	\mathbf{CPS}	CPO	CPM	CPR	URE	URL	URM	CHS	CHV
VB-O	VBO: Definitive	7	38	:	:	3	18	1	:	:	:	•	:	:	:	:	17	9	3
VB-P	VBO: Preliminary	12	÷	33	÷	4	÷	÷	1	÷	÷	÷	÷	÷	÷	÷	18	÷	÷
VB-U	VBO: Photocentric Motion	1	÷	÷	21	14	1	1	÷	÷	÷	÷	÷	7	÷	÷	10	5	÷
SB-1	SB1: Orbital Solution	16	ŝ	4	14	48	÷	÷	÷	÷	÷	÷	÷	1	÷	÷	21	1	÷
SB-2	SB2: Orbital Solution	4	18	÷	1	÷	27	÷	÷	÷	÷	÷	÷	1	1	÷	10	9	4
SB-V	RVV: RV Variations	1	1	÷	1	÷	÷	6	÷	÷	÷	1	1	5	÷	÷	4	÷	÷
CP-T	CPM: Matching π_{trig}	29	÷	1	:	÷	÷	÷	30	÷	÷	÷	:	÷	÷	:	÷	:	:
CP-P	CPM: Matching π_{phot}	39	÷	÷	÷	÷	÷	÷	:	42	÷	÷	:	÷	÷	:	ŝ	:	:
CP-S	CPM: Matching π_{spec}	Ŋ	:	÷	:	÷	÷	÷	÷	÷	9	÷	÷	÷	÷	÷	1	÷	÷
CP-O	CPM: Orbital Motion	n	÷	÷	÷	÷	÷	1	÷	÷	÷	10	÷	÷	÷	÷	9	:	÷
CP-M	CPM: Matching μ	20	÷	÷	÷	÷	÷	1	÷	÷	÷	÷	26	÷	÷	÷	ъ	÷	:
CP-R	CPM: Other (Published)	7	÷	÷	4	1	1	7	÷	÷	÷	÷	÷	11	÷	÷	9	1	÷
UR-E	Unres: Eclipsing Binary	1	÷	÷	÷	÷	1	÷	÷	÷	÷	÷	÷	÷	5	÷	•	÷	÷
UR-L	Unres: Over-luminous	1	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	1	•	÷	:
UR-M	Unres: Accelerating μ	×	17	18	10	21	10	4	÷	ŝ	1	9	Ŋ	9	÷	÷	85	5	:
CH-S	CHARA: SFP	:	9	:	2	1	9	÷	:	÷	:	÷	:	1	:	:	2	×	:
CH-V	CHARA: Visibility	÷	ŝ	•	:		4	•	:	:	:		:	:	:	•		:	4

Stars
Type
Solar-
454
e of
e sampl
the
ıs ir
ompanior
ŭ
Candidate
25
and
Confirmed
258
of
: Classification
7.4
TABLE

Code	Description	D	VBO	VBP	VBU	SB1	SB2	\mathbf{SBV}	CPT	CPP	\mathbf{CPS}	CPO	CPM	CPR	URE	URL	URM	CHS	CHV
VB-O	VBO: Definitive	7	38	:	:	0	18	1	:	:	:	:	:	:	:	:	17	9	3
VB-P	VBO: Preliminary	12	÷	33	÷	4	÷	÷	1	÷	÷	:	÷	÷	÷	:	18	÷	÷
VB-U	VBO: Photocentric Motion	n	÷	÷	23	14	1	1	÷	÷	÷	÷	÷	7	÷	÷	10	7	÷
SB-1	SB1: Orbital Solution	16	ŝ	4	14	48	÷	÷	÷	÷	÷	÷	:	1	÷	÷	21	1	:
SB-2	SB2: Orbital Solution	4	18	÷	1	÷	27	÷	÷	÷	÷	÷	÷	1	1	÷	10	9	4
SB-V	RVV: RV Variations	œ	1	÷	1	÷	÷	16	÷	÷	÷	1	1	7	÷	÷	4	÷	:
CP-T	CPM: Matching π_{trig}	29	÷	1	:	÷	÷	:	30	÷	÷	÷	÷	÷	÷	:	:	÷	÷
CP-P	CPM: Matching $\pi_{\rm phot}$	39	÷	÷	÷	÷	:	:	:	42	÷	:	:	÷	÷	:	e	:	÷
CP-S	CPM: Matching π_{spec}	9	÷	÷	÷	÷	÷	÷	÷	÷	7	÷	÷	÷	÷	÷	1	÷	•
CP-O	CPM: Orbital Motion	e	:	÷	:	÷	÷	1	÷	÷	÷	10	÷	÷	÷	÷	9	÷	
CP-M	CPM: Matching μ	20	:	÷	:	÷	÷	1	÷	÷	÷	÷	26	÷	÷	÷	ю	÷	
CP-R	CPM: Other (Published)	6	÷	÷	4	1	1	7	÷	÷	÷	÷	÷	19	÷	÷	7	1	:
UR-E	Unres: Eclipsing Binary	1	÷	÷	÷	÷	1	÷	÷	÷	÷	÷	:	÷	7	÷	÷	÷	:
UR-L	Unres: Over-luminous	7	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	÷	7	÷	÷	÷
UR-M	Unres: Accelerating μ	14	17	18	10	21	10	4	:	e	1	9	5	7	÷	:	92	7	÷
CH-S	CHARA: SFP	÷	9	÷	2	1	9	:	:	÷	÷	:	:	1	÷	:	5	×	:
CH-V	CHARA: Visibility	÷	n	÷	÷	÷	4	÷	:	÷	÷	÷	÷	:	:	:	÷	÷	4
																			I

7.4 Notes on Individual Systems

Following are the notes on individual systems that require an explanation for confirmed, candidate, and refuted companions.

HD 123: Triple. This system is composed of a 106-year visual-orbit pair, the secondary of which was shown to be the more massive component from the absolute astrometry (see Griffin 1999, for a full description of the history of this system), suggesting that it itself was an unresolved binary. Brettman et al. (1983) reported a periodic variation in the component's brightness over roughly a 1-day period, which Griffin (1999) later disproved based on *Hipparcos* photometry and instead showed it to be a spectroscopic binary with a 47.7-day period, estimating component masses of 0.98, 0.95, and 0.22 M_{\odot} for the three components.

HD 166: Single. A possible companion, 311'' away at 147° was identified by blinking archival images. The proper motion of the candidate companion from Høg et al. (1998) is $\mu_{\alpha} = 0''.104 \text{ yr}^{-1}$ and $\mu_{\delta} = -0''.080 \text{ yr}^{-1}$, directionally similar to, but significantly smaller than the primary's *Hipparcos* values of $\mu_{\alpha} = 0''.380 \text{ yr}^{-1}$ and $\mu_{\delta} = -0''.178 \text{ yr}^{-1}$, refuting a physical association.

HD 1237: *Binary, one planet.* In a systematic search for faint companions to planet hosts, Chauvin et al. (2006) discovered a CPM companion to this star using VLT NACO adaptive optics and demonstrated orbital motion. Chauvin et al. (2007) characterized the companion as $M4\pm1V$.

HD 3651: *Binary, one planet.* Luhman et al. (2007) reported the discovery of a T-dwarf companion 43" away from this planet-host star using Spitzer IRAC images, and confirmed

CPM using 2MASS images. The brown dwarf's infrared colors are consistent with the distance to the primary, confirming companionship. They estimate the companion's mass as $0.051 \pm 0.014 \text{ M}_{\odot}$ and age as 7 ± 3 Gyr by comparing luminosity with evolutionary tracks. This was the first substellar object imaged around an exoplanet host.

HD 4391: Triple. The WDS lists three detections of a pair with separations ranging 10?0–16?6 over 98 years. Bailey might have under-estimated the separation as 10" in 1895, but the 1901 measure of 14?1 at 306?6 and the 1993 measure of 16?6 at 307?0 are consistent with a bound pair. In addition to this candidate, I discovered another possible companion, 49" away by blinking archival images (see § 4.1). With the help of the SMARTS Consortium, I obtained VRI images in 2007 July and October to investigate these two candidates. The images clearly revealed the closer companion 16?2 away at 307?6, confirming CPM. The candidate was saturated in all but one V-band image, but the single unsaturated image allowed me to extract its V magnitude via differential photometry, and enabled confirmation as a companion (see Table 5.5). Absolute photometry for the wider candidate was also extracted from these images and the resulting VRI magnitudes along with 2MASS JHK_S magnitudes enabled confirmation of this newly discovered companion as well (see Table 4.2).

HD 4628: Single, maybe Binary. Heintz & Borgman (1984) resolved a pair, separated by 2.7, on 11 exposures over two nights, but did not see the companion on 164 other plates or on multiple visual checks with a micrometer. Their two observations about 25 days apart show evidence of variation in the companion's brightness of about 1 magnitude. Heintz (1994) notes an acceleration in proper motion for the primary and speculates that this

might be caused by the companion reported earlier. Roberts et al. (2005) did not detect a companion using AO down to $\Delta I \lesssim 10$ and note that only a white dwarf companion could have escaped detection, while the flaring companion as seen by Heintz should have been detected. Moreover, the *Hipparcos* and Tycho-2 proper motions of HD 4628 match to within 2σ , and the FIC lists several null results with speckle interferometry and adaptive optics. B. Mason and I observed this target using the KPNO 4-m telescope in 2008 June, and while the separation was too wide and the Δm too large for speckle observations, the finder TV showed a faint source about 5" away at about 230°. Could this be the companion seen by Heintz after about 30 years of orbital motion? While a possibility, follow-up observations by Elliott Horch two days later with the WIYN 3.6m telescope on KPNO failed to identify the source seen by us. Additionally, 5-second exposures in VRI taken by T. Henry at the CTIO 0.9-m telescope in 2008 June also failed to identify any companion, although saturation around the primary could hide the companion in these images. Nidever et al. (2002) report that the primary shows no variation in radial-velocity. At this time, I do not feel that I have sufficient information to confirm or refute this pair, although, chances of a physical companion appear slim.

HD 4676: Binary. Boden et al. (1999) presented a visual orbit based on LBI observations for this 14-day SB2 and derived component masses of 1.223 ± 0.021 M_{\odot} and 1.170 ± 0.018 M_{\odot}. Earlier, Nadal et al. (1979) had speculated on the presence of a third companion based on temporal changes in the spectroscopic orbital elements. While this suspected companion has been mentioned in subsequent literature (Fekel 1981; Tokovinin et al. 2006), Boden et al. (1999) refute it based on imaging and spectroscopic evidence.

HD 4813: Single. The CNS (Gliese & Jahreiß 1991) and DM91 list this star as "SB?". Duquennoy et al. (1991) list 12 velocities measured over nine years, and while there is a hint of a trend, the measured velocity of 8.17 ± 0.44 km s⁻¹ is constant within their measurement errors. Abt & Willmarth (2006) present 23 observations over three years with a radial velocity of 8.07 ± 0.14 km s⁻¹, consistent with the Duquennoy et al. measures 10 years earlier. There seems to be enough evidence to refute the earlier claim of binarity.

HD 5133: Single. Abt & Biggs (1972) and CNS list this star as SB, based on a 34 $\rm km\,s^{-1}$ variation over three observations presented by Joy (1947). More recent observations do not support this conclusion (Abt 1970; Nidever et al. 2002; Gontcharov 2006), refuting the earlier claim of a spectroscopic companion.

HD 9770: *Quadruple*. This system consists of a 4.6-year visual binary, whose secondary is a 11.5-hour eclipsing binary (Cutispoto et al. 1997). The fourth component is in a 112-year visual orbit.

HD 9826: Binary, three planets. Lowrance et al. (2002) discovered an M4.5V companion, 55" away from planet host v And, and confirmed its physical association by demonstrating CPM and showing that its spectral type is consistent with its magnitudes at the primary's distance. They estimate a mass of ~ 1.2 M_{\odot} for the primary and 0.2 M_{\odot} for the secondary using on spectral-type mass relations from Cox (2000) and Henry & McCarthy (1993).

HD 10476: Single. Hipparcos lists a photocentric orbit for this star with a period of

207 days and an inclination of $89^{\circ} \pm 23^{\circ}$. This is not consistent with the constant radial velocities observed by Nidever et al. (2002) and CfA (see § 6.1). Several studies have used this as a single star (Mazeh et al. 2002; Frasca et al. 2006), and there is no other published evidence of binarity. The WDS lists two additional pairs for this star, both of which are optical pairs (see § 4.2).

HD 13445: *Binary, one planet.* This planet-host star with a 4 M_J planet exhibits a longterm trend in radial velocity, consistent with a stellar companion beyond 20 AU (Queloz et al. 2000). Later work has resolved this companion and demonstrated orbital motion (Lagrange et al. 2006). The companion was initially misidentified as a T dwarf (Els et al. 2001) and later shown to be a white dwarf based on spectroscopy (Mugrauer & Neuhäuser 2005) and a dynamical analysis of astrometry and radial velocities (Lagrange et al. 2006).

HD 17925: Single. Listed as "RV-Var" in the CNS (Gliese 1969; Gliese & Jahreiß 1991), but Nidever et al. (2002) show that RMS scatter is less than 0.1 km s^{-1} in observations over 10 years, a conclusion supported by the CfA velocities (see § 6.1).

HD 20010: *Binary*, maybe *Triple*. This 5" CPM pair has a preliminary visual orbit. The secondary is listed in the CNS (Gliese 1969) as "RV-Var?". Eggen (1956) mentions that there is a strong evidence of variability of the faint companion, but the quoted reference (van den Bos 1928) could not be found. With insufficient evidence to confirm or refute a physical association, the additional companion to the secondary remains a possibility.

HD 20630: Single. Listed as "SB?" in the CNS (Gliese & Jahreiß 1991), but Nidever et al. (2002) show that the RMS scatter is less than 0.1 km s⁻¹ in observations over five

years, a conclusion supported by the CfA velocities (see $\S 6.1$).

HD 20794: Single. CNS (Gliese & Jahreiß 1991) lists this star as "SB", but radial velocity data from various catalogs (Abt & Biggs 1972; Duflot et al. 1995; Gontcharov 2006) show a fairly stable velocities with no mention of variability or binarity.

HD 20807: *Binary.* The wide CPM companion HD 20766 lies 309" away and is confirmed by matching proper motions and parallax. Additionally, the WDS lists a single speckle interferometry measure of a companion in 1978, 0".046 away at 11° (Bonneau et al. 1980). However, Bonneau et al. failed to resolve the companion in 1979 and da Silva & Foy (1987) mention that the 1978 measure was in fact an artifact in the diffraction pattern of the telescope spider.

HD 21175: *Binary*. While this pair only has three ground-based measurements, they span over 50 years and are consistent with a bound pair. Söderhjelm (1999) presents a visual orbit combining *Hipparcos* and ground-based measures, confirming a physical association.

HD 22049: Single, one planet. This is ϵ Eri, the well-studied exoplanet host. The single measure listed in WDS is from speckle observation by Blazit et al. (1977). This system has subsequently been observed 13 other times by speckle and AO, and no companion was identified (e.g., McAlister 1978a; Hartkopf & McAlister 1984; Oppenheimer et al. 2001). Presumably, the Blazit measure is spurious. The WDS lists 10 additional pairs, all of which were confirmed as optical by blinking archival images (see § 4.2).

HD 22484: Single. Listed as "SB?" in the CNS (Gliese & Jahreiß 1991), but Nidever et al. (2002) show that the RMS scatter is less than 0.1 km s^{-1} in observations over seven

years, a conclusion supported by the CfA velocities (see $\S 6.1$).

HD 23484: Single, maybe Binary. The CNS (Gliese 1969; Gliese & Jahreiß 1991) list this as "RV-Var", but no radial velocity data could be found in modern surveys. Catalogs (Abt & Biggs 1972; Duflot et al. 1995; Gontcharov 2006) list velocities with RMS scatter of about 3 km s^{-1} , but this could be due to measurement errors or zero-point variances.

HD 24496: Binary. The two measurements with $\rho = 2.6-2.7$ and $\theta = 254^{\circ}-256^{\circ}$ listed in the WDS are by Wulff Heintz, nine years apart and consistent with a bound pair. The first measure is based on observations over three nights and the second on observations over two additional nights. Given the observer, the quality of observations ($\Delta m = 4-5$ measured) and the reasonably high proper motion of the primary, this is likely a physical pair, but one that could use new measurements.

HD 25457: *Single*. The CNS (Gliese & Jahreiß 1991) lists this star as "RV Var", but several catalogs (Abt & Biggs 1972; Duflot et al. 1995; Gontcharov 2006) show a fairly stable radial velocity over many decades.

HD 25665: Single. A possible companion, 287" away at 134° was identified by blinking archival images but refuted based on proper motion differences between the components. The candidate companion has $\mu_{\alpha} = 0.093 \text{ yr}^{-1}$ and $\mu_{\delta} = -0.138 \text{ yr}^{-1}$ (Lépine & Shara 2005), which is directionally similar to but significantly different from the corresponding values for the primary of $\mu_{\alpha} = 0.074 \text{ yr}^{-1}$ and $\mu_{\delta} = -0.299 \text{ yr}^{-1}$.

HD 25680: *Binary*. A companion 0.2 away was discovered by McAlister et al. (1993) with speckle interferometry and confirmed by the same technique by Hartkopf et al. (2008).

These measures show evidence of orbital motion, and given the 0.2 yr^{-1} proper motion of the primary and an elapsed time of 15 years between them, this pair can be confirmed as physical. Given the constant radial velocity of the primary (see § 6.1), this might be close to a face-on orbit. The WDS also lists a potential companion 177" away, which I also identified by blinking archival images. This candidate (HIP 19075) was however refuted based on its significantly different proper motion in *Hipparcos* from the corresponding value of the primary. The two additional WDS entries are clearly optical.

HD 26491: *Binary.* Astrometric and spectroscopic evidence indicate an unresolved companion. A comparison of *Hipparcos* and Tycho-2 proper motions shows a significant difference suggesting a companion (see § 5.1.2), which was confirmed by radial velocity measurements (Jones et al. 2002). Preliminary orbital elements are presented in Table 6.2 (H. Jones 2008, private communication).

HD 32923: *Single*. The WDS lists 19 measurements at roughly 0".1 separation over 76 years, and Eggen (1956) even derived two preliminary visual orbits from these measures. However, Heintz & Borgman (1984) suggest that this is likely spurious and show that the observations are not consistent with orbital motion of any period. Three additional speckle observations exist since the Heintz & Borgman publication, from 1984–1987 (Tokovinin 1985; Tokovinin & Ismailov 1988; McAlister et al. 1993), but there are 17 null detections listed in the FIC by speckle interferometry as well as by AO. This star has a stable radial velocity based on CfA and Nidever et al. (2002) observations (see §6.1). It appears that these multiple, but sporadic, measures are spurious.

HD 34721: Single. The CNS (Gliese 1969) lists this as "SB", but this is not consistent with constant radial velocities observed by Nidever et al. (2002) to within 0.1 km s⁻¹ and CfA (see § 6.1).

HD 35296: *Binary.* DM91 noted the primary of a 12' CPM pair as "SB", but one that was not confirmed by their work. Nidever et al. (2002) show that this star has a stable radial velocity, and the CfA measures support this conclusion (see § 6.1), refuting the earlier claim.

HD 36705: Quadruple. The WDS lists two measurements of this 10" pair (AB Dor AB), separated by 69 years and consistent with a bound pair. The first measurement by Rossiter (1955) measured a $\Delta m \sim 6$, explaining the lack of many more measures. Close et al. (2005) recovered this pair with AO at the VLT and it is also seen in VRI images obtained by T. Henry in 2008 September at the CTIO 0.9-m telescope using VRI filters. While the photometric distance estimate is a match within only 7σ (see Table 5.5), the V magnitude from Rossiter (1955) is likely approximate. Given the high proper motion of the primary, the consistent measures over 79 years indicate a physical association. The 2MASS colors indicate an M-dwarf with a V magnitude estimate of about 12.0, in good agreement with the measure of Rossiter (1955) and consistent with the primary's *Hipparcos* distance. High-contrast AO efforts have split each of these components into binaries themselves. The primary was identified by *Hipparcos* as showing accelerating proper motion, indicating an unseen companion, and this is supported by the significant difference between *Hipparcos* and Tycho-2 proper motions (see $\S5.1.2$). The suspected companion has since been revealed by LBI (Guirado et al. 1997), resolved by AO (Close et al. 2005), and confirmed as a physical association by photometry and spectroscopy (Close et al. 2005, 2007; Boccaletti et al. 2008, and references therein). Close et al. (2005) also split the secondary into a 0.0000 pair, which was later confirmed by Janson et al. (2007) who measured it at a separation of 66.1 mas at 238.5.

HD 40397: Triple. The five measures in the WDS for AB between 1902 & 1932 are consistent with a bound pair and include observations by Aitken and Burnham. The measured $\Delta m \sim 7$ makes this a difficult target for classical techniques and out of the reach of speckle interferometry. Given that more than 70 years have passed since the latest measure, this is a good candidate for follow-up AO observations. This pair also has a wide CPM companion, NLTT 15867, which was confirmed by photometric distance estimates (see Table 5.5).

HD 42807: *Single.* The CNS (Gliese & Jahreiß 1991) list this as "SB?", but this is not consistent with constant radial velocities observed by CfA (see \S 6.1).

HD 43834: *Binary.* Eggenberger et al. (2007) resolved this 3" pair three times over three years with AO at the VLT, demonstrating CPM and showing a hint of orbital motion. They also mention a linear trend in CORALIE radial velocities consistent with this companion, confirming a physical association, and estimate the companion to be M3.5–M6.5 with a mass of $0.14 \pm 0.01 \text{ M}_{\odot}$.

HD 45270: Single, maybe Binary. The WDS lists three measurements spanning 43 years of a $\Delta m \sim 4$ pair separated by about 16", which are consistent with a bound pair. Curiously, no additional measurements exist. This pair was listed in the *Hipparcos* input catalog, but not resolved by *Hipparcos*. 2MASS lists a source near this candidate companion,

but it is clearly not the same star because its infrared colors are more than three magnitudes fainter than the visual magnitude of 10.6 from the *Hipparcos* input catalog. No additional information was found on this pair and hence it is retained as a candidate.

HD 46588: Single. The CNS (Gliese & Jahreiß 1991) lists this as "SB?" but this star was not included in the Nidever et al. (2002) or CfA surveys. Abt & Biggs (1972) lists two velocity measures, 55 years apart and roughly consistent. In a follow-up search, McAlister (1978a) did not resolve any companions via speckle interferometry. Batten's Seventh Catalog of Spectroscopic Binary Orbits lists this as SB with a 60-day period from Abt & Levy (1976), but the orbit is flagged as "very poor". Gomez & Abt (1982) lists this star as a possible negative result. The SB9 catalog does not have an orbit for this star.

HD 48189: *Binary.* The WDS lists 19 measurements over 105 years that are consistent with a bound pair. During this time, the separation has closed in from about 3" to about 0".3 and the position angle has changed by about 15°. Given the small projected separation of 6–40 AU, one might expect a greater change in position angle as evidence of orbital motion. The change of only 15° indicates that the semimajor axis is larger than the observed separations, perhaps due to a high inclination. While a more robust confirmation is not available, the primary has moved about 9" during the measures, and the companion seems to be moving along with it, indicating a physical association.

HD 64606: Binary. For the primary of an SB1 pair, the WDS lists two measures of another $\Delta m \sim 4$ pair separated by 4".9, one each from *Hipparcos* and Tycho. The *Hipparcos* solution is flagged as "poor" quality, and there is no independent confirmation of this

pair. T. Henry observed this star using the CTIO 0.9-m telescope in 2008 September and obtained 1-second exposure images in VRI. No source was found at the expected position in these images, whereas a companion of $\Delta m \sim 4$ should easily have been seen above the background. While the SB1 pair is real, this astrometric detection is refuted.

HD 65907: *Triple*. LHS 1960 is a companion to this star, separated by about 60", and confirmed by photometric distance estimates (see Table 4.2). The WDS lists four measures of an additional companion to LHS 1960, observed 1930–1983, indicating that this component itself is a 3" CPM binary. No further evidence of companionship could be found, but given the high proper motion of the primary and the four consistent measurements on four different telescopes over 53 years, this system can be confirmed as a triple.

HD 67199: Single, maybe Binary. The Hipparcos and Tycho-2 proper motions differ by greater than a 3σ significance indicating an unseen companion (see § 5.1.2), but in the absence of other conclusive evidence, this companion is retained as a candidate.

HD 68017: Single, maybe Binary. The Hipparcos and Tycho-2 proper motions differ by greater than a 3σ significance indicating an unseen companion (see § 5.1.2), but in the absence of other conclusive evidence, this companion is retained as a candidate. Two other WDS components are clearly optical.

HD 68257: Quintuple, maybe Sextuple. The three brightest roughly solar-type components (ζ Cancri A, B, and C) are supported by over 1000 visual measurements each, corresponding to two visual orbits. Component C has been noted to have an irregular motion for most of its history and was identified as an SB1 with an orbit of 6302 ± 59 days

(Griffin 2000), consistent with earlier astrometric orbits. However, earlier efforts (Heintz 1996) had noted a mass ratio for the C component binary of about 1, and with C being a G0 star, the non-detection was puzzling and attributed to the companion being a white dwarf or itself a binary. Hutchings et al. (2000) finally resolved this pair (Ca,Cb) via AO observations at infrared wavelengths, designated it (Cb) as an M2 dwarf based on its infrared colors, and argued on the basis of prior mass-ratio estimates that it itself is an unresolved binary (Cb1,Cb2). Richichi (2000) confirmed the presence of Cb via lunar occultation measures. While she could not confirm its binary nature, her K-band photometry supported the binary M-dwarf hypothesis, for which she determined an upper-limit for projected separation of 20-30 mas. Further, Richichi reports the potential discovery of a sixth component in this system. While seen just above her detection limit and hence retained as a candidate for this work, she nonetheless confirmed its presence by three independent data analysis methods and excluded it from being the unresolved companion Cb2 noted above. This exclusion is primarily due to its larger separation of about at least 1.6 AU from the lunar occultations, and tentatively identified it as an M2–M4 dwarf. In addition to all this, I identified a potential wide companion, 372" away at 107°, but a physical association could be ruled out based on a significantly different proper motion of the companion of $\mu_{\alpha} = 0.000$ g yr⁻¹ and μ_{δ} = -0''.091 yr⁻¹ from UCAC2, compared to the primary's values of $\mu_{\alpha} = 0''.283 \text{ yr}^{-1}$ and $\mu_{\delta} =$ -0".150 yr⁻¹ from *Hipparcos*.

HD 72760: *Binary*. This companion was identified based on a significant difference in *Hipparcos* and Tycho-2 proper motions. Recently, Metchev & Hillenbrand (2008) resolved

the companion in a Palomar/Keck AO survey, confirmed companionship based on color and magnitude measurements, and estimated the companion's mass as $0.13 M_{\odot}$.

HD 73350: Single. The WDS lists a B component 60" away with a C component about 10" from B. While the DSS images were taken over just a two-year interval, the SSS image provides a longer time baseline and helps confirm component B (HD 73351) as a field star. Component C is a CPM companion of B based on three consistent measures separated by over 100 years, and hence also physically unassociated with HD 73350.

HD 73667: Single. While the blinking of archival images allow confirmation of the two WDS entries as field stars, it revealed a possible CPM companion 335'' away at 207° . However, its photometric distance estimate refuted this candidate as well (see § 4.1).

HD 73752: Binary. The CNS (Gliese 1969) lists the primary of the 1"3 visual binary as "SB" and notes that there are suspected perturbations in its proper motion. The reference detailing the perturbations (Hirst 1943) presents a 35-year inner orbit, which is noted as very preliminary with several different orbits equally permissible. The author also states that systematic effects alone may explain the residuals. His outer 214-year visual orbit was later revised to 145 years by Heintz (1968) with no mention of a third component. In fact Heintz pointed out that that the observed range of radial velocities could be ascribed to scatter. Adopting a parallax of 0"058, he derived a mass-sum of 1.1 M_{\odot}, and noted that at least one component must be over-luminous. If we adopt the HIP parallax of 50.2 mas, we get a mass-sum of 1.9, so the components are likely not over-luminous. Radial velocity catalogs (Abt & Biggs 1972; Gontcharov 2006) lists velocities in the range 40–48 km s⁻¹,

but the differences could be due to zero-point offsets between observers. The early claim of a possible companion is not supported by subsequent observations, which in fact question it. While the visual binary is real, the third component is refuted. An additional wide component listed in the WDS and measured 113.77 away in 1999 is clearly optical.

HD 75767: Quadruple. Tokovinin et al. (2006) reported the discovery of a wide $\Delta m=4.3$ companion to a 10.3-day SB1 binary with NACO adaptive optics and confirmed CPM using a partial resolution in 2MASS images. This companion was independently discovered by Fuhrmann et al. (2005), who obtained two observations four years apart, demonstrating CPM, and confirmed companionship by showing consistent radial velocity with the primary. Their spectra also enabled them to identify the companion itself as a double-lined binary, as evidenced by its H-alpha emission and near-infrared absorption lines appearing as pairs with an offset of about 21 km s⁻¹. Using composite-spectrum analysis, they derived spectral types of M3 and M4. Blinking archival images revealed a possible fifth companion 385" away, and its photometric distance estimate matches the primary's *Hipparcos* value within 2σ . However, the Lépine & Shara (2005) proper motions of the two stars are significantly different, indicating that this might be a comoving star perhaps born out of the same cloud as HD 75767, but one that is not gravitationally bound to it.

HD 79096: *Quadruple*. Wilson et al. (2001) discovered a L8V comp (Gl 337C) 43" from the SB2VB pair from 2MASS images. The two images, separated by 2.5 years, allowed confirmation of CPM. They also showed that the magnitudes are consistent with the primary's distance to within 1σ , confirming companionship. Burgasser et al. (2005) resolved Gl

337C as a nearly equal-magnitude binary (separated by 0.53 ± 0.03 at $291^{\circ} \pm 8^{\circ}$) using Lick natural guide star AO. Companionship was confirmed based on proximity and CPM, which was demonstrated by the absence of a source in 2MASS images at the expected position of a background star.

HD 82885: *Binary*. This is a visual binary with a period of some 200 years. A potential wide companion (NLTT 22106), 328" away at 333° was identified by blinking archival images but refuted based on significantly different proper motions.

HD 84117: *Single*. A potential wide companion (NLTT 22384), 722" away at 331° was identified by blinking archival images but refuted based on significantly different proper motions.

HD 86728: Binary, maybe Triple. Gizis et al. (2000) identified a wide CPM companion from 2MASS and confirm it via a spectral type of identification M6.5. However, based on it being over-luminous ($M_K = 8.19$ using 2MASS magnitudes and Hipparcos parallax versus $M_K = 9.60$ for an M6.5 dwarf) and having high activity (emission observed twice), they argued that it is an unresolved equal-mass binary, or even possibly a triple. I could not find any follow-up work confirming or refuting this claim, so while this system is confirmed as a binary, I retain a third component as a candidate.

HD 90839: *Binary*, maybe *Triple*. The primary of a wide CPM pair is listed in the CNS (Gliese & Jahreiß 1991) as "SB?" and the secondary (HD 237903, GJ 394) is listed in an earlier version of the catalog (Gliese 1969) as "RV-Var". The primary is a constant velocity star, as evidenced by Nidever et al. (2002) and the CfA survey (see § 6.1). These modern

surveys did not observe the secondary. DM91 listed this companion with a constant velocity of 8.24–8.62 km s⁻¹ over 700 days. Heintz (1981) listed velocities of 7.7–8.4 km s⁻¹ over four days and noted that coverage is too weak to show whether the velocity varies. He also noted that the spectrum had emission features. Wilson (1967) listed a velocity of 7.8 km s⁻¹ over three plates with a range of 7.7 km s⁻¹ and standard deviation of 3.1 km s⁻¹. Radial velocity catalogs (Abt & Biggs 1972; Duflot et al. 1995) list values that range over many km s⁻¹, but this could be due to zero-point differences between observers, and these catalogs do not note any variation. While the wide binary is confirmed based on matching parallax and proper motion and the primary's SB claim is refuted, the possible radial velocity variation of the secondary is inconclusive and hence retained as a candidate.

HD 96064: *Triple*. This system is composed of a CPM pair separated by 11.^{'8}, the secondary of which is a 23-year visual binary. Additionally, the blinking of archival images revealed a possible wide companion about 5' away, but this was refuted based on photometric distance estimates (see § 4.1).

HD 97334: Triple. Kirkpatrick et al. (2001) discovered an L4.5V CPM companion (Gl 417B) 90" away at 245° from the primary using 2MASS images and confirmed a physical association by demonstrating CPM and consistent parallaxes. Bouy et al. (2003) resolved this brown dwarf into a binary (0".070 \pm 0".0028 at 79°.6 \pm 1°.2) using HST WFPC2. While companionship of this pair has not been established conclusively, proximity argues for a physical association.

HD 97658: Single. A possible CPM companion was discovered by blinking archival images but refuted based on inconsistent photometric distance estimates (see \S 4.1).

HD 98230: Quadruple. ξ UMa is a quadruple system composed of a 60-year visual binary, the primary of which is a SB1VB and the secondary is an SB1. Mason et al. (1995) reported a possible fifth companion detected via speckle interferometry near the secondary. While the single detection reported is quite convincing, this companion has never again been seen, despite multiple attempts. My efforts with CHARA, while limited to $\Delta K \leq 2.5$, also failed to resolve any additional components. Given only one measure and about a dozen null results with the same technique, this new companion is probably spurious. For the purposes of this survey, I retain ξ UMa as a quadruple system.

HD 100180: *Binary*, maybe *Triple*. The primary of the 15" CPM binary has two speckle interferometry measurements of a close companion, observed 0".035 away at 6°.8 in 2001 and 0".122 at 355°.8 in 2004. One of the two attempts by B. Mason and me at the KPNO 4-m telescope in 2008 June resulted in an "uncertain" measure of 0".218 at 14°.6. Given the 0".378 yr⁻¹ proper motion of the primary, these measures are consistent with a bound pair, but further observations are warranted to obtain a definitive confirmation, especially given the constant radial velocity reported by Nidever et al. (2002). While the wide binary is confirmed by photometric distance estimates, the third component is retained as a candidate.

HD 100623: *Binary*. The WDS lists only a single measure of this large Δm pair discovered by Luyten in 1960. While the proximity and large magnitude difference make follow-up observations difficult, Henry et al. (2002) obtained spectra of this 15th magnitude

companion and showed that it is a DC or DQ white-dwarf, not an M-dwarf as reported in the CNS (Gliese & Jahreiß 1991). The second observation confirms CPM, and the spectral type and photometry are consistent with a physical association.

HD 102365: *Binary*. Luyten first resolved this pair in 1960. The companion is LHS 313 with a proper motion that matches the primary's $1\%6 \text{ yr}^{-1}$. Hawley et al. (1996) identified the companion as a M4V, which was recovered by 2MASS at a similar position angle and separation as Luyten and its infrared colors are consistent with an M4 dwarf at the primary's distance.

HD 102870: *Single*. The CNS (Gliese & Jahreiß 1991) lists this as "SB?", but Nidever et al. (2002) and the CfA survey show it to be a constant velocity star (see § 6.1). The WDS lists two additional components, B and C, which are clearly optical.

HD 103095: Single. The CNS (Gliese 1969) and DM91 listed a companion with separation 2" at 175°. DM91 mentioned that the companion was flaring with magnitudes of 8.5–12 and also mention that it is normally not seen. The FIC lists four null measurements with speckle interferometry and as shown in Table 6.1, there are no radial velocity variations. Three attempts by B. Mason and me in 2008 June at the KPNO 4-m telescope failed to identify a companion. Recently, Schaefer et al. (2000) have shown that the brightness enhancements observed are likely due to superflares on the stellar surface rather than due to a companion.

HD 109358: Single. The WDS lists a single speckle measure (Bonneau & Foy 1980) of a 0''.1 pair, but the FIC has over 20 null speckle detections. B. Mason and I observed this star

with the KPNO 4-m telescope in 2008 June and failed to resolve the suspected companion. Given the mention of telescope artifacts as being responsible for some detections by this observer (da Silva & Foy 1987), I side with the many null detections, including one by the same observer. Additionally, the CNS (Gliese & Jahreiß 1991) listed this star as "SB", and Abt & Levy (1976) presented a preliminary orbital solution, but that was later shown to be spurious (Morbey & Griffin 1987). CfA radial velocity coverage of 16 years, and the high precision measurements of Nidever et al. (2002) confirm this star as a stable radial velocity star. The AB pair in the WDS is clearly optical. This star appears to be single.

HD 111312: Binary, maybe Triple. This is a 2.7-year SB2, for which a new orbital solution is presented here based on CfA velocities. The WDS lists a single speckle measure with a separation of 0".089 at 91° in 2001, and the pair was seen again in 2006 with a separation of 0".050 at 44°.6 (B. Mason 2008, private communication). These measurements are consistent with the spectroscopic binary and more observations are needed to develop a visual orbit. The WDS lists an additional companion, 2".7 away with $\Delta m \sim 4$ pair based on Hipparcos and Tycho measures. The Hipparcos solution is flagged as "poor" quality, and there is no independent confirmation of this pair. Its orbital period, if real, would be too long to affect the velocities obtained over some 7 years. With no conclusive evidence to confirm or refute this companion, it is retained as a candidate requiring further observations.

HD 112758: Triple. This is triple system with an inner SB1 pair and a wider visual component which was first resolved by van den Bos in 1945 and then again in 1960 with $\Delta m \sim 5$. McAlister et al. (1987) recovered this pair in 1983, and the three observations show

evidence of orbital motion. The McAlister et al. measurement with speckle interferometry implies $\Delta m \lesssim 3$, suggesting that the companion may be variable. B. Mason and I attempted to resolve this pair at the KPNO 4-m telescope on 2008-06-13, but could not see it, perhaps because of the large magnitude difference.

HD 113283: Single, maybe Binary. The Hipparcos and Tycho-2 proper motions differ to greater than a 3σ significance indicating an unseen companion (see § 5.1.2), but in the absence of other conclusive evidence, this companion is retained as a candidate.

HD 113449: Binary. Hipparcos presents a photocentric orbit for this star with a period of about 231 days. Moore & Paddock (1950) noted this star as a radial velocity variable and Gaidos et al. (2000) mentioned that the velocity changed by 20 km s⁻¹ over 10 months, but no definitive orbit exists. The companion was resolved at the Palomar 200-inch telescope with aperture masking in 2007 January, 35.65 ± 0.6 mas away at $225^{\circ}2 \pm 0^{\circ}5$ with $\Delta H \sim$ 1.6, and confirmed at the Keck telescope more than a year later (M. Ireland 2008, private communication). With consistent astrometric and spectroscopic evidence, this is a bound pair.

HD 114613: Single. The CNS (Gliese & Jahreiß 1991) list this star as "RV-Var", but Murdoch et al. (1993) show it to be a constant velocity with a scatter less than 0.1 km s⁻¹ over almost three years. Radial velocity catalogs also list stable velocities with no mention of variability.

HD 114783: Single, one planet. A potential CPM companion was identified 240" away at 46° by blinking archival images but subsequently refuted based on photometric distance
estimates (see $\S4.1$).

HD 120136: Binary, one planet. τ Boo hosts a 4.13 M_J minimum-mass planet in a 3day orbit and has a visual orbit stellar companion, measured 2".8 away at 33° in 2000. While the visual orbit is very preliminary, physical association is confirmed by 56 observations in the WDS over 170 years which demonstrate not only CPM, but also orbital motion. Furthermore, Wright et al. (2007) and references therein mention a longterm drift in radial velocity which is likely consistent with this visual companion.

HD 120780: Triple. The WDS lists two measures of a 6" pair. The measurements are 51 years apart and consistent with a bound pair, but follow-up observations have been difficult due to a magnitude difference of ~ 5.5. With the help of the SMARTS Consortium, I obtained I-band images in 2006 July and 2007 June. The companion was seen at both epochs about 5".6 away at 89° with $\Delta I \sim 3.3$. These three observations demonstrate CPM with a fast-moving primary (0".6 yr⁻¹), and, in fact, hint at orbital motion, confirming companionship. Additionally, Hipparcos identifies this star as an accelerating proper-motion binary, and the Tycho-2 proper motion differs from the Hipparcos value to a 19 σ significance. While the Tycho-2 proper motion, averaged over about 100 years, is no doubt affected by the wide pair mentioned above, whose orbital period could be about 1000 years, the Hipparcos observations are over some three years and indicate a closer companion. I conclude that this is a triple star system.

HD 124850: *Single*, maybe *Binary*. The VB6 catalog lists a photocentric orbit for this star with a period of 55 years and an inclination of 60° from Gontcharov et al. (2003, A&A

in press), but this reference could not be found. This star was not included in the Nidever et al. (2002) or CfA radial velocity surveys, and radial velocity catalogs (Abt & Biggs 1972; Duflot et al. 1995; de Medeiros & Mayor 1999; Gontcharov 2006) do not indicate variation.
B. Mason and I could not resolve any companion via speckle interferometry on the KPNO 4-m telescope in 2008 June. Retained as a candidate.

HD 125276: Single, maybe Binary. The Hipparcos and Tycho-2 proper motions differ to greater than a 3σ significance indicating an unseen companion (see § 5.1.2), but in the absence of other conclusive evidence, this companion is retained as a candidate. The WDS lists two companions for this star, the wider one of which, 83" away in 1999, is clearly a background star as seen by blinking archival images. The closer pair is separated by 3"– 8" over four measures listed in the WDS between 1891 and 1936. Some of these measures indicate a $\Delta m \sim 8$, which might explain the several non-detections also included in the WDS. B. Mason and I observed this star at the KPNO 4-m telescope in 2008 June. While the pair is too wide and too high in contrast for detection via speckle, the finder TV at the telescope revealed a faint stellar source about 60" away at 130°. No other source was found 6"-80" away from the primary.

HD 125455: *Binary*. This pair was discovered by Kuiper in 1937 and has measurements in 1960 and 1987 that are consistent with as bound pair. The companion is LHS 2895 with a proper motion that matches the primary's, and its 2MASS colors indicate a late M-dwarf at approximately the primary's distance.

HD 128620: Triple. This is the closest star system, α Centauri, which is composed of a

an SB2VB pair and a wide companion, Proxima Centauri, about 2° away. While the angular separation is extreme for bound systems, it translates to a linear projected separation of 10,000 AU, which is well within the limits of gravitationally bound pairs. Wertheimer & Laughlin (2006) used kinematic and radial velocity data to show that Proxima Centauri is bound to α Centauri. A possible new companion to Proxima Centauri was reported by Schultz et al. (1998) 0".34 away using the HST FOS as a coronagraphic camera. In a followup effort, Golimowski & Schroeder (1998) use HST WFPC2 to show that the FOS feature seen was likely not a substellar companion to Proxima Centauri, but rather an instrumental effect of the FOS, noting that if the prior observation was real, it would have easily been identified by WFPC2, given its expected separation and contrast. They exclude any stellar or substellar companion within 0".09–0".85 of Proxima Centauri.

HD 130948: Triple. Potter et al. (2002) discovered a pair of brown dwarf companions using AO on the Gemini North 8-m telescope. They demonstrated CPM with observations over seven months and confirmed companionship based on their infrared colors, spectral-type of dL2 \pm 2 and a consistent age with the primary of less than 0.8 Gyr derived by placing these stars on an HR diagram and comparing with theoretical models. They also noted that the young age is consistent with the high X-ray activity, Li abundance, and fast rotation. Additionally, this star is listed as SB in the CNS (Gliese 1969), but clearly seen as a constant velocity star in Nidever et al. (2002) and the CfA survey (see § 6.1). This system apparently has one star and two brown dwarfs.

HD 137107: Triple. Kirkpatrick et al. (2001) discovered this wide L8V companion (Gl

584C) to the SB2VB binary using 2MASS images, and confirmed as a physical association with additional measures and spectroscopy.

HD 137763: *Quadruple*. The primary of a 52" CPM binary is itself SB2, and also has a wide companion about 20' away that was first mentioned by CNS and confirmed based on a spectral type of M4.5 and a distance estimate of 21.6 ± 1.9 pc (Reid et al. 1995).

HD 140901: *Binary.* The WDS lists seven measures for this pair from 1897 to 1960 which are consistent with a bound pair. With the help of the SMARTS Consortium, I obtained *VRI* images from the CTIO 0.9-m telescope in 2006 July and 2007 July, which reveal the companion at the expected position, confirming CPM. The magnitude difference of over six makes photometry difficult, but given the large matching proper motion and proximity, this pair can be confirmed as physical. The WDS C component is clearly optical.

HD 141004: *Single.* Two candidate companions were refuted for this star. A CPM companion identified 235" away by blinking archival images was refuted by photometric distance estimates (see § 4.1). Additionally, the CNS (Gliese & Jahreiß 1991) designates this star as "SB1", which is spurious as indicated on the CNS website and by constant velocities observed by Nidever et al. (2002) and the CfA survey. This star appears to be single.

HD 141272: *Binary.* The WDS lists four micrometer observations of this pair over 56 years that are consistent with a bound pair. Eisenbeiss et al. (2007) confirm companionship based on photometry and spectroscopy and derive an estimated mass for the dM3 \pm 0.5 companion of $0.26^{+0.07}_{-0.06}$ M_{\odot}.

HD 142373: Single. Hipparcos lists a photocentric orbit for this star with a period of 51

days and an inclination of $132^{\circ} \pm 28^{\circ}$. This is not consistent with constant radial velocities observed by Nidever et al. (2002) to within 0.1 km s⁻¹ and CfA (see § 6.1). There is no other published evidence of a companion to this star.

HD 143761: Single planet-host or binary with no known planets. ρ CrB definitely has a companion, but it is not clear if it is planetary or stellar in nature. *Hipparcos* identified a photocentric orbit with a period of 78 days, exactly twice that of the planetary companion reported by Noyes et al. (1997). Gatewood et al. (2001) used *Hipparcos* and ground-based observations to conclude that the photocentric orbit is of the same period as the planet, and in fact the "planet" is an M-dwarf companion with a mass estimate of 0.14 M_{\odot} in a nearly face-on orbit. Bender et al. (2005) failed to identify such a companion using highresolution infrared spectroscopy, and placed an upper limit on the companion's mass of 0.11– 0.15 M_{\odot}. Baines et al. (2008b) attempted to resolve this question with LBI observations at the CHARA Array, and could not settle the issue once again. While interferometric visibilities did not perfectly fit a single-star solution, they indicate that additional data is required for a definitive conclusion. This system has a stellar or planetary companion, but not both. Further observations are warranted.

HD 144284: *Binary.* Mazeh et al. (2002) presented a 3-day SB2 orbit for this star using infrared spectroscopy to measure the faint companion, deriving a mass ratio of $0.380 \pm$ 0.013. Mayor & Mazeh (1987) had identified this system as a possible triple based on a 1.7 km s⁻¹ variation in the velocity semiamplitude between their solution and that of Luyten (1936). While the velocity semiamplitude does seem to vary for the different orbital solutions presented for this pair (Luyten 1936; Mazeh et al. 2002, DM91) and the CfA SB1 orbital solution has residuals of up to 2 km s⁻¹ on each side, there is no obvious periodic pattern or longterm drift over the 4.8 years of velocity coverage (D. Latham 2008, private communication). The most recent CfA velocity measure of this star is from 1990, and additional observations are warranted.

HD 144579: *Binary*. The proper motion of the candidate from LSPM is quite different from the primary's *Hipparcos* or LSPM value. Further, the distance estimate has a large error and matches the primary's at a greater than 1σ significance (see Table 4.2). However, given the proximity of these two stars in the sky, the very large and similar proper motions and similar distances, this appears to be a physical companion, as it has been previously recognized (DM91, Gliese & Jahreiß 1991). The differences in the proper motions might indicate that the companion (or primary) has a close unresolved companion and warrants further observations. For this work, I will consider this a binary.

HD 145958: *Binary*, maybe *Triple*. The primary of a 4" visual binary has two additional possible companions, one of which was refuted by this effort and the other remains a candidate. The WDS lists a nearby companion, 0"2 away, detected by H. McAlister in 1983. The FIC catalog lists this as a weak detection and possibly spurious, and includes a null detection using the same technique. B. Mason and I failed to resolve a companion at the KPNO 4-m telescope in 2008 June. Nidever et al. (2002) identifies this as a constant velocity star. Evidence seems to be mounting against this candidate companion, which is considered refuted for this work. Separately, the Dwarf Archives² includes a T6 object about

 $^{^{2}} http://spider.ipac.caltech.edu/staff/davy/ARCHIVE/index.shtml$

27' away from this star. Looper et al. (2007) discovered this T dwarf in the 2MASS survey, obtained spectra, typing it as T6, and estimated its proper motion and distance. Their proper motion of 0".48 \pm 0".05 yr⁻¹ at 139°.0 \pm 0°.2 is similar to the primary's FvL07 value of 0".46 yr⁻¹ at 156°. Additionally, their distance estimate of 29.0 \pm 2.3 pc is within 3 σ of the primary's FvL07 distance of 23.6 pc. While the projected linear separation is very large at about 40,000 AU, this could be a loosely bound companion to HD 145958, and is retained as a candidate.

HD 146361: *Quintuple*. As part of this effort, I derived a visual orbit for the central 1.14-day binary of this quintuple system. For a comprehensive treatment of all components, see *Appendix* E.

HD 147776: Binary, maybe Triple. The WDS lists three candidate companions, but the details actually correspond to four stars. The $\Delta m \sim 4$ pair 103" away at 281° is clearly a field star, as seen by blinking the archival images (see § 4.2). Three additional pairs were reported by Sinachopoulos (1988) – 6".4 separation at 173° with $\Delta m \sim 3$, 9".7 separation at 14°, and 71".9 separation at 28°. The latter two pairs do not have a magnitude difference measurement. Sinachopoulos measured these pairs using a 1.5-m telescope by combining 4–16 exposures of a few seconds each. The wide companion 72" away should have been seen in the DSS images, but no stellar source was seen at the expected position. The closest star to this position in the 1995 DSS image is 83" away at 15° and is clearly a field star. The other two sources seen by them would be buried in the saturation around the primary in the DSS images, so I obtained VRI frames in 2008 May and August from the CTIO 0.9-m telescope which is operated by the SMARTS consortium. The images clearly show a faint companion about 9" away at 19°. This is likely the 9".7 companion seen by Sinachopoulos (1988), exhibiting CPM, and given the proximity, is likely physical. The closest source seen by Sinachopoulos is not detected in the CTIO images and remains a candidate. Additionally, the CNS (Gliese 1969) lists a companion for this star 3" away at 281° in 1909. This is likely the same as the pair measured by Burnham as listed in the WDS, which was seen 103" away at 281° in 1909 and as discussed above, is clearly optical.

HD 148704: Binary. This is a 32-day SB2 binary for which Hipparcos and Tycho identified another companion 4".1 away at 221°. I obtained VRI images from the CTIO 0.9m telescope in 2008 October with the help of the SMARTS Consortium. No companion was seen at the expected position, while a $\Delta m \sim 3$ companion as indicated by Hipparcos should have been seen above the tail of the primary's PSF. However, given the proper motion of the primary, a field star would have moved closer and possibly could be buried within the primary's PSF. Gray et al. (2006) list the spectral type of the companion as G9V and its coordinates imply a separation of 2".4 at 55°, the exact position where a field star would be fifteen years since the Hipparcos measure. The Gray et al. spectral type, along with the Tycho-2 V-magnitude of 10.5 imply a significantly larger distance to this star compared to the primary, enabling me to refute this candidate.

HD 149806: *Binary.* This pair was first reported by Rossiter (1955) 5.9 away at 22° and has two additional measurements in the WDS over the next 54 years, which are consistent with a bound pair. While the photometric distance estimate is not a good match

(see Table 4.2), the *R* magnitude listed is likely approximate. Given the moderate proper motion of the primary, the consistent measures over 54 years indicate a physical association. The 2MASS colors indicate an M-dwarf with a V magnitude estimate of about 12, in fair agreement with the measure of Rossiter (1955). On In 2008 June, B. Mason and I attempted to observe this pair at the KPNO 4-m telescope. While the separation and Δm are too large to be resolved using speckle, the finder image at the telescope showed a source at the expected position with a Δm similar to those of prior observations.

HD 153557: Triple. The WDS lists 17 measurements over 95 years with separations ranging from 1."9–4."9, which are consistent with a bound pair, and given the 0."3 yr⁻¹ proper motion of the primary, imply a physical association. This pair also has a wider companion, HD 153525, about 2' away, which is confirmed by matching proper motion and parallax.

HD 155885: Triple. This system comprises a 470-year VB and a wide companion (HD 156026) about 12' away. Additionally, the CNS (Gliese 1969) list this star as "RV-Var", but the CfA survey shows that this is a constant velocity star (see \S 6.1), allowing me to refute this component. WDS components D and E are optical.

HD 158633: Single. The CNS (Gliese 1969; Gliese & Jahreiß 1991) list this as "SB" and note a radial velocity range of 28 km s⁻¹. CfA observations span 28 years and indicate a constant velocity star with an RMS scatter of 0.46 km s^{-1} , in line with typical measurement errors.

HD 165341: *Binary.* CNS lists the component A of a 88-year SB2VB as a possible binary with a period of about 17 years. Radial velocity data from CfA do not show any

evidence of variation in 26 observations over 13 years. Heintz (1988) presents a revised orbit of the SB2 and points out that previous efforts to explain the residuals by a third unseen companion have led to contradictory results on the nature of the companion. He excludes the possibility of any period with P < 55 years and states that the once suspected velocity variation of A is disallowed by the more precise recent measurements.

HD 165908: *Binary*, maybe *Triple*. This is a 56-year VBO. Additionally, the WDS lists one speckle measure of a pair with a separation of 0.228 at 50.2 from Scardia et al. (2008), who list this new discovery as "faint". They also resolved the known VB companion about 1″ away, and noted it as "very faint". In the absence of additional measures that can help confirm CPM, this close pair is retained as a candidate.

HD 178428: *Binary.* The primary of a 22-day SB1 has a single 1987 measure listed in the WDS with a separation of 0.2. Repeated attempts by speckle interferometry have failed to confirm this pair. The FIC lists six null results and attempts by B. Mason and I at the KPNO 4-m telescope in 2008 June once again failed to reveal any companion.

HD 182572: *Single.* The CNS (Gliese 1969; Gliese & Jahreiß 1991) lists this is as "SB?" and "RV-Var?" in the two versions of the catalog. Nidever et al. (2002) show this to a constant velocity star with a scatter less than 0.1 km s^{-1} in observations over four years, which is confirmed by other high-precision radial-velocity campaigns (A. Hatzes 2008, private communication).

HD 184385: Single. The CNS (Gliese 1969) list this as "RV-Var", but Nidever et al. (2002) and the CfA survey show this to be a constant velocity star (see § 6.1).

HD 186408: Triple, one planet. This close companion to 16 Cyg A was first resolved 3" away by Turner et al. (2001) with AO at the Mount Wilson Observatory and confirmed by Patience et al. (2002b), who demonstrated CPM and measured infrared magnitudes consistent with the primary's distance. Four CfA velocity measures over 25 years show a slow downward drift, consistent with this companion. This system also has a wide companion, 16 Cyg B, which is a planet host. The WDS lists an additional source, 16" away from 16 Cyg B, but Patience et al. (2002b) measured the infrared magnitudes of this candidates, demonstrating that it is a background star. This is the only planetary system in this study with more than two stars.

HD 190067: *Binary*. This pair was discovered by Turner et al. (2001) with AO at the Mount Wilson Observatory, but the single-epoch measure with no color information does not allow confirmation of a physical association. B. Mason and I observed this pair at the KPNO 4-m telescope in 2008 June. While the separation and Δm are too large for speckle observations, a stellar source was seen at the expected position, confirming CPM, and given the proximity to a large proper motion (0".6 yr⁻¹) primary, the physical association of this pair is very likely.

HD 190406: *Binary.* Liu et al. (2002) discovered a faint companion 0.8 from this star with AO at the Gemini North and Keck II telescopes and confirmed a physical association by demonstrating CPM, consistent spectroscopy, and longterm radial-velocity trend. They determined a spectral type for the companion of L4.5 \pm 1.5, estimated its mass to be 55– 78 M_J and age as 1–3 Gyr. This is the first substellar object imaged so close to a solar-type star and indicates that brown dwarfs can exist in extrasolar systems at positions comparable to the gas giants in our solar system.

HD 191499: Binary. The WDS lists 51 measurements of this pair between 1782 and 2003, which are all consistent with a bound pair. There is little evidence of orbital motion during the roughly 200 years of observations, possibly because the companion is near apastron or the orbit is highly inclined. *Hipparcos* and Tycho-2 proper motions differ by 5.6σ , indicating some orbital motion (see Table 5.2). The photometric distance estimate is not a very good match (see Table 4.2), but photometry would be tricky for this close pair as indicated by the large uncertainties of the 2MASS magnitudes. Given the evidence of consistent WDS measures, proper motion differences between *Hipparcos* and Tycho-2, and similar distance estimates, this pair likely has a physical association.

HD 192310: Single. The CNS (Gliese 1969) designates this star as "SB?", but there is no independent confirmation. On the contrary, published velocities show little variation over a long period of time (Kennedy & Przybylski 1963; Abt & Biggs 1972; Beavers & Eitter 1986; Duflot et al. 1995; Gontcharov 2006).

HD 193664: Single, maybe Binary. The Hipparcos and Tycho-2 proper motions differ to greater than a 3σ significance indicating an unseen companion (see § 5.1.2), but in the absence of other conclusive evidence, this companion is retained as a candidate.

HD 195564: *Binary*. The WDS lists 16 measures over 110 years that are consistent with a bound pair. While proximity to the primary and $\Delta m \sim 5$ (from WDS) make photometry of the companion difficult, the proximity and CPM implied by the measures argue for a physical association.

HD 200525: Triple. The CNS (Gliese 1969) and Hipparcos identified the closer pair as a possible binary (stochastic solution) and Goldin & Makarov (2006) derived a photocentric orbit using the *Hipparcos* intermediate astrometry data. Their orbital solutions using data from the two independent *Hipparcos* reduction methods, Fundamental Astronomy by Space Techniques (FAST) and the Northern Data Analysis Consortium (NDAC), are consistent. The FAST data yielded $a_0 = 68^{+82}_{-27}$ mas, $P = 2145^{+3036}_{-914}$ days, $e = 0.70^{+0.10}_{-0.08}$, $\Omega = 13^{+180\circ}_{-4}$, $\omega = 148^{+32}_{-135}$ °, $i = 90^{\circ} \pm 2^{\circ}$, and $\pi = 51 \pm 1$ mas, and the NDAC orbit resulted in $a_0 = 88^{+106}_{-41}$ mas, $P = 2429^{+3089}_{-1104}$ days, $e = 0.65^{+0.11}_{-0.07}$, $\Omega = 17^{+180\circ}_{-3}$, $\omega = 147^{+32}_{-136}$ °, $i = 91^{\circ} \pm 1^{\circ}$, and $\pi = 51 \pm 1$ mas. They tested their orbit determination method satisfactorily against 235 known binaries and derived a better than 99% confidence level based on simulations. The WDS lists four measurements from 1898 to 1932 during which time the separation reduced from about 1'' to 0''.16. The respective position angles of the measurements fall in the first, second, and third quadrants, and are consistent with a high-inclination orbit. Given these independent measurements leading to similar results, I conclude that while the orbital elements may be preliminary, this pair is physically bound. For the wider pair, the WDS has five measures over 88 years that are also consistent with a bound pair. The companion is NLTT 50542 with a proper motion that matches the primary's, and the notes in the catalog recognize this pair as gravitationally bound. 2MASS magnitudes have a large error, but, within 1σ , are consistent with a mid-K dwarf companion at approximately the primary's distance.

HD 200560: *Binary.* The WDS has seven measurements of a pair with separations ranging from 2."8–3."3 over 28 years and are consistent with a bound pair. This is especially significant given the primary's large proper motion of 0."4 yr⁻¹. The companion, GJ 816.1B, is recognized in the CNS as bound, although no conclusive evidence is presented. The 2MASS photometry has large errors, and hence is not very useful. The *Hipparcos* and Tycho-2 proper motions are different to about 8σ , providing evidence of orbital motion and lending credibility to a physical association. The WDS lists this pair as the CD component of the B3V binary HD 200595 AB, but there is clearly no physical association between HD 200560 and HD 200595 as seen by blinking archival images.

HD 202275: *Binary.* This is a 5.7-year SB2VB. Tokovinin et al. (2006) gives an additional orbit with a period of 5.7 days, which is in fact the former orbit listed with an incorrect unit (A. Tokovinin 2007, private communication). This system is a binary with mass estimates of 1.2 M_{\odot} and 1.1 M_{\odot} by Pourbaix (2000).

HD 203244: Single. Hipparcos lists a photocentric orbit for this star with a period of 2.9 years and an inclination of $110^{\circ} \pm 3^{\circ}$. Hipparcos and Tycho-2 proper motions match to within 1σ . Radial velocities in Abt & Biggs (1972) and Gontcharov (2006) indicate a constant velocity star.

HD 206860: *Binary*. Luhman et al. (2007) report the discovery of a T2.5 \pm 0.5 companion using Spitzer IRAC images and confirm CPM using 2MASS images. The infrared colors are consistent with the distance to the primary, confirming companionship. By comparing the luminosity with evolutionary tracks, they estimate the companion's mass as 0.021 \pm 0.009 M_{\odot} and age as 0.3 ± 0.2 Gyr. Additionally, a potential wide companion, 591" away at 16°, was identified by blinking archival images but refuted because its proper motion of $\mu_{\alpha} = 0$. (111 yr⁻¹ and $\mu_{\delta} = -0$. (089 yr⁻¹ from Ducourant et al. (2006) is significantly different from the primary's *Hipparcos* value of $\mu_{\alpha} = 0$. (231 yr⁻¹ and $\mu_{\delta} = -0$. (114 yr⁻¹). Photometric distance estimates to the candidate companion show it to be a distant field star.

HD 217107: Single star with two planets or a binary with one planet. The WDS lists two measurements, fifteen years apart, of a companion $0^{\prime\prime}_{...3}$ away from this star, which also hosts two planets. These speckle interferometry detections could however not be confirmed by the same technique on at least three other occasions, indicating that this pair might have a large or varying Δm . Interestingly, the farther planet is one of the most widely separated planets reported, lying at least 5 AU from the star. Vogt et al. (2005) present orbital solutions with periods of 7–9 years, but mention that it could be three times larger. Wright et al. (2008) present an updated orbit with $P = 11.5 \pm 0.5$ years and $a = 5.27 \pm 0.36$ AU. At the 20-pc distance to the star, these separations are consistent with the speckle observations. Given the inconsistent measures, if we assume a ΔV near the speckle limit of about 3, the companion to the G8 IV-V primary (Gray et al. 2003) could be an early M-dwarf. The mass-sum of such a binary is consistent with the Wright et al. (2008) orbital elements. Vogt et al. (2005) note that an AO image obtained with the Keck telescope did not reveal any stars beyond 0.1 from the primary, and Chauvin et al. (2006) confirm this null result with VLT and CFHT AO observations. The M-dwarf companion would also imply a significantly larger velocity semi-amplitude for the primary, but that possibility is not convincingly excluded by the orbital plot in Wright et al. (2008). While it appears that this "planetary" companion could be a star, further observations are warranted.

HD 220140: Triple. The WDS has five measurements over 100 years for the closer visual companion at separations of about 10"that are consistent with a bound pair. The companion is NLTT 56532 with a proper motion matching that of the primary. 2MASS colors of the companion indicate an early M-dwarf at approximately the primary's distance. The wide CPM companion, 16' away, was first identified by Lépine & Shara (2005). Makarov et al. (2007) confirm companionship by obtaining a trigonometric parallax of 51.6 \pm 0.8 mas for the companion, which agrees well with the primary's *Hipparcos* parallax of 50.7 \pm 0.6 mas. Their BVRI photometry along with 2MASS near-infrared magnitudes show that this star is over-luminous in the K_s band, confirming its suspected pre-main-sequence status, and enabling an age estimate of 12–20 Myr.

HD 222368: Single. The CNS (Gliese & Jahreiß 1991) lists this as "SB?", but Nidever et al. (2002) confirms it as a constant velocity star.

Astronomy compels the soul to look upwards and leads us from this world to another.

---- Plato

DISCUSSION AND ANALYSIS

- 8 -

8.1 Comparison with DM91 Multiplicity Statistics

DM91 studied 164 solar-type stars in the solar neighborhood and derived observed multiplicity percentages for Single:Double:Triple:Quadruple systems of 57:38:4:1. They also identified candidate companions for stars with $P(\chi^2) < 0.01$, and when these are also included, the percentages change to 50:41:7:2. The current study with a sample of 454 primaries yields corresponding numbers of 57:33:8:3 for confirmed systems and 54:34:10:3 when candidates are included. To compare these results meaningfully, I first estimated the uncertainties for each of these numbers using bootstrap analysis with 10,000 iterations, as explained in §7.2. The results of the current study are listed in Table 7.2, and the corresponding fractions for the DM91 confirmed pairs are $57 \pm 4 : 38 \pm 4 : 4 \pm 1 : 1 \pm 1$, for which Figure 8.1 shows the distribution of the multiplicity frequencies. The uncertainties are larger for the DM91 results because their sample is about a third the size of the current work.

These results reveal the following interesting points:

(i) The percentage of triple and quadruple systems in the current study is double that of the DM91 results. Additionally, the current study includes two quintuple systems while DM91 had none. These differences are significant to 3σ for triple systems and 2σ for quadruples. The current results show that 24% of non-single stars are higher-order than binaries, compared to 13% in DM91, confirming their prediction that additional multiple systems were



FIGURE 8.1: The DM91 frequency of single, binary, triple, and quadruple systems. Symbols and methods are same as Figure 7.2

likely to be detected among nearby solar-type stars. Apparently, the attention garnered by this sample of stars has indeed revealed these multiple systems missing in the earlier study. Could the present work still underestimate the percentage of higher-order multiples? That is certainly a possibility, especially given my approach of matching suspected or confirmed companions by different methods. Whenever companions revealed by different methods were consistent with indicating the same physical companion, I assumed that they did. Sometimes this could be confirmed, as in the case of spectroscopic and visual orbits with the same orbital period, but often all I had was a consistency check. For example, does a slow linear drift in radial velocities over, say, 20 years indicate a new companion or is it measuring the known 250-year visual pair? The nature of hierarchical systems indicates that these two measurements relate to the same companion, and this is the assumption I make. Similarly, proper motion acceleration seen for a component of a visual orbit pair is assumed to not indicate an additional component. It is important to note that even if this approach misidentifies two different companions as one, a later rectification can only enhance the multiplicity order of binaries or higher-order systems, but will not change the estimated percentage of single stars.

(ii) While the percentage of multiple systems has doubled, it has come solely at the expense of binary systems. It is indeed remarkable that despite almost twenty years of monitoring since DM91, the observed percentage of single stars remains steadfastly fixed at 57%. This, despite not only a great deal of additional monitoring with techniques available at the time of DM91, but also with far improved sensitivity levels and entirely new methods. This could either be because the current study has missed many companions to presumed single stars or because the majority of Sun-like stars are indeed single. One way to answer this question is to see if binary systems have received greater attention in finding additional companions while single stars have been neglected. While it is true that some studies focus on higher-order multiplicity (e.g. Tokovinin et al. 2006), single stars have also received a lot of attention, and perhaps a lot more attention, given that the high-precision radial velocity search efforts avoid close binaries. I will return to this in the next section on incompleteness

analysis, but the data seem to indicate that the fraction of single Sun-like stars estimated from observations may be coming close to the real answer.

(iii) The reduction in the percentage of single stars when candidates are included is smaller in the current study (3%) when compared to DM91 (7%). This is also a consequence of the varied efforts in monitoring these Sun-like stars and the comprehensive nature of this study. The DM91 candidate list was mostly comprised of suspected spectroscopic binaries based on $P(\chi^2) < 0.01$, which I have shown to be an unreliable metric in Chapter 6. In contrast, the candidate list of this study is what remains so after inspecting every suspected companion revealed from many different methods, by using other means to decide their ultimate fate. A majority of the initial candidates were thus confirmed or refuted during the course of this effort. The details of this process have been covered in preceding chapters, but I mention a couple of examples here. Most suspected radial-velocity variables in DM91 or in the CNS were identified as either bonafide binaries or spurious entries using the results from modern monitoring surveys. On the other hand, ongoing measurement of visual pairs and the blinking of digitized images helped refute many suspected companions as they clearly did not exhibit CPM or helped confirm suspected pairs through observed orbital motion. Section 7.4 describes the rationale for making a lot of these decisions, which have led to a shorter, more qualified list of candidates.

To further check my results with the DM91 study, I investigated the 106 systems common to the two surveys. Results of bootstrap analysis with 10,000 iterations for confirmed companions, as explained in $\S7.2$, are listed in Table 8.1. The first row lists the DM91 results for

their entire sample of 164 stars, and the second row only includes their multiplicity results for the 106 stars in common with this work. As expected, these results are fairly consistent. The third row lists the overall ratios of this work for the sample of 454 stars, and the fourth row lists the statistics for the same set of 106 stars as in the second row, but using the multiplicity results of this work. While the results for triples and quadruples are consistent with the full sample, the single star percentage drops by 8% and the binaries increase by 7%. At first glance this appears to be significant as the deviations are significant to about 1.5σ . Why did this work result in a noticeable change in the fraction of single stars for these 106 stars compared to DM91, while such a result was not seen in the overall sample? An investigation of the individual systems revealed that three of the DM91 companions are now refuted, two by blinking archival images to show that suspected companions are field stars and one (HD 103095) by leveraging recent work which showed that superflares intrinsic to the star were incorrectly interpreted as a varying companion (see $\S7.4$). On the other hand, new companions have been confirmed in 16 of these systems, three by identifying CPM companions by blinking multi-epoch images, two SB1 with $K_1 = 4 - 10 \text{ km s}^{-1}$, two by speckle interferometry, two more by visual orbits, six faint companions by AO or infrared imaging, and one by accelerating proper motion. I will return to this discussion in the next section on incompleteness analysis as it appears that many companions suspected by DM91 as missed have indeed been found. The puzzle being considered here is why these 106 stars are rich in companions compared to the overall sample. Is the difference merely statistical scatter or indicative of something more significant? To test this, I selected 10,000 random subsets of 106 stars from the sample for which the results are listed on the fifth row of Table 8.1. These results are very consistent with the overall results suggesting that the particular set of 106 stars overlapping with DM91 is simply statistical wander.

Sample	Ν	Single	Binary	Triple	Quadruple+
DM91, overall	164	57 ± 4	38 ± 4	4 ± 1	1 ± 1
DM91, common	106	55 ± 5	38 ± 5	3 ± 2	1 ± 1
This work, overall	454	57 ± 2	33 ± 2	8 ± 1	2 ± 1
This work, common	106	49 ± 5	40 ± 5	9 ± 3	2 ± 1
This work, subset	106	55 ± 5	34 ± 4	8 ± 3	3 ± 2
This work, dec $>-15^\circ$	307	54 ± 3	35 ± 3	8 ± 2	3 ± 1
DM91, overall DM91, common This work, overall This work, common This work, subset This work, dec $> -15^{\circ}$	104 106 454 106 106 307	57 ± 4 55 ± 5 57 ± 2 49 ± 5 55 ± 5 54 ± 3	36 ± 4 38 ± 5 33 ± 2 40 ± 5 34 ± 4 35 ± 3	4 ± 1 3 ± 2 8 ± 1 9 ± 3 8 ± 3 8 ± 2	$ \begin{array}{c} 1 \pm 1 \\ 1 \pm 1 \\ 2 \pm 1 \\ 2 \pm 1 \\ 3 \pm 2 \\ 3 \pm 1 \end{array} $

 TABLE
 8.1: Comparison of Multiplicity Statistics with DM91

Are there other factors that could explain the richness of companions in stars common to DM91? I explored two other avenues and did not find anything significant. First, the current sample is all-sky, but DM91 was limited to declination north of -15°. Could it be that the southern hemisphere stars included in this study are less studied and hence missing companions? The last row of Table 8.1 shows the multiplicity fractions for the 307 stars of this study with declination > -15° . Indeed, the percentage of single stars drops by 3%, a 1σ departure, but not enough account for the 8% difference seen for overlapping stars. Moreover, the correction for missed systems that is handled in the next section addresses this gap in deriving the true multiplicity fractions. The second factor I consider is the difference in color range between the samples of this effort and that of DM91. This is covered in §8.3.1 and again seen to not explain the difference seen for the overlapping stars. Finally, I considered if the DM91 sample was perhaps biased to a selection of binaries? While their work specifically selected a volume-limited sample to avoid such effects, we now know that their source for parallax, the CNS catalog (Gliese 1969), has significant errors as seen in Figure 2.5. While there is no reason to believe that the Gliese catalog would systematically overestimate the parallax of brighter stars, Figure 8.2 shows that almost all stars significantly above the main sequence are ones now known to lie outside their criteria, illustrating that the distributions around the main sequence for stars selected from *Hipparcos* and Gliese (1969) are substantially different, in contrast to the results presented in Halbwachs et al. (2003). However, most of these outliers appear to be evolved stars which are expected to have fewer detected companions because older stars have a greater chance of losing companions through dynamical interactions and the increasing Δm as one component evolves makes it harder to find companions. The percentage of binaries among these outliers is under 35%, so confirming the above suspicion. Hence, the extraneous stars in the DM91 sample do not introduce a bias in their multiplicity results.

So, we are left with two possibilities. Either the thoroughness with which the 106 overlapping stars have been studied is higher than the rest of the stars in the sample, causing us to underestimate the binary fraction, or the difference seen is strictly statistical scatter. The coverage of the stars presented in the previous chapters does not suggest any bias towards a better analysis of any particular subset of stars, which is also supported by the data on the fifth row of Table 8.1, leading me to conclude that the difference in the fourth row of the table is simply within the expected distribution.



FIGURE 8.2: HR diagram for the DM91 sample using magnitudes and parallax from *Hipparcos*. Small filled circles represent stars in the DM91 sample that are still consistent with their criteria using *Hipparcos* data while large open circles are now known to fall outside the DM91 selection criteria. The dashed line is the main sequence from Cox (2000).

8.2 Incompleteness Analysis & True Stellar Multiplicity

This survey of nearby solar-type stars benefits from the tremendous attention these stars have garnered over the years from observers using many different techniques. Various published results from many surveys have augmented the observations of this survey, resulting in excellent coverage for spectroscopic and high-resolution astrometric techniques suited for short-period systems, to systematic searches for the widest companions out to the limits of gravitational binding. A usual deterrent in collecting comprehensive information from prior efforts is that while positive results are generally published promptly, null results often remain unpublished, keeping the rest of us guessing as to the true nature of unreported systems. The unpublished radial velocity data obtained from D. Latham, A. Hatzes, and W. Cochran and treated in Chapter 6, along with the published list of stable velocity stars in Nidever et al. (2002), significantly address this gap and permit reliably estimating the population of stable velocity stars. The high-precision radial velocity data of the CCPS efforts obtained from G. Marcy for statistical analyses and discussed below in §8.2.1 help further constrain the number of spectroscopic companions missed. Given the high precision of these velocity measures and the systematic monitoring over the past 13 years, the resulting estimates include not only the lowest-mass stellar companions but also brown dwarfs.

Since DM91, significant progress has also been made in the observations of visual pairs with various techniques. Almost every star of the sample has been observed with speckle interferometry, probing for the closest visual companions, and AO surveys have unearthed close high-contrast pairs. The *Hipparcos* mission identified nearby companions out to a fewarcsecond separations, and the blinking of archival images as described in Chapter 4 identifies companions out to about 10,000 AU. I have also included published and unpublished results of AO surveys and the results of various searches for wide and close brown dwarf companions (see §8.2.3). Much progress has also been made in searches for nearby field white dwarfs (Holberg et al. 2008; Gatewood & Coban 2009; Lepine et al. 2009; Subasavage et al. 2009), which preferentially select high proper motion targets, and hence improve the chances of finding companions to the sample stars, which also have high transverse motions. These results have augmented early efforts in identifying five white dwarf companions to the stars of this study (HD 13445, 26965, 63077, 100623, and 147513), while DM91 had none, and estimated eight undetected white dwarfs. Given all these observations covering a wide range of methods, the survey results presented here are relatively complete, allowing a more simple extrapolation approach to reliably estimate the number of companions missed. The following sections carries out this analysis for spectroscopic and visual pairs.

8.2.1 Missing Spectroscopic Companions

DM91 performed a simulation based on the criteria for missed spectroscopic binaries outlined in Abt & Levy (1976), which specified that no SB1 with $K_1 < 2 \text{ km s}^{-1}$ and no SB2 with $K_1 < 20 \text{ km s}^{-1}$ could be detected by their surveys. They generated a large sample of fictitious binaries for various trial values of M_2 and P with randomly distributed T, ω , and i, and an eccentricity distribution which assumed circularization for period below 10 days, the Hyades dwarf distribution for intermediate periods, and f(e) = 2e for for periods larger than 1000 days. Their basis for this eccentricity distribution for long-period systems was a theoretical study by Ambartsumian (1937), but our data, presented in the following section, seems to indicate a more random distribution of eccentricities. For each fictitious binary, they estimated the radial velocities at the epochs of actual observations of their stars, and computed a $P(\chi^2)$. The resulting detection probability curves for various periods and mass ratios led them to conclude that the average detection probability for $-1 < \log(P) < 4$ was 0.75, yielding an incompleteness factor of 1.33.

Ongoing efforts with increased sensitivity have effectively plugged the detection gaps

mentioned above. Six of the 48 single-lined solutions listed in Table 6.2 have $K_1 < 2 \text{ km s}^{-1}$, two of which have $K_1 < 1 \text{ km s}^{-1}$. All 50 of the exoplanets reported in Table 6.5 have $K_1 < 0.5 \text{ km s}^{-1}$, 42 of which have $K_2 < 0.1 \text{ km s}^{-1}$, and many less than 10 m s⁻¹. Nearly half of the double-lined solutions listed in Table 6.3 have $K_1 < 20 \text{ km s}^{-1}$, two thirds of which are below 10 km s^{-1} . These results show that we can now reliably detect companions to a much higher sensitivity than was available to DM91. Further, as shown in $\S6.1$, I do not believe that the $P(\chi^2)$ test provides reliable results and likely contributes many false positives. In fact, of the 26 entries in their Table 1 flagged as "SB?", i.e. candidate companions, 15 overlap with this study and 14 of them appear to be constant velocity stars based on the analyses in $\S6.1$. The one exception is HD 165908 for which DM91 might have detected the reflex motion of the 56-year VBO in their 14 measures over 10 years. In addition to an individual inspection of the velocity and power spectrum plots conducted by this work, I have greatly benefited from analysis of the high-precision velocities from Geoff Marcy obtained as part of the CCPS program and made available to this effort for statistical analyses. Figure 8.3 shows the excellent coverage in terms of both the number of velocity measures and their time span for the 255 common stars of the CCPS with this effort. Having access to these unpublished data extended the time horizon on the Nidever et al. (2002) results greatly, enabling a thorough search for hidden stellar companions.

Contrary to my expectations, this search resulted in detecting few new binaries. Thirteen of the 255 stars have either too few observations or too brief a time coverage to say anything meaningful. Of the remaining 242 stars, 168 (69%) appear to be of stable radial velocities



FIGURE 8.3: The distributions of the number of radial velocity measurements per star (left) and their time span (right) in the CCPS survey.

with external errors well below the 0.1 km s^{-1} threshold used by Nidever et al. (2002) and this effort as signaling stellar companions. As expected, this percentage is higher than the frequency of single stars in the entire sample because the CCPS effort avoids known close binaries. Figure 8.4 shows the external error distribution of these stars from the CCPS observations. Of the 25 stars exhibiting rms scatter greater than 0.1 km s⁻¹, three have published planets (Butler et al. 2006) and seven have published stellar companions (Nidever et al. 2002; Vogt et al. 2002), six of which also have corresponding orbits from the CfA data. Fourteen of the remaining 15 show variations consistent with known visual companions, yielding only two new companions from this effort that were not detected by any of the other methods.

This information is very useful in constraining the incompleteness of this work as it relates to spectroscopic companions. With two new companions revealed in 242 stars, more than half the size of the overall sample of 454, we could perhaps be missing a total of four new spectroscopic stellar companions. However, as the published solution for HD 4747 is included



FIGURE 8.4: External error distribution in the CCPS RV measurements.

in this study, the net additional contribution is three new companions. In addition to this, the results presented in Chapter 6 include five new SB1, two new SB2, and one new RVvariable solutions. Two of the new SB1 orbits correspond to known *Hipparcos* photocentric motion orbits and an additional one corresponds to a *Hipparcos* double star. One new SB2 solution is a promotion of an SB1 binary. So, this effort yielded four new binaries that were not known before. These new detections were obtained from the larger than 70% overlap of the sample stars with CfA data, implying possibly two additional undetected companions among the stars not studied by CfA. These two resources of radial velocities are limited by declination constraints and cannot reach close to the south pole, but the extrapolation above should adequately address this gap, implying that a total of only five spectroscopic binaries are missing from these observed results.

8.2.2 Missing Visual Companions

As seen for the spectroscopic companions above, this work is relatively complete in the visual regime as well. DM91 assumed that their work could not detect any visual companions with a < 0? 3 and $\Delta V > 0.4$ mag, half the VBs with 0? 3 < a < 1? 0 and 2.0 $< \Delta V < 3.5$, and essentially all of those with $\Delta V > 3.5$. Significant progress with speckle interferometry and AO have filled these gaps. Comprehensive speckle interferometry programs (McAlister 1978a; McAlister et al. 1993; Mason et al. 2001; Hartkopf et al. 2008; Horch et al. 2008) on telescopes with apertures up to 4 meters can reliably detect sources with $\Delta V < 3$ down to the diffraction limit of the telescopes, and such efforts have studied all but three (HD 2151, 16765, 219538) of the 454 stars of this sample. Coronagraphic and AO efforts on telescopes with apertures up to 10 meters can detect close companions with magnitude differences as large as 10 (Liu et al. 2002; Luhman & Jayawardhana 2002; Potter et al. 2002; Roberts et al. 2005; Turner et al. 2001; Chauvin et al. 2006; Lagrange et al. 2006). An added aspect of completion of these close companions comes from the spectroscopic studies mentioned in the previous section, which reliably detect velocity variations from close low-mass companions.

DM91 pointed out that their effort was likely missing faint, wide companions in the largest numbers. Their incompleteness study estimates that they missed 22 systems with q > 0.1, mainly with long orbital periods. As discussed in §8.1, this work adds two spectroscopic and 14 visual companions to the 106 common stars between DM91 and this effort. Four of these are brown dwarfs discovered by AO and two more are wide brown dwarfs discovered by infrared imaging. So, we have added eight new companions with q > 0.1, which extrapolates to 12 new detections in their full sample of 164 stars. Including brown dwarfs, we have added a total of 25 new companions, confirming DM91's incompleteness analysis and effectively addressing it.

The SFP survey $(\S 3.1)$ revealed no new companions, confirming that the presumed gap between spectroscopic and visual efforts for the stars of this sample is nonexistent, at least for modest Δm pairs. At wider separations, this effort has revealed four previously unknown companions, identified as CPM candidates by blinking the archival images and confirmed by follow-up photometry (see $\S4.1$). While every star of the sample was blinked, 44 of them did not reveal any motion of the primary between the images blinked, thwarting an attempt to identify CPM candidates, and an additional 43 had marginal movement. Conservatively assuming that only half of the marginal frames were adequately examined results in about 85% completion with this technique, yielding one possible companion missed. This technique is quite effective in detecting companions down to R = 17 mag, as evidenced by the refuted candidate to HD 141004 which was measured as R = 16.9 and stands well above the background in the images. Assuming an R limit of 17, we can detect down to late M dwarfs at our distance limit of 25 pc. Confirmed companions include a white dwarf companion to HD 63077 and an M6.5 companion to HD 86728, supporting these detection limit estimates. Hence, this technique is well suited for wide companions down to late M dwarf companions. However, it fails for nearby sources which fall within the primary's saturation, which typically is about 15'', but can be as large as 30''. Fortunately, the other techniques mentioned above including traditional visual techniques and the *Hipparcos* mission addressed that parameter space.

8.2.3 Very Low-Mass Companions

The brown dwarf desert at close separations of under ~ 5 AU has now been firmly established by the high-precision radial velocity surveys, which have revealed many planets but few brown dwarfs, even though they are more sensitive to detecting the more massive companions. Based on the discovery of three brown dwarf companions, Gizis et al. (2001) suggested that the brown dwarf desert applicable to radial velocity regimes may not extend out to wider separations, for which they ruled out the low 0.5% brown dwarf frequency seen in radial velocity studies. Many systematic searches for wider brown dwarf companions have been carried out using high-resolution techniques such as coronagraphy, AO, and space-based observations, yielding a few companions to the stars of this sample (e.g. Potter et al. 2002; Bouy et al. 2003; Burgasser et al. 2005; Luhman et al. 2007), which are included in the results presented in the previous chapter. McCarthy & Zuckerman (2004) conducted a comprehensive infrared coronagraphic search for substellar companions in a sample of 280 GKM stars, looking for companions with mass greater than 30 M_{J} and separations in the range 75–1200 AU. They found only one brown dwarf companion. Grether & Lineweaver (2006) studied the nature of close companions (P < 5 years) to Sun-like stars and concluded that while 16% of the primaries have companions more massive than Jupiter, $11 \pm 3\%$ are stellar, $5 \pm 2\%$ are planets, and less than 1% are brown dwarfs. Other studies have suggested a somewhat higher percentage of wide brown dwarf companions, albeit with large uncertainties. Neuhäuser & Guenther (2004) searched for brown dwarfs wider than 50 AU around 79 stars in three stellar associations using high-resolution imaging techniques, and reported only a single brown dwarf in each group, concluding that $6\% \pm 4\%$ of stars may have wide brown dwarf companions, consistent with the Gizis et al. (2001) conclusion assuming that binaries have the same mass function as field stars. Metchev & Hillenbrand (2008) discovered only two brown dwarf companions in an AO search around 266 F5–K5 stars, and based on an incompleteness analysis, estimated a most-likely brown dwarf companion frequency of 3.2%. Their 1σ confidence interval confines brown dwarf frequencies to 1.5 - 6.3% for separations of 28–1590 AU and mass range of 12–72 M_J. They note that this is a factor of eight smaller than stellar companions, and present a universal companion mass function consistent with their results.

What about the widest of brown dwarf companions at separations of many thousand AU? While the blinking of archival images discussed in the previous section is sensitive to late M dwarfs, it cannot detect brown dwarfs, but this work benefits from other systematic searches for wide brown dwarfs to nearby stars (e.g. Kirkpatrick et al. 2001; Lowrance et al. 2002; Looper et al. 2007). Taking advantage of the steep V - K gradient of substellar sources, these efforts looked for sources around Sun-like stars in 2MASS images with no corresponding source in USNO-A visible images. Candidates thus revealed were followed up to determine if they were low-mass stellar or substellar sources at the distance to the primary. While a number of detections are published, confirming the viability of the technique, these efforts have primarily yielded null results (D. Kirkpatrick 2009, private communication). A

specific search for low-mass companions out to separations of 10,000 AU around solar-type stars within 10 pc resulted in no new detections (D. Looper 2009, private communication). Allen et al. (2007) searched for wide companions in a sample of 132 M7–L8 primaries and detected none, indicating that less than 3% of these low-mass primaries have wide companions. Burgasser et al. (2003) conducted an HST search for brown dwarf binaries, finding a binary fraction of 9^{+15}_{-4} %, most of which were tight binaries with separations ≤ 10 AU. These conclusions were reaffirmed by Kraus et al. (2006). Dieterich et al. (2009) conducted an HST/NICMOS search for companions to 233 stars within 10 pc. Their search, sensitive to early L dwarf companions at separations of 0"5 – 10" and T dwarfs at separations of 1"– 10", recovered two known brown dwarfs and split an additional one into a nearly equalmass binary, but found no other brown dwarf companions. The published and unpublished results discussed above indicate a true paucity of brown dwarf companions rather than an incompleteness of search efforts.

This work includes 12 brown dwarf companions in 9 systems (HD 3651, 72760, 79096, 97334, 130948, 137107, 145958, 190406, and 206860). Three of these systems have projected separations of less than 100 AU, and three more have projected separations less than 1000 AU. The implied fractions of stars with brown dwarf companions are consistent with prior studies (McCarthy & Zuckerman 2004; Metchev & Hillenbrand 2008). In contrast, none of the radial-velocity-detected companions to stars in this sample has a minimum mass of over 13 M_J, and only two have a minimum mass of greater than 8 M_J (see Table 6.5), consistent with the expectations of the brown dwarf desert. The widest separation of a confirmed

brown dwarf is 3800 AU (HD 137107E), while the brown dwarf with the extreme separation of 40,000 AU from HD 145958 is retained as a candidate (see § 7.4). Interestingly, three of the nine systems with brown dwarf companions have a pair of closely orbiting brown dwarfs separated relatively widely from the primary. This shows that brown dwarf binaries are overabundant as companions to main sequence stars than for field brown dwarfs, as pointed out by Burgasser et al. (2005) who speculate on the possibility that a pair of brown dwarfs are better able to withstand the dynamical ejection that may be responsible for the formation of these wide pairs.

The above analysis suggests that this work is missing few, if any, brown dwarf companions, and that the true percentage of very low mass companions to solar-type stars is close to the lower limit of 1.9% estimated by DM91 and certainly much less than their upper limit of 14% or the 20% they estimate from the null results of Campbell et al. (1988).

8.2.4 Results from DM91 Incompleteness Analysis

DM91 made two levels of corrections to the fraction of single stars. The adjustment from their observed 57% single stars to 43% single stars was largely based on wide faint companions missed. This works lends credibility to this estimate, identifying 14 visual companions to the subsample of overlapping stars against their estimate of a total of 22 missed, but the current observed fraction of single stars, including these companions, stands at 57% – exactly what DM91 observed! The second adjustment they made was based on the mass-ratio distribution analysis combined with the $P(\chi^2)$ simulations, and this led them to conclude that only onethird of solar-type stars may be truly single, i.e. without stellar or brown dwarf companions.
The current effort shows that this adjustment, although consistent with their analysis and prevailing expectations, was a significant overestimate. There have been other efforts which claimed that DM91 significantly under-estimated the binary frequency of solar-type stars (Söderhjelm 2000; Quist & Lindegren 2000; Kouwenhoven 2006), but the current results were unable to confirm these predictions. Apparently, a majority of solar-type stars are, in fact, single like our Sun.

8.2.5 Missed Companions and True Multiplicity

The above discussion strongly suggests that the results of this survey are relatively complete, and likely represent the most complete survey yet of the sample of nearby solar-type stars. The estimates of five missed spectroscopic and one missed visual companion only suggest a total of six companions missed, representing 1.3% of the stars studied. If we assume that these missed detections are distributed according to the ratio of single to non-single stars presented in §7.2, the percent of single stars would decrease from 56.7% to 56.0%, a shift well within the 1 σ uncertainties, but one that is used to present the final numbers below. In §8.1, I discussed a possible bias in my results because of an all-sky sample, which may include historically less studied southern-hemisphere systems. The difference in the percentage of single stars between the overall study of 454 stars and the 307 stars with declination above -15° was 2.4%, and the incompleteness analysis performed here, which, given my extrapolation approach, mostly addresses the southern hemisphere stars, accounts for over half this difference, raising the confidence in these numbers. The remaining difference is well within any expected statistical margin of error.

305

Finally, let us evaluate the candidate companions identified by this effort. Of the 25 candidates, 7 are RV variables, 4 from CfA and 3 from CNS. The four candidates from CfA all show a possible linear trend over a few observations, but the variations are small and well within measurement errors. While more observations are warranted, each of these has a $P(\chi^2)$ larger than 0.01, suggesting that they may have stable velocities. The three candidates from CNS are also questionable. While no independent observations are available to definitively refute them, no confirmation exists either. As discussed in $\S5.5$, over 50% of the spectroscopic binaries identified in the CNS have later been refuted. So, perhaps one or two of these seven candidates may prove to be real. The remaining candidates include seven close visual pairs, six proper motion accelerations, one *Hipparcos* photocentric orbit, one over-luminous companion, and one (HD 145958) wide brown dwarf companion at 40,000 AU separation with similar proper motion, but perhaps too wide to be bound (see $\S7.4$). It is hard to say which of these will be confirmed and which refuted, but assuming that half of them are real, the percentage of single:double:triple:quadruple, taking the mid-point of the confirmed and possible fractions in Table 7.2 and including the incompleteness analysis is $55 \pm 3: 34 \pm 2: 9 \pm 2: 2 \pm 1.$

Additionally, the CCPS data revealed 33 radial-velocity variables with rms scatter less than the threshold of 0.1 km s^{-1} used in §8.2.1. Seventeen of these can be explained by known stellar companions within a few arcseconds, leaving 16 possible new discoveries in these data. These could be long-period planets or yet undiscovered low-mass stellar or brown dwarf companions at a separation of a few arcseconds. Given the extensive monitoring of

these stars with various visual techniques, an estimate that at most half of these turn out to be stellar or brown dwarf companions appears reasonable. Assuming that the implied eight companions missed are distributed across single stars and those with companions in their respective ratios, an additional 1% of the presumed single stars may have companions. Hence, $54\% \pm 3\%$ of Sun-like stars are expected to be single, i.e. with no stellar or brown dwarf-companions.

8.3 Multiplicity Dependency on Physical Parameters

Let us now see if and how the stellar multiplicity is effected by physical parameters such as temperature, age, and elemental abundance. As dynamical masses have only been obtained for a small fraction of the stars, a direct study of the relation of multiplicity to mass is difficult. However, because all stars of the sample fit in a band around the main sequence, the temperature analysis can be interpreted as a dependence of multiplicity on stellar mass. Unless otherwise specified, the following sections limit analyses to confirmed companions. With the large sample studied here, and the proportionally few candidate systems, the patterns seen and conclusions drawn are not expected to materially change by focusing on the confirmed systems.

Before looking at the specific results, let us review the sources of the various physical parameters, which are listed in *Appendix* A along with their references. For the primary stars, spectral types were obtained from Gray et al. (2003, 2006) or *Hipparcos*. Primary star masses, for the few with dynamical estimates, were extracted from the relevant sources.

For the remaining stars, the table in Appendix A lists estimates extracted from Valenti & Fischer (2005) or Nordström et al. (2004), in that order of preference. However, the analyses below use dynamical mass estimates, when available, and interpolated masses from spectral types using the relations in Cox (2000) when not, in order to follow a consistent approach. Metallicity, as measured by [Fe/H], was extracted from Valenti & Fischer (2005), Nordström et al. (2004), Gray et al. (2003), and Gray et al. (2006), and the chromospheric activity index, $\log(R'_{HK})$, from Wright et al. (2004), Gray et al. (2003), Gray et al. (2006), or from B. Mason (2008, private communication). For the companions, mass estimates were extracted from the various discovery or characterization publications, when available. Otherwise, spectral types were extracted when available in these publications. For double-lined spectroscopic systems, the measured mass ratio led to an estimate of the companion's mass and spectral type. As a next step, multi-color photometry of the companion was used to estimate its spectral type, otherwise Δm information and the primary's spectral type led to an estimate of the companion's spectral type, and thereby to its mass, using the relations from Cox (2000).

8.3.1 Multiplicity by Spectral Type and Color

It is now well established that a greater percentage of bluer, more massive stars have companions when compared to their redder, less massive counterparts (Burgasser et al. 2003; Mason et al. 2009a, DM91). Let us now test this dependence on the sample studied here and review these results in the overall context across the entire spectral sequence. The reasonably wide color range of $0.5 \leq B - V \leq 1.0$ for the sample of stars studied here enables a check for the dependence of multiplicity on spectral type or color, and hence indirectly, on mass. Indeed, the bluer stars of this study seem to favor binaries to a 2σ significance when compared to the redder stars, consistent with the overall trend seen for O–M stars. The data seem to suggest a relatively steep drop in the multiplicity fraction at B - V of about 0.63, corresponding to a ZAMS star with the Sun's spectral type. Table 8.2 summarizes the multiplicity fractions by color and spectral type, and Figure 8.5 shows the distribution for four equal color bins.

TABLE 8.2: Multiplicity Statistics by Spectral Type and Color

Sample	Ν	Single	Binary	Triple	Quadruple+
$F6 \le SpT < G5$	179	52 ± 4	37 ± 4	8 ± 2	3 ± 2
$G5 \le SpT \le K3$	275	60 ± 3	31 ± 3	7 ± 2	2 ± 1
$0.500 \le B - V \le 0.625$	131	50 ± 4	38 ± 4	8 ± 2	3 ± 2
$0.625 < B - V \le 1.000$	323	59 ± 3	31 ± 3	7 ± 2	3 ± 1
$F7 \le SpT \le G9$	281	56 ± 3	34 ± 3	7 ± 2	3 ± 2

The dependence of multiplicity on color raises the question about directly comparing the results of this effort with those of DM91. The color range of $0.5 \le B - V \le 1.0$ used for this study selects stars from F6–K3 types. In contrast, the DM91 study was limited to F7–G9. As 167 (37%) of the stars in the current sample have a K spectral type, could they influence the higher percentage of single stars derived here? As the last line in Table 8.2 shows, a selection of F7–G9 stars from this study includes 281 stars, and yields observed multiplicity frequencies consistent with my entire sample, showing that the color-range difference between the two samples does not bias the results presented here.



FIGURE 8.5: Multiplicity Statistics by B - V Color.

Let's now look at these results in the context of multiplicity fractions across the entire range of O stars to T brown dwarfs. Most O-type stars seem to form in binary or multiple systems, with an estimated lower limit of 75% in clusters and associations having companions (Mason et al. 1998a, 2009a). Studies of OB-associations show that these high percentages (over 70%) of companionship also apply to B and A type stars (Shatsky & Tokovinin 2002; Kobulnicky & Fryer 2007; Kouwenhoven et al. 2007). M-dwarfs have companions in significantly fewer numbers, with estimates of 30–40% with companions (Henry & McCarthy 1990; Fischer & Marcy 1992; Reid & Gizis 1997). Finally, estimates for late M-dwarfs and brown dwarfs show that only 10–30% of them have companions (Burgasser et al. 2003; Siegler et al. 2005; Allen et al. 2007; Maxted et al. 2008; Joergens 2008). Including the results of this effort, Figure 8.6 shows the dependence of multiplicity on spectral type for the entire range of stars and brown dwarfs. While the revised results are consistent with the overall trend that multiplicity reduces with reducing mass, they indicate a sharper drop-off at solar-type stars than previously believed.



FIGURE 8.6: Multiplicity Statistics by Spectral Type. The thin solid lines represent stars and brown dwarfs beyond the spectral range of this study, and their sources are listed in the text. The OBA stars have only lower limits for the fraction of stars with companions. For the FGK stars studied here, the thick solid lines show the observed results of this work for spectral types F6–G5 and G5–K3, and the thick dotted lines show the corresponding incompleteness adjusted numbers for the F6–K3 range. The uncertainties are estimated by bootstrap analysis as explained in § 7.3. The thin dashed lines show the observed statistics from DM91 with uncertainties estimated here, and the dotted lines show the revised DM91 statistics for incompleteness including only companions more massive than 0.1 M_{\odot} (lower line) and all companions more massive than 10 M_J (higher line). The observed and corrected results of this study include all stellar and brown dwarf companions.

8.3.2 Multiplicity by Chromospheric Activity

Multiplicity studies of young stars (Ghez et al. 1997; Kouwenhoven et al. 2007), of nearby solar-type stars (Mason et al. 1998b), and of aging stars in globular clusters (Sollima et al. 2007) suggest that binaries and multiple systems tend to get disrupted with age, presumably due to dynamical interactions. The selection of this current sample of solar-type stars is limited to a band around the main sequence to focus on spectral classes IV, V, and VI. This sample contains stars over a wide age range, from stars such as HD 146361, which is believed to be a few 100 Myr old, to stars like the Sun, which are about 5 Gyr old, enabling such a check, but age estimates are tricky. The prolonged evolutionary process of solar-type stars as well as the diversity of abundances in the sample result in a considerable width for the main sequence, preventing the departure from the main sequence as a reliable age indicator. Fortunately, chromospheric emission, as measured by $\log(R'_{\rm HK})$, the ratio of the flux in the cores of a star's Ca[II] H and K lines relative to the star's bolometric flux, is a good age indicator for solar-type stars (Henry et al. 1996), and has been used in prior multiplicity studies (Mason et al. 1998b). This index was extracted for the current sample from Wright et al. (2004); Gray et al. (2003, 2006, and B. Mason 2008, private communication), leaving out only 12 stars without a measurement.

Figure 8.7 shows that the current sample seems to be divided into two subsets, 156 active (younger) stars with $\log(R'_{\rm HK}) \geq -4.75$, and 286 inactive (older) stars with $\log(R'_{\rm HK}) < -4.75$. The figure also helps confirm that the metric used here does not correlate with color, i.e. mass, and seems to be an effective age indicator (Mason et al. 2009b). As seen in



FIGURE 8.7: The distribution of chromospheric activity by B - V color. The dashed horizontal line separates the active subset of stars above the line from the inactive subset below it.

Figure 8.8, the more active stars do have a higher binarity fraction to a 2σ significance.

8.3.3 Multiplicity by Metallicity

Fischer & Valenti (2005) showed that stars with planets preferentially have higher metallicity than those without planets, and concluded that this is because planets are more likely to condense in the circumstellar disks around stars that form out of metal-rich clouds (the "nature" alternative), rather than a result of stars accreting metal-rich planets during the formation process (the "nurture" alternative). The current results allow us to verify the



FIGURE 8.8: Multiplicity Statistics by Chromospheric Activity. The percentage of stars with companions among the 156 active (young) and 286 inactive (old) subsets of the sample for the 442 stars with chromospheric activity data. The percentages are for confirmed companions and the uncertainties are estimated from bootstrap analysis as explained in § 7.3.

planet-metallicity correlation and see if they apply to more massive companions such as brown dwarfs and stars.

Figure 8.9 shows the metallicity distribution for stars with and without stellar and substellar companions. Metallicity values for 416 stars of this sample were extracted from Valenti & Fischer (2005), Nordström et al. (2004), and Gray et al. (2003, 2006) in that order of preference. Metallicity is plotted against the B - V colors, showing that there is no correlation between these parameters, and allowing us to focus on multiplicity as it relates to



FIGURE 8.9: Multiplicity Statistics by Metallicity. The plot on the top shows stars with (filled triangles) and without (small filled circles) stellar companions plotted with respect to their metallicity and B - V color. The bottom plot shows stars with planetary (filled upward triangles), brown dwarf (large, filled downward triangles) and no substellar (small filled circles) companions.

metallicity. The top panel shows stars with (filled triangles) and without (small filled circles) stellar companions, clearly demonstrating that there is no relationship between metallicity and the tendency of a star to have stellar companions. This is not surprising because while planets may require the presence of higher elements to trigger the condensation process by which they form, the primordial matter that can form one star should be equally likely to form others as well (i.e. this merely illustrates that stars form like stars). But perhaps this clear difference between stars and planets allows one to check if brown dwarfs are more like stars or planets.

The bottom panel clearly demonstrates that planets preferentially form around highermetallicity stars, as shown by Fischer & Valenti (2005) and confirmed by Figure 8.10. Data on brown dwarf companions is sparse, but the nine systems with such companions are plotted on the bottom panel as the large, downward filled triangles. The lowest metallicity star with a brown dwarf companion is only slightly below that with a planetary companion, and no stars with [Fe/H] below -0.3 dex have either planetary or brown dwarf companions, while many of them have stellar companions. While more data are required to make a definitive conclusion, these preliminary results suggest that brown dwarfs, at least when they are companions to stars, form like planets rather than like stars.



FIGURE 8.10: Planet metallicity correlation for the well-populated [Fe/H] bins of -0.3 to +0.3 dex, containing a total of 341 stars. The histogram shows the percentage of stars with planets, and "N" values are the total number of stars in each bin.

8.4 The Distribution of Orbital Elements

8.4.1 Period Distribution

For some binaries, such as those with spectroscopic, visual, or combined orbital solutions, we have a reliable period estimate. For others, such as radial-velocity variations, proper motion accelerations, or wide CPM companions, the period was estimated as described below. For CPM companions with good measurements of separations, I used the statistical relation $\log(a'') = \log(\rho'') + 0.13$ from DM91 to estimate the semimajor axis. Then, using the FvL07 parallax, I converted this to a linear semimajor axis in AU and used Newton's generalization of Kepler's Third Law to estimate the period. Mass estimates for the components were obtained as described in \S 8.3, taking into account the hierarchical nature of multiple systems. For only two confirmed pairs (HD 25680 AB and HD 147776 AD), companion mass estimates were not available as described above, and they were estimated assuming a mass-ratio of 0.2. For 14 confirmed pairs with unresolved companions, mostly radial velocity variables or proper motion accelerations, no period or separation information was available. For the radial velocity variables, I assumed periods in the range of 30-200 years, reasonable because a shorter period would likely have an orbital solution due to the extensive radial velocity coverage for stars of this sample and periods longer than 200 years are unlikely to be detected with the few decades of velocity measurements to a precision of $\sim 0.5 \mathrm{km \, s^{-1}}$. The accelerating proper motion pairs indicate an observed curvature in proper motions with observations over a few decades, so I assumed periods between 10 and 25 years for these unresolved pairs. The specific value of the assumed period within the above ranges does not impact the following analysis because the entire range fits within one bin in each case.

Figure 8.11 shows the period distribution of all 258 confirmed companions to the sample of solar-type stars, with an identification of the technique used to discover and/or characterize the pair. To provide better context, the axis at the top shows the semimajor axis corresponding to the period below assuming a mass sum of 1.47 M_{\odot}, which is the average value of all the confirmed pairs. The period distribution seems to follow a roughly Gaussian profile with a peak at log(P) = 5.03, corresponding to a period of 293 years, slightly larger than that of Pluto around the Sun. The median of the period distribution is 261 years, similar to the Gaussian peak. This compares with a corrected peak and median of 180 years from DM91. The larger value of the current survey is a result of more robust companion information for wide CPM companions. The overall profile of the curve is quite similar to the corrected plot in DM91, suggesting that the companions they estimated as missed have now been found.



FIGURE 8.11: Period distribution for the 258 confirmed companions plotted by the companion detection method. Unresolved companions such as proper motion accelerations are identified by horizontal line shading, spectroscopic binaries by positively sloped lines, visual binaries by negatively sloped lines, companions found by both spectroscopic and visual techniques by crosshatching, and CPM pairs by vertical lines. The AU separations shown at the top correspond to the periods below for a system with a mass-sum of 1.47 M_{\odot} , the average value for all the pairs. The dashed curve shows a Gaussian fit to the distribution. The overlap of multiple techniques for all but the longest-period bins suggests that the coverage is complete. The fall-off of the long-period systems is consistent with canonical limits for gravitational binding.

The overlap of the various detection methods for each period bin, except for the longest bins which, as expected are dominated by CPM companions alone, indicates that there is no gap in the parameter space between the complementary approaches used for detecting companions. The robust overlap between spectroscopic and visual techniques out to periods longer than 20 years shows why the CHARA SFP survey produced null results. While the overall distribution of the various techniques follows the expected pattern, a few notes are warranted. The single unresolved companion of period under 1 day is the 11.5-hour eclipsing binary secondary in the quadruple HD 9770 system, and the only other orbit with less than a one day period is the 6.4-hour SB2 secondary to HD 133640. The shortest-period visual orbit that is not known spectroscopically is the 231-day photocentric motion orbit of HD 113449 mapped by *Hipparcos* which has also been resolved with aperture masking on large-aperture telescopes (see § 7.4). All the nine visual orbits with P < 100 days also have spectroscopic solutions, and are probably follow-on visual efforts after the binaries were detected by radial velocity. Three of these (HD 14241, 147584, and 160346) are photocentric-motion orbits with corresponding spectroscopic solutions, and the remaining are all LBI resolutions of double-lined spectroscopic binaries. The three such systems with the shortest periods were all resolved by this work with the CHARA array.

The two spectroscopic binaries (HD 16765 and 186408) with $\log(P) > 5$ are both radialvelocity variations seen in the CfA data consistent with a 3 - 4'' visual pair. The longestperiod spectroscopic orbital solutions are for HD 24409 and HD 43587, both of which have preliminary solutions from partial orbital coverage. The other five counts of spectroscopic solutions in this bin are all radial velocity variations with the assumed period as described in the first paragraph of this section. The joint spectroscopic and visual solutions in this bin have radial-velocity-based solutions that rely on orbital parameters, such as period, from the visual orbits to present preliminary solutions. The nine unresolved pairs listed in the $3 < \log(P) < 4$ bin all have assumed periods as explained in the first paragraph of this section. Overall, the plot supports earlier conclusions that the results of this work are relatively comprehensive and complete.

8.4.2 Period-Eccentricity Relationship

Figure 8.12 shows the period-eccentricity relationship for the 129 pairs with estimates of these parameters from visual and/or spectroscopic orbital solutions. Pairs with periods below 12 days seem to be well circularized with eccentricities close to zero, with one notable exception. The 7-day SB2 pair in HD 45088 seems to have an unusually high eccentricity of 0.1471 ± 0.0034 for its short period, and the longer 600-year orbit has an eccentricity of 0.25. A likely explanation is that this system is relatively young and hence not yet circularized. The log(R'_{HK}) of the primary of -4.266 (Gray et al. 2003) is among the highest for the stars plotted in Figure 8.7, indicating relative youth, which is confirmed by a high rotationalvelocity and emission features in its spectra (Mishenina et al. 2008).

The figure does show a general trend that was pointed out by DM91, namely, components of multiple systems have generally higher eccentricity. While the overall results of this survey show a 60–40 split between pairs in binary versus multiple systems, 9 of the 14 highesteccentricity pairs are components of higher order multiples, yielding a corresponding ratio



FIGURE 8.12: Planet-eccentricity relationship for the 129 pairs with estimates of those parameters from visual and/or spectroscopic orbital solutions. Components of binaries are plotted as filled circles, of triples as open triangles, and of quadruple systems as open squares. The horizontal dashed line marks a zero-eccentricity limit and the vertical dashed line marks the 12 day period, which roughly corresponds to the circularization period for this population of stars. The exceptions with notable eccentricities to the left of this line are discussed in the text.

of 35–65. The four binary systems with high eccentricity (HD 57095, 82885, 120136, and 161198) may contain the telltale signs of an interesting dynamical past or harbor yet unseen companions. With the exception of HD 161198, the others are all preliminary visual orbits with periods of 93–2000 years, indicating either that the eccentricity estimate is preliminary and approximate, or that they have perhaps endured dynamical ejection of other components. The shortest period among these is the 7-year SB1VB, HD 161198.

Figure 8.13 shows the eccentricity distribution for the 117 systems with periods greater than the circularization limit of 12 days. The dotted line shows the distribution for the 35 systems with periods below 1000 days, and the dashed line for the 82 systems with periods longer than 1000 days. The two distributions look fairly similar, in contrast to the results in DM91, which claimed a "bell shaped" distribution for the shorter period and a f(e) = 2epattern when corrected for missing systems. They also point out that the f(e) = 2e relation is expected from theoretical considerations from Ambartsumian (1937). Looking at Figure 6b of DM91, I do not see the conformity to the above relation in their plot, even after adding in their estimated missed systems of high eccentricity based on the simulations of Harrington & Miranian (1977). While the current effort shows that many systems missed by DM91 have now been found, eccentricity measurements are not available for the wide CPM companions, and hence, a similar correction may be warranted for the data analyzed here. Figure 8.13 shows a roughly uniform distribution for eccentricities below 0.6, followed by a drop-off for larger eccentricities. Compensating for missed systems at high eccentricities as done by DM91, the distribution appears flat for all eccentricities. It certainly does not follow the f(e) = 2e distribution, confirming the conclusion reached by Shatsky (2001) based on 174 objects in the Multiple Star Catalog (Tokovinin 1997).

8.4.3 Mass-Ratio Distribution

Figure 8.14 shows the mass-ratio distribution for the 204 confirmed pairs with primary and secondary mass estimates, obtained as described in §8.3. For hierarchical systems, appropriate mass sums were considered for the successive "pairs". For example, a triple



FIGURE 8.13: Eccentricity distribution for the 117 systems with periods longer than the 12day circularization limit with estimated eccentricities from visual or spectroscopic solutions. The dotted line represents the 35 systems with periods below 1000 days, and the dashed line for the 82 systems with periods longer than 1000 days.

with Aa, Ab, and B has two pairs. For the AB pair, the masses used were that of Aa + Ab for the primary, and B for the secondary. For the Aa, Ab pair, the masses used were for the individual components. Also, for the purposes of the figures in this section, M₁ was always taken to be the larger mass. So, in a A,BC triple, if the mass of B and C added up to more than that of A, the larger mass of BC was considered as M₁ and that of A was considered as M₂ for these plots.

The fall-off of the number of systems at mass-ratios below 0.2 is consistent with the



FIGURE 8.14: Mass-ratio distribution for the 204 confirmed pairs with primary and secondary mass estimates.

observed results of DM91 and probably close to the true distribution, as discussed in §8.2.3. There is a definite and pronounced peak at a mass-ratio of 1, which was not seen in DM91, but has been observed for *Hipparcos* doubles (Söderhjelm 2007). While Abt & Levy (1976) suggested different distributions for orbital periods shorter and longer than 100 years, I do not see that trend in the data. DM91 concluded that no peak was observed at a unit massratio, and suggested as a result that binaries can form by random associations of stars from the IMF. The current results lead me to a different conclusion.

The peak for equal-mass binaries appears real upon closer inspection. Of the 27 systems

at that peak, only two are triples where the mass of the primary is nearly equal to the sum of the other two components. Three more are the wide brown-dwarf pairs mentioned in §8.2.3. The remaining 22 are two-star pairs, some in hierarchical multiple systems, that are made of nearly equal-mass components. Let us also consider how the companion masses for these were obtained. Five of these have dynamical SB2VB mass estimates for each component, two more have SB2 mass ratios of nearly one, five are wide pairs with independent measurements of similar spectral types, and ten more have multiple resolutions in the WDS with a nearly zero magnitude difference. So, binaries do seem to disproportionately favor equal-mass components.



FIGURE 8.15: Mass-ratio distribution as a function of orbital period.

Figure 8.15 shows the mass-ratios plotted by $\log(P)$ to see if there is a period dependence, illustrating two points. First, there is an overall trend that systems with high mass ratios prefer shorter periods. Only 4% of the systems with mass ratios below 0.45 have periods shorter than 100 days, but this percentage doubles to 8% for mass ratios between 0.45 and 0.9, and doubles again to 16% for mass ratios above 0.9. These results have a greater than 1 σ significance using Poisson statistics, indicating that nearly equal-mass pairs prefer shorter periods, perhaps suggesting a fission formation process for some systems. Second, only four of the 27 systems with mass ratios above 0.95 have periods less than 100 days and twins are seen to have periods as large as $\log(P) = 5.5$ (900 years). This indicates that while twins may form preferentially due to fission when compared to pairs of disparate stars, the vast majority of them (85%) still have long periods. This suggests that there are multiple processes responsible for the formation of binaries – fission, that accounts for as many as 16% for similar-mass pairs, and fragmentation or random associations, which account for the majority of binaries which prefer longer periods.

Figure 8.16 shows the distribution of the secondary mass on the left, and as a function of the primary mass on the right. The patterns seem to match expectations, with a drop-off at the very low mass end. The drop-off at masses above 0.8 M_{\odot} are more a function of the sample selection, with primary masses in the $0.7-1.4 \text{ M}_{\odot}$ range, and the definition above that M₂ is always the smaller mass. The right panel shows that systems with the lowest total mass tend to be equal-mass pairs, which likely is because primaries of small mass have only a small mass-range left over for secondaries.



FIGURE 8.16: Secondary mass distribution (left) and as a function of the primary mass (right).

These results, especially the distribution in Figure 8.14, suggest that binaries may form by different mechanisms, at least some of which favor the formation of "twins", and that the brown dwarf desert extends to wide separations and perhaps even to the very low-mass stars, when considered as companions to Sun-like stars.

8.5 Multiplicity in Exoplanetary Systems

Multiplicity among exoplanet systems can tell us a great deal about the formation and stability of solar systems, but unfortunately is thwarted by selection effects. Most planet-search efforts avoid known close binaries, and recent efforts to discover planets around binaries (e.g., Konacki 2005) have not yet yielded any positive results. However, searches for stellar companions to exoplanet systems have yielded surprising results. Contrary to earlier expectations, Raghavan et al. (2006) showed that almost 23% of the 131 exoplanet systems studied contained stellar companions at wide-enough separations to leave the solar system intact around one star. A recent report by Mugrauer & Neuhäuser (2009) states that 43 of the 250 exoplanet hosts (17%) are members of binary or multiple systems, implying that while the number of stellar companions to exoplanet hosts has been growing, the number of exoplanet systems detected is growing proportionally faster. The 34 exoplanet systems of this study are composed of 23 (67.6%) single stars, 10 (29.4%) binaries, and 1 (2.9%) triple systems. The higher percentage of binaries or multiples among exoplanet-hosts obtained here is likely because the current volume-limited sample is probably better-studied for stellar companions as compared to the exoplanet sample, which covers a larger volume of space, and hence, less-studied stars.

Do these results suggest that stellar companions are more common in stars without planets? They do not, because the likely explanation for the relative paucity of stellar companions to stars with planets is likely due to the selection effect outlined above. The noncorrelation between metallicity and stellar companionship discussed in §8.3.3 also supports the possibility that planetary systems are as likely to form around single stars as they are around components of binary or multiple star systems. We can test the dependence of planetary and stellar companions with the current results. Twenty one of the 257 single stars, 10 of the 150 binaries, and 1 of the 34 triple systems of this study have planets. Using Poisson statistics, the frequency of planet-hosts is $8.2\% \pm 1.8\%$ for singles, $6.7\% \pm 2.1\%$ for binaries, and $2.9\% \pm 2.9\%$ for triples. These frequencies are statistically indistinguishable, suggesting that the presence of stellar companions does not affect planet formation or stability, as has been seen in other studies (Bonavita & Desidera 2007). In fact, Boss (2006) presents the results of a modeling effort, showing that binary stars are quite capable of forming solar systems like ours, and that the presence of a nearby (~ 50 AU) star might in fact help trigger planet formation.

However, sufficiently shot-period binaries will disrupt the protoplanetary disk, hampering planet formation about either star(e.g., see Desidera & Barbieri 2007). The evolution of our own solar system would have undoubtedly taken a different course, although one that might still foster life, if the Sun had a stellar companion within 50 AU. We can see from Figure 8.11 that 132 orbits have separations under 100 AU, assuming a mass-sum of 1.47 M_{\odot}. These pairs represent 125 systems, some of which are tight pairs in a wide triple, still leaving room for a solar system to form around the third star. I adopt a conservative estimate of 100 (22%) stars in this study that may not provide suitable environments for the formation of a solar system like ours. More importantly, as seen in Figure 8.11, stellar companions seem to prefer a relatively wide separation between them, leaving the regions around their respective habitable zones intact for planets, and perhaps life. In this context, binary star systems may provide more real-estate conducive to the formation of life, not less, as has been previously thought.

8.6 Hierarchy of Multiple Systems

This section presents the hierarchical or "mobile diagram" for each system with three or more stellar and brown dwarf components. Figures 8.17–8.19 show the hierarchy of triple systems, and Figures 8.20–8.21 shows the hierarchy of higher order systems.

8.7 Conclusions

In this effort, I have attempted a comprehensive evaluation of the multiplicity of solartype stars. The sample studied consists of 454 stars, including the Sun, which serve as representatives of the tens of billions of such stars in the Galaxy. All the sample stars are within 25 pc of the Sun, selected from the *Hipparcos* catalog based on the following criteria: $\pi_{\text{trig}} > 40$ mas with an uncertainty of less than 5%, $0.5 \leq B - V \leq 1.0$, and positioned on an HR diagram within a band extending 1.5 magnitudes below and 2 magnitudes above an iterative best-fit main sequence (Chapter 2). The resulting sample is an exhaustive set of stars with a V-band flux of one-tenth to ten times that of the Sun, providing a physical basis for the term "solar-type".

This work is an update to the seminal effort of DM91, utilizing a larger and more accurate sample and targeting new observations to augment the vast amount of data available from extensive multiplicity studies of these stars by many different techniques. In fact, a key motivator of this effort was the chance to present results from the first nearly-complete survey of these stars. I believe that the robust monitoring of these stars over the past several decades with virtually every available method for finding and characterizing companions, and the comprehensive synthesis effort of this study, have enabled such a result. The primary observational efforts of this work include a survey for SFP companions using the CHARA Array (\S 3.1), a search for wide companions using archival images (Chapter 4), and improving the completeness of speckle interferometry coverage. The null results of the SFP survey (\S 3.1.1) show that the expected gap between short-period spectroscopic companions and

longer-period visual companions is effectively closed for nearby solar-type stars. The long baselines of the CHARA Array were also used to monitor four known short-period binaries of this sample and develop visual orbits (§ 3.2), leading to component mass estimates for HD 8997, 45088, 146361, and 223778. The search for wide companions by blinking multiepoch archival images yielded four new CPM companion discoveries (around HD 4391, 43162, 157347, and 218868), which were confirmed as physical associations based on follow-up photometry yielding distance estimates consistent with the primary's *Hipparcos* parallax. The blinking method also helped identify many WDS pairs as optical because they clearly were field stars that did not share the primary's proper motion (§ 4.2). The photometric follow-up also confirmed several previously suspected CPM companions as physical associations have ensured that all but three of the sample stars have been observed with this technique at least once.

The observational efforts of this work have been tremendously augmented by synthesizing previous results, enabling more robust multiplicity statistics. Visual companions unearthed by the *Hipparcos* mission were individually investigated using other data to determine if they were physical (\S 5.1). The photocentric-motion and resolved-pair visual orbits in the VB6 catalog were analyzed and included (\S 5.2). While all visual orbits of resolved pairs, excluding one exception (HD 32923), are physical, several of the photocentric motion pairs were deemed spurious based on the availability of high-precision radial velocities. The WDS (\S 5.3), FIC (\S 5.4), and MSC catalog entries for all overlapping stars were investigated and included. The CNS catalog entries for common stars were also individually checked and found to be very reliable for CPM pairs, but not so for entries flagged as spectroscopic pairs (§ 5.5). Faint companions found by high-contrast coronagraphic and AO techniques have also been included from the WDS or by searching publications. Unpublished null results of the many searches looking for low-mass stellar and brown dwarf companions have been included in the incompleteness analysis (§ 8.2.3). Known spectroscopic companions listed in the SB9 catalog or in publications have also been included (Chapter 6). Publications of highprecision velocity measurements (Nidever et al. 2002) were very useful not only in improving the statistics of physical companions, but also in helping test the veracity of claims in other sources like the CNS and *Hipparcos* catalogs.

While the above efforts provide a fairly complete coverage, this work would not have been nearly as comprehensive without the results from modern radial-velocity surveys. I obtained and included unpublished results from D. Latham for over 300 of the sample stars from the systematic monitoring of radial velocities at the CfA over more than 20 years. These data include five new SB1 orbits (\S 6.2.1), two new SB2 orbits (\S 6.2.2), and one new binary identified by radial-velocity variations. Two of the SB1 orbits (HD 185414 and HD 224465) are new binary detections, while the remaining three (HD 24409, 32850, and 128642) are the first spectroscopic orbits for known *Hipparcos* doubles. One SB2 solution represents a new companion detection (HD 111312) while the other was previously known as a single-lined system (HD 148704). These results also include improved orbital solutions for 16 SB1 orbits and 7 SB2 orbits. Radial velocity variations for HD 16673 indicate a previously unknown companion, and four more candidates have been identified for future observations (§ 6.2.3). A. Hatzes and W. Cochran also helped me check for stellar companions in their planet-search data (§ 6.1). Finally, I obtained data from the systematic monitoring of high-precision radial velocities for 255 of the sample stars from G. Marcy for statistical analyses. These data included only two stellar companions not detected by any other means, and this unexpected results greatly enhances the robustness of my incompleteness analysis (§ 8.2.1).

This effort finds that the majority of solar-type stars are in fact even more Sun-like, for they are single. This revises prior expectations from DM91, who predict that only 43%of solar-type stars lack companions with masses greater than 0.1 M_{\odot} and only 33% are without companions more massive than $10 M_{J}$. My results include all stellar and brown dwarf companions, yielding the observed percentage of single, double, triple, and quadruple or higher order systems as $57\% \pm 3\%$, $33\% \pm 2\%$, $8\% \pm 1\%$, and $3\% \pm 1\%$, respectively (§7.3). As predicted by DM91, these results double the percentage of triple and higher order systems compared to their fractions. Remarkably, however, the observed percentage of single stars of this effort is identical to that of DM91, and given all the observational effort since their work, suggest that most of them may indeed be single. If all of the candidates revealed by this work are found to be real, the corresponding percentages would change to $54\% \pm 2\%$, $34\% \pm 2\%$, $9\% \pm 2\%$, and $3\% \pm 1\%$. The incompleteness analysis (§8.2) shows that only a few companions are likely missed by this effort, resulting in only about a 1% reduction in the percentage of single stars. The current sample has a more extended coverage when compared to the sample of DM91 in two areas - it includes stars out to K3 spectral type while DM91 stopped at G9, and it extends the declination coverage to all-sky from their limit of $> -15^{\circ}$. However, checks of subsamples from this work matching their criteria yield statistically indistinguishable results with the overall sample, showing that these do not introduce an undue bias (§ 8.3.1 and § 8.2.5). Including incompleteness analysis and estimating the fraction of candidate companions that are real, the probable percentages of single, double, triple, and quadruple or higher-order systems derived here are $55\% \pm 3\%$, $34\% \pm 2\%$, $9\% \pm 2\%$, and $2\% \pm 1\%$, and the percentage of single stars could drop by another percent if we account for possible new companions in the CCPS radial-velocity data (§ 8.2.5).

The large sample of this study enables the creation of subsamples to check the dependence of multiplicity ratios on physical parameters such as temperature (or mass, as checked using colors and spectral types), age, and metallicity. The recognized trend that more massive stars have a higher percentage of binaries is seen to hold even within the stars of this sample. The B - V color as well as spectral type analyses show a clear divide at around B - V = 0.63and spectral type of G5 (§ 8.3.1). Roughly 50% of the stars bluer than this limit, but only about 40% of stars redder than this limit, have stellar companions. Figure 8.6 shows how the stars of this sample fit the overall trend of multiplicity from O stars to T brown dwarfs, but a steeper drop-off than previously thought is observed at solar-type stars. In terms of age, represented by chromospheric emission and measured as $\log(R'_{\rm HK})$, the current sample divides into a relatively young, more active subgroup, 47% of which have companions, and a relatively old, less active subgroup, only 40% of which have companions (§ 8.3.2). Finally, while it has been shown that stars with higher metallicity are more likely to have planets, the same is not true for stellar companions (§8.3.3). This is not surprising because, while planets may require the presence of heavier elements to trigger the condensation process by which they form, the primordial matter that can form one star should be equally likely to form others as well. But this clear difference between stars and planets allows one to check which of them brown dwarfs resemble. While there are only a few data points and more work is needed before drawing definitive conclusions, these results tend to show that brown dwarfs, like planets, are more likely to form around stars with higher metallicity. This leads us to a preliminary conclusion that brown dwarfs, at least when they are companions to stars, form like planets rather than like stars (Figure 8.9).

Looking at orbital elements, binaries among solar-type stars have a period range from a few hours to millions of years. The period distribution seems to follow a roughly Gaussian pattern (Figure 8.11) with a peak and median period of about 300 years, slightly larger than that of Pluto around the Sun. This is significantly larger than the 180-year incompletenesscorrected value in DM91, showing that the current effort is more complete with respect to wide companions. The significant overlap of the various techniques in each period bin (Figure 8.11) shows that there are no major gaps in parameter space that are not adequately addressed by current techniques, a testament to the exacting methods developed over the years. The period-eccentricity relation derived here shows that beyond the expected circularization for short-period systems with periods below 12 days, the distribution is largely flat out to e = 0.6, followed by a fall-off for large eccentricities. I was not able to confirm either the roughly Gaussian distribution seen by DM91 for periods below 1000 days, or the f(e) = 2e distribution they claim for the longer-period systems. Finally, the companion mass-ratio distribution shows a clear preference for nearly-equal-mass pairs (Figure 8.14), suggesting that there is likely some formation mechanism that favors twins. This result is consistent with the results of Abt & Levy (1976), but another departure from the conclusion of DM91, who saw no such peak. A closer look at the mass-ratio-period relationship (Figure 8.15) reveals that higher-mass ratio pairs are more likely to have shorter periods, suggesting that they may preferentially form by fission. However, the vast majority of the pairs have long periods, even in the highest-mass-ratio bins, demonstrating that binaries form by multiple mechanisms – fission, which may be confined to like pairs, and fragmentation or random associations, which prefer longer periods and apply to the majority of stellar systems The mass-ratio distribution also shows a deficiency of low-mass companions, in contrast with some earlier results (Scarfe 1986; Trimble 1987, 1990), but consistent with the DM91 observations. However, DM91 presumed that the deficiency of lower-mass companions was due to missed pairs. We now have enough evidence that the paucity of low-mass companions is not a detection gap, but rather physical, showing that the brown dwarf desert extends to wide separations $(\S 8.2.3)$.

Two-thirds of the 34 exoplanet hosts of this sample are single stars, significantly lower than those in prior studies (Raghavan et al. 2006; Mugrauer & Neuhäuser 2009), and approaching the overall statistics of the volume-limited sample. This is surprising, because planet search efforts avoid known binaries, yielding a higher percentage of single stars among planet hosts, and suggests that the assessment of stellar companions of this study is comprehensive. The fraction of planet hosts among single, binary, and triple systems are statistically indistinguishable, indicating that planets are as likely to form around single stars as they are around components of binary or multiple systems. The period distribution in Figure 8.11 also shows that stellar companions prefer a wide separation. Only 34% of the pairs have periods below 27 years, which corresponds to a separation of 10 AU for an mass-sum of 1.47, the average values for all pairs of this study. This could be interpreted as good news for planets and life, as only a few stellar companions are close enough to disrupt objects in the habitable zones around these stars. So, binary systems may offer more places for planets, and perhaps life, not less as has been thought.



FIGURE 8.17: Mobile diagrams of triple systems (1 of 3). Solid lines connect confirmed companions and dashed lines connect candidate companions.



FIGURE 8.18: Mobile diagrams of triple systems (2 of 3). Solid lines connect confirmed companions and dashed lines connect candidate companions.


FIGURE 8.19: Mobile diagrams of triple systems (3 of 3). Solid lines connect confirmed companions and dashed lines connect candidate companions. The "p" subscript for HD 186408 B indicates that it is a planet-host star.



FIGURE 8.20: Mobile diagrams of quadruple systems. Solid lines connect confirmed companions and dashed lines connect candidate companions.



FIGURE 8.21: Mobile diagrams of quintuple and higher order systems. Solid lines connect confirmed companions and dashed lines connect candidate companions.

If I have seen farther than others, it is because I was standing on the shoulder of giants.

—— Sir Isaac Newton

REFERENCES

- Abt, H. A. 1970, ApJS, 19, 387
- Abt, H. A., & Biggs, E. S. 1972, Bibliography of stellar radial velocities (New York: Latham Process Corp.)
- Abt, H. A., & Levy, S. G. 1976, ApJS, 30, 273
- Abt, H. A., & Willmarth, D. 2006, ApJS, 162, 207
- Aitken, R. G. 1964, The Binary Stars (New York: Dover Publication)
- Aitken, R. G., & Doolittle, E. 1932, New general catalogue of double stars within 120 of the North pole ... (Washington, D.C.: Carnegie institution of Washington)
- Albrow, M. D., Gilliland, R. L., Brown, T. M., Edmonds, P. D., Guhathakurta, P., & Sarajedini, A. 2001, ApJ, 559, 1060
- Allen, P. R., Koerner, D. W., McElwain, M. W., Cruz, K. L., & Reid, I. N. 2007, AJ, 133, 971
- Ambartsumian, V. A. 1937, Astron. Zh., 14, 207
- Bagnuolo, Jr., W. G. et al. 2006, AJ, 131, 2695
- Baines, E. K., McAlister, H. A., ten Brummelaar, T. A., Turner, N. H., Sturmann, J., Sturmann, L., Goldfinger, P. J., & Ridgway, S. T. 2008a, ApJ, 680, 728
- Baines, E. K., McAlister, H. A., ten Brummelaar, T. A., Turner, N. H., Sturmann, J., Sturmann, L., & Ridgway, S. T. 2008b, ApJ, 682, 577

- Bakos, G. Á., Lázár, J., Papp, I., Sári, P., & Green, E. M. 2002, PASP, 114, 974
- Ball, B., Drake, J. J., Lin, L., Kashyap, V., Laming, J. M., & García-Alvarez, D. 2005, ApJ, 634, 1336
- Baranne, A., Mayor, M., & Poncet, J. L. 1979, Vistas in Astronomy, 23, 279
- Baranne, A. et al. 1996, A&AS, 119, 373
- Barlow, R. J. 1999, Statistics: A Guide to the Use of Statistical Methods in the Physical Sciences (Chichester: John Wiley & Sons Ltd.)
- Batten, A. H., Fletcher, J. M., & MacCarthy, D. G. 1989, Publications of the Dominion Astrophysical Observatory Victoria, 17, 1
- Batten, A. H., Fletcher, J. M., & Mann, P. J. 1978, Publications of the Dominion Astrophysical Observatory Victoria, 15, 121
- Beavers, W. I., & Eitter, J. J. 1986, ApJS, 62, 147
- Bender, C., Simon, M., Prato, L., Mazeh, T., & Zucker, S. 2005, AJ, 129, 402
- Blazit, A., Bonneau, D., Koechlin, L., & Labeyrie, A. 1977, ApJ, 214, L79
- Boccaletti, A., Chauvin, G., Baudoz, P., & Beuzit, J.-L. 2008, A&A, 482, 939
- Boden, A. F. et al. 1999, ApJ, 527, 360
- Bonavita, M., & Desidera, S. 2007, A&A, 468, 721
- Bonneau, D., Blazit, A., Foy, R., & Labeyrie, A. 1980, A&AS, 42, 185
- Bonneau, D., & Foy, R. 1980, A&A, 86, 295
- Bopp, B. W., Evans, D. S., Laing, J. D., & Deeming, T. J. 1970, MNRAS, 147, 355
- Boss, A. P. 2006, ApJ, 641, 1148

- Bouvier, J., Rigaut, F., & Nadeau, D. 1997, A&A, 323, 139
- Bouy, H., Brandner, W., Martín, E. L., Delfosse, X., Allard, F., & Basri, G. 2003, AJ, 126, 1526
- Branch, D. 1976, ApJ, 210, 392
- Brandner, W., & Koehler, R. 1998, ApJ, 499, L79
- Brettman, O. H., Fried, R. E., Duvall, W. M., Hall, D. S., Poe, C. H., & Shaw, J. S. 1983, Information Bulletin on Variable Stars, 2389, 1
- Burgasser, A. J., Kirkpatrick, J. D., & Lowrance, P. J. 2005, AJ, 129, 2849
- Burgasser, A. J., Kirkpatrick, J. D., Reid, I. N., Brown, M. E., Miskey, C. L., & Gizis, J. E. 2003, ApJ, 586, 512
- Butler, R. P., Marcy, G. W., Williams, E., McCarthy, C., Dosanjh, P., & Vogt, S. S. 1996, PASP, 108, 500
- Butler, R. P. et al. 2006, ApJ, 646, 505
- Campbell, B., Walker, G. A. H., & Yang, S. 1988, ApJ, 331, 902
- Carney, B. W., & Latham, D. W. 1987, AJ, 93, 116
- Catala, C., Forveille, T., & Lai, O. 2006, AJ, 132, 2318
- Charbonneau, D., Brown, T. M., Latham, D. W., & Mayor, M. 2000, ApJ, 529, L45
- Chauvin, G., Lagrange, A.-M., Udry, S., Fusco, T., Galland, F., Naef, D., Beuzit, J.-L., & Mayor, M. 2006, A&A, 456, 1165
- Chauvin, G., Lagrange, A.-M., Udry, S., & Mayor, M. 2007, A&A, 475, 723
- Close, L. M. et al. 2005, Nature, 433, 286

- Close, L. M., Thatte, N., Nielsen, E. L., Abuter, R., Clarke, F., & Tecza, M. 2007, ApJ, 665, 736
- Cox, A. N. 2000, Allen's astrophysical quantities (New York:Springer-Verlag)
- Cutispoto, G., Kuerster, M., Messina, S., Rodono, M., & Tagliaferri, G. 1997, A&A, 320, 586
- da Silva, L., & Foy, R. 1987, A&A, 177, 204
- de Medeiros, J. R., & Mayor, M. 1999, A&AS, 139, 433
- Desidera, S., & Barbieri, M. 2007, A&A, 462, 345
- Dieterich, S. B., Henry, T. J., Golimowski, D. A., Krist, J. E., & Raghavan, D. 2009, AJ, in preparation
- Dommanget, J., & Nys, O. 2000, A&A, 363, 991
- Ducourant, C. et al. 2006, A&A, 448, 1235
- Duflot, M., Figon, P., & Meyssonnier, N. 1995, A&AS, 114, 269
- Duquennoy, A., & Mayor, M. 1991, A&A, 248, 485 (DM91)
- Duquennoy, A., Mayor, M., Andersen, J., Carquillat, J. M., & North, P. 1992, A&A, 254, L13+
- Duquennoy, A., Mayor, M., & Halbwachs, J.-L. 1991, A&AS, 88, 281
- Duquennoy, A., Tokovinin, A. A., Leinert, C., Glindemann, A., Halbwachs, J.-L., & Mayor,M. 1996, A&A, 314, 846
- Dyck, H. M., Benson, J. A., & Schloerb, F. P. 1995, AJ, 110, 1433
- Edwards, D. A., Evans, D. S., Fekel, F. C., & Smith, B. W. 1980, AJ, 85, 478

Eggen, O. J. 1956, AJ, 61, 405

- Eggenberger, A., Udry, S., Chauvin, G., Beuzit, J.-L., Lagrange, A.-M., Ségransan, D., & Mayor, M. 2007, A&A, 474, 273
- Eggenberger, A., Udry, S., & Mayor, M. 2004, A&A, 417, 353
- Eisenbeiss, T., Seifahrt, A., Mugrauer, M., Schmidt, T. O. B., Neuhäuser, R., & Roell, T. 2007, Astronomische Nachrichten, 328, 521
- Els, S. G., Sterzik, M. F., Marchis, F., Pantin, E., Endl, M., & Kürster, M. 2001, A&A, 370, L1
- Evans, D. S. 1950, Monthly Notes of the Astronomical Society of South Africa, 9, 69
- ——. 1952, Monthly Notes of the Astronomical Society of South Africa, 11, 71
- Fabrycky, D., & Tremaine, S. 2007, ApJ, 669, 1298
- Falin, J. L., & Mignard, F. 1999, A&AS, 135, 231
- Farrington, C. D. 2008, PhD thesis, Georgia State University
- Fekel, F. C., Williamson, M., & Pourbaix, D. 2007, AJ, 133, 2431
- Fekel, Jr., F. C. 1981, ApJ, 246, 879
- Fischer, D. A., & Marcy, G. W. 1992, ApJ, 396, 178
- Fischer, D. A., & Valenti, J. 2005, ApJ, 622, 1102
- Frankowski, A., Jancart, S., & Jorissen, A. 2007, A&A, 464, 377
- Frasca, A., Guillout, P., Marilli, E., Freire Ferrero, R., Biazzo, K., & Klutsch, A. 2006, A&A, 454, 301

- Fricke, W. et al. 1988, Veroeffentlichungen des Astronomischen Rechen-Instituts Heidelberg, 32, 1
- Fuhrmann, K., Guenther, E., König, B., & Bernkopf, J. 2005, MNRAS, 361, 803
- Gaidos, E. J., Henry, G. W., & Henry, S. M. 2000, AJ, 120, 1006
- Gatewood, G., & Coban, L. 2009, AJ, 137, 402
- Gatewood, G., Han, I., & Black, D. C. 2001, ApJ, 548, L61
- Ghez, A. M., McCarthy, D. W., Patience, J. L., & Beck, T. L. 1997, ApJ, 481, 378
- Ghez, A. M., Neugebauer, G., & Matthews, K. 1993, AJ, 106, 2005
- Gizis, J. E., Kirkpatrick, J. D., Burgasser, A., Reid, I. N., Monet, D. G., Liebert, J., &
 Wilson, J. C. 2001, ApJ, 551, L163
- Gizis, J. E., Monet, D. G., Reid, I. N., Kirkpatrick, J. D., & Burgasser, A. J. 2000, MNRAS, 311, 385
- Gliese, W. 1969, Veroeffentlichungen des Astronomischen Rechen-Instituts Heidelberg, 22, 1
- Gliese, W., & Jahreiß, H. 1979, Bulletin d'Information du Centre de Donnees Stellaires, 16, 92
- ——. 1991, Preliminary Version of the Third Catalogue of Nearby Stars, Tech. rep.
- Goldberg, D., Mazeh, T., & Latham, D. W. 2003, ApJ, 591, 397
- Goldberg, D., Mazeh, T., Latham, D. W., Stefanik, R. P., Carney, B. W., & Laird, J. B. 2002, AJ, 124, 1132
- Goldin, A., & Makarov, V. V. 2006, ApJS, 166, 341
- Golimowski, D. A., & Schroeder, D. J. 1998, AJ, 116, 440

- Gomez, A. E., & Abt, H. A. 1982, PASP, 94, 650
- Gontcharov, G. A. 2006, Astronomy Letters, 32, 759
- Gontcharov, G. A., Andronova, A. A., & Titov, O. A. 2000, A&A, 355, 1164
- Gray, R. O., Corbally, C. J., Garrison, R. F., McFadden, M. T., Bubar, E. J., McGahee,C. E., O'Donoghue, A. A., & Knox, E. R. 2006, AJ, 132, 161
- Gray, R. O., Corbally, C. J., Garrison, R. F., McFadden, M. T., & Robinson, P. E. 2003, AJ, 126, 2048
- Grether, D., & Lineweaver, C. H. 2006, ApJ, 640, 1051
- Griffin, R. F. 1967, ApJ, 148, 465
- ——. 1991, Bulletin of the Astronomical Society of India, 19, 183
- ——. 1998, The Observatory, 118, 273
- ——. 1999, The Observatory, 119, 27
- ——. 2000, The Observatory, 120, 1
- ——. 2002, The Observatory, 122, 329
- ——. 2004, The Observatory, 124, 258
- ——. 2008, The Observatory, 128, 176
- Guirado, J. C. et al. 1997, ApJ, 490, 835
- Halbwachs, J. L. 1986, A&A, 168, 161
- ——. 1987, A&A, 183, 234
- Halbwachs, J. L., Arenou, F., Mayor, M., Udry, S., & Queloz, D. 2000, A&A, 355, 581
- Halbwachs, J. L., Mayor, M., Udry, S., & Arenou, F. 2003, A&A, 397, 159

- Hambly, N. C., Irwin, M. J., & MacGillivray, H. T. 2001, MNRAS, 326, 1295
- Harrington, R. S. 1977, AJ, 82, 753
- Harrington, R. S., & Dahn, C. C. 1980, AJ, 85, 454
- Harrington, R. S., & Miranian, M. 1977, PASP, 89, 400
- Hartkopf, W. I., & Mason, B. D. 2004, in Revista Mexicana de Astronomia y Astrofísica, vol.
 27, Vol. 21, Revista Mexicana de Astronomia y Astrofísica Conference Series, ed. C. Allen
 & C. Scarfe, 83–90
- Hartkopf, W. I., Mason, B. D., & McAlister, H. A. 1996, AJ, 111, 370
- Hartkopf, W. I., Mason, B. D., & Rafferty, T. J. 2008, AJ, 135, 1334
- Hartkopf, W. I., Mason, B. D., & Worley, C. E. 2001, AJ, 122, 3472
- Hartkopf, W. I., & McAlister, H. A. 1984, PASP, 96, 105
- Hawley, S. L., Gizis, J. E., & Reid, I. N. 1996, AJ, 112, 2799
- Heacox, W. D. 1995, AJ, 109, 2670
- ——. 1998, AJ, 115, 325
- Heintz, W. D. 1968, AJ, 73, 512
- ——. 1969, JRASC, 63, 275
- ----. 1978
- ——. 1981, ApJS, 46, 247
- ——. 1988, JRASC, 82, 140
- ——. 1990, AJ, 99, 420
- ——. 1994, AJ, 108, 2338

——. 1996, AJ, 111, 408

- Heintz, W. D., & Borgman, E. R. 1984, AJ, 89, 1068
- Henry, T. J., & McCarthy, Jr., D. W. 1990, ApJ, 350, 334
- ——. 1993, AJ, 106, 773
- Henry, T. J., Soderblom, D. R., Donahue, R. A., & Baliunas, S. L. 1996, AJ, 111, 439
- Henry, T. J., Subasavage, J. P., Brown, M. A., Beaulieu, T. D., Jao, W.-C., & Hambly,N. C. 2004, AJ, 128, 2460
- Henry, T. J., Walkowicz, L. M., Barto, T. C., & Golimowski, D. A. 2002, AJ, 123, 2002
- Hertzsprung, E. 1922, Bull. Astron. Inst. Netherlands, 1, 149
- Hirst, W. P. 1943, Monthly Notes of the Astronomical Society of South Africa, 2, 100
- Hoffleit, D., & Jaschek, C. 1982, The Bright Star Catalogue 4th Ed. (New Haven: Yale University Observatory)
- Hoffleit, D., Saladyga, M., & Wlasuk, P. 1983, Bright star catalogue. Supplement (New Haven: Yale University Observatory, 1983)
- Høg, E. et al. 2000, A&A, 355, L27
- Høg, E., Kuzmin, A., Bastian, U., Fabricius, C., Kuimov, K., Lindegren, L., Makarov, V. V.,& Roeser, S. 1998, A&A, 335, L65
- Holberg, J. B., Oswalt, T. D., & Sion, E. M. 2002, ApJ, 571, 512
- Holberg, J. B., Sion, E. M., Oswalt, T., McCook, G. P., Foran, S., & Subasavage, J. P. 2008, AJ, 135, 1225
- Holman, M. J., & Wiegert, P. A. 1999, AJ, 117, 621

- Horch, E. P., van Altena, W. F., Cyr, W. M., Kinsman-Smith, L., Srivastava, A., & Zhou, J. 2008, AJ, 136, 312
- Hummel, C. A., Armstrong, J. T., Buscher, D. F., Mozurkewich, D., Quirrenbach, A., & Vivekanand, M. 1995, AJ, 110, 376
- Hummel, C. A., Armstrong, J. T., Quirrenbach, A., Buscher, D. F., Mozurkewich, D., Simon,R. S., & Johnston, K. J. 1993, AJ, 106, 2486
- Hutchings, J. B., Griffin, R. F., & Ménard, F. 2000, PASP, 112, 833
- Hynek, J. A. 1938, Contributions of Perkins Observatory, 1, 10
- Imbert, M. 1979, A&AS, 38, 401
- ——. 2006, Romanian Astronomical Journal, 16, 3
- Innes, R. T. A. 1901, MNRAS, 61, 414
- Ivanova, N., Belczynski, K., Fregeau, J. M., & Rasio, F. A. 2005, MNRAS, 358, 572
- Janson, M., Brandner, W., Lenzen, R., Close, L., Nielsen, E., Hartung, M., Henning, T., & Bouy, H. 2007, A&A, 462, 615
- Jaschek, C., & Gómez, A. E. 1970, PASP, 82, 809
- Jaschek, C., & Jaschek, M. 1957, PASP, 69, 546
- Joergens, V. 2008, A&A, 492, 545
- Jones, H. R. A., Paul Butler, R., Marcy, G. W., Tinney, C. G., Penny, A. J., McCarthy, C.,
 & Carter, B. D. 2002, MNRAS, 337, 1170
- Joy, A. H. 1947, ApJ, 105, 96
- Kähler, H. 1999, A&A, 346, 67

- Kennedy, P. M., & Przybylski, A. 1963, MNRAS, 126, 381
- Kharchenko, N. V. 2001, Kinematika i Fizika Nebesnykh Tel, 17, 409
- Kirillova, T. S., & Pavlovskaya, E. D. 1963, Soviet Astronomy, 7, 99
- Kirkpatrick, J. D., Dahn, C. C., Monet, D. G., Reid, I. N., Gizis, J. E., Liebert, J., & Burgasser, A. J. 2001, AJ, 121, 3235
- Kobulnicky, H. A., & Fryer, C. L. 2007, ApJ, 670, 747
- Konacki, M. 2005, ApJ, 626, 431
- König, B., Fuhrmann, K., Neuhäuser, R., Charbonneau, D., & Jayawardhana, R. 2002, A&A, 394, L43
- Kouwenhoven, M. B. N. 2006, PhD thesis, University of Amsterdam, The Netherlands
- Kouwenhoven, M. B. N., Brown, A. G. A., Portegies Zwart, S. F., & Kaper, L. 2007, A&A, 474, 77
- Kraus, A. L., White, R. J., & Hillenbrand, L. A. 2006, ApJ, 649, 306
- Kuiper, G. P. 1935a, PASP, 47, 15
- ——. 1935b, PASP, 47, 121
- ——. 1942, ApJ, 95, 201
- Kunkel, W. E., Liebert, J., & Boroson, T. A. 1984, PASP, 96, 891
- Lagrange, A.-M., Beust, H., Udry, S., Chauvin, G., & Mayor, M. 2006, A&A, 459, 955
- Lasker, B. M. et al. 2008, AJ, 136, 735
- Latham, D. W., Davis, R. J., Stefanik, R. P., Mazeh, T., & Abt, H. A. 1991, AJ, 101, 625

- Latham, D. W., Stefanik, R. P., Torres, G., Davis, R. J., Mazeh, T., Carney, B. W., Laird,
 J. B., & Morse, J. A. 2002, AJ, 124, 1144
- Lawson, P. R., ed. 2000, Principles of Long Baseline Stellar Interferometry (Pasadena, Jet Propulsion Laboratory)
- Leinert, C., Zinnecker, H., Weitzel, N., Christou, J., Ridgway, S. T., Jameson, R., Haas, M., & Lenzen, R. 1993, A&A, 278, 129
- Lépine, S., & Shara, M. M. 2005, AJ, 129, 1483
- Lepine, S., Thorstensen, J. R., Shara, M. M., & Rich, R. M. 2009, ArXiv e-prints
- Liu, M. C., Fischer, D. A., Graham, J. R., Lloyd, J. P., Marcy, G. W., & Butler, R. P. 2002, ApJ, 571, 519
- Looper, D. L., Kirkpatrick, J. D., & Burgasser, A. J. 2007, AJ, 134, 1162
- Lowrance, P. J., Kirkpatrick, J. D., & Beichman, C. A. 2002, ApJ, 572, L79
- Lu, W., Rucinski, S. M., & Ogłoza, W. 2001, AJ, 122, 402
- Luhman, K. L., & Jayawardhana, R. 2002, ApJ, 566, 1132
- Luhman, K. L. et al. 2007, ApJ, 654, 570
- Luyten, W. J. 1927, Harvard College Observatory Bulletin, 852, 14
- ——. 1930a, Proceedings of the National Academy of Science, 16, 252
- ——. 1930b, Proceedings of the National Academy of Science, 16, 257
- ——. 1936, ApJ, 84, 85
- Luyten, W. J. 1979, in New Luyten Catalogue of stars with proper motions larger than two tenths of an arcsecond, 2, 0 (1979), 0

- Luyten, W. J., & Shapley, H. 1930, Annals of Harvard College Observatory, 85, 73
- Makarov, V. V., & Kaplan, G. H. 2005, AJ, 129, 2420
- Makarov, V. V., Zacharias, N., Hennessy, G. S., Harris, H. C., & Monet, A. K. B. 2007, ApJ, 668, L155
- Malkov, O. Y., Oblak, E., Snegireva, E. A., & Torra, J. 2006, A&A, 446, 785
- Marcy, G., Butler, R. P., Fischer, D., Vogt, S., Wright, J. T., Tinney, C. G., & Jones,H. R. A. 2005, Progress of Theoretical Physics Supplement, 158, 24
- Marcy, G. W., & Butler, R. P. 1996, ApJ, 464, L147
- Marcy, G. W., Butler, R. P., Fischer, D. A., & Vogt, S. S. 2004, in Astronomical Society of the Pacific Conference Series, Vol. 321, Extrasolar Planets: Today and Tomorrow, ed. J. Beaulieu, A. Lecavelier Des Etangs, & C. Terquem, 3
- Mason, B. D., Gies, D. R., Hartkopf, W. I., Bagnuolo, Jr., W. G., ten Brummelaar, T., & McAlister, H. A. 1998a, AJ, 115, 821
- Mason, B. D., Hartkopf, W. I., Gies, D. R., Henry, T. J., & Helsel, J. W. 2009a, AJ, 137, 3358
- Mason, B. D., Hartkopf, W. I., Holdenried, E. R., & Rafferty, T. J. 2001, AJ, 121, 3224
- Mason, B. D., Henry, T. J., Hartkopf, W. I., Ten Brummelaar, T., & Soderblom, D. R. 1998b, AJ, 116, 2975
- Mason, B. D., Henry, T. J., & Soderblom, D. R. 2009b, AJ, in preparation
- Mason, B. D. et al. 1999, AJ, 117, 1890
- Mason, B. D., McAlister, H. A., Hartkopf, W. I., & Shara, M. M. 1995, AJ, 109, 332

- Maxted, P. F. L., Jeffries, R. D., Oliveira, J. M., Naylor, T., & Jackson, R. J. 2008, MNRAS, 385, 2210
- Mayor, M., & Mazeh, T. 1987, A&A, 171, 157
- Mazeh, T. 1990, AJ, 99, 675
- Mazeh, T., Goldberg, D., Duquennoy, A., & Mayor, M. 1992, ApJ, 401, 265
- Mazeh, T., Martin, E. L., Goldberg, D., & Smith, H. A. 1997, MNRAS, 284, 341
- Mazeh, T., Prato, L., Simon, M., Goldberg, E., Norman, D., & Zucker, S. 2002, ApJ, 564, 1007
- Mazeh, T., Simon, M., Prato, L., Markus, B., & Zucker, S. 2003, ApJ, 599, 1344
- McAlister, H. A. 1978a, PASP, 90, 288
- ——. 1978b, ApJ, 225, 932
- ——. 1979, ApJ, 230, 497
- McAlister, H. A., Hartkopf, W. I., Hutter, D. J., & Franz, O. G. 1987, AJ, 93, 688
- McAlister, H. A., Mason, B. D., Hartkopf, W. I., & Shara, M. M. 1993, AJ, 106, 1639
- McCarthy, C., & Zuckerman, B. 2004, AJ, 127, 2871
- Melo, C. H. F. 2003, A&A, 410, 269
- Metchev, S., & Hillenbrand, L. 2008, ArXiv e-prints
- Michelson, A. A. 1890, Phil.Mag., 30, 1
- ——. 1920, ApJ, 51, 257
- Mishenina, T. V., Soubiran, C., Bienaymé, O., Korotin, S. A., Belik, S. I., Usenko, I. A., & Kovtyukh, V. V. 2008, A&A, 489, 923

- Monet, D. G. et al. 2003, AJ, 125, 984
- Moore, J. H., & Paddock, G. F. 1950, ApJ, 112, 48
- Morbey, C. L., & Griffin, R. F. 1987, ApJ, 317, 343
- Moutou, C. et al. 2008, A&A, 488, L47
- Mugrauer, M., & Neuhäuser, R. 2005, MNRAS, 361, L15
- ——. 2009, A&A, 494, 373
- Mugrauer, M., Neuhäuser, R., Seifahrt, A., Mazeh, T., & Guenther, E. 2005, A&A, 440, 1051
- Murdoch, K. A., Hearnshaw, J. B., & Clark, M. 1993, ApJ, 413, 349
- Musielak, Z. E., Cuntz, M., Marshall, E. A., & Stuit, T. D. 2005, A&A, 434, 355
- Muterspaugh, M. W. 2005, PhD thesis, Massachusetts Institute of Technology
- Nadal, R., Ginestet, N., Carquillat, J.-M., & Pedoussaut, A. 1979, A&AS, 35, 203
- Nather, R. E., & Evans, D. S. 1970, AJ, 75, 575
- Neuhäuser, R., & Guenther, E. W. 2004, A&A, 420, 647
- Nidever, D. L., Marcy, G. W., Butler, R. P., Fischer, D. A., & Vogt, S. S. 2002, ApJS, 141, 503
- Nordström, B. et al. 2004, A&A, 418, 989
- Noyes, R. W., Jha, S., Korzennik, S. G., Krockenberger, M., Nisenson, P., Brown, T. M., Kennelly, E. J., & Horner, S. D. 1997, ApJ, 487, L195
- Oppenheimer, B. R., Golimowski, D. A., Kulkarni, S. R., Matthews, K., Nakajima, T., Creech-Eakman, M., & Durrance, S. T. 2001, AJ, 121, 2189

- Patience, J., Ghez, A. M., Reid, I. N., & Matthews, K. 2002a, AJ, 123, 1570
- Patience, J. et al. 2002b, ApJ, 581, 654
- Perryman, M. A. C., & ESA, eds. 1997, ESA Special Publication, Vol. 1200, The HIPPAR-COS and TYCHO catalogues. Astrometric and photometric star catalogues derived from the ESA HIPPARCOS Space Astrometry Mission
- Petrie, R. M. 1960, Annales d'Astrophysique, 23, 744
- Pilat-Lohinger, E., Funk, B., & Dvorak, R. 2003, A&A, 400, 1085
- Pollacco, D. L. et al. 2006, PASP, 118, 1407
- Potter, D., Martín, E. L., Cushing, M. C., Baudoz, P., Brandner, W., Guyon, O., & Neuhäuser, R. 2002, ApJ, 567, L133
- Pourbaix, D. 2000, A&AS, 145, 215
- Pourbaix, D., Neuforge-Verheecke, C., & Noels, A. 1999, A&A, 344, 172
- Pourbaix, D. et al. 2002, A&A, 386, 280
- Poveda, A., Herrera, M. A., Allen, C., Cordero, G., & Lavalley, C. 1994, Revista Mexicana de Astronomia y Astrofisica, 28, 43
- Queloz, D. et al. 2000, A&A, 354, 99
- Quist, C. F., & Lindegren, L. 2000, A&A, 361, 770
- Raghavan, D., Henry, T. J., Mason, B. D., Subasavage, J. P., Jao, W.-C., Beaulieu, T. D.,
 & Hambly, N. C. 2006, ApJ, 646, 523
- Raghavan, D. et al. 2009, ApJ, 690, 394
- Rapaport, M. et al. 2001, A&A, 376, 325

- Reid, I. N. et al. 2004, AJ, 128, 463
- Reid, I. N., & Gizis, J. E. 1997, AJ, 114, 1992
- Reid, I. N., Hawley, S. L., & Gizis, J. E. 1995, AJ, 110, 1838
- Richichi, A. 2000, A&A, 364, 225
- Roberts, Jr., L. C. et al. 2005, AJ, 130, 2262
- Rossiter, R. A. 1955, Publications of Michigan Observatory, 11, 1
- Rovithis, P., & Rovithis-Livaniou, H. 1990, A&AS, 86, 523
- Salim, S., & Gould, A. 2003, ApJ, 582, 1011
- Salim, S., Rich, R. M., Hansen, B. M., Koopmans, L. V. E., Oppenheimer, B. R., & Blandford, R. D. 2004, ApJ, 601, 1075
- Scardia, M. et al. 2008, Astronomische Nachrichten, 329, 54
- Scarfe, C. D. 1986, JRASC, 80, 257
- Schaefer, B. E., King, J. R., & Deliyannis, C. P. 2000, ApJ, 529, 1026
- Schultz, A. B. et al. 1998, AJ, 115, 345
- Shatsky, N. 2001, A&A, 380, 238
- Shatsky, N., & Tokovinin, A. 2002, A&A, 382, 92
- Siegler, N., Close, L. M., Cruz, K. L., Martín, E. L., & Reid, I. N. 2005, ApJ, 621, 1023
- Sinachopoulos, D. 1988, A&AS, 76, 189
- Skuljan, J., Ramm, D. J., & Hearnshaw, J. B. 2004, MNRAS, 352, 975
- Söderhjelm, S. 1999, A&A, 341, 121
- 2000, Astronomische Nachrichten, 321, 165

- Soderhjelm, S., & Mignard, F. 1998, The Observatory, 118, 365
- Sollima, A., Beccari, G., Ferraro, F. R., Fusi Pecci, F., & Sarajedini, A. 2007, MNRAS, 380, 781
- Subasavage, J. P., Jao, W.-C., Henry, T. J., Bergeron, P., Dufour, P., Ianna, P. A., Costa, E., & Mendez, R. A. 2009, ArXiv e-prints
- Takeda, G., Kita, R., & Rasio, F. A. 2008, ApJ, 683, 1063
- ten Brummelaar, T., Mason, B. D., McAlister, H. A., Roberts, Jr., L. C., Turner, N. H.,
 - Hartkopf, W. I., & Bagnuolo, Jr., W. G. 2000, AJ, 119, 2403
- ten Brummelaar, T. A. et al. 2005, ApJ, 628, 453
- The, C. D. 2005, VizieR Online Data Catalog, 2263, 0
- Tokovinin, A. 2008, MNRAS, 389, 925
- Tokovinin, A., Thomas, S., Sterzik, M., & Udry, S. 2006, A&A, 450, 681
- Tokovinin, A. A. 1985, A&AS, 61, 483
- ——. 1991, A&AS, 91, 497
- ——. 1997, A&AS, 124, 75
- ——. 2000, A&A, 360, 997
- Tokovinin, A. A., & Ismailov, R. M. 1988, A&AS, 72, 563
- Torres, G., Boden, A. F., Latham, D. W., Pan, M., & Stefanik, R. P. 2002, AJ, 124, 1716
- Trimble, V. 1987, Astronomische Nachrichten, 308, 343
- ——. 1990, MNRAS, 242, 79

- Turner, N. H., ten Brummelaar, T. A., McAlister, H. A., Mason, B. D., Hartkopf, W. I., & Roberts, Jr., L. C. 2001, AJ, 121, 3254
- Turner, N. H., ten Brummelaar, T. A., Roberts, L. C., Mason, B. D., Hartkopf, W. I., & Gies, D. R. 2008, AJ, 136, 554
- Udalski, A. et al. 2002, Acta Astronomica, 52, 1
- Udry, S., Eggenberger, A., Mayor, M., Mazeh, T., & Zucker, S. 2004, in Revista Mexicana de Astronomia y Astrofisica, vol. 27, Vol. 21, Revista Mexicana de Astronomia y Astrofisica Conference Series, ed. C. Allen & C. Scarfe, 207–214
- Udry, S. et al. 1998, in Astronomical Society of the Pacific Conference Series, Vol. 154, Cool Stars, Stellar Systems, and the Sun, ed. R. A. Donahue & J. A. Bookbinder, 2148–+
- Udry, S., & Santos, N. C. 2007, ARA&A, 45, 397
- Valenti, J. A., & Fischer, D. A. 2005, ApJS, 159, 141
- van Altena, W. F., Lee, J. T., & Hoffleit, E. D. 1995, The general catalogue of trigonometric stellar parallaxes 4th Ed. (New Haven, CT: Yale University Observatory)
- van Leeuwen, F., ed. 2007a, Astrophysics and Space Science Library, Vol. 250, Hipparcos, the New Reduction of the Raw Data
- van Leeuwen, F. 2007b, A&A, 474, 653 (FvL07)
- Verrier, P. E., & Evans, N. W. 2007, MNRAS, 382, 1432
- Vogt, S. S., Butler, R. P., Marcy, G. W., Fischer, D. A., Henry, G. W., Laughlin, G., Wright,J. T., & Johnson, J. A. 2005, ApJ, 632, 638

- Vogt, S. S., Butler, R. P., Marcy, G. W., Fischer, D. A., Pourbaix, D., Apps, K., & Laughlin,
 G. 2002, ApJ, 568, 352
- Weis, E. W. 1996, AJ, 112, 2300
- Wertheimer, J. G., & Laughlin, G. 2006, AJ, 132, 1995
- Wilson, J. C., Kirkpatrick, J. D., Gizis, J. E., Skrutskie, M. F., Monet, D. G., & Houck,J. R. 2001, AJ, 122, 1989
- Wilson, O. C. 1967, AJ, 72, 905
- Woitas, J., Leinert, C., & Köhler, R. 2001, A&A, 376, 982
- Wright, J. T., Marcy, G. W., Butler, R. P., & Vogt, S. S. 2004, ApJS, 152, 261
- Wright, J. T. et al. 2007, ApJ, 657, 533
- Wright, J. T., Upadhyay, S., Marcy, G. W., Fischer, D. A., Ford, E. B., & Johnson, J. A. 2008, ArXiv e-prints
- Zacharias, N., Monet, D. G., Levine, S. E., Urban, S. E., Gaume, R., & Wycoff, G. L. 2004, in Bulletin of the American Astronomical Society, Vol. 36, Bulletin of the American Astronomical Society, 1418–+

Zinnecker, H. 1984, Ap&SS, 99, 41

Appendices

Physical Parameters for Primaries and Companions

This appendix lists the physical parameters extracted for the primary stars and extracted or estimated values for many companions. These data are used in the analyses presented in Chapter 8 (see the discussion in § 8.3 for more information).

tars	
iple s	
san	
f the	
mass o	
and	
parameters	
A.1: Physical]	
TABLE	

HD	HIP	Spec		Mass					
Name	Name	Type	Ref	(M_{\odot})	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
Sun	:	G2V	:	1.00	:	0.00	:	:	
000123	000518	G4V	Gra2003	0.98	Nor2004	-0.01	Nor2004	-4.644	Gra2003
000166	000544	G8V	Gra2003	0.96	Val2005	0.12	Val2005	-4.458	Gra2003
000870	001031	$\mathrm{K}0\mathrm{V}$	Gra2006	0.81	Nor2004	-0.20	Nor2004	-4.824	Gra2006
001237	001292	G8.5V	Gra2006	0.85	Nor2004	-0.09	Nor2004	-4.496	${ m Gra2006}$
001273	001349	G5V	Gra2006	0.80	Nor2004	-0.65	Nor2004	-4.802	Gra2006
001461	001499	G3V	Gra2003	1.14	Val2005	0.16	Val2005	-5.030	Wri2004
001562	001598	G1V	Gra2003	0.85	Nor2004	-0.34	Nor2004	-4.979	Gra2003
001581	001599	F9.5V	Gra2006	1.17	Val2005	-0.18	Val2005	-4.855	Gra2006
001835	001803	G5V	Gra2006	0.98	Val2005	0.22	Val2005	-4.440	Wri2004
002025	001936	K3V	Gra2006	0.70	Val2005	-0.27	Val2005	-4.933	Gra2006
002151	002021	G0V	Gra2006	1.43	Val2005	-0.09	Val2005	-5.006	Gra2006
003196	002762	F8.5V	Gra2003	1.13	Nor2004	-0.07	Nor2004	-4.461	Gra2003
003443	002941	G7V	Gra2006	0.94	Pou2000	-0.14	Nor2004	-4.940	Wri2004
003651	003093	K0V	Gra2003	0.79	Val2005	0.16	Val2005	-5.020	Wri2004
003765	003206	K2.5V	Gra2003	0.84	Val2005	0.12	Val2005	-5.101	Gra2003
004256	003535	K3IV-V	Gra2003	1.36	Val2005	0.22	Val2005	-5.042	Gra2003
004308	003497	G6V	$\operatorname{Gra2006}$	1.47	Val2005	-0.18	Val2005	-4.853	Gra2006
004391	003583	G5V	Gra2006	0.88	Nor2004	-0.25	Nor2004	-4.669	Gra2006
004614	003821	G0V	HIP	1.10	Val2005	-0.17	Val2005	-4.930	Wri2004
004628	003765	K2V	Gra2003	0.63	Val2005	-0.19	Val2005	-5.071	Gra2003
004635	003876	K2.5V+	Gra2003					-4.670	Wri2004
004676	003810	F8V	Gra2003	1.22	$\operatorname{Bod1999}$	-0.06	Nor2004	-4.917	Gra2003
004747	003850	G9V	$\operatorname{Gra2006}$	0.99	Val2005	-0.25	Val2005	-4.720	Wri2004
004813	003909	F7V	Gra2003	1.10	Nor2004	-0.15	Nor2004	-4.780	Gra2003
Continu	ed on Nex	t Page							

ЦЦ	НГР	Spec		Mass					
Name	Name	\mathbf{Type}	Ref	(M_{\odot})	Ref	$[\rm Fe/H]$	Ref	$\log(R'_{HK})$	Ref
004915	003979	G6V	Gra2003	1.02	Val2005	-0.18	Val2005	-4.860	Wri2004
005015	004151	F8V	HIP	1.25	Nor2004			-5.02	MAs2009b
005133	004148	K2.5V	Gra2006	1.19	Val2005	-0.13	Val2005	-4.751	Gra2006
006582	005336	K1V	Gra2003	0.75	Nor2004	-0.83	Nor2004	-5.031	Gra2003
007570	005862	F9V	Gra2006	1.16	Val2005	0.14	Val2005	-4.861	Gra2006
007590	005944	G0-V	Gra2003	1.12	Val2005	-0.07	Val2005	-4.530	Wri2004
007693	005842	K2+V	Gra2006	0.92	$\operatorname{Sod}1999$	0.05	Val2005	-4.580	Gra2006
007924	006379	$\mathrm{K0}$	HIP	0.80	Val2005	-0.12	Val2005	-4.830	Wri2004
008997	006917	K2.5V	Gra2003	:		-0.59	Nor2004	-4.557	Gra2003
009407	007339	G6.5V	Gra2003	1.13	Val2005			-5.010	Wri2004
009540	007235	G8.5V	Gra2006	0.91	Val2005	-0.04	Val2005	-4.600	Wri2004
022600	007372	K2V	Gra2006	0.74	$\operatorname{Sod1999}$	-0.65	Nor2004	-4.354	Gra2006
009826	007513	F8V	HIP	1.2	Low 2002	0.12	Val2005	-5.040	Wri2004
010008	007576	G9V	Gra2003	0.86	Nor2004	-0.03	Gra2003	-4.530	Gra2003
010086	007734	G5V	Gra2003	0.98	Val2005	0.09	Val2005	-4.600	Wri2004
010307	007918	G1V	Gra2003	1.01	Nor2004	-0.05	Nor2004	-5.017	Gra2003
010360	007751	K2V	Gra2006	0.44	Val2005	-0.20	Val2005	-4.899	Gra2006
010476	007981	K0V	Gra2003	0.81	Val2005	-0.07	Val2005	-4.950	Wri2004
010647	007978	F9V	Gra2006	0.98	Val2005	-0.08	Val2005	-4.675	Gra2006
010700	008102	G8.5V	Gra2006	0.99	Val2005	-0.36	Val2005	-4.980	Wri2004
010780	008362	G9V	Gra2003	0.88	Val2005	-0.06	Val2005	-4.690	Wri2004
012051	009269	G9V	Gra2003	1.37	Val2005	0.15	Val2005	-5.050	Wri2004
012846	009829	G2V-	Gra2003	0.87	Val2005	-0.20	Val2005	-4.980	Wri2004
013445	010138	K1V	Gra2006	0.92	Val2005	-0.20	Val2005	-4.768	Gra2006
013974	010644	G0V	HIP	0.85	Nor2004	-0.39	Nor2004	-4.710	Wri2004
014214	010723	G0IV-	$\operatorname{Gra}2003$	1.14	Nor2004	0.06	Nor2004	-5.114	${ m Gra2003}$
014412	010798	G8V	Gra2006	1.03	Val2005	-0.45	Val2005	-4.850	Wri2004
Continu	ted on Next	t Page							

TABLE	A.1 - Coi	ntinued							
HD Name	HIP Name	Spec Type	Ref	Mass (M _☉)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
014802	011072	G0V	Gra2006	1.15	Nor2004	-0.08	Nor2004	-5.050	Wri2004
016160	012114	K3V	Gra2003	0.90	Val2005	0.08	Nor2004	-5.094	Gra2003
016287	012158	K2.5V	Gra2003	0.97	Val2005	0.08	Val2005	-4.504	Gra2003
016673	012444	F8V	Gra2003	1.12	Nor2004	-0.11	Nor2004	-4.586	Gra2003
016739	012623	F9IV-V	Gra2003	1.38	$\operatorname{Bag2006}$	0.14	Nor2004	-5.012	Gra2003
016765	012530	F7V	Gra2003	1.06	Nor2004	-0.24	Nor2004	-4.400	Gra2003
016895	012777	F7V	HIP	1.37	Val2005	0.02	Val2005	-4.970	Wri2004
017051	012653	F9V	Gra2006	1.07	Val2005	0.09	Val2005	-4.625	Gra2006
017382	013081	$\mathrm{K}0\mathrm{V}$	Gra2003	:	:	:	:	-4.450	Wri2004
017925	013402	K1.5V	Gra2006	0.81	Val2005	0.11	Val2005	-4.357	${ m Gra2006}$
018143	013642	K2IV	Gra2003	1.05	Val2005	0.21	Val2005	-5.119	Gra2003
018632	013976	K2.5V	Gra2003	0.55	Val2005	0.18	Val2005	-4.418	Gra2003
018757	014286	G1.5V	Gra2003	0.84	Nor2004	-0.44	Nor2004	-4.987	Gra2003
018803	014150	G6V	Gra2003	0.97	Val2005	0.09	Val2005	-4.880	Wri2004
019373	014632	F9.5V	Gra2003	1.35	Val2005	0.13	Val2005	-5.020	Wri2004
019994	014954	F8.5V	Gra2003	1.81	Val2005	0.17	Val2005	-4.880	Wri2004
020010	014879	F6V	Gra2006	1.30	Nor2004	-0.16	Nor2004	-4.901	${ m Gra2006}$
020165	015099	K1V	Gra2003	0.66	Val2005	-0.04	Val2005	-4.860	Wri2004
020407	015131	G5V	Gra2006	0.90	Nor2004	-0.46	Nor2004	-4.734	${ m Gra2006}$
020619	015442	G2V	Gra2003	1.06	Val2005	-0.18	Val2005	-4.830	Wri2004
020630	015457	G5Vvar	HIP	0.94	Val2005	0.10	Val2005	-4.47	MAs2009b
020794	015510	G8V	Gra2006	1.30	Val2005	-0.23	Val2005	-4.998	${ m Gra2006}$
020807	015371	G0V	Gra2006	1.19	Val2005	-0.23	Val2005	-4.827	${ m Gra2006}$
021175	015799	K1V	Gra2006	0.85	Nor2004	0.12	Nor2004	-4.773	Gra2006
022049	016537	K2V	Gra2006	0.70	Val2005	-0.15	Gra2006	-4.510	Wri2004
022484	016852	F9V	HIP	1.46	Val2005	-0.03	Val2005	-5.120	Wri2004
022879	017147	F9V	HIP	1.12	Val2005	-0.76	Val2005	-4.920	Wri2004
Continu	ed on Nex	tt Page							

Continued	
A.1 –	
BLE	

TABLE	A.1 - Coi	ntinued							
HD	HIP	Spec	Rof	Mass	Rof	[Fa /H]	Rof	$\log(R')$	Rof
		Type	TAGT		TOT		TOT	(XH) r)Sot	TOOT
023356	017420	K2.5V	Gra2006	0.74	Val2005	-0.07	Val2005	-4.807	Gra2006
023484	017439	K2V	Gra2006	0.66	Val2005	0.04	Val2005	-4.534	${ m Gra2006}$
024238	018324	K2V	Gra2003	0.87	Val2005	-0.32	Val2005	-4.980	Wri2004
024409	018413	G3V	Gra2003	0.87	Nor2004	-0.25	Nor2004	-4.927	Gra2003
024496	018267	G7V	Gra2003	1.08	Val2005	-0.01	Val2005	-4.870	Wri2004
025329	018915	K3Vp	Gra2003	0.58	Nor2004	-1.69	Nor2004	-4.940	Wri2004
025457	018859	F7V	Gra2003	1.15	Nor2004	-0.10	Nor2004	-4.390	Wri2004
025665	019422	K2.5V	Gra2003	0.66	Val2005	-0.06	Val2005	-4.847	Gra2003
025680	019076	G1V	Gra2003	1.15	Val2005	0.04	Val2005	-4.608	Gra2003
025998	019335	F8V	Gra2003	1.22	Nor2004	0.02	Nor2004	-4.468	Gra2003
026491	019233	G1V	Gra2006	1.06	Val2005	-0.06	Val2005	-4.889	Gra2006
026923	019859	G0V	Gra2006	1.01	Nor2004	-0.13	Nor2004	-4.521	Gra2006
026965	019849	K0.5V	Gra2006	0.89	Val2005	-0.08	Val2005	-4.900	Wri2004
029883	021988	K5III	HIP	0.89	Val2005	-0.16	Val2005	-4.87	MAs2009b
030495	022263	G1.5V	Gra2006	0.86	Val2005	-0.01	Val2005	-4.600	Wri2004
030501	022122	K2V	Gra2006	0.82	Nor2004	-0.09	Gra2006	-4.762	Gra2006
030876	022451	K2V	HIP	0.85	Val2005	-0.11	Val2005	-4.55	MAs2009b
032778	023437	G7V	Gra2006	0.95	Val2005	-0.48	Val2005	-4.870	Gra2006
032850	023786	G9V	Gra2003	0.80	Nor2004	-0.19	Nor2004	-4.600	Wri2004
032923	023835	G1V	Gra2003	1.44	Val2005	-0.13	Val2005	-5.030	Wri2004
033262	023693	F9V	Gra2006	1.07	Nor2004	-0.20	Nor2004	-4.373	Gra2006
033564	025110	F7V	Gra2003	1.25	Nor2004	-0.06	Nor2004	-4.949	Gra2003
034411	024813	G1V	Gra2003	1.36	Val2005	0.09	Val2005	-5.050	Wri2004
034721	024786	F9-V	Gra2006	1.17	Val2005	-0.08	Val2005	-5.030	Wri2004
035112	025119	K2.5V	Gra2003	:		-0.27	Nor2004	-4.879	Gra2003
035296	025278	F8V	Gra2003	1.06	Nor2004	-0.15	Nor2004	-4.353	Gra2003
035854	025421	K3-V	Gra2006	0.77	Val2005	-0.04	Val2005	-4.922	Gra2006
Continu	ted on Nex	t Page							

Continued
A.1
ABLE

TABLE	A.1 – Cor	ntinued							
HD Name	HIP Name	${ m Spec} { m Type}$	Ref	${ m Mass}$ (${ m M}_{\odot}$)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
036435	025544	G9V	Gra2006	0.83	Nor2004	-0.18	Nor2004	-4.499	Gra2006
036705	025647	K2V	Gra2006	0.87	Clo2005	-0.59	Nor2004	-3.880	Gra2006
037008	026505	K1V	Gra2003	0.80	Val2005	-0.31	Val2005	-4.960	Wri2004
037394	026779	$\mathrm{K}0\mathrm{V}$	Gra2003	0.88	Val2005	0.16	Val2005	-4.553	Gra2003
037572	026373	K1.5V	Gra2006	0.76	Nor2004	-0.49	Nor2004	-4.234	Gra2006
038230	027207	$\mathrm{K}0\mathrm{V}$	Gra2003	0.90	Val2005	-0.02	Val2005	-4.990	Wri2004
038858	027435	G2V	Gra2003	1.05	Val2005	-0.21	Val2005	-4.950	Wri2004
039091	026394	G0V	Gra2006	1.11	Val2005	0.04	Val2005	-4.941	Gra2006
039587	027913	G0V	Gra2006	1.10	Cat2006	-0.16	Nor2004	-4.426	Gra2006
039855	027922	G8V	Gra2006	0.80	Nor2004	-0.49	Nor2004	-4.932	Gra2006
040307	027887	K2.5V	Gra2006	0.77	Val2005	-0.25	Val2005	-5.037	Gra2006
040397	028267	G7V	Gra2003	1.50	Val2005	-0.05	Val2005	-4.980	Wri2004
041593	028954	G9V	Gra2003	:	:	:	:	-4.390	Wri2004
042618	029432	G3V	Gra2003	0.84	Val2005	-0.09	Val2005	-4.940	Wri2004
042807	029525	G5V	Gra2003	0.87	Nor2004	-0.21	Nor2004	-4.465	Gra2003
043162	029568	G6.5V	Gra2006	0.88	Nor2004	-0.10	Nor2004	-4.400	Wri2004
043587	029860	G0V	Gra2006	1.17	Val2005	-0.08	Val2005	-5.000	Wri2004
043834	029271	G7V	Gra2006	1.10	Val2005	0.05	Val2005	-4.940	Gra2006
045088	030630	$\mathrm{K}3\mathrm{V}$	Gra2003	0.83	This work	-0.85	Nor2004	-4.266	Gra2003
045184	030503	G1.5V	Gra2006	0.97	Val2005	0.03	Val2005	-4.950	Wri2004
045270	030314	G0Vp	Gra2006	0.96	Nor2004	-0.18	Nor2004	-4.378	Gra2006
046588	032439	F8V	Gra2003	1.03	Nor2004	-0.25	Nor2004	-4.885	Gra2003
048189	031711	G1V	Gra2006	0.94	Nor2004	-0.23	Nor2004	-4.268	Gra2006
048682	032480	F9V	Gra2003	1.12	Val2005	0.09	Val2005	-4.850	Wri2004
050692	033277	G0V	Gra2003	0.98	Val2005	-0.13	Val2005	-4.940	Wri2004
051419	033537	G5V	Gra2003	1.00	Val2005	-0.33	Val2005	-4.870	Wri2004
051866	033852	K3V	Gra2003	:			•	-4.889	Gra2003
Continu	ted on Nex	\overline{t} Page							

per
ntin
Coj
A.1
ABLE

TABLE	A.1 – Cor	ntinued							
HD Name	HIP Name	Spec Type	Ref	${ m Mass}$ (${ m M}_{\odot}$)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
052698	033817	K1V	Gra2006					-4.590	Wri2004
052711	034017	G0V	Gra2003	1.13	Val2005	-0.10	Val2005	-4.960	Wri2004
053143	033690	K0IV-V	Gra2006	0.85	Nor2004	-0.01	Nor2004	-4.547	Gra2006
053705	034065	G0V	Gra2006	1.31	Val2005	-0.15	Val2005	-4.981	Gra2006
053927	034414	K2.5V	Gra2003	0.74	Nor2004	-0.27	Nor2004	-4.984	Gra2003
054371	034567	G6V	Gra2003	0.88	Nor2004	-0.10	Nor2004	-4.462	Gra2003
055575	035136	F9V	Gra2003	0.95	Nor2004	-0.35	Nor2004	-4.950	Wri2004
057095	035296	K2.5V	Gra2006		:			-4.538	Gra2006
059468	036210	G6.5V	Gra2006	1.04	Val2005	0.01	Val2005	-4.946	Gra2006
059747	036704	K1V	Gra2003	0.74	Val2005	-0.03	Val2005	-4.370	Wri2004
059967	036515	G2V	Gra2006	0.88	Nor2004	-0.24	Nor2004	-4.372	Gra2006
060491	036827	K2.5V	Gra2003	0.86	Val2005	-0.26	Val2005	-4.430	Wri2004
061606	037349	K3-V	Gra2003	0.97	Val2005	-0.05	Val2005	-4.390	Wri2004
062613	038784	G8V	HIP	0.83	Nor2004	-0.23	Nor2004	-4.840	Wri2004
063077	037853	F9V	Gra2006	0.84	Nor2004	-0.75	Nor2004	-4.970	Wri2004
063433	038228	G5V	Gra2003	0.95	Val2005	0.02	Val2005	-4.390	Wri2004
064096	038382	G0V	Gra2006	0.93	Pou2000	-0.18	Nor2004	-4.883	Gra2006
064468	038657	K2.5V	Gra2003	0.66	Val2005	0.09	Val2005	-5.146	Gra2003
064606	038625	$\mathrm{K}0\mathrm{V}$	Gra2003	0.74	Nor2004	-0.80	Nor2004	-4.940	Wri2004
065430	039064	$\mathrm{K}0\mathrm{V}$	Gra2003	0.87	Val2005	-0.04	Val2005	-5.010	Wri2004
065583	039157	$\mathrm{K}0\mathrm{V}$	Gra2003	1.29	Val2005	-0.48	Val2005	-4.950	Wri2004
065907	038908	F9.5V	Gra2006	1.77	Val2005	-0.15	Val2005	-4.846	Gra2006
067199	039342	K2V	Gra2006	0.77	Val2005	0.01	Nor2004	-4.843	Gra2006
067228	039780	G2IV	Gra2003	1.22	Val2005	0.17	Val2005	-5.120	Wri2004
068017	040118	G3V	Gra2003	1.55	Val2005	-0.30	Val2005	-4.920	Wri2004
068257	040167	F8V	Gra2003	1.52	Nor2004	0.08	Nor2004		
069830	040693	G8+V	Gra2006	0.86	Val2005	-0.08	Val2005	-4.950	Wri2004
Continu	ed on Nex	t Page							

HD	HIP	Spec		Mass					
Name	Name	Type	Ref	(\mathbf{M}_{\odot})	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
071148	041484	G1V	Gra2003	1.03	Val2005	-0.01	Val2005	-4.950	Wri2004
072673	041926	G9V	Gra2006	0.98	Val2005	-0.33	Val2005	-4.950	Wri2004
072760	042074	$\mathrm{K0-V}$	Gra2003	0.92	Val2005	0.03	Val2005	-4.380	Wri2004
072905	042438	G1.5Vb	HIP	0.90	Nor2004	-0.25	Nor2004	-4.400	Wri2004
073350	042333	G5V	Gra2003	1.01	Val2005	0.04	Val2005	-4.490	Wri2004
073667	042499	K2V	Gra2003	0.91	Val2005	-0.36	Val2005	-4.970	Wri2004
073752	042430	G5IV	Gra2006	1.25	$\operatorname{Sod1999}$	0.31	Nor2004	-5.031	Gra2006
074385	042697	K2+V	Gra2006	0.79	Nor2004	-0.03	Nor2004	-4.902	Gra2006
074576	042808	K2.5V	Gra2006	0.77	Nor2004	-0.24	Nor2004	-4.402	Gra2006
075732	043587	K0IV-V	Gra2003	0.92	Val2005	0.25	Val2005	-5.040	Wri2004
075767	043557	G1.5V	Gra2003	0.91	Nor2004	-0.18	Nor2004	-4.638	Gra2003
076151	043726	G3V	Gra2006	1.24	Val2005	0.07	Val2005	-4.853	Gra2006
076932	044075	G2V	Gra2006	0.89	Nor2004	-0.70	Nor2004	-4.781	Gra2006
078366	044897	G0IV-V	Gra2003	1.34	Val2005	0.03	Val2005	-4.555	Gra2003
079028	045333	G0IV-V	Gra2003	1.08	Nor2004	-0.10	Nor2004	-5.073	Gra2003
079096	045170	G9V	Gra2003	0.89	Pou2000	-0.30	Nor2004	-4.846	Gra2003
079969	045617	K3V	Gra2003	0.74	$\operatorname{Sod1999}$	-0.22	Nor2004	-4.736	Gra2003
080715	045963	K2.5V	Gra2003		:	-0.58	Gra2003	-4.099	Gra2003
082342	046626	K3.5V	Gra2006	:	:	:		-5.131	Gra2006
082443	046843	K1V	Gra2006		:		:	-4.234	Gra2006
082558	046816	$\mathrm{K0}$	HIP	2.05	Val2005	-0.21	Val2005	-4.09	MAs2009b
082885	047080	K0V	VJHK	:		:	: : :	-4.68	MAs2009b
084117	047592	F8V	Gra2006	1.11	Val2005	-0.08	Val2005	-4.862	Gra2006
084737	048113	G0IV-V	Gra2003	1.42	Val2005	0.14	Val2005	-5.230	Wri2004
086728	049081	G4V	Gra2003	0.73	Val2005	0.11	Val2005	-5.060	Wri2004
087424	049366	K2V	Gra2006	0.78	Val2005	-0.14	Val2005	-4.440	Wri2004
087883	049699	K2.5V	Gra2003	0.78	Val2005	0.04	Val2005	-4.999	Gra2003
Continu	ted on Ney	xt Page							

TABLE	A.1 - Cor	ntinued							
HD Name	HIP Name	Spec Tvne	Ref	Mass (M.)	Ref	[Fe/H]	Ref	$\log(R'_{min})$	Ref
								(VII)0	
088742	<i>c1</i> 00c0	607	Gra2006	0.99	Val2005	-0.04	Val2005	-4.806	Gra2006
089125	050384	F6V	Gra2003	1.01	Nor2004	-0.44	Nor2004	-4.832	${ m Gra2003}$
089269	050505	G4V	Gra2003	0.94	Val2005	-0.18	Val2005	-4.940	Wri2004
090156	050921	G5V	Gra2006	1.25	Val2005	-0.21	Val2005	-4.950	Wri2004
090343	051819	$\mathrm{K0}$	HIP	:	:	:		-4.58	MAs2009b
090508	051248	G0V	Gra2003	0.85	Nor2004	-0.40	Nor2004	-5.005	Gra2003
090839	051459	F8V	Gra2003	0.99	Val2005	-0.05	Val2005	-4.860	Wri2004
091324	051523	F9V	Gra2006	1.16	Nor2004	-0.28	Nor2004	-4.766	Gra2006
091889	051933	F8V	Gra2006	1.04	Nor2004	-0.27	Nor2004	-4.849	Gra2006
092719	052369	G1.5V	Gra2006	0.91	Nor2004	-0.21	Nor2004	-4.826	Gra2006
092945	052462	K1.5V	Gra2006	1.18	Val2005	-0.12	Val2005	-4.320	Wri2004
094765	053486	K2.5V	Gra2003	1.19	Val2005	-0.03	Val2005	-4.546	Gra2003
095128	053721	G0V	HIP	1.29	Val2005	0.02	Val2005	-5.020	Wri2004
096064	054155	G8+V	Gra2003	0.83	Nor2004	-0.13	Nor2004	-4.373	Gra2003
096612	054426	K3-V	Gra2003		:			-4.836	Gra2003
097334	054745	G1V	Gra2003	0.93	Val2005	0.08	Val2005	-4.368	Gra2003
097343	054704	G8.5V	Gra2006	1.11	Val2005	-0.05	Val2005	-5.000	Wri2004
097658	054906	K1V	Gra2003	0.90	Val2005	-0.27	Val2005	-4.920	Wri2004
098230	055203	G2V	Bal2005	0.96	Nor2004	-0.30	Nor2004	:	:
098281	055210	G8V	HIP	1.00	Val2005	-0.17	Val2005	-4.940	Wri2004
099491	055846	K0IV	HIP	1.15	Val2005	0.24	Val2005	-4.840	Wri2004
100180	056242	F9.5V	Gra2003	1.01	Val2005	-0.02	Val2005	-4.940	Wri2004
100623	056452	K0-V	Gra2006	0.96	Val2005	-0.32	Val2005	-4.890	Wri2004
101177	056809	F9.5V	Gra2003	0.99	Val2005	-0.17	Val2005	-4.930	Wri2004
101206	056829	K5V	HIP	:	:	:	:	-4.52	MAs2009b
101501	056997	G8V	Gra2003	0.73	Val2005	-0.03	Val2005	-4.550	Wri2004
102365	057443	G2V	Gra2006	1.22	Val2005	-0.26	Val2005	-4.957	Gra2006
Continu	ted on Nex	t Page							

Continued	
A.1 -	
ABLE	

HD	HIP	Spec		Mass					
Name	Name	\mathbf{Type}	Ref	(${ m M}_{\odot}$)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
102438	057507	G6V	Gra2006	1.01	Val2005	-0.23	Val2005	-4.924	Gra2006
102870	057757	F8V	HIP	1.61	Val2005	0.16	Val2005	-4.940	Wri2004
103095	057939	K1V	Gra2003	0.71	Val2005	-1.16	Val2005	-4.850	Wri2004
104067	058451	K3-V	Gra2006	0.91	Val2005	0.04	Val2005	-4.751	Gra2006
104304	058576	G8IV	Gra2006	1.34	Val2005	0.16	Val2005	-4.920	Wri2004
105631	059280	G9V	Gra2003	0.91	Val2005	0.14	Val2005	-4.650	Wri2004
108954	061053	F9V	Gra2003	1.04	Nor2004	-0.13	Nor2004	-4.921	${ m Gra2003}$
109200	061291	K1V	Gra2006	0.79	Val2005	-0.23	Val2005	-5.124	Gra2006
109358	061317	G0V	Gra2003	1.05	Val2005	-0.10	Val2005	-4.920	Wri2004
110463	061946	K3V	HIP					-4.47	MAs2009b
110810	062229	K2+V	Gra2006	0.76	Val2005	-0.03	Val2005	-4.441	Gra2006
110833	062145	K3V	HIP	0.84	Nor2004	0.08	Nor2004	-4.70	MAs2009b
110897	062207	F9V	Gra2003	0.83	Nor2004	-0.59	Nor2004	-4.869	Gra2003
111312	062505	K2.5V	Gra2006					-4.571	Gra2006
111395	062523	G7V	Gra2003	1.08	Val2005	0.06	Val2005	-4.580	Wri2004
112758	063366	G9V	Gra2006	0.76	Nor2004	-0.38	Nor2004	-5.067	Gra2006
112914	063406	K3-V	Gra2003	0.70	Val2005	-0.26	Val2005	-5.043	Gra2003
113283	064690	G5V	Gra2006	0.87	Nor2004	-0.15	Nor2004	-4.720	Gra2006
113449	063742	K1V	Gra2003	:		:	:	-4.340	Gra2003
114613	064408	G4IV	Gra2006	1.80	Val2005	0.16	Val2005	-5.118	Gra2006
114710	064394	G0V	HIP	1.54	Val2005	0.04	Val2005	-4.760	Wri2004
114783	064457	K1V	Gra2003	0.75	Val2005	0.10	Val2005	-5.056	Gra2003
114853	064550	G1.5V	Gra2006	1.04	Val2005	-0.24	Val2005	-4.936	Gra2006
115383	064792	G0Vs	HIP	2.31	Val2005	0.21	Val2005	-4.400	Wri2004
115404	064797	K2.5V	Gra2003	:		-0.49	Gra2003	-4.640	Gra2003
115617	064924	G7V	Gra2006	0.99	Val2005	0.09	Val2005	-5.040	Wri2004
116442	065352	G9V	Gra2003	0.75	Val2005	-0.30	Val2005	-4.940	Wri2004
Continu	ted on Next	t Page							

TABLE A.1 – Continued

TABLE	A.1 - Coi	ntinued							
HD Name	HIP Name	Spec Type	Ref	Mass (M _o)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
116956	065515	G9V	Gra2003	:		-0.13	Nor2004	-4.447	Gra2003
117043	065530	G6V	HIP	0.89	Nor2004	0.06	Nor2004	-4.96	MAs2009b
117176	065721	G5V	HIP	1.48	Val2005	-0.01	Val2005	-4.990	Wri2004
118972	066765	K0V	Gra2006	0.77	Val2005	-0.09	Val2005	-4.439	${ m Gra2006}$
119332	066781	K0IV-V	HIP			:	:	-4.70	MAs2009b
120136	067275	F7V	HIP	1.33	Val2005	0.25	Val2005	-4.700	Wri2004
120559	067655	G7V	Gra2006			-0.95	Nor2004	-5.029	Gra2006
120690	067620	G5+V	Gra2006	1.11	Val2005	0.05	Val2005	-4.750	Wri2004
120780	067742	K2V	Gra2006	0.59	Val2005	-0.26	Val2005	-4.888	Gra2006
121370	067927	G0IV	HIP	1.62	Nor2004	0.27	Nor2004	•	•
121560	068030	F6V	HIP	0.96	Val2005	-0.41	Val2005	-4.920	Wri2004
122742	068682	G6V	Gra2003	1.01	Val2005	-0.08	Nor2004	-4.955	Gra2003
124106	069357	K1V	Gra2006	0.79	Val2005	-0.13	Val2005	-4.630	Wri2004
124292	069414	G8+V	Gra2003	1.30	Val2005	-0.10	Val2005	-4.970	Wri2004
124580	069671	G0V	Gra2006	0.92	Nor2004	-0.28	Nor2004	-4.597	Gra2006
124850	069701	F7V	HIP	1.52	Nor2004	-0.04	Nor2004		:
125276	069965	F9V	Gra2006	0.97	Nor2004	-0.62	Nor2004	-4.641	Gra2006
125455	070016	K1V	HIP	0.78	Val2005	-0.15	Val2005	-4.930	Wri2004
126053	070319	G1.5V	Gra2003	1.14	Val2005	-0.29	Val2005	-4.940	Wri2004
127334	070873	G5V	Gra2003	1.17	Val2005	0.22	Val2005	-5.060	Wri2004
128165	071181	K3V	HIP	0.91	Val2005	-0.09	Val2005	-4.67	MAs2009b
128311	071395	K3-V	Gra2003	1.32	Val2005	0.01	Val2005	-4.489	Gra2003
128400	071855	G5V	Gra2006	0.86	Nor2004	-0.14	Nor2004	-4.518	Gra2006
128620	071683	G2V	Gra2006	1.11	Pou2002	0.19	Val2005	-5.059	Gra2006
128642	070857	G5	HIP	:	• • •	-0.39	Nor2004	-4.910	Wri2004
128987	071743	G8V	Gra2006	0.92	Nor2004	0.02	Nor2004	-4.439	Gra2006
130004	072146	K2.5V	Gra2003	•		•	•	-4.919	Gra2003
Continu	ted on Nex	tt Page							
TABLE	A.1 – Coi	ntinued							
------------	-------------	-------------------------	----------------------	----------------------------------	--------------------------	--------	----------------------	-----------------	----------------------
HD Name	HIP Name	Spec Type	Ref	${ m Mass}$ (${ m M}_{\odot}$)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
130042	072493	K1V	Gra2006	:	• •	-0.21	Nor2004	-4.949	Gra2006
130307	072312	K2.5V	Gra2003	0.81	Val2005	-0.20	Val2005	-4.560	Wri2004
130948	072567	G2V	HIP	1.35	Val2005	-0.19	Nor2004	-4.500	Wri2004
131156	072659	G7V	Gra2003	0.94	$\operatorname{Sod1999}$	-0.07	Val2005	-4.472	Gra2003
131511	072848	$\mathrm{K}0\mathrm{V}$	Gra2003	0.97	Val2005	0.11	Val2005	-4.510	Gra2003
131582	072875	K3V	HIP	0.69	Nor2004	-0.54	Nor2004		
131923	073241	G4V	Gra2006	1.56	Val2005	0.09	Val2005	-5.059	Gra2006
132142	073005	K1V	HIP	1.03	Val2005	-0.30	Val2005	-5.000	Wri2004
132254	073100	F8-V	Gra2003	1.23	Nor2004	0.02	Nor2004	-5.030	Gra2003
133640	073695	G2V	HIP	0.85	Nor2004	-0.42	Nor2004	-4.61	MAs2009b
134060	074273	G0V	Gra2006	1.14	Val2005	0.08	Val2005	-5.042	${ m Gra2006}$
135204	074537	G9V	Gra2003	0.85	Nor2004	-0.06	Nor2004	-5.107	Gra2003
135599	074702	$\mathrm{K}0\mathrm{V}$	Gra2003	0.61	Val2005	-0.09	Val2005	-4.520	Wri2004
136202	074975	F8III-IV	HIP	1.34	Nor2004	-0.04	Nor2004		• •
136352	075181	G2-V	Gra2006	1.51	Val2005	-0.23	Val2005	-5.013	Gra2006
136713	075253	K3IV-V	Gra2003	0.86	Val2005	0.15	Val2005	-4.878	Gra2003
136923	075277	G9V	Gra2003	0.93	Val2005	-0.07	Val2005	-4.770	Wri2004
137107	075312	G2V	HIP	1.19	Pou2000	-0.10	Nor2004	-4.76	MAs2009b
137763	075718	G9V	Gra2003	0.85	Nor2004	0.02	Nor2004	-4.970	Wri2004
139341	076382	K1V	Gra2003	0.85	$\operatorname{Sod1999}$:	-5.102	Gra2003
139777	075809	G1.5V(n)	Gra2003	0.86	Nor2004	-0.31	Nor2004	-4.405	Gra2003
140538	077052	G5V	HIP	0.98	Val2005	0.06	Val2005	-4.830	Wri2004
140901	077358	G7IV-V	Gra2006	1.07	Val2005	0.08	Val2005	-4.802	Gra2006
141004	077257	G0IV-V	Gra2003	1.27	Val2005	0.09	Val2005	-4.970	Wri2004
141272	077408	G9V	Gra2003	0.83	Nor2004	-0.05	Nor2004	-4.390	Wri2004
142267	077801	G0IV	HIP	1.09	Val2005	-0.38	Val2005	-4.850	Wri2004
142373	077760	G0V	Gra2003	1.46	Val2005	-0.39	Val2005	-5.110	Wri2004
Continu	ed on Nex	tt Page							

TABLE	A.1 - Co.	ntinued							
HD Name	HIP Name	${ m Spec} { m Type}$	Ref	${ m Mass}$ (${ m M}_{\odot}$)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
143761	078459	GOV	Gra2003	1.04	Gat2001	-0.14	Val2005	-5.080	Wri2004
144284	078527	F8IV-V	HIP	1.2	Maz2002	0.04	Nor2004	•	
144287	078709	G8+V	Gra2003	0.84	Nor2004	-0.13	Nor2004	-5.020	Wri2004
144579	078775	$\mathrm{K}0\mathrm{V}$	Gra2003	1.20	Val2005	-0.49	Val2005	-4.970	Wri2004
144628	079190	K1V	Gra2006	0.66	Val2005	-0.30	Val2005	-5.128	Gra2006
144872	078913	K3V	Gra2003	0.72	Nor2004	-0.30	Nor2004	-4.804	Gra2003
145417	079537	K3V	Gra2006	0.63	Nor2004	-1.37	Nor2004	-5.205	Gra2006
145675	079248	K0IV-V	Gra2003	1.07	Val2005	0.41	Val2005	-5.060	Wri2004
145825	079578	G2V	Gra2006	1.07	Val2005	0.03	Val2005	-4.793	Gra2006
145958	079492	G8V	Gra2003	0.88	Val2005	-0.07	Val2005	-4.940	Wri2004
146233	079672	G2V	Gra2003	0.98	Val2005	0.03	Val2005	-4.950	Wri2004
146361	709670	G1IV-V	Gra2003	1.14	m Rag2009	-0.26	Nor2004	-3.827	Gra2003
147513	080337	G1V	Gra2006	1.30	Val2005	0.07	Val2005	-4.656	Gra2006
147584	080686	F9V	Gra2006	1.06	Nor2004	-0.19	Nor2004	-4.585	Gra2006
147776	080366	K3-V	Gra2006	0.82	Val2005	-0.26	Val2005	-4.810	Gra2006
148653	080725	K2V	Gra2003	0.79	$\operatorname{Sod1999}$:		-4.655	Gra2003
148704	080925	K1V	Gra2006	:	:	-0.60	Nor2004	-5.101	Gra2006
149612	081520	G5V	Gra2006	1.02	Val2005	-0.40	Val2005	-4.954	Gra2006
149661	081300	$\mathrm{K}0\mathrm{V}$	Gra2006	0.85	Val2005	0.05	Val2005	-4.570	Wri2004
149806	081375	$\mathrm{K}0\mathrm{V}$	Gra2003	1.00	Val2005	0.17	Val2005	-4.830	Wri2004
151541	081813	K1V	HIP	0.96	Val2005	-0.18	Val2005	-4.990	Wri2004
152391	082588	G8.5V	Gra2006	0.92	Val2005	-0.05	Val2005	-4.440	Wri2004
153557	083020	K3V	Gra2003		:	-0.33	Nor2004	-4.508	Gra2003
154088	083541	K0IV-V	Gra2006	0.93	Val2005	0.28	Val2005	-5.020	Wri2004
154345	083389	G8V	HIP	0.90	Val2005	-0.10	Val2005	-4.910	Wri2004
154417	083601	F9V	Gra2003	0.84	Val2005	0.03	Val2005	-4.590	Wri2004
154577	083990	K2.5V	Gra2006	0.65	Val2005	-0.56	Val2005	-5.080	Gra2006
Continu	ted on Ney	tt Page							

Continued
Å.
ABLE

TABLE	A.1 - Coi	ntinued							
HD Name	HIP Name	${ m Spec} { m Type}$	Ref	Mass (M _☉)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
155712	084195	K2.5V	Gra2003	:	:	:		-4.988	Gra2003
155885	084405	K1.5V	$\operatorname{Gra2006}$:	•		:	-4.711	Gra2006
156274	084720	M0V	HIP	0.90	Val2005	-0.27	Val2005		
157214	084862	G0V	HIP	1.54	Val2005	-0.15	Val2005	-5.040	Wri2004
157347	085042	G3V	Gra2003	1.19	Val2005	0.03	Val2005	-5.040	Wri2004
158614	085667	G8IV-V	HIP	0.98	Pou2000	-0.01	Nor2004	-5.12	MAs2009b
158633	085235	$\mathrm{K}0\mathrm{V}$	HIP	0.94	Val2005	-0.33	Val2005	-4.930	Wri2004
159062	085653	G9V	Gra2003		:	-0.51	Nor2004	-5.030	Wri2004
159222	085810	G1V	Gra2003	0.80	Val2005	0.09	Val2005	-4.900	Wri2004
160269	086036	G0V	HIP	1.08	Sod 1999	-0.22	Nor2004	-4.61	MAs2009b
160346	086400	K2.5V	Gra2003	0.78	Nor2004	-0.09	Nor2004	-4.956	Gra2003
160691	086796	G3IV-V	Gra2006	1.25	Val2005	0.26	Val2005	-5.101	Gra2006
161198	086722	G9V	Gra2003	0.78	Nor2004	-0.39	Nor2004	-4.970	Wri2004
161797	086974	G5IV	HIP	1.34	Val2005	0.24	Val2005	-5.110	Wri2004
162004	086620	G0V	HIP	1.05	Nor2004	-0.18	Nor2004	-4.86	MAs2009b
164922	088348	G9V	Gra2003	1.05	Val2005	0.17	Val2005	-5.050	Wri2004
165185	088694	G0V	Gra2006	0.93	Nor2004	-0.25	Nor2004	-4.545	Gra2006
165341	088601	$\mathrm{K0-V}$	Gra2003	0.90	Pou2000	-0.29	Nor2004	-4.698	Gra2003
165401	088622	G0V	Gra2003	0.83	Nor2004	-0.50	Nor2004	-4.668	Gra2003
165499	089042	G0V	Gra2006	1.00	Nor2004	-0.17	Nor2004	-4.935	Gra2006
165908	088745	F7V	HIP	0.91	Nor2004	-0.55	Nor2004	-5.020	Wri2004
166620	088972	K2V	Gra2003	0.66	Val2005	-0.05	Val2005	-4.970	Wri2004
167425	089805	F9.5V	Gra2006	1.11	Nor2004	0.02	Nor2004	-4.606	Gra2006
168009	089474	G1V	Gra2003	0.98	Val2005	-0.02	Val2005	-5.080	Wri2004
170657	062060	K2V	Gra2006	0.79	Val2005	-0.15	Val2005	-4.650	Wri2004
172051	091438	G6V	Gra2006	0.91	Val2005	-0.24	Val2005	-4.900	Wri2004
175073	092858	K1V	Gra2006	0.80	Nor2004	-0.14	Nor2004	-4.889	G_{ra2006}
Continu	ed on Nex	t Page							

TABLE	A.1 - Coi	ntinued							
HD Name	HIP Name	Spec Type	Ref	${ m Mass}$ (${ m M}_{\odot}$)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
175742	092919	K0V	HIP	:	:	:	:	:	:
176051	093017	G0V	HIP	1.07	Sod 1999	-0.19	Nor2004		:
176377	093185	G1V	Gra2003	0.81	Val2005	-0.23	Val2005	-4.870	Wri2004
177474	093825	F8V	Gra2006	1.29	Nor2004	-0.22	Nor2004	-4.890	Gra2006
177565	093858	G6V	Gra2006	1.08	Val2005	0.07	Val2005	-4.973	Gra2006
178428	093966	G4V	Gra2006	0.97	Nor2004	0.05	Nor2004	-5.110	Gra2006
179957	094336	G3V	Gra2003	0.73	Val2005		• • •	-5.050	Wri2004
180161	094346	G8V	HIP	:	:		•	-4.520	Wri2004
181321	095149	G1V	Gra2006	0.89	Nor2004	-0.26	Nor2004	-4.372	Gra2006
182488	095319	G9+V	Gra2003	1.28	Val2005	0.12	Val2005	-4.940	Wri2004
182572	095447	G8IVvar	HIP	1.38	Val2005	0.36	Val2005	-5.100	Wri2004
183870	096085	K2.5V	Gra2006	0.96	Val2005	-0.05	Val2005	-4.512	Gra2006
184385	096183	G8V	Gra2003	0.88	Val2005	0.11	Val2005	-4.560	Wri2004
184467	095995	K2V	Gra2003	0.8	Pou2000	-0.22	Nor2004	-5.047	Gra2003
185144	096100	G9V	Gra2003	0.75	Val2005	-0.16	Val2005	-4.850	Wri2004
185414	096395	G0	HIP	0.95	Nor2004	-0.16	Nor2004	-4.88	MAs2009b
186408	096895	G1.5V	Gra2003	1.25	Val2005	0.08	Val2005	-5.100	Wri2004
186858	097222	K3+V	Gra2003	0.69	Sod 1999			-4.726	Gra2003
187691	097675	F8V	HIP	1.37	Val2005	0.12	Val2005	-5.050	Wri2004
189340	098416	F9V	Gra2003	1.12	Sod 1999	-0.05	Nor2004	-4.951	Gra2003
189567	098959	G2V	Gra2006	1.08	Val2005	-0.18	Val2005	-4.857	Gra2006
189733	098505	K2V	Gra2003	0.78	Nor2004	-0.12	Nor2004	-4.553	Gra2003
190067	779860	$\mathrm{K}0\mathrm{V}$	Gra2003	0.95	Val2005	-0.30	Val2005	-4.880	Wri2004
190248	099240	G8IV	Gra2006	1.04	Val2005	0.26	Val2005	-5.092	Gra2006
190360	098767	G7IV-V	Gra2006	1.16	Val2005	0.19	Val2005	-5.090	Wri2004
190404	098792	K1V	Gra2003	0.89	Val2005	-0.44	Val2005	-4.980	Wri2004
190406	098819	G0V	Gra2006	1.13	Val2005	0.02	Val2005	-4.770	Wri2004
Continu	ed on Nex	tt Page							

TABLE	A.1 - Coi	ntinued							
НD	HIP	Spec		Mass					,
Name	Name	Type	Ref	(M_{\odot})	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
190422	099137	F9V	Gra2006	1.04	Nor2004	-0.21	Nor2004	-4.458	Gra2006
190470	098828	K2.5V	Gra2003	:		-0.16	Gra2003	-4.828	Gra2003
190771	098921	G2V	Gra2003	0.96	Val2005	0.14	Val2005	-4.430	Wri2004
191408	099461	K2.5V	Gra2006	0.76	Val2005	-0.33	Val2005	-5.079	Gra2006
191499	099316	G9V	Gra2003			•		-5.076	Gra2003
191785	099452	K0V	Gra2003	1.08	Val2005	-0.09	Val2005	-5.030	Wri2004
192263	099711	K2.5V	Gra2003	0.83	Val2005	-0.07	Val2005	-4.676	Gra2003
192310	099825	K2+V	Gra2006	0.84	Val2005	0.14	Nor2004	-5.048	Gra2006
193664	100017	G0V	Gra2003	1.13	Val2005	-0.11	Val2005	-4.927	Gra2003
194640	100925	G8V	Gra2006	1.01	Val2005	-0.06	Val2005	-4.924	Gra2006
195564	101345	G2V	Gra2003	1.34	Val2005	0.01	Val2005	-5.130	Wri2004
195987	101382	G9V	Gra2003	0.84	Tor 2002	-0.38	Nor2004	-4.970	Wri2004
196378	101983	G0V	Gra2006	1.41	Val2005	-0.32	Val2005	-4.837	Gra2006
196761	101997	G8V	Gra2006	0.90	Val2005	-0.25	Val2005	-4.920	Wri2004
197076	102040	G1V	Gra2003	1.03	Val2005	-0.09	Val2005	-4.920	Wri2004
197214	102264	G6V	$\operatorname{Gra2006}$	0.81	Nor2004	-0.50	Nor2004	-4.920	Wri2004
198425	102766	K2.5V	Gra2003	:				-4.726	Gra2003
199260	103389	F6V	Gra2006	1.12	Nor2004	-0.17	Nor2004	-4.402	Gra2006
199288	103458	G2V	Gra2006	1.51	Val2005	-0.47	Val2005	-4.847	Gra2006
199509	104436	G1V	Gra2006	1.05	Val2005	-0.27	Val2005	-4.925	Gra2006
200525	104440	F9.5V	Gra2006	1.01	Nor2004	-0.13	Nor2004	-4.667	Gra2006
200560	103859	K2.5V	Gra2003	:		0.01	Nor2004	-4.512	Gra2003
200968	104239	G9.5V	Gra2006	0.86	Val2005	0.02	Val2005	-4.650	Gra2006
202275	104858	F7V	Gra2003	1.19	Pou2000	-0.07	Nor2004	-4.905	Gra2003
202628	105184	G1.5V	Gra2006	1.04	Val2005	-0.01	Val2005	-4.782	Gra2006
202751	105152	K3V	Gra2003	0.88	Val2005	-0.09	Val2005	-5.111	Gra2003
202940	105312	G7V	Gra2006	:		-0.34	Nor2004	-4.988	Gra2006
Continu	ted on Nex	tt Page							

led
ntin
Col
-
A
ABLI

ЦН	НІР	Spec		Mass					
Name	Name	Type	Ref	(M_{\odot})	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
203244	105712	G8V	Gra2006	0.82	Nor2004	-0.32	Nor2004	-4.555	Gra2006
203850	105905	K2.5V	Gra2006	0.68	Nor2004	-0.67	Nor2004	-5.033	${ m Gra2006}$
203985	105911	K2III-IV	Gra2006		:				:
205390	106696	K1.5V	Gra2006	0.78	Val2005	-0.18	Val2005	-4.702	${ m Gra2006}$
205536	107022	G9V	Gra2006	0.98	Val2005	-0.03	Val2005	-5.084	${ m Gra2006}$
206826	107310	F6V	Gra2003	1.39	Nor2004	-0.20	Nor2004	-4.783	Gra2003
206860	107350	G0V	Gra2006	1.08	Val2005	-0.01	Val2005	-4.400	${ m Gra2006}$
207129	107649	G0V	Gra2006	1.06	Val2005	-0.04	Val2005	-5.020	${ m Gra2006}$
207144	107625	K3V	Gra2006		:			-4.990	Gra2006
208038	108028	K2.5V	Gra2003	:	:			-4.569	Gra2003
208313	108156	K2V	Gra2003	0.74	Val2005	-0.08	Val2005	-4.987	${ m Gra2003}$
210277	109378	G8V	Gra2003	1.27	Val2005	0.20	Val2005	-5.060	Wri2004
210667	109527	G9V	Gra2003	0.95	Val2005	0.16	Val2005	-4.500	Wri2004
210918	109821	G2V	Gra2006	1.20	Val2005	-0.07	Val2005	-5.121	${ m Gra2006}$
211415	110109	G0V	Gra2006	0.85	Nor2004	-0.36	Nor2004	-4.918	${ m Gra2006}$
211472	109926	$\mathrm{K}0\mathrm{V}$	Gra2003	0.87	Nor2004	0.03	Nor2004	-4.538	Gra2003
212168	110712	G0V	Gra2006	1.13	Val2005	-0.04	Val2005	-4.981	${ m Gra2006}$
212330	110649	G2IV-V	Gra2006	1.40	Val2005	0.01	Val2005	-5.157	Gra2006
214683	111888	K3V	Gra2003	:	:			-4.554	Gra2003
214953	112117	F9.5V	Gra2006	0.81	Val2005	0.03	Val2005	-4.988	Gra2006
215152	112190	K3V	Gra2003	:		-0.17	Gra2003	-4.925	Gra2003
215648	112447	F7V	HIP	1.24	Val2005	-0.16	Val2005	-5.070	Wri2004
216259	112870	K2.5V	Gra2003	0.94	Val2005	-0.47	Val2005	-4.950	Wri2004
216520	112527	$\mathrm{K}0\mathrm{V}$	Gra2003	:		:		-4.980	Wri2004
217014	113357	G2V+	Gra2006	1.33	Val2005	0.15	Val2005	-5.080	Wri2004
217107	113421	G8IV-V	Gra2003	1.49	Val2005	0.27	Val2005	-5.080	Wri2004
217813	113829	G1V	Gra2003	1.15	Val2005	0.02	Val2005	-4.470	Wri2004
Continu	ted on Ney	xt Page							

TABLE A.1 – Continued

HD	HIP	Spec		Mass					
Name	Name	Type	Ref	(${ m M}_{\odot}$)	Ref	[Fe/H]	Ref	$\log(R'_{HK})$	Ref
218868	114456	G8V	Gra2003	0.93	Val2005	0.19	Val2005	-4.750	Wri2004
219134	114622	K3V	Gra2003	0.74	Val2005	0.09	Val2005	-5.089	Gra2003
219482	114948	F6V	Gra2006	1.07	Nor2004	-0.20	Nor2004	-4.434	Gra2006
219538	114886	K2V	Gra2003	0.66	Val2005	-0.05	Val2005	-4.840	Wri2004
219623	114924	F7V	HIP	1.07	Nor2004	-0.09	Nor2004	-4.84	MAs2009b
220140	115147	K2V	Gra2003		:	-0.64	Nor2004	-4.074	Gra2003
220182	115331	G9V	Gra2003	0.85	Nor2004	0.01	Nor2004	-4.370	Wri2004
220339	115445	K2.5V	Gra2003	0.65	Val2005	-0.24	Val2005	-4.896	Gra2003
221354	116085	$\mathrm{K}0\mathrm{V}$	Gra2003	0.75	Val2005	-0.01	Val2005	-5.149	Gra2003
221851	116416	K1V	Gra2003	0.80	Nor2004	-0.13	Nor2004	-4.735	Gra2003
222143	116613	G3V	Gra2003	0.98	Val2005	0.12	Val2005	-4.555	Gra2003
222237	116745	K3+V	Gra2006	0.88	Val2005	-0.20	Val2005	-4.959	Gra2006
222335	116763	G9.5V	Gra2006	0.81	Val2005	-0.16	Val2005	-4.909	Gra2006
222368	116771	F7V	HIP	1.42	Val2005	-0.08	Val2005	-4.76	MAs2009b
223778	117712	K3V	Gra2003	0.79	This work	-0.71	Nor2004	-4.518	Gra2003
224228	118008	K2.5V	Gra2006	:		:		-4.468	Gra2006
224465	118162	G4V	Gra2003	0.95	Nor2004	-0.01	Nor2004	-4.969	Gra2003
224930	000171	G5V	Gra2003	0.91	$\operatorname{Sod1999}$	-0.78	Nor2004	-4.880	Gra2003
232781	015673	K3.5V	Gra2003		:			-4.691	Gra2003
263175	032423	K3V	Gra2003			-0.59	Gra2003	-4.903	Gra2003
	036357	K2.5V	Gra2003		:	-0.61	Gra2003	-4.526	Gra2003
	040774	G_5	HIP					-4.38	MAs2009b
	087579	K2.5V	Gra2003					-4.529	Gra2003
:	091605	K2.5V	Gra2003	:		-0.68	Gra2003	-4.909	Gra2003

TABLE A.1 – Continued

NOTES.—Reference codes for columns 4 and 6 are as follows: Bod1999 = Boden et al. (1999); Clo2005 = Close et al. (2005); Gra2003 = Gray et al. (2003); Gra2006 = Gray et al. (2006); HIP = *Hipparcos*; Low 2002 = Low rance et al. (2002); MAs2009b = Mason et al. (2009b); Nor2004 = Nordström et al. (2004); Pou2000 = Pourbaix (2000); Pou2002 = (2002); Pou2002 = (20Pourbaix et al. (2002); Rag2009 = Raghavan et al. (2009); Sod1999 = Söderhjelm (1999); Tor2002 - Torres et al. (2002); Val2005 = Valenti & Fischer (2005); Wri2004 = Wright et al. (2004).

Comp	Alt	Spec		Mass	
Name	Name	\mathbf{Type}	\mathbf{Ref}	(${ m M}_{\odot}$)	\mathbf{Ref}
HD 000123 Ba				0.95	Gri1999
HD 000123 Bb				0.22	Gri1999
HD 001237 B	HD 1237B	$M4\pm1V$	Cha2007	0.13	Cha2006
HD 001273 Ab					
HD 003196 Ab				0.40^{a}	DM91
HD 003196 B		G2V	ΔV	1.00	DM91
HD 003443 B	$GJ \ 25 \ B$			0.70	Pou2000
HD 003651 B	HD 3651B	$T7.5\pm0.5$	Luh2007	0.05	Luh2007
HD 004391 B		M4V	VJHK		
HD 004391 C		M5V	VJHK		
HD 004614 B	LHS 122	K7V	VJHK		
HD 004628 Ab					
HD 004676 Ab				1.17	$\operatorname{Bod1999}$
HD 004747 Ab					
HD 006582 Ab	mu Cas B	M3V	ΔV		
HD 007693 A	HD 7788	F5V	Gra 2006	1.32	Nor2004
HD 007693 B	GJ 55.3B	K1V	Gra 2006		
HD 007693 D	GJ 55.1B			0.86	Sod1999
HD 008997 Ab		G7V	SB2q		
HD 009770 Ba				0.73	Cut1997
HD 009770 Bb $$				0.67	Cut1997
HD 009770 C		M3V	ΔV		
HD 009826 D	ups And B	M4.5V	Low 2002	0.2	Low2002
HD 010307 Ab				0.2	$\operatorname{Sod1999}$
HD 010360 A	HD 10361	K2V	Gra 2006		
HD 013445 B	GJ 86 B	WD	Mug2005a	0.5	Lag2006
HD 013974 Ab		K3V	SB2q		
HD 014214 Ab					
HD 014802 B		M2V	ΔV		
HD $016160 {\rm ~Ab}$					
HD 016160 B	NLTT 8455	M3.5V	Hen 2002		
HD 016287 Ab					
HD 016739 Ab				1.24	Bag2006
HD 016765 B		K2V	ΔV		
HD 016895 B	NLTT 8787	M2V	BRJHK		

TABLE A.2: Spectral type and mass of the companions

TABLE A.2 – Continued

Comp	Alt	Spec	Dí	Mass	
Iname	Iname	Type	Rei	(M_{\odot})	Rei
HD 017382 Ab					
HD 017382 B $$	NLTT 8996	M7V	VRJHK		
HD 018143 B	HD 18143 B	K9V	ΔV		
HD 018143 C	NLTT 9303	M7V	VRJHK		
HD 018757 C	NLTT 9726	M2V	VIJHK		
HD 019994 B		M2V	ΔV		
HD 020010 Bb $$					
HD 020010 B	GJ 127 B	K2V	ΔV		
HD 020807 B $$	HD 020766	G2V	Gra2006	1.12	Val2005
HD 021175 B		M3V	ΔV		
HD 023484 Ab					
HD 024409 D					
HD 024409 E		M2V	VJHK		
HD 024496 B		M2V	ΔV		
HD 025680 Ab					
HD 025998 A	HD 25893	G9V	Gra 2003		
HD 025998 B		K2V	ΔV		
HD 025998 Eb					
HD 026491 Ab					
HD 026923 B	HD 26913	G6V	Gra2003	0.87	Nor2004
HD 026965 B	LHS 25	M4.5	Rei2004		
HD 026965 C	HD 26976	DA3	Hol2002		
HD 032778 B	NLTT 14447	M0V	VRIJHK		
HD 032850 Ab					
HD 032923 B		G1V	ΔV		
HD 035112 B		M1V	ΔV		
HD 035296 C	HD 35171	m K7V	VJHK		
HD 036705 Ab		M8V	Boc2008	0.09	Clo2007
HD 036705 Ba		M5V	Jan2007	0.16	Jan2007
HD 036705 Bb $$		$\mathrm{M4V} \pm 1$	Jan2007	0.14	Jan2007
HD 037394 B	HD 233153	M1V	VJHK		
HD 037572 B	HIP 26369	K5V	Gra2006		
HD 039587 Ab				0.15	Kon2002
HD 039855 B	BD-19 1297B	K9V	VIJHK		
HD 040397 B		M4V	ΔV		
HD 040397 D	NLTT 15867	M5V	RJHK		
HD 043162 B		M4V	VRIJHK		

TABLE A.2 – Continued

Comp Name	Alt Name	Spec Type	Ref	${ m Mass}$	Ref
UD 042507 Ab		01		0.54	Cat2006
ПD 043387 AD	 NI TT 16999	 M 4 V	 VDIIUK	0.04	Cat2000
HD 043337 L	NLI I 10555	$M5V \perp 1.5$	$V MIJII \Lambda$	 0.1 <i>4</i>	 Faa2007
HD 045054 AD	• • •	$MOV \perp 1.0$	Egg2007	$0.14 \\ 0.71$	This work
HD 045088 RD	• • •	 M4M	ΔV	0.71	THIS WOLK
HD 045000 D HD 045270 B		MOV	ΔV ΔV	•••	
HD 049210 D HD 048189 B		K7V	ΔV ΔV	•••	
HD 052698 Ab		IX / V		•••	
HD 053705 B	HD 053706	K0 5V	Gra2006	0.78	 Val2005
HD 053705 Ca	HD 053680	K6V	Gra2006	0.10	Va12000
HD 053705 Cb	112 000000	110 1			
HD 054371 Ab					
HD 057095 B		K6V	ΔV		
HD 061606 B	NLTT 18260	K9V	VRIJHK		
HD 063077 Ab					
HD 063077 B	NLTT 18414	DC	Kun1984	0.56	Hol2008
HD 064096 B				0.9	Pou2000
HD 064468 Ab					
HD 064606 Ab					
HD 065430 Ab					
HD 065907 B	LHS 1960	M0V	IJHK		
HD 065907 C		M5V	ΔV		
HD 067199 Ab					
HD 068017 Ab					
HD 068257 B	$HD \ 068255$	F8V	Gra 2003	1.52	Nor2004
HD 068257 Ca	$HD \ 068256$	G0 IV-V	Gra2003		
HD 068257 Cb1		M2V			
HD $068257 \text{ Cb}2$		M2V			
HD 068257 Cb3		M2V-M4V			
HD 072760 Ab				0.13	Met2008
HD 073752 B	• • •	K2V	ΔV	1.07	$\operatorname{Sod1999}$
HD 074385 B	NLTT 20102	M2V	VRIJHK	•••	
HD 075732 B	LHS 2063	M6V	VRIJHK		
HD 075767 Ab	• • •	••••			
HD 075767 Ba		M3V	Fuh2005		
HD 075767 Bb		M4V	Fuh2005		
HD 079028 Ab					

TABLE A.2 – Continued

Comp	Alt	Spec	5.4	Mass	5.4
Name	Name	Type	Ref	(M_{\odot})	Ref
HD 079096 Ab				0.85	Pou2000
HD 079096 Ea	Gl 337C	L8	Wil2001		
HD 079096 Eb		L8	Bur2005		
HD 079969 B		K4V	ΔV	0.74	$\operatorname{Sod1999}$
HD 080715 Ab		K3V	SB2q		
HD 082342 B		M3.5V	Haw1996		
HD 082443 B	NLTT 22015	M5.5V	Haw1996		
HD 082885 B		M8V	ΔV		
HD 086728 Ba	GJ 376 B	M6.5V	Giz2000		
HD 086728 Bb $$		M6.5V	Giz2000		
HD 089125 B	GJ 387 B	M1V	VJHK		
HD 090508 B $$	LHS 2266	M2V	ΔV		
HD 090839 Ba	HD 237903	K5	RJHK		
HD 090839 Bb	•••				
HD 096064 B	NLTT 26194	$\mathbf{K7}$	ΔV		
HD 096064 C	BD-033040C	$\mathbf{K7}$	ΔV		
HD 097334 Ea	Gl 417B	L4.5V	Kir2001	0.1	Bou2003
HD 097334 Eb		L4.5V	Bou2003	0.1	Bou2003
HD 098230 Ab		M3V	Bal2005		
HD 098230 A	HD 98231	F9V	ten 2000		
HD 098230 Bb $$		K7V	ΔV		
HD 099491 B	HD 099492	K2V	Hipparcos	1.24	Val2005
HD 100180 Ab					
HD 100180 B	NLTT 27656	K5V	RIJHK		
HD 100623 B	LHS 309	DC	Hen 2002		
HD 101177 Ba	LHS 2436	K3V	VRJHK		
HD 101177 Bb		M2V	SB2q		
HD 101206 Ab					
HD 102365 B	LHS 313	M4V	Haw1996		
HD 102438 Ab					
HD 110463 Ab					
HD 110833 Ab					
HD 111312 Ab		M2V	ΔV	0.58	sb2q
HD 111312 B	• • •	K8V	SB2q		
HD 112758 Ab					
HD 112758 B	• • •	M2V	ΔV		
HD 112914 Ab		•••			

TABLE A.2 – Continued

Comp	Alt	Spec		Mass	
Name	Name	Type	Ref	(M_{\odot})	Ref
HD 113283 Ab					
HD 113449 Ab					
HD 115404 B	LHS 2714	M0.5V	Haw1996		
HD 116442 B	HD 116443	K2V	Gra2003	0.78	Val2005
HD 120136 B	HD 120136B	M2V	ΔV		
HD 120690 Ab					
HD 120780 Ab					
HD 120780 B		M4V	ΔV		
HD 121370 Ab					
HD 122742 Ab					
HD 124850 Ab					
HD 125276 Ab					
HD 125455 B	LHS 2895	M6	VJHK		
HD 128620 B	HD 128621	K2IV	Gra2006	0.93	Pou2002
HD 128620 C	HIP 070890			0.11	Wer2006
HD 128642 Ab					
HD 130042 B		K8V	ΔV		
HD 130948 B	HD 130948 B	$L2 \pm 2$	Pot2002	0.07	Pot2002
HD 130948 C	HD 130948 C	$L2 \pm 2$	Pot2002	0.07	Pot2002
HD 131156 B	HD 131156B			0.67	$\operatorname{Sod1999}$
HD 131511 Ab					
HD 131582 Ab					
HD 131923 Ab					
HD 133640 Ba	NLTT 39210	K2V	Lu2001		
HD 133640 Bb		M2V	SB2q		
HD 135204 B		G9V	ΔV		
HD 136202 B	LHS 3060	K9V	VJHK		
HD 137107 B	HD 137108			1.05	Pou2000
HD 137107 E	GJ 584 C	L8V	Kir2001	0.06	Kir2001
HD 137763 Ab	K9	SB2q		0.57	SB2
HD 137763 B	HD 137778	K2V	Gra 2003	0.87	Val2005
HD 137763 C	GJ 586C	M4.5V	Rei1995		
HD 139341 B		K1V	ΔV	0.83	$\operatorname{Sod1999}$
HD 139341 C	HD 139323	K2IV-V	Gra 2003	1.13	Val2005
HD 139777 B	HD 139813	K2V	RJHK	1.15	Val2005
HD 140538 B		M5V	ΔV		
HD 140901 B	NLTT 41169	M2	ΔV		

TABLE A.2 – Continued

Comp	Alt	Spec	Def	Mass	Def
Iname	Iname	Type	Rei	(M_{\odot})	Ref
HD 141272 B		$\mathrm{M3V}\pm0.5$	Eis2007	0.26	Eis2007
HD 142267 Ab					
HD 143761 Ab				0.14	Gat2001
HD 144284 Ab				0.46	Maz2002
HD 144287 Ab		K4V	ΔV		
HD 144579 B	LHS 3150	M4V	VRIJHK		
HD 145825 Ab					
HD 145958 B	NLTT 42272	G9V	Gra2003		
HD 145958 D		T6	Loo 2007		
HD 146361 Ab				1.09	Rag2009
HD 146361 B	HD 146362			1.0	Rag2009
HD 146361 Ea	HIP 079551	M2.5V	Rei1995		
HD 146361 Eb	sig CrB D			0.1	Hei1990
HD 147513 B		DA2	Hol2002		
HD 147584 Ab					
HD 147776 C					
HD 147776 D					
HD 148653 B	LHS 3204			0.77	$\operatorname{Sod1999}$
HD 148704 Ab		K1V	SB2q		
HD 149806 B		M6V	RJHK		
HD 153557 B		M2V	ΔV		
HD 153557 C	HD 153525			0.73	Nor2004
HD 155885 B		K1.5V	ΔV		
HD 155885 C	HD 156026	K5V	Gra2006		
HD 156274 Ab					
HD 156274 B	NLTT 44525	m K7V	Haw1996		
HD 157347 B	HR 6465	M3V	This work		
HD 158614 B				0.90	Pou2000
HD 160269 B				0.65	$\operatorname{Sod1999}$
HD 160269 C	HIP 86087	M0.5V	Rei2004		
HD 160346 Ab					
HD 161198 Ab					
HD 161797 Ab		M8V	ΔV		
HD 161797 B	NLTT 45430	M3V	ΔV		
HD 161797 C		M3V	ΔV		
HD 162004 A	HD 162003	F5V	RJHK	1.38	Nor2004
HD 165341 B	NLTT 45900		•••	0.78	Pou2000

TABLE A.2 – Continued

Comp Name	Alt Name	Spec Type	Ref	Mass	Ref
		- <i>J</i> P ^o	1001	(
HD 165401 Ab	• • •			• • •	•••
HD 165499 Ab			•••		
HD 165908 Ab		···· VCV	· · · ·		
HD 165908 B		K6V MOV	ΔV		
HD 167425 B	• • •	MOV	VJHK		•••
HD 175742 Ab			· · ·		 G 11000
HD 176051 B	• • •	K3V Doly	ΔV	0.71	Sod1999
HD 177474 B	• • •	F8V	ΔV	• • •	•••
HD 178428 Ab				•••	•••
HD 179957 B	HD 179958	G4V	ΔV	•••	• • •
HD 181321 Ab					···
HD 184467 B				0.8	Pou2000
HD 185414 Ab					
HD 186408 Ab		MOV	ΔV	•••	
HD 186408 B	HD 186427	G3V	Gra2006	1.10	Val2005
HD 186858 A	HD 187013	F5.5IV-V	Gra 2003	1.24	Nor2004
HD 186858 B		K3V	ΔV	0.68	$\operatorname{Sod1999}$
HD 186858 G	HD 225732	K4V	ΔV		
HD 187691 C		M3V	VRIJHK		
HD 189340 B				0.94	Sod1999
HD 190067 B		K7V	ΔV		
HD 190360 B	LHS 3509	M4.5V	Rei2004		
HD 190406 Ab	HD 354613	$L4.5\pm1.5$	Liu2002	0.06	Liu2002
HD 190771 Ab					
HD 191408 B	LHS 487	M5V	ΔV		
HD 191499 B	ADS 13434B	K5V	ΔV		
HD 191785 E		M3.5V	This work		
HD 193664 Ab					
HD 195564 B	LTT 8128	M2V	ΔV		
HD 195987 Ab				0.67	Tor2002
HD 197076 C	NLTT 49681	M2.5V	Rei2004		
HD 198425 Ab					
HD 198425 B	NLTT 49961	M6V	VJHK		
HD 200525 B		GOV	ΔV		
HD 200525 C	NLTT 50542	M3V	$\frac{-}{\Delta V}$		
HD 200560 D	GJ 816 1B	M3V	$\frac{-}{\Delta V}$		
HD 200968 B	GJ 819B	K7V	H_{aw} 1996		

TABLE A.2 – Continued

Comp	Alt Namo	Spec Type	Dof	Mass	Dof
Inallie	Ivallie	Type	ner	(IVI ₀)	nei
HD 202275 B $$				1.12	Pou2000
HD 202940 Ab					
HD 202940 B	LHS 3656	K9V	ΔV		
HD 203985 Ab					
HD 203985 B	LTT 8515	M3.5V	This work		
HD 206826 B	HD 206827	G5V	ΔV		
HD 206860 B	HN Peg B	$T2.5\pm0.5$	Luh2007	0.02	Luh2007
HD 211415 B		K9V	ΔV		
HD 211472 T	GJ 4269	M4V	VJHK		
HD 212168 B	HIP 110719	G0V	Gra2006	1.13	Val2005
HD 212330 Ab					
HD 214953 B	NLTT 54607	M0.5V	Haw1996		
HD 215648 B		M4V	ΔV		
HD 217107 B					
HD 218868 B		M5V	VJHK		
HD 220140 B	NLTT 56532	M3V	VRIJHK		
HD 220140 C		M7V	VRIJHK	0.1	Mak2007
HD 223778 Ab				0.75	This work
HD 223778 B		M6V	ΔV		
HD 224465 Ab					
HD 224930 B		$K5\pm1 V$	ten 2000	0.58	$\operatorname{Sod1999}$
HD 263175 B	HD 263175B	M0.5V	Rei2004		
HIP 036357 A	HD 58946	F1V	Gra 2003	1.40	Nor2004
HIP 036357 B	GJ 274B	M7V	ΔV		
HIP 091605 B	LHS 3402	M2V	VJHK		

NOTES.—Reference codes for columns 4 and 6 are as follows: Bag2006 = Bagnuolo et al. (2006); Bal2005 = Ball et al. (2005); Boc2008 = Boccaletti et al. (2008); Bod1999 = Boden et al. (1999); Bou2003 = Bouy et al. (2003); Bur2005 = Burgasser et al. (2005); Cat2006 = Catala et al. (2006); Cha2006 = Chauvin et al. (2006); Cha2007 = Chauvin et al. (2007); Cut1997 = Cutispoto et al. (1997); Egg2007 = Eggenberger et al. (2007); Eis2007 = Eisenbeiss et al. (2007); Fuh2005 = Fuhrmann et al. (2005); Gat2001 = Gatewood et al.

(2001); Giz2000 = Gizis et al. (2000); Gra2003 = Gray et al. (2003); Gra2006 = Gray et al. (2006); Gri1999 = Griffin (1999); Haw1996 = Hawley et al. (1996); Hei1990 = Heintz (1990); Hen2002 = Henry et al. (2002); Hol2002 = Holberg et al. (2002); Hol2008 = Holberg et al.(2008); Jan2007 = Janson et al. (2007); Kir2001 = Kirkpatrick et al. (2001); Kon2002 = König et al. (2002); Kun1984 = Kunkel et al. (1984); Lag2006 = Lagrange et al. (2006); $Liu_{2002} = Liu$ et al. (2002); $Loo_{2007} = Looper$ et al. (2007); $Low_{2002} = Low_{rance}$ et al. (2002); Lu2001 = Lu et al. (2001); Luh2007 = Luhman et al. (2007); Mak2007 = Makarov et al. (2007); Maz2002 = Mazeh et al. (2002); Met2008 = Metchev & Hillenbrand (2008); Mug2005a = Mugrauer & Neuhäuser (2005); Nor2004 = Nordström et al. (2004); Pot2002= Potter et al. (2002); Pou2000 = Pourbaix (2000); Rag2009 = Raghavan et al. (2009); Rei1995 = Reid et al. (1995); Rei2004 = Reid et al. (2004); Sod1999 = Söderhjelm (1999);ten2000 = ten Brummelaar et al. (2000); Tor2002 = Torres et al. (2002); Val2005 = Valenti &Fischer (2005); Wer2006 = Wertheimer & Laughlin (2006); Wil2001 = Wilson et al. (2001); $\Delta V = \text{estimated using primary's spectral type and measured } \Delta V; \text{SB2q} = \text{estimated using}$ primary's mass or spectral type and mass-ratio from SB2 solution; VRIJHK etc. colors = estimated from colors.

^a Estimated using mass-sum of Aa+Ab from DM91 and mass of Aa from Nor2004